AGARDograph 109 AGARDograph 109 ADVISORY GROUP FOR AEROSPACE RESEARCH & DEVELOPMENT, AD 657092 64 RUELDE VARENNE PARIS 75 FRANCE **Subsonic Wind Tunnel** Wall Corrections · 01 by H. C. Garner, E. W. E. Rogers 0013 W. E. A. Acum and E. C. Maskell  $\star$ AUE: 3 1967 fr **OCTOBER 1966** NORTH ATLANTIC TREATY ORGANIZATION

5.

AGARDograph 109

語語でしょうです

# NORTH ATLANTIC TREATY ORGANIZATICN ADVISORY GROUP FOR AEROSPACE RESEARCH AND DEVELOPMENT (ORGANISATION DU TRAITE DE L'ATLANTIQUE NORD)

and heathershipsen it was so and a second

- Marine

k

1.4.4.4

י האורים אוריאראלו לאלאלאלא לא אוריים אוריים אוריים אוריים איני איני אוריים אוריים אוריים אוריים אוריים אוריים

## SUBSONIC WIND TUNNEL WALL CORRECTIONS

by

H.C. Garner, E.W.E.Rogers, W.E.A. Acum and E.C. Naskell

Editor

H.C.Garner Aerodynamics Division. National Physical Laboratory, Teddington, Middlesex, England.

October 1966



and a start

This AGARDograph is one of a series of publications sponsored by the NATO-AGARD Fluid Dynamics Panel under the editorship of Professor Wilbur C.Nelson

٤.

.,

#### SUMMARY

is the states

The subject of wall interference spans nearly half a century and remains a field of active research. The classical theories of lift and blockage interference in two-dimensional and three-dimensional tunnels are now highly developed for steady subsonic flows without separation in fully closed or open tunnels. Oscillatory experiments, cases of separated flow, and tunnels with slotted or perforated walls can be treated by more recent analysis, but results are relatively limited.

The theoretical background is reviewed and modern developments in the formulation, calculation and application of interference corrections are discussed in seven chapters. Chapter I gives a general review and lists areas in which further research is needed. Each of the more specialized chapters contains selected graphs of numerical data and a summary or table of principal formulae.

#### RESUME

L'étude de la question de l'interférence due à la paroi remonte à presque un demi-siècle et continue à faire l'objet de recherches actives. Les théories classiques concernant l'interférence due à la portance ou au blocage dans les souffleries bi- et tridimensionnelles ont mainténant été élaborées à un haut degré pour les écoulements subsoniques stationnaires sans séparation dans des souffleries à circuit entièrement fermé ou ouvert. Les expériments oscillatoires. les écoulements séparés et les souffleries à parois à fentes ou perforées peuvent être traités par des méthodes d'analyse plus réceutes, mais les résultats ainsi fourais ne sont que relativement limités.

Les sept chapitres constituant la présente Agardographie passeut en revue l'historique des théories à ce sujet et examinent des développements modernes dans la formulation, calcul et application des corrections de parois. Le premier chapitre donne un aperçu general et indique les domaines dans lesquels de nouvelles recherches sont nécessaires. Chacun des chapitres plus spécialisés comporte des graphiques choisis de données numériques, ainsi qu'un résamé ou tableau récepitulatif des principales formules utilisées.

533.6.071.4

## CONTENTS

いた。中

,i

ż

1. 4. 400

.

		rage
SUMMARY		115
RESURE		<b>{</b> 11
CHAPTER I	STATUS OF INTERFERENCE EFFECTS	
	by H.C.Garter	1
1.1	Introduction	5
1.2	General References	6
1.3	Prezent Knowledge	8
1.4	Outstanding Problems for Research	14
1. 5	Acknowledgements	16
CEAPTER II	LIFT INTERFERENCE ON TWO-DIMENSIONAL WINGS	
	by B.C. Garner	21
2.1	Introduction	29
2.2	Wings of Zero Thickness in Closed Tunnels	31
2.3	Wings of Finite Thickness in Closed Tunnels	37
2.4	Aerofoils with Hiaged Flaps	44
2. 5	Open-Jet Tunnels	46
2.6	Experimental Considerations	53
CHAPTER III	LIFT INTERFERENCE ON THREE-DIMENSIONAL WINGS	
	by H.C.Garner	73
3. 1	Introduction	87
3.2	Small Wings in Closed and Open Tunnels	\$0
3.3	General Theory of Lift Interference	103
3.4	Evaluation of Interference Parameters	117
3.5	Numerical Interference Corrections	141
3.6	Special Configurations	153
3.7	Experimental Considerations	167
3.8	Index of Formulae and Date	173
CHAPTER IV	INTERFERENCE EFFECTS IN UNSTEADY EXPERIMENTS	
	by W.E.A. Acus	215
4.1	Introduction	223
4.2	Two-Dimensional Wings in Incompressible flow	232
4.3	Two-Dimensional Wings in Compressible Flow	Z <b>36</b>
<b>4.4</b>	General Theory for Three-Dimensional Wings	239
4.5	Application to Three-Dimensional Wings in Closed	
	Tunnels	247
4.6	Perforated and Slotted Walls	236
4.7	Steadily Rotating Models	258
4.8	Applications to Common Experimental Situations	254

i

ŧ

ŝ

ł

2.0

12

· · · · · ·

~

p	•	•	•

And the second second

Charles and a second second

}..

CHAPTER V	BLOCKAGE EFFECTS IN CLOSED OR OPEN TUNNELS	
	by E.W.E.Rogers	279
5.	1 Introduction	287
5.	2 Solid Blockage Factors in Closed Rectangular Tunnels	289
5.	3 Solid Blockage Factors in Closed Non-Rectangular	
	Turnels	299
5.	4 Wake Blockage Factors in Closed Tunnels	304
5.	5 Blockage Factors in Open-Jet Tunnels	308
5.	S Estimation of Blockage from Wall Measurements	311
5.	7 Blockage Corrections at Very High Subsonic Speeds in	
_	Closed Tunnels	315
5.	8 Use of Blockage Factors in Correcting Measured	~ ~ ~
_	Quantities	317
5.	9 SLABARY OF Principal Blockage Formulae	321
CHAPTER V	1 WALL INTERFERENCE IN TURNELS WITH VENTILATED WALLS	241
	Dy E. W. E. Kogeis	741
8	1 Introduction	351
6.	2 Roundary Conditions for Walls with Longitudinal Slots	353
6.	3 Tro-Dirensional Blockage in Tunnels with Longitudinal	
•••	Slots	357
6.	4 Three-Dimensional Blockage in Tunnels with Longitudinal	
	Slots	362
6.	5 Lift Interference in Tunnels with Longitudinal Slots	369
6.	6 Some General Remarks on Interference Effects in	
	Tunnels with Longitudinal Slots	379
6.	7 The Perforated Wall	380
5.	8 Two-Dimensional Blockage in Tunnels with Perforated	
	Walls	383
6.	9 Three-Dimensional Blockage in Tunnels with Perforated	
	Walls	385
6.	10 Lift Interference in Tunnels with Perforated Walls	385
<b>6</b> .	11 Application of Correction Factors	389
CRAPTER 1	II BLUPF BODIES AND HIGH-LIFT SYSTEMS	425
	by E.C. Haskeli	731
7.	1 Introduction	437
7.	2 Tlockage Effects on Bluff Bodies	435
7.	3 Lifting Wings with Separated Flow	441
7.	4 V/STOL Configuration3	444

Į

## DISTRIBUTION

PORTES I

----

. .

ſ

Contraction of the second seco

V

CHAPTER I

## STATUS OF INTERFERENCE EFFECTS

by

H.C.Garner

Law of the second s

i.

Aerodynamics Division, National Physical Laboratory, Teddington, Middlesex, England

1

1.50

è

i

## CONTENTS

and the start of the

nuomene erandukten naresoret, attittukkinnikin mennikitan 1922 bishir Richardshiret II. UN Bishir Kushiran and

:

at the state of the

Ż

an and speed and

- Aller

		Page
1.1	INTRODUCTION	5
1.2	GENERAL REFERENCES	6
1.3	PRESENT KNOWLEDGE	8
	1.3.1 Lift Interference	8
	1.3.2 Blockage Interference	10
	1.3.3 Ventilated Tunnels	11
	1.3.4 Unsteady Interference	13
	1.3.5 High-Lift Systems	14
1.4	OUTSTANDING PROBLEMS FOR RESEARCH	14
1.5	ACKNOWLEDGENENTS	16
REFE	RENCES	16

#### STATUS OF INTERFERENCE EFFECTS

+ H.C.Garner

#### 1.1 INTRODUCTION

By its very nature the most perfect subsonic wind tunnel cannot reproduce an unconstrained flow past a model. The problem of wall interference has been of lasting 'concern to experimenters and theoreticians while tunnel design, model shapes and experimental techniques have been developing through the years. The decision to prepare a monograph on subsonic wind-tunnel wall corrections was made during the planning of an AGARD meeting on interference effects in aerodynamic test facilities, held in Brussels in March 1959. Goethert's<sup>1, 1</sup> monograph on transonic wind-tunnel testing, then in course of preparation, concentrates more especially on flows at high subsonic and low supersonic Mach numbers. As regards wall interference, this and the present AGARDograph are largely complementary. The slight overlap on the subject of ventilated tunnels is not inappropriate as there have been recent developments in this important field. A review of progress and current problems was presented by Regers<sup>1, 2</sup> in the introductory paper to the AGARD meeting. Further advances and new problems have since emerged, and such facts confirm the continuing importance of a study that has already occupied nearly half a century.

The foundation of research on tunnel-wall interference is attributed to Prandtl (Ref. 1.3; 1919), because his lifting-line theory led to many experimental investigations with the object of verifying the theory. Moreover, the basic principles of the liftingline theory are essential to an understanding of the simplest calculations of wall interference on finite lifting wings. The method of analysis for closed and open tunnels is established in Reference 1.3, where Prandtl develops the concept of trailing vortices so that the problem of wall interference at a lifting line involves two dimensions only. Theoretical and empirical studies then followed in quick succession, and after some ten years of research the elements of wall interference had been built into a practical framework so as to influence model testing and the design of wind tunnels. A comprehensive account of these early developments is given in Glauert's classic monograph (Ref. 1.4; 1933).

Interference effects in subsonic wind tunnels may arise from the influence of tunnelwall boundary layers, disturbances from measuring gea: and model supports within the airstream, and irregularities of the airstream itself due to non-uniformity, unsteadiness or small-scale turbulence. In different ways these can all be important, but they are mainly outside the scope of the present monograph. The nature of tunnel-wall constraint can be deduced from physical principles of streamline flow. It is also associated directly with the theoretical consideration that, although the differential equations of the flow are the same in the tunnel as in free air, the outer boundary conditions are different. A logical and precise evaluation of the interference flow near the

model by linearized theory would involve firstly a solution for the velocity potential in the presence of the tunnel boundaries, and secondly a solution for the velocity field near the model when all the boundaries, including that of the model, are removed and the velocity potential at the surface of the model from the first solution is preserved. The resulting perturbations of velocity-potential gradient normal to the surface of the model represent the interference-flow field which may be regarded as a correction to the free stream. In practice this correction has significant vertical and streamwise components. If the model is small enough, these can be interpreted as respective corrections to the direction and speed of the stream: if the model has enpreciable span, average values may suffice: if the model has appreciable length, the respective components involve a streamline curvature and a longitudinal pressure gradient with consequent corrections to pitching moment and drag. The correction to stream direction and the streamline curvature are known as lift interference, since they are usually associated with circulation or vorticity round the model; in most applications they are considered to be independent of the changes in longitudinal velocity, known as blockage interference and usually arising from the volume occupied by the model and its wake. Although the basic problem is to determine the interferenceflc= field, this meeds to be interpreted in the form of corrections to nearly all the measured aerodynamic quantities.

Prom Reference 1.4 it is clear that many of the early developments were associated with lift interference, and that the concept of streamline curvature. though implicit in tailplane interference, first arose as the essence of two-dimensional interference. Glauert's monograph is of more than historical importance; it continues to provide a comparative back-ground for most of the subsequent developments. Above all it gives a full and lucid account of the many principles underlying lift and blockage interference. Some more recent works of reference are briefly Feviewed in Section 1.2 from the standpoint of tunnel-wall interference in subsonic flow. Section 1.3 discusses the contents of Chapters II to VII of the AGARDograph, and conclusions regarding important fields for further research are presented in Section 1.4.

#### 1.2 GENERAL REFERENCES

References 1.1 and 1.5 to 1.12 comprise a representative collection of treatises concerning, among other important topics, the theory and application of subsonic wind-tunnel wall corrections. These are considered below in chronological order.

The contribution in Reference 1.5 by von Kármán and Burgers followed shortly after Glauert's<sup>1.\*</sup> monograph. They consider only lift interference and give the analysis for two-dimensional flows with closed boundaries and for a lifting line in closed and open rectangular and circular tunnels. By far the most important contribution is the general solution, due to Burgers, for the three-dimensional flow field of a lifting element in a closed or open circular tunnel. Under the editorship of Durand miss, there is the chapter in Reference 1.6 by Toussaint. This includes a discussion of blockage interference fuller than that of Reference 1.4 without improving its practical content. The treatment of two-dimensional lift interference in Reference 1.6 is unreliable, but the uniformly loaded lifting line in rectangular tunnels is considered in more detail than in Reference 1.5. In particular, the correct expression is given

6

for the case of solid side-walls and open floor and roof, for which Glauert quoted an erroneous result<sup>\*</sup>.

In an AVA monograph Riegels<sup>1, 7</sup> gives a most comprehensive collection of graphs to illustrate many aspects of lift interference in incompressible flow. These include off-centre models, streamline curvature along the tunnel axis, and types of mixed boundary to achieve zero interference. With brief discussion of some sixty references of later date than Rcferences 1.4 to 1.6, the monograph represents a major advance, especially with regard to circular and elliptical tunnels with closed, open or mixed boundaries. Solid and wake blockage are also considered in some detail. Of the same period there is a short informative article by Katzoff<sup>1, 8</sup> that deserves to be read by all wind-tunnel users. These two references highlight the impact of research on wall interference in the dozen years following the preparation of References 1.4 to 1.6.

Two valuable works on wind-tunnel practice by Pankhurst and Holder<sup>1.9</sup> and Pope<sup>1.10</sup> appeared early in the following decade, both of which give full accounts of wall interference corrections. Three-dimensional lift interference is regarded primarily as a function of tunnel shape and wing span, spart from the treatment of tailplane interference. Blockage interference effects, including the buoyancy correction to drag due to longitudinal pressure gradient, are well discussed. In many configurations of the present day, however, the influence of streamline curvature on lifting wings is usually much more important than the buoyancy correction, and in this respect References 1.9 and 1.10 have both been overtaken by developments. By way of contrast, Pankhurst and Holder<sup>1.9</sup> give more information on octagonal tunnels and alternative methods of applying the two-dimensional lift interference corrections, with particular attention to the neighbourhood of the stall. Pope<sup>1.10</sup>, however, gives a fuller account of corrections to downwash in the wake of a lifting wing and includes some numerical data on streamline curvature; moreover, he gives more discussion of reflection-plane models and the case of asymmetrical spanwise loading. Pope gives considerably more graphical data, but perhaps concentrates too much on lift interference for uniform spenwise loading in conjunction with the equivalent span of the rolled-up trailing vortex; however, elliptic spanwise loading is by no means neglected. Fankhurst and Holder include scme particularly useful tables and reinforce their more limited graphical data with tabulated values; by contrast, Pope includes no tables but does give illustrative worked examples of calculated interference corrections. Both works serve the wide needs of the windtunnel operator most admirably and cannot be judged from the standpoint of wall interference alone.

It is unfortunate that ventilated tunnels were subject to security classification when References 1.9 and 1.19 were prepared; their most serious limitation arises from the intensive development and wide use of ventilated tunnels for transonic testing. Goethert's<sup>1.1</sup> monograph deals comprehensively with this subject for the first time. Subsonic wall interference is reviewed in some detail. There is a good physical discussion of solid blockage in open and closed tunnels preparatory to a chapter on tunnels with longitudinal slots. The replacement of the singular boundary conditions of discrete slots by a homogeneous one is discussed, and there follows the general conclusion that with inviscid flow the open area ratic of slotted tunnels for zero interference decreases exponentially as the number of slots increases. The homogeneous boundary condition is generalized to account for slot depth and viscous effects such as friction or local separation inside the slots. Goethert argues that the slots are

• This, the only error the author has detected in Reference 1.4, is discussed in Chester III (Section 3.2.2).

in the second second

treated as in slender-wing theory, so that the boundary condition is essentially independent of Mach number, although the viscous effects may be altered by compressibility. The more detailed theoretical treatment of lift and blockage interference is confined to two-dimensional and circular slotted tunnels. There is a physical discussion of the flow near perforated walls and of the existence of an isage system when the model and the tunnel are two-dimensional; again the theoretical results are limited to twodimensional and circular tunnels. Finally Grethert points out a basic difference between perforated tunnels and ideal slotted tunnels without viscous effects. The boundary condition for a perforated turnel (and indeed for a slotted tunnel with viscous effects) behaves like that for a closed tunnel in the distant wake of a lifting wing. Hence the Trefftz condition that the downwash at large distances behind the wing is twice as large as the downwash at the plane of the wing does not hold true. The importance of ventilated tunnels is emphasised in the account of subscale wall interference by Allen and Spiegel<sup>1,11</sup>. Closed and open houndaries are regarded virtually as special cases of slotted or perforated walls. The theoretical lift and blockage interference in tunnels of circular section are used to illustrate the fundamental characteristics of the two types of ventilated tunnel and the manner in which they are intermediste between closed and open tunnels and can achieve zero interference.

The previous references give very little attention to wind-tunnel interference in unsteady experiments. Molyneux<sup>1. 12</sup> has recently written a useful review of the problems. For dynamic investigations there is a more extensive range of interference effects which include resonances associated with transverse acoustic waves, the tunneldrive system and induced flows external to an open jet. There is also the possibility of random disturbances in a ventilated working section from turbulent mixing at the downstream end of slotted or perforated walls. Reference 1.12 gives a brief discussion of other aspects of wall constraint on oscillating models and provides a practical introduction to the subject.

#### **1.3 PRESENT KNOWLEDGE**

From many aspects of wall interference there have been considerable advances since the preparation of the important works of reference discussed in Section 1.2. Chapters II to VII of this AGARDograph incorporate such advances and also include original contributions and unpublished numerical data. Some of these recent developments are outlined below in relation to earlier achievements.

#### 1.3.1 Lift Interference

For two-dimensional tests in closed tunnels the most well-known corrections apply to small, thin wings and neglect terms in aerofoil thickness to chord ratio t/c and in  $(c/b)^{4}$  where h is the tunnel height. But Goldstein's<sup>1,13</sup> analysis includes interactions of thickness and incidence in wall interference, such as the effect of thickness on lift interference and that of incidence on solid blockage. In Chapter II his theory is presented in a simplified form for symmetrical metofoils, so that the corrections include terms in  $(c/h)^{2}$  and  $(c/h)^{4}$  with coefficients linearly dependent on t/c. The formulae are easily generalized to incorporate first-order effects of compressibility and some allowance for viscous effects. The complete corrections are applied to the systematic experiments of Knechtel<sup>1, 14</sup> on the NACA 4412 profile with encouraging success.

There is a tendency in theoretical work to express wall interference in terms of correction factors to the measured forces. In practice the interference on a lifting aerofull is interpreted more appropriately as incremental corrections to the measured quantities, including the incidence. This is particularly true of aerofulls with hinged flaps. The theory of Preston and Manwell<sup>1, 13</sup> is developed to give the necessary corrections to incidence, lift, pitching moment and hinge moment. A later theory by de Jager and van de Vooren<sup>1, 16</sup>, also discussed in Chapter II, considers the non-linear characteristics of flaps at large angles of deflection for which the interference corrections are shown to change sign.

It seems profitless to treat two-dimensional testing in open tunnels in great detail. Bither the flow is far from two-dimensional, or else with the model between solid side-walls the interference corrections are large, of order c/h, and of dubious accuracy so as to place a very restricted limit on model size. However, there is considerable theoretical interest in the configuration of an aerofoil spanning a rectangular tunnel with solid side-walls, when the working section has open floor and roof between a closed entrance mozzle and a collector downstream. The usual boundary condition imposes the same constant pressure at the two free boundaries. Katzoff<sup>1,17</sup> introduces differential pressures to prevent spillage at the collector, and under such conditions the wall interference of order c/b and the streamline curvature can both be nearly eliminated.

•

Chapter III draws more heavily than other chapters from classical theory as compiled by Glagert<sup>1, 4</sup>. By 1935 the basic theory for simple lifting-line wings with a tail are was available for most tunnel shapes. Improved methods have been developed subsequently for calculating the interference upwash in rectangu'ar and circular tunnels, but the basic theory survived. New tunnel configurations have necessitated the theoreticki development for closed octagonal tunnels by Batchelor<sup>1, 18</sup>, and Katsoff et al.<sup>1,17</sup> have analysed the problet of open circular tunnels of finite jet length. The testing of half-models mounted on a reflection plane has in effect introduced other chapes of tunnel; the bipolar shapes corresponding to circular tunnels have been treated with closed brondaries by Sivells and Salmi<sup>1, 19</sup> and with open boundaries by Davison and Rosenhead<sup>1, 20</sup>. Gther developments for the lifting-line model of a wing with tail include off-centre models considered by Silverstein and White<sup>1, 21</sup> and interference corrections on ground effect by Brown<sup>1, 22</sup> where the floor of the tunnel is used to simulate the ground. The ultimate refinement of the lifting-line treatment is that of Sanders and Pounder<sup>1, 23</sup> for closed rectangular tunnels.

The main developments of three-dimensional lift interference in the past twonty years have concorned the sweptback, and later sleeder, wings designed for high-speed flight. The lifting-line model no longer applies, so that a different representation of the lifting wing by distributed vortices is needed, together with more elaborate wethods of interpreting the interference upwash. Difficulties of presentation in Chapter III arise from the wide variety of vortex models, notations and procedures for obtaining the interference corrections. The models vary from distributions of lifting elements along the locus of sectional streamwise centres of pressure used by Acum<sup>1, 28</sup> to the point concentrations of lift used by Katzoff and Haunah<sup>1, 25</sup>. The ferentiation varies from the admirable approximate expressions in Reference 1.19 to that of Reference 1.26 which requires the use of one of the lifting-surface theories that have come to replace the inadequate lifting-line theory. Allowance for Mach number M is included, wherever possible, by using the Prandtl-Glavert analogy and insorting the appropriate power of  $\beta = \sqrt{(1 - M^2)}$  as a factor in the geometric and aerodynamic perpendents of an equivalent incompressible flow. In this way the results can usually

9

なったらは、小田などのなどがないののなどでもないないできないないできたから、たくなくてきないとなるとなるとなっていましょうです。

be presented for a general subsonic Mach Augher and in a unified Solution. The special case of slender wings involves a distribution of lifting elements along the axis of the tannel, and is this sense can be regarded as an extension of the theory for small wings. Berndt<sup>1.27</sup> has given an exact theory for slender wings, to which a useful simplifying approximation can be made.

There are some aspects of lift-interference correction that are much less satisfactory. These arise primarily through deficiencies or complexities in the theoretical treatment of the corrections are quite large, are control-surface hinge moments, vortex-induced yaving moment and wing-body combinations. The hinge moments involve uncertainties of large viscous effects, and the corrections of Bryant and Gerner<sup>1, 28</sup> allow for these on a crude two-dimensional basis. More serious are thought to be the shortcomings of the interference corrections to yaving moment. Apparently all existing formulae for these are derived from lifting-line theory. Likewise the theory of wall interference on lifting wing-body combinations is inadequate, and the author concludes that the literature on this aspect of lift interference is misleading.

It is observed experimentally by Knight and Harris<sup>1, 25</sup> on wings of moderately high asynct ratio, that the interference parameter for the drag coefficient  $C_{\rm B}$  is closer to the lifting-line correction than is the interference parameter for correcting the incidence a; that is to say, the streamline curvature associated with the streamwise extent of the model has a larger effect on  $\Delta a$  than on  $\Delta C_{\rm D}$ . However, circumstances are different for slender wings with leading-edge separation. With the loss of leading-adge suction the effects of streamline curvature on  $\Delta a$  and  $\Delta C_{\rm D}$  become equally important, but the evaluation poses an unsolved non-linear problem.

The diversity of methods and notations is coupled with scarcity of published numerical enumples. An attempt is made in Chapter III to digest numerical data in the most general form, with particular emphasis on formulae that are applicable to general subsonic Mach number, general planform and a range of tunnel shape. Simplified dethods of evaluating the interference corrections are developed where possible. Specific comparisons between the results of different methods are quite encouraging and show which of them are useful approximations and which give primarily a rapid estimate of the pider of magnitude of the corrections.

#### 1.3.2 Blockage Interference

Chepter V is based on assumptions that the lift is not large and that solid and wake tlockage effects are independent. It further assumes that only longitudinal flow is subject to blockage interference, which is only true if the model is mounted in the centre of the tunnel. For closed boundaries the condition of zero normal velocity is reliable. The corresponding assumption for an open-jet boundary is one of constant pressure; mixing effects are ignored and a linearized condition is applied at the undisturbed boundary. Moreover, it is frequently assumed that the jet is of infinite length, but in practice the nozzle and collector are relatively near the model and may strengly influence the constraint corrections<sup>1, 17</sup>. It follows that the open-jet corrections, though usually smaller in magnitude, are less soundly based than those for a closed tunnel; furthermore, there are few experimental data from which to assess the validity of the estimated corrections for open tunnels.

With these limitations the range of available corrections is fairly complete. Two-dimensional solid and wake blockage are discussed by Thom in Reference 1.30, and Goldstein<sup>1, 13</sup> gives a more general treatment including terms in  $(c/h)^4$ . With particular reference to the work of Herriot<sup>1, 31</sup>, the solid and wake blockage corrections to the usual aerodynamic coefficients may be calculated with some confidence for three-dimensional models in tunnels having a wide range of rectangular cross-sections. The circular tunnel also presents no great problems, but for elliptical and octagodal cross-sections certain simplifying assumptions need to be made to relate these to corresponding simpler shapes. Allen and Vincenti<sup>1, 32</sup> give a sound basis on which the effect of compressibility at subcritical Mach numbers can be incorporated simply by linear-perturbation techniques; even the presence of local regions of supersonic flow about the model, with attendant shock waves, may be allowed for in an empirical manner.

Thus the topic of blockage interference on small streamline models is in a reasonably satisfactory state, though there are instances in Chapter V of differences of approach which can lead to somewhat different answers. These, however, are the exceptions in an area which after nearly forty years is, within its limitations, approaching finality.

The bulk of established interference theory is concerned with streamline flow. It is implicit that the wake is thin and that it can be taken to originate from the trailing edge of a wing or from the rearmost point of a body. However, there is a growing interest in flows which depart significantly from this classical pattern and also some evidence<sup>1.33</sup> to show that new, and sometimes surprising, interference effects may arise.

It is perhaps worth emphasising that high lift does not, in itself, imply a departure from the classical flow pattern. The principles of the classical interference theory of Chapter V can, therefore, remain valid, although wind-tunnel models may give rise to larger corrections than are strictly within the scope of current theories. Highorder calculations of the interference field are straightforward in principle, but it may well be impossible to interpret such calculations usefully in terms of modified free-stream conditions. In any event, it is plainly dangerous to assume that large corrections can be estimated satisfactorily. To do so implies that the entire flow field could have been calculated with fair precision at the outset.

The extension of classical interference theory to a non-streamline flow is never straightforward, even though the appropriate corrections may be small. The distinguishing feature of a problem of this kind is that it requires, as a first step, the establishment of a suitable mathematical model of the given flow. The dominant physical characteristics of the flow must be described adequately by this model, but it must remain simple enough to admit of further analysis. Some typical examples are considered in Chapter VII, in particular Maskell's<sup>1, 33</sup> theory of blockage effects on two-dimensional and three-dimensional bluff bodies and on stalled wings in a closed wind tunnel.

#### 1.3.3 Ventilated Tunaels

The development of the ventilated-wall tunnel arose from the desire to exploit the opposing interference effects associated with closed and open boundaries; by combining these in some judicious manner zero wall interference might be obtained. The subsequent attainment of transonic test speeds seems to have been a most fortunate hy-product, with its own rapid exploitation. The development of general theories for the calculation of

11

િંદ્રે કેર્પ્સિટેટ કે સંસ્થાયક્ષ્ય કેર્પ્સાયક અડ કાર કેર્પ્સા પ્રસાળ પ્રયાગ પ્રચલ કે અવર્ત્સ સ્પર્ધવૃત્તિ જ અગલ વર્ષ્ય વ્યવસાય અવ્ય

Contraction and the second

wall interference stems from the concept of replacing the wixed boundary conditions by a single homogeneous condition valid over the entire wall, as discussed in Section 1.2 (Refs. 1.1 and 1.11). Baldwin et al.<sup>1.38</sup> have constructed a condition for inviscid or viscous flow at a ventilated boundary. The actual geometry of a slotted wall and the degree of porosity of a slotted or perforated wall appear as two parameters in the homogeneous condition. The validity of this simplification has been established by comparison with earlier and more exact calculations of Fistolesi<sup>1.35</sup> and Katthews<sup>1.36</sup> for 2 circular tunnel with discrete slots, and by examination of experimental data for the poresity parameter as in Reference 1.37. Thus the way is open for the calculation of the interference corrections due to lift and blockage; the latter assumes great importance as the Mach number approaches the region of high subsonic fl-m.

During the preparation of Chapter VI some calculations of two-dimensional lift interference have become available; Holder<sup>1, 35</sup>, using the boundary condition of Reference 1.34, evaluates the influence of slot geometry and porosity for a small two-dimensional model in a slotted rectangular tunnel. Two-dimensional blockage interference is covered fairly completely in References 1.29 and 1.34. A similar situation exists for small three-dimensional models in circular slotted tunnels (Sefs. 1.40 and 1.37), but the more important cases of rectangular tunnels with slotted floor and roof are rather more difficult to calculate. Davis and Moore<sup>1. 20</sup> give a few results for lift and blockage interference when there is inviscid flow near the slots, but important extensions to their analysis have appeared recently in References 1.41 and 1.42. Holder<sup>1.41</sup> has given analytical expressions and representative calculations of lift interference to satisfy the homogeneous condition including the porosity parameter. Acum<sup>1,42</sup> has sumplifed the expression for the solid-blockage factor in Reference 1.40 and gives results for a wide range of rectangular crosp-sections. The influence of viscous slot-flow on blockage interference does not seen to have been considered for this type of tunnel.

The theoretical approach to interference corrections is a rectangular tunnel with ail four walls slotted is less satisfactory. Lift and blockage interference can be estimated from Chapter VI on the basis of limiting values for closed and completely open tunnels by allowing an analogy between the effect of slot gcometry in circular and other tunnels. A similar basis wight be used to estimate streamline curvature, but there is insufficient reliable information for any type of slotted tunnel. Solutions by electrical analogue should prove useful here, and a promising start has been made by Rushton in Reference 1.43 and the subsequent development of a threedimensional network. Little experimental information exists on the value of the porosity parameter defining viscous effects near the slots, and in the absence of this the theoretical results cannot be applied with any precision. More experiments are essential, because estimates of both lift and blockage interference made for ideal wall conditions may be greatly in error.

The perforated wall has distinct characteristics, mentioned in Section 1.2 and discussed more fully in relation to experiment in Reference 1.1. The boundary condition is a limiting form of that used for a slotted wall with viscous flow as the slot spacing tends to zero. Farametric analysis suggests that the perforated tunnel behaves more and more like an open jet az sonic flow is approached. Theoretical calculations of interference corrections in terms of the porocity parameter are available for lifting and non-lifting two-dimensional models and fairly small wings in circular tunnels from the work of Goodman in References 1.44 and 1.45 and later developments given in Reference 1.46.

THE REAL PROPERTY OF THE PROPERTY OF THE REAL PROPE

Blockage effects, however, have not been considered in a rectangular tunnel having two opposite or all four walls perforated. A lifting model may be treated as a special case of Reference 1.41 when the floor and roof are perforated. If the side-walls are perforated also, a graphical procedure in Chapter VI may be followed, but this is somewhat unsatisfactory as the square tunnel is considered by analogy with a circular tunnel.

#### 1.3.4 Unsteady Interference

Nost of Chapter IV is concerned with escillatory problems, but some consideration is given to rotating models. Rolling wings of high aspect ratio can be treated fairly satisfactorily; for more slender wings only a quasi-steady theory is available, but the corrections are small. The interference corrections to propeller tests in Reference i.4 have been standardized for many years.

There are rarely formulae or tables for corrections in oscillatory experiments, and Chapter IV is much more concerned with methods, some of which involve heavy computation. The subject is in a fairly satisfactory state numerically as regards two-dimensional closed tunnels. Timman<sup>1.47</sup> gives the basic theory for incompressible flow; compressibility introduces the phenomenon of acoustic resonance which is included in the general method of Runyan et al. 1. 46. For closed three-dimensional tunnels, also, there is a convenient theory, provided that the product of Mach number and frequency parameter is small; the basic idea stems from the work of Goodman<sup>1, 49</sup>, by which the unsteady interference upwark can be derived from its distribution upstream of the model in steady flow. The respective applications to closed rectangular and circular tunnels by Acua in References 1.50 and 1.51 suffice for incompressible flow. Chapter IV includes a generalization for subsonic Mach numbers and small frequency parameters. Ac a1.52 gives more explicit results for slowly oscillating slenger wings, but even for slender wings there is no method when the Mach number and the frequency parameter are both fairly large, whatever the tunnel cross-section may be. Fortunately the frequency parameter is usually small for experiments on rigid models in subsonic compressible flow, but corrections to flutter tests pose a difficult problem.

The problems for vantilated tunnels are also difficult and perhaps more pressing. Wight<sup>1,53</sup> has reviewed experimental data obtained in slotted-walk tunnels with slots open and with slots scaled, which reveal some particularly large interference effects on damping derivatives. Unfortunately the steady interference upwash in ventilated tunnels has not jet been formulated in sufficient detail to be used on the basis of Reference 1.49. Novertheless, a likely explanation of the observed differences in wall interference between closed and slotted tunnels is reported in Chapter IV through an extension of the classical theory of lift interference.

Further progress along the lines of investigation described in Chapter IV seems likely to lead to increased analytical, algebraic and computational complexity. One is tempted to wonder whether a better alternative would not be to use direct numerical solutions of the differential equations or analogue experiments, such as those envisaged in Reference 1.17 or in course of development as an extension to Reference 1.43.

13

1. v v 1.

「日本のです」

#### 1.3.5 High-Lift Systems

A common feature of the examples dealt with in Chapter VII is that they are concerned with flows differing materially from the essentially stream-line flow, with small lift, of the previous chapters. The flow past jet flaps or slender wings with leadingedge vortices can be regarded as an extension of the classical streamline flow. The appropriate mathematical model is still a vortex sheet, but one which includes nevel features that require special attention when the effects of the interference field are to be interpreted. Maskell's<sup>1.5%</sup> treatment of a three-dimensional jet-flap in a closed tunnel illustrates this group of problems. In others, for example bluff bodies and stalled wings (Ref. 1.33), a thick wake of uncertain structure is the dominant characteristic of the flow patterns. The main problem is to identify those properties of the wake that are most significant, and then to devise a mathematical model that reproduces them with sufficient accuracy.

A third group of problems is concerned with configurations for vertical or short take-off and landing, in which lifting roters, fans or jets may be combined in various ways with wings and bodies to form systems in which the interference between the various constituent flows is fundamental to the performance of the system as a whole. Heyson has given linearized theories of tunnel-wall corrections for lifting rotors (Ref. 1.55) and other high-lift systems (Ref. 1.56). In the latter case there are serious unresolved difficulties, both in the construction of a suitable mathematical model of the flow and in the interpretation of the effects of wall constraint on the mutual interference between the different elements of the system. The subject is well reviewed by Templin in Reference 1.57 which is complementary to Chapter VII.

#### 1.4 OUTSTANDING PROBLEMS FOR RESEARCH

There remain unresolved difficulties in subsonic wind-tunnel wall interference, only a few of which are likely to be overcome by mathematical analysis alone. In some instances the most promising method of solution may well be one of numerical analysis by a finite-difference technique: experiments by electrical analogue offer an alternative approach. In others, the primary handlesp is the lack of a mathematical model or definitive boundary conditions, and progress may necessarily have to stem from experimental research. Apart from the thirteen general problems grouped below, there is scope for wind-tunnel investigations to check the applicability of existing theoretical methods of interference correction; notable exemples are the non-linear two-dimensional theory of merofolls with trailing-edge flaps at large angles of deflection<sup>1, 15</sup>, lifting element wings in closed tunnels<sup>1, 27</sup>, hinge moments from three-dimensional control testing<sup>1, 29</sup>, and blockage interference on bluff models for non-meronantical purposes<sup>1, 33</sup>.

#### Closed and c en tunnels

1. Two-dimensional cambered serofoils are often tested at high sub-critical Mach numbers. In such cases the affects of t/c and c/h of wall interference are aggravated by compressibility. It is desirable to seek a better solution to the hon-linear sathematical problem than the relatively simple extension to Reference 1.13 suggested in Gaspter II.

2. The conventional correction to yawing moment is calculated on the basis of lifting-line theory. In Chapter III this is shown to be problematical for wings of moderately small aspect ratio, and a precise treatment of the interference correction demands a reliable lifting-surface calculation of the spanwise distribution of leading-edge suction. Such a development for subsonic flow would have many applications.

3. For slender wings with leading-edge separation the correction to drag needs clarification; it is likely to be larger than in cases of attached flow. There is scope for a non-linear theoretical treatment of constrained flow with leading-edge vortices.

4. There exists no satisfactory method of applying residual interference corrections to, say, the pitching moment on a lifting wing-body combination; the theories discussed in Chapter III are thought to be of little help. As the corrections are unlikely to be very large, there may be justification for an adaption of slender-wing theory.

5. The uncertainties of blockage interference in open tunnels impose considerable restrictions on model size (Chapter V). The problem becomes more acute when bluff bodies have to be tested in open tunnels. It is important to establish whether the corrections for non-streamline flow increase as dramatically in open tunnels as in closed tunnels<sup>1.33</sup>.

6. The discussion of Reference 1.55 in Chapter VII suggests that a convincing theory of wall interference on high-lift systems may require an improved mathematical model and further insight into the interpretation of corrections. This most novel aspect of wall interference may require inspiration from experimental sources.

#### Ventilated tunnels

7. There are insufficient numerical data on the streamline curvature induced by slotted walls. Even for rectangular tunnels there is no image system; the problem may best be solved by means of a three-dimensional electrical analogue (Chapter VI).

8. It has been demonstrated theoretically, and is illustrated in Chapter VI, that viscous effects near slots, represented by a constant porosity parameter, have large influence on lift and blockage interference. More experimental work is needed to establish an empirical formula for the porosity parameter appropriate to various Mach numbers and types of slot geometry, and to verify that its variation along the tunnel is small enough for the theory to apply.

9. There appears to be no information on three-dimensional solid-blockage interference in perforated rectangular tunnels. Each arrangement of perforated walls poses a distinct problem that should be amenable to mathematical analysis.

10. The problem of lift interference in rectangular tunnels with all four walls perforated is considered by a crude graphical procedure in Chapter VI. The interference upwash in the transverse plane of a lifting element is not expressible in terms of a two-dimensional flow, and solution by electrical analogue is recommended.

#### Unsteady interference

11. One mathematical definition of the linearized problem of an oscillating threedimensional wing in a subsonic closed rectangular tunnel is given in Chapter IV, but

15

Ŷ

per the local and and

¥

numerical analysis may well be prohibitive when neither Mach number nor frequency parameter is small. A new approach to this problem might be sought, say, for slender models.

12. As discussed in Chapter IV, there have been mathematical formulations of the interference on an oscillating model in a perforated tunnel, but the possibility of a phase difference between the pressure drop and the normal flow through the perforations has not been considered. This possibility is worth examination theoretically and experimentally in two-dimensional flow.

13. More theoretical work is needed to establish a method of evaluating the large interference effects on oscillating wings in slotted-wall tunnels<sup>1, 53</sup>. It is desirable to examine experimentally whether or not the effects are smaller in perforated tunnels and how they can be minimized in existing ventilated tunnels.

#### 1.5 ACKNOWLEDGEMENTS

The preparation of Chapters I to VI of the AGARDograph formed part of the research programme of the National Physical Laboratory. The authors wish to acknowledge the assistance of Mrs S.Lucas with the calculations and diagrams. Acknowledgements are due to the members of the Fluid Dynamics Panel of the Advisory Group for Aerospace Research and Development, and to Mr A.B.Haines and Dr R.C.Pankhurst who offered constructive criticism of the draft chapters.

#### REFERENCES

1.1	Goethert, B.H.	Transonic Wind Tunnel Testing. AGARDograph 49. Pergamon Press, 1961.
1.2	Rogers, E.W.E.	A Background to the Problems of Wind-Tunnel Interference. NPL paper presented at ACARD meeting on Interference Effects. AGARD Report 292, 1959.
1.3	Prandtl, L.	Tragflügeltheorie, Part II. Nachrichten der K.Gesellschaft der Wissenschaften zu Göttingen, 1919. Reprinted in Vier Abhandlungen zur Hydrodynamik und Aerodynamik by L.Prandtl and A.Betz. Göttingen, 1927.
1.4	Glavert, H.	Sind Tunnel Interference on Wings, Bodies and Airscrews. ARC R & M 1566, 1933.
1.5	von Kármán, T. "urgers, J.M.	General Aerodynamic Theory - Perfect Fluids. Airfoils and Airfoil Systems of Finite Span. Aerodynamic Theory (ed. W.F.Durand), Vol.II, Div.E, Chap.IV, Julius Springer, Berlin, 1935, pp.236-280.

1.6	Tousseint, A.	Experimental Methods - Wind Tunnels. Influence of the Dimensions of the Air Stream. Aerodynamic Theory (ed. W.F.Durrid), Vol.III, Div.I, Chap.III, Julius Springer, Berlin, 1935, pp.280-319.
1.7	Riegels, F.W.	Wind Tunnel Corrections for Incompressible Flow. AVA Monographs (ed. A.Betz), Section D <sub>3</sub> 4.1. MAP Völkenrode VG-258. Reports and Trans.958, 1947.
1.8	Katzoff, S.	Wind-Tunnel-Wall Corrections. NACA-University Conference on Aerodynamics. 21st - 23rd June, 1948. (NACA/TIB/1783).
1.9	Pankhurst, R.C. Holder, D.W.	Wind-Tunnel Technique, Chap.8. Tunnel Interference Effects. Pitman, London, 1952. (Reprinted in 1965).
1.10	Рорс, А.	Wind-Tunnel Testing, Chap.VI. Wind-Tunnel-Boundary Conditions. Second Edition. Wiley, New York, 1954.
1.11	Allen, H.J. Spiegel, J.M.	Wind Tunnel Measurements. Test-Section Flow Uniformity. High Speed Aerodynamics and Jet Propulsion, Vol.VIII, High Speed Problems of Aircraft and Experimental Methods, Part 2, Wind Tunnel Techniques, Section K (ed. F.E.Goddard). Oxford University Press, 1951.
1.12	Molyneux, W.G.	Wind Tunnel Interference in Dynamic Measurements. RAE Tech. Report 64069. ARC Report 26,673, 1964. Presented at a short course on the Use of Flexible Models in Aeroelastic Research at the Von Kármán Institute of Fluid Dynamics, May 1964.
1.13	Goldstein, S.	Two-Dimensionai Wind-Tunnel Interference. ARC R & M 1902, 1942.
1.14	Knechtel, E.D.	Experimental Investigation of Two-Dimensional Tunnel-Wall Interference at High Subsonic Speeds. NACA Tech. Note 3087, 1953.
1. 15	Preston, J.H. Manwell, A.R.	Calculation of the Interference on a Thin Symmetrical Aerofoil with Hinged Flap Spanning a Closed Wind-Turnel. ARC R & M 2465, 1941.
1. 16	de Jager, E.M. van de Vooren, A.I.	Tunnel-Wall Corrections for a Wing-Flap System between Two Parallel Walls. NLR (Netherlands) Report W.7, 1961.
1.17	Katzoff, S. et al.	Linear Theory of Boundary Effects in Open Wind Tunnels with Finite Jet Lengths. NACA Report 976, 1950.
1. 18	Batchelor, G.K.	Interference in a Wind Tunnel of Octagonal Section. Report ACA-1 (Australia), 1944.

÷

ž

1

-----

÷

ł

•

;

and a second second

and and any state of the state

÷

South States of the second

1

1

- 4

ì

うちまたろうち たちかいろい ちくちしまい マンス

1.19	Sivells, J.C. Salmi, R.M.	Jet-Boundary Corrections for Complete and Semispan Swept Wings in Closed Circular Wind Tunnels. NACA Tech. Note 2454, 1951.
1.20	Davison, B. Rosenhead, L.	Wind Tunnel Correction for a Circular Open Jet Tunnel with a Reflexion Plate. Proc. Roy, Soc. Series A, Vol. 177, 1941, pp. 366-382.
1.21	Silverstein, A. White, J.A.	Wind-Tunnel Interference with Particular Reference to Off-Centre Positions of the Wing and to the Downwash at the Tail. NACA Report 547, 1935.
1.22	Brown, W.S.	Wind Tunnel Corrections on Ground Effect. ARC R & M 1865, 1938.
1.23	Sanders, J. Poander, J.R.	Wall Interference in Wind Tunnels of Closed Rectangular Section. NRC (Canada) Report AR-7, 1949.
1.24	Acum, W.E.A.	Corrections for Symmetrical Swept and Tapered Wings in Rectangular Wind Tunnels. ARC R & M 2777, 1950.
1.25	Katzoff, S. Hannah, M.E.	Calculation of Tunnel-Induced Upwash Velocities for Swept and Yawed Wings. NACA Tech. Note 1748, 1948.
1.26	Garner, H.C. Acum, W.E.A.	Interference Corrections for Asymmetrically Loaded Wings in Closed Rectangular Wind Tunnels. ARC R & M 2948, 1953.
1.27	Berndt, S.B.	Wind Tunnel Interference Due to Lift for Delta Wings of Small Aspect Ratio. KTH (Sweden) Tech. Note Aero TN 19. 1950.
1.28	Bryant, L.W. Garper, H.C.	Control Testing in Wind Tunnels. AKC R & M 2881, 1950.
1. 29	Knight, M. Harris, T.A.	Experimental Determination of Jet Boundary Corrections for Airfoil Tests in Four Open Wind Tunnel Jets of Different Shapes. NACA Report 361, 1930.
1.30	Tho?. A.	Blockage Corrections in a Closed High-Speed Tunnsl. ARC R & M 2023, 1943.
1.31	Herriot, J.G.	Blockage Corrections for Three-Dimensional Flow Closed- Throat Wind Tunnels, with Consideration of the Effect of Compressibility. NACA Report 995, 1950.
1.32	Allen, H.J. Vincenti, W.G.	Wall Interference in a Two-Dimensional-Flow Wind Tunnel, with Consideration of the Effect of Compressibility. NACA Report 78%, 1944.
1.33	Maskell, E.C.	A Theory of the Blockage Effects on Bluff Bodies and Stalled Wings in a Closed Wind Tunnel. ABC R & M 3490, 1963.

## ļ

:

.

1.34 Baldwin, E.S. et al.	Wall Interference in Wind Tunneîs wich Slotted and Porcus Boundaries at Subsonic Speeds. NACA Tech. Note 3176, 1954.
1.35 Pistolesi, E.	<ul> <li>(1) On the Interference of a Wind Tunnel with a Mized Boundary. Pont. Accad. Sci. Vol.4, No.9, 1940.</li> <li>(Unpublished translation by Cornell Aero. Lab., 1949.)</li> <li>(11) On the Interference of a Wind Tunnel with Mixed Contour. Pont. Accad. Sci. Vol.7, No.5, 1943.</li> <li>(Unpublished translation by NACA, 1951.)</li> </ul>
1.36 Matthews, C.W.	Theoretical Study of the Tunnel-Bundary Lift Interference Due to Slotted Walls in the Presence of the Trailing-Vortex System of a Lifting Wodel. NACA Report 1221, 1955.
1.37 Reynolds, W.G. et al.	Boundary Interference Studies in the Slotted Test Section of the Southern California Cooperative Wind Tunnel for the Mach Number Range 0.4 to 1.3. CWT Report P-19, 1959.
1.38 Helder, D.K.	Uppash Interjerence in a Rectangular Wind Tungel with Closed Side Walls and Porous Slotted Floor and Rocf. ARC R & ¥ 3322, 1962.
1.39 Maeder, P.F.	Theoretical Investigation of Subsonic Wall Interference in Rectangular Slotted Test Sections. Brown Univ. Div. of Eng. Tech. Report WT-11, 1953.
1.40 Davis, D.D. Moore, D.	Analytical Study of Blockage – and Lift-Interference Corrections for Slotted Tunnels Obtained by the Substitution of an Equivalent Homogeneous Boundary for the Discrete Slots. NACA RM LESEOTD (NACA/TIB/3792), 1953.
1.41 Holder, D.R.	Upwash Interference on Wings of Finite Span in a Rectangular Wind Tunnel with Closed Side Walls and Porous-Slotted Floor and Roof. ARC R & W 3395, 1963.
1.42 Acum, W.E.A.	Note on the Evaluation of Solid-Blockage Corrections for Rectangular Wind Tunnels with Slotzed Walls. ARC R & N 3297, 1961.
1.43 Rushton, K.S.	Studies of Slotted-Wall Interference Using an Electrical Analogue. Purt II. Particular Examples of Slotted-Wall Tunnel Interference. ARC R&W 3452, 1965.
1.44 Goodman, T.R.	The Porous Kall Wind Tunnel, Pt. V. Subsonic Interference. on a Lifting Wing between Two Infinite Walls. Cornell Aero. Lab. Report AD-706-A-3, 1951.
1.45 Goodman, T.R.	The Porous Wall Wind Tunnel, Pt. IV. Subsonic Interference Problems in a Circular Tunnel. Cornell Aero. Lab. Report AD-706-A-2, 1951.

÷

19

States a weather of which a second

1

• \_

1.46 Wright, R.H.

1.47 Timean, R.

1.48 Sunyan, H.S.

1.49 Goodman, T.R.

1.50 Acum, W.E.A.

1.51 ACUS, W.E.A. Gerner, H.C.

1.52 Acom W.E.A.

1.53 Wight, K.C.

1.54 Maskell, E.C.

1.56 Heyson, H.H.

1.57 Templin, R.J.

The Effectiveness of the Transonic Wind Tunnel as a Device for Minimizing Tunnel-Boundary Interference for Model Tests at Transonic Speeds. NASA paper presented at AGARD meeting on Interference Effects. AGARD Report 294, 1959.

The Aerodynamic Forces on an Oscillating Aerofoil between Two Farallei Wulls. App. sci. Res. A (Hegue), Vol.3, 1951. pp. 31-57.

Theoretical and Experimental Investigation of the Effect of Tunnel Walls on the Forces on an Oscilloting Airfoil et #1. in Two-Dimensional Subsonic Compressible Flow. NACA Report 1262, 1956.

The Upwash Correction for an Oscillating Wing in a Wind Tunnel. J. aero. Sci. Vcl. 29, 1953, pp. 383-386, 406.

Wall Corrections for Wings Oscillating in Wind Tunnels of Closed Rectangular Section. ARC R & M 3312, 1958.

Approximate Wall Corrections for an Oscillating Swept Wing in a Wind Tunnel of Closed Circular Section. ARC CP 184, 1954.

A Note on the Estimation of the Effect of Wind Tunnel Walls on the Forces on Slowly Oscillating Slender Kings. ARC CP 707, 1963.

A Review of Slotted-Wall Wind Tunnel Interference Effects on Oscillating Models in Subsonic and Transonic Flows. J. Roy. seron. Soc. Vol.68, 1964, pp. 670-674.

The Interference on a Three-Dimensional Jet-Flap Wing in a Closed Wind Tunnel. ARC. R & ¥ 3219, 1959.

Jet-Boundary Corrections for Lifting Rotors Centered in 1.55 Heyson, H.H. Rectangular Wind Tunnels. MASA Tech. Report R-71, 1960.

Linearized Theory of Wind-Tunnel Jet-Boundary Corrections and Ground Effect for VIOL-STOL Aircraft. NASA Tech. Report R-124, 1982.

The Choice of Working Section Size and Shape for V/STOL Wind Tunnels. NEC (Canada) Quarterly Bulletin DME/NAE 1965 (4), 1965.

CHAPTER 13

and the second of the second of the second secon

5200¢755

= (j

ž

まった。 - 14月3日日 Hatter Add Addina (1) きってんかんしょうべんがいためがくない Partic Activity (1) (1) Jan Particity (1) (1) Particity マン・マン・マ

1999) 1999)

marces N

## LIFT INTERFERENCE ON TWO-DIMENSIONAL WINGS

by

E.C.Garner

Aerodynamics Division, National Physical Laboratory, Teddington, Middlesex, England

# **BLANK PAGE**

CONTENTS

;

		Page
LIST	T OF TABLES	24
LIST	OF FIGURES	24
NOTA	ATION	25
2.1	INTRODUCTION	29
2.2	WINGS OF ZERO THICKNESS IN CLOSED TUNNELS	31
	2.2.1 Corrections to Order (c/h) <sup>2</sup>	31
	2.2.2 Corrections to Order (c/h) <sup>4</sup>	34
	2.2.3 Non-Rectangular Tunnels	36
2.3	WINGS OF FINITE THICKNESS IN CLOSED TUNNELS	37
	2.3.1 Effect of One Wall	37
	2.3.2 Goldstein's Theory	37
	2.3.3 Allowance for Compressibility	42
2.4	AEROFOILS WITH HINGED FLAPS	44
2.5	OPEN-JET TUNNELS	46
	2.5.1 Boundary Conditions	46
	2.5.2 Fully Open Jets	47
	2.5.3 Two-Dimensional Tests	49
	2.5.4 Spillage Behind the Model	52
2.6	EXPERIMENTAL CONSIDERATIONS	53
	2.6.1 Viscous Effects	53
	2.6.2 Closed Tunnels	54
	2.6.3 Open-Jet Tunnels	56
REFE	ERENCES	57
ADDI	ITIONAL REFERENCES	59
<b>TABI</b>	L£	61
FIGU	URES	62

23

. \_\_\_:

. .

٠ŧ

**t** 5

1 12 C

and the second second

and the

## LIST OF TASLES

\* \*\*\*

4

Page

 TABLE 2.1
 Summary of Principal References, Figures and Formulae
 61

## LIST OF FIGURES

Fig. 2, 1	Thin two-dimensional serofoll and its image system for lift interference	62
Fig. 2, 2	Spanwise distribution of lift and pitching moment on an merofoil spanning a closed rectangular tunnel	63
Fig. 2. 3	Closed-tunnel to free-strame ratios of lift and moment on a fist-place merofoil	64
Fig. 3.4	Wall corrections for an RAE 102 acrofoll	65
<b>Pig. 2.5</b>	Closed-tunnel to free-stream ratios of derivatives for merofoils with kinged flaps (Ref. 2.14)	66
Fig. 2. 6	Wall corrections to flap hinge moments	67
Fig. 2.7	Wall corrections due to flap setting	68
Fig. 3. 8	Effect of large flap setting on lift interference $(h = 0)$	69
Fig. 2, 9	Overections to pitching moment due to a deflected flap $(8 = 0)$	70
Fig. 2. 10	Lift interference in rectangular open-jet tunnels	71
Fig. 2 11	Lift interference in circular open-jet tunacls	72
Fig. 2. 12	Functions from equations (2.71) to (2.75) for closed-open-closed tangels	73
Pig. 2. 13	Experimental variation of $C_L$ with N for various c/h	74

24

23.5

## NOTATION

ş

a		speed of sound
<b>a</b> 1		lift slope = $\partial G / \partial \alpha$
8 <sub>2</sub>	z	ુદ <sup>1</sup> /9ર્ટ
К <sub>а</sub>		coefficient in chordwise loading from Equation (2.12)
р		breadth of tunnel
b'		twice the horizont g projection of a corn r fillst
<b>b</b> 1	z	9 <b>૯</b> મેં.9¤
b <sub>2</sub>	z	કુર્યોઝ્ટ
b <sub>E</sub>		coefficient in spanwise loading from Equation (2.51)
c		chord of serofoil
с		cross-sectional area of tunnel
€ <sub>D</sub>		drag coefficient = (drag per unit span) $/\frac{1}{t}\rho U^2$
с <sub>н</sub>		(hinge moment per unit spen) $/\frac{1}{2}\rho U^2 E^2 c^2$
C <sub>L</sub>		lift coefficient = $L/\frac{1}{2}\rho U^2 c$
C <sub>m</sub>		pitching-moment coefficient = $\frac{1}{2}/\frac{1}{2}\rho g^2 c^2$
C <sub>n</sub>		function of servicil profile in Equation (2.37) $(n = 0, 1, 2,)$
đ		distance of scrofoil from floor of tunnel (Pig. 2, 1)
o <sub>n</sub>		function of serofoil camber line in Equation $(2,34)$ $(a = 0, 1, 2,)$
Б		ratic of flap chard to scrofoil chord
G <sub>o</sub>		function defined in Equation (2.73)
G <sub>1</sub>		function defined in Equation (2.73)
2		height of tunad
₹,		twice the vertical projection of a conser fillet
b <sub>e</sub>		equivalent height of tuanel (Section 2.2.3)
k		vortex strength per unit leugth

•

City.

L \. >

2F.

Ż

Considerably which are the

14 Mar

.....

14 - 541 14 C.M. WAL 40

 Ē.

26 K strength of vortex l non-disensional aerodynamic loading in Equation (2.14) Ł Length of open working section (Section 2.5) lift per unit span Ł  $\partial C_{\partial \alpha}$ ₹, ∂**c\_**∕∂<u>₹</u> 3, = Mach number M h pitching moment per unit span about axis  $x = \frac{1}{2}c$ R.tc pitching moment per unit span about axis  $x = \frac{1}{7}c$ p pressure radius of sirce's tennel R ares of planform of model 8 2 thickness of acrofoil x-component of velocity U u' perturbation in horizontal velocity at upper boundary of jet (Section 2.5.4" velocity of undisturbed stream 11 z-component of velocity value of w in distant wake ¥+0 distance downstream of leading edge of aerofoil X 7 centre of pressure of aerofoil in Equation (2.8) distance of model from entrance nozzle of open-jet tunnel xa x in transformed plane Х spanwise distance from centre of tunnel F distance upwards from leading edge of sercioii z transformed co-ordinate =  $\beta_z$ Z incidence of aerofoil (in radians onless otherwise stated) α

discussion of

IN SET THE STATE WAS THE PARTIN SAME IN THE SAME AND THE SET OF

β	=	$(1 - M^2)^{\frac{1}{2}}$
Г		virculation around aerofoil
۵		prefix denoting increment due to wall correction
ζ		complex variable = X + iZ
θ		chordwise parameter defined in Equation (2.13)
λ		parameter $3b/\pi a_1 c$ used in Section 2.5.2
$\mu_{\mathbf{R}}$		viscous currection factor in Equation (2.77)
<i>⊦</i> ъ		viscous currection factor in Equation (2.78)
ξ		angle of flep setting (in radianc)
p		density of undisturbed stream
¢		velocity potential of two-dimensional flow
Ψ		strear function of two-dimensional flow

The new a selected and a selected

ڊ ڊ

.....

27

## Subscripts

•

.!

「たいていいい」なないというとないとなどと

i	denotes	quantity	induced	by tun	rel	wellis.	
0	denotes	free-stre	an aerod	lynazic	: c0e	efficien	nt.

se.

7 denotes theoretical value.



### LIFT INTERFERENCE ON TWO-DIMENSIONAL WINGS

#### H. C. Garner

#### 2.1 INTRODUCTION

In any two-dimensional, steady, isentropic and irrotational compressible flow the velocity potential  $\Phi$  satisfies the differential equation

$$\left(1-\frac{u^2}{a^2}\right)\frac{\partial^2\Phi}{\partial x^2}+\left(1-\frac{w^2}{a^2}\right)\frac{\partial^2\Phi}{\partial z^2}-\frac{2uw}{a^2}\frac{\partial^2\Phi}{\partial x\partial z}=0, \qquad (2.1)$$

where  $u = \partial \Phi / \partial x$ ,  $w = \partial \Phi / \partial z$  and a is the local speed of sound. The linearized form of this equation

$$(1 - M^2) \frac{\partial^2 \Phi}{\partial x^2} + \frac{\partial^2 \Phi}{\partial z^2} = 0 \qquad (2.2)$$

follows from the simplifying assumption that terms of second order in the component perturbations of a uniform velocity U = Ma are negligible throughout the field of flow.

The problem of wall interference arises because the differential equation is subject to outer boundary conditions dependent on the working section of the tunnel. For example, the flow must be tangential to the wall of a closed tunnel and of constant speed at the boundary of an open jet. More complicated boundary conditions are used in cases of ventilated tunnels which are treated separately in Chapter VI.

There are two distinct mathematical approaches to the problem. One is to obtain solutions for  $\Phi$  for a particular model both with and without the boundary conditions imposed by the tunnel. Through neglect of viscous forces and many other approximations the potential solution that satisfies the extra boundary conditions cannot be identified with the real flow in the tunnel; large differences between these would cause major uncertainty in the interpretation of wall interference. The more realistic approach is to assume distributed doublets and vortices within the model compatible with its shape and the aerodynemic forces measured on it. The interference is then identified with the potential flow which, when added to the field of the model, satisfies the outer boundary conditions.

The lift interference is that associated with the circulation round the model. When the model is a thin aerofoil, its field can be represented by that of a distribution of vorticity along the length of the chord. The resulting potential flow within a closed tunnel thus corresponds to the infinite array of vortices partly illustrated

in Figure 2.1. The interference will then depend on the height h of the turnel, the Mach number M, the chord c and lift distribution of the aerofoil, and its incidence  $\alpha$  and location d between the walls of the tunnel. When the aerofoil is thick, its field involves a chordwise distribution of doublets as well. The lift interference and blockage corrections (Chapter V) then interact.

When the linearized differential Equation (2.2) is used, the transformation

$$X = x$$
,  $Z = z(1 - M^2)^{\frac{1}{2}} = \beta z$  (2.3)

is made, so that the potential in the new co-ordinate system satisfies Laplace's equation for incompressible flow

$$\frac{\partial^2 \Phi}{\partial x^2} + \frac{\partial^2 \Phi}{\partial z^2} = 0 \qquad (2.4)$$

It is convenient to regard the perturbation potential  $(\Phi - Ux)$  as unchanged in the transformation. Thus chord and vortex strength are invariant, while the incidence of the aerofoil and all lateral dimensions are reduced by the factor  $\beta$ . The component perturbations of velocity in the (X,Z) plane are unchanged in the longitudinal direction and increased by the factor  $\beta^{-1}$  in the lateral dimension. In accord with the linearized boundary condition the velocity ratio w/U on the aerofoil is reduced to match the inclination of the transformed surface dZ/dX =  $\beta(dz/dx)$ , so that in the (X,Z) plane the undisturbed velocity and the local upward component are respectively  $\beta^{-2}U$  and  $\beta^{-1}w$ . Thus the linearized solution is readily expressed in terms of that of an equivalent incompressible flow. The problem involves the evaluation of  $\beta^{-1}w_i$ , the vertical velocity induced by the image system of Figure 2.1 with  $\Gamma$ , h, d and  $\alpha$  replaced by  $\beta^{-2}U$ ,  $\beta$ h,  $\beta$ d and  $\beta c$  respectively.

The tunnel-induced vertical velocity  $w_i$  has to be interpreted as a number of corrections to measured quantities, such as incidence, lift and pitching moment. The incremental correction to incidence is a somewhat arbitrary average value of  $w_i/U$  radians. After this correction has been applied, the residual upwash field is converted into "residual" incremental corrections to the aerodynamic forces. The evaluation of these corrections, which are as important as the incidence correction, is greatly simplified if the residual upwash field of the image system can be expressed as a uniform streamline curvature along the length of the model. When this is too approximate, a fuller mathematical treatment is necessary; further analysis is then required to isolate the individual corrections and to relate them to the measured quantities.

Provided that the incidence and aerofoil thickness are small enough and the whole flow field is subsenic, then linearized theory is valid and the problem of wall interference is relatively simple. When, however, the incidence of the zerofoil is no longer small, it is necessary to use conformal transformation of the general equations of inviscid flow, even though the aerofoil is thin (Section 2.2). Such applications are restricted to incompressible flow, for which Equation (2.1) reduces to Laplace's equation. Likewise the treatment of a thick aerofoil between parallel walls involves conformal transformation (Section 2.3); some allowance for compressibility may still be incorporated by means of the equivalent incompressible flow. The more difficult non-linear problem of compressible flow has received little attention.

Other instances of wall corrections on two-dimensional models arise with regard to control surfaces (Section 2.4) and open-jet tunnels (Section 2.5). Problems of interference on merofolds with hinged flaps are treated either by means of the linearized equations or by a non-linear theory for incompressible flow. The linear treatment includes corrections to measured hinge moments, which may be subject to a semi-empirical factor to take account of large viscous effects. For very large flap deflections the non-linear theory shows a marked reduction, and even a reversal in sign, in the corrections to lift and pitching moment. An aerofoil spanning an open tunnel is strictly a three-dimensional configuration, since the lift falls to zero at the boundary of the jet; nevertheless, work on this problem is reviewed in Section 2.5. It is also necessary to consider the interference on an aerofoil mounted on solid side walls or large end plates for which it is usual to ignore any spanwise variation in lift distribution. The analytical solution and its physical interpretation are complicated by the finite length of open working section which is bounded by a closed entrance nozzle upstream and a collector downstream of the model.

The final Section 2.6 considers viscous effects and experimental approaches to the problems of two-dimensional wall corrections and the conclusions that may be drawn. It might be helpful to remark in advance that the boundary layers on the side walls do not have extensive influence on the pressure distribution. From their investigation Mendelsohn and Polhamus (Ref. 2.1; 1947) conclude that the loss of total load on an aerofoil spanning a closed rectangular tunnel is unlikely to exceed 1% at incidences below the stall. Figure 2.2, reproduced from Reference 2.1, shows that very close to the wall ( $y = -\frac{1}{2}b$ ) the local lift is less than 16% below that at the centre of the tunnel (y = 0), while there is little effect on pitching mement. Large changes in side-wall boundary-layer thickness are found to produce only small changes in the loading. One may therefore have confidence in a purely two-dimensional theoretical analysis.

A summary of the principal references, figures and equations and their fields of application is contained in Table 2.1.

#### 2.2 FINGS OF ZERO TRICKNESS IN CLOSED TUNNELS

## 2.2.1 Corrections to Order (c/h)<sup>2</sup>

The classical work of Glavert provides the basis on which two-dimensional wall interference is evaluated for thin aerofoils at small incidence and having moderately small chord. The lifting aerofoil can then be replaced by a single vortex of strength K at the centre of pressure to obtain the flow field induced by the walls. Batchelor (Ref. 2.2; 1944) considers the incompressible flow past a vortex situated at an arbitrary distance d from the floor of the tunnel. With allowance for compressibility (Section 2.1), the primary interference near the model then appears as a vertical velocity

$$\mathbf{w}_{1} = \frac{\beta \mathbf{x}}{2\pi} \left[ \sum_{n=-\infty}^{\infty} \frac{\mathbf{x} - \bar{\mathbf{x}}}{(\mathbf{x} - \bar{\mathbf{x}})^{2} + 4\beta^{2}(2h - d)^{2}} - 2 \sum_{n=1}^{\infty} \frac{\mathbf{x} - \bar{\mathbf{x}}}{(\mathbf{x} - \bar{\mathbf{x}})^{2} + 4\beta^{2}n^{2}h^{2}} \right] . \quad (2.5)$$

**S1** 

----

32

which, st small streamwise distance  $(x - \bar{x})$  from the vortex, becomes

$$\mathbf{w}_{\mathbf{i}} = \frac{\pi \mathbf{x} (\mathbf{x} - \bar{\mathbf{x}})}{8\beta h^2} \left(\frac{2}{3} + \infty t^2 \frac{\pi d}{h}\right) + 0(\mathbf{x} - \bar{\mathbf{x}})^3 \quad . \tag{2.6}$$

If the lift and pitching moment shout the guarter-chord axis are

$$\mathbf{L} = \rho \mathbf{U} \mathbf{K} = \frac{1}{2} \rho \mathbf{U}^2 \mathbf{C}_{\mathbf{L}}$$
, (2.7)  
$$\mathbf{R} = \frac{1}{2} \rho \mathbf{U}^2 \mathbf{C}_{\mathbf{L}}$$

then the centre of pressure is at a distance

$$\overline{\lambda} = c \left( \frac{1}{4} - \frac{C_{\rm m}}{C_{\rm L}} \right)$$
(2.8)

downstream of the loading edge. From Equations (2.6) to (2.8) the induced upwash angle in radians is

$$\frac{\mathbf{w}_{\mathbf{i}}}{\mathbf{U}} = \frac{\pi c^2}{16\beta h^2} \left\{ \left( \frac{\mathbf{x}}{c} - \frac{1}{4} \right) C_{\mathbf{L}} + C_{\mathbf{w}} \right\} \left( \frac{2}{3} + \cot^2 \frac{\pi d}{h} \right) \quad . \tag{2.9}$$

Batchelor<sup>2, 2</sup> also snows that there is an induced horizontal velocity

$$u_1 = -\frac{K}{4\beta_h} \cot \frac{\pi d}{h} = -\frac{UcC_L}{3\beta_h} \cot \frac{\pi d}{h} \qquad (2.10)$$

st the wing, which vanishes when the vortex is placed centrally  $(d = \frac{1}{2}h)$ .

The aerofoil is usually taken to be in the central plane of the tunnel. This problem is considered by Allen and Vincenti (Ref. 2.3; 1944), who represent the field of the model by continuously distributed vortices and reglect terms in  $(c/h)^4$ . To carry out the analysis, points on the aerofoil are defined by the angular co-ordinate  $\theta$  such that the distance from the leading edge

$$\mathbf{x} = \frac{1}{2}\mathbf{c}(1 - \cos\theta) \quad . \tag{2.11}$$

The vortez strength in the tunnel is expressed as

$$\mathbf{k} = 2\mathbf{i} \left\{ \mathbf{A}_{\mathbf{n}} \cot \frac{\mathbf{i}}{2} \theta + \sum_{n=1}^{\infty} \mathbf{A}_{n} \sin n \theta \right\}$$
(2.12)

per unit chordwise distance. From Equation (2,6) the vertical velocity induced by the system of targets is

10 Mar 10 Mar
$$H_{1} = \int_{0}^{C} \frac{\pi k (x - x_{0})}{12\beta h^{2}} dx$$
  
=  $\frac{\pi^{2} c^{2} U}{48\beta h^{2}} \{ (A_{0} + \frac{1}{2}A_{2}) - (2A_{0} + A_{1}) \cos \theta \}$   
=  $\frac{\pi c^{2} U}{96\beta h^{2}} \{ (C_{L} + 4C_{E}) - 2C_{L} \cos \theta \}$ , (2.13)

which varies linearly and confirms the accuracy of Equation (2.9) to order  $(c/h)^2$  in the special case  $d = \frac{1}{2}h$ .

The non-dimensional aerodynamic loading is defined as

Southers.

$$l = \frac{p_{l} - p_{u}}{\frac{1}{2}\rho u^{2}}$$
(2.14)

33

where  $p_l$  and  $p_u$  are the respective pressures on the lower and upper surfaces of the aerofoil. To the approximation of linearized theory l = 2k/U, and the upwash in Equation (2.13) is equivalent to a theoretical increment

$$l_1 = \frac{\pi}{24} \left( \frac{c}{\beta h} \right)^2 \left\{ (C_L + 4C_R) \cot \frac{1}{2}\theta + 3C_L \sin \theta \right\} , \qquad (2.15)$$

by which the loading in the tunnel would exceed that in the free stream. Alish and Vincenti argue that the data obtained in a wind tunnel should be corrected in such a way that the plack loading near the leading edge is unaltered. Accordingly

$$\Delta \alpha = \frac{\pi c^2}{96\beta h^2} (C_L + 4\tilde{C}_R)$$

$$\Delta I = -\frac{\pi}{12} \left(\frac{c}{\beta h}\right)^2 C_L \sin \theta$$
(2.16)

are applied as corrections to the measured incidence  $\alpha$  and loading l.  $\Delta \alpha$  (rad) is precisely the value of  $w_1/U$  at mid-chord ( $\theta = \frac{1}{2}\pi$ ), since the symmetrical strandline curvature equivalent to the second term of Equation (2.13) introduces no incremental singularity in the loading at the leading edge. The loading  $\Delta l$  leads to residual corrections

$$\Delta \mathbf{c}_{\mathbf{L}} = -\frac{\pi^2}{48} \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^2 \tilde{\mathbf{c}}_{\mathbf{L}}$$

$$\Delta \mathbf{c}_{\mathbf{L}} = \frac{\pi^2}{192} \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^2 \mathbf{c}_{\mathbf{L}}$$
(2.17)

With the respective increments in Equation (2.17) the measured coefficients of lift and pitching moment relate to unconstrained flow of velocity U past the actual serofoil at the corrected incidence ( $\alpha + \Delta \alpha$ ).

# 2.2.2 Corrections to Order (c/h)<sup>4</sup>

In Reference 2.4 (1934) and Reference 2.5 (1938), Tomotika uses a series of conformal transformations to obtain the lift and pitching moment on a flat-plate zerofoil inclined at an arbitrary angle  $\alpha$  to an incompressible stream between parallel walls. The midchord of the plate is assumed to lie on the centre line of the tunnel. His results as expansions in powers of e/h are

$$L = L_{0} \left\{ 1 + \frac{\pi^{2}}{24} \left( \frac{c}{h} \right)^{2} (1 + \sin^{2} \alpha) - \frac{\pi^{3}}{7690} \left( \frac{c}{h} \right)^{4} (11 - 53 \sin^{2} \alpha - 22 \sin^{4} \alpha) + 0 \left( \frac{c}{h} \right)^{6} \right\} , \qquad (2.18)$$

where  $L_n = \frac{1}{2} \partial \theta^2 \sigma (2n \sin \alpha)$ , and the pitching moment about the mid-cherd is

$$\tilde{h}_{\frac{1}{2}c} = (\tilde{h}_{\frac{1}{2}c})_{0} \left\{ 1 + \frac{\pi^{2}}{48} \left(\frac{c}{h}\right)^{2} (1 + 6 \epsilon i n^{2} \alpha) - \frac{\pi^{4}}{23040} \left(\frac{c}{h}\right)^{8} (11 - 174 \sin^{2} \alpha - 170 \sin^{4} \alpha) + 0 \left(\frac{c}{h}\right)^{6} \right\}.$$
 (2.19)

where  $(\mathbb{B}_{1c})_0 = \frac{1}{2}00^2 c^2(\frac{1}{2}\pi \sin \alpha \cos \alpha)$ . The tunnel to free-air ratios  $L/L_0$  and  $\mathbb{B}_{\frac{1}{2}c}/(\mathbb{B}_{\frac{1}{2}c})_0$  are plotted against c/h for selected values of  $\alpha$  in Figure 2.3. The broken curves for  $\alpha = 0$  and  $\alpha = 20^{\circ}$  are obtained when terms in  $(c/h)^4$  are omitted. The discrepancies are apparent when c = 0.3h and succeed  $\frac{1}{2}$ , when c = 0.5h; it should therefore be recognized that Equations (2.16) and (2.17) are subject to significant inaccuracy when c > 0.4/2h.

Havelock (Ref. 2.6; 1938) considers the same problem by treating the flat plate as the limiting case of an elliptical cylinder. He confirms Tomotika's<sup>2.4</sup> result in Equation (2.18) and derives Equation (2.19) independently. Havelock neglects terms of higher order than  $(c/h)^3$  and gives formulae for  $L/L_0$  and  $\frac{E_1}{2}c/(E_{1c})_0$  for a plate whose midpoint is at an arbitrary distance d from the floor of the tunnel (Fig. 2.1). His results can be written as

$$L = L_0 \left[ 1 - \frac{\pi c}{2h} \cot\left(\frac{\pi d}{h}\right) \sin \alpha + \frac{\pi^2 c^2}{16h^2} \left\{ \left(\frac{2}{3} + \cot^2 \frac{\pi d}{h}\right) + \left(\frac{2}{3} + 3 \cot^2 \frac{\pi d}{h}\right) \sin^2 \alpha \right\} + 0 \left(\frac{c}{h}\right)^3 \right]$$
(2.20)

and

4

$$\mathbb{E}_{\frac{1}{2}c} = \left(\mathbb{E}_{\frac{1}{2}c}\right)_{3} \left[1 - \frac{\pi c}{2h} \cot\left(\frac{\pi d}{h}\right) \sin \alpha + \frac{\pi^{2}c^{2}}{32h^{2}} \left\{\left(\frac{2}{3} + \cot^{2}\frac{\pi d}{h}\right) + \left(4 + 10 \cot^{2}\frac{\pi d}{h}\right) \sin^{2}\alpha\right\} + 0\left(\frac{c}{h}\right)^{3}\right].$$
(2.21)

and the second se

Equation (2.9) would lead to the ratios  $L/L_0$  and  $\mathbb{E}_{1,0}/(\mathbb{E}_{1,0})_0$  obtained by substituting  $\alpha = 0$  in Equations (2.20) and (2.21). Havelock's values for c = 0.2h and  $\alpha = 0$  and  $10^0$  are compared below.

d/h		0.3	0.4	0.5	D. 6	0.7
L/L <sub>o</sub>	for $\alpha = 0$	1.028	1.019	1.616	1.019	1.028
L/L <sub>o</sub>	for $\alpha = 10^{G}$	0.989	1.002	1.017	1.037	1.071
hto/(hto) o	for $\alpha = 0$	1.014	1.010	1.008	1.010	1.014
™tc/(™tc) o	for $\alpha = 10^{\circ}$	0.977	0.994	1.010	1.030	1.059

By contrast with Figure 2.3 the effect of incidence on wall interference can be of primary importance when the mid-chord is off the centre line. Moreover, a domnward displacement of an aerofoil at positive incidence can change the sign of the interference. It should also be noted that, even for c = 0.2h, the terms in  $(c/h)^{4}$  are becoming significant in the above table for  $\alpha = 10^{0}$  and d/h = 0.3 and 0.7, when the meries converges comparatively slowly.

We have already seen that Equations (2.16) and (2.17) represent a practical form of wall correction, but become inaccurate when  $c > 0.4\beta h$ ; this condition is restrictive at the high Mach numbers. The theoretical ratios  $L/L_0$  and  $\tilde{E}_{1,c}/(\tilde{E}_{1,c})_0$  from Equations (2.18) and (2.19) must then be used to obtain terms in  $(c/\beta h)^3$ . The pitching-moment coefficient about the guarter-chord axis is

$$C_{\rm m} = \frac{\hbar_{\rm Hc} - \frac{1}{4}cL\cos\alpha}{\frac{1}{2}\rho U^2 c^2}$$
  
=  $\pi \sin \alpha \cos \alpha \left\{ -\frac{\pi^2}{96} \left(\frac{c}{h}\right)^2 (1 - 4\sin^2\alpha) + \frac{\pi^4}{46060} \left(\frac{c}{h}\right)^4 (22 + 15\sin^2\alpha + 104\sin^4\alpha) \right\}.$  (2.22)

When  $\alpha$  is small,

$$C_{L} = \frac{2\pi\alpha}{\beta} \left\{ 1 + \frac{\pi^{2}}{24} \left( \frac{c}{\beta h} \right)^{2} - \frac{11\pi^{4}}{7600} \left( \frac{c}{\beta h} \right)^{4} \right\}$$

$$C_{R} = \frac{\pi\alpha}{\beta} \left\{ -\frac{\pi^{2}}{96} \left( \frac{c}{\beta h} \right)^{2} + \frac{11\pi^{4}}{23040} \left( \frac{c}{\beta h} \right)^{4} \right\}$$
(2.23)

where the linearized compressibility factor  $\beta$  has been inserted.

It remains to convert the results in Equations (2.23) into incremental corrections to the measured incidence, wift and pitching moment. The previous discussion leaves no doubt that, to order  $(c/h)^2$ , Equation (2.16) is the host definition of  $\Delta \alpha$ . Although this could still be used, we prefer the definition

$$\Delta \alpha = \beta \Im_{c} \left\{ \frac{\pi}{96} \left( \frac{c}{\beta_{\rm E}} \right)^2 - \frac{41\pi^3}{92160} \left( \frac{c}{\beta_{\rm E}} \right)^4 + \Im \left( \frac{c}{\beta_{\rm R}} \right)^6 \right\} \qquad (2.24)$$

which conforms to that of Equation (2.36) in the following Section 2.3.2. Then

$$\Delta \mathbf{C}_{\mathbf{L}} = \left(\frac{2\pi}{\beta}\right) (\alpha + \Delta \alpha) - \mathbf{C}_{\mathbf{L}} \quad .$$

which by Equations (2.23) and (2.24) becomes

$$\Delta C_{L} = C_{L} \left\{ -\frac{\pi^{2}}{48} \left( \frac{c}{\beta h} \right)^{2} + \frac{7\pi^{4}}{3072} \left( \frac{c}{\beta h} \right)^{4} + 0 \left( \frac{c}{\beta h} \right)^{6} \right\} . \qquad (2.25)$$

Since C\_ is zero in unconstrained flow, it follows from Equations (2.23) that

$$\Delta \mathbf{C}_{\mathbf{g}} = -\mathbf{C}_{\mathbf{g}} = \mathbf{C}_{\mathbf{L}} \left\{ \frac{\pi^2}{192} \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^2 - \frac{7\pi^4}{15360} \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^4 + 0 \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^6 \right\} . \tag{2.33}$$

An alternative and more practical form of Equation (2.24) is

$$\Delta \alpha :: \frac{\pi c^2}{96/\hbar^2} (C_L + 4C_R) - \frac{7\pi^3 c^4 C_L}{30720\beta^3 h^4} , \qquad (2.27)$$

and this should be used whenever  $C_{m}$  has been measured. In two-dimensional subsonic flow Equations (2.25) to (2.27) give the wall interference on a thin merofoil with its mid-chord on the centre line of a closed tunnel as incremental corrections to the measured quantities  $\alpha$ ,  $C_{\rm L}$  and  $C_{\rm m}$ . If the merofoil were displaced vertically, corresponding formulae could be derived from Reference 2.6 with neglect of  $0(c/\beta h)^5$ ; there would include the effect of a non-uniform streamwise flow that is indicated by Equation (2.10).

#### 2.2.3 Non-Exctangular Tunnels

There is no exact treatment of flat-plate aerofoils spanning olosed tunnels of non-rectangular section. The mathematical difficulties involved in the case of a circular tunnel are well discussed and treated approximately by Vincenti and Graham (Ref. 2.7: 1946). Their assumption of uniform sponwise loading on a two-dimensional model is amply justified by experiment. Even with this simplification it only appears possible to obtain the interference upwesh to order  $(c/h)^2$  at the centre of the turnel. To this approximation the first effect is that in a two-dimensional rectangular tunnel of height

$$h_{a} = 0.843 \times (dismeter of tunnel)$$
. (2.28)

A lower equivalent height would be expected, if it were possible to evaluate the average interference across the span. Nevertheless the result (2.28) should apply to pressure distributions at the centre section, which may be corrected by Equation (2.16). For tunnels of octrgonal section Satchelor<sup>2, 2</sup> assumes that the flow in sach plane normal to the span is two-dimensional, so that the interference has to be averaged across the span. His simple procedure leads to an equivalent height

$$h_e = h \left\{ 1 + \frac{b'h'}{b(h-h')} \right\}^{-\frac{1}{2}}$$
, (3.32)

where  $\frac{1}{2}b'$  and  $\frac{1}{2}b'$  are the projections of each corner fillet on the horizontal and vertical calls respectively. Equation (2.29) must seriously underestimate  $h_g$  unless the fillets are fairly coall. For balance measurements on a full-span model it may be safer to use, in place of Equations (2.28) and (2.29), an equivalent height C/b, where ( is the cross-sectional area of the tunnel. Then respectively

$$h_{e} = 0.785 \times (dismeter of tunnel)$$

$$h_{e} = h - \frac{1}{2}b'h'/b$$
(2.36)

# 2.2 VINGS OF FINITE THICKNESS IN CLOSED TURNELS

## 2.3.1 Effect of One Wall

By way of introduction it is interesting to consider a thick aerofoil in the presence of one wall. Tomotika et al. (Ref. 2.8; 1951) investigate theoretically the forces acting on certain thick cambered aerofoils at incidence in a two-dimensional incompressible flow bounded by one plane wall which can be taken to represent the ground. The exact solution for the lift is derived as an intrastable integral of an infinite series and is evaluated numerically. It can be said at once that corresponding exact solutions for flow in a straight channel are exceedingly laborious and that they defy numerical analysis if the fluid is compressible.

For positive incidences  $L/L_0$ , the ratio of lift with grownid to the unconstrained life, decreases at first from unity as c/d increases but later increases rapidly as the chord greatly exceeds the distance of the aerofoil from the ground. Then, by Figure 6 of Keference 2.8, both thickness and positive camber of the aerofoil can cause largo reductions in  $L/L_0$ , although these reductions do not appear to exceed  $\frac{145}{25}$  for the values of c/d that normally occur in problems of small interference corrections.

#### 2.3.2 Goldstein's Theory

The general two-dimensional problem of a thick, cambered aerofoil at incidence in a low-speed closed tunnel has been solved by Goldstein (Ref. 2.9; 1942) as an algebraic power series in c/h. He first derives the transformation of the aerofoil into a circle and then considers the velocity-potential field at large distances from the aerofoil in a uniform free stream. Superposition of corresponding potential fields from the infinite system of images gives a first approximation to the non-uniform flow field generoted by the walls of the tunnel. This flow is calculated in the seighbourhood of the serofoil, and hence Geldstein evaluates a modified velocity distribution on the aerofoil hy the theory described in the first paper of Reference 2.9. This provides a second approximation to the distant field of the aerofoil and hence to the nonuniform interference flow field from which the aerofynamic forces are obtained correct

1991 H - 1992

ころうちのないというできょうです。

to the order  $(c/h)^5$ . The tedicus algebra is restricted to cases in which the origin of the transformation close to the mid-chord point lies in the middle of the tunnel. Goldstein points out that, although simpler calculations. Such as those described in Section 2.2, are normally adequate for correcting lift and moment, his method allows us to decide whether this remains true in exceptional circumstances when c/h, thickness/chord ratio of the zerofoil, its camber or lift coefficient is larger than usual.

When all second-order terms in thickness, camber and incidence are neglected, the lift interference is independent of serofoil thickness, just as the blockage corrections (Chapter V) are independent of serofoil camber and incidence. Goldstein defines a correction to incidence to be the interference upwash angle at mid-chord. In the present notation Equations (76), (77) and (79) from the second paper of Reference 2.9 become

$$\Delta \alpha = \frac{\pi^2}{96} \left(\frac{c}{h}\right)^2 (2\alpha + D_2) + \frac{\pi^2}{92160} \left(\frac{c}{h}\right)^4 (-\frac{2\alpha}{3} + 20D_2 - 21D_4) \quad . \tag{2.31}$$

$$\Delta C_{L} = -\frac{\pi a_{1}C_{L}}{96} \left(\frac{c}{h}\right)^{2} + \frac{7\pi^{3}a_{1}}{30720} \left(\frac{c}{h}\right)^{2} \left\{ 3C_{L} + 2\pi(2\alpha - D_{1} + D_{2} + D_{3}) \right\} , \quad (2.32)$$

and

$$\Delta C_{\rm m} = -\frac{1}{4} \Delta C_{\rm L} + \frac{7\pi^5}{61440} \left(\frac{c}{h}\right)^4 (2\alpha + D_2) \qquad (2.33)$$

respectively. Here  $a_1$  is the two-dimensional lift slope and

$$D_{n} = \frac{4}{\pi} \int_{0}^{\pi} \frac{z_{c}}{c} \frac{\sin n\theta}{\sin \theta} d\theta , \qquad (2.34)$$

where  $z_c$  is the ordinate of the camber line of the aerofoil and  $\theta$  is defined in Equation (2.11). Coldstein shows that

$$D_2 = \frac{4}{\pi} C_{\rm p} + \frac{2}{a_1} C_{\rm L} - 2\alpha + 0 \left(\frac{c}{h}\right)^2$$
 (2.35)

;

Substitution of  $D_2$  from Equation (2.35) and  $a_1 = 2\pi$  in Equation (2.31) gives

$$\Delta \alpha = \frac{\pi}{96} \left(\frac{c}{h}\right)^2 (C_L + 4C_R) + 0 \left(\frac{c}{h}\right)^4$$

therefore, to order  $(c/h)^2$ . Equations (2.2.) to (2.27) for a flat plate scrofoll are not influenced by first-order effects of camber.

. If the aerofoil is uncambered, i.e.  $D_n = 0$ , then with the aid of the relationship

$$(\mathbf{C}_1 + \Delta \mathbf{C}_1) = \mathbf{a}_1(\alpha + \Delta \alpha) = 2\pi(\alpha + \Delta \alpha)$$

38

12

1. . . . . .

Bricks Action a three the

「商業」及正

PERSON AN MICHARMAN

Equations (2.31) to (2.33) reduce to

$$\frac{\Delta \alpha}{C_{\rm L}} = \frac{\pi}{95} \left(\frac{c}{\rm h}\right)^2 - \frac{41\pi^3}{92160} \left(\frac{c}{\rm h}\right)^4 \\ \frac{\Delta C_{\rm h}}{C_{\rm L}} = -\frac{\pi^2}{48} \left(\frac{c}{\rm h}\right)^2 + \frac{7\pi^4}{3072} \left(\frac{c}{\rm h}\right)^4 \\ \frac{\Delta C_{\rm g}}{C_{\rm L}} = \frac{\pi^2}{192} \left(\frac{c}{\rm h}\right)^2 - \frac{7\pi^4}{15360} \left(\frac{c}{\rm h}\right)^4$$
(2.38)

These expressions are consistent with Equations (2.24) to (2.26).

For a symmetrical seroioil it is straightforward to retain terms of second order in  $\alpha$ ,  $C_{L}$  and functions of the aerofoil profile

$$C_{e} = \frac{2}{\pi} \int_{0}^{\pi} \frac{z_{s}/e}{\sin \theta} d\theta$$

$$C_{n} = \frac{4}{\pi} \int_{0}^{\pi} \frac{z_{s}}{c} \frac{\cos n\theta}{\sin \theta} d\theta (n \ge 1)$$

$$(2.37)$$

where  $z_8$  is the ordinate of its upper surface. By Appendix 5 of the first paper of Reference 2.9 and Equations (69), (72) and (74) of the second paper with the appropriate theoretical lift slope  $a_1 = 2\pi(1 + C_0)$ , we have

$$\Delta \alpha = \frac{\pi}{96} \left(\frac{c}{h}\right)^2 \left\{ 2\pi \alpha (1+C_2) + C_L C_1 \right\} - \frac{\pi^3}{92163} \left(\frac{c}{h}\right)^4 \left\{ 2\pi \alpha (1-20C_0 - 9C_2 + 42C_3) + C_L (11C_1 + 31C_2) \right\} , \qquad (2.38)$$

$$\Delta C_{L} = -\frac{\pi^{2}}{35} \left(\frac{c}{h}\right)^{2} C_{L} (2 + 14C_{0} - 2C_{1} - 5C_{2}) + \\ + \frac{\pi^{2}}{92159} \left(\frac{c}{h}\right)^{4} \{2\pi\alpha \{84 + 496C_{3} - 120C_{1} - 18C_{2} + 129C_{3}\} + \\ + C_{L} (125 + 632C_{0} - 70C_{1} - 78C_{2} - 14C_{3} - 69C_{3})\}$$

$$(2.39)$$

11 A

and

$$\Delta C_{\mathbf{g}} = -\frac{1}{4} \Delta C_{\mathbf{L}} + \frac{\pi^2}{384} \left(\frac{c}{h}\right)^2 \left\{ 2\pi\alpha(-8C_0 + 4C_2) + C_{\mathbf{L}}(-3C_1 + 3C_3) \right\} + \frac{\pi^4}{363640} \left(\frac{c}{h}\right)^4 \left\{ 2\pi\alpha(42 + 212C_0 + 84C_1 + 20C_2 + 36C_4) + \frac{\pi^4}{363640} \left(\frac{c}{h}\right)^4 \left\{ 2\pi\alpha(42 + 212C_0 + 84C_1 + 20C_2 + 36C_4) + \frac{\pi^4}{363640} \left(\frac{c}{h}\right)^4 \right\}$$

$$C_{L}(156C_{1} - 79C_{3} - 35C_{5})\}$$
 (2.40)

(2.43)

To determine the theoretical relationship between  $\alpha$  and  $C_L$ , it is necessary to equate  $(C_L + \Delta C_L)$  with  $2\pi(1 + C_0)(\alpha + \Delta \alpha)$ . Thus we substitute

$$2\pi\alpha = C_L \left[ (1 - C_0) \frac{\pi^2}{96} \left( \frac{c}{h} \right)^2 (4 + 10C_0 - 3C_2) \right]$$
(2.41)

to give

$$\frac{\Delta \alpha}{C_{L}} = \frac{\pi}{96} \left(\frac{c}{h}\right)^{2} (1 - C_{0} + C_{1} + C_{2}) - \frac{\pi^{3}}{92160} \left(\frac{c}{h}\right)^{4} (41 + 79C_{0} + 11C_{1} + C_{2} + 31C_{3} + 42C_{4}) , \quad (2.42)$$

$$\frac{\Delta C_{L}}{C_{L}} = -\frac{\pi^{2}}{96} \left(\frac{c}{h}\right)^{2} (2 + 14C_{0} - 2C_{1} - 5C_{2}) + \frac{\pi^{4}}{92160} \left(\frac{c}{h}\right)^{4} (210 + 1004C_{0} - 190C_{1} - 96C_{2} + 106C_{3} - 59C_{4})$$

and

$$\frac{\Delta C_{m}}{C_{L}} = \frac{\pi^{2}}{384} \left(\frac{c}{h}\right)^{2} (2 + 6C_{0} - 5C_{1} - C_{2} + 3C_{3}) - \frac{\pi^{4}}{368640} \left(\frac{c}{h}\right)^{4} (163 + 514C_{0} - 430C_{1} + 44C_{2} + 185C_{3} - 105C_{3} + 35C_{5}) \quad (2.44)$$

It is interesting to note that Equation (2.42) again reduces to

$$\Delta \alpha = \frac{\pi}{96} \left(\frac{c}{h}\right)^2 (C_{\underline{t}} + 4C_{\underline{s}}) + 0 \left(\frac{c}{h}\right)^{\alpha}$$

since, to first order in thickness, the theoretical centre of pressure occurs at a distance  $\frac{1}{4}c(1 + C_0 - C_1 - C_2)$  behind the leading edge (Ref. 2.10).

Equations (2.42) to (2.44) have been evaluated for a 10-per-cent-thick RAE 102 aerofoil (Ref. 2.11). By Equations (27) of Reference 2.10, this merofoil may be defined approximately by the coefficients

which lead to the curves of  $2\hbar \Delta c/C_L$ ,  $\Delta C_L/C_L$  and  $\Delta C_R/C_L$  against  $(c/h)^2$  in Figure 2.4. The terms in  $(c/h)^{\pi}$  are seen to reduce the magnitude of each quantity and cannot safely be ignored unless  $c \leq 0.3h$ . The effect of thickness is especially important in the case of  $\Delta C_L/C_L$  which is increased by 50% or more. Even for a 5-per-cent-thick service,  $-\Delta C_L/C_L$  would be seriously underestimated by Equation (2.36) unless  $c \leq 0.3h$ .

Goldstein derives to order  $(c/h)^{4}$  the residual corrections to the distribution of velocity on the aerofoil; his formulae in Equation (97) of Reference 2.9 can be of importance in special investigations, when the crude alternative.  $\Delta l$  in Equation (2.16), is unlikely to suffice. The problem of a thick aerofoil placed anywhere in the tunnel is solved to order  $(c/h)^{2}$  in Reference 2.9, and the results are consistent with Equations (2.20) and (2.21) of Section 2.2. Unfortunately their usefulness is restricted by the rather slow convergence of the series in powers of c/h, when the aerofoil is far from the middle of the tunnel.

A method of conformal mapping has been applied by Moses (Ref. 2. 12; 1949) to calculate the velocity distributions on an arbitrary aerofoil at incidence in a closed wind tunnel. The method provides a satisfactory numerical solution. If it were necessary to include terms of higher order than  $(c/h)^{\pm}$ , then it would appear to be more convenient co use Reference 2.12 than to extend the analysis of Reference 2.9. As an example Moses considers a 12-per-cent-thick serofoil at  $\alpha = 4^{\circ}$  and a ratio c/h = 0.5, which provides a severe test of the formulae (2.42) and (2.43). From the ordinates of the aerofoil in Table I of Reference 2.12, we obtain, by Reference 2.10,

 $C_{0} = 0.0903$   $C_{1} = 0.057.$   $C_{2} = -0.0229$   $C_{3} = 0.00663$   $C_{4} = 0.0052s$ 

(2.46)

and a theoretical value  $(a_1)_T = 6.848$ , which gives a free-stream  $C_L = 6.848 \times 0.06981 = 0.478$  in agreement with the result in Reference 2.12. The value calculated by Moses for the aerofoil in the tunnel is  $C_L = 0.537$ , when  $\alpha = 0.06981$  radians. When these values are corrected to free-stream conditions by Equations (2.42), (2.43) and (2.46), we have

$$\begin{array}{l} C_{L} + \Delta C_{L} = 0.5922 \\ \alpha + \Delta \alpha = 0.07340 \end{array} \right\}.$$

$$a_1 = (C_1 + \Delta C_1)/(\alpha + \Delta \alpha) = 5.842$$

which is virtually a perfect check. If  $(c/h)^4$  is neglected in Equation (2.42) and (2.43), then the lift slope is over-corrected by nearly 3% to  $a_1 = 6.65$ : if, on the other hand, aerofoil thickness is ignored  $(C_n = 9)$ , then the corrected value  $a_1 = 7.01$  is about 24% too high.

# 2.3.3 Allowance for Compressibility

42

Hence

Solutions for the symmetrical compressible flow past aerofoils have yielded formulae for blockage corrections (Chapter V), but the difficult non-linear problem of a lifting aerofoil in a channel remains unsolved. Franke and Weinig (Ref. 2. 13: 1939) use the transformed linearized differential Equation (2.4) and represent the aerofoil by its velocity-potential field at large distances

$$\Phi + i\Psi = i\lambda, \log \zeta - (\lambda_2 + i\lambda_2)\zeta^{-1} , \qquad (2.47)$$

where  $\lambda_1, \lambda_2$  and  $\lambda_3$  are real and  $\zeta = X + iZ$ . The accuracy of the resulting simple formulae is comparable with that of Reference 2.3.

Since there is no practicable alternative to the evaluation of the interference upwash by linearized subscnic theory, the recommended corrections are those of Equations (2.42) to (2.44) modified in accord with the remarks in Section 2.1. We first construct the solution for the equivalent incompressible flow of density  $\rho$  and velocity  $\beta^{-2}$  with boundary conditions set by the geometrical parameters listed below. The aerodynamic quantities so obtained can then be equated as follows.

Geometrical Parameters	Aerodynamic Quantities		
Tunnel beight = ßh	Lift, Moment $= \beta^{-2}L, \beta^{-2}R$		
Aerofoil chord = c	Force coefficients = $\beta^2 C_L$ , $\beta^2 C_3$		
Aercfoil incidence = $\hat{\beta}\alpha$	Upwash velocity $= \beta^{-1} w$		
Aerofoil thickness = $\beta t$	Upwash angle = $\beta \Delta \alpha$		

where h, t,  $C_L$ ,  $C_m$ ,  $\Delta \alpha$ , etc. refer to the real compressible flow.

Tous

$$\frac{\Delta \alpha}{C_{L}} = \frac{\pi \beta}{96} \left(\frac{c}{\beta h}\right)^{2} \{1 + \beta(-C_{0} + C_{1} + C_{2})\} - \frac{\pi^{3} \beta}{92160} \left(\frac{c}{\beta h}\right)^{2} \{41 + \beta(79C_{0} + 11C_{1} + C_{2} + 31C_{3} < 42C_{4})\}$$
(2.48)

$$\frac{\Delta C_{\rm L}}{C_{\rm L}} = -\frac{\pi^2}{96} \left(\frac{c}{\beta h}\right)^2 \left\{2 + \beta (14C_0 - \mathcal{D}_1 - 5C_2)\right\}; + \frac{\pi^4}{92160} \left(\frac{c}{\beta h}\right)^4 \left\{210 + \beta (1004C_0 - 15C_1 - 5C_2 + 105C_3 - 69C_3)\right\}$$
(2.48)

NAME OF

$$\frac{\Delta C_{g}}{C_{L}} = \frac{\pi^{2}}{384} \left(\frac{c}{\beta h}\right)^{2} \left\{2 + \beta (6C_{0} - 5C_{1} - C_{2} + 3C_{3})\right\} - \frac{\pi^{4}}{368640} \left(\frac{c}{\beta h}\right)^{4} \left\{168 + \beta (514C_{0} - 430C_{1} + 44C_{2} + 155C_{3} - 105C_{2} + 35C_{5})\right\}$$

where  $C_n$  is given by Equations (2.37) or may be identified with  $A_n$  in Equations (27) of Reference 2.10. In Figure 2.4 with abscissa relabelled  $(c/\beta_n)^2$  the appropriate curves will give  $2\pi\Delta\alpha/(\beta c_L)$ ,  $\Delta c_L/c_L$ ,  $\Delta c_a/c_L$  for a typical symmetrical aerofoil of thickness  $t = 0.1c/\beta$ . The implicit assumption, that  $e_1 = 2\pi(\beta^2 + C_0)$ , sust be recognized. In the absence of \_ more exact theory of two-dimensional wall interference there is some justification in that this lift slope will normally lie below the exact theoretical value and above the experimental one.

Then Equations (2.16) and (2.17) will not suffice, the wall corrections for a thick cambered aerofoil should be obtained in two parts. If the measured lift coefficient is written as

$$C_{L}(\alpha) = C_{L}(0) + [C_{L}(\alpha) - C_{L}(0)]$$

then the contribution to interference from  $C_L(0)$  is found by setting  $\alpha = 0$  in Equations (2.31) to (2.33) as modified in accord with the above table. Thus

$$\Delta_{\alpha} = \frac{\pi^{2}}{96} \left(\frac{c}{\beta h}\right)^{2} D_{2} + \frac{\pi^{4}}{92160} \left(\frac{c}{\beta h}\right)^{4} (20D_{2} - 21D_{4})$$

$$\Delta C_{L} = -\frac{\pi^{2}}{48} \left(\frac{c}{\beta h}\right)^{2} C_{L}(0) + \frac{7\pi^{4}}{15360} \left(\frac{c}{\beta h}\right)^{4} \left\{ 3C_{L}(0) + \frac{2\pi}{\beta} (-D_{1} + D_{2} + D_{3}) \right\}$$

$$\Delta C_{H} = -\frac{1}{4} \Delta C_{L} + \frac{7\pi^{5}}{611440} \left(\frac{c}{\beta h}\right)^{4} \frac{D_{2}}{\beta}$$
(2.49)

where  $D_n$  is given by Equation (2.34). To Equations (2.49) are added contributions from Equations (2.48) with the substitution  $C_L = [C_L(\alpha) - C_L(0)]$ . It may be necessary to reduce these corrections on account of the aerofcil boundary-layer; a simple semi-expirical procedure for this is discussed in Section 2.6.1.

**£**3

#### 2.4 AEBOFOILS WITH BINGED FLAPS

44

Sections 2.2 and 2.3 have been concerned with lifting aerofoils without fisps; they apply to aerofoils with undeflected flaps, but an additional formula is needed for the residual correction  $\Delta C_{\rm g}$  to the hinge-moment coefficient. The interference on a thin ayrmetrical percfoil with deflected trailing-edge flap spanning a closed two-dimensional tunnel has been obtained to order  $(c/h)^3$  by Preston and Manwell (Ref. 2. 14; 1941). Theoretical values of the non-dimensional derivatives of lift, quarter-chord pitching moment and hinge moment with respect to incidence  $\alpha$  and flap setting  $\xi$ 

$$\mathbf{a}_{1} = \partial \mathbf{c}_{L} / \partial \alpha , \qquad \mathbf{a}_{2} = \partial \mathbf{c}_{L} / \partial \xi ,$$
$$\mathbf{a}_{1} = \partial \mathbf{c}_{H} / \partial \alpha , \qquad \mathbf{a}_{2} = \partial \mathbf{c}_{H} / \partial \xi ,$$
$$\mathbf{b}_{1} = \partial \mathbf{c}_{H} / \partial \alpha \qquad \text{and} \qquad \mathbf{b}_{2} = \partial \mathbf{c}_{H} / \partial \xi ,$$

are given in the respective Equations (38), (39), (43a), (44a), (50) and (51) of Reference 2.14. The theory takes no account of finite thickness, and it is verified that the expressions for  $a_1$  and  $a_1$  are consistent with the full curves in Figure 2.3 for  $\alpha = 0$ . Ratios of the other derivatives to their free-stream values,  $b_1/(b_1)_0$ ,  $a_2/(a_2)_0$ ,  $a_2/(a_2)_0$  and  $b_2/(b_2)_0$  are plotted in Figure 2.5 as functions of  $c/(\beta n)$ and  $B = c_f/c$ , where  $c_f$  is the chord of the flap. When c = 0.5/2n and B = 0.3, the inverted ratios are as follows.

Tunnel derivative	<sup>2</sup> 1	b <sub>1</sub>	8 <u>.</u>	72 <sub>2</sub>	b <sub>2</sub>
Correction factor	0.914	0.837	0.939	0.917	0.930

The correction factors to  $a_2, a_2$  and  $b_2$  decrease as E increases and are of the same order as that to  $a_1$ . The interference corrections to  $b_1$  are corporatively large and such less dependent on E.

In conformity with Sections 2.2 and 2.3, we express the results of Reference 2.14 as theoretical incremental corrections to incidence and the aerodynamic coefficients. The corrected hinge moment on the undeflected flap is  $\frac{1}{2}OU^2c_g^2(C_H + \Delta C_H)$  with

$$C_{H} + \Delta C_{H} = b_{1} \alpha + \Delta C_{H} = (b_{1})_{\alpha} (\alpha + \Delta \alpha) , \qquad (2.50)$$

where  $b_1$  and  $(b_1)_0$  in Equation (50) of Reference 2.14 require the usual modification for compressibility, and  $\Delta \alpha$  is defined in Equation (2.24) for a flat-plate zerofoll. Substituting

$$x = \frac{\beta C_{\rm L}}{2\pi} \left\{ 1 - \frac{\pi^2}{24} \left( \frac{c}{\beta h} \right)^2 + 0 \left( \frac{c}{\beta h} \right)^4 \right\}$$
(2.51)

from Equation (2.23), py obtain

$$\frac{4\mathbf{E}^{2}(\Delta \mathbf{C}_{\mathrm{P}})}{\mathbf{C}_{\mathrm{L}}} = \frac{\pi}{288} \left(\frac{\mathbf{c}}{\beta \mathbf{h}}\right)^{2} \left\{12(\pi - \theta_{1})\cos\theta_{1} + 9\sin\theta_{1} + \sin 3\theta_{1}\right\} - \frac{7\pi^{3}}{921609} \left(\frac{\mathbf{c}}{\beta \mathbf{h}}\right)^{8} \left\{-60(\pi - \theta_{1}) + 600(\pi - \theta_{1})\cos\theta_{1} \div 520\sin\theta_{1} - 20\sin^{2}\theta_{1} + 25\sin^{2}\theta_{1} - 5\sin^{2}\theta_{1} + \sin^{2}\theta_{1}\right\}, \quad (2.52)$$

where  $E = c_f/c = \frac{1}{2}(1 + \cos\theta_i)$ . In Figure 2.6(a) curves of  $\Delta c_H/C_L$  against  $c/\beta h$  are drash for several fixed values of E. There is no reason to suppose that the effect of zerofoil thickness on these curves is any greater than on  $\Delta C_H/C_L$  in Figure 2.4. However, the terms in  $(c/\beta h)^6$  may become especially important in the case of  $\Delta c_H/C_L$ ; it appears likely that for  $c > 0.4\beta h$  these higher-order terms are nore significent than thickness effect.

For the compressible flow past a thin aerofoil at raro ...cidence with deflected flap, we write

$$C_{L} + \Delta C_{L} = \mathbf{s}_{2} \boldsymbol{\xi} + \Delta C_{L} = (\mathbf{s}_{2})_{\alpha} \boldsymbol{\xi} + (\mathbf{s}_{1})_{\alpha} \Delta \boldsymbol{\alpha}$$

$$C_{R} + \Delta C_{R} = \mathbf{s}_{2} \boldsymbol{\xi} + \Delta C_{R} = (\mathbf{s}_{2})_{\alpha} \boldsymbol{\xi} + (\mathbf{s}_{1})_{\alpha} \Delta \boldsymbol{\alpha}$$

$$C_{R} + \Delta C_{R} = \mathbf{b}_{\alpha} \boldsymbol{\xi} + \Delta C_{R} = (\mathbf{b}_{2})_{\alpha} \boldsymbol{\xi} + (\mathbf{b}_{1})_{\alpha} \Delta \boldsymbol{\alpha}$$

$$(2.53)$$

45

where  $(a_1)_0 = 2\pi\beta^2$ ,  $(a_1)_0 = 0$ ,  $(b_1)_0$ ,  $(a_2)_0$ ,  $(a_2)_0$  and  $(b_2)_0$  are  $\beta^{-1}$  times their values in Reference 2.14 for M = 0. The correction  $\Delta z$  to incidence is chosen from Equation (2.27), whence

$$\frac{2\pi\Delta\alpha}{\beta c_{\perp}} = \frac{\pi^2}{48} \left(\frac{c}{\beta_{\rm il}}\right)^2 \left\{1 + \frac{4a_2}{a_2}\right\} - \frac{7\pi^8}{15350} \left(\frac{c}{\beta_{\rm il}}\right)^4 \qquad (2.34)$$

By Equations (2.53) and (2.54)

$$\frac{\Delta C_{L}}{C_{L}} = \frac{1}{a_{2}} \{(a_{2})_{0} - a_{2}\} + \frac{2\pi\Delta\alpha}{\beta C_{L}}$$

$$\frac{\Delta C_{C}}{C_{L}} = \frac{1}{a_{2}} \{(a_{2})_{0} - a_{2}\} + \frac{2\pi\Delta\alpha}{\beta C_{L}}$$

$$(2.55)$$

$$\frac{\Delta C_{H}}{C_{L}} = \frac{1}{a_{2}} \{(b_{2})_{0} - b_{2}\} + (b_{1})_{0} \frac{\Delta\alpha}{C_{L}}$$

These corrections are plotted as functions of  $c/\beta h$  and E in Figures 2.6(b) and 2.7;  $\Delta c/C_L$  falls sharply as E decreases, and to order  $(c/h)^2$  it tends to zero as  $E \rightarrow 0$ . To this order  $\Delta C_L/C_L$  and  $\Delta C_R/C_L$  are independent of E and  $\Delta C_R/C_L$ is given by Equation (2.52); for large enough  $c/\beta h$ , however, all these corrections for a deflected flop are numerically larger than the corresponding values in Figures 2.4 and 2.5(a) for the lift interference due to zerofoil incidence. In the case of a symmetrical servicil at incidence with deflected flap,  $\Delta \alpha$  should first be evaluated directly from Equation (2.27). Then the residual corrections  $\Delta C_L$ ,  $\Delta C_R$  and  $\Delta C_R$  should be estimated both from Figures 2.4 and 2.6(a) and from Figures 2.6(b) and 2.7. Whenever these differ appreciably, a mean value should be weighted according to the contributions to  $C_L$  from zerofoil incidence and flap setting. According to the contributions to  $C_L$  from zerofoil incidence and flap setting. According to the contributions to  $C_L$  from zerofoil incidence and flap setting. According to the treated separately as in Equation (2.49). Semi-empirical correction factors to  $\Delta C_R$  will normally be required to take account of the boundary layer (Section 2.6.1).

The results in Figures 2.6 and 2.7 only apply to unbalanced two-dimensional flaps. To order  $(c/h)^2$ , Miss Lyon (Ref. 2.15; 1942) has extended the analysis to flaps with sealed abroaded balance on the supposition that the chordwise pressure distribution is that of an unbalanced flap of increased chord. The hinge moments are further modified, since the binge is now set back from the nose of the flap. If required, the results for balanced flaps could be derived to order  $(c/h)^4$  with the aid of Reference 2.14.

De Jager and van de Vooren (Ref. 2. 16; 1961) have considered the non-linear problem of a hinged plate between parallel walls in incompressible flow. The forward portion of the plate lies along the centre of the channel, but the rear portion or flap is deflected through large angles  $\xi$  so that the vorticity may no longer be assumed to lie entirely in the central plane. Numerical solutions with six terms to represent the vorticity are obtained for a wide range of  $\xi$  when E = 0.2, 0.25, 0.3 and c/h = 0.2, 0.3, 0.4. Because of the non-linearity, the results cannot be presented as in Equations (2.53) to (2.55) with a correction  $\Delta \alpha$ , nor can any reliable allowance be made for compressibility. The corrections  $\Delta C_L/C_L$  from Reference 2.16, reproduced in Figure 2.8, are therefore not comparable with those in Figure 2.7. The most striking foature of the results is the rapid decrease in the correction as  $\xi$  increases above  $30^\circ$ ; there is indeed a reversal in the sign of the correction near  $\xi = 81^\circ$ ,  $70^\circ$  and  $58^\circ$  for E = 0.2, 0.25 and 0.3 respectively. With the aid of the graphical data in Figures 2, 3, 5 and 8 of Reference 2.16 it is simple to calculate the ratio

$$\frac{\Delta C_{\mathbf{n}}}{C_{\mathbf{n}}} = \frac{\Delta C_{\mathbf{n}}}{C_{\mathbf{n}}} \frac{(C_{\mathbf{n}})_{6} \{1 + (\Delta C_{\mathbf{n}}/C_{\mathbf{n}})\}}{(C_{\mathbf{n}})_{6} \{1 + (\Delta C_{\mathbf{n}}/C_{\mathbf{n}})\}}$$

to obtain the curves in Figure 2.9 for B = 0.25. In this case, since  $(B_1)_0 = 0$ , the results for small  $\xi$  are seen to be consistent with the corresponding Equation (2.55) and the curve in Figure 2.7 that is practically independent of B. For each value of c/h the correction is halved near  $\xi = 40^\circ$  and vanishes near  $\xi = 67^\circ$ . It would be interesting to have experimental confirmation.

# 2.5 OPEN-JET TUNNELS

#### 2.5.1 Roundary Conditions

At one time it was common practice to test two-dimensional serofoils spanning an open jet; such models could be supported conveniently outside the stream. The problem of jet-boundary interference has to be considered from several points of view. The flow is far from two-dimensional, since the lift falls to zero where the merofoil crosses the boundary; the wing therefore experiences a large downwash due to lift interference and also some lift-dependent drag which is absent in purely two-dimensional

flow. In a flow of finite cross-section with free upper and lower boundaries, a finite lift on the aerofoil induces a finite downwash in the distant wake and a consequent interference at the model. Both of these considerations lead to large corrections  $\Delta \alpha$  proportional to the chord of the aerofoil. Furthermore, the flow is accompanied by some distortion as well as the deflection of the jet, and these effects have never been incorporated in the boundary conditions. Therefore the interference corrections are not only larger but less soundly based than those for closed tunnels.

The boundary condition to be satisfied at the edge of a jet is that the pressure has a constant value equal to that of the surrounding air. Although linear velocity perturbations are hard to justify where the aerofoil emerges from the jet, this assumption is made; it then follows from Bernouilli's equation and the conditions of constant pressure, that  $u = \bar{u}$  over the entire boundary of the open jet. If it is further assumed that the jet is infinitely long and undistorted by the model, then on the boundary the velocity potential is  $\bar{\Psi} = Ux$  and the perturbations in velocity have zero tangential components.

## 2.5.2 Fully Open Jets

We first consider a lifting aerofoll spanning an incompressible jet of breadth  $b(-\frac{1}{2}b \leq y \leq \frac{1}{2}b)$  and infinite height. Stüper (Ref. 2. 17; 1932) has treated this problem as an application of the classical lifting-line theory to a wing of infinite span at incidence  $\alpha(y) = (-1)^{R}$ , where N is the integer nearest to y/b. This unit function of periodically changing sign is expanded as a Fourier series

$$x(y) = \frac{4}{\pi} \sum_{n=0}^{\infty} \frac{(-1)^n}{2c+1} \cos\left\{\frac{\pi y}{b} (2n+1)\right\} .$$
 (2.56)

Hence, for  $|y| \leq \frac{1}{2}b$ , the lift per unit span is  $\rho \overline{u}(y)$  with

$$\Gamma(y) = \frac{1}{2}a_{1} \operatorname{cc}\left[1 - \frac{4}{\pi} \sum_{n=0}^{\infty} \frac{(-1)^{n}}{2n \div 1 \div \lambda} \cos\left\{\frac{\pi y}{b} (2n \div 1)\right\}\right] , \qquad (2.57)$$

where a, is the two-dimensional lift slope and

$$\lambda = 3b/(\pi a_1 c) = 4b/(\pi^2 c)$$
 when  $a_1 = 2\pi$ .

The total lift

$$L = \int_{-\frac{1}{2}b}^{\frac{1}{2}b} \rho G'(y) \, dy$$
  
-  $L_0 \left[ 1 - \frac{2}{\pi} \int_{-1}^{1} \sum_{n=1}^{\infty} \frac{(-1)^n}{2n+1+\lambda} \cos \left\{ \frac{1}{2} \pi \eta (2n+1) \right\} \, d\eta \right]$ 

(2, 58)

i

47

and all of algebrands dated determines where is not as the still been able to the determine the still all the still and the still all the still a

ś

& Broken a hande after the second

$$= L_0 \left[ 1 - \frac{8}{\tau^2} \sum_{n=0}^{\infty} \frac{1}{(2n+1)(2n+1+\lambda)} \right].$$
 (2.58)

This is easily summed when  $\lambda$  is a positive integer; in general

$$\mathbf{L} = \mathbf{L}_{0} \left[ 1 - \frac{4}{\pi^{2} \lambda} \left\{ \{0.5772 \cdots + 2\frac{1}{2}(1+\lambda) - \frac{1}{2}(1+\frac{1}{2}\lambda)\} \right\} \right].$$
 (2.59)

where 0.5772... is Euler's constant and values of the psi function  $\Psi(\mathbf{x})$  will be found in Reference 2.18. The boundary interference will depend on whether the experimental data are taken from measurements of total force or from pressure plotting at the centre section. In the forcer case  $L/L_0$  is given by Equation (2.59) and is plotted in Figure 2.10(a) scainst  $c/b = 4/(\pi^2 \lambda)$ ; even for a small wing of chord c = 0.1b the correction factor  $L_0/L$  is as high as 1.37.

Stüper<sup>2.17</sup> deals with open rectangular tunnels for which he obtains as a generalization of Scation (2.57),

$$\frac{1}{\Gamma_0} = \frac{4}{\pi} \sum_{n=0}^{\infty} \frac{(-1)^n \cos\{(2n+1)\pi y/b\}}{(2n+1)[1+P_0/k]} \quad (2.63)$$

mere

$$\Gamma_{0} = \frac{1}{2}a_{1}Uc$$

$$F_{n} = (2n + 1) \operatorname{coth} \{(2n + 1)\pi h/2h\}$$

He provides evidence from experimental pressure distributions for  $h/b = \frac{1}{2}$  and  $\hat{i}$  in support of the spanwise loadings predicted by Digition (3.60). Although there is less interference on the lift at the centre section that we the sotal lift, the curves of  $\hat{\Gamma}(0)/\hat{\Gamma}_{,i}$  against c/b in Figure 2.10(c) show that the correction factors for a wing of chord 0.1b are  $\hat{\Gamma}_0/\hat{\Gamma}(0) = 1.18$ , 1.20 and 1.55 for  $h = \infty$ , b and  $\frac{1}{2}b$  respectively.

Another configuration that has received considerable attention is a rectangular wing spanning a circular open jet. In Reference 2.17 there appear to be errors on the lefthand sides of the final equations for the coefficients of the spanwise logling. Horeover, Squire (Ref. 2.19; 1039) points out that Stüper's form of solution misrepresents the mathematical singularity in downwash at the wing tips. Sprive gives a more exact treatment of the lifting-line problem with a distribution of lift

$$\frac{\Gamma(\mathbf{y})}{\Gamma_{0}} = \frac{8\pi R}{\frac{1}{2}c} \left[ \frac{2}{\pi^{2}} \left\{ 2\log_{e} 2 - \left(1 - \frac{y}{R}\right)\log_{e} \left(1 - \frac{y}{R}\right) - \left(1 - \frac{y}{R}\right)\log_{e} \left(1 + \frac{y}{R}\right) \right\} - \frac{\sum_{n=0}^{\infty} \frac{b_{n}}{2n+2} \left\{ 1 - \left(\frac{y}{R}\right)^{2n+2} \right\} \right].$$
(2.61)

where R is the radius of the jet. Equation (2.61) yields

$$\frac{L}{L_0} = \frac{8\pi R}{s_1 c} \left[ \frac{2}{\pi^2} - \sum_{n=0}^{\infty} \frac{b_n}{2n+3} \right], \qquad (2.62)$$

where  $b_n (n = 0, 1 \text{ and } 2)$  are obtained approximately by collocation. The numerical results are plotted in Figure 2.11(s) and are not greatly different from the values of  $\Gamma(0)/\Gamma_0$  and  $L/L_0$  for a square jet.

As a consequence of the non-uniform spanwise loading across the jet, the wing experiences lift-dependent drag which is absent in purely two-dimensional flow. Quite small errors in the spanwise loading are significant in the evaluation of this drag; it is not surprising, therefore, that Stüper's values are appreciably greater than those from Reference 2.19 which are used in Figure 2.11(b). Here the corrections to incidence and drag are given by

$$\frac{\Delta \alpha}{C_{\rm L}} = -\frac{1}{a_{\rm i}} \left( \frac{{\rm L}_0}{{\rm L}} - 1 \right)$$
(2.63)

and by  $\Delta C_D/C_L^2$ , where  $\Delta C_D$  is equal and of opposite sign to the theoretical drag coefficient of Reference 2.19 with  $a_1 = 2\pi$ ; it is easily shown that the results are not sensitive to the choice of  $a_1$ .

The treatment by lifting-line theory ignores the effects of induced curvature of flow, which will be of the same order as the corrections for closed tunnels in Figure 2.4. By comparison with the positive values of  $\Delta a/C_L$  for closed tunnels, the quantity defined by Equation (2.63) in conjunction with either Figure 2.10(a) or Figure 2.11(a) is negative and of such higher magnitude. The likely accuracy of the interference corrections for open tunnels does not warrant allowance for induced curvature. Similarly there is little justification for applying lifting-surface theory, although Reference 2.20 (Rethorst, 1958) could be adapted for the purpose.

#### 2.5.3 Two-Dimensional Tests

It is now supposed that the breadth of the tunnel is very large, or that with the aid of solid side-walls or end plates a two-dimensional flow is preserved. On the simple regument, that the interference downwash at the wing is half that in the distant wake, we have

$$\Delta \alpha = \frac{1}{2} \frac{\pi_{+\infty}}{U} = -\frac{1}{4} \frac{S}{C} C_{L} , \qquad (2.64)$$

since from considerations of momentum the lift force  $\frac{1}{2}OU^2SC_L$  balances the rate of growth of downward momentum -  $\rho UCw_{+\infty}$ . An exact solution of the incompressible problem by Sasaki (Ref. 2. 21; 1928), quoted in Glauert's monograph, has given numerical values of  $L/L_0$  for a flat place placed midway between the boundaries of the jet at on incidence  $\alpha = 10^{\circ}$ . These are plotted against c/h in Figure 2. 10(b), and for c = 0.2h the correction factor  $L_0/L$  has reached 1.34. From consideration of Equation (2.64) alone we would have

 $\frac{L_0}{L} = 1 - \frac{2\pi(\Delta \alpha)}{C_c} - 1 + \frac{\pi S}{2C} = 1 + \frac{\pi c}{2h} = 1.31 ,$ (2.65)

which is a fair approximation to the accurate theory.

We now suppose that the lifting aerovoil is of small chord and can be replaced by a single vortex at an arbitrary position between the upper and lower boundaries of the jet. Corresponding to Equations (2.6) and (2.10) for a closed tunnel, the induced upward and streamwise velocities for a tunnel with open roof and floor are

$$\mathbf{w}_{i} = \frac{\pi \mathbf{k} (\mathbf{x} - \bar{\mathbf{x}})}{8\beta h^{2}} \left(\frac{4}{3} + \cot^{2} \frac{\pi d}{h}\right)$$
(2.66)

and

$$v_1 = \frac{K}{4\beta h} \cot \frac{\pi d}{h} = \frac{V c C_L}{8\beta h} \cot \frac{\pi d}{h} . \qquad (2.67)$$

Equations (2.56) and (2.67) arise from an image system comprising a column of vortices, each of the same sense; these contribute a finite upwash  $w_{-\infty}/U = +\frac{1}{4}cC_L/h$  upstream of the model and an equal and opposite downwash  $w_{+\infty}/U = -\frac{1}{4}cC_L/h$  in the distant wake. In order to eliminate the undesired upstream upwash, a uniform downwash can be added to the solution in Equation (2.66) without violating the condition that the tangential velocity is unperturbed at the boundary of the jet. Hence, by Equations (2.7) and (2.8), the induced upwash angle in the neighbourhood of the model is

$$\frac{\Psi_{i}}{\Im} = -\frac{1}{4} \frac{c}{h} C_{L} - \frac{\pi c^{2}}{16\beta h^{2}} \left\{ \left( \frac{x}{c} - \frac{1}{4} \right) C_{L} + C_{R} \right\} \left( \frac{4}{3} + \cot^{2} \frac{\pi d}{h} \right) \quad . \tag{2.68}$$

Glauert has shown that this result with  $\beta = 1$  and d = 0 is in very good agreement with the exact solution of Reference 2.21.

Vandrey (Ref. 2. 22; 1942) has considered a scall model in a two-dimensional open tunnel with a closed entrance mozzle, and he gives a physical discussion and mathematical solution of this mixed-boundary problem. The model is represented by a combined wortex and source, and the interference velocities are expressed in terms of lift and drag. The lift interference of Reference 2.22 can be regarded as a special case of the work of Gardner and Diesendruck in the second part of Reference 2.33 (1950). However far downstream the model may be, the closed entrance nozzle takes care of the upstream flow condition. One configuration of tunnel in Reference 2.23 has a closed entrance nozzle and a single lower exit lip. i.e. a roofless collector; for a lifting model midway between the roof and floor, the induced velocities at the model are

$$u_{1} = -\frac{K}{4\beta h} e^{\pi X_{0}/\beta h} P$$

$$\pi_{1} = -\frac{X}{4h} \left[ e^{\pi I/\beta h} P - 2(1 + e^{\beta \pi X_{0}/\beta h})^{-1} \right]$$
(2.69)

Let construction of the and substantian construction with the second statement of the second s

AP12-2010 5

where

$$P = \frac{(e^{2\pi i/\beta h} + e^{2\pi x_0/\beta h})^{\frac{1}{2}} + e^{\pi i/\beta h}}{e^{2\pi i/\beta h} + e^{2\pi x_0/\beta h}}$$

ţ

STATE DON

 $x_0$  and l are the respective distances of the model and the exit lip downstream of the entrance nozzle. As  $l \to \infty$ ,  $u_i \to 0$  and

$$\mathbf{x}_{i} \rightarrow -\frac{K}{2h} \left(1 + e^{-2\pi X_{0}/\hat{\rho}h}\right)^{-1}$$
, (2.70)

which is consistent with Reference 2.22.

Of greater practical interest is the two-dimensional closed-open-closed tunnel with an open working section of length l and a closed entrance nozzle and collector. In Reference 2.23 the interference velocity is derived for an arbitrarily placed vortex in such a tunnel. The streamwise component  $u_i$  venishes when the vortex is in the horizontal plane of symmetry, and

$$w_{1} = -\frac{K}{2h} \left[ G_{0} + \frac{\pi(x-\bar{x})}{\beta h} G_{1} + O(x-\bar{x})^{2} \right] , \qquad (2.71)$$

where

$$G_0 = \frac{(e^{2\pi l/\beta h} - 1)e^{2\pi x_0/\beta h}}{(1 + e^{2\pi x_0/\beta h})(e^{2\pi l/\beta h} + e^{2\pi x_0/\beta h})}$$
(2.72)

$$G_{1} = -\frac{1}{6} + \frac{(e^{2\pi l/\beta h} - 1)e^{2\pi x_{0}/\beta h}}{(1 + e^{2\pi x_{0}/\beta h})^{2}(e^{2\pi l/\beta h} + e^{2\pi x_{0}/\beta h})} \times \left\{ e^{2\pi l/\beta h}(1 + \frac{1}{2}e^{2\pi x_{0}/\beta h}) - e^{2\pi x_{0}/\beta h}(\frac{1}{2} + e^{2\pi x_{0}/\beta h}) \right\}$$
(2.73)

and  $(x - \bar{x})$  denotes the distance downstream of the vortex. We substitute

$$\vec{x} = \frac{1}{2} UCC_{L}$$
$$x - \bar{x} = \frac{1}{4} c + cC_{R}/C_{L}$$

to obtain the upwash angle at mid-chord

$$\Delta \alpha = -\frac{cc_{L}}{4h} \left[ G_{0} + \frac{\pi c}{\beta h} \left( \frac{1}{4} + \frac{c_{a}}{c_{L}} \right) G_{1} \right] \qquad (2.74)$$

ł

Then the residual corrections to lift and moment are given by

$$\frac{\Delta C_{\rm L}}{C_{\rm L}} = -\frac{\pi c}{2\beta u} \frac{\partial \mathbf{x}_{\rm I}}{\partial \mathbf{x}} = \frac{\pi^2 G_{\rm I}}{8} \left(\frac{c}{\beta b}\right)^2 \left\{ \frac{\Delta C_{\rm R}}{C_{\rm L}} = \frac{\pi c}{8\beta u} \frac{\partial \mathbf{x}_{\rm I}}{\partial \mathbf{x}} = -\frac{\pi^2 G_{\rm I}}{32} \left(\frac{c}{\beta b}\right)^2 \right\} , \qquad (2.75)$$

where  $G_1$  is defined in Equation (2.73). If  $x_0 \to \pm \infty$ ,  $G_1 \to 0$  and  $G_1 \to -1/6$  and we have the results for a closed tunnel in Equations (2.16) and (2.17). If  $l \to \infty$  and then  $x_0 \to \infty$ ,  $G_0 \to 1$  and  $G_1 \to \pm 1/3$ , so that for an open tunnel the residual corrections in Equation (2.75) are of twice the magnitude and of the opposite signs. The functions  $G_0$  and  $G_1$  are plotted in Figure 2.12. In a typical case l = 1.5/3h,  $x_0 = 0.5/3h$ ,  $G_0 = 0.96$  and  $G_1 = 0.33$ ; such a closed-open-closed tunnel has interference characteriptics very similar to those of a completely free jet.

## 2.5.4 Spillage Behind the Model

In the first part of Reference 2.23, Katzoff remarks that considerable uncertainty exists with regard to conditions at the exit and the mathematical equivalent of these conditions. Accordingly, certain compromises are justified when idealized downstream conditions are used in a determination of boundary interference. In the example  $l = 1.5\beta h$ ,  $x_0 = 0.5\beta h$  quoted above, it is found from integration of the vertical component of velocity at either free surface, that the downward displacement of the jet at the collector is  $1.20\beta K/U = 0.60\beta c C_L$ . The physical implications of this have been ignored; we may suppose, however, that the errors in the interference as given in Equations (2.74) and (2.75) would become serious, were the downward displacement of the jet to exceed 0.1 $\beta h$ . Thus tests on an aerofoil of chord  $c = 0.2\beta h$  would need to be restricted to incidences below 8<sup>0</sup>.

Katzoff also points out that the velocities at the two free boundaries need not be equal. If the space below the lower boundary is sealed off, the pressure at this free surface will adjust itself so that the jet attaches smoothly at the lower lip of the collector. By modifying the boundary conditions so that there are horizontal perturbation velocities  $\pm u'$  on the upper free boundary and -u' on the lower one, increments to the interference velocities are obtained in Reference 2.23. The downward components of induced velocity along the central axis and the free boundaries are given by simple functions of elliptic integrals. In the numerical example with  $l = 1.5\beta_h$ and  $x_c = 0.5\beta_h$ , the incremental upward displacement of the jet is  $3.89\beta^2u'h/U$ ; the original downward displacement  $1.20\beta_K/U$  is exactly cancelled if  $u' = 0.31K/(\beta_h)$ . Now if the interference near the model is expressed in the form of Equation (2.71), the increment due to the unequal velocities at the free boundaries is

$$f_1 = 1.44\beta u' + 1.98u'(x - x_0)/h$$
  
= {0.45 + 0.61(x - x\_0)/\beta h}K/h . (2.76)

In other words, the increments to  $G_0 = 0.96$  and  $G_1 = 0.33$  are -0.90 and -0.39 respectively. This demonstrates theoretically that lift interference may be nearly eliminated if spillage at the collector is prevented by enclosing the space into which the spillage would normally occur.

On account of the large and unreliable interference corrections the use of a completely open tunnel for accurate testing of two-dimensional models is not recommended. The likely inaccuracies in the corrections predicted by Figures 2.10 and 2.11 set too low a limit on the chord of the model. If, however, two-dimensional flow is preserved by means of solid side walls or large end plates, then the chief uncertainty is the condition at the collector downstream of the jet. The linearized theoretical results in Equations (2.74) and (2.75) with Figure 2.12 ignore any effect of the jet ueflection

at the collector, though Reference 2.23 permits the evaluation of the vertical displacement of the jet. A condition that this should not exceed 0.1h, say, might restrict the tests to low incidences. Finally, there is the theoretical merit of a two-dimensional closed-open-closed tunnel with straight unflared collector and with self-adjusting unequal pressures at the free surfaces to prevent spillage.

# 2.6 EXPERIMENTAL CONSIDERATIONS

## 2.6.1 Viscous Effects

Unless there are extensive regions of separated flow, the theoretical interference corrections to lift and pitching moment can easily he modified to take some account of viscous effects; the correction to incidence would remain unchanged. From the semi-empirical reasoning in Section 4.2 of Reference 2.24 (Bryant and Garner, 1951), the incremental lift and pitching moment due to streamline curvature are influenced by boundary layers in roughly the same ratio as the lift slope. Viscous effects on two-dimensional wall interference can therefore be incorporated by applying to  $\Delta C_L/C_L$  and  $\Delta C_m/C_L$  the approximate correction factor

$$\mu_{a} = \frac{1}{(a_{1})_{T}} \{ \text{Corrected experimental } \partial C_{L} / \partial \alpha \} , \qquad (2.77)$$

where  $(a_1)_T = 2\pi(\beta^{-1} + C_0)$  with  $C_0 = 0$  unless aerofoil thickness has been taken into account (Section 2.3). The author (Ref. 2.25; 1957) has shown that this factor will become important at low Reynolds numbers, especially if the trailing-edge angle is large.

In the case of hinge moments, however, viscous effects will always be important and cause considerable uncertainty in the residual corrections  $\Delta C_{\rm H}/C_{\rm L}$ , as is discussed in detail in Reference 2.26 (1950). There is limited evidence that the hinge moment due to induced curvature of flow is influenced by boundary layers in the same ratio as the derivative  $b_1$ . It is advisable therefore to apply to  $\Delta C_{\rm H}/C_{\rm L}$  the correction factor

$$\mu_{\rm b} = \frac{1}{(b_1)_{\rm T}} \{ \text{Corrected experimental} \ \partial C_{\rm H} / \partial \alpha \} , \qquad (2.78)$$

where, with  $\cos \theta_1 = 2E - 1$ ,

$$4\beta E^{2}(b_{1})_{T} = -2(\pi - \theta_{1})(1 - 2\cos\theta_{1}) + 4\sin\theta_{1} - \sin 2\theta_{1}$$

This factor may well reduce  $\Delta C_{\mu}/C_{L}$  to less than one half of its value in Figure 2.6.

As discussed in Section 2.1, Mendelsohn and Polhamus<sup>2.1</sup> have shown experimentally that side-wall boundary layers can be ignored in problems of two-dimensional interference correction. Further evidence to this effect is given by Vincenti and Graham in Figures 6 and 7 of Reference 2.7. However, Barbieux (Ref. 2.27; 1955) has suggested that a reduced height of tunnel should be taken to allow for the displacement thickness of boundary layers on the roof and flocr; such a correction would only become important if the ratio c/h were unusually large. -----

# 2.6.2 Closed Tunnels

Barbieux<sup>2.27</sup> has provided important experimental data on wall corrections in incompressible flow. An aerofoil (p. 137 of Reference 2.27) was pressure plotted in a two-dimensional closed tunnel whose height was varied systematically in the range  $c \leq h \leq 3c$ . He applied an original theory of interference correction, which is described in Chapter I of the second part of Reference 2.27. This differs from Goldstein's theory in that the model is represented more simply by the combination of a streamwise doublet at its centroid, a finite vortex at its aerodynamic centre and a transverse doublet corresponding to the pitching moment at zero lift, each of which is assumed to lie midway between the walls of the tunnel. The resulting formulae, given as power series up to  $(c/h)^6$ , are summarized in Tables IX to XII of Reference 2.27. The lift interference is expressed as an incremental correction to incidence which is chosen such that there is a residual correction to pitching moment but not to lift. By following the sequence of Figures 62, 70, 76 and 79 (legend in Figure 82), it can be seen that Barbieux's results collapse on to a unique lift curve against incidence within about ± 2%; his fully corrected experimental pitching-moment curve in Figure 80 shows some small systematic residual effect of c/h . The results are remarkably good in view of the extensive range of c/h.

Nevertheless, the formulation of Goldstein's theory in Section 2.3 is preferable, since this involves a rigorous treatment of the first-order effects of aerofoil profile and a rigorous representation of the interference flow field to the order  $(c/h)^4$ . The resulting corrections in Equations (2.48) will now be considered in the light of viscous effects (Section 2.6.1) and an experimental investigation of Knechtel (Ref. 2. 28; 1953) of two-dimensional wall interference in subsonic compressible flow on an NACA 4412 aerofoil. By varying the effective height of a closed tunnel, four ratios c/h in the range  $0.119 \le c/h \le 0.95$  were obtained fithout change of Reynolds number. The two highest values c/h = 0.357 and 0.595 were obtained by using respectively two and four image aerofoils so as to simulate tunnels of 1/3 and 1/5 the true height. Tests were also carried out on the same model in a large two-dimensional open tunnel with c/h = 0.026 and negligible wall interference. The main conclusion from Figures 2(a) and 2(b) of Reference 2.28 is that wall corrections by the method of Allen and Vincenti<sup>2.3</sup> give satisfactory comparisons provided that c < 0.15h; for larger values of c/h, wall interference becomes progressively greater and results corrected by this method become increasingly questionable. The evidence on lift coefficient at  $\alpha = 4^{\circ}$  is reproduced in Figures 2.13(a) and 2.13(b) for Mach numbers in the range  $0.3 \leq M \leq 0.8$ .

We now suppose that accurate blockage corrections and approximate lift interference from Equations (2.16) and (2.17)

$$\Delta \alpha = \frac{\pi c^2}{96\beta h^2} (C_L + 4C_g)$$

$$\Delta C_L = -\frac{\pi^2}{48} \left(\frac{c}{\beta h}\right)^2 C_L$$
(2.79)

have been applied in Figure 2.13(b). It will be seen that the curve for c/h = 0.595 lies roughly 7% below the others. Equations (2.48) with the viscous correction factor  $\mu_{a}$  from Equation (2.77) give more comprehensive corrections than Equation (2.79).

To estimate the effect of aerofoil thickness, we use the values of  $C_n$  from Equation (2.45) with a generalizing factor 10t/c to obtain the numerical formulae

$$\Delta \alpha = \frac{\pi c^2 C_L}{96\beta h^2} \left\{ 1 - 0.3397 \frac{\beta t}{c} \right\} - \frac{\pi^3 c^4 C_L}{92160\beta^3 h^4} \left\{ 41 + 67.33 \frac{\beta t}{c} \right\}$$
$$\Delta c_L = \mu_a C_L \left[ -\frac{\pi^2}{48} \left( \frac{c}{\beta h} \right)^2 \left\{ 1 + 5.253 \frac{\beta t}{c} \right\} + \frac{\pi^2}{92160} \left( \frac{c}{\beta h} \right)^4 \left\{ 210 + 689.4 \frac{\beta t}{c} \right\} \right] \right\}. \quad (7 \ 80)$$

We have seen in Section 2.3 that to order  $(c/h)^2$  Equations (2.79) and (2.80) for  $\Delta \alpha$  are identical, and we shall regard the term in  $(c/h)^{\alpha}$  in Equation (2.80) as an increment to the correction  $\Delta \alpha$  given in Equation (2.79). This increment must be converted to an equivalent lift coefficient by means of the factor  $-a_1$ , since we wish to compare results at a fixed incidence  $\alpha = 4^{\circ}$ . Thus we apply as a further correction to  $C_L$ , as given in Figure 2.13(b), the increment

$$(\Delta C_{\rm L})' = \frac{a_{\rm I} \pi^3 c^{\rm u} C_{\rm L}}{92160\beta^3 h^{\rm u}} \left\{ 41 + 67.33 \frac{\beta t}{c} \right\} + (1 - \mu_{\rm a}) \frac{\pi^2}{48} \left( \frac{c}{\beta h} \right)^2 C_{\rm L} + \mu_{\rm a} C_{\rm L} \left[ -\frac{\pi^2}{48} \left( \frac{c}{\beta h} \right)^2 \left\{ 5.253 \frac{\beta t}{c} \right\} + \frac{\pi^{\rm u}}{92160} \left( \frac{c}{\beta h} \right)^{\rm u} \left\{ 210 + 689.4 \frac{\beta t}{c} \right\} \right], \quad (2.81)$$

where

$$\mu_{\mathbf{a}} = \frac{\mathbf{a}_{1}}{2\pi(\beta^{-1} + C_{0})} = \frac{\beta \mathbf{a}_{1}}{2\pi} \left\{ \mathbf{1} \div \mathbf{0}.773 \frac{\beta \mathbf{t}}{\mathbf{c}} \right\}^{-1}$$

and a, is estimated from experiment.

The four terms of Equation (2.81) may be described as

- (i) the equivalent of the  $(c/h)^4$  term in  $\Delta \alpha$
- (ii) the viscous correction to  $\Delta C_{L}$  in Equation (2.79)
- (iii) the thickness correction of order  $(c/h)^2$
- (iv) the  $(c/h)^{4}$  term in  $\Delta C_{L}$  in Equation (2.80)

For the thickness to chord ratio t/c = 0.12 the correction factor  $1 + (\Delta C_L)'/C_L$  has been evaluated with c/h = 0.119, 0.156, 0.357, 0.595 and M = 0.30, 0.45, 0.60, 0.70, 0.80; typical results are given in the following table, where the separate contributions (i) to (iv) often tend to cancel.

888 ay

		μ <sub>ε</sub>	Contributions to ( $ riangle C_L$ )'/CL				$1 + \frac{(\Delta C_L)'}{2}$
			(i)	(ii)	(iii)	(iv)	сĽ
0. 156	0.60	0.827	0.000	0.001	- 0.003	0.000	0.998
0.357 0.357 0.357	0.30 0.60 0.80	0.840 0.827 0.413	0.002 0.004 0.005	0.005 0.007 0.043	- 0.015 - 0.017 - 0.011	0.005 0.010 0.014	0.997 1.004 1.051
0.595 0.595 0.595	0.30 0.60 0.80	0.840 0.827 0.413	0.014 0.027 0.041	0.013 0.020 0.119	- 0.04C - 0.047 - 0.032	0.039 0.074 0.110	1.026 1.074 1.238

The results so corrected are plotted in Figure 2.13(c), where, for the four values of c/h and Mach numbers in the range  $0.3 \le M \le 0.7$ , the discrepancies appear to be random and never exceed  $\pm 2\%$ . Although the results for the interference-free case c/h = 0.026 are up to 4% higher, this discrepancy can be attributed to effects of extraneous flow in the large open-jet tunnel with side-walls installed<sup>2.28</sup>.

It may be inferred from Figure 2.13 that effects of shock-induced separation on the NACA 4412 aerofoil at  $\alpha = 4^{\circ}$  are felt at Mach numbers above 0.6. The theory of wall interference is based on the velocity-potential field at large distances from the aerofoil where the linearized equations continue to give a good approximation to the flow, and so to  $\Delta \alpha$ . Although the residual corrections  $\Delta C_L$  and  $\Delta C_m$  are less certain, the evaluation of wall interference remains justified until M approaches 0.8.

There can be general confidence in methods of applying two-dimensional interference corrections for tests in subsonic closed rectangular tunnels. The method of Reference 2.3 will normally be adequate for correcting lift and pitching moment; Equations (2.79) to (2.81) indicate some of the inaccuracies in such an approximation. When these are too large, Goldstein's theory<sup>2.9</sup>, as formulated in Section 2.3, is recommended.

#### 2.6.3 Open-Jet Tunnels

An experimental investigation by Adamson (Ref. 2.29; 1941) includes tests of two rectangular wings of different chord spanning an open-jet circular tunnel. The measurements of lift and drag have been corrected according to Squire's theory<sup>2.19</sup> as given here in Figure 2.11(b). These corrections are too small in magnitude on account of the approximate lifting-line theory. It is found that the corrected two-dimensional lift slopes  $a_1$  from Figure 1 of Reference 2.29 are rather lower than would be predicted by Equation (22) of Reference 2.25. The following table shows that the inaccuracy in lifting-line theory is of the correct order of magnitude and sign to account for the discrepancies in  $a_1$ .

Aerofoil .:	2R	Reynolds number	Value of <sup>8</sup> 1		Estimated inaccuracy	
	c		Ref. 2.25	Ref. 2.29	a <sub>1</sub> (Ref. 2.29)	ll. theory
RAF 34 RAF 48	7.5 5.0	$0.50 \times 10^{6}$ $0.75 \times 10^{6}$	5.05 5.29	4.95 4.88	- 2% - 8%	+ 4% <sup>.</sup> + 7%

Nevertheless it would be unwise to expect a result to better accuracy than 5% ficm such experiments, whatever care were taken.

# REFERENCES

2.1	Mendelsohr, R.A. Polhemus, J.F.	Effect of the Tunnel-Wall Boundary Layer on Test Results of a Wing Protruding from a Tunnel Wall. NACA Tech. Note 1244, 1947.
2. 2	Batchelor, G.K.	Interference on Wings, Bodies and Airscrews in a Closed Tunnel of Octagonal Section. Report ACA-5 (Australia), 1944.
2.3	Allen, H.J. Vincenti, W.G.	Wall Interference in a Two-Dimensional-Flow Wind Tunnel, with Consideration of the Effect of Compressibility. NACA Report 782, 1944.
2.4	Tomotika, S.	The Lift of a Flat Plate Placed in a Stream between Two Parallel Walls and Some Allied Problems. Aero. Rem. Inst. (Tokyo) Report 101, 1934.
2. 5	Tomotika, S.	The Moment of the Fluid Pressures Acting on a Flat Plate in a Stream between Two Parallel Walls and Some Allied Problems. Aero. Res. Inst. (Tokyo) Report 170, 1938.
2.6	Havelock, T.H.	The Lift and Moment on a Flat Plate in a Stream of Finite Width. Proc. Roy. Soc., Series A, Vol.166, 1938, pp.178-196.
2.7	Vincenti, W.G. Grahan, D.J.	The Effect of Wall Interference upon the Aerodynamic Characteristics of an Airfoil Spanning a Closed-Throat Circular Wind-Turnel. NACA Report 349, 1946.
2.8	Tomotika, S. et al.	The Forces Acting on an Aerofoil of Approximate Joukowski Type in a Stream Bounded by a Plane Wall. Quart. J. Nech. appl. Math., Vol.4, 1951, pp. 289-307.

The second second second

and the second

012

57

tiger the

•

فكرزه

58	
2.9 Goldstein, S.	Two-Dimensional Wind-Tunnel Interference. ARC R & ¥ 1902, 1942.
2.10 Garmer, N.C.	Simple Evaluation of the Theoretical Lift Slope and Aerodynamic Centre of Symmetrical Aerofoils. ARC R & M 2847, 1951.
2.11 Pankhurst, R.C. Squire, M.B.	Calculated Pressure Distributions for the RAE 100 to 104 Aerofuil Sections. ARC CP 80, 1950.
2.12 Moses, H.E.	Velocity Distributions on Arbitrary Airfoils in Closed Tunnels by Conformal Happing. NACA Tech. Note 1899, 1949.
2.13 Franke, A. Weinig, P.	Die Korrektur der Anströmgeschwindigkeit und des Anströmwinkels in einem Hochgeschwindigkeitskanal mit geschlossener Meßstrecke infolge der Verdrängungsströmung in Tragflügelmodellen. Deutsche Luftfahrtforschung Forschungsbericht 1171, 1939. (Translated as MAP R & T 255. ARC Report 10,268.)
2.14 Preston, J.H. Manwell, A.R.	Calculation of the Interference on a Thin Symmetrical Aerofoil with Hinged Flap Spanning a Closed Wind-Tunnel. ARC R & M 2465, 1941.
2.15 Lyon, H.M.	An Approximate Method of Calculating the Effect of the Interference of a Closed Tunnel on Hinge Moments of Control Surfaces. ARC Report 5803, 1942.
2.16 de Jager, E.M. van de Vooren,	Tunnel-Wall Corrections for a Wing-Flap System between A.I. Two Parallel Walls. NLR (Netherlands) Report W.7, 1961.
2.17 Stüper, J.	Der durch einem Freistrahl hindurchgesteckte Tragflügel. Ing. Arch. Vol.3, 1932, pp.338-355. (Translated as NACA Tech. Memo.723.)
2.18 Davis, H.T.	Tables of the Higher Mathematical Functions. Volume I. The Principal Press Inc., Indiana, USA, 1933.
2.19 Squire, H.B.	The Lift and Drag of a Rectangular Wing Spanning a Free Jet of Circular Section. Phil. Mag., Vol.27, 1939, pp.229-239.
2.20 Rethorst, S.	Aerodynamics of Non-Uniform Flows as Related to an Airfoil Extending through a Circular Jet. J. aero. Sci., Vol.25, 1958, pp.11-28.
2.21 Sasaki, T.	; On the Effect of the Wall of a Wind Tunnel upon the Lift Coefficient of a Model. Aero. Res. Inst. (Tokyo) Report 46, 1928.

;

1

1

and a second set of the second se

1

and a subsection of the subsec

L'unit

.

. .

2.22 Vandrey, F.	Der Düseneinfluss auf die Windkanalkorrekturen bei ebener Strömung. Jahrbuch der Deutschen Luftfahrtforschung, 1942, pp.1766-1793.
2.23 Katzoff, S. et al.	Linear Theory of Bowndary Effects in Open Wind Tunnels with Finite Jet Lengths. NACA Report 976, 1950.
2.24 Bryant, L.W. Garner, H.C.	Control Testing in Wind Tunnels. ARC R & M 2881, 1950.
2.25 Garner, H.C.	Charts for Low-Speed Characteristics of Two-Dimensional Trailing-Edge Flaps. ARC R & M 3174, 1957.
2.26 Garner, H.C.	Note on Aerodynamic Camber. ARC R & M 2820, 1950.
2.27 Barbieŭx, J.	Contribution a l'Étude de l'Effet de Paroi en Écoulement Plan Incompressible. Publ. Sci. Tech. Min. Air France 304, 1955.
2.28 Knechtel, E.D.	Experimental Investigation of Two-Dimensional Tunnel-Wall Interference at High Subsonic Speeds. NACA Tech. Note 3087, 1953.
2.29 Adamson, J.E.	An Experimental Investigation of Wind-Tunnel Interference in the RAE 5 ft Open-Jet Circular Tunnel. ARC R & M 1897, 1941.

# ADDITIONAL REFERENCES FOR CHAPTER II

Inai, I.	On the Deformation of Free Boundary Due to Line Vortices. Aero. Res. Inst. (Tokyo) Report 183, 1939.
de Jager, E.M. van Spicgel, E.	Calculated Tunnel-Wall Corrections jor Two-Dimensional High-Lift Wings. MLL (Netherlands) Report WP.181, 1959.
Katzoff, S. et al.	Interference Method for Obtaining the Potential Flow Past an Arbitrary Cascade of Airfoils. NACA Tech. Note 1252, 1947.
Pankhurst, R.C. Pearcey, H.H.	Camber Derivatives and Two-Dimensional Tunnel Interference at Maximum Lift. ARC CP 28, 1950.
Preston, J.H.	Note on the Application of Interference Corrections for a Large-Chord Wing with Control Spanning a Closed Tunnel. ARC Report 5802, 1942. (Unpublished.)

ALL ALL

100 CAL

59

200

-2

i

Sanders, J.

Stüper, J.

Wijker, H.

£

ñ. 2

Wall Corrections for a Two-Dimensional Electrical Tank. NRL (Canada) Report MA-213, 1948.

Beitrag zum Problem des durch einen Freistrahl hindurchgesteckten Tragflügels. Luftfahrtforschung, Vol.12, 1935, pp.267-281. (Translated as NACA Tech. Memo., 796.)

Experiments on the Two-Dimensional Flow over a NACA 0018 Profile at High Angles of Attack. Appendix II: Influence of Tunnel Walls on the Pressure Distribution. NLL (Netherlands) Report A. 1263, 1953.

TABLE 2	2. ]	Ι
---------	------	---

ē

Sumary	ΟĨ	Principal	References,	Figures	and	Formul	lae
--------	----	-----------	-------------	---------	-----	--------	-----

		Velocity u <sub>i</sub> or w <sub>i</sub>		Γ. Lor M in tunnel		$\begin{array}{c} \text{Correction} \\ \Delta \alpha \text{ or } \Delta C_L \end{array}$		$\begin{array}{c} \textit{Correction} \\ \Delta C_{\underline{m}} \text{ or } \Delta C_{\underline{H}} \end{array}$	
Subject	Ref.	Fig.	Eqn.	Fig.	Eqn.	Fig.	Eqn.	Fig.	Eqn.
Thin aerofoil in closed tunnel	2.3 2.4 2.5 2.6		2.9 2.10	2.3	2. 18 2. 19 2. 20 2. 21		2. 16 2. 17 2. 24 2. 25 2. 27		2. 17 2. 26
Thick aerofoil in closed tunnel	2.9 2.12 2.27 2.28			2. 13		2. 4	2.31 2.32 2.48 2.49 2.81	2.4	2. 33 2. 48 2. 49
Aerofoil with flap in closed tunnel	2. 14 2. 15 2. 16 2. 24			2.5		2.7 2.8	2. 54 2. 55	2.6 2.7 2.9	2. 52 2. 55
Open-jet tunnel	2. 17 2. 21 2. 22 2. 23	2. 12	2.67 2.68 2.69 2.71	2. 10	2. 60		2. 74 2. 75		2.15
Non- rectangular tunnel	2. 2 2. 7 2. 19			2.11	2. 61 2. 62	2.11			

61

7

ł

1

1

ī

1

10 ----



A COMPANY AND A COMPANY AND A COMPANY



Reproduced from Fig. 6 of Ref. 2.1



63

्र



64

题早 ~~

23-448-575

Contravier,



:

1.

States Dentes

١.

Fig. 2.4 Wall corrections for an RAE 102 serofoil

65

and the second second

1. . . .

1

32.5



Fig. 2. 5 Closed-tunnel to free-stream ratios of derivatives for aerofulls with high (Ref. 2.14)

•

66

W

i

No.1 Y

200

and and and a second

Ε (a) Lift interference due to incidence. Eqn. (2.52) 0.016 0.5 0.4 0.014 0.3 0.012 0-2 0.010 ACH /CL 0.008 0 · 1 0.006 0.004 0.002 0 (b) Lift interference due to flap setting. Eqn.(2.55) Ε 0-018 0.5 0-4 0.016 Q·3 0.014 0·2 0.012 0.010 ACH/CL Q · I 0.008 3.006 0-004 0.002 Û 0.3 0-2 0.1 0.5 0 0.4 c/ßh

÷

A STATE AND A STAT

. . . . . . . . .

Pig. 2.6 Wall corrections to flap hinge moments

and fis

Salar Succession

< <u>.</u>

: ·



Fig. 2.7 Wall corrections due to flap setting
Based on Ret. 2.16

and the standing of the state o

ł



Pig.2.8 Effect of large flap setting on lift interference (M = 0)



- ~



Alexandre and a second

24.5

(a) h= $\infty$ . Equation (2.59) (b)  $b = \infty$ . Ref. 2.21 1.0 0.9 0.8  $\frac{L}{L_0}$  0.7 0.6 0.5 0-4 0-2 0.5 0.3 0-3 0 0.1 0.4 0.1 0.4 05 c/h c/b Equation (2.60) (c) Centre section. 1.0 0.9 0.8 (<u>-</u> 0.7

ł



Fig. 2, 16 Lift interference in rectangular open-jet tunnels

71

5-. \*

1

,

ŧ

1. N. N. 1



;



• -



ł



ъ.

73

.....





A CONTRACT OF A CONTRACT OF

The second second second

## CHAPTER III

# LIFT INTERFERENCE ON THREE-DIMENSIONAL WINGS

by

H.C.Garner

Aerodynamics Division, National Physical Laboratory. Teddington, Middlesex, England

75

- -

-

ch.

-----

. . .

- -

\$



CONTEN/S

÷

ł

ł

LIST	F OF TABLES	79
LIST	C OF FIGURES	79
NOTA	ATION	81
3.1	INTRODUCTION	87
3.2	SHALL WINGS IN CLOSED AND OPEN TUNNELS	90
	3.2.1 Nethod of Images	91
	3.2.2 Rectangular Tunnels	92
	3.2.3 Non-Rectangular Tannels	97
	3.2.4 Applications	106
3.3	GENERAL THEORY OF LIFT INTERFEBENCE	103
	3.3.1 Representation of Model	103
	3.3.2 Basic Interference Parameters	105
	3.3.3 Corrections to Measured Quantities	108
	3.3.4 Complete Spanwise Symmetry	119
	3.3.5 Non-Symmetrical Spanwise Loading	112
	3.3.6 Asymmetrically Placed Wings	115
3,4	EVALUATION OF INTERFERENCE PARAMETERS	117
	3.4.1 Rectangular Turnels	118
	3.4.2 Octagonal Tunnels	125
	3.4.3 Circular and Elliptical Tunnels	139
	3.4.4 Half-Wing Nodels	138
3.5	NUMERICAL INTERFERENCE CORRECTIONS	141
	3.5.1 Rectangular Tunnels	143
	3.5.2 Non-Rectangular Tannels	147
	3.5.3 Comparisons of Nethods	149
3.6	SPECIAL CONFIGURATIONS	153
	3.6.1 Slender Wings	154
	3.6.2 Wing-Body-Tail Combinations	157
	3.6.3 Interference on Sround Effect	163
	3.6.4 Hodels Spanning Closed Tunnels	163
3.7	EXPERIMENTAL CONSIDERATIONS	167
	3.7.1 Wigh Lift	168
	3.7.2 Closed Tunnels	170
	3.7.3 Open Tunnels	171

77

Page

in the second

L

÷.

1

٠.

: :

3.8 INDEX	OF FORMULAE AND DATA	173
3.8.1	Simple or Approximate Interference	
	Velocity	174
3.8.2	Elaborate or Accurate Interference Upwash	175
3.8.3	Corrections to Measured Quantities	176
REFERENCES		177
ADDITIONAL REFERENCES		182
TABLES		183
FIGURES		194

# LIST OF TABLES

ł

÷ ...

Page	

79

**Sine** 

•

where is a specifical of a second of a second

10

-----

La Pasta La

، مة، تركل :

TABLE 3. I	Equivalent Incompressible Flow from Equation (3.5) when $M = (1 - \beta^2)^{\frac{1}{2}}$	183
TABLE 3.II	Function $S_1$ from Equation (3.18) for the Evaluation of Streamline Curvature in Rectangular Tunnels	184
TABLE 3.III	Values of $\lambda$ and $\lambda x_1/\overline{c}$ from Lifting-Surface Theory and Strip Theory	185
TABLE 3. IV	Summary of Numerical Data on Interference Upwash for Large Wings	186
TABLE 3.V	Interference Parameters for a Closed Square Tunnel	187
TABLE 3.VI	Interference Parameters $\zeta_0$ and $\zeta_1$ for Closed Rectangular Tunnels	188
TABLE 3.VII	Comparative Calculations of $\Delta \alpha / C_L$ and $\Delta C_{g} / C_L$ for Wings in Closed Rectangular Tunnels	189

# LIST OF FIGURES

<b>Fig. 3.</b> 1	Typical loage systems	190
Fig.3.2	Lift interference on small wings in rectangular tunnels	191
Fig.3.3	Upwash and stream-velocity interference on a small lifting wing displaced vertically from the centre of a closed rectangular tunnel	192
Fig. 3.4	Ratios $\delta_1/\delta_c$ for small models in closed and open rectangular tunnels	193
Fig. 3.5	Lift interference on small wings in elliptical tunnels	19 <u>4</u>
Fig.3.5	Axial distributions of interference upwash for several positions of a small lifting wing in a closed-open-closed circular tunnel	195
Fig. 3.7	Lift-interference correction to measured incidence of small models of various planforms	196
F15.3.8	Different vortex models of lifting wings	197
Fig. 3.9	Lift interforence on wings with elliptic spanwise loading in closed rectangular tunnels	198

Fig. 3. 10	Approximate representations of doubly infinite arrays of doublets to evaluate $W_A$ for rectangular tunnels	199
Fig. 3. 11	Lift interference on off-centre models in a closed duplex tunnel with and without ground simulation	200
Fig. 3. 12	Lift interference on wings with elliptic spanwise loading in closed octagonal tunnels	201
Pig. 3. 13	Interference parameter for elliptically loaded wings in elliptical tunnels	202
Pig. 3. 14	Vortex configurations defining $\Phi_{\bf R}$ , $\Phi_1^{(2)}$ and $\Phi_1^{(2)}$ for circular tunnels	203
Fig. 3. 15	Lift interference on uniformly lowied half-wings in closed and open circular tunnels with variable reflection planes	204
Fig. 3. 16	Approximate interference corrections to incidence for wings of various spans and aspect ratios in three closed rectangular tunnels ( $M = 0$ )	205
<b>Fig. 3. 17</b>	Interference corrections for wings of high aspect ratio in a closed rectangular tunnel ( $b/h = 10/7$ )	206
Fig. 3. 18	Approximate curves of $\delta$ against h/h for wings of different area ratios in closed rectangular tunnels	207
<b>Pig. 3. 19</b>	Comparative calculations of residual corrections to pitching moment on wings in closed rectangular tunnels for a range of Mach number	268
Fig. 3. 20	Lift interference on slender delta wings in a closed duplex tunnel	209
Fig. 3. 21	Interference correction to tailplane incidence on models of varying size in a closed duplex tunnel $(M = 0)$	210
Fig. 3. 22	Effect of wing span on lift interference with ground simulation in three closed rectangular tunnels	211
Fig. 3. 23	Lift interference on ground effect on unifermly londed wings in three closed rectangular tunnels	212
Pig. 3. 24	Experiments on lift interference on models spanning closed tunnels	213
Fig. 3. 25	Lift curves for an elliptical wing in two closed circular tunnels	214
Fig. 3. 26	Comparison of aileron effectiveness of a complete model and a part-span model in a closed circular tunnel with reflection plane	215
Fig. 3. 27	Experimental determination of lift interference in two open tunnels	216
Fig. 3. 28	Interforence corrections to drag on three similar delta wings in an open circular tunnel	217

.

80

٠.

- .

.

127

Fage

and the second

# NOTATION

:

a	speed of sound
2	length of corner fillet of octagonal tunnel
A	aspect ratio of wing = $2s/\overline{c}$
b	breadth of tunnel
B	hinge moment parameter in Equation (3.77)
c	chord of wing
ē	geometric mean chord of wing = S/2s
Ē	aerodynamic mean chord of wing in Equation (3.60)
° <sub>f</sub>	chord of control surface
°r	root chord of wing
° <sub>t</sub>	tip chord of wing
C	cross-sectional area of tunnel
C <sub>D</sub>	drag coefficient = $D/\frac{1}{2}\rho U^2 S$
C <sub>H</sub>	hinge moment coefficient
c,	rolling moment coefficient = $L/\rho 0^2 Ss$
C <sub>L</sub>	lift coefficient = $L/\frac{1}{2}\rho U^2 S$
C <sub>LL</sub>	(lift per unit span) $\frac{1}{2}\rho U^2 c$
C <sub>m</sub>	pitching moment coefficient = $\hbar/\frac{1}{2}\rho U^2 S\overline{c}$
C <sub>mL</sub>	(pitching moment per unit span) $\frac{1}{2}\rho U^2 c^2$
C <sub>a</sub>	yawing moment coefficient = $N/\rho U^2 Se$
C <sub>p</sub>	pressure coefficient
đ	distance of wing from floor of tunnel
D	drag
E <sub>1</sub>	complete elliptic integral

81

~ #

7

سيد

1

Ľζ,

82 ſ function defined in Equation (3.177) or (3.178) functions (q = 1, 2, 3, 4) defined in Equations (3.90)1<sub>a</sub> integral defined in Equation (3.177) P function defined above Equation (3, 121) Ī, G function defined in Equation (3.193) height of tunnel h modified Bessel function of the first kind I, sign according to the appropriate Equation (3.14) S Bessel function (n = 1, 2, ...)Jn X strength of vortex strength of equivalent horse-shoe vortex in Equations (3.41) K, modified Bessel function of the second kind (n = 1, 2)K, 1 length of open working section non-dimensional wing loading =  $(p_1 - p_n)^{\frac{1}{2}} \rho U^2$ 1 l semi-focal distance of elliptical tunnel L lift L(x) lift ahead of position x C rolling moment integer defining column of images y = mb (Fig. 3.1) Mach number of undisturbed stream ¥ b. nose-up pitching moment outward normal distance from tunnel boundary n integer defining row of images z = nh (Fig. 3.1) n n yawing moment . pressure on upper or lower surface of wing D strength of doublet P

Statistics.

:

;

3

## 

83

STP 1

A Constant of the second

1.

and a state of the second of t

. .

z

r	radial distance
R	radius of circular tunnel
R <sub>N</sub>	Reynolds number
\$	semi-span of wing
s(I)	local semi-span of wing
<sup>S</sup> e	semi-span of equivalent horse-shoe vortex in Equations (3.41)
S	area of planform of wing
s <sub>q</sub>	functions $(q = 1, 2, 3)$ in Equations (3.18), (3.19), (3.26)
t	thickness of wing
t	semi-span of elementary horse-shoe vortex
u	x-component of velocity perturbation
U	velocity of undisturbed stream
٧	y-component of velocity
۳ <sub>0</sub>	velocity component of image system in Figure 3.1(b) normal to fillet
W	z-component of velocity
w <sub>i</sub>	interference upwash velocity
¥	weight factor in Equation (3.55)
W	upwash function relating to horse-shoe worter in Equation (3.79)
W <sub>A</sub>	upwash function in Equations (3.96)
Wa	upsash function in Equations (3.96)
¥ <sub>c</sub>	upwash function in Equation (3.97)
x	streamwise distance
x	local centre of pressure
x <sub>o</sub>	distance of model from entrance nozzle of open-jet turnel
x <sub>1</sub>	centre of lift defined below Equation (3.60)
x <sub>f.</sub>	ordinate of leading edge

a 🗲

84	
Ī	integral defined in Equation (3.60)
x <sub>t</sub>	distance between three-quarter-chord points of wing and tail surfaces
x	x in transformed plane
J	spanwise distance from centre of tunnel
<b>y</b> 1	inboard end of aileron
<b>y</b> <sub>2</sub>	outboard end of aileron
¥	transformed co-ordinate = $\beta y$
Z	upward distance from centre of tunnel
Z	transformed co-ordinate = $\beta z$
α	incidence of wing (in radians unless otherwise stated)
α <sub>è</sub>	incidence of tailplane
β =	$(1 - M^2)^{\frac{1}{2}}$
γ	non-dimensional circulation = $\Gamma/2sU$
Г	circulation
δ	lift interference parameter in Equation (3.150)
۶'	incremental upwash interference in Equation (3.176)
δ <sub>o</sub>	upwash interference at a lifting line
δ	upwash interference associated with streamline curvature
\$ A	additional upwash interference downstream of wing
δ <sub>D</sub>	drag interference parameter in Equation (3.215)
δ <sub>w</sub>	upwash interference parameter at wing
δ <sub>α</sub>	incidence interference parameter in Equation (3.215)
ε	non-dimensional correction to stream velocity
ζο	interference parameter in Equations (3.156)
ζ	interference parameter in Equations (3.156)
η.ζ	elliptical co-ordinates in Equation (3.113)

17.1

202

,

> رم م

.

η <sub>0</sub> =	$tanh^{-1}(h/b)$ for elliptical tunnel (b > h)	
θ	angular cylindrical polar co-ordinate	
ĸo	antisymmetrical upwash interference at a lifting line	
ĸ <sub>1</sub>	effect of streamline curvature in Equation (3.49)	
λ	planform parameter defined below Equation (3.60)	
Λ	angle of sweepback	
۸ <sub>L</sub>	sweepback of leading edge	
μ	planform parameter in Equation (3.62)	
ξ	angle defined in Figure 3.15	
ξ	angular deflection of aileron	
ρ	density of undisturbed stream	
σ	span ratio = 2s/b	
$\sigma_{e}$	equivalent span ratio = 2s <sub>e</sub> /b	
Φ	velocity potential	
$\Phi_1$	velocity potential in Figure 3.14(b) or 3.14(c)	
$\Phi_2$	incremental velocity potential in Equation (3.119)	
Х	function defined in Equation (3.104)	
Ý	angle defined in Figure 3.15	
Ψ	stream function in transverse plane	
ω	complex (y.z) plane	
Superscripts		
(1)	closed tunnel	
(2)	open tunnel	
Subscripts		
0.25	quarter-chord	
0.5	sid-chord	

Ŷ

÷

0.75	three-quarter-chord
ື	wake $(x = \infty)$
A	antisymmetrical part
8	elliptic spanwise loading
P	free-stream
g	ground effect
i	induced by tunnel walls
1	from antisymmetrical loading
L	from symmetrical loading
3	zodel
N	point concentration of lift
R	rectangular tunnel
S	symmetrical part
t	tailplane
T	with tunnel constraint
U	wniforg spanwise loading
7	vortex-induced
Prefixes	

δ	contribution	due	to	wall	interference

△ increment due to wall correction

ΣΣ̈́Σ'

ŧ .....

denotes that (m, n) takes all possible integral pairs except (0, 0).

'(prime) denotes differentiation unless otherwise stated.

### LIFT INTERFERENCE ON THREE-DIMENSIONAL WINGS

#### H.C.Garner

## 3.1 INTRODUCTION

In Chapter II various aspects of lift interference on two-dimensional models have been covered. The wall interference corrections on three-dimensional lifting models are not only more numerous and greater in magnitude, but they involve considerably more geometrical parameters to represent the wind tunnel and the model. There is a corresponding diversity of mathematical techniques, and often the analysis demands simplifying assumptions that are unnecessary in two-dimensional flow. Glauert's classic monograph (Ref. 3. 1; 1933) illustrates both these points and provides a comparative background for most subsequent developments.

The governing linearized equation for the velocity potential in three-dimensional steady ideal compressible flow is

$$(1 - M^2) \frac{\partial^2 \Phi}{\partial x^2} + \frac{\partial^2 \Phi}{\partial y^2} + \frac{\partial^2 \Phi}{\partial z^2} = 0$$
(3.1)

in rectangular co-ordinates; alternatively, in cylindrical co-ordinates.

$$(1 - \mathbf{M}^2) \frac{\partial^2 \Phi}{\partial x^2} + \frac{\partial^2 \Phi}{\partial r^2} + \frac{i}{r} \frac{\partial \Phi}{\partial r} + \frac{1}{r^2} \frac{\partial^2 \Phi}{\partial \theta^2} = 0. \qquad (3.2)$$

It is assumed that squares and products of the component perturbations of a uniform velocity U = Ma are negligible throughout the field of flow. The differential Equation (3.1) or (3.2) is subject to outer conditions

$$\partial \Phi / \partial n = 0$$
 at a closed boundary  
 $\partial \Phi / \partial x = 0$  at an open boundary  $\}$ : (3.3)

the former is exact, but the latter follows from the linearized approximation to constant pressure at an undistorted free boundary (Ref.3.1. p.3). More complicated boundary conditions are used in cases of ventilated tunnels (Chapter VI). There are in addition the upstream and downstream conditions that the flow is undisturbed at  $x = -\infty$  and independent of x for large positive values of x.

The first-order effects of compressibility can be obtained from solutions of the linearized differential Equation (3 1) with the aid of the transformation

$$\mathbf{X} = \mathbf{x}, \quad \mathbf{Y} = \beta \mathbf{y}, \quad \mathbf{Z} = \beta \mathbf{z}, \quad (3.4)$$

とうというないのに見ていたると

where  $\beta = (1 - M^2)^{\frac{1}{2}}$ . In the new co-ordinate system the potential satisfies Laplace's equation for incompressible flow

$$\frac{\partial^2 \Phi}{\partial \mathbf{x}^2} + \frac{\partial^2 \Phi}{\partial \mathbf{x}^2} + \frac{\partial^2 \Phi}{\partial \mathbf{z}^2} = 0 . \qquad (3.5)$$

It is convenient to regard the perturbation potential ( $\Phi$ -Ux) as unchanged in the transformation. Then, by the argument in Section 2.1. the linearized solution in compressible flow is readily expressed in terms of an equivalent incompressible flow with the geometric and aerodynamic parameters listed in Table 3.1. That is to say, the equivalent incompressible flow gives an upwash velocity  $\beta^{-1} w$  or an upwash angle  $\beta$ #/U corresponding to a model of chord c, aspect ratio  $\beta$ A and thickness  $\beta$ t in a tunnel of breadth  $\beta$ b and height  $\beta$ n at an incidence  $\beta$ a to a stream of density ho and velocity  $eta^{-2}U$  . There are alternative methods of allowing for compressibility in Reference 3.2, where the free-stream velocity rather than the perturbation velocity potential is kept invariant. As Goldstein and Young point out, the linear perturbation theory of compressible flow is not intended to be applied when shock waves are present, and it must clearly fail in the neighbourhood of a stagnation point. Nevertheless, the theory of wall interference only involves the potential field at appreciable distances from the model, and the linearized equations can still be used to satisfy Equations (3.3) at fairly high subsonic Mach numbers. The effect of compressibility is obtained by substituting the generalized parameters of Table 3.1 into any formulae or numerical data for wall interference in low-speed tunnels.

The classical approach due to Prandtl is to regard the model as a lifting line, when the problem of wall interference reduces to a solution of the two-dimensional Laplace equation in the transverse (y,z) plane containing the model (Ref.3.1, pp.3 to 5). The perturbation potential in this plane is exactly haif that in the distant wake. For a given spanwise lift distribution and tunnel geometry, the interference upwash velocity  $w_i(y)$  along the wing span can usually be obtained analytically; hence the simple corrections

$$\Delta \alpha = \delta_{0} SC_{L} / C$$

$$\Delta C_{D} = \delta_{0} SC_{L}^{2} / C$$
(3.6)

to the measured incidence and drag coefficient of the model are derived (Ref.3.1, pp.3 to 11). The interference parameter  $\delta_0$  is a non-dimensionalized mean value of  $w_i$  weighted proportionally to the spanwise lift distribution and given by

$$\delta_{g} = \frac{1}{L} \int_{-s}^{s} \left( \frac{\mathbf{w}_{i}}{\mathbf{v}} \frac{\mathbf{C}}{\mathbf{SC}_{L}} \right) \frac{d\hat{\mathbf{L}}}{d\mathbf{y}} d\mathbf{y} . \qquad (3.7)$$

Since the drag coefficient is determined by the conditions in the distant wake, the expression for  $\Delta C_D$  in Equation (3.6) is usually regarded as exact according to linearized theory. The corresponding expression for  $\Delta \alpha$  is often too approximate and underestimates the magnitude of the correction.

Unless the wing chord is small, it is necessary to consider the streamwise variation of  $w_i$ , which is interpreted as a curvature of the free stree due to wall constraint. For wings of fairly small span the results of Section 3.2 . • often be used and the interference correction to incidence is approximately

$$\Delta \alpha = \left( \delta_0 + \frac{\overline{c}}{2\beta h} \delta_1 \right) \frac{sc_L}{c} . \qquad (3.8)$$

مه المعطنية تعاليه

where  $\overline{c}$  is the aerodynamic mean chord and  $\delta_1$  is independent of Mach number. Thus, as Göthert remarked (Ref. 3.3; 1940), the streamline curvature correction is of growing importance in compressible flow. A more general treatment is necessary for sweptback wings of moderate spen, which must be considered as lifting surfaces. Many different vortex representations of the model have been suggested (Section 3.3.1). For most purposes it is sufficient to regard  $w_1(x,y)$  in the plane of the model as a linear function of x (Section 3.3.2). Indeed, with the exception of interference effects on long slender wings (Section 3.6.1) and tail planes (Section 3.6.2), a more detailed representation of the model than appears practicable would be required to justify a precise evaluation of the streamwise variation of the interference upwash.

There is a miscellary of literature concerning wall interference on lifting wings, ranging from the excessively numerical to the excessively mathematical. Some papers deal exclusively with one particular tunnel section and rely on extensive tables of interference upwash and approximate computations; the numerical results are only useful for the single tunnel shape, and it is often difficult to assess the ultimate accuracy. At the other extreme, it is not always expedient to carry the mathematical analysis so far that the complexity of the resulting formulae prohibits their use. An attempt is made in Section 3.3.2 to steer a middle course in presenting the basic interference parameters.

Broadly the cases to be considered fall into four categories

- (a) complete spanwise symmetry, e.g., a wing at incidence,
- (b) symmetrical planform but asymmetrical spanwise loading, e.g., a wing with deflected ailerons,
- (c) asymmetrical planform relative to the tunnel, e.g., a yawed wing,
- (d) special configurations to be considered in Section 3.6.

Even with complete spanwise symmetry there is no unique procedure for converting the tunnel-induced upwash distribution into corrections to the measured quantities. The incremental correction to incidence, such as Equation (3.8), is a somewhat arbitrary average value of  $\pi_i/U$  redians, and the subtraction of the residual upwash field is made equivalent to incremental corrections to the serodynamic forces. The complication of asymmetry introduces residual corrections to the lateral moments (Section 3.3.5). The mathematical theories leading to the basic interference parameters for the various types of tunnel section are discussed in Sections 3.4.1 to 3.4.4, and some available sources of numerical data are listed in Table 3.IV. The final incremental corrections are considered numerically in Section 3.5, and in Section 3.5.3 an attempt is made to assess the accuracy of alternative methods where comparative calculations exist.

Experimental considerations of high lift and some practical confirmation of wallinterference theory are given in Section 3.7. The former is largely empirical and beyond the scope of linearized theoretical methods; the latter usually involves tests in different tunnels, on different models or at different or rather small Reynolds numbers and lacks firm conclusions. Unfortunately it is no easy matter to amalgamate the tunnel corrections for wall interference and Reynolds number. Sometimes large models or half models are tested to achieve the highest possible Reynolds number, since the uncertainties of scale effect exceed those of wall interference. In Reference 3.4, Bryant and the present author have suggested a scheme whereby the estimation of wall interference includes a correlation between experiment and an approximate wingloading theory incorporating sectional wing characteristics dependent on Reynolds number. This ideal philosophy envisages corrections to full-scale conditions after wall interference has been taken into account.

There are many problems of lift interference, where a mathematical solution must involve simplifying assumptions regarding tunnel cross-section, wing thickness or location of model. In such cases the use of an electrical analogy is worth consideration. In particular, Reference 3.5 discusses the basic concepts of such analogies and their technical difficulties in relation to open wind tunnels of finite jet lengths.

Attention is drawn to the final Section 3.8 which serves as an index to the more important equations, tables and figures. It is intended as a preliminary guide firstly to the more approximate formulae and numerical data, and secondly to the more complicated and accurate formulae for the interference upwash. A third table lists the various expressions and results for the interference corrections themselves.

#### 3.2 SMALL WINGS IN CLOSED AND OPEN TUNNELS

Useful estimates of lift interference can often be made when the model is assumed to be small. The interference upwash at the wing and the streamline curvature are given respectively by

 $\begin{cases} \delta_{0} = \frac{C \mathbf{w}_{1}}{USC_{L}} \\ \delta_{1} = \frac{\beta Ch}{USC_{L}} \frac{\partial \mathbf{w}_{1}}{\partial \mathbf{x}} \end{cases}$  (3.9)

Both parameters are readily evaluated for closed or open rectangular tunnels (Section 3.2.2). For more general cross-sections a method of conformal transformation will usually determine  $\delta_0$ . When the analysis for  $\delta_1$  becomes intractable, an approximate formula may be used to relate  $\delta_1/\delta_0$  to its known value for a rectangular tunnel (Section 3.2.3). Several types of semi-closed tunnel with mixed boundary conditions are also discussed.

The applications in Section 3.2.4 are restricted to wings of fairly small span; if accuracy within  $\pm 10\%$  is necessary, then the wing span should not exceed half the tunnel breadth or half the tunnel height. For many purposes the formulae and graphs

will give the order of magnitude of the interference corrections with a minimum of effort.

#### 3.2.1 Nethod of Images

The small wing is represented as a semi-infinite vortex pair trailing from the origin of co-ordinates. The doublet strength is

$$\lim_{s \to 0} (2Ks) = \frac{1}{2} USC_{L} , \qquad (3.10)$$

and the corresponding velocity potential is

$$\Phi_{\rm m} = U_{\rm X} + \frac{U_{\rm SC}_{\rm L} z}{8\pi ({\rm y}^2 + z^2)} \left[ 1 + \frac{{\rm x}}{({\rm x}^2 + \beta^2 {\rm y}^2 + \beta^2 z^2)^{\frac{1}{2}}} \right] . \qquad (3.11)$$

where x is in the horizontal direction of the undisturbed stream, y spanwise and z upwards. The boundary conditions (3.3) to be satisfied by the potential  $\Phi = \Phi_{\mu} + \Phi_{i}$  are supposedly independent of x. Like  $\Phi_{\mu}$ , the interference potential is then of the form

$$\Phi_i = f(y,z) + F(x,y,z) ,$$

where F(x,y,z) is an odd function of x. From Prandtl's argument (Ref.3.1, pp.3 to 5), the solution for  $\Phi_i$  in the plane x = 0 is given by the two-dimensional Laplace equation

$$\frac{\partial^2 \Phi_i}{\partial y^2} + \frac{\partial^2 \Phi_i}{\partial z^2} = 0$$

by Equations (3.3) with the upstream condition and Equation (3.11) the boundary conditions on  $\Phi_i$  are



where  $\partial/\partial n$  denotes differentiation along the outward normal to solid portions of the tunnel boundary.

For special shapes and types of tunnel a convenient system of images can be constructed to give the required potential  $\Phi_i$  in the plane x = 0, or a good approximation to it. Typical examples are illustrated in Figure 3.1. In the case of a closed circular tunnel the image system is particularly simple, but it is necessary to consider the wing as a uniformly loaded lifting line of finite span 2s and then take the limit as  $s \to 0$  with the aid of Equation (3.10). The images for a closed rectangular tunnel are doubly infinite in number and are evenly distributed over the (y, z) plane; as will be seen in Section 3.2.2, image systems of this type are not always valid. When image systems fail or do not exist, special analytical treatment and the state of the

is sought. The third illustration of Figure 3.1 represents an incremental image system to the previous one, to give an approximate solution for an octagonal tunnel obtained by the addition of isosceles corner fillets (Section 3.2.3).

In general, the method of images can only be applied to the interference parameter  $\delta_0$  in Equation (3.9). For the important class of rectangular tunnels, however, the images can be regarded as semi-infinite vortex pairs trailing from the plane x = 0, which by symmetry provide a complete three-dimensional system from which  $\Phi_i$ , and hence  $\delta_1$ , can be evaluated. This is not true of the circular and octagonal tunnels.

#### 3.2.2 Rectangular Tunnels

As early as 1931, Theodorsen<sup>3.6</sup> studied different types of rectangular tunnel from the standpoint of minimizing lift interference. Four particular types of breadth b and height h.

(1) completely closed tunnel
(2) completely open tunnel
(3) open sides, closed floor and roof
(4) closed sides, open floor and roof

will be considered. In each case there is a complete image system comprising semiinfinite doublets of strength  $\frac{i}{2}$ USC<sub>L</sub> and the appropriate sign; for a central model they are situated at (x, y, z) = (0, mb, nh) where m and n are integers. Thus, by Equation (3.11), the interference potential for a small lifting wing is

$$\Phi_{i} = \sum_{-\infty}^{\infty} (j) \frac{USC_{L}(z-nh)}{8\pi \{(y-ab)^{2} + (z-nh)^{2}\}} \left[1 + \frac{x}{\{x^{2} + \beta^{2}(y-ab)^{2} \div \beta^{2}(z-nh)^{2}\}^{\frac{1}{2}}}\right], (3.13)$$

where  $\sum_{\infty}^{\infty}$  denotes that (m, n) takes all possible integral pairs except (0, C). Following Glauert (Ref. 3. 1, Figs. 7 and 8, pp. 21 to 24), we take

 $\begin{array}{l} j = j^{(1)} = (-1)^{n} \\ j = j^{(2)} = (-1)^{n} \\ j = j^{(3)} = (-1)^{n+n} \\ j = j^{(3)} = 1 \end{array}$  (3.14)

respectively for the four types of tunnel, the first of whi: is illustrated in Figure 3.1(b). From the definitions in Equations (3.9) the . .t-interference parameters at the origin are obtained as

$$\delta_{0} = \frac{bh}{USC_{1}} \frac{\partial \Phi_{1}}{\partial z} = \frac{bh}{8\pi} \sum_{-\infty}^{\infty} (j) \frac{\pi^{2}b^{2} - n^{2}h^{2}}{(\pi^{2}b^{2} + n^{2}h^{2})^{2}}$$
(3.15)

 $\delta_{1} = \frac{\beta bh^{2}}{USC_{f}} \frac{\partial^{2} \Phi_{1}}{\partial z \partial x} = \frac{bh^{2}}{8\pi} \sum_{-\infty}^{\infty} (j) \frac{m^{2}b^{2} - 2n^{2}h^{2}}{(m^{2}b^{2} + n^{2}h^{2})^{5/2}}.$  (3.16)

Theodorsen<sup>3.6</sup> derived expressions for the interference-upwash parameter  $\delta_0$  for the four types of rectangular tunnel. It was later remarked by Rosenhead (Ref. 3.7; 1933) that the doubly infinite series of Equation (3.15) is not absolutely convergent. He therefore re-examined the problem rigorously and confirmed Theodorsen's result for types (1), (2) and (3), but not for type (4) having closed sides and open floor and roof. Van Schliestett (Ref.3.8; 1934) attempted to verify Theodorsen's values of  $\delta_n$  by experimental means and revealed serious discrepancies for a square tunnel of type (4); he likewise re-examined Theodorsen's work and gave a third expression for  $\delta_0^{(4)}$  that differs from those of References 3.6 and 3.7, but is consistent with his experiment. A full theoretical discussion of this problem is given in Reference 3.9. An snalytical treatment in Fourier series and as independent calculation by relaxation show that the correct result for  $\delta_0^{(4)}$  is obtained when the summation (3.15) is carried out column by column (first with respect to n). However, Equations (3.13) and (3.15) are strictly divergent in this case. The image system fails, since the velocity gradient at the tunnel boundaries is satisfied, but not the upstream condition for undisturbed flow at  $x = -\infty$ . When this last condition is correctly applied, Rosenhead's rigorous analysis is reconciled with the others. Some expressions for  $\delta_0$  are given below.

For a completely closed rectangular tunnel

$$\delta_0 = \delta_0^{(1)} = \frac{\pi h}{24b} + \frac{\pi h}{b} \sum_{n=1}^{\infty} \frac{n}{e^{2\pi n k/b} + 1}.$$

For a completely open rectangular tunnel

$$\delta_{c} = \delta_{0}^{(2)} = -\frac{\pi b}{24h} - \frac{\pi b}{h} \sum_{n=1}^{\infty} \frac{n}{e^{2\pi n b/h} + 1}$$
$$= -\frac{\pi h}{48b} - \frac{\pi h}{2b} \sum_{n=1}^{\infty} \frac{2n - 1}{e^{(2n - 1)\pi h/b} - 1}.$$
(3.17)

For open sides, closed floor and roof

$$\delta_0 = \delta_0^{(3)} = -\frac{\pi h}{48b} + \frac{\pi h}{2b} \sum_{n=1}^{\infty} \frac{2n-1}{e^{(2n-1)\pi h/b} + 1}.$$

For closed sides, open floor and roof

$$\hat{\delta}_0 = \hat{S}_0^{(4)} = -\frac{1}{4} + \frac{\pi h}{24b} - \frac{\pi h}{b} \sum_{n=1}^{\infty} \frac{n}{e^{2\pi a h/b} - 1}$$

and

In Reference 3.9, Rosenhead's discrepancy in  $\delta_0^{(n)}$  was resolved by the superposition of a uniform upwach velocity to restore undisturbed flow at  $x = -\infty$  without violating the conditions (3.3). The streamline-curvature parameter  $\delta_1$  in Equation (3.16) is unaffected thereby; moreover, this doubly infinite series is seen to be absolutely convergent. To evaluate the series, we write

$$\delta_{1} = \frac{h^{2}}{8\pi b^{2}} \left[ 2 \sum_{m=1}^{\infty} (j)_{h=0} \left( \frac{1}{m^{3}} \right) + 2 \sum_{n=1}^{\infty} \left\{ \sum_{m=-\infty}^{\infty} (j) \frac{m^{2} - 2(nh/b)^{2}}{[m^{2} + (nh/b)^{2}]^{5/2}} \right\} \right]$$

and consider the summations in curly brackets. It can be shown that

$$\sum_{n=-\infty}^{\infty} \frac{n^2 - 2\lambda^2}{(n^2 + \lambda^2)^{5/2}} \quad \text{tends asymptotically to } -\frac{2}{\lambda^2} \text{ shew } \lambda \text{ is large.}$$

We therefore put

$$S_{1}(\lambda) = \frac{2}{\lambda^{2}} + \sum_{\mathbf{a}^{\pi} - \infty}^{\infty} \frac{\mathbf{a}^{2} - 2\lambda^{2}}{(\mathbf{a}^{2} + \lambda^{2})^{5/2}}$$
  
=  $-16\pi^{2} \sum_{p=1}^{\infty} p^{2} K_{0}(2\pi p\lambda) - \frac{8\pi}{\lambda} \sum_{p=1}^{\infty} p K_{1}(2\pi p\lambda)^{*}$ , (3.18)

. 7

which behaves like  $-8\pi^2\lambda^{-\frac{1}{4}}e^{-2\pi\lambda}$  for large  $\lambda$ . Next we consider

$$S_{2}(\lambda) = \sum_{\underline{s}^{2} - \infty}^{\infty} (-1)^{\underline{s}} \frac{\underline{s}^{2} - 2\lambda^{2}}{(\underline{s}^{2} + \lambda^{2})^{5/2}}$$
$$= -S_{1}(\lambda) + \frac{1}{4}S_{1}(\frac{1}{4}\lambda), \qquad (3.19)$$

which is identified with the derivative coefficient  $f'(\lambda)$  of the function in Equation (AR) of the Appendix to Reference 3.10. When the appropriate values of j are substituted from Equations (3.14),  $\delta_1$  is obtained as follows.

For completely closed rectangular tunnels

$$\delta_1 = \delta_1^{(1)} = \frac{\pi}{24} + \frac{\hbar^2}{4\pi b^2} \left[ \sum_{n=1}^{\infty} \frac{1}{n^3} + \sum_{n=1}^{\infty} (-1)^n S_1 \left( \frac{n\hbar}{b} \right) \right].$$

For completely open rectangular tunnels

$$\delta_1 = \delta_1^{(2)} = \frac{h^2}{4mb^2} \left[ \sum_{n=1}^{\infty} \frac{(-1)^n}{n^3} + \sum_{n=1}^{\infty} \Re_2 \left( \frac{nh}{b} \right) \right].$$
(3.20)

\* The functions K<sub>0</sub> and K<sub>1</sub> are defined and tabulated in "A Treatise on the Theory of Bessel Functions" by G.N. Watson (Cambridge University Press). For open sides, closed floor and roof

$$\delta_{1} = \delta_{1}^{(3)} = \frac{\hbar^{2}}{4\pi b^{2}} \left[ \sum_{n=1}^{\infty} \frac{(-1)^{n}}{n^{3}} + \sum_{n=1}^{\infty} (-1)^{n} S_{2} \left( \frac{nh}{b} \right) \right] .$$
(3.20)

For closed sides, open floor and roof

$$\delta_{1} = \delta_{1}^{(4)} = -\frac{\pi}{12} + \frac{h^{2}}{4\pi b^{2}} \left[ \sum_{B=1}^{\infty} \frac{1}{B^{3}} + \sum_{L=1}^{\infty} S_{1} \left( \frac{nb}{b} \right) \right].$$

Here

$$\sum_{m=1}^{\infty} \frac{1}{m^3} = 1.2020_5 \quad \text{and} \quad \sum_{m=1}^{\infty} \frac{(-1)^m}{m^3} = -0.9015_5;$$

 $S_1(nh/b)$  from Equation (3.18) is evaluated in Tuble 3.11 and  $S_2(nh/b)$ , given by Equation (3.19), is identically f'(nh/b) from Table AI of Reference 3.10.

Glauert (Ref. 3. 1, pp. 23 and 24) records numerous relationships between  $\delta_0^{(1)}$ ,  $\delta_0^{(2)}$ ,  $\delta_0^{(3)}$  and  $\delta_0^{(4)}$  regarded as functions of h/b, the most important of which is

$$\delta_0^{(1)}(k/b) + \delta_0^{(2)}(b/h) = 0;$$
 (3.21)

owing to the correction to Rosenhead's work, Equation (9.06) of Reference 3.1 becomes in the present notation

$$\delta_0^{(4)}(h/b) + \delta_0^{(4)}(t/h) = -0.25$$
 (3.22)

with the corollary that  $\delta_0^{(1)} \approx -0.125$  for a square tunnel. The following relationships between  $\delta_1^{(1)}$ ,  $\delta_1^{(2)}$ ,  $\delta_1^{(3)}$  and  $\delta_1^{(4)}$  are deduced from superposition of the image systems:

$\delta_1^{(1)}(h/b) + \hat{c}_1^{(3)}(h/b)$	=	$\delta_{1}^{(1)}(\frac{1}{2}h/b)$		
$\delta_{i}^{(1)}(b/b) + \delta_{i}^{(3)}(b/b)$	=	$\frac{1}{2}S_{1}^{(*)}(2b/b)$		~~`
$\delta_{1}^{(2)}(h/b) + \hat{\delta}_{1}^{(3)}(h/b)$	Ħ	$\frac{1}{2}\delta_{1}^{(2)}(2h/b)$	· (3.	.23)
$\delta_{1}^{(2)}(b/b) + \delta_{1}^{(a)}(b/b)$	=	$\delta^{(a)}(\frac{1}{2h}/b)$		

Special interest attaches to rectangular tunnels of types (2) and (4) in Chapter VI. The completely open tunnel is a limiting case of a tunnel with longitudinal slots on all four walls (Section 6.5.4): similarly, type (4) is a limiting case of rectangular tunnels with slotted floor and roof (Section 6.5.3). Figures 3.2 gives all the lift interference parameters of Equations (3.17) and (3.20) for the range of tunnel shape  $0.5 \le h/h \le 2.0$ .

95

If, however, a lifting model is displaced vertically from the centre of the tunnel, there arise both streamwise and upward interference velocities. For a completely closed tunnel with the model (and origin of co-ordinates) at a distance d from the floor, the interference potential corresponding to Equation (3.13) is

$$\Phi_{i} = \sum_{-\infty}^{\infty} \int_{-\infty}^{\pi} (j) \frac{USC_{L}(z-\zeta)}{8\pi\{(y-mb)^{2} + (z-\zeta)^{2}\}} \left[1 + \frac{x}{\{x^{2} + \beta^{2}(y-mb)^{2} + \beta^{2}(z-\zeta)^{2}\}^{\frac{1}{2}}}\right],$$
or  $i = \pm 1$  with  $\zeta = 2mb$  and all integral pairs  $(m, n)$  except  $(0, 0)$ 

where  $j = \pm 1$  with  $\zeta = 2nh$  and all integral pairs (n, n) except (0, 0)j = -1 with  $\zeta = 2(nh-d)$  and all integral pairs (n, n)

Then the upwash and stream-velocity interference at the model are respectively

$$\delta_{0} = \frac{C}{USC_{L}} \frac{\partial \Phi_{1}}{\partial z} = \frac{bh}{8\pi} \sum_{-\infty}^{\infty} (j) \frac{\mathbf{u}^{2}b^{2} - \zeta^{2}}{(\mathbf{u}^{2}b^{2} + \zeta^{2})^{2}} \\ \frac{C\epsilon}{SC_{L}} = \frac{C}{USC_{L}} \frac{\partial \Phi_{1}}{\partial x} = \frac{bh}{8\pi\beta} \sum_{-\infty}^{\infty} (j) \frac{-\zeta}{(\mathbf{u}^{2}b^{2} + \zeta^{2})^{3/2}}$$
(3.24)

where in the latter equation the terms involving j = +1 cancel. Since

$$\sum_{\mathbf{m}^2 = -\infty}^{\infty} \frac{\mathbf{m}^2 - \lambda^2}{(\mathbf{m}^2 + \lambda^2)^2} = -\pi^2 \operatorname{cosech}^2(\pi\lambda),$$

we obtain

$$\delta_0 = \frac{\pi h}{24b} + \frac{\pi h}{8b} \operatorname{cosech}^2 \left( \frac{2\pi d}{b} \right) + \frac{\pi h}{8b} \sum_{n=1}^{\infty} \left\{ \operatorname{cosech}^2 \frac{2\pi (nh+d)}{b} + \operatorname{cosech}^2 \frac{2\pi (nh-d)}{b} - 2 \operatorname{cosech}^2 \frac{2\pi nh}{b} \right\}.$$
 (3.25)

Similarly to Equation (3.18), we put

$$S_{3}(\lambda) = -\frac{2}{\lambda} + \sum_{m=-\infty}^{\infty} \frac{\lambda}{(m^{2} + \lambda^{2})^{3/2}} = 8\pi \sum_{p=1}^{\infty} pK_{1}(2\pi p\lambda). \qquad (3.26)$$

It then follows from the second of Equations (3.24) that

$$\frac{\beta c_{\epsilon}}{sc_{L}} = \frac{h}{8\pi b} \left[ \sum_{n=-\infty}^{\infty} \frac{b}{nh-d} + \sum_{n=-\infty}^{\infty} s_{3} \frac{2(nh-d)}{b} \right]$$
$$= -\frac{1}{6} \cot \frac{\pi d}{h} + \frac{h}{8\pi b} \sum_{n=-\infty}^{\infty} s_{3} \frac{2(nh-d)}{b} . \qquad (3.27)$$

and the second second

Equations (3.25) and (3.27) with (3.26) have been evaluated for b/h = 0.5, 1.0 and 2.0, and both interference parameters are plotted against d/h in Figure 3.3. For example, if b/h = 2.0 and the model is displaced from the central position d = 0.5h to d = 0.625h, then the upwash interference is increased by 24%. The corresponding contribution to the stream velocity at the model is given by

$$\frac{\delta \mathbf{U}}{\mathbf{U}} = \epsilon = 0.095 \frac{\mathrm{SC}_{\mathrm{L}}}{\beta \mathrm{C}}.$$

which could exceed the solid-blockage factor  $\epsilon_s$  (Section 5.2.3).

## 3.2.3 Non-Rectangular Tunnels

۶,

Many closed working sections are of basic rectangular shape, but have triangular corner fillets. Batchelor (Ref.3.11; 1944) devised an image system to obtain  $\delta_0$ when the fillets are isosceles. As illustrated in Figure 3.1(c), the fillets and their images form a doubly infinite array of squares. To the image system for a closed rectangular tunnel in Figure 3.1(b) can be added an appropriate distribution of vorticity round the perimeter of each square, so as to cancel the normal velocity across the fillets and preserve the other boundaries as streaglines. Batchelor's aethod provides a good approximation to this, and his formulae for a particular tunnel have been generalized in Reference 3.12. By a conformal transformation, Gent (Ref. 3.13; 1944) obtains  $\delta_0$  exactly for a closed tunnel of regular octagonal section. These theories are outlined in more detail in Section 3.4.2, where wings of finite span are considered. The interference upwash is increased by the presence of the fillets, but the increment is only about half that which would arise if  $w_i$ were inversely proportional to the cross-sectional area C The parameter  $\delta_{0}$ of Equation (3.9) is therefore reduced by the fillets. An approximate formula, suggested in Appendix II of Reference 3.4, gives

$$\delta_0 = \delta_0^{(1)} \frac{bh + C}{2bh},$$
 (3.28)

where  $\delta_0^{(1)}$  is the value for a closed rectangular tunnel in the first of Equations (3.17). The results of References 3.11 to 3.13 indicate the likely accuracy of this formula as follows.

Turnel	C bh	Theory		s(1)	Francisco (2.92)
Tunnet		Ref.	δ <sub>ο</sub>	00	Equation (3.28)
9 × 7 (Welbourne)	0.9028	3.11	0.1138	0.1204	0.1145
9 × 7 (NPL)	0.9048	3.12	0.1140	0.1204	0. 1147
13 × 9 (NPL)	0.8932	3.12	0.1125	0. 1191	0.1127
Regular octagon	0.8284	3.13	0.1262	0.1368	0. 1251

97

Jahren . . . . . . . . . .

فحذف

The uncertainty in the decrement  $(\delta_0^{(1)} - \delta_0)$  is of the c<sup>-</sup>der ±10%, but  $\delta_0$  itself is within ±1%. When the corner fillets are scalene, the squares of Figure 3.1(c) become rhorbuses and Batchelor's approach is found to be less satisfactory. In such cases Equation (3.28) may continue to be a useful guide. Conformal transformations from the (y, z) plane can simplify the boundary conditions

(3.12) and in effect reduce the problem of evaluating  $\delta_{e}$  to one having a simple image system. Examples are the Schwarz-Christoffel theorem whereby Gent<sup>3.13</sup> transforms the interior of the regular octagon into a half plane, the bipolar co-ordinates whereby Kondo<sup>3.14</sup> investigates tunnels with boundaries of circular arcs, and the simple transformation whereby Glauert (Ref.3.1, p.32) relates elliptical and rectangular boundaries.

In this last instance  $\delta_0$  for closed or open elliptical tunnels is reduced to double summations similar to Equation (3.15). In the present notation Glauert gives for a small wing in a closed elliptical tunnel

$$\delta_{0} = \frac{1}{2} \sinh \theta \cosh \theta \sum_{p=1}^{\infty} \frac{2p-1}{e^{2\theta (2p-1)} + 1}$$

$$= \frac{1}{\theta} \sinh \theta \cosh \theta \left[ \frac{\pi^{2}}{24\theta^{2}} + \frac{1}{6} - \frac{\pi^{2}}{\theta^{2}} \sum_{p=1}^{\infty} \frac{2p-1}{e^{(2p-1)\pi^{2}/2\theta} + 1} \right]$$
(3.29)

where  $b/h = \coth \theta \ge 1$ ; numerical values of  $\delta_0$  due to Sanuki are published in the Appendix to Reference 3.14. When the wing span lies along the minor axis of the ellipse, there are corresponding expressions from Equations (10.05) of Reference 3.1;

$$\delta_{0} = \frac{1}{2} \sinh \theta \cosh \theta \sum_{p=1}^{\infty} \frac{2p-1}{e^{2\theta(2p-1)}-1}$$
$$= \frac{1}{8} \sinh \theta \cosh \theta \left[ \frac{\pi^{2}}{12\theta^{2}} - \frac{1}{6} + \frac{\pi^{2}}{\theta^{2}} \sum_{p=1}^{\infty} \frac{2p}{e^{p\pi^{2}/\theta}+1} \right]$$
(3.30)

where  $b/h = tanh \ \theta \le 1$ . The second of Equations (3.29) or (3.30) is rapidly convergent, unless the ellipse approaches very closely to a circle. In the limit as  $\theta \to \infty$ , the first of Equations (3.29) or (3.30) tends to the result  $\delta_0 = 0.125$ for a closed circular tunnel and is consistent with the image system of Figure 3.1(2). Results for open elliptical tunnels follow from Glauert's interference theorem; as expressed in Equation (3.21),  $\hat{\epsilon}_0$  has the same magnitude, but opposite sign, as that in a closed elliptical tunnel of height b and breadth h .

To the accuracy envisaged in Section 3.2.1 for fairly small models, it is considered satisfactory to assume that the ratio of  $\beta \partial w_i / \partial x$  to  $w_i / C$  is the same as for a rectangular tunnel of the same breadth to height ratio, completely closed or completely open as the case may be. Thus we approximate to  $\delta_1$  by the formula

$$\delta_{1} = \delta_{0} \sqrt{\frac{bh}{C}} \frac{\delta_{1}^{(1)}}{\delta_{0}^{(1)}} \quad \text{or} \quad \delta_{0} \sqrt{\frac{bh}{C}} \frac{\delta_{1}^{(2)}}{\delta_{0}^{(2)}}, \qquad (3.31)$$

where  $\delta_0^{(1)}$ ,  $\delta_0^{(2)}$ ,  $\delta_1^{(1)}$ ,  $\delta_1^{(2)}$  are given in Equations (3.17) and (3.20). The ratios  $\delta_1^{(1)}/\delta_0^{(1)}$  and  $\delta_1^{(2)}/\delta_0^{(2)}$  are plotted against b/h in Figure 3.4. The values  $\delta_1 = 0.248$  and -0.209 so obtained for closed and open circular tunnels are acceptably close to the respective exact values  $\delta_1 = 0.2497$  and -0.1992 computed from Equations (3.136) and (3.137) of Section 3.4.3 and shown in Figure 3.5. The lift interference parameters  $\delta_0$  and  $\delta_1$  for closed and open elliptical tunnels can thus be calculated from Equations (3.29) to (3.31) and are given in Figure 3.5 for the range of shape  $0.5 \le b/h \le 2.0$ . Equations (3.28) and (3.31) combine to provide a rough formula

$$\delta_{1} = \delta_{1}^{(1)} \frac{bh + C}{2\sqrt{bhC}} \approx \delta_{1}^{(1)}$$
(3.32)

for small wings in closed octagenal tunnels,  $\delta_i^{(1)}$  being given by the full curve against b/h in the lower panel of Figure 3.2.

Generally speaking, the problem of lift interference in wind tunnels having partly closed and partly open boundaries is more complex. Tunnels with slotted walls (Chapter VI) and open-jet tunnels of finite length are cases in point. The latter problem is considered in Part III of Reference 3.5 by Eisenstadt, who gives a solution for a circular tunnel involving expansions in Bessel functions. In this particular problem the linearized condition in Equation (3.3) for constant pressure on the open boundary must in general be replaced by

$$\partial \Phi / \partial x = U + u \quad (u \neq 0),$$

to achieve continuity of velocity at the lip of the closed entrance nozzle. However u = G in the special case of a small model situated on the axis of the tunnel, and numerical results are obtained by satisfying the boundary condition at ter points along the free surface. The distribution of interference upwash along ti unnel axis is reproduced in Figure 3.6 for a jet length l = 3R with the wing at various distances  $x_0$  downstream of the entrance nozzle, the broken portion of each curve corresponding to the extent of the open boundary. Provided  $R \leq x_0 \leq (l-R)$ , the corrections at the model (x = 0) are essentially those for an infinitely long open jet, but the streamline-curvature parameter  $\delta_1$  may only apply over a restricted axial distance. As  $x_c$  decreases below R, the upwash-interference parameter  $\delta_0$  changes rapidly towards the average (zero in this case) of that for an open tunnel and that for a closed tunnel. The greatest streamline curvature occurs in the region of the collector  $x = (l - x_0)$ .

For tunnels of infinite length, Kondo<sup>3.18</sup> has obtained exact values of  $\delta_c$  in two particular cases of mixed boundaries, namely semi-closed tunnels of circular and arch-shaped sections. Kondo's examples serve to illustrate an interesting general theorem proposed by Glauert<sup>3.7</sup>:

"The interference on a very small aerofoil in a tunnel, whose boundaries are partly rigid walls and partly free surfaces, is of the same magnitude, but opposite in sign, as that on the same serofoil rotated through a right angle in a tunnel of the same shape as the previous one but where rigid walls replace free surfaces, and free surfaces replace rigid walls."

Glauert has proved a corresponding theorem for completely open tunnels, of which Equation (3.21) is a special case. In the notation of Section 3.2.2, consequences of the general theorem are that

$$\begin{cases} \delta_{0}^{(3)}(h/b) + \delta_{0}^{(3)}(b/h) = 0 \\ \delta_{0}^{(4)}(h/b) + \delta_{0}^{(4)}(b/h) = 0 \end{cases};$$

the former is true, but the latter conflicts with Equation (3.22) and is false. The present author<sup>3.15</sup> has recently proved the theorem for a wing arbitrarily placed in any tunnel having one closed portion and one open portion, as is true of Kondo's examples; apart from an additive constant, the interference velocity potential of the first half of the theorem can be identified with the interference stream function of the second half. With the exception of rectangular tunnels of type (3) having open side-walls and closed floor and roof, the proof cannot be extended to tunnels whose boundaries consist of two closed and two open portions.

#### 3.2.4 Applications

The values of  $\delta_0$  and  $\delta_1$ , discussed in Sections 3.2.2 and 3.2.3, will often suffice for the purpose of rapid estimates of tunnel wall interference. Equations (3.6) give the corrections to the measured incidence and drag of a small model. Although streamline curvature does not normally influence the latter correction, it does introduce a variation in the interference upwash

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \left(\delta_{c} + \frac{\mathbf{x}}{\beta_{h}}\delta_{1}\right) \frac{\mathrm{SC}_{L}}{\mathrm{C}}$$
(3.33)

THE REAL PROPERTY AND

along the centre line of a wing. The first term of Equation (3.33) is equivalent to a lift coefficient

$$(\delta C_L)_0 = \delta_0 \frac{S C_L}{C} \frac{\partial C_L}{\partial \alpha}$$
 acting at the centre of lift  $x = 0$ ;

the second term contributes an increment in lift coefficient

$$(\delta C_L)_1 = \frac{\lambda \overline{c} \delta_1}{2\beta h} \frac{S C_L}{\overline{c}} \frac{\partial C_L}{\partial \alpha}$$
 acting at some position  $x = x_1$ .

where  $\lambda$  and  $x_i$  are to be determined.

The correction  $\Delta \alpha$  is a somewhat arbitrary average value of  $w_i/U$  over the planform. We may simply take

and the second second

~ J

à,

10ì

$$\Delta \alpha = \delta_{0} \frac{\delta c_{L}}{c}$$

$$\Delta c_{L} = -\frac{\lambda \overline{c} \delta_{1}}{2\beta h} \frac{S c_{L}}{c} \frac{\partial c_{L}}{\partial \alpha}$$

$$\Delta c_{m} = \frac{\lambda x_{1} \delta_{1}}{2\beta h} \frac{S c_{L}}{c} \frac{\partial c_{L}}{\partial \alpha}$$
(3.34)

where  $C_{a}$  is referred to an axis through the uncorrected centre of lift. But it is often more convenient to choose  $\Delta \alpha$  such that there is no residual correction to lift. If

A STATE OF A

$$\Delta \alpha = \left( \delta_0 + \frac{\lambda \bar{c}}{2\beta h} \delta_1 \right) \frac{SC_L}{C}. \qquad (3.35)$$

then the residual interference upwash (w\_i/0) -  $\Delta\alpha$  gives rise to increments in lift coefficient

$$(\delta C_L)_C - \Delta \alpha \frac{\partial C_L}{\partial \alpha} = -(\delta C_L)_1 \text{ acting at } \mathbf{x} = 0$$

$$(\delta C_L)_1 = \frac{\lambda \overline{c} \delta_1}{2\beta h} \frac{S C_L}{C} \frac{\partial C_L}{\partial \alpha} \text{ acting at } \mathbf{x} = \mathbf{x}_1$$

Together these give zero lift, but a pure couple or pitching moment coefficient

$$\delta C_{g} = -\frac{x_{1}}{\overline{c}} \left( \delta C_{L} \right)_{1}$$

which has to be subtracted from the measured  $C_{a}$  to give a residual correction

 $\Delta C_{\underline{s}} = \frac{x_1}{\overline{c}} (\delta C_{\underline{L}})_1 = \frac{\lambda z_1 \delta_1}{2\beta \hbar} \frac{S C_{\underline{L}}}{C} \frac{\partial C_{\underline{t}}}{\partial \alpha}$ (3.36)

independent of pitching axis. As in the second of Equations (3.6) the correction to the drag coefficient is

$$\Delta c_{\rm D} = \delta_{\rm g} \frac{{\rm S} c_{\rm L}^2}{{\rm C}} \,. \tag{3.37}$$

Any residual correction to spanwise loading would involve a correction to drag, but this is usually ignored.

It is sometimes suggested, by analogy with two-dimensional models, that  $\lambda = 1$ and  $x_1 = (1/8)\overline{c}$ . When the wing is regarded as a three-dimensional lifting surface, both  $\lambda$  and  $\underline{x}_1$  usually take larger values. On the approximate basis of strip theory, in which each streamwice section of a wing is treated as if the flow were two-dimensional,

and the second s

į

1 mergenin wer

(3.38)

 $\frac{\lambda \mathbf{x}_1}{\mathbf{\tilde{c}}} = \frac{1}{8} \left( \frac{\mathbf{\tilde{c}}}{\mathbf{\tilde{c}}} \right)^2 + \frac{1}{24} \left( \mathbf{A} \tan \Lambda_{0,5} \right)^2 \left( 2 - \frac{\mathbf{\tilde{c}}}{\mathbf{\tilde{c}}} \right)^4$ where  $A = 2s/\bar{c}$  is the aspect ratio of the wing and  $A_{0.5}$  is its midchord sweepback

(assumed constant). Thus  $\lambda x_1$  would vary from  $(1/8)\overline{c}$  for rectangular wings and  $(1/3)\overline{c}$ for delta wings to larger values for some wings with sweptback trailing edges. The interference corrections resulting from Equations (3.35) to (3.38) are

 $\lambda = \int_{-\infty}^{1} \left(\frac{c}{\overline{c}}\right)^2 d\left(\frac{y}{s}\right) = \frac{\overline{c}}{\overline{c}}$ 

$$\Delta \alpha = \left( \delta_{0} + \frac{\overline{c}\delta_{1}}{2\langle h \rangle} \frac{SC_{L}}{C} \right)$$

$$\Delta c_{D} = \delta_{0} \frac{SC_{L}^{2}}{C} \qquad (3.39)$$

$$\Delta c_{R} = \frac{\overline{c}\delta_{1}}{16\langle h |} \left[ \left( \frac{\overline{c}}{\overline{c}} \right)^{2} + \frac{1}{3} \langle A \tan A_{0.5} \rangle^{2} \left( 2 - \frac{\overline{c}}{c} \right) \right] \frac{SC_{L}}{C} \frac{\partial C_{L}}{\partial \alpha}$$

where  $\overline{c}$  is given in Equation (3.38); the terms involving  $\delta$ , are approximate, the values of  $\lambda$  and  $x_1$  from Equations (3.38) being poor substitutes for values from lifting-surface theory.

Typical theoretical values of  $\lambda$  and  $\lambda x_i/\overline{c}$  are compared with the corresponding values from Equations (3.38) in Table 3.III. Apart from rectangular wings and some slender wings,  $\overline{C}/\overline{C}$  lies within 16% of the appropriate theoretical value of  $\lambda$ . The accuracy of strip theory for various planforms is illustrated for a range of leading-edge sweepback  $\Lambda_{\rm L}$  in Figure 3.7, which suggests that  $\overline{c}/\overline{c}$  is likely to overestimate or underestimate  $\lambda$  according as A tan  $\Lambda_{t}$  is greater or less than about 3. Table 3.III shows the unreliability of the Equation (3.38) for  $\lambda x_1/\bar{c}$ ; typically it is low by a factor of order 2 for wings without trailing-edge sweepback, whatever the taper parameter  $c_r/\bar{c}$ , but rather too large for wings of high trailingedge sweepback. Whilst the residual correction  $\Delta C_{\mu}$  is usually fairly small, it seens difficult to estimate without recourse to lifting-surface theory. Nevertheless, Table 3. III gives some guidance for uniformly tapered wings.

The interference upwash at the tail of a complete aircraft model may be estimated from Equation (3.33). This early application was made by Glauert and Hartshorn (Ref. 3, 16; 1924); after a correction  $\Delta \alpha$  is made to incidence, there remains a residual correction to tailsetting

$$\Delta \alpha_{\rm t} = \frac{\mathbf{x}_{\rm t} \delta_{\rm I}}{\beta_{\rm B}} \frac{\mathrm{SC}_{\rm L}}{\mathrm{C}} \,. \tag{3.40}$$

where  $x_t$  is approximately the distance between the centroids of the wing and tail surfaces. This will often be quite farge and rather inaccurate. For example, in a closed circular tunnel with  $x_t = R$  and  $\beta = 1$ , Equation (3.40) gives

as compared with the correct value 0.099 from the full curve of Figure 3.6.

In practice the formulae (3.39) and (3.40) may be of more use in open than in closed tunnels, since it is inadvisable to test large models in open tunnels. The free boundaries of the jet will distort and spillage may occur at the collector downstream of the model (Section 2.5.4); for such reasons the more accurate interference corrections for large models become unreliable. In closed tunnels, however, the more accurate interference corrections will often be reliable and yet differ appreciably from those for small models on account of either large span or large chord. Present formulae for  $\delta_0$  will give accuracy within about 10%, provided that the wing span does not exceed half the tunnel height or breadth, i.e.,  $\sigma < \frac{1}{2}h/b$  and  $\sigma < \frac{1}{2}$ . These conditions will often be satisfied for slender models, when the chord length becomes  $\downarrow$  witing factor. However, Equation (3.33) is a useful approximation to the more exact analysis considered in Section 3.6.1.

 $\frac{\pi R^2}{SC_{\star}} \Delta_{\alpha_t} = 0.125$ 

Half-models mounted on the side-wall of a tunnel are treated as complete wings in a tunnel of twice the area with the side-wall as reflection plane (Section 3.4.4). In view of the great difficulty in obtaining  $\delta_1$  for non-rectangular configurations, it may be expedient to use the simple formula of Equation (3.31) in conjunction with a mean value of  $\delta_0$  appropriate to the span of the model. The corrections to measured quantities in either Equations (3.34) or Equations (3.35) to (3.39) may then suffice.

#### 3.3 GENERAL THEORY OF LIFT INTERFERENCE

In general the formulations of Section 3.2 will be too approximate. When the wing is large, there are many different approaches to the evaluation of interference upwash depending mainly on the representation of the model (Section 3.3.1). The choice of vortex model will be diffuenced by aspect ratio, sweepback and yaw of the wing, the symmetry or asymmetry of its spanwise lift distribution and by the availability of tabulated interference parameters for any particular tunnel (Section 3.4).

In Section 3.3.2, the interference parameters are defined for different basic types of model. irrespective of tunnel cross-section. General expressions for the distribution of interference upwash are then derived. The procedures for converting such distributions into corrections to measured quantities are far from standardized. Section 3.3.3 summarizes the general approach to this problem. The various cases are then classified according to the degree of symmetry about the centre line. Sections 3.3.4 to 3.3.6 summarize approximate formulae for the final lift interference corrections relating to complete and half-wing models.

#### 3.3.1 Representation of Model

In problems of two-dimensional lift interference it may occasionally be desirable to represent the model precisely as a thin merofoil is represented in classical theory. The same only remains true in three dimensions in the case of unswept wings of high aspect ratio to which the classical lifting-line theory applies. As illustrated in

103

- - A. Maynumumumumumumum

Figure 3.8(a), an elementary strip of the wing  $(\delta y)$  having circulation  $\Gamma$  is represented by a trailing vortex element of strength  $-(d\bar{a}/dy)\delta y$ . At the other extreme, a slender wing can be divided into elementary strips  $(\delta x)$  supporting a lift  $\delta L = (d\bar{a}/dx)\delta x$ . The lift interference (Section 3.6.1) is regarded as that due to a distribution of trailing vortex pairs of doublet strength  $\delta L/\rho U$  along the centre line; Figure 3.8(b) is the generalization of the representation of a small wing in Section 3.2.1. Thus even a slender wing is replaced by a simpler vortex model than that used in the corresponding free-stream theory.

This is still more evident for a general planform, as a complete vortex sheet is needlessly complicated. It is desirable to let the vortex model depend upon as many of the aerodynamic characteristics as are measured or can be estimated easily. Figure 3.8 includes three typical representations, but there is no universal opinion as to which is best. In Reference 3.17, Acum uses in effect a lifting line through the local chordwise centres of pressure with the appropriate trailing vorticity. The same model is used in Reference 3.18 for wings with asymmetrical spanwise loading, but it is then desirable to split the loading into its symmetrical and antisymmetrical parts, as indicated in Figures 3.8(c) and (d). Eisenstadt (Ref.3.19; 1947) uses discrete vortices in the form of swept uniformly loaded lifting lines, while Katzoff and Hannsh (Ref.3.20; 1948) use point concentrations of lift, each representing a portion of the wing. These two vortex models, illustrated in Figures 3.8(e) and (f), are more adaptable, being readily applied to yawed wings.

It is sometimes sufficient to replace a trailing vortex sheet by a single vortex pair. By expanding the velocity field of the trailing vortex sheet in inverse powers of the lateral distance, Loos (Ref.3.21; 1951) derives the equivalent semi-span  $s_e$  and vortex strength  $K_e$ , such that

$$K_{e}z_{e} = -\int_{0}^{s} \frac{d\Gamma}{dy} y \, dy = \int_{0}^{s} \Gamma \, dy$$

$$K_{e}s_{e}^{3} = -\int_{0}^{s} \frac{d\Gamma}{dy} y^{3} \, dy = 3\int_{0}^{s} \Gamma y^{2} \, dy$$
(3.41)

To a useful approximation a horse-shoe vortex of strength K<sub>e</sub> and semi-span s<sub>e</sub> gives the same interference upwash distribution as the unswept lifting line of Pigure 3.8(a). For elliptic spanwise loading  $s_e/s = \frac{1}{2}\sqrt{3}$ . Since the interference apwash has to be averaged over the complete span, it may be expedient to replace the wing by a uniformly loaded one of semi-span slightly greater than  $s_e$ . In this respect Swanson and Toll (Ref.3.22; 1943) have suggested that for their particular application the effective span ratios  $s_e/s = 0.93$ , 0.28 and 0.83 may be used for wings of taper ratios  $c_t/c_r = 1$ , 0.5 and 0.25 respectively.

Certain types of model may require more complicated representation. Consideration of tail-plane interference and wing-body combinations will be deferred until Section 3.6.2. Wings with deflected control surfaces may be treated by the procedure suggested in Section 4.5 of Reference 3.4. Basically this amounts to the superposition of discrete elements of the type in Figure 3.8(e). The introduction of
control surfaces affects the loci of the lifting lines in Figures 3.8(c) and (d), but otherwise leaves models (a) to (d) unchanged. Model (f) might well call for greater subdivision of the wing, but would serve for symmetrical and asymmetrical loading alike.

## 3.3.2 Basic Interference Parameters

In the first place the wing and its load distribution are supposed to be symmetrical about the vertical plane through the axis of the tunnel. The vortex models of the wing in Figures 3.8(a) and (c) can be constructed by superposition of symmetrically placed horse-shoe vortices of strength K and semi-span t. The interference upwash angle in the plane of the wing is then expressed as

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{4\mathbf{x}t}{\mathbf{U}\mathbf{C}} \left\{ \delta_{0}(\mathbf{y},t) + \frac{z}{\beta h} \delta_{1}(\mathbf{y},t) + 0 \left(\frac{\mathbf{x}}{\beta h}\right)^{3} \right\}.$$
 (3.42)

where, as in Acum's<sup>3.17</sup> theory,  $\delta_0(y,t)$  and  $\delta_1(y,t)$  are functions to be determined for a particular tunnel (Section 3.4) and terms involving the third and higher powers of  $x/\beta h$  are neglected. Equation (3:42) is a generalization of Equation (3.33) for a small wing; both  $\delta_0$  and  $\delta_1$  may be regarded as also dependent on any vertical displacement of the wing from the tunnel axis.

The well known theorem of Prandtl shows that, in the limit as  $x \to \infty$ , Equation (3.42) becomes

$$\frac{\mathbf{w}_i}{\mathbf{U}} = \frac{4\mathbf{K}t}{\mathbf{U}\mathbf{C}} \left\{ 2\delta_0(\mathbf{y}, \mathbf{t}) \right\},$$

which may be calculated on a two-dimensional basis. Thus, if the chordwise extent of the model is small, as envisaged in Figure 3.8(a),  $\delta_1$  may be neglected and for certain tunnels the expressions for  $w_1/U$  are guite simple. Sanders and Pounder<sup>3.23,3.24</sup> exploit this for closed rectangular tunnels on the basis of liftingline theory. For an untwisted, uncambered wing at incidence, lifting-line theory yields the simple result in Equations (3.6) and (3.7) with the mean interference parameter

$$\delta_{g} = \frac{1}{L_{p}} \int_{-g}^{g} \left( \frac{v_{1}}{v} \frac{C}{SC_{L}} \right) \frac{dL_{p}}{dy} dy , \qquad (S.43)$$

where  $d\hat{u}_p/dy$  denotes the lift per unit span in free air. For a uniformly loaded wing it is particularly easy to obtain

 $(\delta_0)_{ij} = \int_{\sqrt{2}}^{1} \delta_0(\mathbf{y}, \mathbf{s}) d\left(\frac{\mathbf{y}}{\mathbf{s}}\right)$  (3.44)

Alternatively  $(5_0)_{ij}$  is given directly by the two-dimensional solution in the transverse plane for the interference stream function  $\frac{\Phi_i}{2}$  corresponding to  $\Phi_i$ , since

105

-----

$$(\delta_0)_{U} = \frac{C}{SC_L} \int_0^1 \frac{w_1}{U} d\left(\frac{y}{s}\right) = -\frac{C}{SC_L} \int_0^1 \frac{1}{U} \frac{\partial \psi_1}{\partial y} d\left(\frac{y}{s}\right)$$
$$= \frac{C}{USC_L s} \left[\psi_1\right]_{y=s}^{y=0}; \qquad (3.45)$$

in practice s would be replaced by a reduced effective semi-span, as discussed in Section 3.3.1. A better approximation is to assume that the spanwise loading is elliptic over the full span, when Equation (3.43) becomes

$$(\delta_0)_{\mathbf{E}} = \frac{4}{\pi} \int_0^1 \left( \frac{\mathbf{v}_1}{\mathbf{U}} \frac{\mathbf{C}}{\mathbf{SC}_{\mathbf{L}}} \right) \left\{ 1 - \left( \frac{\mathbf{y}}{\mathbf{s}} \right)^2 \right\}^{\frac{1}{2}} d\left( \frac{\mathbf{y}}{\mathbf{s}} \right) , \qquad (3.46)$$

$$\frac{\mathbf{v}_1 \mathbf{C}}{\mathbf{U} \mathbf{SC}_{\mathbf{L}}} = \frac{4}{\pi} \int_0^{\mathbf{1}} \delta_0(\mathbf{y}, \mathbf{t}) \left( \frac{\mathbf{t}}{\mathbf{s}} \right)^2 \left\{ 1 - \left( \frac{\mathbf{t}}{\mathbf{s}} \right)^2 \right\}^{-\frac{1}{2}} d\left( \frac{\mathbf{t}}{\mathbf{s}} \right) .$$

where

This has been calculated for a wide range of tunnels (Section 3.4).

For sweptback wings of moderate aspect ratio the interference upwash due to the vortex model of Figure 3.8(c) is readily expressed in terms of  $\delta_0(\mathbf{y}, t)$  and  $\delta_1(\mathbf{y}, t)$ . Following Reference 3.17, we replace the elementary vortices by a horse-shoe vortex of strength  $\Gamma$  and semi-span (t +  $\delta t$ ) together with one of strength  $-\Gamma$  and semi-span t. Hence

$$\frac{\delta \mathbf{w}_{i}}{\mathbf{v}} = \frac{4\Gamma}{\mathbf{v}C} \left[ \frac{\partial}{\partial t} \left\{ t \delta_{0}(\mathbf{y}, t) \right\} + \frac{\mathbf{x} - \overline{\mathbf{x}}}{\beta \mathbf{h}} \frac{\partial}{\partial t} \left\{ t \delta_{1}(\mathbf{y}, t) \right\} \right] \delta t.$$

where  $x = \bar{x}(t)$  is the curve through the local centres of pressure. Integration over the span of the wing  $-s \le t \le s$  gives

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \int_{0}^{1} \frac{4\Gamma \mathbf{s}}{\mathbf{U}\mathbf{c}} \left[ \frac{\partial}{\partial t} \left\{ \mathbf{t} \delta_{0}(\mathbf{y}, t) \right\} + \frac{\mathbf{x} - \overline{\mathbf{x}}(t)}{\beta \mathbf{h}} \frac{\partial}{\partial t} \left\{ \mathbf{t} \delta_{1}(\mathbf{y}, t) \right\} \right] d\left( \frac{t}{\mathbf{s}} \right); \quad (3.47)$$

for elliptic spanwise loading Equation (3.47) becomes

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{4SC_{L}}{\pi C} \int_{C}^{T} \left[ \frac{\partial}{\partial t} \left\{ t\delta_{a}(\mathbf{y}, t) \right\} + \frac{\mathbf{x} - \overline{\mathbf{x}}(t)}{\beta h} \frac{\partial}{\partial t} \left\{ t\delta_{1}(\mathbf{y}, t) \right\} \right] \left\{ 1 - \frac{t^{2}}{s^{2}} \right\}^{\frac{1}{2}} d\left(\frac{t}{s}\right), \quad (3.48)$$

as obtained in Equatica (20) of Reference 3.4.

If the model remains symmetrically situated but its spanwise loading is asymmetrical, then the load distribution can be split into symmetrical and antisymmetrical portions represented by the vortex models (c) and (d) of Figure 3.8. In Reference 3.18 (Garner and Acum; 1953), the symmetrical part of the interference is obtained in the form of Equation (3.47). In addition to Equation (3.42) we now define

a bestagning age that

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{4\mathbf{x}t}{\mathbf{U}\mathbf{C}} \left[ \kappa_{0}(\mathbf{y}, \mathbf{t}) \div \frac{\mathbf{x}}{\beta \mathbf{h}} \kappa_{1}(\mathbf{y}, \mathbf{t}) + 0 \left(\frac{\mathbf{x}}{\beta \mathbf{h}}\right)^{2} \right]$$
(3.49)

as the interference upwash angle due to a pair of asymmetrically placed horse-shoe vortices, one of strength -K and span  $-t \le y \le 0$  and the other of strength K and span  $0 \le y \le t$ . It then follows that, corresponding to Figure 3.8(d),

$$\frac{\mathbf{w}_{1}}{\mathbf{u}} = \int_{0}^{1} \frac{4\Gamma_{s}}{\mathbf{u}C} \left[ \frac{\partial}{\partial t} \left\{ t\kappa_{0}(\mathbf{y},t) \right\} + \frac{\mathbf{x} - \bar{\mathbf{x}}(t)}{\beta h} \frac{\partial}{\partial t} \left\{ t\kappa_{1}(\mathbf{y},t) \right\} \right] d\left(\frac{t}{s}\right). \quad (3.50)$$

an antisymmetrical function of y in which  $\Gamma$  and  $\bar{x}(t)$  refer to the antisymmetrical portion of the wing loading. This approach to wall interference demands the tabulation of the basic parameters

$$\frac{\partial}{\partial t} \left( t \, \delta_{t}(\mathbf{y}, t) \right) , \quad \frac{\partial}{\partial t} \left( t \delta_{1}(\mathbf{y}, t) \right) , \quad \frac{\partial}{\partial t} \left\{ t \mathbf{x}_{0}(\mathbf{y}, t) , \quad \frac{\partial}{\partial t} \left\{ t \mathbf{x}_{1}(\mathbf{y}, t) \right\} ,$$

which can be evaluated fairly easily for rectangular tunnels (Section 3.4.1). There is no equivalent procedure for asymmetrically situated wings.

The more flexible method of Katzoff and Hannah in Reference 3.29 uses point sources of lift, which are considered to Lie in the horizontal plane through the centre line of the tunnel. As indicated in Figure 3.8(f), it is necessary to segment the wing and to estimate the lift  $L_N$  and its centroid  $(x_N, y_N)$  on each sogment. The total interference upwash is obtained as

$$\frac{\mathbf{x}_{i}}{\mathbf{U}} = \sum_{\mathbf{x}} \frac{\mathbf{L}_{\mathbf{x}}}{\rho \mathbf{U}^{2} \mathbf{b}^{2}} \, \mathbf{x}_{i} \left( \frac{\mathbf{x} - \mathbf{x}_{\mathbf{y}}}{\beta \mathbf{b}} \, , \, \frac{\mathbf{y}}{\mathbf{b}} \, , \, \frac{\mathbf{y}_{\mathbf{y}}}{\mathbf{b}} \right) \, , \qquad (3.51)$$

where b is the breadth or some typical length of the tunnel cross-section. In general, charts of

 $\mathbf{W}_{i}\left(\frac{\mathbf{x}-\mathbf{x}_{\mathbf{N}}}{\beta \mathbf{b}}\cdot\frac{\mathbf{y}}{\mathbf{b}}\right)$ 

must be constructed for a series of spanwise locations of the doublet  $L_{\rm M}^{\prime}/\rho U$ . For rectangular tunnels, however, the function  $\Xi_{\rm i}$  can be defined by two charts independent of  $y_{\rm M}^{\prime}/b$ .

Both the foregoing procedures for sweptback wings are economical as regards the basic computations for any particular tunnel cross-section. Both imply some knowledge of the lift distribution;  $\Gamma$  and  $\bar{x}$  in Equation (3.47) or (3.50) are normally obtained by lifting-surface theory, while  $L_N$  and  $(x_N, y_N)$  in Equation (3.51) must be estimated likewise. The former procedure of Reference 3.18 neglects terms of order  $[(x-\bar{x})/\beta h]^3$  and is restricted to unyawed wings. These limitations do not apply to that of Reference 3.20, which suffers from the inaccuracies of a graphical method and the poorer representation of the model. Between these two extremes are the intermediate schemes of References 3.19 and 3.25, which apply to closed circular and

rectangular tuncels respectively. For each half of the vortex model of Figure 3.8(e) Eisenstadt tabulates, in the present notation,

$$\frac{4Cr_1}{3t} \quad \text{as a function of } \quad \frac{2r}{\beta b}, \quad \frac{2r}{b}, \quad \frac{2t}{b} \quad \text{and } \cot^{-1}(\beta \cot h),$$

the list parameter being the equivalent sweepbeck angle of the bound vortex line (Table 3.1). Swanson's approach in Reference 3.25 is for a particular model, but it would in general involve the non-dimensional interference upwash in terms of the same frar parameters. These accesses are more curbersome than the three-parameter system of Equation (3.51) or the two-parameter system of Equation (3.47). On the other hand Katwoff and Hannah have pointed out in Figure 9 of Reference 3.20, that a series of calculations for one sweep angle may suffice for computing the wall interference on a wing of different sweep.

## 3.3.3 Corrections to Measured Quantities

. . . . .

We first suppose that z suitable vortex model of the wing may be identified with some theoretical approximation to the non-dimensional wing loading  $l_p(x,y)$  in the free strear. The basic interference parameters then determine an upwash angle  $\delta \alpha(x,y)$  from which an equivalent incremental loading  $\delta l(x,y)$  may be computed theoretically; this introduces additions to be superposed on the vortex model. The procedure may then be repeated to obtain the corresponding  $\delta^2 \alpha$  and  $\delta^2 l$ , and so on until the final theoretical picture gives distributions of load and interference upwash

$$l_{T} = l_{F} + \delta l + \delta^{2} l + \dots$$
$$\mathbf{w}_{i} / \mathbf{U} = \delta \alpha + \delta^{2} \alpha + \delta^{3} \alpha + \dots$$

in the tunnel. For many reasons  $l_{\rm T}$  will not be identical to the measured distribution l, and there will be inevitable discrepancies between the theoretical coefficients of lift and rolling moment  $\mathcal{C}_{\rm LT}$  and  $\mathcal{C}_{\rm IT}$  corresponding to  $l_{\rm T}$  and the measured  $\mathcal{C}_{\rm L}$  and  $\mathcal{C}_{\rm I}$ . It becomes necessary to subdivide  $l_{\rm T}$  into  $l_{\rm S} + l_{\rm A}$ , the sum of symmetrical and antisymmetrical functions of y, and to apply respective correction factors  $\mathcal{C}_{\rm L}/\mathcal{C}_{\rm LT}$  and  $\mathcal{C}_{\rm R}$  and  $l_{\rm R}$  and the appropriate parts of  $w_{\rm I}/U$ .

The magnitude of the discrepancies, so removed, would be expected to be as great as that of  $\delta C_L$  or  $\delta C_l$  corresponding to  $\delta l$ . This would justify the minor simplifying assumption that  $\delta l$  is directly proportional to  $l_p$ , i.e.,

$$\hat{\mathcal{L}}(\mathbf{x},\mathbf{y}) \equiv \mathbf{k}l_{\mathbf{p}}(\mathbf{x},\mathbf{y})$$
.

It then follows that

$$l_{T} = l_{p}(1 + k + k^{2} + ...) = l_{p}/(1 - k)$$

$$u_{n}/U = S\alpha(1 + k + k^{2} + ...) = S\alpha/(1 - k)$$

The original vortex model, based on the serodynamics of the free stream, will suffice, provided that it is separated into its symmetrical and antisymmetrical parts with

adjustable factors to ensure consistency with the measured lift and rolling moment respectively. We therefore express

$$\frac{\mathbf{w}_{i}}{\mathbf{U}} = \left[\frac{\left(\mathbf{w}_{i}\right)_{L}}{\mathbf{UC}_{L}}\right] \mathbf{C}_{L} + \left[\frac{\left(\mathbf{w}_{i}\right)_{l}}{\mathbf{UC}_{l}}\right] \mathbf{C}_{l} . \qquad (3.52)$$

For symmetrically placed wings  $[(w_i)_L/UC_L]$  will be a symmetrical function of y and  $[(w_i)_L/UC_l]$  antisymmetrical, but in general they will both be asymmetrical. The associated factors are the measured lift and rolling moment coefficients; the subscript T can now be cmitted without ambiguity.

The distribution of incidence  $w_i/U$  has to be converted into corrections to measured quantities, both geometric and aerodynamic. In principle there is no difficulty in using lifting-surface theory to compute the aerodynamic coefficients  $\delta C_L$ ,  $\delta C_l$  and  $\delta C_m$  corresponding to  $w_i/U$  and also the increments  $\delta C_{DV}$  and  $\delta C_{nV}$  in the vortexinduced drag and yawing moment coefficients. Wall corrections could simply amount to the subtraction of these interference quantities from those measured; for exemple, in the notation of Equation (3.52) we could take

$$\Delta \mathbf{C}_{\mathbf{m}} = -\delta \mathbf{C}_{\mathbf{m}} = -\left[\frac{\left(\delta \mathbf{C}_{\mathbf{m}}\right)_{\mathbf{L}}}{\mathbf{C}_{\mathbf{L}}}\right] \mathbf{C}_{\mathbf{L}} - \left[\frac{\left(\delta \mathbf{C}_{\mathbf{m}}\right)_{\mathbf{i}}}{\mathbf{C}_{\mathbf{i}}}\right] \mathbf{C}_{\mathbf{i}} , \qquad (3.53)$$

where the second term would vanish if the wing were symmetrically placed. An arbitrary correction may be applied to any measured geometric quantity, in particular a correction  $\Delta \alpha$  to the incidence of the model. It is then necessary to calculate the appropriate free-stream derivatives  $\partial C_L / \partial \alpha$ ,  $\partial C_l / \partial \alpha$  and  $\partial C_m / \partial \alpha$ , and, for example, to add the quantity  $\Delta \alpha (\partial C_m / \partial \alpha)$  to the right hand side of Equation (3.53). It is often convenient to choose  $\Delta \alpha$  such that  $\Delta C_L = 0$ . Then Equation (3.53) would become

$$\Delta \mathbf{C}_{\mathbf{m}} = \left\{ \frac{\partial \mathbf{C}_{\mathbf{m}}}{\partial \mathbf{C}_{\mathbf{L}}} \left[ \frac{\left( \delta \mathbf{C}_{\mathbf{L}} \right)_{\mathbf{L}}}{\mathbf{C}_{\mathbf{L}}} \right] - \left[ \frac{\left( \delta \mathbf{C}_{\mathbf{m}} \right)_{\mathbf{L}}}{\mathbf{C}_{\mathbf{L}}} \right] \right\} \mathbf{C}_{\mathbf{L}} + \left\{ \frac{\partial \mathbf{C}_{\mathbf{m}}}{\partial \mathbf{C}_{\mathbf{L}}} \left[ \frac{\left( \delta \mathbf{C}_{\mathbf{L}} \right)_{l}}{\mathbf{C}_{l}} \right] - \left[ \frac{\left( \delta \mathbf{C}_{\mathbf{m}} \right)_{l}}{\mathbf{C}_{l}} \right] \right\} \mathbf{C}_{l} . \quad (3.54)$$

The expressions in curly brackets are theoretical quantities dependent on the vortex model, the tunnel cross-section and the wing planform. Thus, like  $w_i/U$ , each interference correction becomes the product or sum of products of theoretically and experimentally determined quantities.

Difficulties arise when greater precision or, as is more often the case, rapid calculation is required. Reference 3.23 is an eminent example of the former; in effect Sanders and Pounder shun the use of Equation (3.52), for their treatment implies that the load distribution cannot be regarded as independent of wall interference apart from factors proportional to the measured  $C_L$  and  $C_l$ . Even within the frazework of lifting-line theory their analysis becomes exceedingly complicated. The more common difficulty is that, because corrections are fairly small and therefore not required to great accuracy, the use of lifting-surface theory is considered to be an unwarranted computation. Section 3.2.4 has already illustrated the trouble. For a small symmetrical wing Equation (3.54) reduces to Equation (3.36), but the theoretical quantities  $\lambda$  and  $x_1$  must be estimated. Figure 3.7 in conjunction with strip theory can perhaps give  $\lambda$  within ±10% and so  $\Delta \alpha$  in

Equation (3.35) to a fair accuracy. However, Table 3.III suggests that in the absence of lifting-surface theory the residual correction  $\Delta C_{\rm m}$ , proportional to  $\lambda x_1$ , is liable to be excessively inaccurate. In the following Sections 3.3.4 to 3.3.6 suitable approximations are considered for different types of problem when the wing is not small.

# 3.3.4 Complete Spanwise Symmetry

For large wings having complete spanwise symmetry there are approximate methods by which rapid interference corrections may be estimated. It is usual to take  $\Delta \alpha$ as a weighted mean of  $w_i/U$  along the three-quarter-chord line, such that the residual correction  $\Delta C_L$  is negligible. Thus

$$\frac{\Delta_{\alpha}}{C_{L}} = \int_{0}^{1} \left(\frac{\overline{w_{i}}}{UC_{L}}\right)_{0.75} W d\left(\frac{y}{s}\right) , \qquad (3.55)$$

where\* from

Reference 3.26, Equation (29), 
$$W = cC_{LL}/\overline{c}C_L$$
  
Reference 3.17, Equation (11),  $W = \frac{2.4}{\pi} \left\{ 1 - \left(\frac{y}{s}\right)^2 \right\}^{\frac{1}{2}} + 0.4 \frac{c}{\overline{c}}$   
elliptic loading  $W = \frac{4}{\pi} \left\{ 1 - \left(\frac{y}{s}\right)^2 \right\}^{\frac{1}{2}}$   
strip theory  $W = c/\overline{c}$ 

The first weighting is the spanwise loading factor for the wing at uniform incidence; charts of  $cC_{LL}/\bar{c}C_L$  are given in Figure 6 of Reference 3.26 for various sweep angles, aspect ratios and taper ratios. The second, equivalent to 0.6 times the elliptic weighting added to 0.4 times the chord weighting, can be evaluated when the factor  $cC_{LL}/\bar{c}C_L$  is uncertain. In many cases the elliptic weighting is accurate enough, and the basic interference parameter  $(\delta_0)_E$  in Equation (3.46) is useful. The chord weighting from strip theory has little to commend it.

• By application of the reverse-flow theorem C.R. Taylor<sup>3.70</sup> of the Aerodynamics Department of the Royal Aircraft Establishment has pointed cut the precise result that

$$\mathcal{L}_{\alpha} = \frac{1}{S} \iint_{S} \frac{\mathbf{w}_{1}}{UC_{L}} \hat{l} \, dxdy ,$$

where  $\hat{l}$  is the non-dimensional wing loading on the reversed planform at uniform incidence; the corresponding spanwise loading factor  $c\hat{c}_{LL}/c\hat{c}_{L}$  would therefore seem to be an appropriate choice of W in Equation (3.55).

The drag correction is usually taken to be consistent with lifting-line theory and Equations (3.6) and (3.7). If  $(w_i^{/UC}_L)_{\infty}$  denotes the value of  $w_j^{/UC}_L$  in the distant wake, then

$$\frac{\Delta C_{\rm D}}{C_{\rm L}^2} = \frac{1}{2} \int_0^1 \left( \frac{\mathbf{w}_1}{UC_{\rm L}} \right)_{\infty} \frac{cC_{\rm LL}}{\bar{c}C_{\rm L}} d\left( \frac{\mathbf{y}}{\mathbf{s}} \right) . \qquad (3.56)$$

where  $cC_{LL}/\bar{c}C_L$  now represents the actual spanwise loading that would be measured on the wing. As in Equation (3.37), a small term from a residual correction to spanwise loading is neglected.

Since  $\Delta C_L = 0$ ,  $\Delta C_m$  will be a pure couple independent of pitching axis. In both References 3.26 and 3.17 the residual correction may be written as

$$\Delta \mathbf{C}_{a} = (\Delta \mathbf{C}_{a})_{1} + (\Delta \mathbf{C}_{a})_{2}, \qquad (3.57)$$

where  $(\Delta C_m)_i$  results from the shift in spanwise centre of lift associated with the spanwise variation of  $(w_i/UC_i)_{0.75}$ , and  $(\Delta C_m)_2$  arises from the streamline curvature. Alternative formulae for  $(\Delta C_m)_i$  are given in References 3.26 and 3.17; the former is preferred as it avoids a calculation by lifting-line theory and in the present notation, gives directly

$$\frac{(\Delta C_{\underline{y}})_{1}}{C_{\underline{L}}} = \frac{\frac{1}{2}\pi A^{2} (\partial C_{\underline{L}}/\partial \alpha) \tan \Lambda_{0.25}}{\pi A + 2 \partial C_{\underline{L}}/\partial \alpha} \int_{0}^{1} \left[ \left( \frac{\overline{y}_{1}}{UC_{\underline{L}}} \right)_{0.75} - \frac{\Delta \alpha}{C_{\underline{L}}} \right] \overline{y} \frac{y}{s} d\left( \frac{y}{s} \right) . \quad (3.58)$$

where A is the aspect ratio and  $\Lambda_{0.25}$  is the sweepback of the quarter-chord line. ( $\Delta C_{m}$ )<sub>2</sub> is incorrectly derived in Reference 3.17: with allowance for compressibility Equation (35) of Reference 3.26 becomes

$$\frac{(\Delta C_{\mathbf{y}})_2}{C_{\mathbf{L}}} = \frac{\pi \cos \Lambda_{0.5}}{4\sqrt{1 - \mathbf{y}^2 \cos^2 \Lambda_{0.5}}} \int_0^1 \left[ \left( \frac{\mathbf{w}_1}{\mathbf{U}C_{\mathbf{L}}} \right)_{0.75} - \left( \frac{\mathbf{w}_1}{\mathbf{U}C_{\mathbf{L}}} \right)_{0.25} \right] \left( \frac{\mathbf{c}}{\mathbf{c}} \right)^2 d\left( \frac{\mathbf{y}}{\mathbf{s}} \right) . \quad (3.59)$$

where  $\Lambda_{0.5}$  is the half-chord sweepback.

Equations (3.58) and (3.59) are not applicable to planforms with curved or cranked edges. In place of Equation (3.57) strip theory would give the less accurate expression

$$\frac{\Delta C_{ga}}{C_{L}} = \frac{\partial C_{L}}{\partial \alpha} \int_{0}^{1} \left[ \left( \frac{w_{1}}{UC_{L}} \right)_{0.75} (x_{L} - \frac{\overline{x}}{\overline{x}} + \frac{3}{8}c - \frac{1}{\overline{x}}\overline{c}) - \left( \frac{w_{1}}{UC_{L}} \right)_{0.25} (\frac{1}{\overline{c}^{2}} d\left( \frac{y}{s} \right), \quad (3.60)$$

where

and the second state of the second seco

$$\overline{\overline{c}} = \int_{0}^{1} \frac{c^{2}}{\overline{c}} d\left(\frac{y}{s}\right)$$
$$\overline{\overline{x}}_{L} = \int_{0}^{1} x_{L}\left(\frac{c}{\overline{c}}\right) d\left(\frac{y}{s}\right)$$

It may sometimes be convenient to use Equation (3.60) with a correction factor based on lifting-surface theory. In the light of Equation (3.36) for a small model, it may be simple to evaluate the quantity  $\lambda x_1/\bar{c}$  by both strip theory and lifting-surface theory. In general  $\lambda \partial c_1/\partial \alpha$  is the lift coefficient and  $x = x_1$  is the centre of lift corresponding to an incidence  $2x/\bar{c}$ , where x is measured from the theoretical aerodynamic centre. Thus a lifting-surface solution for the wing is steady pitching motion will suffice to determine an accurate value of  $\lambda z_1/\bar{c}$ . By strip theory

$$\left(\frac{\lambda_{L}}{c}\right)_{st} = 2 \int_{0}^{1} \left(\frac{x_{L} + \frac{1}{2}c}{\overline{c}}\right)^{2} \frac{c}{\overline{c}} d\left(\frac{y}{\varepsilon}\right) - 2\left(\frac{\overline{x}_{L} + \frac{1}{2}\overline{c}}{\overline{c}}\right)\left(\frac{\overline{x}_{L} + \frac{3}{2}\overline{\overline{c}}}{\overline{c}}\right), \quad (3.61)$$

which reduces to the second of Equations (3.33) for straight tapered wings. The ratio  $\lambda x_i/\tilde{c}$  to  $(\lambda x_i/\tilde{c})_{st}$  may be inserted as a correction factor on the right hand side of Equation (3.60); typical values may be obtained from Table 3.III. The relative accuracy of this method and of Equations (3.57) to (3.59) is discussed in Section 3.5.3.

Other measured quantities may require interference corrections. Although control surfaces are considered under asymmetrical configurations in Section 3.3.6, the approximate residual correction to hinge moment in Equation (3.77) also applies when there is spanwise symmetry. To apply a residual correction to surface pressure, it is necessary to use lifting-surface theory and to replace  $C_{\mu}$  by  $C_{p}$  in Equation (3.54). Reference 3.25 includes expressions for the interference corrections to downwash angle and wake displacement. Swanson and Schuldenfrei<sup>3.27</sup> consider these corrections in the presence of a slipstream behind a powered model.

## 3.3.5 Non-Symmetrical Spanwise Loading

We next consider symmetrically placed wings with antisymmetrical spanwise loading. The case of deflected allerons is treated approximately in References 3.26 and 3.28. Since Reference 3.28 is restricted to unswept wings, the expression from Equation (41) of Reference 3.26 is of greater generality. Hence

$$\frac{\Delta C_l}{C_l} = -\frac{\pi A \mu \partial C_l / \partial \alpha}{2(\pi A + \mu \partial C_l / \partial \alpha)} \int_c^1 \left( \frac{\mathbf{w}_1}{U C_l} \right)_{0.75} \mathbf{w} \frac{\mathbf{y}}{\mathbf{s}} d\left( \frac{\mathbf{y}}{\mathbf{s}} \right) . \quad (3.62)$$

where

and a super state of the state

$$\mu = \frac{2 + \sqrt{A^2(\beta^2 + \tan^2 \Lambda_{0.25})} \div 4}{2 + \sqrt{A^2(\beta^2 + \tan^2 \Lambda_{0.25})} + 16}$$

and W is the weighting factor in Equation (3.55) given by charts in Figure 6 of Reference 3.26. There will be a corresponding correction to the drag given by

$$\frac{\Delta C_{\rm D}}{C_l^2} = \frac{1}{2} \int_0^1 \left( \frac{\overline{\mathbf{w}}_1}{UC_l} \right)_{\infty} \frac{cC_{\rm LL}}{\overline{c}C_l} d\left( \frac{\overline{\mathbf{y}}}{\overline{\mathbf{s}}} \right) . \qquad (3.63)$$

It is now inappropriate to apply a correction to any measured geometric quantity such as aileron angle. In the case of a slowly rolling wing, however, 21: Regative

Contracting Contraction and an and

and the second s

correction  $\Delta C_1$  can be replaced by a positive correction to the rate of roll<sup>3.29</sup>. Similar considerations could apply to wings having smooth antisymmetrical modes of deformation.

Interference corrections for symmetrically placed, asymmetrically loaded wings are derived in Reference 3.18. In place of Equation (3.52) we may then write

$$\delta \alpha = \frac{\mathbf{w}_{i}}{\mathbf{y}} = \left(\frac{(\mathbf{w}_{i})_{S}}{\mathbf{w}_{L}}\right) \mathbf{C}_{L} + \left(\frac{(\mathbf{w}_{i})_{A}}{\mathbf{y}_{L}}\right) \mathbf{C}_{l} ; \qquad (3.64)$$

likewise the spanwise distribution of circulation may be written as

s.

$$\Gamma = 2sU\gamma = 2sU(\gamma_s + \gamma_s) , \qquad (3.65)$$

where

$$C_{L} = 2A \int_{0}^{1} \gamma_{s} d\left(\frac{y}{s}\right)$$

$$C_{l} = A \int_{0}^{1} r_{A} \frac{y}{s} d\left(\frac{y}{s}\right)$$
(3.66)

By the application of lifting-surface theory the symmetrical interference upwash angle  $(w_i)_S/U$  gives increments  $\delta C_L$  and  $\delta C_m$  to lift and pitching moment, and the antisymmetrical part  $(w_i)_A/U$  gives an increment  $\delta C_l$  to the rolling moment coefficient. Hence

$$\frac{\Delta \alpha}{C_{L}} = \frac{\delta C_{L}}{(\partial C_{L} / \partial \alpha) 2A \int_{0}^{1} \gamma_{s} d(y/s)} \left\{ \frac{\Delta C_{m}}{C_{L}} = -\frac{\delta C_{m}}{2A \int_{0}^{1} \gamma_{s} d(y/s)} + \frac{\Delta \alpha}{C_{L}} \frac{\partial C_{m}}{\partial \alpha} \right\}.$$

$$(3.67)$$

$$\frac{\Delta C_{l}}{C_{l}} = -\frac{\delta C_{l}}{A \int_{0}^{1} \gamma_{A}(y/s) d(y/s)}$$

The second of Equations (3.67) is equivalent to Equation (3.54), since

$$(\delta C_{L})_{I} = (\delta C_{R})_{I} = 0$$

for symmetrically placed wings. Equations (3.55), (3.57) and (3.62) are working approximations to Equations (3.67) and are recommended whenever it is inexpedient to calculate the increments  $\delta C_L$ ,  $\delta C_m$  and  $\delta C_l$  by lifting-surface theory.

The evaluation of the usual interference corrections to drag and yawing moment do not require calculations by lifting-surface theory, provided that the distributions of  $\gamma_{\rm S}$  and  $\gamma_{\rm A}$  can be estimated. These corrections are calculated on the principle<sup>\*</sup> that the local lift per unit span acts at right angles to a local effective stream direction, and that wall interference causes this to deflect upwards through the angle  $\frac{1}{2}(\delta \alpha)_{\infty}$ . Under linear conditions this deflection does not influence the lift itself, but reduces the drag per unit span by an amount

$$\frac{1}{2}\rho U \Gamma (\delta \alpha)_{\varpi} = \frac{1}{4}\rho U^2 S \left(\frac{A\gamma}{s}\right) (\delta \alpha)_{\varpi} .$$

Thus there are increments

$$\delta C_{DY} = -\frac{1}{2} \Lambda \int_{-1}^{1} \gamma (\delta \alpha)_{\infty} d\left(\frac{y}{s}\right)$$

$$\delta C_{DY} = \frac{1}{4} \Lambda \int_{-1}^{1} \gamma (\delta \alpha)_{\infty} \frac{y}{s} d\left(\frac{y}{s}\right)$$
(3.68)

to the wortex-induced drag and yawing moment coefficients. When  $\gamma$  and  $(\delta \alpha)_{\omega}$  are separated into their symmetrical and antisymmetrical parts, Equations (3.68) become

$$\delta C_{DV} = -\Lambda \int_{0}^{1} \left[ \gamma_{S} \left( \frac{(w_{1})_{S}}{I \mathcal{C}_{L}} \right)_{\infty}^{C} C_{L} + \gamma_{A} \left( \frac{(w_{1})_{A}}{U \mathcal{C}_{l}} \right)_{\infty}^{C} C_{l} \right] d\left( \frac{y}{s} \right)$$

$$\delta C_{DV} = \frac{1}{2} \Lambda \int_{0}^{1} \left[ \gamma_{A} \left( \frac{(w_{1})_{S}}{U \mathcal{C}_{L}} \right)_{\infty}^{C} C_{L} + \gamma_{S} \left( \frac{(w_{1})_{A}}{U \mathcal{C}_{l}} \right)_{\infty}^{C} C_{l} \right] \frac{y}{s} d\left( \frac{y}{s} \right)$$
(3.69)

When  $\gamma$  is replaced by  $cC_{LL}/4s$ , it is seen that  $-\delta C_{D\gamma}$  is equivalent to the sum of Equations (3.56) and (3.53). Only in the case of yawing moment do the symmetrical and an isymmetrical parts interact; the second of Equations (3.69) becomes

$$\frac{\delta C_{nv}}{C_L C_l} = \frac{1}{4} \int_0^1 \left[ \left( \frac{(w_1)_S}{U C_L} \right)_{\infty} \frac{c(C_{LL})_A}{\overline{c} C_l} + \left( \frac{(w_1)_A}{U C_l} \right)_{\infty} \frac{c(C_{LL})_S}{\overline{c} C_L} \right] \frac{y}{s} d\left( \frac{y}{s} \right), \quad (3.70)$$

where the right hand side is determined theoretically. It can easily be shown that  $\delta C_{DV}$  and  $\delta C_{DV}$  are unaffected by the application of the interference correction  $\Delta \alpha$ , since the reductions in magnitude due to the substitution  $\left(\frac{1}{2}(\delta \alpha)_{\infty} - \Delta \alpha\right)$  for  $\frac{1}{2}(\delta \alpha)_{\infty}$  in Equations (3.68) exactly cancel the increases due to the change of wind axes. But the result of residual corrections to the load distribution needs to be considered. As  $\Delta C_L = 0$ , any correction to  $\gamma_S$  will have a negligible effect on vortex-induced drag and yawing moment. The correction  $\Delta C_l$  implies a correction to  $\gamma_A$ , which may be regarded crudely as a factor  $\{1 + (\Delta C_l/C_l)\}$ ; this will increase  $(C_{DV})_A$ , the

• By consideration of the total energy of flow in the wake this principle can be justified in the case of drag. For yawing moment, however, the principle is not rigorous and numerical values of the interference correction are in some domits (Section 3.5 3).

vortex drag corresponding to the antisymmetrical part of the spanwise loading, by the factor  $\{1 + (\Delta C_l/C_l)\}^2$  and the total  $C_{nv}$  by the factor  $\{1 + (\Delta C_l/C_l)\}$ . The final corrections are therefore

$$\Delta C_{\rm D} = -\delta C_{\rm Dv} + 2 \frac{\Delta C_l}{C_l} (C_{\rm Dv})_{\rm A}$$

$$\Delta C_{\rm n} = -\delta C_{\rm nv} + \frac{\Delta C_l}{C_l} C_{\rm nv}$$
(3.71)

where  $\delta C_{DV}$  and  $\delta C_{nV}$  are given by Equations (3.69) and (3.70). The last terms in Equations (3.71) are approximate and relatively small and can usually be neglected, as in Equation (3.63). A full derivation of Equations (3.71) is found in Reference 3.18, but the latter is given incorrectly in Equation (8.13) of Reference 3.18. The magnitude of the corrections is discussed in Section 3.5.1.

References 3.23, 3.26 and 3.28 also consider the interference corrections for symmetrically placed, asymmetrically loaded wings. The rigorous treatment by liftingline theory is Reference 3.23 is beyond the scope of rapid calemiation. Like Reference 3.23, Grahamis<sup>3.26</sup> method is restricted to lifting-line theory and unswept wings in closed rectangular tunnels, but with acceptable approximations he derives corrections to rolling and yawing moment in a practical form. For wings with sweepback or large chord the interference corrections derived by Sivells and Salmi<sup>3.26</sup> are practically consistent with those of the previous paragraph, except that  $w_1/\bar{u}$  along the quarter-chord line replaces  $\frac{1}{2}(\delta \alpha)_m$  in the formulae concerning drag and yawing moment.

## 3.3.6 Asymmetrically Placed Wings

Nore difficult is the case of yaved wings when Equation (3.52) applies in all its generality. The walks also induce an interference sidewash unless the wing is situated in the horizontal plane of symmetry, but this is usually ignored. Provided that the angle of yaw is fairly small, simple generalizations of Equations (3.67) to (3.71) are possible. Added to Equations (3.67) there will be contributions  $\Delta \alpha/C_l$ ,  $\Delta C_m/C_l$  and  $\Delta C_l/C_L$ , which may be estimated by inserting the symmetrical part of  $(w_1/UC_1)$  in place of  $(w_1/UC_1)$  in Equations (3.55), (3.58) and (3.59) and also the antisymmetrical part of  $(w_1/UC_1)$  in place of  $(w_1/UC_1)$  in Equation (3.62). Both  $\Delta C_p$  and  $\Delta C_n$  will contain contributions proportional to  $C_L^2$ ,  $C_L C_l$  and  $C_l^2$ , since  $(\delta \alpha)_{\infty}$  in Equation (3.68) must be written in four parts, viz.,

$$(\delta \alpha)_{\infty} = \left[\frac{\{(\mathbf{w}_{1})_{L}\}_{S}}{UC_{L}}\right]_{\infty} C_{L} + \left[\frac{\{(\mathbf{w}_{1})_{L}\}_{A}}{UC_{L}}\right]_{\infty} C_{L} + \left[\frac{\{(\mathbf{w}_{1})_{l}\}_{S}}{UC_{l}}\right]_{\infty} C_{l} + \left[\frac{\{(\mathbf{w}_{1})_{l}\}_{A}}{UC_{l}}\right]_{\infty} C_{l} . \quad (3.72)$$

If, however, the angle of yaw is large, the information on lift distribution may be so scanty that a very simple vorter model will be adopted (Section 3.3.1). In Reference 3.25, Swanson represents the yawed wing as a number of skewed horse-shoe vortices corresponding to the equivalent wing semi-span of Equation (3.41) or the spanwise extent of the dihedral or aileron. Lengthy, but manageable, expressions are obtained for the interference corrections to incidence, drsg, rolling and yawing moments, but it is hard to assess their accuracy for a typical sweptback wing. The uncertainty is such that there may be little point in applying corrections to the moment coefficients  $C_1$  and  $C_m$ . It may suffice to calculate the interference upwash angles  $(\delta \alpha)_{0.75}$  and  $\frac{1}{2}(\delta \alpha)_{\infty}$ . Then, in place of Equation (3.55).

$$\Delta \alpha = \frac{1}{2} \int_{-1}^{1} (\delta \alpha)_{0.75} \, \mathbb{V} \, \mathrm{d} \left( \frac{y}{s} \right) \, . \tag{3.73}$$

and in place of Equations (3.56), (3.63) and (3.68),

$$\Delta C_{\rm D} = \frac{1}{US} \int_{-S}^{S} \Gamma(\delta \alpha)_{\infty} \, dy \qquad (3.74)$$
$$\Delta C_{\rm D} = \frac{1}{2USS} \int_{-S}^{S} \Gamma(\delta \alpha)_{\infty} y \, dy$$

where the distribution of circulation  $\Gamma = \Gamma(y)$  should be consistent with the measured lift and rolling moment. The extreme alternative is to calculate the distribution of  $\delta \alpha = w_i/U$  over the yawed planform and then to apply the best available theory, as envisaged in Section 3.3.3; this would seldem be justified.

Half-wing or reflection-plane models introduce further types of interference correction. All corrections to be applied for symmetrically loaded conditions are precisely those for a complete model mounted in a wind tunnel of the same height as the original tunnel and twice the breadth (Section 3.4.4); Equations (3.55) to (3.59) remain just as valid. But, as discussed in Reference 3.22, there is a second group of corrections for antisymmetrically loaded conditions when the reflection would not be present on a complete model. In such cases there may be large corrections to

half-wing rolling moment = 
$$\frac{1}{2}\rho U^2(\frac{1}{2}S)2sC_1$$
  
half-wing yawing moment =  $\frac{1}{2}\rho U^2(\frac{1}{2}S)2sC_n$  (3.75)

where  $\frac{1}{2}$ S is the area of the half-wing planform and 2s is the span of the complete wing. The corrections are determined in two parts. Firstly, the model and its reflection are corrected for symmetrical interference effects on the half wing. This involves the main correction to incidence from Equation (2.55) and corrections to lateral coefficients

$$\frac{\Delta C_{I}}{C_{L}} = -\frac{\frac{1}{2}\pi A \partial C_{L} / \partial \alpha}{\pi A + 2 \partial C_{L} / \partial \alpha} \int_{0}^{1} \left[ \left( \frac{\mathbf{w}_{I}}{UC_{L}} \right)_{0.75} - \frac{\Delta \alpha}{C_{L}} \right] \mathbf{w} \frac{\mathbf{y}}{\mathbf{s}} d\left( \frac{\mathbf{y}}{\mathbf{s}} \right) \right]$$

$$\frac{\Delta C_{n}}{C_{L}^{2}} = -\frac{1}{4} \int_{0}^{1} \left( \frac{\mathbf{w}_{I}}{UC_{L}} \right)_{\infty} \frac{cC_{LL}}{cC_{L}} \frac{\mathbf{y}}{\mathbf{s}} d\left( \frac{\mathbf{y}}{\mathbf{s}} \right)$$

$$(3.76)$$

Apart from the effects of streamline curvature. the residual spanwise loading is distributed along the quarter-chord line according to the assumptions of Reference 3.26. Since the corresponding moment on the half wing about an axis parallel to the quarter-chord line should vanish, Equations (3.58) and (3.76) are consistent in that

$$2s\Delta C_1 = -\tilde{c}(\Delta C_n), \cot \Lambda_{0,25}$$

116

Secondly and of greater importance, factors independent of the tunnel cross-section must be applied to the quantities  $(C_l + \Delta C_l)$  and  $(C_n + \Delta C_n)$  to account for the reflection plane, and these demand special theoretical calculations. Simplified treatments are discussed in References 3.4, 3.22 and 3.26.

Large half-wing models are often used to test control surfaces, for which the streamline curvature is particularly significant<sup>3.\*</sup>. In the present notation the approximate residual correction to hinge moment becomes

$$\frac{\Delta C_{H}}{C_{L}} = -\frac{\partial C_{H}}{\partial \alpha} \frac{\int \left[ B\left(\frac{\mathbf{w}_{i}}{UC_{L}}\right)_{0.75} - (B-1)\left(\frac{\mathbf{w}_{i}}{UC_{L}}\right)_{0.25} - \frac{\Delta \alpha}{C_{L}} \right] \mathbf{w} c_{f}^{2} d\left(\frac{\mathbf{y}}{\mathbf{s}}\right)}{\int \mathbf{w} c_{f}^{2} d\left(\frac{\mathbf{y}}{\mathbf{s}}\right)}, \quad (3.77)$$

where the weighting<sup>\*</sup> W is selected from Equation (3.55), the integrais are taken over the span of the control surface,  $c_1(y)$  is the local control chord, and for unbalanced control surfaces B is tabulated below.

c <sub>f</sub> /c	0	0.65	ē. 19	0.15	0.20	0.25	0.30	0.35	0.40	0.45	0.50
В	2.50	2.46	2.41	2.37	2.33	2.28	2.24	2. 19	2. 15	2. 10	2.05

Values of B for balanced control surfaces having set-back hinges may be evaluated as  $\frac{1}{2} + \frac{1}{4}(b'/b_1)$  from Table 2 in Section 4.2 of Reference 3.4. For satisymmetrically loaded conditions a factor should be applied to  $(C_H + \Delta C_H)$  to account for the reflection plane. For outboard allerons this will be less important than the corresponding factors for relias and yawing moments, so that an estimate by lifting-line theory may suffice.

The interpretation of an interference correction may need adjustment if the lift or rolling moment is not measured. The uncertainty concerns which part of the expression is calculated theoretically and which part is determined experimentally. For example, Equation (3.77) envisages that  $C_{1}$  on the left hand side and  $\partial C_{H}/\partial \alpha$  on the right hand side are measured, while the integrals are predetermined. If, however, only the control hinge noment is measured, it would be necessary to multiply both sides of the equation by an estimated value of  $C_{L}/C_{H}$ ; then the ratio of  $\Delta C_{H}/C_{H}$  to  $\partial C_{H}/\partial \alpha$ is determined by calculation.

#### 3.4 EVALUATION OF INTERFERENCE PARAMETERS

In order the apply the general methods of Section 3.3, it is necessary to obtain numerical data for the interference upwash  $w_i$  due to lifting elements in a particular tunnel. A wide range of tunnel cross-section has been employed in research centres,

<sup>6</sup> Equation: (3.77) gives a rough average value of the expression in square brackets. Although by similarized the registing Y should be replaced by the two-dimensional  $\partial c_H/\partial u$  in each integrand, the major influence of boundary layers may override this consideration.

117

MONTAN

القشمه فالملية وكالمنظف شكايطه وبالمناق أأرابه وبالمشب

and the real of a land we have a first free of the second of the

the most common shapes having closed rectangular boundaries (Section 3.4.1) or closed octagonal boundaries (Section 3.4.2); the latter are usually regarded as rectangular tunnels with small corner fillets. Much less attention has been given to open rectangular tunnels, as the rounded circular or elliptical shape is more appropriate for a free jet; these are considered as well as closed circular and elliptical tunnels in Section 3.4.3. As in the case of rectangular tunnels (Section 3.2.2), some curved shapes with mixed boundaries have been devised to reduce or eliminate the theoretical lift interference, but these semi-closed tunnels are of passing academic interest. Wind-tunnel testing of half models mounted on a reflection plane is common practice. Although this merely doubles the effective breadth to height ratio of rectangular tunnels and extends the practical upper limit of b/h. in the case of octagonal and circular tunnels (Section 3.4.3).

Table 3.1V summarizes the numerical data available for wings of moderate or large span. In many cases the information is limited to quantities such as  $(\delta_0)_{ij}$ ,  $(\delta_2)_{g}$  and  $\delta_0(\mathbf{y}, \mathbf{t})$ , as defined in Equations (3.44), (3.46) and (3.42), which are only sufficient if the wing can be regarded as an unswept lifting line. A tick in the final column of Table 3.1V indicates that the streamwise variation of  $\mathbf{w}_i$  has been culculated, whether through the quantity  $\delta_1(\mathbf{y}, \mathbf{t})$  of Equation (3.42) or some more general tabulation or graph.

# 3.4.1 Bectangular Tannels

The evaluation of upwash interference on large lifting models in closed rectangular tunnels is basically straightforward, since there is a complete image system corresponding to any vortex representation of the model. The upwash field of a horse-shoe vortex of strength X surrounding the area x > 0. |y| < t, z = 0 is

$$\mathbf{w} = \mathbf{w}_{\mathbf{y}}(\mathbf{x}, \mathbf{y}, \mathbf{z}, \mathbf{t}) = \frac{\mathbf{K}}{4\pi} \left[ \mathbf{w}(\beta^{-1}\mathbf{x}, \mathbf{y} + \mathbf{t}, \mathbf{z}) - \mathbf{w}(\beta^{-1}\mathbf{x}, \mathbf{y} - \mathbf{t}, \mathbf{z}) \right], \quad (3.73)$$

where

$$\overline{z}(x,y,z) = -\frac{y}{y^2 + z^2} - \left(\frac{1}{x^2 + z^2} + \frac{1}{y^2 + z^2}\right) \frac{xy}{(x^2 + y^2 + z^2)^{\frac{1}{2}}} \quad . \quad (3.79)$$

Its image system is a simple generalization of Figure 3.1(b), so that the interference upwash is

$$\mathbf{w}_{1} = \frac{\pi}{4\pi} \sum_{-\infty}^{\infty} \sum_{-\infty}^{\tau} (-1)^{n} [\pi(\beta^{-1}\mathbf{x}, \mathbf{y} - \mathbf{ab} + \mathbf{t}, \mathbf{z} - \mathbf{nh}) - \pi(\beta^{-1}\mathbf{x}, \mathbf{y} - \mathbf{ab} - \mathbf{t}, \mathbf{z} - \mathbf{nh})], \quad (3.89)$$

where, as in Equation (3.13).  $\sum_{\infty}^{\infty} \sum_{n=0}^{\infty} i^n$  denotes that  $(\mathbf{x}, \mathbf{n})$  takes all possible integral pairs except (0, 0).

If the wing is of large span 2s but of small chord, it may suffice to put x = z = 0 in Equation (3.80), so that in the notation of Equation (3.42)

$$\mathbf{x}_{i} = \frac{4Kt}{bb} \delta_{0}(\mathbf{y}, \mathbf{t}) . \qquad (3.81)$$

CONSCIENTS?

שביילואראשושי איאוי דיאופן ע לינוט אישי

$$\delta_{0}(y,t) = \frac{bh}{16\pi t} \sum_{-\infty}^{\infty} (-1)^{n} \left[ -\frac{y-ab+t}{(y-ab+t)^{2}+n^{2}h^{2}} + \frac{y-ab-t}{(y-ab-t)^{2}+n^{2}h^{2}} \right]$$
  
=  $-\frac{bh}{16\pi t} \left[ \frac{1}{y-t} - \frac{1}{y+t} \right] + \frac{b}{16t} \sum_{a=-\infty}^{\infty} \left[ \operatorname{cosech} \frac{\pi(y-ab-t)}{h} - \operatorname{cosech} \frac{\pi(y-ab+t)}{h} \right].$  (3.82)

Hence the mean interference parameters  $(\delta_0)_U$  and  $(\delta_0)_E$  for uniform and elliptic spanwise loading in respective Equations (3.44) and (3.46) can be evaluated. As described in Section 9 of Reference 3.1, the early literature has shown that the results can be derived analytically, viz..

$$(\delta_0)_{ij} = \frac{h}{4\pi b\sigma^2} \log_e \frac{\pi\sigma}{\sin\pi\sigma} + \frac{\pi h}{b} \sum_{n=1}^{\infty} \frac{n}{e^{2\pi nb/b} + 1} \left(\frac{\sin\pi n\sigma}{\pi n\sigma}\right)^2 \qquad (3.83)$$

and

where

$$(\delta_0)_{g} = \frac{h}{2b} P(\sigma) + \frac{4\pi h}{b} \sum_{n=1}^{\infty} \frac{n}{e^{2\pi nh/b} + 1} \left( \frac{J_1(\pi n\sigma)}{\pi n\sigma} \right)^2$$
 (3.84)

where  $\sigma = 2s/b$ ; numerical values of F( $\sigma$ ) and  $\{J_1(\pi\sigma)/\pi\sigma\}^2$  are found in Tables 4 and 5 of Reference 3.1. A good description of the mathematical analysis is given by Sanders and Pounder in Section 2.1 of Reference 3.23. They also give useful graphs of  $\{\delta_0\}_{\mathbb{D}}$  and  $\{\delta_0\}_{\mathbb{E}}$  against h/b for  $\sigma = 0, 0.2, 0.4, 0.6, 0.8$  and 0.9; the latter is given in carpet form in Figure 3.9. The factor  $\{\delta_0\}_{\mathbb{E}}(\sigma)/(\delta_0)_{\mathbb{E}}(\theta)$  provides a simple improvement on the corrections to incidence and drag coefficient for a small wing in Equations (3.39); thus

$$\Delta \alpha = (\delta_0)_{E}(\sigma) \left( 1 + \frac{\bar{c}\delta_1}{2\beta h \delta_0} \right) \frac{SC_L}{C}$$

$$\Delta C_D = (\delta_0)_{E}(\sigma) \frac{SC_L^2}{C}$$
(3.85)

where  $(\delta_c)_E(\sigma)$  is given in Figure 3.9,  $\delta_1/\delta_0$  is given by the upper curve of Figure 3.4, and Figure 3.7 indicates a correction factor  $\lambda \overline{c}/\overline{c}$  to the term involving  $\delta_1$ .

It is often necessary to calculate the parameter  $\delta_1(y,t)$  of Equation (3.42). By Equation (3.80) this is readily expressed as

$$\delta_{1}(y,t) = \frac{\beta ch^{2}}{4Kt} \left( \frac{\partial w_{1}}{\partial x} \right)_{x=0}$$

$$= \frac{bh^{2}}{16\pi t} \sum_{-\infty}^{\infty} (-1)^{n} \left[ \frac{\partial w}{\partial x} (0, y - sb + t, -nh) - \frac{\partial w}{\partial x} (0, y - sb - t, -nh) \right], \quad (3.86)$$

where

1000

$$\frac{\partial \mathbf{w}}{\partial \mathbf{x}} (0, \mathbf{y}, -\mathbf{nh}) = - \frac{\mathbf{y}(\mathbf{y}^2 + 2\mathbf{n}^2\mathbf{h}^2)}{\mathbf{n}^{2}\mathbf{h}^{2}(\mathbf{y}^2 + \mathbf{n}^{2}\mathbf{h}^{2})^{3/2}}.$$

The singularity from n = 0 is illusory, since in the limit

T CARDONAL -

$$\frac{\partial \mathbf{w}}{\partial \mathbf{x}} (0, \mathbf{y} - \mathbf{ab} + \mathbf{t}, 0) - \frac{\partial \mathbf{w}}{\partial \mathbf{x}} (0, \mathbf{y} - \mathbf{ab} - \mathbf{t}, 0) = \frac{1}{2} \left[ \frac{1}{(\mathbf{y} - \mathbf{ab} + \mathbf{t})^2} - \frac{1}{(\mathbf{y} - \mathbf{ab} - \mathbf{t})^2} \right]$$

according as a is positive or negative. The evaluation of  $w_i$  for a duplex tunnel (b = 2h) was accomplished by Cowley and McMillan (Ref.3.30; 1934), who tabulated upwash angle at the tail of an elliptically loaded model without the approximation that W is linear in x. They obtained the equivalent of

$$\frac{4}{\pi}\int_{0}^{1}\delta_{1}(0,t)\left(\frac{t}{s}\right)^{2}\left\{1-\left(\frac{t}{s}\right)^{2}\right\}^{-\frac{1}{2}}d\left(\frac{t}{s}\right)$$

for a range of  $\sigma = 2\pi/b$  by expanding  $\delta_1(0, t)$  in even powers of t/s and neglecting contributions from  $|\mathbf{n}| > 3$  and from  $|\mathbf{n}| > 5$  in Equation (3.86). As it stands the double series converges slowly, but it can be transformed into the rapidly convergent expression in Equation (3.25) of Reference 3.23. Hence

$$\delta_{1}(\mathbf{y},t) = \frac{\pi}{24} + \frac{2h^{2}}{bt} \sum_{\mathbf{y}=1}^{\infty} \sum_{q=1}^{\infty} (-1)^{q-1} \mathbf{y} \sin \frac{2\pi p t}{b} \cos \frac{2\pi p \mathbf{y}}{b} \chi \left(\frac{2\pi p q h}{b}\right) + \frac{h^{2}}{32\pi bt} \left[ \Psi' \left(1 + \frac{\mathbf{y} - t}{b}\right) - \Psi' \left(1 + \frac{\mathbf{y} + t}{b}\right) + \Psi' \left(1 - \frac{\mathbf{y} + t}{b}\right) - \Psi' \left(1 - \frac{\mathbf{y} - t}{b}\right) \right].$$
(3.87)

where  $\tilde{\Psi}'(1 + \eta) = \sum_{m=1}^{\infty} \frac{1}{(\eta + m)^2}$  is the Trigannia function<sup>\*</sup> and  $\chi(\eta) = \tilde{\kappa}_q(\eta) + \frac{1}{\eta} \kappa_1(\eta)$ 

in terms of modified Bessel functions of the second kind. It can be seen that Equation (3.87) reduces to the first of Equations (3.20) as both y and t tend to zero. A rigorous derivation of the rapidly convergent series has been given by Olver<sup>3.33</sup>, who was responsible for the tabulation of  $\delta_1(3, t)$  for b/h = 1, 9/7, 2,18/7 in Beference 3.17.

When the interference upwash is to be calculated from Equation (3.47) or (3.48). it is convenient to tabulate the quantities

$$\frac{\partial}{\partial t} \{ t \delta_0(y,t) \} \text{ and } \frac{\partial}{\partial t} \{ t \delta_1(y,t) \}.$$

 P has been tabulated by H.T.Davis in Tables of the Higher Nathenatical Functions, Volume II (The Principal Press Inc., Indiana, USA, 1935).

い、法語

It follows from Equations (3.82) and (3.87) that both quantities are even functions of y and t; moreover, they are unaltered when y and t are interchanged. The diagonal symmetry is illustrated in Table 3.V, where both quantities are given for a closed square tunnel. Another account of the theory is given in Reference 3.18 by the present author and Acum. It is shown that

~

$$\frac{\partial}{\partial t} \left\{ t \delta_0(\mathbf{y}, \mathbf{t}) \right\} = \frac{b}{16h} \left[ \sum_{\mathbf{n}=1}^{\infty} \left\{ -\tilde{\mathbf{f}}_1 \left( \frac{\mathbf{y} - \mathbf{t} - \mathbf{n} \mathbf{b}}{h} \right) - \mathbf{f}_1 \left( \frac{\mathbf{y} - \mathbf{t} + \mathbf{n} \mathbf{b}}{h} \right) - \mathbf{f}_1 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{n} \mathbf{b}}{h} \right) - \mathbf{f}_1 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{n} \mathbf{b}}{h} \right) \right] - \mathbf{f}_1 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{n} \mathbf{b}}{h} \right) - \mathbf{f}_2 \left( \frac{\mathbf{y} - \mathbf{t}}{h} \right) - \mathbf{f}_2 \left( \frac{\mathbf{y} - \mathbf{t}}{h} \right) \right]$$
(3.88)

and

$$\frac{\partial}{\partial t} \{t\delta_{1}(y,t)\} = \frac{b}{16\pi h} \left[ \sum_{R=1}^{\infty} \left\{ -f_{3}\left(\frac{y-t-ab}{h}\right) - f_{3}\left(\frac{y-t+ab}{h}\right) - f_{3}\left(\frac{y+t-ab}{h}\right) - f_{3}\left(\frac{y+t-ab}{h}\right) \right\} - f_{3}\left(\frac{y+t-ab}{h}\right) \right\} - f_{3}\left(\frac{y-t}{h}\right) - f_{4}\left(\frac{y+t}{h}\right) \right], \quad (3.89)$$

where

$$f_{1}(\lambda) = \frac{d}{d\lambda} \{ \operatorname{cosech} \pi \lambda \}$$

$$f_{2}(\lambda) = \frac{d}{d\lambda} \left\{ \operatorname{cosech} \pi \lambda - \frac{1}{\pi \lambda} \right\}$$

$$f_{3}(\lambda) = \frac{1}{\lambda} f(\lambda) + \frac{d}{d\lambda} \{ f(\lambda) \}$$

$$f_{4}(\lambda) = \frac{1}{\lambda} f(\lambda) + \frac{d}{d\lambda} \{ f(\lambda) \} + \frac{1}{|\lambda|^{3}}$$

$$f(\lambda) = 4\pi \{ K_{1}(\pi \lambda) + 3K_{1}(3\pi \lambda) + 5K_{1}(5\pi \lambda) - \ldots \}$$

$$(3.90)$$

with

are found in Tables 1, 2 and 3 of Reference 3.18. The corresponding interference parameters from the antisymmetrical part of the spanwise loading are defined in Equation (3.49), viz.,

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{4\mathbf{K}\mathbf{t}}{\mathbf{U}\mathbf{b}\mathbf{h}} \left\{ \kappa_{0}(\mathbf{y},\mathbf{t}) + \frac{\mathbf{x}}{\beta\mathbf{h}} \kappa_{1}(\mathbf{y},\mathbf{t}) \div 0 \left(\frac{\mathbf{x}}{\beta\mathbf{h}}\right)^{3} \right\} .$$

Then the antisymmetrical interference upwash in Equation (3.50) corresponding to the vortex model of Figure 3.8(d) involves the parameters



$$\frac{\partial}{\partial t} \{ t\kappa_0(\mathbf{y}, \mathbf{t}) \} = \frac{b}{16h} \left[ \sum_{\mathbf{z}=1}^{\infty} (-1)^{\mathbf{z}} \left\{ -f_1 \left( \frac{\mathbf{y} - \mathbf{t} - \mathbf{z} \mathbf{b}}{h} \right) - f_1 \left( \frac{\mathbf{y} - \mathbf{t} + \mathbf{z} \mathbf{b}}{h} \right) + f_1 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{z} \mathbf{b}}{h} \right) + f_1 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{z} \mathbf{b}}{h} \right) \right] + f_1 \left( \frac{\mathbf{y} + \mathbf{t} + \mathbf{z} \mathbf{b}}{h} \right) \right] - f_2 \left( \frac{\mathbf{y} - \mathbf{t}}{h} \right) + f_2 \left( \frac{\mathbf{y} + \mathbf{t}}{h} \right) \right]$$
and

and

$$\frac{\partial}{\partial t} \left\{ t \kappa_1(\mathbf{y}, \mathbf{t}) \right\} = \frac{b}{16\pi h} \left[ \sum_{\mathbf{y}=1}^{\infty} (-1)^{\mathbf{y}} \left\{ -f_3 \left( \frac{\mathbf{y} - \mathbf{t} - \mathbf{m}b}{h} \right) - f_3 \left( \frac{\mathbf{y} - \mathbf{t} + \mathbf{m}b}{h} \right) + \hat{f}_3 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{m}b}{h} \right) + f_3 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{m}b}{h} \right) + f_3 \left( \frac{\mathbf{y} + \mathbf{t} - \mathbf{m}b}{h} \right) \right] + f_3 \left( \frac{\mathbf{y} + \mathbf{t} + \mathbf{m}b}{h} \right) \left\{ -f_4 \left( \frac{\mathbf{y} - \mathbf{t}}{h} \right) + f_4 \left( \frac{\mathbf{y} + \mathbf{t}}{h} \right) \right\}, \quad (3.92)$$

which may be computed for any rectangular tunnel from Equations (3.90) for  $f_1(\lambda)$  and  $f_2(\lambda)$  and Table 3 of Reference 3.18 from which, with the use of second differences,  $f_u(\lambda)$  may be obtained to four decimal places.

For models in the central horizontal plane z = 0,  $w_i$  can be evaluated quite easily from Equations (3.79) and (3.80) for a general value of x. The terms of order  $(x/\beta h)^3$  will become significant if the model is of great streamwise extent, for example, a large sweptback wing or a complete aircraft. Moreover, the errors in assuming a linear upwash will increase hearly sevenfold as the Mach number increases from 0 to 0.85. Furthermore, the foregoing method cannot be applied to yawed or asymmetrical models. In such cases, or when higher order terms are required, it is convenient to replace the distributed vorticity by point concentrations as in Figure 3.8(f). Following Katzoff and Hannah<sup>3.20</sup>, we replace Equation (3.78) by the upwash field due to a line doublet of strength

 $P = \lim_{t \to 0} (2Kt) = L/(\rho t)$ 

extending along the x-axis from the origin to infinity. This upwash field is

$$\mathbf{r} = \frac{\mathbf{P}}{4\pi} \frac{\partial \mathbf{\bar{x}} (\beta^{-1} \mathbf{z}, \mathbf{y}, \mathbf{z})}{d\mathbf{y}} , \qquad (3.93)$$

where, from Equation (3.79),

$$\frac{\partial}{\partial y} \left[ W(x, y, z) \right] = \frac{y^2 - z^2}{(y^2 + z^2)^2} + \frac{x \left[ (y^2 + z^2) (x^2 + y^2 - 2z^2) - 2x^2 z^2 \right]}{(y^2 + z^2)^2 (x^2 + y^2 + z^2)^{3/2}}, \quad (3.94)$$

Now consider a number of lifting elements  $L_N = \rho DP_N$  at positions  $(x_N, y_N, 0)$  in a closed rectangular tunnel. In accord with the image system in Figure 2(a) of Reference 3.20, the interference upwash is expressed as

$$W_{1}(\mathbf{x},\mathbf{y},0) = \sum_{N} \frac{L_{N}}{4\pi\rho U} \left\{ \sum_{-\infty}^{\infty} (-1)^{n} \frac{\partial}{\partial y} \left[ W \left( \frac{\mathbf{x}-\mathbf{x}_{N}}{\beta}, \mathbf{y}-2\mathbf{n}\mathbf{b}-\mathbf{b}+\mathbf{y}_{N}, -\mathbf{n}\mathbf{h} \right) \right] + \sum_{-\infty}^{\infty} (-1)^{n} \frac{\partial}{\partial y} \left[ W \left( \frac{\mathbf{x}-\mathbf{x}_{N}}{\beta}, \mathbf{y}-2\mathbf{n}\mathbf{b}-\mathbf{y}_{N}, -\mathbf{n}\mathbf{h} \right) \right] \right\}$$

This is conveniently written in non-dimensional form

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \sum_{\mathbf{N}} \frac{\mathbf{L}_{\mathbf{N}}}{\rho \mathbf{U}^{2} \mathbf{b}^{2}} \left[ \mathbf{w}_{\mathbf{A}} \left( \frac{\mathbf{x} - \mathbf{x}_{\mathbf{N}}}{\rho \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{b} + \mathbf{y}_{\mathbf{N}}}{\mathbf{b}} \right) + \mathbf{w}_{\mathbf{A}} \left( \frac{\mathbf{x} - \mathbf{x}_{\mathbf{N}}}{\rho \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{y}_{\mathbf{N}}}{\mathbf{b}} \right) + \mathbf{w}_{\mathbf{B}} \left( \frac{\mathbf{x} - \mathbf{x}_{\mathbf{N}}}{\rho \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{b} + \mathbf{y}_{\mathbf{N}}}{\mathbf{b}} \right) \right].$$

$$(3.95)$$

where

The second second second second second

$$\mathbb{W}_{A}\left(\frac{x}{b}, \frac{y}{b}\right) = \frac{b^{2}}{4\pi} \sum_{-\infty}^{\infty} (-1)^{n} \frac{\partial}{\partial y} [\mathbb{W}(x, y - 2nb, -nh)] \\
\mathbb{W}_{B}\left(\frac{x}{b}, \frac{y}{b}\right) = \frac{b^{2}}{4\pi} \frac{\partial}{\partial y} [\mathbb{W}(x, y, 0)]$$
(3.96)

Thus  $w_i/U$  may be evaluated from charts of the functions  $W_A$  and  $W_B$ . The former chart A is illustrated for ratios b/h = 2/5, 10/7 and 2 respectively in Figures 7(b), 7(c) and 7(a) of Reference 3.20. The latter chart B in Figure 5 of Reference 3.20 is independent of tunnel shape. Katzoff and Hannah describe in detail their graphical procedure for the computation of  $w_i$  on a gaved lifting wing. They also give useful advice on the calculation and preparation of charts with particular reference to rectangular tunnels with one or more of their sides open. Any such cunfiguration can be treated by means of chart A dependent cn b/h and the floor and roof condition and the universal chart B, but there may be changes of sign in Equation (3.95). To expedite the calculation of chart A, a horizontal row of sufficiently distant image doublets  $(|\mathbf{n}| \ge 2, \text{ say})$  may be replaced by a row of horse-shoe vortices of span 2b, so that all the trailing vortices except the innermost ones cancel in pairs; for large enough n ( $|n| \ge 2$ , say) the upwash field from the complete row of doublets is approximately that of a two-dimensional bound vortex extending from  $y = -\infty$ to  $y = +\infty$ , as illustrated in Figure 3.10(a). In the case of closed side-walls and open floor and roof (type (4) of Section 3.2.2), when all the images are of the same sign, Katzoff and Kannah suggest that, instead of being extended horizontally, the distant image doublets are extended vertically into a source line and a sink line at a distance h spart. The source and sink lines in any column cancel in pairs, and only those at a distance  $\frac{1}{2}h$  above or below the inner group of doublets ( $|\mathbf{m}| \leq 1$ ,  $|\mathbf{m}| \leq 1$ ) remain, as illustrated in Figure 3.10(b). Although the image system for a rectangular tunnel of type (4) is not valid, the procedure adopted here amounts to an approximate summation column by column (first with respect to n ); it follows from the discussion above Equation (3.17) that this leads to a correct approximation while the alternative array of Figure 3.10(c) would fail.

A further complication may arise when the displacement of the model from the central horizontal plane z = 0 can no longer be ignored. This can result from long models at high incidence or when the floor of the tionel is used to simulate the ground. In the former case the interference corrections can be large and changes in the distance d of the lifting element from the floor can be significant. as has already been seen for a small wing in Figure 3.3. In the latter case  $(\frac{1}{2}h - d)$  may be even larger, and the interference corrections, though smaller than usual because the principal image in the floor is not included, will again depend on d. Silverstein and White (Ref. 3.32; 1935) made an early contribution in this field, mainly with regard to interference effects on the downwash at a tail plane (Section 3.6.2). Brown<sup>3.33</sup> derives general formulae for  $(\delta_0)_{\rm U}$  and  $(\delta_0)_{\rm Ug}$ , the corresponding corrections to wind-tunnel experiments in ground effect, and he gives results for b/h = 2 and 4. Some values for the duplex tunnel are reproduced in Figure 3.11. In a typical case (d = 0.25h), the principal image in the floor accounts for a large proportion.  $\{(\delta_0)_U - (\delta_0)_{Ug}\}/(\delta_0)_U$ . of the interference, but the ratio  $(\delta_{g})_{Ug}/(\delta_{g})_{U}$  increases from 0.036 to 0.118 as  $\sigma$  increases from 0 to 0.6. Provided that d/h is fairly small, crude estimates of  $(\delta_{5})_{\text{Ug}}$  should suffice (Section 3.6.3).

The quantities  $\delta_0(y,t)$  and  $\delta_1(y,t)$  for off-centre models are formulated in Appendix II of Reference 3.4, and a simple approximation to the double summation for  $\delta_1(y,t)$ , based on Reference 3.33, is included. A full mathematical discussion of  $\delta_0(y,t)$  and interference effects at a lifting line displaced vertically from the centre line of the tunnel is given by Sanders and Pounder in References 3.23 and 3.24. They give expressions in Equations (1.51) and (1.52 to 1.57) of Reference 3.23 for the interference velocities  $v_i$  and  $u_i$  respectively. The effect of the sidewash  $v_i$ on the aerodynamic forces is unknown, but  $u_i$  will give a correction  $\Delta U$  to the stream velocity akin to that in Figure 3.3(b).

In practice, the distance d between model and floor will depend on x, so that the complete evaluation of streamline curvature would involve displacements in z. This effect has been considered in References 3.32 and 3.27 in relation to the downwash field near a tail by introducing two extra parameters d/h and d'/h corresponding to the vertical locations of the lifting element and the required  $w_i$ . In the approximate representation of Figure 3.10(a), the odd rows would be lowered by the distance  $(d - \frac{1}{2}h)$  and the even rows would be raised by the same amount. Although it is inadmissible so to split the image system in two parts, just as Figure 3.10(c) is incorrect, it is possible to construct the required upwash field from that of an isolated doublet and that of Figure 3.10(b) with double the vertical spacing. We need the complete upwash field of a doublet placed centrally in a rectangular tunnel of breadth 2b, height 2h, with closed sides but open roof and floor

$$W_{C}\left(\frac{x}{b}, \frac{y}{b}, \frac{z}{b}\right) = \frac{b^{2}}{4\pi} \sum_{-\infty}^{\infty} \frac{\partial}{\partial y} \left[W(x, y-2nb, z-2nb)\right] , \qquad (3.97)$$

where  $\partial W/\partial y$  is defined in Equation (3.94) and the summation is made first with respect to n. The final generalization of Equation (3.95) is

$$\frac{\mathbf{Y}_{1}}{\mathbf{U}} = \sum_{N} \frac{\mathbf{L}_{N}}{\beta \mathbf{U}^{2} \mathbf{b}^{2}} \left\{ -\frac{\mathbf{b}^{2}}{4\pi} \frac{\partial}{\partial \mathbf{y}} \left[ \frac{\mathbf{W}}{\mathbf{W}} \left( \frac{\mathbf{x} - \mathbf{x}_{N}}{\beta} \cdot \mathbf{y} - \mathbf{y}_{N}, \mathbf{z} - \mathbf{z}_{N} \right) \right] + \frac{\mathbf{W}_{C}}{\mathbf{v}} \left( \frac{\mathbf{x} - \mathbf{x}_{N}}{\beta \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{y}_{N}}{\mathbf{b}} \cdot \frac{\mathbf{z} - \mathbf{z}_{N}}{\mathbf{b}} \right) - \mathbf{W}_{C} \left( \frac{\mathbf{x} - \mathbf{x}_{N}}{\beta \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{y}_{N}}{\mathbf{b}} \cdot \frac{\mathbf{z} - \mathbf{h} + \mathbf{z}_{N}}{\mathbf{b}} \right) + \frac{\mathbf{W}_{C}}{\mathbf{v}} \left( \frac{\mathbf{x} - \mathbf{x}_{N}}{\beta \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{y}_{N}}{\mathbf{b}} \cdot \frac{\mathbf{z} - \mathbf{z}_{N}}{\mathbf{b}} \right) - \mathbf{W}_{C} \left( \frac{\mathbf{x} - \mathbf{x}_{N}}{\beta \mathbf{b}} \cdot \frac{\mathbf{y} - \mathbf{y} + \mathbf{y}_{N}}{\mathbf{b}} \cdot \frac{\mathbf{z} - \mathbf{h} + \mathbf{z}_{N}}{\mathbf{b}} \right) \right\}. \quad (3.98)$$

125

Since an approximation to Equation (3.97) may be made on the basis of Figure 3.10(b) with double the vertical spacing, it may well be practicable and desirable to use Equation (3.98) for  $\sigma$  very long and slender model at high incidence with the simplification that  $y = y_N = 0$ , while  $(z - z_N)/b$  may be far from small.

## 3.4.2 Octagonal Tunnels

Unlike rectangular tunnels, an octagonal tunnel has no exact image system. Any attempt to construct such a system for a closed boundary containing an obtuse angle is thwarted sooner or later by the demand for images within the boundary. The treatment of octagonal tunnels is necessarily approximate, and artificial methods have to be introduced.

In principle, the two-dimensional problem of obtaining  $\delta_0(\mathbf{y}, \mathbf{t})$  can be solved by transforming the interior of the octagon into the helf plane; the Schwarz-Christoffel theorem achieves this, but the algebra is prohibitive for a general octagonal boundary. Gent<sup>3-13</sup> has derived the appropriate transformation for a regular octagon of breadth b.

$$\omega = 1.7191 \left(\frac{1}{2}b\right) \int_{0}^{\Omega} \prod_{p=1}^{4} \left(\Omega_{p}^{2} - \Omega^{2}\right)^{-\frac{1}{4}} d\Omega \qquad (3.99)$$

with  $\Omega_1 = \Omega_3^{-1} = 0.1989$  and  $\Omega_2 = \Omega_3^{-1} = 0.6682$ , so that the real axis in the  $\Omega = X + iZ$  plane represents the octagon in the  $\omega$  plane. A convenient alternative is to map the octagon on a unit circle by the related transformation

$$\omega = \frac{ib}{\kappa} \int_{-1}^{\Omega} \frac{d\Omega}{(1+\Omega^8)^{\frac{1}{4}}} .$$
 (3.100)

where  $\kappa = 1.9565$ . This is written as a rapidly converging power series in  $\Omega$ , which is inverted to give

$$\Omega = \left(\frac{\omega\kappa}{1b} - \frac{\kappa}{2}\right) + \frac{1}{36}\left(\frac{\omega\kappa}{1b} - \frac{\kappa}{2}\right)^9 - \frac{11}{4896}\left(\frac{\omega\kappa}{1b} - \frac{\kappa}{2}\right)^{17} + \dots$$

A vortex pair of strengths  $\pm K$  at  $\omega = i(\frac{1}{2}b \pm t)$  and a point  $\omega = i(\frac{1}{2}b \pm y)$  transform into a vortex pair of strengths  $\pm K$  at  $\Omega = \pm T$  and a point  $\Omega = X$  in the plane of the unit circle. Hence the upwash interference parameter in the  $\omega$  plane of the octagon becomes

$$\delta_{0}(\mathbf{y}, \mathbf{t}) = \frac{C\mathbf{w}_{1}}{4Kt} = \frac{C}{8\pi t} \left[ \left\{ -\frac{T}{T^{2} - Y^{2}} + \frac{T^{-1}}{T^{-2} - Y^{2}} \right\} \left| \frac{d\Omega}{d\omega} \right| + \frac{t}{t^{2} - y^{2}} \right], \quad (3.101)$$

where

$$\frac{d\Omega}{d\omega} = \frac{\kappa}{b} (1+Y^{6})^{\frac{1}{4}} = \frac{\kappa}{b} \left(1 + \frac{1}{4}Y^{5} - \frac{3}{32}Y^{16} + \ldots\right)$$

$$T = \frac{\kappa t}{b} + \frac{1}{36} \left(\frac{\kappa t}{b}\right)^{9} - \frac{11}{4896} \left(\frac{\kappa t}{b}\right)^{17} + \ldots$$

$$Y = \frac{\kappa y}{b} + \frac{1}{36} \left(\frac{\kappa y}{b}\right)^{9} - \frac{11}{4896} \left(\frac{\kappa y}{b}\right)^{17} + \ldots$$

It follows from Equations (3.100) and (3.101) that

$$\delta_0(0,0) = \frac{C\kappa^2}{8\pi b^2} = \frac{(\sqrt{2}-1)\kappa^2}{4\pi} = 0.1262$$

consistent with the table in Section 3.2.3. Gent also derives this result from the transformation (3.99). The mean parameters  $(\delta_0)_U$  and  $(\delta_0)_E$  for uniform and elliptic loading are given in Tables 1 and 2 of Reference 3.13 for  $\sigma = 23$ /b up to 0.8. Larger values of  $\sigma$  would require an increasing number of terms in the power series. By a simple adjustment to Equation (3.100) a similar analysis could be developed for any regular polygonal tunnel; the result corresponding to Equation (3.101) would rapidly approach that for a circular tunnel (Section 3.4.3), if the number of sides were increased.

For a more general class of closed octagonal tunnels Batchelor<sup>3.11</sup> has supplemented the image system for the rectangular tunnel by superposing the doubly infinite system illustrated in Figure 3.1(c). Here each vortex represents a quadratic distribution of vorticity increasing from zero at each corner to a maximum  $k_{\rm m}$  at the centre of each fillet. Each fillet of length a is assumed to be the hypotenuse of an isosceles triangle, so that their images form an array of squares. Batchelor's method exploits the fact that the image system for the rectangular tunnel in Figure 3.1(b) gives approximately an antisymmetric linear variation in normal velocity  $v_{\rm m}$  across a fillet with distance along the fillet. The same is true of the corresponding normal velocity from the vorticity round the image square, and the autual effect of any two squares can be ignored. The vorticity  $k_{\rm m}$  is chosen to cancel the velocity  $v_{\rm m}$  as best it can. Batchelor's formulae have been generalized for arbitrary a, b and h in Reference 3.12, and the result may be expressed as

$$\delta_{0}(\mathbf{y}, t) = \frac{C}{bh} \delta_{CR}(\mathbf{y}, t) - \frac{CG}{15.36P} \sum_{m=0}^{\infty} [\chi(S_{m}^{+}, T_{m}^{+}) + \chi(S_{m}^{-}, T_{m}^{-})] , \qquad (3.102)$$

where the cross-sectional area  $C = bh - a^2$ .  $\delta_{0R}(y,t)$  is  $\delta_0(y,t)$  for a closed rectangular tunnel as given in Equation (3.82), P is the doublet strength 2Kt and the functions G and X are defined below. G/P is a constant dependent on the tunnel

æ. ..

4 X

Martin . And and a

geometry and is proportional to the gradient of  $v_n$  at the centre of the fillet due to a small wing<sup>\*</sup> at the tunnel centre; from Equation (5) of Reference 3.12,

$$\frac{G}{P} = \frac{2\pi^2 a}{b^3} \sum_{n=-\infty}^{\infty} (-1)^n \left[ \frac{(1 + \cos 2\pi k \cosh 2\pi N + \sin 2\pi N \sinh 2\pi N)(\sin 2\pi N + \sinh 2\pi N)}{(\cos 2\pi M + \cosh 2\pi N)^3} - \frac{\sin 2\pi M \cosh 2\pi N}{(\cos 2\pi M + \cosh 2\pi N)^2} \right], \quad (3.103)$$

where

ŧ

Attended Brannissen Barris

$$2\pi M = \frac{\pi a}{b_{1}/2}$$

$$2\pi N = \frac{\pi a}{b_{1}/2} - \frac{(2n + 1)\pi h}{b}$$

The function X represents the upwash field of a column of image squares and is given by Equation (9) of Reference 3.12 plus an extra term, viz.,

$$\begin{split} \ddot{\chi}(S,T) &= \frac{2}{15} \left(\frac{\pi a}{h}\right)^3 (-2S^3 \div S) + \\ &+ \frac{\sqrt{2}}{15} \left(\frac{\pi a}{h}\right)^a (-6S^3 \div S)T + \\ &+ \frac{1}{30} \left(\frac{\pi a}{h}\right)^5 (24S^5 - 20S^3 + S) + \\ &+ \frac{\sqrt{2}}{180} \left(\frac{\pi a}{h}\right)^6 (120S^5 - 60S^3 + S)T + \\ &+ \frac{8}{5670} \left(\frac{\pi a}{h}\right)^7 (-720S^7 + 840S^5 - 182S^3 \div S) + \\ &+ \frac{17\sqrt{2}}{113400} \left(\frac{\pi a}{h}\right)^8 (-5040S^7 + 4200S^5 - 546S^3 + S)T , \end{split}$$
(3.104)

where S and T take the values

$$S_{a}^{\pm} = \operatorname{sech} \frac{\pi(a\sqrt{2} + b \pm 2y + 2ab)}{2h}$$
$$T_{a}^{\pm} = \tanh \frac{\pi(a\sqrt{2} + b \pm 2y + 2ab)}{2b}$$

\* Note, however, the remarks above and below Equation (3.141), that G/P will depend on t when b/h is large.



The parameter  $(\delta_0)_E$ , as defined in Equation (3.46), has been derived approximately in References 3.11 and 3.12. For three particular octagonal tunnels  $(\delta_0)_E$  as a function of  $\sigma$  may be identified with the quantity  $\frac{1}{2}\delta$  from Table 4 of Reference 3.11 and from Tables 1 and 2 of Reference 3.12.  $(\delta_0)_E$  for  $\sigma = 0$  is given by Equations (3.102) to (3.104) with y = t = 0. This has been computed for the regular octagon  $a = b(\sqrt{2} - 1)$ , b = h. Equation (3.103) gives  $Gb^2/P = -1.307$ , so that Equation (3.102) becomes

$$\delta_0(0,0) = \frac{C}{b^2} \left[ 0.13678 - \frac{2 \times -1.307}{15.36} (0.0819 + 0.0053 + ...) \right]$$
  
= 0.82843 × 0.15147 = 0.1255

which is to be compared with the exact value  $\delta_0(0,0) = 0.1262$  from Reference 3.13. The expansion in Equation (3.104) must fail if a/h is too large, but it appears to be satisfactory provided that a < 0.35h. The regular octagon and the NPL 13 x 9 tunnel have slightly larger fillets than this.

A more elegant, but less accurate method of dealing with the same problem has been formulated by  $\text{Loos}^{3}$ ;<sup>21</sup>. He replaces the distributed vorticity around Batchelor's image squares by a semi-infinite trailing quadrupole vortex at the centre of each square. The strength of the quadrupole is determined from the condition that there is zero flow across each half fillet. Unfortunately, no numerical results are given.

The accuracy to which  $v_{\rm p}$  is cancelled by the squares of vorticity is discussed in Reference 3.12. With reference to Figure 5 of Reference 3.12, the accuracy deteriorates as wing span increases; for a typical span,  $\sigma = 2/3$ , it is found that the maximum uncertainty in G/P is of order 20%, and this can well be larger if the fillets no longer form isosceles triangles. In view of the consequent uncertainty in the second term of Equation (3.102) for wings of large span, it is pertinent to consider the following artifice. Equation (3.28) gives a good approximation to  $(\delta_0)_{\rm E}$  when  $\sigma$  is small; it may be rewritten as

$$(\delta_0)_E = (\delta_{0R})_E \frac{bh + C}{2bh}$$
 when  $c = 0$ . (3.105)

When  $\sigma$  is very large, the fillets will have a negligible effect compared with those of the principal images in the side-walls, and we may write

$$(\delta_0)_E = (\delta_{0R})_E \frac{C}{bh}$$
 when  $\sigma = 1$ , (3.106)

since the influence of the fillets on  $w_i$  may be neglected. Since  $(\delta_0)_E$  is an even function of  $\sigma$ , we combine Equations (3.105) and (3.106) to give

$$(\delta_0)_{\rm E} = (\delta_{0\rm E})_{\rm E} \left[ \left( \frac{1}{2} + \frac{\rm C}{2\rm bh} \right) - \left( \frac{1}{2} - \frac{\rm C}{2\rm bh} \right) \sigma^2 \right] , \qquad (3.107)$$

where  $(\delta_{0R})_E$  is given by Equation (3.84) or Figure 3.9 for the rectangular tunnel of the same breadth and height. The success of Equation (3.107) is demonstrated in Figure 3.12 for the regular octagon of Reference 3.13 and the NPL 9 x 7 and 13 x 9 tunnels corresponding to the respective Tables 1 and 2 of Reference 3.12.

The treatment of rectangular tunnels in Section 3.4.1 includes an analysis of the streamline curvature through the parameter  $\delta_1(y,t)$  of Equation (3.42). There is no

$$\delta_1(\mathbf{y}, \mathbf{t}) = \delta_c(\mathbf{y}, \mathbf{t}) \sqrt{\frac{bh}{c}} \frac{\delta_{1R}(\mathbf{y}, \mathbf{t})}{\delta_{0R}(\mathbf{y}, \mathbf{t})} , \qquad (3.108)$$

where  $\delta_0(y,t)$  is given by Equation (3.102) and the parameters  $\delta_{0R}(y,t)$  and  $\delta_{1R}(y,t)$ are those for the basic rectangular tunnel from Equations (3.82) and (3.87). More simply than by Equation (3.103), it may suffice to evaluate the upwash interference as if the tunnel were rectangular and then to apply the correction factor

corresponding theory for octagonal tunnels, and we can suggest nothing better than

Equation (3.31), which is now written as

$$\left(\frac{bh}{c}\right)^{\frac{3}{2}} \frac{\left(\delta_{0}\right)_{E}}{\left(\delta_{0R}\right)_{E}} \approx \left(\frac{bh}{c}\right)^{\frac{3}{2}} \left[\left(\frac{1}{2} + \frac{c}{2bh}\right) - \left(\frac{1}{2} - \frac{c}{2bh}\right)\sigma^{2}\right]$$
(3.109)

to all contributions involving  $\delta_1(y,t)$ . With reference to Equation (3.55), it is suggested that the elliptic loading factor  $\overline{w}$  be used, and that  $\Delta\alpha \Delta c_i$  be evaluated as if the fillets were not present and split into two parts

$$\left(\frac{\Delta\alpha}{C_{L}}\right)_{R} = \frac{S}{bh} \left(\delta_{0R}\right)_{E} + \left[\left(\frac{\Delta\alpha}{C_{L}}\right)_{R} - \frac{S}{bh} \left(\delta_{0R}\right)_{E}\right].$$

The first part would be replaced by the corresponding quantity for the octagonal tunnel, and the second multiplied by the factor (3.109). Thus

$$\frac{\Delta \alpha}{C_{L}} = \left(\frac{bh}{c}\right)^{\frac{3}{2}} \frac{\left(\delta_{0}\right)_{E}}{\left(\delta_{0R}\right)_{E}} \left(\frac{\Delta \alpha}{C_{L}}\right)_{R} - \frac{S}{c} \left(\delta_{0}\right)_{E} \left\{\left(\frac{bh}{c}\right)^{\frac{1}{2}} - 1\right\} .$$
(3.110)

The drag correction of Equation (3.56) does not involve  $\delta_i(\mathbf{y}, \mathbf{i})$  and would simply become

$$\frac{\Delta c_{\rm D}}{c_{\rm L}^2} = \frac{s}{c} \left(\delta_{\rm p}\right)_{\rm E} \tag{3.111}$$

for elliptic spanwise loading. The principle of Equation (3.109) can be applied to the residual correction  $\Delta C_g/C_L$  from Equations (3.57) to (3.59).  $(\Delta C_g)_2$  depends only on streamline curvature, but  $(\Delta C_g)_1$  has to be split into two parts

$$(\Delta C_{g})_{i} = (\Delta C_{g})_{ii} + (\Delta C_{g})_{i2} .$$

where  $(\Delta C_{n})_{11}$  corresponds to

$$\left[\left(\frac{w_1}{UC_L}\right)_{0.75} - \frac{\Delta \alpha}{C_L}\right] = \frac{4S}{\pi C} \int_0^1 \frac{\partial}{\partial t} \left\{t\delta_0(y,t)\right\} \left\{1 - \frac{t^2}{s^2}\right\}^{\frac{1}{2}} d\left(\frac{t}{s}\right) - \frac{S}{C} \left(\delta_0\right)_{\mathbf{E}}$$

129

in Equation (3.58).  $(\Delta C_{\rm m})_{11}$  has to be evaluated for the octagenel tunnel as if  $\delta_1(y,t) \equiv 0$ , and the remainder  $\Delta C_{\rm m} - (\Delta C_{\rm m})_{11} = (\Delta C_{\rm m})_{12} + (\Delta C_{\rm m})_2$  for the rectangular tunnel. Thus

$$\frac{\Delta C_{\rm B}}{C_{\rm L}} = \frac{(\Delta C_{\rm B})_{11}}{C_{\rm L}} + \left(\frac{bh}{C}\right)^{\frac{3}{2}} \frac{(\delta_0)_{\rm R}}{(\delta_{0\rm R})_{\rm E}} \left[\frac{\Delta C_{\rm B}}{C_{\rm L}} - \frac{(\Delta C_{\rm B})_{11}}{C_{\rm L}}\right]_{\rm R} \quad (3.112)$$

The principle can of course be used with arbitrary spanwise loading end with any method of correction to measured quantities suggested in Sections 3.3.3 to 3.3.6. provided that  $\delta_0(y,t)$  is calculated from Equations (3.102) to (3.104) and the factor (3.103) is applied to all the upwash interference associated with streamline curvature.

Interference effects on off-centre models in a particular octagonal tunnel have been investigated by Batchelor in Reference 3.34. He has shown that fillets have a negligible effect on the incremental interference upwash due to a vertical displacement  $(d - \frac{1}{2}h)$  of the model from the central position for all practical values of d. It therefore means that Equation (3.102) may be used with  $\delta_{oR}(y,t)$  as given in Appendix II of Reference 3.4. The extent to which this holds for values of d outside the range 0.4h < d < 0.6h, say, could be checked by an extension of the analysis for the regular octagonal tunnel in Reference 3.13.

# 3.4.3 Circular and Elliptical Tunnels

While rectangular and octagonal tunnels are plentiful, there are not many elliptical tunnels in regular use today. Closed elliptical working sections are not particularly convenient, but the rounded shape is more appropriate for an open jet and a few open elliptical tunnels have survived. Only the simplest wall interference corrections are likely to be needed, and this may explain the lack of theoretical developments for elliptical tunnels in the past twenty years. On the other hand, the special case of a closed circular tunnel has received full development.

A brief description of the early work will suffice. Sanuki and Tani (Ref.3.35; 1932) use the transformation

$$\overline{y} + iz = l \cosh(\eta + i\zeta) , \qquad (3.113)$$

so that a segment of  $\eta = \eta_0$  corresponds to an 'lliptical boundary of breadth  $b = 2l \cosh \eta_0$  and height  $h = 2l \sinh \eta_0$ . They solve for the stream function in the  $(\eta, \zeta)$  plane and evaluate the parameter  $(\delta_0)_U$  for uniform spanwise loading from Equation (3.45). For 3 wing situated at

in a closed and open tunnel respectively

$$(\delta_{0})_{U} = \frac{\sinh \eta_{0} \cosh \eta_{0}}{2 \cosh^{2} \eta' \cos^{2} \zeta'} \left[ \sum_{\substack{n=1 \\ \text{odd}}}^{\infty} \frac{\cosh^{2} n \eta' \cos^{2} n \zeta'}{n(e^{2n\eta_{0}} + 1)} + \sum_{\substack{n=2 \\ \text{even}}}^{\infty} \frac{\sinh^{2} n \eta' \sin^{2} n \zeta'}{n(e^{2n\eta_{0}} - 1)} \right]$$
(3.114)

$$(\delta_0)_{ij} = -\frac{\sinh \eta_0 \cosh \eta_0}{2 \cosh^2 \eta' \cos^2 \zeta'} \left\{ \sum_{\substack{n=1 \\ \text{odd}}}^{\infty} \frac{\cosh^2 n \eta' \cos^2 n \zeta'}{n(e^{2n\eta_0} - 1)} + \sum_{\substack{n=2 \\ \text{even}}}^{\infty} \frac{\sinh^2 n \eta' \sin^2 n \zeta'}{n(e^{2n\eta_0} + 1)} \right\} . \quad (3.114)$$

For a centrally placed wing  $\eta' = 0$  and the summations in even n disappear. In the limiting case of small span,  $\zeta' = \frac{1}{2}\pi$ , the result for a closed tunnel may be identified with the first of Equations (3.29). Similarly the second of Equations (3.114) for an open tunnel reduces to the first of Equations (3.30) with a change of sign and  $\theta = \eta_0 = \tanh^{-1}(h/b)$ . For wings of finite span and b/h < 1 there would be a corresponding analysis with the transformation

$$y + iz = l \sinh(\eta + i\zeta)$$
.

Curves of  $(\delta_0)_U$  against  $\sigma = 2s/b = (\cosh \eta' \cos \zeta')/\cosh \eta_c$  are given in Figures 4 and 5 of Reference 3.35 for wings lying along the major axis of the elliptical section. It is found that the minimum interference parameter for small wings in a closed tunnel,  $\delta_c = 0.147$ , occurs when  $b/h = \sqrt{2}$ , as does the minimum  $\delta_0 = 0.1190$  for closed rectangular tunnels; the corresponding minima in  $-\delta_0$  for open tunnels occur when  $b/h = 1/\sqrt{2}$ . For any particular closed or open elliptical tunnel the minimum interference parameter occurs when the wing tips are situated at the foci of the ellipse, so that  $\eta' = \zeta' = 0$  and

$$\sigma = 2l/b = \operatorname{sech} \eta_0 = (b^2 - h^2)^{\frac{1}{2}}/b$$

This configuration is of particular interest when the wing has elliptic spanwise loading. Glauert has proved the remarkable result that the interference upwash is constant along the wing span, whether the tunnel is closed or open (Ref. 3. 1, pp. 29 to 31); respectively

$$(\delta_0)_E = \frac{h}{4(b+h)}$$
 or  $-\frac{b}{4(b+h)}$ , when  $\sigma = \frac{(b^2 - h^2)^{\frac{1}{2}}}{b}$ . (3.115)

The important general solution for  $(\delta_0)_{\Xi}$  has been obtained by Rosenhead (Ref. 3. 36; 1933) in terms of elliptic functions. The resulting formulae are rather complicated and comprise four expressions for  $(\delta_0)_{\Xi}$  according as b > h or h > b and the boundary is closed or open: his useful numerical data are presented in Figure 3. 13. The special case of a circular tunnel gives a simple result for the parameter  $\delta_0(y, t)$ of Equation (3.42); for closed and open tunnels respectively

$$\delta_0(y,t) = \pm \frac{R^4}{8(R^4 - t^2y^2)}$$
, (3.116)

where R is the radius (Ref. 3.1, p. 13). It can be shown (cf., Ref. 3.4, Appendix II) that from Equations (3.46) and (3.116)

\* Sanuki's tabulated values appear in the Appendix to Reference 3.14.

$$(\delta_{0})_{g} = \frac{16}{\pi^{2}} \int_{0}^{1} \int_{0}^{1} \delta_{0}(y,t) \left(\frac{t}{s}\right)^{2} \left\{1 - \left(\frac{t}{s}\right)^{2}\right\}^{-\frac{1}{2}} \left\{1 - \left(\frac{y}{s}\right)^{2}\right\}^{\frac{1}{2}} d\left(\frac{t}{s}\right) d\left(\frac{y}{s}\right)$$

$$= \pm \frac{1}{\pi\sigma^{2}} \left[\frac{1}{2}\pi - E_{1}(\sigma^{2})\right] , \qquad (3.117)$$

where  $\sigma = s/R$  and the complete elliptic integral

$$E_1(\sigma^2) = \int_0^{\frac{1}{2}\pi} (1 - \sigma^4 \sin^2 \phi)^{\frac{1}{2}} d\phi$$
.

There is the convenient expansion of Equation (3.117)

$$\left(\delta_{0}\right)_{E} = \pm \frac{1}{8} \left[ 1 + \frac{3}{4^{2}} \sigma^{4} + \frac{3^{2} \cdot 5}{4^{2} \cdot 6^{2}} \sigma^{8} + \frac{3^{2} \cdot 5^{2} \cdot 7}{4^{2} \cdot 6^{2} \cdot 8^{2}} \sigma^{12} + \dots \right] . \qquad (3.118)$$

positive for a closed and negative for an open circular tunnel, in agreement with Equation (6.8) of Reference 3.1.

Analytical expressions for  $(\delta_0)_U$  have been obtained when the tunnel boundary is partly open and partly closed. Kondo<sup>3.14</sup> considers circular tunnels with a symmetrical closed portion below the wing by transforming the interior of the circle into an infinite strip with one straight boundary open and the other closed. For a small wing, zero lift interference is found when 36.4% of the circular boundary is closed, and the percentage increases slowly with wing span. Riegels<sup>3.37</sup> has extended Kondo's analysis to elliptical tunnels and gives numerical results for  $b/h = \sqrt{2}$ . For example, the closed portion can be chosen so that  $|(\delta_0)_U| < 0.008$  for a range of span  $0 \le c \le 0.6$ . Riegels also treats elliptical tunnels with closed portions above and below the wing by transforming the interior into a rectangular boundary with open sides and closed floor and roof; but this arrangement is a less effective means of minimizing lift interference for a range of wing span.

Before we consider further numerical results, it is convenient to outline the more elaborate mathematical methods that are needed to obtain the complete interference upwash field due to the lifting element. General theories for closed and open circular and elliptical tunnels are developed by Lotz (Ref.3.38; 1935). An independent treatment of the circular tunnels by Burgers<sup>3.39</sup> also appeared in 1935.

For the circular tunnels Lot $z^{3,33}$  solved the differential Equation (3.2) in cylindrical co-ordinates for

$$\dot{\Psi} = \Phi_{m} \div \Phi_{i} = \Phi_{m} \div \Phi_{1} \div \Phi_{2} , \qquad (3.119)$$

where  $\Phi_{\rm m}$  is the potential field of the horse-shoe vortex.  $\Phi_1$  corresponds to the image L-vortices of Figure 3.14 so that  $(\Phi_{\rm m} + \Phi_1)$  satisfies the boundary conditions at x = 0 and  $\pm \infty$ , and  $\Phi_2$  is to be determined. With allowance for compressibility the assumed series to satisfy the differential Equation (3.2) is

-

$$\Phi_2 = \frac{K}{4\pi} \sum_{n=1}^{\infty} \cos n\hat{\tau} \sum_{k=1}^{\infty} D_{nk} \sin \left(\frac{k\tau x}{\beta pR}\right) J_n \left(\frac{1k\tau r}{pR}\right) . \qquad (3.120)$$

where  $\beta^2 = 1 - M^2$ ,  $D_{nk}$  are unknown coefficients,  $J_n$  are Bessel functions of the first kind, and the periodicity in x will disappear as p is allowed to increase indefinitely. For the closed tunnel, zero normal velocity on the boundary r = R gives the condition

$$\frac{\partial \Phi_2^{(1)}}{\partial r} = -\frac{\partial}{\partial r} \left( \Phi_n + \Phi_1^{(1)} \right) = -\frac{\kappa}{4\pi R} \sum_{n=1}^{\infty} g_n \left( \frac{x}{\beta R} \right) \cos n\theta , \quad \text{say,}$$

whence, by Equation (3.120).

$$D_{nk} = -\frac{1}{ik\pi J_n'(ik\pi/p)} \int_{-p}^{p} g_n\left(\frac{x}{\beta R}\right) \sin\left(\frac{k\pi x}{\beta p R}\right) d\left(\frac{x}{\beta R}\right) : \qquad (3.121)$$

for the open tunnel, zero tangential velocity on the boundary gives the condition

$$\frac{\partial \Phi_2^{(2)}}{\partial x} = -\frac{\partial}{\partial x} \left( \Phi_{\underline{m}} + \Phi_1^{(2)} \right) = -\frac{K}{4\pi\beta R} \sum_{\underline{n=1}}^{\infty} l_{\underline{n}} \left( \frac{x}{\beta R} \right) \cos n\hat{c} \, . \qquad \text{say.}$$

whence, by Equation (3.120).

$$D_{nk} = -\frac{1}{k\pi J_n(ik\pi/p)} \int_{-p}^{p} l_n\left(\frac{x}{\beta R}\right) \cos\left(\frac{k\pi x}{\beta p R}\right) d\left(\frac{x}{\beta R}\right) . \qquad (3.122)$$

After the coefficients from Equation (3.121) or (3.122) have been substituted in Equation (3.120), the substitution  $k\pi/p = q$  is made as  $p \to \infty$ , so that

$$\sum_{k=1}^{\infty} \frac{1}{k} F\left(\frac{k\pi\varsigma}{p}\right) \quad \text{is replaced by} \quad \int_{0}^{\infty} \frac{1}{q} F(q\varsigma) \, dq \quad .$$

Thus the respective expressions for the upwash  $w_2 = \pm (1/r) \partial \Phi_2 / \partial \hat{\gamma}$  in the plane of the wing ( $\theta = \pi/2$  or  $3\pi/2$  in Figure 3.14) become

$$\begin{split} \pi_{2}^{(1)} &= \pm \frac{K}{4\pi r} \sum_{n=1}^{\infty} n \sin n\ell \int_{0}^{\infty} \frac{\sin (qx/\beta R) J_{n}(iqr/R)}{i\pi q J_{n}^{\prime}(iq)} \left[ \int_{-\infty}^{\infty} g_{n}(\zeta) \sin (q\zeta) d\zeta \right] dq \\ \pi_{2}^{(2)} &= \pm \frac{K}{4\pi r} \sum_{n=1}^{\infty} n \sin n\ell \int_{0}^{\infty} \frac{\sin (qx/\beta R) J_{n}(iqr/R)}{\pi q J_{n}(iq)} \left[ \int_{-\infty}^{\infty} l_{n}(\zeta) \cos (q\zeta) d\zeta \right] dq \end{split}$$

$$(3.122)$$

Along the axis of the tunnel r = 0 Equations (3.123) become relatively simple expressions, since

$$\frac{J_1(iqr/R)}{iqr/R} \rightarrow \frac{1}{2} \quad \text{and, for } n > 1 , \qquad \frac{J_n(iqr/R)}{iqr/R} \rightarrow 0 ;$$

for closed and open circular tunnels respectively

$$w_{2}^{(1)} = \frac{K}{3\pi R} \int_{0}^{\infty} \frac{\sin (qx/\beta R)}{\pi J_{1}^{\prime}(iq)} \left[ \int_{-\infty}^{\infty} g_{1}(\xi) \sin (q\xi) d\xi \right] dq$$

$$w_{2}^{(2)} = \frac{K}{8\pi R} \int_{0}^{\infty} \frac{i \sin (qx/\beta R)}{\pi J_{1}(iq)} \left[ \int_{-\infty}^{\infty} l_{1}(\xi) \cos (q\xi) d\xi \right] dq \qquad (3.124)$$

where

$$\frac{K}{8\pi R} g_1 \left(\frac{x}{\beta R}\right) = \frac{1}{\pi} \int_0^{\pi} \frac{\partial}{\partial r} \left(\Phi_{\underline{n}} + \Phi_1^{(1)}\right)_{r=R} \cos\theta \, d\theta \text{ for the closed tunnel}$$
$$\frac{K}{8\pi R} l_1 \left(\frac{x}{\beta R}\right) = \frac{\beta}{\pi} \int_0^{\pi} \frac{\partial}{\partial x} \left(\Phi_{\underline{n}} + \Phi_1^{(2)}\right)_{r=R} \cos\theta \, d\theta \text{ for the open tunnel}$$

Lengthy expressions for the velocity potentials  $\Phi_{\rm m}$ ,  $\Phi_1^{(1)}$  and  $\Phi_1^{(2)}$  are defined by the vortex configurations in Figure 3.14. The equal and opposite interference upwashes corresponding to  $\Phi_1^{(1)}$  and  $\Phi_1^{(2)}$  are readily defined in terms of the function W(x,y,z)of Equation (3.79), and the limit as  $z \to 0$ 

$$w_{1}^{(1)} = -w_{1}^{(2)} = \frac{\partial \Phi_{1}^{(1)}}{\partial z} = \frac{Kt}{4\pi (R^{2} + yt)} \left[ 1 + \frac{xt/\beta}{R^{2} + yt + \{(R^{2} + yt)^{2} + (xt/\beta)^{2}\}^{\frac{1}{2}}} \right] + \frac{Kt}{4\pi (R^{2} - yt)} \left[ 1 + \frac{xt/\beta}{R^{2} - yt + \{(R^{2} - yt)^{2} + (xt/\beta)^{2}\}^{\frac{1}{2}}} \right]. \quad (3.125)$$

The total interference upwashes in the plane of the wing are

$$\begin{array}{l} w_{1}^{(1)}(x,y) = x_{1}^{(1)} + w_{2}^{(1)} & \text{for the closed tunnel} \\ \\ \pi_{1}^{(2)}(x,y) = w_{1}^{(2)} + w_{2}^{(2)} & \text{for the open tunnel} \end{array} \right\} . \tag{3.126}$$

It will be observed that  $w_2^{(1)}$  and  $w_2^{(2)}$  in Equations (3.123) vanish when x = 0and the whole interference upwash comes from Equations (3.125). On the other hand, for a small wing  $(t \to 0)$  Equations (3.125) give zero upwash gradients  $\partial w_1^{(1)}/\partial x$  and  $\partial w_1^{(2)}/\partial x$  at x = 0, so that the stressline curvature comes entirely from Equations (3.123).

In Reference 3.39 Burgers considers point concentrations of lift at arbitrary positions in the tunnel: for an element of lift  $L_N$  at  $(x_N, y_N)$  in the plane z = 0 the upwash velocity in a closed circular tunnel is obtained as

a second a s

Number (

$$\frac{\partial \Phi_{\rm N}}{\partial z} = w_{\rm N}^{(1)}(x,y) = \frac{L_{\rm N}}{\pi \rho U} \sum_{\rm n=1}^{\infty} \sum_{\rm s} \frac{n^2 J_{\rm n}(\kappa_{\rm s} y) J_{\rm n}(\kappa_{\rm L} y_{\rm N})}{(\kappa_{\rm s}^2 R^2 - n^2) y y_{\rm N} [J_{\rm n}(\kappa_{\rm s} R)]^2} e^{\kappa_{\rm s} (x - x_{\rm N})/\beta}$$
(3.127)

when  $(x - x_N) \le 0$ ; here  $\sum_{S}$  denotes that, for each n, s extends over all positive roots of  $J'_n(\kappa_S R) = 0$ . When  $(x - x_N) > 0$ 

 $\mathbf{w}_{N}(\mathbf{x},\mathbf{y}) = \mathbf{w}_{N}(\mathbf{x},\mathbf{y}) - \mathbf{w}_{N}(2\mathbf{x}_{N} - \mathbf{x},\mathbf{y})$  (3.128)

The corresponding result for an open circular tunnel, when  $(x - x_{y}) \leq 0$ , is

$$\mathbf{w}_{N}^{(2)}(\mathbf{x},\mathbf{y}) = \frac{L_{N}}{\pi\rho U} \sum_{n=1}^{\infty} \sum_{s} \frac{n^{2} J_{n}(\vec{\kappa}_{s}\mathbf{y}) J_{n}(\vec{\kappa}_{s}\mathbf{y}_{N})}{(\vec{\kappa}_{s}R)^{2} \mathbf{y}_{N} [J_{n}'(\vec{\kappa}_{s}R)]^{2}} e^{\vec{\kappa}_{s} (\mathbf{z}-\mathbf{x}_{N})/\beta} .$$
(3.129)

where  $\sum_{S}$  now denotes that, for each n, s extends over all positive roots of  $J_n(\bar{\kappa}_S R) = 0$ . Again Equation (3.128) holds for  $(\mathbf{x} - \mathbf{x}_N) > 0$ . The interference upwash velocity is obtained by subtracting from  $\mathbf{w}_N(\mathbf{x}, \mathbf{y})$  the uncenstrained upwash

$$\mathbf{w}_{N0}(\mathbf{x},\mathbf{y}) = \frac{\mathbf{L}_{N}}{4\pi\rho \mathbf{U}} \frac{\beta^{2} \{(\mathbf{x}-\mathbf{x}_{N})^{2} + \beta^{2} (\mathbf{y}-\mathbf{y}_{N})^{2}\}^{\frac{1}{2}}}{\{(\mathbf{x}-\mathbf{x}_{N})^{2} + \beta^{2} (\mathbf{y}-\mathbf{y}_{N})^{2}\}^{\frac{1}{2}} - (\mathbf{x}-\mathbf{x}_{N})}$$
(3.130)

due to the lifting element. Thus for a number of lifting elements

$$w_i(x,y) = \sum_N [w_N(x,y) - w_{N0}(x,y)]$$
 (3.131)

in accordance with Equations (3.127) to (3.130).

La Have Barry and a starting the

An cutline of the treatment of elliptical tunnels by  $Lotz^{3.38}$  will complete the tasic theory. She uses the elliptical co-ordinates of Equation (3.113), so that the differential Equation (3.2) becomes

$$\beta^2 \frac{\partial^2 \Phi}{\partial \mathbf{x}^2} + \frac{1}{l^2 (\cosh^2 \eta - \cos^2 \zeta)} \left( \frac{\partial^2 \Phi}{\partial \eta^2} + \frac{\partial^2 \Phi}{\partial \zeta^2} \right) = 0 \quad . \tag{3.132}$$

The solution is again obtained in the form of Equation (3.119), where  $\Phi_1$  corresponds to Figure 14, 15 or 16 of Reference 3.38 and

$$\Phi_2 = -\frac{K}{4\pi} \sum_{\nu} \sum_{\mathbf{k}} C_{\nu \mathbf{k}} \sin\left(\frac{2\mathbf{k}\pi \mathbf{x}}{\beta \mathbf{p} \mathbf{b}}\right) \overline{\mathbf{M}}_{\nu \mathbf{k}}(\eta) \mathbf{M}_{\nu \mathbf{k}}(\zeta) , \qquad (3.133)$$

where b, the tunnel breadth, is the major axis of the ellipse. By Equations (3.132) and (3.133), with separation of the variables  $\eta$  and  $\zeta$ ,

$$\frac{d^{2}\overline{M}_{\nu k}}{d\eta^{2}} - \left(\frac{2k^{2}\pi^{2}l^{2}}{p^{2}b^{2}}\cosh 2\eta + \nu\right)\overline{M}_{\nu k} = 0$$

$$\frac{d^{2}M_{\nu k}}{d\zeta^{2}} + \left(\frac{2k^{2}\pi^{2}l^{2}}{p^{2}b^{2}}\cos 2\zeta + \nu\right)M_{\nu k} = 0$$
, (3.134)



where  $\nu$  is determined, for each k, so that the Mathieu function  $\mathbf{A}_{\nu \mathbf{k}}(\zeta)$  is periodic in  $\zeta$ . As for the circular tunnels, p is eventually allowed to tend to infinity so that  $\sum_{\mathbf{k}}$  is replaced by an integral. The final expressions for  $\Phi_2$  for closed and open elliptical tunnels respectively are found in the formidable Equations (73) and (50) of Reference 3.38. By Equations (3.113) and (3.119) the interference upwash becomes

$$\mathbf{w}_{\mathbf{i}} = \frac{\partial}{\partial z} \left( \Phi_{\mathbf{i}} + \Phi_{\mathbf{i}} \right) = \frac{\partial \Phi_{\mathbf{i}}}{\partial z} + \frac{\cosh \eta \sin \zeta}{l(\cosh^2 \eta - \cos^2 \zeta)} \frac{\partial \Phi_{\mathbf{i}}}{\partial \eta} + \frac{\sinh \eta \cos \zeta}{l(\cosh^2 \eta - \cos^2 \zeta)} \frac{\partial \Phi_{\mathbf{i}}}{\partial \zeta} . \quad (3.135)$$

Although Lotz gives graphs of  $w_i$  for both closed and open elliptical tunnels having b = h/2, the calculations only cover small span ( $\sigma = 0$ ) and uniform loading when  $\sigma = \frac{1}{2}\sqrt{2}$  and the wing tips lie at the foci of the ellipse. These difficult calculations (due to Riegels) illustrate a technique for handling Equations (3.133) to (3.135). Nevertheless, for general elliptical tunnels the data in Figures 3.5 and 3.13 may have to serve. The interference corrections  $\Delta \alpha$  and  $\Delta C_0$  must then be evaluated from Equations (3.85) with the aid of  $(\delta_0)_E$  from Figure 3.13 and  $\delta_1/\delta_0$ for a small wing from Figure 3.5 or Equation (3.31). For closed tunnels it seems better to invoke Equations (3.110) to (3.112), where now  $bh/C = 4/\pi$ . The quantities

$$\left(\frac{\Delta \alpha}{C_L}\right)_R$$
 and  $\left[\frac{\Delta C_R}{C_L} - \frac{(\Delta C_R)_{11}}{C_L}\right]_R$ 

have to be evaluated for a closed rectangular tunnel of the same breadth and height as the elliptical tunnel. The ratio  $(\delta_0)_E/(\delta_{0R})_E$  for arbitrary b/h and  $\sigma$  may be evaluated from Figures 3.13(a) and 3.9. Equation (3.112) for the residual correction to pitching moment involves the quantity  $(\Delta C_m)_{11}/C_L$  which is to be calculated from Equation (3.58) as if  $\delta_1(y,t) \equiv 0$ , i.e. as if

$$\begin{bmatrix} \left(\frac{\mathbf{w}_{1}}{UC_{L}}\right)_{0.75} - \frac{\Delta\alpha}{C_{L}} \end{bmatrix} = \int_{0}^{1} (1 - \mathbf{w}) \begin{bmatrix} \int_{0}^{1} \frac{4\Gamma s}{UCC_{L}} \frac{\partial}{\partial t} \left\{ t\delta_{0}(\mathbf{y}, t) \right\} d\left(\frac{t}{s}\right) \end{bmatrix} d\left(\frac{y}{s}\right)$$
$$= \frac{s}{c} \begin{bmatrix} \frac{4}{\pi} \int_{0}^{1} \frac{\partial}{\partial t} \left\{ t\delta_{0}(\mathbf{y}, t) \right\} \left\{ 1 - \left(\frac{t}{s}\right)^{2} \right\}^{\frac{1}{2}} d\left(\frac{t}{s}\right) - \left(\delta_{0}\right)_{E} \end{bmatrix},$$

if the weighting factor W in Equation (3.55) and the circulation  $\Gamma$  in Equation (3.47) are both elliptic. This requires the knowledge of  $\delta_0(y,t)$ , which is formulated on the basis of Reference 3.38 in Equations (2) to (5) of Reference 3.40 by Gavin and Hensel; their function  $\delta_0$  requires the factor 0.125 to be consistent with Equation (3.42).

For circular tunnels, exact values of  $\delta_1$  are available for wings of small span. Von Baranoff<sup>3.41</sup> derives the result

$$\delta_{1} = \frac{1}{4\pi} \int_{0}^{\infty} \frac{q^{2}K_{2}(q) - qK_{1}(q)}{I_{1}'(q)} dq$$
$$= -\frac{1}{4\pi} \int_{0}^{\infty} \frac{q^{2}K_{1}'(q)}{I_{1}'(q)} dq = 0.24975 , \qquad (3.136)$$

when the tunnel is closed for the open circular tunnel

$$\delta_1 = -\frac{1}{4\pi} \int_0^\infty \frac{q^2 K_1(q)}{I_1(q)} dq = -0.19921 , \qquad (3.137)$$

where  $I_1(q) = -iJ_1(iq)$  and  $K_1(q)$  are the modified Bessel functions. Lotz's calculations are consistent with the latter result, but overestimate the former. In Figure 8 of Reference 3.19 Eisenstadt gives further evidence to suggest that Lotz's calculations for the closed circular tunnel are incorrect while those of Burgers are reliable. The relative simplicity of Equations (3.127) and (3.135) as compared with Equations (3.123) and (3.125) commends the method of Burgers as a computational procedure. For the closed tunnel, however, both methods have received important development. While Eisenstadt<sup>3.19</sup> has extended Lotz's method to treat a swept lifting line, Sivells and Salmi<sup>3.26</sup> give extensive tables and charts based on Equation (3.127).

As a lifting element Eisenstadt takes a skew herse-shoe vortex with corners at an origin on the axis of the tunnel and at a point  $(x, r, \partial) = (t \tan \Lambda, t \sec \Lambda, \frac{1}{2}\pi)$  in the co-ordinate system of Figure 3.14(a). Thus  $\Lambda$  is the angle of sweepback of the bound vortex, and t is the perpendicular distance between the trailing vortices. He chooses  $\Phi_1$  to correspond to the L-vortex of Figure 3.14(b) in the half plane  $\theta = \frac{1}{2}\pi$ , so that  $\partial \Phi_1 / \partial z$  is independent of  $\Lambda$  and is given by the second term of Equation (3.125). In an appendix he proves the validity of Lotz's method and discusses the convergence of the series for  $\Phi_2$  and its derivatives. In Reference 3.38,  $\partial(\Phi_m + \Phi_1)/\partial r$  is even in y but not necessarily odd in z, but Eisenstadt's quantity

$$\frac{\partial \Phi_2}{\partial \mathbf{r}} = -\frac{\partial}{\partial \mathbf{r}} \left( \Phi_{\mathbf{n}} + \Phi_1 \right) = -\frac{\mathbf{K}}{4\pi \mathbf{R}} \sum_{\mathbf{n}=1}^{\infty} \tilde{\mathbf{z}}_{\mathbf{n}} \left( \frac{\mathbf{x}}{\beta \mathbf{R}} \right) \sin \left\{ \mathbf{n} \left( \theta - \frac{1}{2} \pi \right) \right\}$$

is odd in z but not necessarily even in y: moreover,  $\Phi_z$  is not in general odd in x. Therefore Equation (3.120) has to be modified and the first of Equations (3.123) is replaced by

$$\frac{\partial \Phi_2}{\partial z} = \mp \frac{\kappa}{4\pi r} \sum_{n=1}^{\infty} n \cos \left\{ n(\theta - \frac{1}{2}\pi) \right\} \int_0^\infty \frac{J_{\mathbf{g}}(iqr/R)}{i\pi q J_{\mathbf{g}}'(iq)} \left[ \int_{-\infty}^\infty \tilde{\mathbf{g}}_{\mathbf{g}}(\zeta) \cos \left( q\zeta - \frac{q\mathbf{x}}{\beta R} \right) d\zeta \right] dq , \qquad (3.138)$$

where the negative sign corresponds to y > 0 ( $\theta = \frac{1}{2}\pi$ ) and the positive sign to y < 0 ( $\theta = 3\pi/2$ ). The integral in square brackets is expressed as

$$\int_{-\infty}^{\infty} \overline{g}_{n}(5) \cos \left(q_{5}^{2} - \frac{qx}{\beta R}\right) d5 = 2l_{n}(q) \sin \frac{qx}{\beta R} + 2k_{n}(q) \cos \frac{qx}{\beta R}$$

and the functions  $l_n(q)$  and  $k_n(q)$  are tabulated for n = 1, 2, 3. Finally in Table 4 of Reference 3.19

$$\frac{4\pi R^2 \mathbf{w}_1}{Kt} = \frac{4\pi R^2}{Kt} \left( \frac{\partial \Phi_1}{\partial z} + \frac{\partial \Phi_2}{\partial z} \right)$$
(3.139)

137

736 J. W. W. W

Supported and the second second second second

and that at here is so in an the

Carry Contraction

# 138

is given as a function of four parameters; in the equivalent incompressible flow (Table 3.1) these parameters become

$$\frac{t}{R} (1 + \beta^{-2} \tan^2 \Lambda)^{\frac{1}{2}}, \quad \cot^{-1}(\beta \cot \Lambda), \quad \frac{x}{\beta R} \quad \text{and} \quad \frac{y}{R} (1 + \beta^{-2} \tan^2 \Lambda)^{\frac{1}{2}}$$

which are somewhat inconvenient unless N = 0.

Sivells and Salmi<sup>3.26</sup> had found the results of Reference 3.19 for swept horse-shoe vortex elements difficult to apply and therefore, following Burgers, used point concentrations of lift. They have simply evaluated the quantity

$$\mathbb{P}\left(\frac{\mathbf{y}_{N}}{\mathbf{P}}, \frac{\mathbf{x}-\mathbf{x}_{N}}{\beta \mathbf{R}}, \frac{\mathbf{y}}{\mathbf{R}}\right) = \frac{\pi \rho U^{2} \mathbf{R}^{2}}{\mathbf{L}_{N}} \left[\mathbb{w}_{N}(\mathbf{x}, \mathbf{y}) - \mathbb{w}_{N0}(\mathbf{x}, \mathbf{y})\right]$$
(3.140)

from Equations (3.127), (3.128) and (3.130) and have given it in Tables 1(a) to 1(e) and Figures 2(a) to 2(e) for y/R = 0, 0.2, 0.5, 0.7 and 0.9. The interference upwash can then be evaluated from Equations (3.131) and (3.140); illustrative calculations are given in Tables 3 to 9 of Reference 3.26.

Off-centre models above or below the plane z = 0 are covered by Equations (3.114) derived from Reference 3.35. The more elaborate theories do not preclude off-centre positions, but the numerical work is complicated further and no calculations appear to have been made. Silverstein and White have evaluated  $(\delta_0)_U$  from Equations (3.114) for closed and open elliptical tunnels with b/h = 1 and 2; their curves in Figures 3. 4, 6 and 10 to 13 of Reference 3.32 indicate that the increased interference is quite as important as for rectangular tunnels (Fig.3.11). The vertical displacement gives a much greater correction factor to  $(\delta_0)_U$  for small wings than for models of large span.

Full-span models are discussed in Section 3.6.4. In their treatment of this problem for a closed circular tunnel, Vincenti and Graham<sup>3.82</sup> consider a simplification of Equation (3.127) on the axis y = 0. In Equations (8) and (11) and Appendix A of Reference 3.42 they transform the series of Bessel functions into a convergent power series in  $(x-x_y)$ .

## 3.4.4 Naif-Wing Models

When the model and the flow conditions have spanwise symmetry, only one half (y > 0, say) need be considered. In such cases there are many advantages in using half models mounted at the tunnel wall, which becomes a reflection plane of symmetry y = 0. Van der Bliek<sup>3, 43</sup> has discussed the practical aspects, noting that half models are cheaper and easier to make and may be combined with shorter pressure leads or a test rig outside the tunnel so as to eliminate sting or strut interference. The Reynolds number can be doubled, and especially for control surfaces the larger size of model should ensure more accurate manufacture<sup>3, 4</sup>. The use of half models introduces a number of interference effects. The modified upwash interference will be discussed in some detail; the other effects of tunnel-wall boundary layer and gap between model and tunnel wall are discussed in Reference 3.48.

Little need be said about rectangular tunnels. The formulae of Section 3.4.1 have simply to be applied to a complete model in a tunnel of the same height and twice the breadth. References 3.22 and 3.44 illustrate the procedure for unswept and swept models respectively.

Half models are frequently tested in octagonal tunnels. Batchelor's<sup>3.11</sup> method, outlined in Section 3.4.2, then requires the modification in Reference 3.45, where calculations are made for the tunnel in the middle diagram of Pigure 3.12. The origin in Figure 1(c) is shifted one half breadth to the lift, and it is necessary to distinguish between the "corner fillets" at  $y = \pm b$ ,  $\pm 3b$ , ... and the "central fillets" at y = 0,  $\pm 2b$ , .... In the first place it is convenient to re-define the quantity G/P in Equation (3.103), so that the gradient of  $v_n$  at the midpoint of the fillet is replaced by the mean gradient over its middle half. For the double tunnel of large breadth to height ratio 2b/h, moreover, G/P is found to depend on the semi-span t of the horse-shoe vortex. For example, from the non-dimensional quantity

$$\Lambda' = 4/2$$
 bt G/P =  $2/2$  b G/K =  $\overline{G}(t/b)$  (3.141)

in Table 2 of Reference 3.45 it is seen that G/P is more than doubled when t increases from 0 to 0.75b. The corresponding quantity for the central fillets is negative and may be identified with  $-\overline{G}[1-(t/b)]$ . It follows that, in the special case  $y = t = \frac{1}{2}b$ , the contributions to the interference apwash from the corner and central fillets cancel each other. In principle,  $\delta_0(y,t)$  is given by a simple extension of Equations (3.102) and (3.104). For the column of central fillets near the reflection plane

$$S = \operatorname{sech} \frac{\pi(a\sqrt{2} + y)}{2h}$$

Unfortunately the expansion for  $\chi(S,T)$  in Equation (3.104) may converge too slowly when y is small, as shown in Table 3(a) of Reference 3.45; the alternative expansion of Equation (25) in the Appendix to Reference 3.45 is more accurate, and it is seen that relatively large negative values of  $\chi$  are found for small y. The main conclusions are that the octagonal tunnel induces considerably more interference upwash near the reflection plane than does the corresponding rectangular tunnel, but that elsewhere the effect of the fillets, whether positive or negative, is likely to be much smaller than for complete models. No simple formula such as Equation (3.107) can be suggested, but it seems typical that  $(\delta_0)_{\rm E}/(\delta_{\rm OR})_{\rm E} > 1$  for small  $\sigma$  and is roughly equal to the area ratio C/bh for fairly small  $(1-\sigma)$ . For half-model testing  $\sigma$  is unlikely to be small; when  $\sigma > \frac{1}{2}$ , it is recommended that the fillets should be neglected for all but the most detailed experiments.

When a reflection plane is installed to test half models in a circular tunnel, the effective boundary is of bipolar cross section as illustrated in Figure 3.15. This configuration was first considered by Kondo (Ref. 3.14; 1935), who used the transformation

 $y + iz = R \sin \psi \tan (\eta + i\zeta)$  (3.142)

to map the interior into the strip  $|\eta| \leq \frac{1}{2}(\pi - \psi)$ , where the angle  $\psi$  is indicated in Figure 3.15. By Equation (3.45), solutions for the interference stream function along a uniformly loaded lifting line give simple expressions

$$(\delta_0)_{U} = \frac{C}{16\pi t^2} \log_e \frac{\pi \sin 2\xi}{(\pi - \psi) \sin \left(\frac{2\pi\xi}{\pi - \psi}\right)}$$
 for closed tunnels  

$$(\delta_0)_{U} = \frac{C}{16\pi t^2} \log_e \frac{\pi \sin 2\xi}{2(\pi - \psi) \tan \left(\frac{\pi\xi}{\pi - \psi}\right)}$$
 for open tunnels  

$$(3.143)$$

where t is the half-wing span,  $\tan \xi = t/(R \sin \psi)$  (Fig.3.15), and the bipolar area  $C = 2R^2(\pi - \psi + \frac{1}{2} \sin 2\psi)$ . Figure 4(b) of Reference 3.14 shows that a closed bipolar tunnel has values of  $(\delta_0)_U$  remarkably close to those for the closed elliptical tunnel of the same breadth to height ratio  $(1 + \cos \psi)$ ; this is much less true of an open bipolar tunnel. Figure 3.15 shows how the interference on a half model of fixed span can be reduced from that on the corresponding complete model in a circular tunnel  $(\psi = 90^\circ)$ . The magnitude of the simple correction to incidence, given by

$$\frac{\pi R^2 \Delta \alpha}{SC_{\rm L}} = \frac{\pi R^2}{C} \left( \delta_0 \right)_{\rm U} \qquad (3.144)$$

falls to a minimum as  $\psi$  decreases. For closed tunnels the minimum  $\Delta \alpha$  is about one half that of a small complete wing or, when t = 0.8R, as little as one third of that for a complete wing. Less impressive reductions are found in the negative correction  $\Delta \alpha$  when the boundary is open.

Davison and Rosenhead<sup>3.%6</sup> have treated the open bipolar tunnel independently by the transformation

$$\pi \tan^{-1} \left( \frac{\mathbf{y} + \mathbf{i} \mathbf{z}}{\mathbf{R} \sin \psi} \right) = 2(\pi - \psi) \tan^{-1}(\eta + \mathbf{i} \zeta) \qquad (3.145)$$

in place of Equation (3.142); Table 2 of Reference 3.46 gives precisely the quantity plotted in Figure 3.15. With an approximation they extend the analysis to obtain a convergent expansion for the corresponding quantity when the spanwise loading is elliptic.

Sivells and Deters<sup>3.47</sup> use the same transformation (3.145) for a closed bipolar tunnel to obtain the interference parameter

$$\delta_{0}(\mathbf{y},t) = \frac{C}{8\pi} \left[ \frac{\pi R(1+\eta^{2}) \sin \psi}{2(\pi-\psi)(\mathbf{y}^{2}+R^{2}\sin^{2}\psi)} \left\{ \frac{\tau/t}{\eta^{2}-\tau^{2}} - \frac{\tau/t}{\eta^{2}\tau^{2}-1} \right\} - \frac{1}{\mathbf{y}^{2}-t^{2}} \right] . \quad (3.146)$$

where

$$\tan^{-1}\tau = \frac{\pi}{2(\pi - \psi)} \tan^{-1} \left(\frac{t}{R \sin \psi}\right)$$
$$\tan^{-1}\eta = \frac{\pi}{2(\pi - \psi)} \tan^{-1} \left(\frac{y}{R \sin \psi}\right)$$
$$C = 2R^{2}(\pi - \psi + \frac{1}{2} \sin 2\psi)$$
141

いっているいで、こので、いないのないないないできょう、いちょうと、ころものできないないないないないできょう

Values of (4Rt/C)  $\delta_0(y,t) = w_i R/K$  at x = 0 are tabulated in Tables 1 and 2 of Reference 3.47 for two configurations  $\psi = 43.09^{\circ}$  and 60.14° that are favourable in Figure 3.15. The streamwise variation of  $w_i$  is represented rather crudely by taking h = 2R and

$$\delta_1(\mathbf{y}, \mathbf{t}) \equiv 2.1 \, \delta_0(\mathbf{y}, \mathbf{t})$$

in Equation (3.42); the factor 2.1 was obtained from the incorrect calculations of Reference 3.38 for a closed circular tunnel, and from Equation (3.136) the value 1.998 might be preferred. Reference 3.47 also examines the consequences of replacing the reflection plane by an end plate; this tends to give a larger interference upwash and a spanwise lift distribution different from that of a complete wing. It is concluded that a reflection plane should be used wherever possible for half-model testing.

No rigorous method is available for determining the general interference upwash field  $w_i(x, y)$  in the plane z = 0 of a closed bipolar tunnel. The best approximation appears to be that of Sivells and Salmi<sup>3.26</sup> using point concentrations of lift. If the element of lift  $L_N$  acts at a position  $(x_N, y_N)$ , then following Reference 3.47 they obtain

$$\mathbf{w}_{N}(\mathbf{x}_{N},\mathbf{y},\mathbf{y}_{N}) = \frac{\mathbf{L}_{N}}{4\pi\rho \upsilon} \left[ \frac{\pi^{2}(1+\eta^{2})(1+\eta^{2}_{N})\mathbf{R}^{2}\sin^{2}\psi}{4(\pi-\psi)^{2}(\mathbf{y}^{2}+\mathbf{R}^{2}\sin^{2}\psi)(\mathbf{y}_{N}^{2}+\mathbf{R}^{2}\sin^{2}\psi)} \left\{ \frac{1}{(\eta-\eta_{N})^{2}} + \frac{1}{(1-\eta\eta_{N})^{2}} \right\} - \frac{1}{(\mathbf{y}-\mathbf{y}_{N})^{2}} \right], \qquad (3.147)$$

where, as in Equation (3.146).

$$\tan^{-1} \eta_{\rm N} = \frac{\pi}{2(\pi - \psi)} \tan^{-1} \left( \frac{\mathbf{y}_{\rm N}}{\mathbf{R} \sin \psi} \right) \, .$$

The plausible assumption is then used that, for any fixed x and  $x_{N}$ , the ratio of  $w_{N}(x,y,y_{N})$  to  $w_{N}(x_{N},y,y_{N})$  is equal to the corresponding ratio for the closed circular tunnel in the  $(\eta,\zeta)$  plane; that is to say.

$$\frac{w_{\rm N}(x,y,y_{\rm N})}{w_{\rm N}(x_{\rm N},y,y_{\rm N})} = \frac{w_{\rm N}^{(1)}(x,\eta) - w_{\rm NG}(x,\eta)}{w_{\rm N}^{(1)}(x_{\rm N},\eta) - w_{\rm NG}(x_{\rm K},\eta)}, \qquad (3.148)$$

where  $\pi_{\rm N}^{(1)}$  and  $w_{\rm H0}$  are defined in Equations (3.127), (3.128) and (3.130) with  $y_{\rm N}$  replaced by  $\eta_{\rm N}$ . Hence the charts in Figure 5 of Reference 3.26 are derived for the particular bipolar tunnel of Reference 3.47 with  $\psi = 60.14^{\circ}$  (b/h = 1.49781). The interference corrections can then be obtained as simply as for the circular tunnel.

## 3.5 NUMERICAL INTERFERENCE CORRECTIONS

Various formulae for interference corrections to measured quantities have been derived earlier. Section 3.2.4 gives expressions suitable for small wings, while more general formulae are discussed in Sections 3.3.3 to 3.3.6. The application of these formulae to a particular tunnel requires knowledge of the appropriate interference

parameters from Sections 3.4.1 to 3.4.4. We consider in Sections 3.5.1 and 3.5.2 the relative importance of the various corrections from available numerical data for rectangular and non-rectangular tunnels respectively. In Section 3.5.3 an attempt is made to cases the accuracy of the simpler methods of estimation.

In the first place the measured quantities that often require interference correction can be divided into three groups.

Geometrical quantity	œ
Force coefficients affecting the interference upwish	C <sub>L</sub> , C <sub>I</sub> , C <sub>m</sub>
Other force coefficients	C <sub>D</sub> , C <sub>n</sub> , C <sub>H</sub>

Although the vortex representation of the model may be governed partly by the theoretical lift distribution, the reliability of interference corrections must also depend on the amount of experimental information in the second group. In half-model tests all three coefficients may be measured, so that the spanwise and streamwise centres of pressure may be known. For complete models there will be separate fields of interference upwash from the symmetrical loading ( $C_L$  and  $C_m$ ) and antisymmetrical loading ( $C_l$ ); the latter, though smaller, will be less well defined, since the streamwise centre of pressure associated with  $C_l$  is unknown. There will be varying degrees of uncertainty in the absence of one or more of these coefficients, especially when only the control hinge moment  $C_{\mu}$  or a restricted pressure distribution is measured. An independent appeal to the theoretical lift distribution or an empirical estimate of  $C_L$  or  $C_l$  is then essential.

The more common interference corrections can be grouped as in the following table.

Principal corrections	$\Delta \alpha$ , $\Delta C_{i}$
Zero correction	∆c <sub>i</sub>
Residual corrections to longitudinal moments	$\Delta C_{H}$ , $\Delta C_{H}$ $\Delta C_{l}$ (half-model)
Residual corrections to vortex-induced coefficients	ΔC <sub>D</sub> . ΔC <sub>n</sub>

The principal corrections are the easiest to estimate quickly;  $\Delta \alpha$  is usually much more important than  $\Delta C_{l}$ . It is convenient to choose  $\Delta \alpha$  such that  $\Delta C_{L} = 0$ . The residual corrections  $\Delta C_{m}$  and  $\Delta C_{H}$  are rather difficult to estimate at all accurately; the latter can be especially important. The vortex-induced coefficients  $C_{DV}$  and  $C_{nV}$ are only part of the measured quantities, but are subject to large percentage interference corrections. It will be assumed that  $\Delta C_{n}$  is governed by the principles of lifting-line theory, but it is suggested in Section 3.5.3 that the resulting formulae are somewhat unreliable.

and the second second

142

A different category of correction arises when lateral aerodynamic characteristics are to be deduced from tests of half-span or part-span models. After the interference corrections, including  $\Delta C_l$ , have been applied to give conditions of unconstrained flow with a reflection plane y = 0, there remains a lifting-surface problem independent of the tunnel shape. The precise details of such a calculation are not discussed, but the resulting correction factor to  $(C_l + \Delta C_l)$  is of major importance.

#### 3.5.1 Bectangular Tunnels

and the Statistic interstation and the

Attention is now given to specific calculations of wall interference on lifting wings in rectangular tunnels. An attempt is made by illustrative charts or formulae or inequalities to indicate the likely magnitude of the various principal and residual corrections relevant to wing models. Brief discussions of the numerical corrections for slender wings, complete aircraft models and other special configurations will be deferred until the appropriate subsections of Section 3.6.

Figure 3.2 has already provided the order of magnitude of the correction  $\Delta \alpha$  for various types of rectangular tunnel. For closed rectangular tunnels Equation (3.55) with some weighting W has been used to calculate the influence of wing span over a wide range of planforms. Polhamus<sup>3, th</sup> has evaluated the mean interference upwash along a swept lifting line with the weighting W = c/ $\bar{c}$ ; this indicates very little influence of sweep angle (-60°  $\leq \Lambda \leq$  60°) for planforms of moderate taper and span ratios  $\sigma = 2s/b \leq 0.8$  in a tunnel of effective shape b/h = 1.4. Acum<sup>3, 17</sup> used the weighting

$$\mathbf{W}_{\mathbf{v}} = \frac{2.4}{\pi} \left\{ 1 - \left(\frac{\mathbf{y}}{\mathbf{s}}\right)^2 \right\}^{\frac{1}{2}} + 0.4 \frac{\mathbf{c}}{\mathbf{c}}.$$

in Equation (3.55) to evaluate the mean interference upwash at three-quarter chord. It is found to a useful approximation that, for any particular breadth to height ratio,

 $\delta = \left(\frac{bh}{SC_L}\right) \Delta \alpha \qquad (3.149)$ 

can be regarded as a function of  $\sigma$  and aspect ratio  $A = (2s)^2/S$  in incompressible flow. Not only is  $\delta$  largely independent of sweepback and wing taper, but it is practically insensitive to spanwise loading, as Appendix I of Reference 3.4 illustrates convincingly for a square tunnel. For elliptic spanwise loading carpets of  $\delta$  are reproduced from Reference 3.17 with allowance for compressibility by plotting against a horizontal scale of  $\sigma + 3(\beta A)^{-1}$  in Figure 3.16; the three diagrams for b/h = 1, 9/7 and 2 show  $\delta$  within the range  $0.11 < \delta < 0.22$  for  $\beta A \leq 8$ ,  $\sigma \leq 0.8$  and  $\sigma + 3(\beta A)^{-1} \leq 2$ . For fixed b/h and  $\sigma$ ,  $\delta$  is represented as a linear function of  $(\beta A)^{-1}$ ; we may therefore write

 $\delta(\beta A) = \delta(6) \left[ 2 - \frac{6}{\beta A} \right] + \delta(3) \left[ \frac{6}{\beta A} - 1 \right] .$ 

 $\frac{\Delta \alpha}{C_{L}} = \frac{S}{bh} \delta(\beta A) , \qquad (3.150)$ 

where

#### 143

あるう とうない かくしょう

441-15 tran 1 "



There is a tendency for Equation (3.150) to underestimate  $\Delta \alpha$  for highly tapered wings when b/h is large (Fig.12 of Ref.3.17) and to overestimate  $\Delta \alpha$  when  $\beta A < 2$ , say.

Independent calculations of  $\Delta \alpha$  by the method of Reference 3.22 for b/h = 10/7 are included in Reference 3.48. The results take the form

$$\frac{\Delta \alpha}{C_{l}} = \frac{S}{bh} \left[ \delta_{ll} + \delta_{sc} \right] , \qquad (3.151)$$

where the 'lifting line' contribution  $\delta_{ll}$  corresponds to  $A = \infty$  and the 'strealline curvature' contribution  $\delta_{sc}$  is roughly inversely proportional to  $\beta A$ , as in Equation (3.150). The curves against  $\sigma$ , reproduced in the top two diagrams of Figure 3.17, show  $\delta_{ll}$  for three values of d/h, the non-dimensional distance from model to floor, and the quantity  $\beta A \delta_{sc}$  subject to the restriction  $\beta A \ge 2$ , say. In Figure 3.18 we try to piece together the approximate information on  $\delta$ , as defined in Equation (3.149), for wings and Mach numbers such that  $\beta A = 2.5$ , i.e.,

$$\sigma^2 = \frac{AS}{b^2} = \frac{2.5h}{b} \left( \frac{S}{\beta bh} \right) .$$

The curves of  $\delta$  against b/h for constant area ratios S/( $\beta$ bh) show how the minimum interference shifts from b/h =  $\sqrt{2}$  for small wings to broader shapes of tunnel as the area ratio increases. Thus the variation in  $\delta$  from tunnel to tunnel is likely to increase with the relative size of model.

Whereas the momentum at right angles to the stream direction is influenced by normal pressures at the tunnel boundary, the streamwise momentum is virtually uninfluenced by external forces. The vortex-induced drag is therefore determined by the spanwise loading on the wing and the cross-section of the tunnel. When an elliptically loaded lifting line can be assumed, a good approximation to the interference correction is given by

$$\Delta C_{\rm D} = \frac{SC_{\rm L}^2}{bh} \left(\delta_{\rm 0}\right)_{\rm E} , \qquad (3.152)$$

where for closed rectangular tunnels  $(\delta_0)_E$  is given as a function of b/h and  $\sigma$  in Figure 3.9. Minor contributions due to residual effects and asymmetrical spanwise loading are noted in Equations (3.63) and (3.71). The parameter  $\delta_{ll}$  at the top of Figure 3.17 illustrates approximately the increase in  $(\delta_0)_E$  with off-centre models. The uncorrected vortex drag of a lifting line with elliptic loading is simply

$$C_{Dv} = \frac{SC_{L}^{2}}{4\pi s^{2}} . \qquad (3.153)$$

By Equations (3.152) and (3.153) the ratio

$$\frac{\Delta C_{\rm D}}{C_{\rm Dy}} = \frac{\pi b \sigma^2}{h} \left( \delta_0 \right)_{\rm E} , \qquad (3.154)$$

The second se

which can be large, e.g., 26% when b/h = 1.6,  $\sigma = 0.7$  and  $(\delta_0)_E = 0.105$ . This may appear slightly exaggerated, since  $C_{Dv}$  does not represent the total drag.

The residual correction to pitching moment, as defined in Equation (3.36) or Equations (3.57) to (3.59) or more generally in the second of Equations (3.67), is independent of pitching axis. The accuracy of the different methods of estimating  $\Delta C_{n}$  will be discussed in Section 3.5.3. As far as can be judged from a few lifting-surface calculations by the Equation (3.67), the correction is likely to lie within the range

$$0.03 \frac{\sigma^2 b}{\beta_{Ah}} < \frac{\Delta C_m}{C_L} < 0.05 \frac{\sigma^2 b}{\beta_{Ah}}$$
(3.155)

for fairly large models in closed rectangular tunnels; Equation (3.36) with Table 3.III would have suggested a much wider variation.

The principal correction  $\Delta C_l$  to rolling moment is never very large. It appears in the form of Equation (3.62) or the last of Equations (3.67). For unswept wings of high aspect ratio, in particular, there are other simple procedures. With a change of sign, Equation (9) of Reference 3.28, with charts for b/h = 10/7 and 3/2, gives  $-\Delta C_l/C_l$  for wings with deflected allerons; Equation (3.62) may be regarded as a generalization of this. Equation (3) of Reference 3.48, based on Reference 3.25, gives the result in the third panel of Figure 3.17 and incorporates the factor (2.5 cos  $\psi - 1.5$ ) for wings yawed through the angle  $\psi$ . We now examine the behaviour of the interference upwash in Equation (3.50). When the wing is small, it can be shown that

$$\frac{\partial}{\partial t} \{ t\kappa_0(y,t) \} = \frac{bty}{8h^3} \zeta_0$$

$$\frac{\partial}{\partial t} \{ t\kappa_1(y,t) \} = \frac{bty}{8\pi h^3} \zeta_1$$
(3.156)

where, by Equations (3.91) and (3.92),

$$\zeta_{0} = f_{2}''(0) + 2 \sum_{n=1}^{\infty} (-1)^{n} f_{1}''(nb/h)$$

$$\zeta_{1} = f_{n}''(0) + 2 \sum_{m=1}^{\infty} (-1)^{n} f_{3}''(nb/h)$$
(3.157)

the second derivatives  $f_n''$  may be evaluated from Reference 3.18. The antisymmetric spanwise loading is taken such that in Equation (3.50)

$$\frac{4s\Gamma}{UC} = \frac{2scC_{LL}}{C} = \frac{32sC_{l}}{\pi bh} \frac{t}{s} \left(1 - \frac{t^2}{s^2}\right)^{\frac{1}{2}} . \qquad (3.158)$$

145

2.20 - 20 2



By Equations (3.50), (3.156) and (3.158),

$$\frac{\overline{w}_{j}}{U} = \frac{4 \operatorname{sysC}_{l}}{\pi h^{4}} \int_{0}^{1} \left[ \zeta_{0} + \frac{x - \overline{z}(t)}{\pi \beta h} \zeta_{1} \right] \left( \frac{t}{s} \right)^{2} \left( 1 - \frac{t^{2}}{s^{2}} \right)^{\frac{1}{2}} d\left( \frac{t}{s} \right) . \quad (3.159)$$

For larger wings it may be appropriate to evaluate  $\zeta_0$  and  $\zeta_1$  in the special case  $y = t = \frac{1}{4}b$ , when there are simple formulae

$$\zeta_{0} = \frac{3h^{2}}{b^{2}} \left[ -f_{2}(0) + f_{2}\left(\frac{b}{2h}\right) - f_{1}\left(\frac{b}{2h}\right) - 2\sum_{m=1}^{\infty} (-1)^{m} f_{1}\left(\frac{mb}{h}\right) \right]$$
  
$$\zeta_{1} = \frac{8h^{2}}{b^{2}} \left[ -f_{q}(0) + f_{q}\left(\frac{b}{2h}\right) - f_{3}\left(\frac{b}{2h}\right) - 2\sum_{m=1}^{\infty} (-1)^{m} f_{3}\left(\frac{mb}{h}\right) \right]$$
. (3.160)

Values of  $\zeta_0$  and  $\zeta_1$  from Equations (3.157) and (3.160) are listed in Table 3.VI. To obtain a rough estimate of  $\Delta C_i$  for wings of high aspect ratio, we omit the second term in the square bracket of Equation (3.159) and use Equations (3.62) and (3.159) to give

$$-\frac{\Delta C_{I}}{C_{I}} = \frac{\pi \dot{A} \mu \dot{\partial} C_{L}}{2(\pi A + \mu \partial C_{L}/\partial \alpha)} \frac{s^{2} s \zeta_{0}}{4 h^{4}} \int_{0}^{1} W \left(\frac{y}{s}\right)^{2} d\left(\frac{y}{s}\right) .$$

With the further approximations

$$u = 1 , \qquad \frac{\partial C_{i}}{\partial \alpha} = \frac{2\tau \Lambda}{\beta \Lambda + 2} , \qquad W = \frac{4}{\pi} \left( 1 - \frac{y^2}{s^2} \right)^{\frac{1}{2}} ,$$

this becomes

5.44

$$-\frac{\Delta C_{l}}{C_{l}} = \frac{\pi s^{*} \zeta_{0}}{4h^{*} (\beta \lambda + 4)}, \qquad (3.161)$$

which has formal similarity with Equation (9) of Reference 3.28. A typical correction to  $C_i$  would be -2.2%, when b/h = 1.6,  $\zeta_0 = 2.9$ ,  $\sigma = 0.7$  and  $\beta A = 6$ . For wings of small span  $\Delta C_i/C_i$  is negligible, being porportional to  $\sigma^4$ . Equation (3.161) with  $\zeta_0$  from Table 3.VI will determine whether the correction can be ignored safely.

The vortex-induced yawing noment involves interaction between symmetrical and antisymmetrical parts of the spanwise loading. Even for unyawed wings, the evaluation of the interference correction  $\Delta c_n$  from Equations (3.70) and (3.71) includes three terms, and further subdivision of the symmetrical part may be needed when control surfaces are deflected. This is illustrated in the methods of Swanson<sup>3.25</sup> and Graham<sup>3.28</sup>. To obtain a rough estimate of  $\Delta c_n$ , we assume a sparwise load distribution

$$\frac{cC_{LL}}{\bar{c}} = \frac{4C_L}{\pi} \left(1 - \frac{y^2}{s^2}\right)^{\frac{1}{2}} + \frac{32C_l y}{\pi s} \left(1 - \frac{y^2}{s^2}\right)^{\frac{1}{2}}$$
(3.162)

and the second second

THE A

「「「「「「「「「「」」」」

and an interference upwash

$$\frac{1}{2} \left( \frac{\mathbf{w}_{1}}{\mathbf{U}} \right)_{\infty} = \left( \frac{SC_{L}}{bh} \right) \delta_{0} \div \left( \frac{r \mathbf{y} SC_{l}}{4h^{4}} \right) \zeta_{0} , \qquad (3.163)$$

where the last term is derived from Equation (3.159). The parameters  $\delta_c$  and  $\zeta_0$  here denote suitable mean values dependent on b/h and  $\sigma$ . Then Equation (3.70) gives

$$\frac{\delta C_{nv}}{C_L C_l} = \frac{S}{bh} \left[ \delta_0 + \left( \frac{bs^2}{32h^3} \right) \zeta_0 \right] . \qquad (3.164)$$

From the lifting-line integral it is found that the vortex-induced incidence is

$$c_{\mathbf{y}} = \frac{1}{4\pi A} \int_{-1}^{1} \frac{s}{\mathbf{y}' - \mathbf{y}} \frac{d}{d(\mathbf{y}/s)} \left[ \frac{cC_{LL}}{\overline{c}} \right]_{\mathbf{y}=\mathbf{y}'} d\left( \frac{\mathbf{y}'}{s} \right)$$
$$= -\frac{1}{\pi A} \left[ C_{L} + 16(\mathbf{y}/s) C_{l} \right] ,$$

whence

$$C_{nv} = \frac{1}{4} \int_{-1}^{n^{1}} \alpha_{v} \frac{cC_{LL}}{\overline{c}} \frac{y}{s} d\left(\frac{y}{s}\right) = -\frac{3C_{L}C_{l}}{\pi A} . \qquad (3.165)$$

By Equations (3.71), (3.164) and (3.165).

$$\frac{\Delta C_{n}}{C_{nv}} = \frac{mb\sigma^{2}}{3h} \left[ \delta_{0} + \left( \frac{\sigma^{2}b^{3}}{128h^{3}} \right) \zeta_{0} \right] + \frac{\Delta C_{l}}{C_{l}} . \qquad (3.166)$$

This lies between the corresponding ratios  $\Delta C_p/C_{DV}$  and  $\Delta C_l/C_l$  from Equations (3.154) and (3.161), e.g., 10% when b/h = 1.6,  $\sigma = 0.7$ ,  $\delta_0 = 0.105$  and  $\zeta_0 = 2.9$ . In practice,  $\Delta C_n$  can be a larger proportion of the measured  $C_n$  (Section 9 of Ref.3.18). It is noted that the second term of Equation (3.164) largely explains the difference between  $\delta_{ll}$  and  $-(C/S)(\Delta C_n/C_LC_l)$  in the top and bottom panels of Figure 3.17.

Hypothetical cases of swept and unswept half-models with control surface in a closed square tunnel are considered in Section 9 and Figures 6 and 7 of Reference 3.4. The residual correction to hinge moment from Equation (3.77) is seen to be important for large models. Further discussion of wall interference on control hinge moments with aerodynamic balance is found in Reference 3.49; Miss Lyon has pointed out that uncorrected wind-tunnel results may be misleading in the design of closely balanced elevators.

## 3.5.2 Non-Rectangular Tunnels

There is evidence in Reference 3.40 of a small effect of spanwise loading on  $\Delta \alpha$  for unswept wings of aspect ratio A = 8 in a closed elliptical tunnel of ratio b/h = 1.37. Given total lift and wing span, the wall interference is less for lift distributions that are more concentrated towards the centre line. But Figure 8 of Reference 3.40 shows that  $\Delta \alpha$  is only slightly less for highly tapered than for

untapered wings. Even smaller changes would be expected of sweptback wings and those of lower aspect ratio; for most purposes elliptic spanwise loading may be assumed. There should be little difficulty in preparing approximate charts for octagonal and elliptical tunnels similar to those for rectangular tunnels in Figure 3.16.

Once the principal correction  $\Delta \alpha/C_L$  is known for rectangular tunnels, the corresponding correction for octagonal or elliptical tunnels can be estimated quickly from Equation (3.110). This only requires the further knowledge of the parameter  $(\delta_0)_E$  for the appropriate value of  $\sigma$ , which is given in Figure 3.12 or 3.13; otherwise Equation (3.107) is available to give a rough estimate for closed octagonal tunnels. Although  $(\delta_0)_E$  will be smaller than the corresponding quantity  $(\delta_{0R})_E$  for a closed rectangular tunnel of the same breadth and height, the interference correction will be larger for the same model. By contrast the bipolar shape, appropriate to a half-model in a closed circular tunnel with reflection plane, can give a much smaller interference (Fig. 3.15) than the half-model in a rectangular tunnel of the same breadth and height.

Equation (3.152) with bh replaced by C will continue to give a good approximation to  $\Delta C_{\rm D}$ . The inequality (3.155) should again indicate whether  $\Delta C_{\rm m}$  is negligible; if not, Equation (3.112) is available. Likewise Equation (3.161) with an appropriate value of  $\zeta_0$  will give the order of magnitude of the correction  $\Delta C_l$ . A crude estimate of  $\delta C_{\rm nv}$  then follows from Equation (3.164), but this correction will seldom be negligible under conditions of asymmetrical spanwise loading. For a closed circular tunnel it can be shown that  $\zeta_0$ , as defined in Equation (3.156), is

$$\zeta_{0} = \frac{32R^{2}}{ty} \frac{\partial}{\partial t} \left\{ \frac{R^{2}t^{2}y}{B(R^{4} - t^{2}y^{2})} \right\} = \frac{8R^{8}}{(R^{4} - t^{2}y^{2})^{2}}; \quad (3.167)$$

for the present purpose this is not appreciably different from the quantity for a closed square tunnel, especially if it is adjusted in the ratio of the parameters  $\delta_0$  for the respective tunnels. In other words, a change in tunnel shape from rectangular to octagonal or elliptical is likely to influence all the corrections by roughly the same factor

$$\frac{\mathrm{bh}(\delta_0)_{\mathrm{E}}}{\mathrm{C}(\delta_{0\mathrm{R}})_{\mathrm{E}}}.$$

For wings of high aspect ratio with part-span ailerons in a closed circular tunnel the correction to yawing moment is conveniently evaluated by means of Reference 3.50. In the present notation

where

{

0

(

$$-\Delta C_{n} = (\delta C_{nv})_{1} + (\delta C_{nv})_{2} + (\delta C_{nv})_{3} , \qquad (3.168)$$

$$\delta C_{nv})_{1} = \frac{SR^{2}(C_{L})_{1}C_{1}}{4s^{2}(y_{2}^{2}-y_{1}^{2})} \left\{ F_{1}\left(\frac{sy_{2}}{R^{2}}\right) - F_{1}\left(\frac{sy_{1}}{R^{2}}\right) \right\}$$

$$\delta C_{nv})_{2} = \frac{S(C_{L})_{1}C_{1}}{y_{2}^{2}-y_{1}^{2}} \left\{ F_{2}\left(\frac{sy_{2}}{R^{2}}\right) - F_{2}\left(\frac{sy_{1}}{R^{2}}\right) \right\}$$

$$\delta C_{nv})_{3} = \frac{SR(C_{L})_{2}C_{l}}{4\pi(y_{2}^{2}-y_{1}^{2})(y_{2}-y_{1})} F_{3}\left(\frac{y_{2}}{R} \cdot \frac{y_{1}}{y_{2}}\right)$$

12 10 10

148

 $(C_L)_1$  and  $(C_L)_2$  denote the contributions to  $C_L$  from the wing incidence and the deflected ailerons respectively, and  $\mathbf{y}_1 < |\mathbf{y}| < \mathbf{y}_2$  denotes the spanwise extent of the ailerons; the functions  $F_1$ ,  $F_2$  and  $F_3$  are formulated and plotted in Reference 3.50. Stewart states that the total correction is likely to be a large percentage of the uncorrected  $C_n$  and may even exceed it when s/R is as large as 0.8.

Special reference must be made to the illustrative computation for a complete sweptback model with tailplane in a closed circular tunnel admirably set out in Tables III to XIII of Reference 3.26. The spanwise distributions of interference upwash at quarter chord, three-quarter chord and the tail are shown in Figures 7 and 8 of Reference 3.26. The corrections to incidence, drag, pitching-moment, rolling-moment and yawing-moment coefficients resemble\* those in Equations (3.55) to (3.59), (3.62) and (3.70), and are evaluated in simple stages with the aid of the charts in Figures 2 and 6 of Reference 3.26, so that similar calculations for other models are reduced to a straightforward routine.

#### 3.5.3 Comparisons of Nethods

We first consider four methods of calculating the principal correction to incidence, viz.,

- (i) Small-wing method. Equation (3.35)
- (ii) Charts of Reference 3.17. Figure 3.16
- (iii) Reference 3.26. Equation (3.55) with  $W = cC_{LL}/CC_L$
- (iv) Lifting-surface theory. Equation (3.67)

The results are compared in Table 3.VII for fairly large models of various planforms in closed rectangular tunnels. The wings are uncambered and without control surfaces. For method (i) the factor  $\lambda$  is taken from lifting-surface theory, as given in Table 3.III. Whether these or the more approximate values  $\lambda = \tilde{c}/\tilde{c}$  are used, it is only when  $\sigma \leq 0.5$  and  $\sigma \leq 0.5h/b$  that the method can always be trusted within ±10%; for nearly square tunnels, however, these conditions on  $\sigma$  can be relaxed a little. In most of the examples one of these limits is violated and larger errors are found. There is little reason to use so crude a calculation when the simple methods (ii) and (iii) exist. In method (ii) extrapolation has been used to cover b/h = 0.911: in method (iii)  $W = cC_{LL}/cC_L$  has been taken from lifting-surface theory and, as for method (iv), Equation (3.47) is used to determine  $w_1/U$ . Method (iii) gives marginally better over-all accuracy and a root-mean-square deviation from method (iv) of 0.0011 in  $\Delta \alpha / C_L$  (0.3 degrees when  $C_L = 0.5$ );  $\Delta \alpha$  tends to be underestimated by method (iii), but in the present examples the error does not exceed 0.5% of the measured incidence.

A comparison of  $\Delta \alpha / C_L$  for a non-rectangular tunnel is made by applying Equation (3.110) to the illustrative example defined in Table III of Reference 3.26. With  $\sigma = s/R = 0.656$ ,

 $(\delta_0)_E = 0.129_5$  for a closed circular tunnel from Figure 3.13  $(\delta_{0R})_E = 0.150_5$  for a closed square tunnel from Figure 3.9  $\delta = 0.169_5$  for b = h and  $\beta A = 5.11$  from Figure 3.16

• In Reference 3.26,  $(w_1/U)_{0.75}$  replaces  $\frac{1}{2}(w_1/U)_{cc}$  in Equations (3.56) and (3.70).

Therefore for the square tunnel

$$\left(\frac{\Delta \alpha}{C_{L}}\right)_{R} = \frac{S}{bh}\delta = 0.0842 \times 0.169_{5} = 0.0143$$
,

and for the circular tunnel Equation (3.110) becomes

$$\frac{\Delta \alpha}{C_L} = \left(\frac{4}{\pi}\right)^{\frac{3}{2}} \frac{0.129_5 \times 0.0143}{0.150_5} - 0.107 \times 0.129_5 \left(\sqrt{\frac{4}{\pi}} - 1\right)$$
$$= 0.0177 - 0.0018 = 0.0159 ,$$

which is in excellent agreement with the value  $\Delta \alpha/C_L = 0.0158$  in Table X of Reference 3.26.

We next consider four methods of calculating the residual correction to pitching moment coefficient, viz.,

- (i) Small-wing method. Equation (3.36)
- (ii) Strip theory. Equation (3.60) with factor
- (iii) Reference 3.26. Equations (3.57) to (3.59)
- (iv) Lifting-surface theory. Equation (3.67)

The results are compared in Table 3.VII for the eleven examples of wings at uniform incidence in closed rectangular tunnels. Nethod (i) is by far the simplest and uses values of the quantity  $\lambda x / \bar{c}$  from lifting-surface theory\* in Table 3.III. As would be expected from the behaviour of  $(\delta_0)_E$  against  $\sigma$  in Figure 3.9, both  $\Delta \alpha / C_L$  and  $\Delta C_L / C_L$  are seriously undersatimated when b/h < 1.0 and  $\sigma$  is large and seriously overestimated when b/h > 1.8 and  $\sigma$  is moderately large. For many cases Equation (3.36) will suffice to show the order of magnitude of the residual correction.  $\Delta C_L / C_L$ amounts to a forward movement of the aerodynamic centre as a fraction of the mean chord  $\bar{c}$  and may not be required to better accuracy than ±0.002, say. Nethod (ii) uses the formula

$$\frac{\Delta C_{g}}{C_{L}} = \frac{(\lambda x_{1}/\bar{c})_{ls}}{(\lambda x_{1}/\bar{c})_{st}} \frac{\partial C_{L}}{\partial \alpha} \int_{0}^{s_{1}} \left[ \left( \frac{w_{1}}{UC_{L}} \right)_{0.75} (x_{L} - \ddot{x}_{L} + \frac{3}{8}c - \frac{1}{4}\ddot{c}) - \left( \frac{w_{1}}{UC_{L}} \right)_{0.25} (\frac{1}{8}c) \right] \frac{c}{\bar{c}^{2}} d\left( \frac{y}{s} \right) .$$

$$(3.169)$$

where  $\overline{c}$  and  $\overline{x}_L$  are defined in Equation (3.60),  $(\lambda x_1/\overline{c})_{15}$  and  $(\lambda x_1/\overline{c})_{st}$  are taken from the last two columns of Table 3.111. The latter is formulated in Equation (3.61) and the former is defined above that equation; their ratio is an essential correction factor in Equation (3.169). Except for the very broad tunnels  $b/h \ge 2$ , method (ii) gives quite as good accuracy as method (iii); furthermore, unlike method (iii), it is not restricted to wings of constant sweepback from root to tip. However, method (iii) does not require the quantity  $(\lambda x_1/\overline{c})_{15}$  and involves only slightly more computation. Its root-mean-square deviation from method (iv) is as small as 9.0010. Finally, from Table 3.VII, with the exception of Case 11  $(\Delta C_m/C_L = 0.054 \sigma^2 b/\beta Ah)$  all the

• Note that Equation (3.38) from strip theory seriously underestimates  $\lambda x_1/\overline{c}$  for unswept wings of moderate or small aspect ratio.

1

150

calculations in the last column by method (iv) satisfy the inequality (3.155). The effect of compressibility on  $\Delta C_{\rm m}/C_{\rm L}$  tends to be greater than the factor  $\beta^{-1}$ ; typical comparisons of methods (iii) and (iv) over the Mach number range  $0.4 \le {\rm M} \le 0.9$  are shown in Figure 3.19.

Turning to antisymmetrical spanwise loading, we now consider three examples of the principal correction to rolling mement coefficient in closed tunnels, viz.,

- (a) b/h = 10/7. Figure 3.17 based on Reference 3.48
- (b) b/h = 2. Tables A3 to A7 of Reference 3.18
- (c) Circular. Tables III and XIII of Reference 3.26

For example (a) the wings are unswept and of high aspect ratio; the quantity

$$\frac{\beta A \div 4}{\beta A} \frac{\beta C \Delta C_1}{SC_1}$$

is calculated by Equation (3.161), by Graham's method<sup>3.28</sup> and from Figure 3.17. Equation (3.161) gives

$$-\frac{(\beta A + 4)C\Delta C_1}{4s^2C_2} = \frac{\pi b^3 \zeta_0 \sigma^2}{64h^3} = 0.1431\zeta_0 \sigma^2 ,$$

where, by Table 3.VI,  $\zeta_0 = 5.0$  and 4.0 for y = t = 0 and  $\frac{1}{4}$ b respectively. When the latter is used, the following comparisons are obtained with results from Figure 3.17 for large values of  $\beta A$ .

σ	0.5	0.6	0.7	0.8
Pigure 3.17	0. 19	0. 23	0.30	0.42
$0.572\sigma^2$	0.14	0.21	0.28	0.37

By Equation (9) of Reference 3.28 with taper parameter  $K_1 = 0.8\overline{c}/\overline{c}$  and section liftcurve slope  $2\pi$ 

$$\frac{(\beta A + 4)C\Delta C_1}{4s^2 C_1} = \frac{0.8h}{2\sigma b} \frac{F_1(y_2/b) - F_1(y_1/b)}{(y_2/b)^2 - (y_2/b)^2}, \qquad (3.170)$$

where  $y_1 < y < y_2$  denotes the span of an aileron and  $F_1$  is plotted for b/h = 10/7and  $\sigma = 0.68$  (0.04) 0.92 in Figure 5 of Reference 3.28. Equation (3.170) has been evaluated for  $\sigma = 0.8$  to investigate the dependence of the interference correction on aileron span

y <sub>1</sub> /s	y <sub>2</sub> /8	Equation (3.170)
0.4	0.8	0.402
0.4	1.0	0. 430
0.6	1.0	0.438
0.8	1.0	0.444

The results only vary within 16% of the value 0.42 from Figure 3.17.

Examples (b) and (c) are for swept wings. In example (b) Equation (3.62) is evaluated with  $(w_i/U)_{0.75}$  from Table A4 of Reference 3.18 and the lift slope  $\partial C_L/\partial \alpha = 2.80$ and weighting W from lifting-surface theory. With A = 2.64,  $\beta = 1$  and  $\tan \Lambda_{0.25} = 1$ , it follows that  $\mu = 0.835$ ; hence Equation (3.62) gives

$$-\frac{\Delta C_1}{C_1} = \frac{0.912 \times 0.00315}{0.1413} = 0.0203$$

in excellent agreement with the value 0.0202 from Table A7 of Reference 3.18 by liftingsurface theory. The rough Equation (3.161) gives

$$-\frac{\Delta C_{l}}{C_{1}} = \frac{\pi s^{4} \zeta_{c}}{4 h^{4} (\beta A + 4)} = 0.01426 \zeta_{0} .$$

when  $\beta A = 2.64$ , b = 2h and s = 0.589h. For the broader rectangular tunnels  $\zeta_0$  varies greatly with wing span. The value  $\zeta_0 = 1.637$  from Table 3.VI(b) gives  $-\Delta C_1/C_2 = 0.023$  of the correct order. In view of the crudity of the approximations leading to Equation (3.161) and the difficulty in assigning a mean value to  $\zeta_0$ , errors of  $\pm 20\%$  are likely. But the interference correction  $\Delta C_1$  is often so small that such errors can be tolerated. When applied to example (c) for a closed circular tunnel, Equations (3.161) and (3.167) with  $t = y = s_0$  give

$$\frac{\Delta C_l}{C_l} = \frac{\pi \sigma^4}{8(\beta A + 4)(1 - \sigma_a^4)^2} = \frac{0.00798}{(1 - \sigma_a^4)^2} \,.$$

when  $\sigma = 0.656$ ,  $\beta A = 5.11$  and  $\sigma_e = s_e/R$ . This agrees with the reliable value 0.01007 from Table XIII of Reference 3.26 when  $\sigma_e = 0.575$ ; the ratio  $\sigma_e/\sigma = 0.88$  is reminiscent of the effective span ratios  $s_e/s$  discussed below Equation (3.41).

Finally we examine the interference corrections to yawing moment coefficient corresponding to the same three examples. The simple expression in Equation (3.164) is evaluated with  $\delta_0 = (\delta_0)_{\rm E}$  and  $\zeta_0$  from Table 3.VI(b).

152

 $\mathbf{T}$ 

Eramle	Tunnel	σ	(8)	7	s/r	$-\Delta c_n / c_l c_l$	
Example	1000100	Ŭ	(°0'E	20		Eq. (3. 164)	Original
(a)	b/h = 10/7	0.600	0.1117	4.00	0.0857	0.0124	0.0118
(b)	b/h = 2	0. 589	0.0980	1.64	0.2631	0.0351	0.0435
(c)	Circular	0.656	0. 1295	10.08	0.1072	0.0175	0.0216

The table compares the approximate values of  $-\Delta C_n/C_L C_l$  with the original values from Figure 3.17, Equation (9.8) of Reference 3.18 and Table XIII of Reference 3.26 respectively. The discrepancies are as much as 20% for both the swept wings. It is instructive to calculate example (c) by the method of Reference 3.50 set out in Equation (3.168) with  $\sigma = 0.656$  and  $(C_1)_2 = 0$ .

y <sub>1</sub> /s	y <sub>2</sub> /8	$-\Delta c_{p}^{\prime}/c_{l}^{\prime}c_{l}^{\prime}$
0.4	0.8	0.0173
0.4	1.0	0.0178
0.6	1,0 ·	0.0181
0.8	1.0	0.0185

The values lie closer to that from Equation (3.164) than to the result in Reference 3.26. A likely explanation is that the latter uses the interference upwash  $(w_i/U)_{0.25}$  in place of  $\frac{1}{2}(w_i/U)_{\infty}$  in Equation (3.70), while the quantity in Equation (3.163), and also in Reference 3.50, is an approximation to  $\frac{1}{2}(w_i/U)_{\infty}$ . There is no reason for preferring either, and the uncertainty of about  $\pm 10\%$  in  $\Delta C_n$  may be fundamental. This is demonstrable in example (b). The original calculations from Table 3b of Reference 3.18 used Equation (3.70), except that  $(\pi_i)_{\mathcal{B}}$  and  $(w_i)_{\mathcal{A}}$  were taken respectively at the local centres of pressure of the antisymmetrical and symmetrical portions of the loading. The calculation has been repeated for the particular alleron considered in Tables A3 to A7 of Reference 3.18, not only with Equation (3.70) as it stands but with  $\frac{1}{2}(w_i/U)_{\infty}$  replaced by  $(w_i/U)_{0.25}$ .

₩ <sub>1</sub> /Ū	Local c.p.	0.25 chord	Half $x = \infty$	Eq. (3. 164)
$-\Delta c_n/c_l c_l$	0.0437	0.0376	0.0331	0.0351

The results convey the necessary warning, that  $\Delta C_n$  is rather unreliable.

#### 3.6 SPECIAL CONFIGURATIONS

Two important aspects of lift interference have not yet been considered. The first of these concerns slender wings, whose span is small compared with the tunnel breadth

153

La sector a sector s

and whose length is of the same order as the tunnel height (Section 3.6.1). The second important configuration is that of wing-body combinations treated in Section 3.6.2 which includes tail-plane interference. Finally there are some remarks on threedimensional interference effects when the tunnel floor is used to simulate the ground (Section 3.6.3) and when models span the tunnel (Section 3.6.4).

## 3.6.1 Sleader Wings

The vortex model of a slender wing is represented in Figure 3.8(b) as a distribution of lifting elements along a portion of the x-axis. Let  $L(x_0)$  denote the lift on the portion of the wing  $0 < x < x_0$ , so that

$$\frac{1}{\rho U}\frac{dL}{dx_0}\,\delta x_0$$

defines the doublet strength of the semi-infinite vortex pair trailing from  $x = x_0$ . Then by Equation (3.13) the interference potential in a rectangular tunnel is

$$\Phi_{1}(\mathbf{x},\mathbf{y},\mathbf{z}) = \frac{1}{4\pi\rho U} \int \psi(\mathbf{x}_{0},\mathbf{x},\mathbf{y},\mathbf{z}) \frac{d\mathbf{L}}{d\mathbf{x}_{0}} d\mathbf{x}_{0} , \qquad (3.171)$$

where

$$\psi(\mathbf{x}_{0},\mathbf{x},\mathbf{y},\mathbf{z}) = \sum_{-\infty}^{\infty} (\mathbf{j}) \frac{\mathbf{z} - \mathbf{n}\mathbf{h}}{(\mathbf{y} - \mathbf{u}\mathbf{b})^{2} + (\mathbf{z} - \mathbf{n}\mathbf{h})^{2}} \left[ 1 + \frac{\mathbf{x} - \mathbf{x}_{0}}{\{(\mathbf{x} - \mathbf{x}_{0})^{2} + \beta^{2}(\mathbf{y} - \mathbf{u}\mathbf{b})^{2} + \beta^{2}(\mathbf{z} - \mathbf{n}\mathbf{h})^{2}\}^{\frac{1}{2}}} \right]$$
(3.172)

and the integral is taken along the length of the wing. It follows from Equation (3.171) and integration by parts that the interference upwash along the axis of the tunnel is

$$\mathbf{w}_{1}(\mathbf{x}) = \left(\frac{\partial \Phi_{1}}{\partial z}\right)_{\mathbf{y}=\mathbf{z}=0} = \frac{\mathbf{L}}{4\pi\rho \mathbf{U}} \int \left(\frac{\partial^{2}\psi}{\partial z\partial \mathbf{x}_{c}}\right)_{\mathbf{y}=\mathbf{z}=0} \left[1 - \frac{\mathbf{L}(\mathbf{x}_{0})}{\mathbf{L}}\right] d\mathbf{x}_{0} , \quad (3.173)$$

provided that L(0) = 0: Equation (3.172) yields

$$\left(\frac{\partial^2 \psi}{\partial z \partial \mathbf{x}_c}\right)_{\mathbf{y}=\mathbf{z}=0} = -\beta^2 \sum_{-\infty}^{\infty} (\mathbf{j}) \frac{(\mathbf{x}-\mathbf{x}_0)^2 + \beta^2 (\mathbf{m}^2 \mathbf{b}^2 - 2\mathbf{n}^2 \mathbf{n}^2)}{\{(\mathbf{x}-\mathbf{x}_0)^2 + \beta^2 (\mathbf{m}^2 \mathbf{b}^2 + \mathbf{n}^2 \mathbf{b}^2)\}^{5/2}} . \quad (3.174)$$

Berndt<sup>3.51</sup> gives the theory for uncambered wings with unswept trailing edges in closed rectangular tunnels. His result for  $w_i/U$  follows from Equations (3.173) and (3.174) with  $j = (-1)^n$  by Equation (3.14) and with the theoretical slender-wing loading

$$\frac{L(x_0)}{L} = \left[\frac{s(x_0)}{s(c_r)}\right]^2 .$$
 (3.175)

where  $s(x_0)$  is the local semi-span and it is assumed that s(0) = 0 and  $ds/dx_0 \ge 0$ for  $0 \le x_0 \le c_r$ . In his final expression

154

1 次語

ì

nametrovana se a antere a se a

4

155

$$\frac{\mathbf{w}_{i}}{\mathbf{U}} = \frac{SC_{L}}{C} \left[ \delta_{\gamma} + \delta'(\mathbf{x}) \right] , \qquad (3.176)$$

 $\delta_0$  is the parameter  $\delta_0^{(1)}$  as defined for a small wing in Equation (3.17) and

$$\delta'(\mathbf{x}) = \int_0^{\mathbf{C}\mathbf{r}} \left[ \frac{\mathbf{s}(\mathbf{x}_0)}{\mathbf{s}(\mathbf{c}_r)} \right]^2 \mathbf{f} \left( \frac{\mathbf{x} - \mathbf{x}_0}{\beta \sqrt{c}} \right) \frac{d\mathbf{x}_0}{\beta \sqrt{c}} - \mathbf{F} \left( \frac{\mathbf{c}_r - \mathbf{x}}{\beta \sqrt{c}} \right) , \qquad (3.177)$$

where

and the state of the state of the second second second

$$f(\xi) = \frac{1}{8\pi} \sum_{-\infty}^{\infty} \int_{-\infty}^{\infty} (-1)^{n} \frac{\xi^{2} + (b/h)n^{2} - 2(h/b)n^{2}}{\{\xi^{2} + (b/h)n^{2} + (h/b)n^{2}\}^{5/2}}$$
$$F(\xi) = \int_{0}^{\xi} f(\xi) d\xi .$$

and

Berndt gives graphs of both  $f(\xi)$  and  $F(\xi)$  ( $0 \le \xi \le 1$ ) for the four tunnel crosssections  $b/h = \frac{1}{2}/2$ . 1,  $\sqrt{2}$  and 2. It may be noted that  $F(\infty) = \delta_0^{(1)}$ ; there is a critical value of b/h above which  $F(\xi)$  ceases to be monotonic. Equations (3.176) and (3.177) also hold for a closed circular tunnel, when  $\delta_0 = 0.125$  and

$$f(\xi) = \frac{1}{8\sqrt{\pi}} \int_0^\infty \int_0^\infty \frac{(2-t^2) \cos ut \cos (u\xi/\pi)}{(1+t^2)^{5/2} I_1'(u)} dt du , \qquad (3.178)$$

where  $I'_1$  is the derivative of the modified Bessel function  $I_1$ . The interference corrections corresponding to the upwash angle of Equation (3.176) are easily formulated from slender-wing theory. In the present notation Berndt's expressions for the incremental corrections to incidence and pitching moment are

$$\Delta \alpha = \frac{SC_L}{C} \left[ \delta_0 + \delta'(c_r) \right] .$$

$$\Delta C_{\mathbf{s}} = \frac{SC_L}{C} \frac{\partial C_L}{\partial \alpha} \int_0^{c_r} \left[ \delta'(c_r) - \delta'(\mathbf{x}) \right] \left[ \frac{\mathbf{s}(\mathbf{x})}{\mathbf{s}(c_r)} \right]^2 \frac{d\mathbf{x}}{\overline{c}}$$
(3.179)

A useful approximation to Berndt's theory is to replace  $f(\xi)$  by a constant  $f(0) = \delta_1 \sqrt{C/h}$  given in Equation (3.16). Then Equation (3.177) becomes

$$\delta'(\mathbf{x}) = \frac{\delta_1}{\beta h} \left[ \int_0^{c_r} \left[ \frac{\mathbf{s}(\mathbf{x}_0)}{\mathbf{s}(\mathbf{c}_r)} \right]^2 d\mathbf{x}_0 - (\mathbf{c}_r - \mathbf{x}) \right] . \qquad (3.180)$$

To this approximation Equation (3.176) can be rewritten as

$$\frac{\mathbf{w}_{i}}{\mathbf{U}} = \frac{SC_{L}}{C} \left[ \delta_{0} + \frac{\mathbf{x} - \mathbf{\bar{x}}}{\beta \mathbf{h}} \delta_{1} \right] . \qquad (3.181)$$

where  $x = \bar{x}$  denotes the centre of lift; it may be noted that Equation (3.181) would apply to cambered slender wings with swept  $trx^{11}$ ing edges when Equation (3.175) is invalid. Equations (3.179) are replaced by the approximations

Construction and the solution of the second

$$\Delta \alpha = \frac{SC_{L}}{C} \left[ \delta_{0} + \frac{c_{r} - \bar{x}}{\beta h} \delta_{1} \right]$$
  
$$\Delta c_{m} = \frac{SC_{L}}{C} \frac{\partial c_{L}}{\partial \alpha} \delta_{1} \int_{0}^{c_{r}} \frac{c_{r} - x}{\beta h} \left[ \frac{s(x)}{s(c_{r})} \right]^{2} \frac{dx}{\bar{c}} \right] . \qquad (3.182)$$

where  $\delta_c$  and  $\delta_j$  are the interference parameters discussed in Sections 3.2.2 and 3.2.3 for all practical tunnel cross-sections; the expression for  $\Delta C_m$  is subject to the same restrictions on planform as Equation (3.175).

The numerical examples in Reference 3.51 concern triangular wings of aspect ratio A = 1 in a low-speed closed duplex tunnel (b = 2h), and Berndt concludes that the corrections do not differ very much from the lifting-line quantities given by Equations (3.182) with  $\delta_1 = 0$ . This is largely a consequence of the corrections being fairly small. Naturally Equations (3.182) as they stand should be even closer to Berndt's values corresponding to Equations (3.179) with  $\delta'(x)$  from Equation (3.177) and

$$\beta = 1$$
,  $b = 2h$ ,  $\frac{s(x_0)}{s(c_r)} = \frac{x_0}{c_r}$ ,  $\frac{S}{C} = \frac{1}{8} \left(\frac{c_r}{h}\right)^2$ ,  $\overline{c} = \frac{1}{2} c_r$ ,  $\frac{\partial C_L}{\partial \alpha} = 1.3$ 

Equations (3.182) with  $\delta_0 = 0.1368$ ,  $\delta_1 = 0.2927$  and  $\bar{\mathbf{x}} = (2/3)c_r$  give the broken curves of  $\Delta \alpha/C_L$  and  $\Delta C_m/C_L$  against  $c_r/h$  in Figure 3.20, which lie very near the full curves obtained from Berndt's calculations. Even when  $c_r/h = 1$ , the discrepancy in  $\Delta \alpha$  is only 0.3% of the measured incidence and the discrepancy in corrected aerodynamic centre is less than  $0.001c_r$ . It seems worth remarking that at  $\alpha = 10$  degrees, when the discrepancy in  $\Delta \alpha$  is 0.03 degrees, the vertical displacement of the wing from apex to trailing edge is  $c_r \sin \alpha = 0.174h$  and is beginning to be significant. That is to say, in rare instances when Equations (3.182) cease to be accurate enough because  $c_r/h$  is large, Equations (3.179) may well be inaccurate through the assumption that the lifting elements are situated in the plane z = 0. The latter defect would require the unattractive remedy of replacing z in Equations (3.171) and (3.172) by  $z - z_0(x_0)$  where  $z_0$  is linear in  $x_0$ .

When the effects of compressibility are calculated, there is greater restriction on the accuracy of Equations (3.182) without prejudice to Equations (3.179). The function  $f(\xi)$  in Equation (3.177) is required over the wider range  $|\xi| < c_r/(\beta/C)$  and the approximation in Equation (3.180) fails for a smaller value of  $c_r/n$ . If the compressibility factors of Table 3.1 are applied at very high subsonic Mach numbers, then both the wing and the tunnel may be regarded as slender<sup>3.52</sup>. It is easily shown that. in place of Equation (3.176),

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{2\mathbf{L}(\mathbf{x})}{c\mathbf{U}^{2}\mathbf{C}} [2\delta_{c}]$$

When (1-M) is small, the interference corrections are therefore

$$\Delta \alpha = \frac{SC_L}{C} [2\delta_0]$$
  

$$\Delta C_m = \Delta \alpha \frac{\partial C_L}{\partial \alpha} \int_0^{C_r} \left\{ 1 - \left[ \frac{s(x)}{s(c_r)} \right]^2 \right\} \left[ \frac{s(x)}{s(c_r)} \right]^2 \frac{dx}{\overline{c}} \right\} .$$
 (3.183)

The state of the state of the

156

いたいであるというないであるとうないのである

Nevertheless it is difficult to envisage practical situations in which Equations (3.183) may be used confidently.

Berndt gives the drag correction

$$\Delta c_{\rm p} = \delta_0 \frac{S c_{\rm L}^2}{c} . \qquad (3.184)$$

This can be justified either by considering the momentum flux across a transverse plane in the distant wake, with the corollary that the vortex drag is independent of any changes in the chordwise load distribution. Alternatively, Taylor<sup>3.70</sup> has shown that Equation (3.184) is compatible with the concept of normal pressures and a suction force distributed along the leading edge. The assumptions of linearized theory are essential to both these arguments.

Unfortunately slender wings usually involve flow separation along the whole leading edge. Such flows, with a free vortex sheet rolling up into concentrated vortices, are beyond the scope of linearized theory; the results in Equations (3.179) or (3.182) and in Equation (3.184) therefore need reconsideration. In the absence of a suitable nonlinear theory the expressions for  $\Delta \alpha$  in Equations (3.179) and (3.182) will be retained, but it is doubtful whether those for  $\Delta C_m$  are worth applying. Equation (3.184) needs to be modified, since the leading-edge suction force disappears and the whole lift acts normal to the planform. Even if the wing is cambered, it seems reasonable to neglect the redistribution of lift associated with the residual interference correction  $\Delta C_m$ ; then the principal correction  $\Delta \alpha$  to incidence is accompanied by a correction

$$\Delta C_{\rm D} = C_{\rm L} \Delta \alpha = \frac{SC_{\rm L}^2}{C} [\delta_0 + \delta'(c_{\rm r})] \qquad (3.185)$$

by the first of Equations (3.179). Equation (3.185) should therefore replace Equation (3.184) whenever there is extensive leading-edge flow separation.

#### 3.6.2 Wing-Body-Tail Combinations

Many fundamental aerodynamic experiments requiring accurate wall interference correction are carried out on wing models, but the majority of wind-tunnel tests are made on more complete aircraft models. Although rather lower accuracy may often be sought, models of wing-body-tail configurations tend to have larger dimensions relative to the tunnel, so that wall interference remains important. The evaluation of interference corrections becomes more complicated in each of its phases, vortex representation, interference upwash field, principal corrections and residual corrections.

In the first place there are more unknowns in the vortex model, as the wing, body and tail may all carry lift and only the total forces may be measured. The usual approximation is to ignore the body lift and to take the theoretical spanwise loading along the quarter-chord lines of the wing and tail so as to be consistent with the measured lift and pitching moment. A more detailed representation is of course desirable, but is only practicable when there are additional theoretical or experimental loading data. Next, the interference upwash field involves large streamwise displacements between the lifting elements and the various parts of the model, so that it is

157

•

ţ



inadmissible to suppose that the upwash is linear in x; this particular complication is aggravated by the effect of compressibility. Furthermore, vertical displacements from the tunnel axis may no longer be negligible.

Then there are corrections to measured quantities to consider, and in this respect it is usual to treat the wing, body and tail quite separately. The principal interference correction  $\Delta \alpha$  is still calculated from Equation (3.55) where  $(\pi_i/U)_{0.75}$ now includes a small contribution proportional to  $(C_i)_t$  the lift coefficient of the tail plane. It is also necessary to calculate a mean value of the interference upwash along the three-quarter-chord line of the horizontal tail surface

$$\delta \alpha_{t} = \int_{0}^{1} \left( \frac{\mathbf{w}_{i}}{U} \right)_{t} \frac{U}{U_{t}} \mathbf{w}_{t} d\left( \frac{s}{s_{t}} \right) , \qquad (3.186)$$

where  $U_t$  is the local streamwise velocity at the tail,  $s_t$  is its semi-spen, and  $W_t$  is the weighting appropriate to the planform of the tail surface. Then the correction

$$\Delta \alpha_{t} = \hat{o} \alpha_{t} - \Delta \alpha \qquad (3.187)$$

may be applied to the incidence of the tail plane. Alternatively Equation (3.187) may be replaced by corrections

$$\Delta C_{L} = -\frac{\partial C_{L}}{\partial x_{t}} (\delta \alpha_{t} - \Delta \alpha)$$

$$\Delta C_{R} = -\frac{\partial C_{R}}{\partial \alpha_{t}} (\delta \alpha_{t} - \Delta \alpha)$$
(3.188)

where  $\partial C_L / \partial \alpha_c$  and  $\partial C_E / \partial \alpha_c$  are the experimentally determined acrodynamic derivatives with respect to tail setting. A further correction must be applied to the vertical location of the vortex wake or positions at which the local flow is measured. The wake displacement is obtained as

$$\Delta z = \int_{TE}^{X} \left( \frac{v_1}{v} \right) \frac{v}{v'} dx . \qquad (3.189)$$

where U' denotes the ioual streamwise velocity and the integration is from the trailing edge to the position concerned. This correction can be particularly important when the aerodynamic characteristics of the tail are strongly influenced by body vortices or trailing vortices from the wing. Difficulties arise from the lack of a suitably simple lifting-surface theory for wing-body-tail configurations. Because the interference upwash is interpreted separately for each component, the residual corrections are liable to be larger than usual and less predictable. This is certainly the case when well interference effects at the body are considered in isolation, as will be discussed later.

In the earlier theories the effect of the body was neglected. Glauert and Hartshorn (Ref.3.16; 1924) were the first to evaluate Equation (3.187) from the simple result in Equation (3.40), viz.,

175

$$\Delta \alpha_{t} = \frac{x_{t} \delta_{1}}{\beta h} \frac{SC_{L}}{C} ,$$

where  $x_t$  is the length of the tail arm from the three-quarter-chord point and  $\delta_1$ is the interference parameter for a small wing. They assume that the tail arm is of the same order of magnitude as the wing semi-span and that the dimensions of the wing are small relative to those of the tunnel. Their correction to tail setting  $\Delta \alpha_{+}$  and tunnel-induced downwash at the tail

$$-\left(\frac{\mathbf{w}}{\mathbf{U}}\right)_{\mathbf{t}} = -\left(\delta_{0} + \frac{\mathbf{x}_{\mathbf{t}}\delta_{1}}{\beta\mathbf{h}}\right)\frac{\mathbf{SC}_{\mathbf{L}}}{\mathbf{C}}$$
(3.190)

are well confirmed by experiments on the same model in closed square tunnels of different sizes, as shown in Figures 3 to 6 of Reference 3.16 (Figs. 17 and 18 of Ref. 3.1). Towards the end of Section 3.2.4 with reference to Figure 3.6, it is shown that so simple a theory fails when the tail are  $x_t$  is as such as half the tunnel height. The following discussion explains the apparent success of this theory when applied to square tunnels.

A more general theory is described in Reference 3.30, where Cowley and McMillan calculate  $\Delta \alpha_{\star}$  with allowance for wing span and the non-linear variation in interference upwash with streamwise distance. For uniform spanwise loading w, is obtained from Equation (3.89) without the terms  $|\mathbf{n}| > 3$  and  $|\mathbf{n}| > 5$ , and the correction may be written as

$$\Delta x_{t} = \frac{SC_{L}}{C} (\delta_{t}')_{t} = \frac{SC_{L}}{C} [\delta_{0}(0,s) + \delta'(\beta^{-1}x_{t},0,s) - (\delta_{0})_{t}] , \qquad (3.191)$$

where  $\delta_{0}(y,t)$  and  $(\delta_{0})_{ij}$  are given by Equations (3.82) and (3.83) and  $\delta'(\beta^{-1}x_{t},0,s)$  corresponds to the increment in  $w_{i}$  along the tail arm. For elliptic spanwise loading they evaluate

$$\Delta \alpha_{t} = \frac{SC_{L}}{C} \left( \delta_{t}^{\prime} \right)_{E} , \qquad (3.192)$$

where

and the second states of the second states and the

$$(\delta_{t}')_{E} = \frac{4}{\pi} \int_{0}^{1} \left[ \delta_{0}(0,t) + \delta'(\beta^{-1}x_{t},0,t) - \{\delta_{0}(t)\}_{U} \right] \left(\frac{t}{s}\right)^{2} \left\{ 1 - \left(\frac{t}{s}\right)^{2} \right\}^{-\frac{1}{2}} d\left(\frac{t}{s}\right)$$

Cowley and McMillan showed that their results for a low-speed duplex tunnel diff greatly from the simpler theory of Reference 3.16 when  $\sigma > 0.5$ . The comparise between  $(\delta'_t)_{t}$ ,  $(\delta'_t)_{g}$  and the linear quantity  $\mathbf{x}_t \delta_t / h$  against  $\sigma$  are reproduced to the second sec in Figure 3.21 for the particular configuration  $x_t = 0.8s$ . The additional mixedbroken curve of

$$\frac{c}{sc_{L}} \Delta \alpha_{t} = \delta'(\beta^{-1}x_{t}, 0, 0) = F\left(\frac{x_{t}}{\beta\sqrt{c}}\right) \qquad (3.193)$$



Margin to Some She

by Equation (3.177) shows that for large models more than half the error from Reference 3.16 is stiributable to the approximation that  $\Delta \alpha_t$  is linear in  $\mathbf{x}_t$ . Now for square tunnels, in which the experiments of Reference 3.16 were made, the effects of nonlinearity in  $\mathbf{x}_t$  and of wing span are in opposition. This cancellation of errors must have played an important part in the otherwise perfect experimental confirmation of the simpler theory mentioned above.

The most comprehensive calculations of  $\Delta \alpha_t$  for rectangular tunnels are those of Silverstein and White (Ref. 3. 32; 1935). The body and the tail loadings are ignored and the wing is assumed to have uniform spanwise loading. The interference upwash in the central plane y = 0 is calculated as

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{SC_{\mathrm{L}}}{C} \delta_{\mathrm{w}} \left( 1 + \frac{\delta_{\mathrm{A}}}{\delta_{\mathrm{w}}} \right) . \qquad (3.194)$$

where  $\delta_{\mathbf{x}}$  is a function of  $\sigma = 2s/b$  and the vertical location of the wing and the ratio  $\delta_{\mathbf{x}}/\delta_{\mathbf{x}}$  is obtained to a good approximation as a function of  $\sigma$ ,  $\mathbf{x}_{\mathbf{t}}/\beta_{\mathbf{b}}$  and the vertical locations of wing and tail. In effect Equation (3.80) is evaluated exactly where  $|\mathbf{x}|$  and  $|\mathbf{n}|$  are both less than 3, but the remainder of the double series is calculated to first order in  $\mathbf{x}$  and  $\mathbf{t} = \mathbf{s}$ . Extensive results are available for four different rectangular tunnels in the following charts of Reference 3.32.

Boundary	b/h	δ	δ <sub>A</sub> /δ <sub>z</sub>
Closed	1	Figure 10	Figures 15 to 17
Cper	1	Figure 11	Figures 18 to 20
Closed	2	Pigure 12	Pigures 21 to 23
Open	2	Figure 13	Figures 24 to 26

It is suggested that the values of  $\delta_A/\delta_w$  for rectangular tunnels may suffice for elliptical tunnels of the same breadth to height ratio and type of boundary, provided that the correct  $\delta_w$  is used in Equation (3.194). For circular tunnels Figure 6 of Reference 3.32 would be used in place of Figure 10 or 11 for  $\delta_w$ , and likewise for elliptical tunnels (b = 2h) Figures 3 and 4 would replace Figures 12 and 13.

A different method of calculation has been given by Brown<sup>3.33</sup>. The wing loading is again represented by a horse-shoe vortex, and the complete expression for the interference upwash at the tail in a closed rectangular tunnel is given by

$$\frac{16\pi s}{sc_{L}} \hat{c}\alpha_{t}(y) = \sum_{-\infty}^{\infty} \sum_{-\infty}^{\prime} \left[ W(\beta^{-1}x_{t}, y - \mathbf{n}b + s, d_{t} - d - 2nh) - W(\beta^{-1}x_{t}, y - \mathbf{n}b - s, d_{t} - d - 2nh) \right] - \sum_{-\infty}^{\infty} \sum_{-\infty}^{\prime} \left[ W(\beta^{-1}x_{t}, y - \mathbf{n}b + s, d_{t} + d - 2nh) - W(\beta^{-1}x_{t}, y - \mathbf{n}b - s, d_{t} + d - 2nh) \right] - \left[ W(\beta^{-1}x_{t}, y + s, d_{t} + d) - W(\beta^{-1}x_{t}, y - s, d_{t} + d) \right], \qquad (3.195)$$

Kanishana kanakana kanakana sanasana

where, as usual,  $\sum_{i=1}^{\infty} denotes that (m, n)$  takes all possible integral pairs except (0,0), W(x,y,z) is defined in Equation (3.79),  $x_t$  is the length of the tail arm, y is the spanwise distance along the tail, d and  $d_t$  denote the respective distances of the wing and horizontal tail surface from the tunnel floor. Brown writes each double summation in the form

$$\sum_{-\infty}^{\infty} \sum_{n=-N}' = \sum_{n=-N}^{N} \sum_{n=-N}^{N-1} + R_{NN}$$

and the remainder  $R_{NN}$  is evaluated on the basis that W(x, y, z) is a nearly linear function of y when y and z are not both small; thus

$$W(x, y+s, z) - W(x, y-c, z) \approx \frac{2s}{b} [W(x, y+\frac{1}{2}b, z) - W(x, y-\frac{1}{2}b, z)]$$

so that

$$\begin{split} \hat{h}_{\mathbf{k},\overline{n}} &= \left(\sum_{-\infty}^{\infty} \int_{-\infty}^{u} -\sum_{-\mathbf{k}}^{\mathbf{k}} \sum_{-\mathbf{N}}^{v}\right) \left[ \mathbb{W}(\mathbf{x}, \ \mathbf{y} - \mathbf{x}\mathbf{b} + \mathbf{s}, \ \mathbf{z} - 2\mathbf{n}\mathbf{h}) - \mathbb{W}(\mathbf{x}, \ \mathbf{y} - \mathbf{x}\mathbf{b} - \mathbf{s}, \ \mathbf{z} - 2\mathbf{n}\mathbf{h}) \right] \\ &= \frac{2\mathbf{s}}{\mathbf{b}} \left[ 2\sum_{\mathbf{n}=-\infty}^{\infty} \mathbb{W}(\mathbf{x}, \ \infty, \ \mathbf{z} - 2\mathbf{n}\mathbf{h}) - \sum_{\mathbf{n}=-\mathbf{N}}^{\mathbf{N}} \mathbb{W}(\mathbf{x}, \ \mathbf{M}\mathbf{b} + \frac{1}{2}\mathbf{b} + \mathbf{y}, \ \mathbf{z} - 2\mathbf{n}\mathbf{h}) - \sum_{\mathbf{n}=-\mathbf{N}}^{\mathbf{N}} \mathbb{W}(\mathbf{x}, \ \mathbf{M}\mathbf{b} + \frac{1}{2}\mathbf{b} - \mathbf{y}, \ \mathbf{z} - 2\mathbf{n}\mathbf{h}) \right] \end{split}$$

$$(3.196)$$

When j = 0, Equations (3.195) and (3.196) simplify to give

$$\frac{C}{SC_{L}} \delta \alpha_{t} = \delta_{t} = \frac{bh}{16\pi B} \left[ \sum_{-N}^{N} \sum_{-N}^{N} \left\{ W(\beta^{-1}x_{t}, mb + s, z - 2nh) - W(\beta^{-1}x_{t}, mb - s, z - 2nh) \right\} + \frac{4s}{b} \left\{ \sum_{-\infty}^{\infty} W(\beta^{-1}x_{t}, \infty, z - 2nh) - \sum_{-N}^{N} W(\beta^{-1}x_{t}, Mb + \frac{1}{2}b, z - 2nh) \right\} \right]_{z=d_{t}+d}^{z=d_{t}+d} - 2W(\beta^{-1}x_{t}, s, d_{t}+d) , \qquad (3.197)$$

where the square bracket notation indicates that, as in the case of an integrated function, the value for  $z = d_t \div d$  is to be subtracted from that for  $z = d_t - d$ . A few calculations for horizontal models  $(d_t = d)$  in two closed rectangular tunnels b = 2h and b = 4h are found in Table 8 of Reference 3.33. A more elaborate method of computation for clozed rectangular tunnels is suggested by Sanders and Pounder in Section 3.4 of Reference 3.23. When the tail span is appreciable, an average value of  $\delta \alpha_t$  is required so that the correction to tail setting becomes

$$\Delta \alpha_t = \int_0^1 \delta \alpha_t(y) d(y/e_t) - \Delta \alpha . \qquad (3.198)$$

161

es j=207

the second second and the second s

initzen.

Batchelor<sup>3.34</sup>, following the method of Reference 3.32 for a centrally placed wing  $(d = \frac{1}{2}h)$  in a closed rectangular tunnel (b/h = 9/7), has devised a rough procedure for estimating the incremental effect of superimposed fillets to form an octagonal tunnel. As an alternative estimate, the assumption of Equation (3.108) may be generalized, so that for octagonal tunnels Equation (3.194) is replaced approximately by

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{\mathbf{SC}_{\mathrm{L}}}{\mathbf{C}} \, \delta_{\mathbf{w}} \left[ 1 + \left( \frac{\delta_{\mathrm{A}}}{\delta_{\mathrm{w}}} \right)_{\mathrm{R}} \sqrt{\frac{\mathrm{bh}}{\mathrm{C}}} \right] \,. \tag{3.199}$$

in which  $(\delta_{i}/\delta_{w})_{R}$  corresponds to the rectangular tunnel. The parameter  $\delta_{w}$  for a centrally placed wing is identified with  $\delta_{0}(0,s)$  from Equations (3.102) to (3.104). For off-centre wings it is shown in Reference 3.34 that the fillets have negligible effect on the variation of  $SC_{L} \delta_{w}/C$  with the vertical location of the wing. Therefore the result for closed rectangular tunnels from Reference 3.33\*

$$b_{w} = \frac{b}{16s} \left[ \frac{2h}{\pi s} - \sin^{2} \frac{\pi d}{h} \right]_{w=-\infty}^{\infty} \frac{\coth \frac{\pi (ba+s)}{2h}}{\sinh^{2} \frac{\pi (ba+s)}{2h} + \sin^{2} \frac{\pi d}{h}} \right]$$
(3.200)

provides the increment

$$\delta_{\mathbf{y}} - \delta_{0}(0, \mathbf{s}) = \frac{C}{bh} \frac{b}{16s} \left[ \sum_{\mathbf{n}=-\infty}^{\infty} \frac{\coth \frac{\pi(b\mathbf{n}+\mathbf{s})}{2h}}{\sinh^{2} \frac{\pi(b\mathbf{n}+\mathbf{s})}{2h} + 1} - \sin^{2} \frac{\pi d}{h} \sum_{\mathbf{n}=-\infty}^{\infty} \frac{\coth \frac{\pi(b\mathbf{n}+\mathbf{s})}{2h}}{\sinh^{2} \frac{\pi(b\mathbf{n}+\mathbf{s})}{2h} + \sin^{2} \frac{\pi d}{h}} \right]$$
$$= \frac{C}{16hs} \cos^{2} \frac{\pi d}{h} \sum_{\mathbf{n}=-\infty}^{\infty} \frac{\tanh \frac{\pi(b\mathbf{n}+\mathbf{s})}{2h}}{\cosh^{2} \frac{\pi(b\mathbf{n}+\mathbf{s})}{2h} - \cos^{2} \frac{\pi d}{h}}$$
(3.201)

to apply to the value of  $\delta_0(0, s)$  from Equations (3.102) to (3.104) for use in Equati-(3.199). The quantity  $(\delta_A/\delta_W)_R$  may be evaluated as  $(\delta_L/\delta_W) - 1$ , where  $\delta_L$  and  $\hat{J}_{J}$ are defined in Equations (3.197) and (3.200). The factor  $\sqrt{(bh/C)}$  in Equation (3.199) is rather speculative when  $x_L$  is large, and indeed is not required in the limit as  $x_L \to \infty$  when  $\delta_A = \delta_W$  whatever the tunnel section. However, if  $\gamma$  as be noted from the results in Table 3 of Reference 3.34 and Table 4 of Reference 3. that the fillets appear to have greater percentage influence on  $w_1$  at the tail than on  $w_1$  at the wing, so that the factor  $\sqrt{(bh/C)}$  greater than unity is compatible.

Several papers have been published on tunnel-wall corrections for wing-body combinations. Smith<sup>3.53</sup> considers a circular body concentric with a closed circular tunnel,

 $\delta_{w} = \frac{1}{2} \delta_{T}$  from the bottom equation on p.4 of Reference 3.33, since the lift coefficient  $k_{T} = \frac{1}{2}C_{L}$  was used in the definition of  $\delta_{T}$ .

represents the lifting wing by a horse-shoe vortex, and solves the two-dimensional problem in the plane of the bound vortex x = 0 subject to tangential flow at the two circular boundaries. The resulting infinite sequence of image vortices is shown to give rise to an interference upwash that may be up to twice as great as for the wing alone. Gorgui<sup>3.54</sup> gives closed expressions for the series in Reference 3.53 by a method of conformal transformation and also derives analytical results when the circular body is symmetrically placed in any closed rectangular tunnel. In most of his numerical examples the interference upwash at the wing-body junction is practically twice what it would have been in the absence of the body. Loos<sup>3.21</sup> has treated the same problem for a closed octagonal tunnel with an elegant approximation to the influence of the fillets. and he points cut that the interference effects have to be interpreted as a principal correction to lift in respect of the vortex pairs added within the body on account of the tunnel boundary. In consequence there is the excessive residual interference wwwash at the wing that would be likely to disturb the stalling characteristics of the model. Saith and Gorgui conclude wrongly that their results cast considerable doubt on the practice of applying the corrections, such as Equations (3.186) to (3.189) to measurements on wing-body combinations. There may be uncertainty as to the interference flow field associated with body lift, but this merely requires a few point concentrations of lift distributed along the body and the uncertainty is purely one of representation. From knowledge of the interference field it is a retrograde step to treat the body in isolation. The principal correction to lift in respect of the added vortex pairs is illusory or at least highly undesirable. Instead the principal corrections  $\Delta \alpha$  and  $\Delta \alpha_{t}$  should be applied, and the difficulties of interpretation may then be confined to the wuch smaller residual interference upwash  $(\pi_1/U) - \Delta \alpha$  in the presence of the complete model.

## 3.6.3 Interference on Ground Effect

The discussion of off-centre models towards the end of Section 3.4.1 includes a brief reference to ground effect. Figure 3.11 shows some results from Reference 3.33 for uniformly loaded wings in a closed duplex tunnel as the model is displaced towards the floor of the tunnel. The primary effect on the mean interference parameter  $(\delta_0)_U$  is the increasing contribution from the principal image in the floor. When the floor is used to simulate the ground, this effect ceases to be part of the wall interference and the remainder  $(\delta_0)_{UE}$  falls rapidly to zero as the model approaches the ground (d = 0) and the other images tend to cancel out. The effect is illustrated by the removal of the contribution from the principal image in Equation (3.200). The remainder represents the upwash interference parameter at the centre y = 0 of the uniformly loaded wing with ground simulation

$$\delta_{\pi g} = \frac{b}{16s} \left[ \frac{\sin^2}{\pi s (s^2 + 4d^2)} - \sin^2 \frac{\pi d}{h} \sum_{n=-\infty}^{\infty} \frac{\coth \frac{\pi (\tan + s)}{2h}}{\sinh^2 \frac{\pi (\tan + s)}{2h} + \sin^2 \frac{\pi d}{h}} \right]. \quad (3.202)$$

which is seen to be  $O(d^2)$  as  $d \to 0$ . Typically, with d = 0.25h in Figure 3.11,  $(\delta_G)_{U_S}$  is smaller than  $(\delta_0)_U$  by an order of magnitude and is not required to great accuracy. Even for wings in the centre of oblong tunnels (h = 0.25b and h = 0.50b), Brown calculates small maximum values of  $|(\delta_0)_{R_S} - (\delta_0)_{U_S}|$  in Figures 2 and 3 of Reference 3.33 (0.013 and 0.005 respectively), so that it should be unnecessary to consider elliptic spanwise loading when d/h is fairly small.

163

「日本」なるの日本のないのであるようま

ų

÷

Land British was and the to war

Sanders<sup>3.55</sup> has derived a fairly simple expression for the mean upwash interference along a uniformly loaded lifting-line model in a closed rectangular tunnel. The correction to incidence with ground simulation is

$$\Delta \alpha_{\mathbf{g}} = \frac{SC_{\mathrm{L}}}{C} \left( \delta_{\mathrm{0}} \right)_{\mathrm{Ug}} , \qquad (3 \ 203)$$

where

$$(\delta_{c})_{Ug} = [(\delta_{0})_{U}]_{d=\frac{1}{2}h} + \frac{\pi h}{b} \sum_{a=1}^{\infty} \frac{n \sinh^{2} \frac{\pi n(h-2d)}{b}}{\sinh (2\pi nh/b)} \left( \frac{\sin \pi n\sigma}{\pi n\sigma} \right)^{2} - \frac{h}{8\pi b\sigma^{2}} \log_{e} \left( 1 + \frac{b^{2}\sigma^{2}}{4d^{2}} \right)$$

and the first term on the right hand side is identified with Equation (3.83). Values of  $(\hat{s}_0)_{Ug}$  are available in Tables 5 and 6 of Reference 3.33 which require the factor  $\frac{1}{2}$ , and in Figure 3 of Reference 3.55 where the ordinate is ten times too large. Some of the results are collected in Figures 3.22 and 3.23 for the three shapes of tunnel, h/b = 0.25, C.35 and 0.50. Figure 3.22 shows the effect of model span for two height ratios d/h = 0.5 and 0.2, and Figure 3.23 shows the effect of model height for  $\sigma = 0$ and 0.6. The series in Equation (3.203) only converges rapidly when (h-2d) is small, but the alternative expression, derived in Equation (A-3) of Appendix A of Reference 3.55, is quite suitable for the smaller values of d. Sanders also formulates the sidewash interference with alternative expressions suitable for the two ranges of d. The images of the bound vortex induce a velocity perallel to the tunnel axis, and the average value of this stream-velocity correction over the span of the wing is obtained as

$$\left( \frac{\Delta U}{U} \right)_{\mathbf{g}} = \epsilon_{U\mathbf{g}} = \frac{\mathbf{SC}_{\mathbf{L}}}{\mathbf{C}} \frac{\mathbf{h}}{\mathbf{8}\pi\beta\mathbf{b}\sigma^2} \sum_{-\infty}^{\infty} \frac{\mathbf{i}}{2\mathbf{n}\mathbf{h}-2\mathbf{d}} \left[ \{(2\mathbf{n}\mathbf{h}-2\mathbf{d})^2 + (\mathbf{m}+\sigma)^2\mathbf{b}^2\}^{\frac{1}{2}} + \{(2\mathbf{n}\mathbf{h}-2\mathbf{d})^2 + (\mathbf{m}-\sigma)^2\mathbf{b}^2\}^{\frac{1}{2}} - 2\{(2\mathbf{n}\mathbf{h}-2\mathbf{d})^2 + \mathbf{m}^2\mathbf{b}^2\}^{\frac{1}{2}} \right] .$$
 (3.204)

The slow convergence of this double series is discussed fully in Appendix A of References 3.55 where Equation (3.204) is reformulated for computation in the two cases when (2h-d) or d is small. The calculations for h = 0.35b indicate that  $(C/SC_L)\epsilon_{vg}$  can be considerably larger than  $(\delta_q)_{vg}$ . Equation (3.204) shows that  $\epsilon_{vg} = 0(d)$  as the model approaches the simulated ground. For the larger values of d, however,  $\epsilon_{vg}$  can be quite as important as the usual blockage corrections discussed in Chapter V. But the compressibility factor is only  $\beta^{-1}$  as compared with  $\beta^{-3}$  for solid blockage.

As regards interference on ground effect at the tail, the parameter  $\delta \alpha_{tg}(y)$  is given by Equation (3.195) with the last term omitted. Since  $\mathbf{x}$  is an even function of z, it follows that  $\delta \alpha_{tg}$  is unaltered if the vertical locations of the wing and tail are interchanged. However, the interference at the wing is different in the two cases, and so therefore is the correction to tail setting

$$\Delta \alpha_{tg} = \int_0^1 \delta \alpha_{tg}(\mathbf{y}) \, d(\mathbf{y}/\mathbf{s}_t) - \Delta \alpha_g \, . \qquad (3.205)$$

The evaluation of  $\Delta \alpha_{te}$  is discussed further in Reference 3.33.

Irrespective of tunnel shape, the interference corrections on ground effect are readily deduced from those for off-centre models. The remarks on closed octagonal tunnels involving Equations (3.199) and (3.291) may be helpful, but it is more straightforward to estimate the correction to incidence as

$$\Delta \alpha_{\mathbf{g}} = \frac{SC_{\mathbf{L}}}{C} \lambda_{\mathbf{g}} (\delta_{\mathbf{g}})_{\mathbf{g}} , \qquad (3.206)$$
$$\lambda_{\mathbf{g}} = (\delta_{\mathbf{g}})_{\mathbf{g}\mathbf{g}} / [(\delta_{\mathbf{g}})_{\mathbf{g}}]_{\mathbf{g} = \frac{1}{2} \mathbf{h}}$$

where

is defined by Equation (3.203) for the basic rectangular tunnel and the parameter  $(\delta_0)_{\rm g}$  for an octagonal tunnel is found from the approximate Equation (3.107). When the original working section is circular, experiments on ground effect require the insertion of a ground plate and the tunnel is effectively of bipolar shape. The parameter  $(\delta_0)_{\rm Ug}$  has been derived for open and closed circular tunnels. In Section 12 of Reference 3.14, Kondo has given the general expression  $(\delta'_{\rm u})$  and tabulated results in Table IX in particular cases when the model lieb along the horizontal diameter of the open circular bourdary. Albritton and Huber<sup>3.56</sup> give an analytical expression for the distribution  $[\delta_0(\mathbf{y},\mathbf{s})]_{\rm g}$  along the span of a horse-shoe vortex in a closed circular tunnel with ground plate. The results in Figure 9 of Reference 3.56 for  $\mathbf{s} = 0.62R$  and  $0.11 \leq d/s \leq 0.41$  correspond to the range  $0.002 \leq (\delta_0)_{\rm Ug} \leq 0.026$  based on the area of the complete circular section. Again  $(\delta_0)_{\rm Ug}$  is an order of magnitude smaller than the contribution from the ground plate

$$\frac{C}{32\pi s^2} \log_e \left(1 + \frac{s^2}{d^2}\right) = 0.082 \log_e \left(1 + \frac{s^2}{d^2}\right) .$$

## 3.6.4 Models Spanning Closed Tunnels

We shall consider two investigations in which two-dimensional models have been tested under conditions when wall interference is three-dimensional. The first configuration of an unswept wing in a closed circular tunnel<sup>3,32</sup> has already been mentioned in Section 2.2.5 and at the end of Section 3.4.3. The other configuration is of a swept wing spanning a closed rectangular tunnel<sup>3,57</sup>. In each case the practical problem is to determine whether pressure measurements at the centre section can be corrected to two-dimensional velues.

The analysis of Vincenti and Graham is based on their manipulation of Equation (3.127) with y = 0 in Appendix A of Reference 3.42. Hence

$$\mathbf{w}_{N}^{(1)}(\mathbf{x}, \mathbf{G}) = \frac{\mathbf{L}_{H}}{4\pi\rho UR} \left[ \frac{\mathbf{R}^{2} + \mathbf{y}_{N}^{2}}{\mathbf{R}\mathbf{y}_{N}^{2}} + \frac{\mathbf{R}\mathbf{x}}{\mathbf{y}_{N}^{2}(\mathbf{x}^{2} + \beta^{2}\mathbf{y}_{N}^{2})^{\frac{1}{2}}} - \frac{\sum_{q=0}^{\infty} \sum_{p=0}^{\infty} \frac{(-1)^{p} \mu_{2}^{\prime}(p+q+1) \mathbf{y}_{N}^{2}(\mathbf{x}/\beta)^{2p+1}}{q!(q+1)!(2p+1)!2^{2q}\mathbf{g}^{2}(p+q+1)} \right] . \quad (3.207)$$

where  $\mu_{21}' = -\frac{1}{(21+1)\pi} \int_{0}^{\infty} \frac{t^{21-2}(1+t^{2})}{[I_{1}'(t)]^{2}} dt$ = -0.999 , -1.627 , -9.78 , -120.8 for f = 1, 2, 3, 4.

165

a set a set of the the the set of the set

1

WISSENSER MAD

Since the spanwise loading is approximately constant,  $L_N$  is replaced by elements  $\frac{1}{2}\rho U^2 l(x_N)$ ,  $\delta x_N$ ,  $\delta y_N$  in Equation (3.207) and the total downwash along the centre line becomes

$$\frac{w(x)}{U} = \int_{0}^{\infty} \frac{i(x_{\rm N})}{8\pi R} \int_{-R}^{R} \frac{4\pi\rho \partial R}{L_{\rm N}} w_{\rm N}^{(1)} (x - x_{\rm N}, 0) \, dy_{\rm N} dx_{\rm N}$$

$$= \int_{0}^{\infty} \frac{l(x_{\rm R})}{8\pi R} \left[ -\frac{2(\beta^{2}R^{2} + (x - x_{\rm N})^{2})^{\frac{1}{2}}}{x - x_{\rm N}} - \frac{\sum_{q=0}^{\infty} \sum_{p=0}^{\infty} \frac{(-1)^{p} \mu_{2(p+q+1)}^{\prime}}{q! (q+1)! (2p+1)! (2q+1)2^{2q-1}} \left( \frac{x - x_{\rm N}}{\beta R} \right)^{2p+1} \right] dx_{\rm N} ; \quad (3.208)$$

special care has been taken with the integration across  $y_{H} = 0$  (Ref.3.42). In general the integral equation (3.208) would have to be solved for the chordwise load distribution  $l(x_{H})$  with w(x)/U corresponding to the camber line of the wing. But the approximation is made that terms of order  $[(x-x_{H})/2R]^{3}$  and higher are negligible, so that only p = 0 is retained and Equation (3.208) reduces to

$$-\frac{\Psi(\mathbf{x})}{U} = \frac{\beta}{4\pi} \int_{0}^{C} l(\mathbf{x}_{\mathrm{N}}) \left\{ \frac{1}{\mathbf{x} - \mathbf{x}_{\mathrm{N}}} + \frac{\mathbf{x} - \mathbf{x}_{\mathrm{N}}}{\beta^{2} \mathrm{R}^{2}} \left( \frac{1}{2} + \sum_{q=0}^{\infty} \frac{\mu_{2(q+1)}^{\prime}}{q! (q+1)! (2q+1)2^{2q}} \right) \right\} d\mathbf{x}_{\mathrm{N}}$$
$$= \frac{\beta}{4\pi} \int_{0}^{C} l(\mathbf{x}_{\mathrm{N}}) \left\{ \frac{1}{\mathbf{x} - \mathbf{x}_{\mathrm{N}}} - \frac{0.579(\mathbf{x} - \mathbf{x}_{\mathrm{N}})}{\beta^{2} \mathrm{R}^{2}} \right\} d\mathbf{x}_{\mathrm{N}} . \qquad (3.209)$$

To this approximation the wall interference is precisely that which would arise in a two-dimensional tunnel of height 1.686R. By a corresponding treatment the interference to the stream velocity is precisely the solid blockage of the same model in a twodimensional tunnel of height 1.558R. The interference corrections are thus

$$\Delta \alpha = \frac{0.289}{8\pi\beta} \left(\frac{c}{R}\right)^2 (C_{LL} + 4C_{BL})$$

$$\Delta C_{LL} = -\frac{0.289c^2C_{LL}}{4\beta^2 R^2}$$

$$\Delta C_{BL} = \frac{0.289c^2C_{LL}}{16\beta^2 R^2}$$
(3.2i0)

where  $C_{LL}$  and  $C_{mL}$  are the local lift coefficient and pitching moment coefficient about the quarter-chord axis and there are additional terms to account for blockage effects in Equations (51) and (52) of Reference 3.42.

The analysis for the swept wing spanning a rectangular tunnel in incompressible flow is more straightforward. Again the spanwise loading is taken to be uniform, but it is represented simply by a bound vortex along the quarter-chord line. Dannenberg<sup>3.57</sup> gives the interference upwash at the centre section in the form

163

1

$$\frac{\mathbf{w}_{i}}{\mathbf{U}} = \frac{SC_{LL}}{8\pi b^{2}} \sum_{-\infty}^{\infty} \frac{(-1)^{\mathbf{B}+\mathbf{n}_{\mathbf{B}}} \tan \Lambda}{\mathbf{m}^{2} \sin^{2}\Lambda + (\mathbf{n}h/b)^{2}} \left[ \frac{\mathbf{m} \cos^{2}\Lambda + \frac{1}{2}}{\left\{\frac{1}{2} \tan^{2}\Lambda + (\mathbf{m} + \frac{1}{2})^{2} + (\mathbf{n}h/b)^{2}\right\}^{\frac{1}{2}}} - \frac{\mathbf{m} \cos^{2}\Lambda - \frac{1}{2}}{\left\{\frac{1}{4} \tan^{2}\Lambda + (\mathbf{m} - \frac{1}{2})^{2} + (\mathbf{n}h/b)^{2}\right\}^{\frac{1}{2}}} \right].$$
(3.211)

As a result of the swept lifting line the interference velocity also has streamwise and sidewash components which are formulated. The final correction to incidence at y = 0 is

$$\Delta \alpha = \frac{\mathbf{k}_1 C_{LL}}{1 + \mathbf{k}_2 C_{LL}} \qquad (3.212)$$

where  $k_i = 0.43^{\circ}$  and  $k_2 = -0.084$  for the sweepback  $\Lambda = 45^{\circ}$ .

Figures 3.24(a) and 3.24(b) show the uncorrected and corrected curves of  $C_{LL}$ against  $\alpha$  from the respective investigations. In Reference 3.42 the corrections of Equations (3.210) are applied with further allowance for solid blockage to measurements on two sizes of model, c/R = 0.71 and 1.25, in low-speed flow. The corrected results in Figure 3.24(a) collapse on to a single curve spart from uncertainties near maximum lift, and Vincenti and Graham also find satisfactory agreement between the two corrected curves of  $C_{mL}$  against  $C_{LL}$ . In Reference 3.57 the experimental lift curve is corrected by Equation (3.212) and compared with that for an unswept wing of the same NACA 63<sub>1</sub>-012 aerofoil section normal to the leading edge with the sole sweep factor  $\cos \Lambda$ . The agreement in Figure 3.24(b) is remarkably good, and the corrected experiments in Reference 3.57 also give satisfactory comparisons with the pitching moment and wake drag of the unswept aerofoil. Both configurations can be justified as means of estimating two-dimensional flow.

## 3.7 EXPERIMENTAL CONSIDERATIONS

ł

Although the study of wind-tunnel wall interference arose directly from experiment, it rapidly became a source of absorbing mathematical problems. In some respects the ensuing theoretical developments have outstripped the needs of many experimenters to whom wall interference is one of several corrections to be applied; these and other factors set a limit to the accuracy of the investigation. Moreover, there are types of experiment in which the air flow is so complicated that only the principal interference corrections are likely to have significance. Flow separation due to high incidence, leading-edge sweepback, strong shock waves or bluffness of the model occur in most programmes of wind-tunnel testing. Any one of these may cause serious violation of the linear assumptions implicit in the preceding formulae. The remarks in Section 3.7.1 concern the experimental evidence of wall interference at the stalling incidence. Other aspects of separated flow form the subject of Chapter VII.

Naturally the precision of calculated interference parameters is far greater than that of any experimental verification of the underlying theory. The available information for closed and open tunnels is summarized in Sections 3.7.2 and 3.7.3 respectively. Although much of the material is inconclusive, the results have some practical interest. It is pertinent that, especially for open tunnels, there is more evidence of underestimation than overestimation of interference effects. Neglect of streamline curvature

is perhaps more prevalent than it should be. Simple approximations cover this and other refinements, and there can be little justification for ignoring them.

### 3.7.1 High Lift

There are many factors to complicate wall interference on wings at high lift. The attitude of the model may introduce off-centre effects due to vertical displacement. If the chord of the wing is large, the wing and its principal image in a closed roof will represent a divergent passage and so tend to cause flow separation from the upper surface of the wing. The high lift will increase the magnitude of spanwise variations in interference upwash and the chordwise variation known as otreamline curvature. The residual effects of these on the stalling characteristics of the wing are usually unpredictable, but could be important. Furthermore, the trailing-vortex system may suffer pronounced distortion at high incidence, and so may the boundary of an open jet. All these uncertainties leave little confidence in the standard interference corrections.

Glauert (Ref.3.1, p. 38) gives a clear qualitative picture of the effect of spanwise variation of interference upweak on the stalling of lifting-line models of large span. A wing of elliptical planform, having a constant effective incidence throughout the span, can be expected to develop a tip stall in a closed tunnel and a root stall in an open tunnel. A rectangular wing normally stalls at the root where the effective incidence is greatest; in a closed tunnel the interference upwash at the root will often be less than the weighted mean  $\Delta \alpha$  so that the corrected stalling incidence is strictly too high, and the reverse is true in an open tunnel. Conversely, with the correction  $\Delta \alpha$ , a highly tapered wing of large span in a closed tunnel should exhibit conservative stalling characteristics, while tests in an open tunnel should be unduly optimistic.

In the derivation of the interference corrections in Section 3.3.3 it is mentioned that the correction to incidence  $\Delta \alpha$  is in a sense quite arbitrary. It is convenient to choose it so that  $\Delta C_L = 0$ ; if some other value of  $\Delta \alpha$  were taken, the residual corrections would be different, but within the framework of linearized theory the corrected aerodynamic data would be consistent. Such considerations no longer apply near the stall, and the theory of wall interference is incapable of determining the effect of wall constraint on maximum lift. Therefore Bradfield, Clark and Fairthorne<sup>3.58</sup> carried out an experimental investigation of  $(C_L)_{MAZ}$  on rectangular wings of various sections in different sizes of tunnel. Their results in closed rectangular tunnels satisfy the empirical negative correction

$$\Delta(C_{\rm L})_{\rm max} = -0.76 \, {\rm S/b^2} \qquad (3.213)$$

The second s

within ±0.005, but there is no systematic evidence of any corresponding correction in an open circular tunnel. Although Equation (3.213) would appear to be consistent with the expected higher stalling incidence in a closed tunnel, a further series of experisents by Adamson, Piper and Brown<sup>3.59</sup> on rectangular wings revealed no evidence of systematic wall interference on maximum lift. They suggest that, if full account of wake blockage were taken in Reference 3.58, the correction to  $(C_L)_{max}$  would practically disappear. Furthermore, tests in Reference 3.59 with off-centre models show little effect on  $(C_L)_{max}$ . Since wake blockage can usually be neglected in open tunnels (Section 5.5.3), the results of Reference 3.58 for closed and open tunnels become compatible.

168

A convincing quantitative study of the effect of spanwise variation of interference upwash has been published by Stewart (Ref.3.60; 1941), who considers a large liftingline model with elliptic spanwise loading in two closed circular tunnels. The interference upwash at the wing root on the tunnel axis

$$(w_i/U)_r = \frac{SC_L}{8C}$$

is independent of  $\sigma = s/R$ , and it is supposed that this incremental correction has been applied to the incidence. Stewart obtains the residual interference at the wing tip

$$(w_{i}/U)_{t} - (w_{i}/U)_{r} = \frac{SC_{L}}{C} \left[ \frac{1}{4\sigma^{u}} \left\{ (1-\sigma^{u})^{-\frac{1}{2}} - 1 \right\} - \frac{1}{8} \right]$$
 (3.214)

which gives

$$\frac{(w_i/U)_t - (w_i/U)_r}{SC_i/C} = 0.047 \text{ and } 0.207$$

for  $\sigma = 0.77$  and 0.93 respectively. In the two corresponding experiments on an elliptical planform of aspect ratio 5.88 the tip stall was observed carefully with tufts to occur suddenly at corrected incidences  $\alpha = 17.3^{\circ}$  and 15.1° respectively. Figure 3.25 shows the results and indicates how perfectly Equation (3.214) accounts for the difference in stalling angle. It also illustrates two important conclusions, that the correction to stalling angle is unlikely to be large unless  $\sigma > 0.8$ , and that no deductions about ( $C_L$ )<sub>max</sub> can be made; the larger observed ( $C_L$ )<sub>max</sub> corresponds to the lower stalling angle. Some additional information for a closed elliptical tunnel (b/h = 1.37) is found in Reference 3.40.

Davis and Sweberg<sup>3.61</sup> have studied interference effects in an open circular tunnel by comparisons of full-scale tunnel and flight measurements of maximum lift for a complete aircraft. The experiments are complicated by the large size of the aircraft and the high  $(C_L)_{max} = 1.99$  obtained in flight with flaps and landing gear extended. . With these retracted,  $(C_L)_{max} = 1.36$  and 1.39 are deduced from tunnel and flight respectively, when allowance is made for Reynolds number and time rate of change of incidence. In the landing condition, however, the corresponding tunnel result  $(C_L)_{max} = 1.80$  is as much as 0.19 below the value in flight. Half this discrepancy has been explained by two unusual corrections. The first is associated with a large spanwise variation in dynamic pressure ahead of the upcraft due to a distortion of the circular jet; weighted in accordance with the spanwise loading, this gives a large negative correction to the dynamic pressure and an increment of 0.05 in  $(C_L)_{max}$ . second correction of  $\Delta(C_L)_{max} = 0.95$  is stated to arise from the negative camber associated with streamline curvature; a corresponding increment of 0.03 eliminates the small discrepancy in  $(C_L)_{max}$  with flaps and leading gear retracted. The only firm conclusion is that the area ratio S/C of the planform to the jet should be well below 0.2. It would be realistic to impose restrictions  $S/C \le 0.10$  for open tunnels,  $3/C \le 0.15$  for closed tunnels, and never to assume better accuracy than  $\pm 0.02$  in (CL) asx .

S. S. S.

١,

## 1...0

### 3.7.2 Closed Tennels

Nucle of the theory of wall interference in closed tunnels is now accepted without question. Nevertheless, experimental checks on the theoretical formulae for  $\Delta \alpha$  and  $\Delta C_{\rm D}$  for wings of high aspect ratio are important in the historical development of the subject. In Figures 11 and 12 of Reference 3.1, Glauert summarizes the experiments of Cowley and Jones<sup>3.62</sup> on a biplane model of span 3 ft tested in closed square tunnels of side 4 ft and 7 ft. Glauert applied the corrections of Equation (3.6) with  $\delta_0 = (\delta_0)_{\rm E}$ to bring the lift and drag curves from the two tunnels into satisfactory agreement. He also describes the results of experiments by Higgins<sup>3.63</sup> on three rectangular wings of aspect ratio 5 ( $\sigma \leq 0.6$ ) at constant Keynolds number in a closed circular tunnel: after trying several empirical corrections without success, Higgins concluded that the theoretical formulae ( $\delta_0 = 0.125$ ) gave the best correlation.

In later experiments the method of analysis is to determine empirically the corrections

$$\Delta \alpha = \delta_{\alpha} \operatorname{sc}_{L} / C$$

$$\Delta c_{p} = \delta_{p} \operatorname{sc}_{L}^{2} / C$$
(3.215)

that reconcile the measurements. It is instructive to compare the results with the theoretical values from Equations (3.85), whence

$$\delta_{\alpha} = (\delta_{0})_{B} \left( 1 + \frac{\delta_{1}}{\delta_{0}} \frac{\overline{c}}{2\beta h} \right)$$

$$\delta_{D} = (\delta_{0})_{E}$$

$$(3.216)$$

where  $\delta_i/\delta_0$  is given in Figure 3.4 for rectangular tunnels. Three examples are taken for square tunnels.

Scurce	a	A	Experiment		Theory	
	, v		δ <sub>α</sub>	δ <sub>D</sub>	δ <sub>α</sub>	8 <sub>0</sub>
Reference 3.8, Case I	J. 40	4	0.140	0.140	0.152	0.140
Reference 3.8, Case 1	0.60	6	0.142	0. 125	0. 160	0. 147
Reference 3.64, Test 20	0.75	δ	0. 170	0.1 <sub>7</sub> ?	·0. 175	G. 158

In the last example the experimental  $S_D$  varied from 0.15 to 0.19. The tabulated results represent a reasonable confirmation of the theory.

The investigation of Sivells and Deters<sup>3.47</sup> justifies the use of part-span models in a closed circular tunnel with a reflection plane. Figure 3.15 illustrates the reduction in lift interference when a half-wing model is used. The rolling moments due to alleron deflection  $\xi$  will also have to be corrected to conditions of antisymmetrical spanwise loading, as discussed briefly in Section 3.3.6. Furthermore,

a the second second and the second second

171

TOMESSION AND AND AND

if a part-span model is used, there will be a third correction to allow for the change in planform. In the example of Reference 3.47, sketched in Figure 3.26, these corrections are respectively

$$\Delta C_1 = -0.0816 C_1$$

and factors 0.949 and 0.934, which combine to give a free-stream rolling moment coefficient

$$(C_1)_{y} = 0.814 C_1$$
.

The success of this large correction is illustrated in Figure 3.26, where corrected results (+) from the part-model test at a Reynolds number  $R_N = 3.9 \times 10^6$  are compared with the full curve against  $\xi$  from experiments on a complete model at  $R_N = 4.7 \times 10^6$ .

Reference 3.4 sets out to put wall interference and scale effect into perspective. Table 5B (Ref. 3.4, p. 45) illustrates how the two corrections may often oppose each other when the tunnel boundary is closed. It may be typical that the wall interference corrections to serodynamic forces and moments due to incidence are as important as corrections to full-scale Reynolds number. On the other hand, the characteristics of deflected control surfaces are likely to be influenced less by wall interference than by scale effect; the case of rolling moment on a part-span or half model is a notable exception.

#### 3.7.3 Open Tunnels

Because open tunnels were used extensively during the period in which the theory of wall interference was developed, more attention has been given to the experimental verification of the theory for open and semi-closed tunnels than for completely closed tunnels. Knight and Harris<sup>3,65</sup> have tested three rectangular wings of aspect ratio 5 with span ratios  $\sigma = 0.45$ , 0.60 and 0.75 in four open tunnels of circular, rectangular (b =  $h\sqrt{2}$ ) and oval section (b =  $h\sqrt{2}$  and b = h with circular sides). Their analysis follows Equations (3.215) on the assumption that the results in the circular tunnel can be extrapolated to zero interference. In the present theoretical analysis Equation (3.216) is used, but  $\delta_1/\delta_0$  is now defined by Equation (3.31) with  $\delta_1^{(2)}/\delta_0^{(2)}$ from Pigure 3.4. The comparisons in Figure 3.27(a) for the open circular tunnel are reasonably satisfactory. There is further experimental confirmation of the theoretical lift interference in open circular tunnels published by Tani and Taima<sup>3,66</sup>; in Figure 3 of Reference 3.66 they make allowance for streamline curvature and obtain consistent lift and drag curves for three rectangular wings ( $\sigma = 0.5$ , 0.6, 0.7) of aspect ratio 5. The results of Reference 3.65 are less convincing for the rectangular and oval sections, but experimentally  $-\delta_{\alpha}$  tends to be larger than  $-\delta_{\alpha}$  and the difference indicates the order of magnitude of the streamline-curvature correction. Figure 3.27(b) compares the experimental results for the oval tunnel (b = h/2) with calculations based on Equations (3.216) and the approximate theory of Sanuki and Tani in Section III of Reference 3.35 for the particular open boundary.

The investigation of van Schliestett<sup>3.8</sup> is particularly interesting. Not only does he give a satisfactory confirmation of the theoretical interference parameters for an open square tunnel, but obtains experimental results that contradict the theory of

Reference 3.6 for a square jet with solid side-walls (type (4) in Section 3.2.2) and a square jet with one solid horizontal boundary. The double achievement of detecting two theoretical errors by experimental means and correcting both successfully deserves special mention. Other results for semi-closed tunnels are found in the experiments described by Tani and Taima in the second part of keference 3.66 for a circular boundary having a symmetrical closed portion below the model. The related theory is discussed briefly in the fourth paragraph of Section 3.4.3; Kondo has applied his theory to the experiments of Reference 3.66 and obtains very consistent lift and drag curves in Figures 13(a) and 13(b) of Reference 2.14 for three sizes of model and five amounts of closed portion. For  $\sigma = 0.5$  he confirms the elimination of lift interference when the lower 38.5% of the circle is closed.

In Paference 3.67, Silverstein and Katzoff describe tests on rectangular wings in a full-scale open circular tunnel and in a 1/15-scale model of the tunnel. The assumption that the free jet is of infinite length is shown to be adequate for the purpose of obtaining interference upwash at the lifting line. Traverses of downwash behind the wings have confirmed that it is essential to consider the finite length of the open tunnel in relation to interference corrections at a tail plane, and qualitative agreement is found with a theory of Weinig (Ref. 3.68; 1936). The experiments in Reference 3.67 led presumably to the theoretical developments in Reference 3.5, discussed in Section 3.2.3 and illustrated in Figure 3.6.

Finally we consider tests on three equilateral delta wings (A = 2.51) by Jones and Miles<sup>3.69</sup> in a low-speed open circular tunnel, to which were applied the simple corrections in Equation (3.6) with  $\delta_0 = -0.125$ . For each of the span ratios  $\sigma = s/R = 0.361$ , 0.500 and 0.664 the interference corrections to incidence based on Equation (3.35) are now calculated from the formula

$$\Delta \alpha = (\delta_0)_{\rm E} \left[ 1 + \frac{\lambda \overline{c}}{4\beta R} \frac{\delta_1}{\delta_0} \right] \frac{{\rm SC}_{\rm L}}{C}$$
(3.217)

with  $\delta_1/\delta_0 = 1.594$  and  $\lambda = 1.25$  interpolated from Table 3.III. The application of this correction in place of the first of Equations (3.6) increases the lift slope by the approximate factor

$$\mathbf{F} = \mathbf{i} - \frac{\mathbf{s}}{\mathbf{c}} \frac{\partial \mathbf{C}_{\mathbf{t}}}{\partial \alpha} \left[ \left( \delta_0 \right)_{\mathbf{g}} \left( \mathbf{1} + \mathbf{0} \cdot \mathbf{498} \frac{\mathbf{\overline{c}}}{\mathbf{p}} \right) + \mathbf{0} \cdot \mathbf{125} \right]$$
(3.218)

which is calculated below with the experimental  $\partial C_{\mu}/\partial \alpha = 2.66$ .

Model	σ	s/C	~(S <sub>0</sub> ) <sub>E</sub>	ē∕R	P
A	0.361	0.072	0.1255	0. 312	1.004
В	0.500	0. 138	0. 1267	0.453	1.011
C	0.664	0. 243	0. 1302	0.575	1.027

It is stated in Reference 3.69 that respective empirical factors 1.01 and 1.05 were applied to  $C_{\rm L}$  on the models B and C to give consistent lift curves. Allowance for streamline curvature through the factor F is now seen to account for half the effect

of these empirical factors. As regards drag, consideration is given to the remarks at the end of Section 3.6.1 and we examine the consequences of applying the correction in Equation (3.185),

$$\Delta C_{\rm p} = C_{\rm L} \Delta \alpha , \qquad (3.219)$$

at the higher values of  $C_L$  when there is flow separation along the whole leading edge. When this speculative correction is used in place of the second of Equations (3.6) for the largest model C, the solid circles are obtained in place of the open circles in Figure 3.28. The results for the three models are seen to be in acceptable agreement while  $C_L$  is fairly small, but the correction in Equation (3.219) reduces the scatter for  $C_L > 0.4$ . Unfortunately  $R_M = 0.59 \times 10^6$  is the highest Reynolds number at which all three models were tested over the full range of  $C_L$ . The results in Figure 3.28 are far from conclusive, but they are included as a warning that the standard corrections to  $C_D$  are too small when leading-edge vortices are present.

### 3.8 INDEX OF FORMULAE AND DATA

The process of evaluating the interference velocity field depends primarily on the tunnel section, but the requirements are influenced greatly by the configuration of the model, by the aerodynamic quantities to be measured and above all by the accuracy of the experiments. When small inaccuracies can be tolerated, many of the simpler formulae for interference parameters are most useful. These are grouped according to tunnel section for various purposes in the first table below; although this includes reference to the numerical data in the present tables and figures, there are important additional sources listed in Table 3.IV. The second table concerns the more elaborate formulae that define more detailed calculations of the distribution of interference upwash for the three main types of tunnel. When the basic numerical data relating to these two tables are known, sany useful approximate sud more accurate formulae for the type of tunnel. These and wore special formulae are listed in the third table together with calculated results.

Each table gives the relevant equations and related tables and figures. For example the entries in the first table under rectangular tunnels for a small wing associate Equation (3.17) with Figure 3.2, Equation (3.20) with Table 3.11 and also with Figure 3.2.

173

したで、「ないた」というない。「ない」には、「ない」には、「ない」である

 $\mathcal{A}$ 

2.4

NS .-

## 3.8.1 Simple or Approximate Interference Velocity

	Rectangular tunnels		Octa tun	gonal nels	Elliş and b	otical ipolar
	Eqn	T&F	Eqn	T & F	Egn	T&F
Small wing	17	<b>F</b> 2	28		29	<b>F</b> 5
	20	<b>T2</b>	32		30	<b>P</b> 5
	20	<b>F</b> 2	•		31	<b>F4</b>
			<b>]</b> -		31	P5 •
					136	
					137	
Lifting line	83	<b>F11</b>	107	<b>F</b> 12	115	<b>F</b> 13
	84	<b>P</b> 2			117	<b>F</b> 13
	156	<b>T6</b>			118	
					143	<b>P</b> 15
Lifting surface	156	76	109			
Slender wing	176				176	
	177				178	
Tail plane	191	<b>F</b> 21	199			<b>F6</b>
1	192		201		•	
	194	•				
Ground effect		<b>F</b> 11	206			
1	203	F22				
	203	F23				
	204					
	205					
Off-centre wing	25	<b>P</b> 3	201		114	
	27	<b>F</b> 3				
	200					

acores

	Rectangular tunnels		Rectangular tunnels		Octagonal tunnels	Cir and l	cular Dipolar	Апу
	Eqn	T&F	Egn	Eça	T&P	Ecn		
Lifting line	82		101	116		43		
	88	<b>T</b> 5	102	146		44		
	91		103	167		48		
			104	214	¥25			
Lifting surface	87		108	138		42		
	89	TS		139		47		
	92				1	50		
Complete sidel	95			124		51		
	96			to		186		
	93			126	1	189		
	195			127 tə				
	196			131				
	2 <b>97</b>			140				
				147				
				148				

# 3.3.2 Elaborate or Accurate Interference Upwash

and the second

.

-

N. W. W.

172

	Δα		۵c		Δc <sub>i</sub>		$\Delta C_{\mathbf{p}} \triangleq \Delta C_{\mathbf{p}}$	
	Sign	T&F	Ego	T & F	Eqn	T&F	Eqn	Υ& F
General sod	6		34		161		6	
approximate	8		35				37	
	34		50				164	
	35		182	F20			166	
	182	<b>F2</b> 0	183				185	
	183		188					
General and	55		57		62		56	
more accurate	67	•	58		67		63	
	73		59		76		70	
	179	₽20	67				71	
			179	<b>F</b> 20			74	
							76	
Rectangular	85		155		170	P17	85	
tunnels	150	P16		<b>F</b> 19			154	
	151	F17		<b>T</b> 7				F17
		F18						į
-		<b>T</b> ?						
Octagonel	110		112				111	
Circular	215	₽27					168	
							215	F27
								F28
Bipolar	144	<b>P</b> 15				<b>F</b> 26		

## 3.8.3 Corrections to Measured Quantities

There are also general formulae for  $\Delta c_{\rm H}$  in Equation (3.77) and  $\Delta \alpha_{\rm t}$  in Equation (3.198).
REFERENCES

27.1

. .

ĺ

3.1	Glauert, H.	Wind Tunnel Interference on Wings, Bodies and Airscrews. ARC R & M 1566, 1933.
3.2	Golástein, S. Young, A.D.	The Linear Perturbation Theory of Compressible Flow, with Applications to Wind-Tunnel Interference. ARC R & M 1909, 1943.
3.3	Göthert, B.H.	Windkanslkorrekturen bei hohen Unterschallgeschwindigkeiten unter besonderer Berücksichtigung des geschlossenen Kreiskanals. Deutsche Luftfahrtforschung Forschungsbericht 1215, 1940. (Translated as NACA Tech. Hemo. 1300.)
3.4	Bryant, L.W. Gærner, H.C.	Control Testing in Wind Tunnels. ARC R & M 2881, 1950.
3, 5	Kstzoff, S. et al.	Linzar Theory of Boundary Effects in Open Wind Tunnels with Finite Jet Lengths. NACA Report 378, 1950.
3.5	Theodorsen, T.	The Theory of Wind-Tunnel Interference. NACA Report 410, 1931.
3.7	Rosenhend, L.	Interference Due to Walls of a Wind-Tunnel. Ploo. Roy. Soc., Series A, Voi. 142, 1933, pp. 308-320.
3.8	van Schliestett, G.	Experimental Verification of Theodorsen's Theoretical Jet-Boundary Correction Factors. NACA Tech. Note 506, 1934.
3.9	Garner, H.C. ct al.	An Anomaly in the Theory of Tyunel-Mall Interference on a Lifting Wing. Aeron. Quart. Vol. 14, 1963, pp. 31-40.
3.10	Acum, W. E. A. Garner, H. C.	Approximate Wall Corrections for an Oscillating Smept Wing in a Wind Tunnei of Closed Circular Section. ARC CP 184, 1954.
3.11	Batchelor, G.K.	Interference in a Wind Tunnel of Octugonal Section. Report ACA-1 (Australia), 1941.
3.12	Garner, R.C.	Note on Interference in a Wind-Tunnel of Octagonal Section. ARC Report 6659, 1943. (Unpublished).
3. 13	Gent, B.L.	Interference in a Wind Tunnel of Regular Octagonal Section. Report ACA-2 (Australia), 1944.
5.14	Kondo, K.	The Wall Interference of Wind Tunnels with Boundaries of Circular Arcs. Aero. Res. inst. (Tokyo) Report 126, 1935.
3, 15	Garner, H.C.	Some Remarks on Glowert's General Interference Theorem. Aeron. Quart. Vol. 14, 1963, pp.41-44.

177

5 MA - E -

-

-

1000

. (1) a X

THE REAL PROPERTY OF

ちょうろう ちょうちょうちょう いちないない ちょうちょう ちょうちょう しょうちょう しょうちょう しょうちょう しょうしょう

-

1

and the second second second in the second secon

Ξ÷.

3. 18	Glauert, H. Hartshorn, A.S.	The Interference of Wind Channel Walls on the Doompash Angle and the Tail-Setting to Trim. ARC R & M 347, 1924.
3.17	Acune, W. E. A.	Corrections for Symmetrical Swept and Tapered Wings in Rectangular Wind Tunnels. ARC R & M 2777, 1950.
3.18	Garner, H.C. Acum, W.E.A.	Interference Corrections for Asymmetrically Loaded Wings in Closed Rectangular Wind Tunnels. ARC R & M 2948, 1953.
3, 19	Eisenstadt, B.J.	Boundary-Induced Upwash for Yawed and Swept-Back Wings in Closed Circular Wind Tunnels. NACA Tech. Note 1265, 1947.
3.20	Katzoff, S. Hannah, N.R.	Calculation of Tunnel-Induced Upwash Velociti <sup>*</sup> : for Swept and Yazed Wings. NACA Tech. Note 1748, 1948.
3, 21	Loos, H.G.	Tunnelwall Interference for Aerofoils and Aerofoil-Fuselage Combinations in a Tunnel with an Octagonal Section for Incompressible Flow. NLL (Netherlands) Report A 1024, 1951.
3. 22	Swanson, R.S. Toll, T.A.	Jet-Boundary Corrections for Reflection-Plane Models in Rectangular Wind Tunnels. NACA Keport 770, 1943.
3.23	Sanders, J. Pounder, J.R.	Wall Interference in Wind Tunnels of Closed Rectangular Section. MRC (Canada) Report AR-7, 1949.
3. 34	Sanders, J. Pounder, J.R.	Wall Interference in Wind Tunnels of Closed Rectangular Section. Development of Equations. NRC (Canada) Report AR-11, 1951.
3. 25	Smanson, R.S.	Jet-Boundary Corrections to a Yamed Model in a Closed Rectangular Wind Tunnel. NACA AKR (WR-L-603), 1943.
3.28	Sivells, J.C. Saïzi, R.N.	Jet-Boundary Corrections for Complete and Semispan Swept Wings in Closed Circular Wind Tunnels. NACA Tech. Note 2454, 1951.
3.27	Sranson, R.S. Schuldenfrei, M.J.	Jet-Boundary Corrections to the Downwash behind Powered Wodels in Rectangular Wind Tunnels with Numerical Values for 7- by 10- Foot Closed Wind Tunnels. NACA ARR (WR-L-711), 1942.
3. 28	Graban, D.J.	Turcel-Wall Corrections to Rolling end Yawing Moments Due to Aileron Deflection in Closed Nectangular Wind Turnels. NACA ARR 4F21 (WR-A-14), 1945.
3, 29	Evans, J.M.	Stability Derivatives. Find Tunnel Interference on the Lateral Derivatives 1 <sub>p</sub> , 1 <sub>r</sub> , and 1 <sub>y</sub> with Particular Deference to 1 — Report 272-33 (Australia), 1947.

. 1

NUM LAN

The second second

Ż

S

3.30	Cowley, W.L. McMillan, G.A.	Effect of Wind Tunnel Wall Interference on the Pitching Moments of Large Models in the Duplex Tunnel. ARC & & M 1639, 1934.
3.31	Olver, P.W.J.	Transformation of Gertain Series Occurring in Aerodynamic Interference Calculations. Quart. J. Mech. appl. Math., Vol.2, 1949, pp.452-457.
3.32	Silveratein, A. White, J.A.	Wind-Tunnel Interference with Particular Reference to Off-Centre Positions of the Wing and to the Downwash at the Tail. MACA Report 547, 1935.
3.33	Brown, W.S.	Wind Tunnel Corrections on Ground Effect. ARC R & M 1865, 1938.
<b>5. 34</b>	Batchelor, G.K.	Interference on Wings, Bodies and Airscrews in a Closed Turnel of Octagonal Section. Report ACA-5 (Australia). 1944.
3.35	Esauki, N. Teni, I.	The Wall Interference of a Wind Tunnel of Elliptic Cross Section. Proc. Phys. Math. Soc. Japan, Vol.14, 1932, pp.592-603.
3.36	Rosenhead, L.	The Aerofoil in a Wind Tunnel of Elliptic Section. Proc. Roy. Soc., Series A. Vol.140, 1933, pp.579-604.
3.37	Riegels, 7.	Korrekturfaktoren für Windkanäle elliptischen Querschnitts mit teilweis- offener und teilweise geschlossener Mebstrecke. Luftfahrtforschung, Vol. 16, 1939, pp. 26-30. (Translated as NACA Tech. Memo. 1310.)
3. 58	Lotz, I	Korrehtur des Abwindes in Windkonälen mit kreisrunden oder elliptischen Querschnitten. Luftführtforschung, Vol. 12, 1935, pp. 250-284. (Translated as NACA Tech. Meso. 801.)
3.39	von Zármán, T. Burgers, J.M.	General Aerodynamic Theory - Perfect Fluids, Airfoils and Airfoil Systems of Finite Span, "Aerodynamic Theory" (ed. W.F. Durand), Vol. II, Div. E, Chap. IV, Sect. 42-44. Julius Springer, Berlin, 1935.
3.40	Gavin, J.G. Hensel, R.W.	Elliptic Tunnel-Wall Corrections on Drag and Stall. J. mero. Sci., Vol.9, 1942, pp.533-537.
3.41	von Baranoff, A.	Zur Frage der Konalkorrektur bei kompressibler Unterschallströmung. Jahrbuch der Deutschen Luftfahrtforschung, 1941, pp.1672-1677.
3.42	Vincenti, W. Graham, D.	The Effect of Wall Interference upon the Aerodynamic Characteristics of an Airfoil Spanning & Closed-Throat Circular Wind-Tunnel. NACA Report 849, 1948.

۲. ۲

..

189	
3.43 van der Bliek, J.A.	Notes on Half Modei Testing in Wind Tunnels. AGARD Report 298, 1959. Canadian Aeronautical Journal, Vol.6, 1960, pp.59-68.
3.44 Polhamus, E.C.	Jet-Boundary-Induced-Upwash Velocities for Swept Reflection Plane Models Mounted Vertically in 7- by 10- Foot, Closed, Rectangular Wind Tunnels. NACA Tech. Noto 1752, 1948.
3.45 Garner, H.C.	Note on the Interference on a Part-Wing Mounted Symmetrically on One Wall of a Wind Tunnel of Octagonal Section. ARC CP 5, 1947.
3.46 Davison, B. Rosenhead, L.	Wind funnel Correction for a Circular Open Jet Tunnel with a Reflexion Plate. Proc. Koy. Soc., Series A, Vol. 177, 1941, pp. 366-382.
3.47 Sivells, J.C. Deters, O.J.	Jet Boundary and Pian-Form Corrections for Partial-Span Models with Reflection Plate, End Plate or no End Plate in a Closed Circular Wind Tunnel. NACA Report 843, 1946.
3.48 Gillis, C.L. et al.	Charts for Determining Jet-Boundary Corrections for Complete Models in 7- by 10- Foot Glosed Rectangular Wind Tunnels. NACA ARR L5G31 (WR-L-123), 1945.
3.43 Lvon, H.M.	An Approximate Method of Calculating the Effect of the Interference of a Closed Tunnel on Hinge Moments of Control Surfaces. ARC Report 5803, 1942. (Unpublished).
3.50 Stewart, H.J.	A Correction to the Yawing Moment Due to Ailerons for Circular Wind Tunnels. J. aero. Sci., Vol.6, 1939, pp.329-331.
3.51 Berndt, S.B.	Wind Tunnel Interference Due to Lift for Delta Wings of Small Aspect Ratio. KTH (Sweden) Tech. Note Aero TN 19, 1950.
3.52 Tirumalesa, D.	Lift Interference for Low Aspect Ratio Wings in Transonic Wind Tunnels. J. aero. Soc. India, Vol.11, 1959, pp.79-82.
3.53 Saith, C.B.	Wind-Tunnel Wall Corrections for Wing-Body Combinations. J. sero. Sci., Vol.16, 1949, pp.237-242.
3,54 Gorgui, M.A.	Wind-Tunnel Interference for Wing-Body Combination. J. aero. Sci., Vol.28, 1961, pp.823-825.
3.55 Sanders, J.	Wind Tunnel Corrections on Ground Effect Tests. NBC (Canada) Report AR-5, 1948.
3.56 Albritton, J.P. Huber, P.J.A.	Wall Corrections for a Wing near a Ground Plate in a Circular Cross Section Wind Tunnel. Wright Air Development Center Tech. Report 55-280, 1955.

6.6. 3.6 Beer

-Comput

5

;

.

LOCH Lint

LY MACHT

3.57 Dannenberg, R.E.	Measurements of Section Characteristics of a 45° Swept Wing Spanning a Rectangular Low-Speed Wind Tunnel as Affected by the Tunnel Walls. NACA Tech. Note 2160, 1950.
3.58 Bradfield, F.B. et al.	Measurement of Maximum Lift in Closed Tunnels of Different Sizes, and in an Open Jet Tunnel. ARC R & M 1363, 1930.
3.59 Adamson, J.R. et al.	The Effect of Tunnel Size on Maximum Lift Coefficients. ARC Report 5436, 1942. (Unpublished).
3.60 Stewart, H.J.	The Effect of Wind Tunnel Wall Interference on the Stalling Characteristics of Wings. 5. mero. Sci. Vol.8, 1941, 79.426-428.
3.61 Davis, D.D. Sweberg, H.H.	Investigation of Some Factors Affecting Comparisons of Wind-Tunnel and Flight Measurements of Maximum Lift Coefficients for a Fighter-Type Airplune. NACA Tech. Note 1639, 1948.
3.62 Cowley, W.L. Jones, L.J.	An Experimental Test of the Prandtl Correction for Tunnel Wall Interference. ARC E & M 898: 1924.
3.63 Higgins, G.J.	The Effect of the Walls in Closed Type Wind Tunnels. NACA Report 275, 1927.
3.64 Theodcrsen, T. Silverstein, A.	Experimental Verification of the Theory of Wind-Tunnel Boundary Interference. NACA Report 478, 1934.
3.65 Knight, M. Harris, T.A.	Experimental Determination of Jet Boundary Corrections for Airfoil Tests in Four Open Wind Tunnel Jets of Different Shapes. NACA Report 361, 1930.
3.66 Tani, I. Taima, M.	Two Notes on the Boundary Influence of Wind Tunnels of Circular Cross Section. Aero. Res. Inst. (Tokyo) Report 121, 1935.
3.67 Silverstein, A. Katzoff, S.	Experimental Investigation of Wind-Tunnel Interference on the Downwash behind an Airfo. <sup>1</sup> . NACA Report 609. 1937.
3.68 Weinig, P.	Der Strahleinfluss bei offenen mindkanälen. Luftfabrtforschung, Vol.13, 1936, pp.210-213.
3.69 Jones, R. Miles, C.J.W.	Tests on Three Equilateral Triongular Plates in the Compressed Air Tunnel. ARC R & ¥ 2518, 1945.
3.70 Taylor, C.R.	Private Communication. 1966.

ł

Contraction and the Contraction of the Contraction

.

181

1

#### ADDITIONAL REFERENCES FOR CHAPTER III

**u**.y. '

Ί

ŧ

Adamson, J.E.	An Experimental Investigation of Wind Tunnel Interference in the RAE 5 ft Open Jet Circular Tunnel. ANC R & M 1897, 1941.
Bradfield, F.B. (editor;	Notes on the T:chnique Employed at the RAE in Low-Speed- Wind-Tunnel Tests in the Period 1939-1945. ARC R & M 2556, 1947.
Garner, H.C. Halliday, A.S.	An Investigation of a Part-Wing Test on an Aileron and Methods of Computing Aileron Characterisiics. ARC Report 5922, 1945. (Unpublished).
Halliday, A.S. Cox, D.K.	Note on the Sffect of Size and Position of End Plates on the Lift of a Rectangular Wing in a Wind Tunnel. ARC CP 305, 1955.
Malavard, L.	Étude de Quelques Problèmes Techniques Relevant de la Théorie des Àiles. Application a Leur Solution de la Méthode Rhéoélectrique. Chaps. IV and V. Publ. Sci. Tech. Win. Air Prance 153, 1939.
Muttrey, H.	Über die Grösse der Berichtigungsbeiwerte für Widerstand und Anstellwinkel bei Freistrahlen von elliptischem Querschnitt. Luftfi <sup>°</sup> rtforschung, Vol.12, 1935, pp.265-265. (Translated as ARC Report 2492.)
Narayana, B.L.	Interference Effects in an Infinite Cascade of Wings of Finite Span. J. Roy. aeron. Soc., Vol.66, pp.460-461, 1962.
Petraulea, N.N. Dumitrescu, L.	Influence of Wails on the Results of Tests on Annular Wings in Tunnels of Circular Cross-Section. Stud. Cercet. Mec. apl. (Rumania). Vol.8, 1957, pp.703-710.
Recant, I.G.	Wind-Tunnel Investigation of Ground Effect on Wings with Fiaps. Appendix: Tunnel-Wall Corrections for Test Installation Used. NACA Tech. Note 705, 1939.
Silverstein, A. Katzoff, S.	Design Charts for Predicting Downwash Angles and Wake Characteristics behind Plain and Flapped Wings. NACA Report 648, 1939.

182

1

÷

AND CONTRACTOR

A NUMBER AND ADDRESS .

ŧ,

1

T

- mon & and the second se

# TABLE 3.I

ł

15

Equivalent Incompressible Flow from Equation (3.5) when  $M = (1-\beta^2)^{\frac{1}{2}}$ 

Geometric Parameters						
Wing chord	c					
Wing semi-span	βs					
Wing area	₿S					
Aspect ratio	βΑ					
Wing thickness	βt					
Incidence	βa					
Aileron setting	βĘ					
Tunnel breadth	βъ					
Tunnel height	βh					
Tunnel area	β²c					
Distance from floor	ßd					

۶,

The equivalent angle of sweepback is  $\cot^{-1}(\beta \cot \Lambda)$ .

Aerodynamic Parameters							
Stream velocity	ρ						
Stream velocity	β <sup>-2</sup> τ						
Velocity perturbations	υ. β <sup>-1</sup> ν. β <sup>-1</sup> τ						
Upwash angle	β(Δα)						
Circulation Vortex strength Pressure difference Lift Vortex-induced drag Pitching moment Rolling moment	$ \vec{\Gamma} \\ \mathbf{K} \\ \beta^{-2}(\mathbf{p}_{u} - \mathbf{p}_{i}) \\ \beta^{-1}\mathbf{L} \\ \mathbf{D}_{v} \\ \beta^{-1}\mathbf{R} \\ \mathbf{L} $						
Lift coefficient	β <sup>2</sup> C <sub>L</sub>						
Noment coefficients	β <sup>2</sup> C <sub>R</sub> ,β <sup>2</sup> C <sub>i</sub>						
Drag coefficient	β <sup>2</sup> C <sub>Dv</sub>						

Ś

183

í,

and the second

Ser and the series of the seri

### TABLE 3.11

nh b	$-S_1\left(\frac{nh}{b}\right)$		nh b	$-S_1\left(\frac{nh}{b}\right)$
0.25	94.1122		1.30	0.0219
<b>0.30</b>	59.1543		1.35	9.0156
0.35	28.8239		1.40	0.9111 <sub>5</sub>
0.40	17.4569		1.45	0.0080
0.45	10.9781		1.50	9.0057
0.50	7.09765		1.55	0.0041
0.55	4.6851		1.60	0.6029
0.60	3. 1419		1.65	0.3921
0.65	2. 1329		1.70	0.0015
0.70	1.4618		1.75	0.0011
J.75	1.0095		1.80	0.9008
0.80	0.7014		1.85	0.0005 <sub>5</sub>
0.85	0.48975		1.90	0.0004
0.90	0.34335	-	1.95	0.0003
0.95	0.2415		2.00	G.0002
1.00	0. 1704		2.05	9.0001 <sub>5</sub>
1.05	0.1205		2.10	0.0001
1.10	0.0854		2.15	0.0001
1.15	0.0606		2, 20	0.0000 <sub>5</sub>
1.20	0.6431		2.25	0.00005
1.25	0.0307		2.30	0.0900

. .

Function S<sub>1</sub> from Equation (3.18) for the Evaluation of Streamline Curvature in Rectangular Tunnels

184

A PERCENTING

; ;

من<u>بر</u>

N.

#### TABLE 3.III

;

Values of  $\lambda$  and  $\lambda x_{1}/\bar{c}$  from Lifting-Surface Theory and Strip Theory

βa	c <sub>r</sub> ∕ē	A tan A <sub>0.5</sub>		λ	λ <b>x</b> 1/c		
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~			L-S Th.	ē̄∕c	L-S Th.	Egn. (3.38)	
0.624	1	0	1.48	1	0.54	0.12,	
1	1	0	1.33	1	0.42	0.12	
2	1	0	1.16	1	0.28	0.12	
4	1	0	1.06	1	0.20	0.12	
2	1	2	1.04	1	0.33	0.29	
4	1	4	0.84	1	0.57	0.79	
0.66	1.44	0.88	1.37	1.06	Ð. <b>4</b> 9	J. 17	
1.32	1.44	0.88	1.26	1.06	0.39	0.17	
1.98	1.44	C.88	1.20	1.06	0.33	0.17	
2.64	1.44	0.88	1.16	1.06	6.29	0.17	
0.66	1.44	2.2	1.12	1.05	0.44	0.33	
1.32	1.44	2.2	1.11	1.06	0.42	0.33	
1.98	1.44	2.2	1.10	1.06	9.39	0.33	
2.64	1.44	2.2	1.08	1.06	0.37	0.33	
0.667	1.44	3.567	0,95	1.00	0.57	9.64	
1.334	1.44	3.567	0.95	1.06	0.55	0.64	
2.001	1.44	3.567	0.96	1.05	0.54	0.64	
2.668	1.44	3.567	0.96	1.05	0.53	0.54	
1.352	1.58	0	1.30	1.11	0.40	0.15	
2. 598	1.58	0	1.20	1.11	0.30	0.15	
4.329	1.58	0	1.12	1.11	0.24	0.15	
1.2	1.75	1.5	1.22	1. 19	0.44	0.25	
1.8	1.75	1.5	1.20	1. 19	0.40	0.25	
3	1.75	1.5	1.17	1.19	0.35	0.25	
0.654	2	2	1.31	1.33	0.59	0.33	
1	2	2	1.28	1.33	0.56	0.33	
1.5	2	2	1.25	1.33	0.53	0.33	
4	2	2	1.22	1.33	C. 42	0.33	
1	Gothi	c wing	1.31	1.12			
1	1 Ogee wing		1.26	1.2 <del>4</del>			

Ξ.

#### TABLE 3. IV

Summary of Numerical Data on Interference Upwash for Large Wings

Tunne l	Ref.	Table(s)	Fig(s)	b/b	(δ <sub>0</sub> ) υ	(δ <sub>0</sub> ) <sub>5</sub>	$\delta_0(\mathbf{y}, \mathbf{t})$	<b>w</b> <sub>1</sub> (x, y, t)
Rectangular	3.1	6	10	1, 2	V	1		
closed	3.17	1 to 8	-	1.2			1	✓
	3.17	9 to 16	-	9/7.18/7			1	1
	3.18	4 to 7	-	2			1	
	3.20	-	5.7	2/5.10/7.2				
	3.22	2	3.4	20/7			✓	
	3.23	-	3.4	0.91 to 2.63			÷	
Ì	3.27	-	3	10/7				}
	3.30	2*.3.9	1.2.3	2				l ▼=0
	3.32	-	15.21	12				<b>x=0</b>
	3.33	5.6	4.5.72.13	24	1			
	3.44	-	4	7/5			J	
O-to-conol	1			0/7				ļ
alored	3.11	1	10	9/1 0/7 12/0				}
CTOBER	0.14	1,4	9, i	9/1,13/9				
	3.13	1,2	-	regular 0/788		I Y		[
L	3.45	4,0	-	3/ 1				
Elliptical	3.1	1	5	1				1
closed	3.1	10	15	½to2		1		]
1	3.14	Appendix	-	½ to 1	.∕.	[		
1	3.19	4	3 20 7	1		}	Ì	
	3.26	1	2	1	l	]		
1	3.35	1 1	-	√2.2		]		1
Į –	3.35	-	4	1 to 2				}
	3.36	2	2	= to 3				<b>j</b>
	3.38	- 1	32 to 34	*2	Į	1		
[	3.39	15		1	í .			y=0
	3.40	-	2	1.37			$\checkmark$	
Elliptical	3.1	11	16	½ to 2		1		
open	3.14	Appendix	- 1	% to 1	√	1		
1	3.35	2	-	√2,2				
1	3.35	- 1	5	1 to 2	√	Į		Ì
	3.36	3	3	= to 3	Į	1		ł
	3.38	- 1	22 to 24	*√2	1	1		↓ ↓
	3.39	11	-	1				y=0
Bipolar	3.14	1	4	1 to 5/3	17			1
closed	3.26	- 1	5	1.50	]			1
	3. 47	1,2	2,3	1.50,1.73		ł	V V	
Bipolar	3.14	2	6	1 to 5/3	7			
open	3.46	2,3	3,4	1 to 2	✓	<b>√</b>		
Semi-closed	 	1	1					1
circular	3.14	3	10	1	1	ł		1
arch-shaped	3.14	7	16	<1				1
elliptical	3.37	1 -	8,7	12		1		1
	1	1	1	}	3			1

• corrected in Table 7 of Reference 3.33 •• half model

r.

and a second second

and the second second state and the second second

### TABLE 3. V

;

Interference Parameters for a Closed Square Tunnel

27/b 2t/b	0	. 0.1	0.2	0.3	0.4	9.5	0.6	0.7	0.8
	0 4000		0.0000		0.0407	0.000			
U	U. 1368	0.1371	9.1353	0.1404	0.1437	U. 1488	0.1561	U. 1664	0.1807
0.1	ə. 1371	0. 1375	0.1387	0.1410	0. 1446	0. 1499	0.1576	0. 1684	0. 1835
0.2	0. 1383	9. 1387	0.1402	0. 1429	0.1472	0.153 <del>4</del>	0. 1622	0. 1746	0. 1919
0.3	0. 1404	0.1410	0.1429	0.1464	0. 1518	0. 1595	0.1704	0. 1857	0.2071
0.4	0. 1437	0.1446	0.1472	0. 1518	0. 1588	0. 1688	0. 1830	0.2029	0. 2313
0.5	0.1438	0. 1499	G. 1534	0. 1595	0. 1688	0. 1822	0.2013	0.2286	0. 2687
0.6	0. 1561	0. 1576	0. 1622	0.1704	0. 1830	0.2013	0. 2278	0.2671	0.3276
0.7	0. 1664	0.1684	0.1746	0. 1857	0.2029	0.2286	C. 2671	0, 3269	0. 4259
0.8	0. 1807	0. 1835	0. 1919	0.2071	0. 2313	0.2687	0.3276	0.4259	0.8065

(a) Values of 
$$\frac{\partial}{\partial t} \{ t \delta_0(\mathbf{y}, t) \}$$

(b) Values of 
$$\frac{\partial}{\partial t} \{ t \delta_1(\mathbf{y}, t) \}$$

2 <b>y/5</b> 2t/b	0	<b>G.</b> 1	0.2	0.3	0.4	0.5	0.6	0.7	0.8
0	0.2401	J. 2407	0.2428	0.2462	0.2525	0.2627	0.2785	J. 3024	0.3379
0.1	0.2407	0.2413	0. 2435	0.2475	0. 2545	0.2655	0.2826	0.3082	0.3464
0.2	9.2 <u>4</u> 26	0.2435	0.2463	0.2517	0.2605	0.2743	0. 2952	0.3266	6.3737
0.3	6.2552	9.2476	0.2517	0.2593	0.2715	0.2902	0.3183	0.3607	0.4253
0.4	0.2225	0.2545	0.2605	0.2715	0.2890	0.3156	J. 3557	0.4171	0. 5140
Ð.5	0.2022	1.2655	0.2743	0.2902	0.3156	0.3544	0. 4143	0.5090	0.6659
0.5	0.2725	6.212	0. 2952	0.3183	0.3557	0.4143	0.5077	0.6631	0.9394
0.7	0.3324	6.30RZ	0.3266	0.3507	0.4171	0.5090	0.6631	0.9381	1.4778
<b>D.</b> 8	0.3379	0.346 <u>4</u>	6.3737	0.4253	0. 5140	0.6659	0.9394	1.4779	2.6943
1					ŧ				í

-----

### TABLE 3, VI

Interference Parameter's  $\zeta_0$  and  $\zeta_1$  for Closed Rectangular Tunnels

<u>b</u> <u>ה</u>	ζ	ζ,	$\frac{b^2\zeta_0}{b^2}$	<u> bζ</u> <u>hζ</u>	
Û.8	14.746	108.34	9.44	5.88	
1.0	9.028	56.24	9.03	6.23	
1.3	5.688	33.42	9.61	7.54	
1.6	4. 426	26.73	11.33	· 9.66	
2.0	3.847	24. 16	15.39	12.56	

(a) y → 0, t → 0. Equations (3.157)

(b)  $y = t = \frac{1}{20}$ . Equations (3.160)

b h	ζ	ζı	b <sup>27</sup> 50 b <sup>2</sup>	<u>bζ</u> <sub>1</sub> <u>hζ</u> <sub>0</sub>	
0, B	19.389	180. 98	12.41	7.47	
1.0	10. 185	74.58	10. 19	7.32	
1.3	5.959	30.55	8. 55	7.85	
1.6	2.934	16.49	7.51	8.99	
2.0	1.637	Q. 97	6. 55	11.08	

### TABLE S.VII

12445

•

ş

# Comparative Calculations of $\Delta\alpha/c_L$ and $\Delta c_g/c_L$ for Wings in Closed Bectangular Tunnels

C	b h	σ	Wing				9C <sup>r</sup>	ĉ	s
Lase			Â	c₁∕ĩ	A tan A <sub>0.5</sub>		δα	Én	c
1	0-911	0.500	2.00	1.00	0	0.8	2.83	0.379	0.1138
2	0.911	0.714	4.00	1.00	0	0.8	4.62	0.271	0.1162
3	0.911	C.684	4.33	1.58	G	0,6	4.33	0. 180	0.0983
4	0.911	0.684	4.33	1.58	0	0.8	4.95	0.240	0.0983
5	1	0.450	1.59	2.00	2.09	0	1.83	C. 300	9.1350
6	1	(. 450	1. 50	2.00	2.00	0.8	2.02	0. 500	9. 1350
7	1.236	0.611	2.64	1.44	2. 20	e	2.80	0.293	0. 1819
8	2	0.393	2.64	1.44	2.20	0	2.80	0.298	0.1170
9	2	0.589	2.64	1.44	2.20	0	2. SO	0. 446	0.2631
19	18/7	0.473	2.54	1.44	2. 20	0	2.89	0.460	0. 2175
11	16/7	9.361	3.00	1.75	1.50	0.8	3.70	0.479	0.1038

ł		Δ	a/C <sub>L</sub>		۵¢, /¢,			
Case	$\sigma \rightarrow 0$ Eqn. (3.35)	Ref. 3. 17 Fig. 3. 16	Ref. 3. 26 Eqn. (3. 55)	L-S Th. Egn. (3.67)	$\sigma \rightarrow 0$ Eqn. (3.36)	Eqn. (3. 169)	Ref. 3. 26 Egn. (3. 57)	L-S Th. Egn. (3.67)
1	0.0238	0.0240	0.0253	0.0253	0.0058	0.0064	0.0978	0.0063
2	0.0216	0.0249	0.0259	0.0257	0.0048	0.0062	0.0058	0.9072
3	0.0171	0.0194	0.0193	0.0194	0.0026	0.0038	0.0035	0.0037
4	0.0121	0.0202	0.0203	0.0205	0.0045	ú. 00 <b>6</b> 3	9.0061	0.0055
5	0.0246	0 0244	0.0241	0.0255	C. 0047	0.0051	0.0056	0.0054
6	0.0269	-	0.0274	C. 0298	0.0093	0.0100	C.0111	0.0096
7	0.0385	0.0283	-	0.0283	0.0063	-	-	0.0065
8	9.9215	0.0176	0.0176	0.0179	0.0052	0.0027	0.0034	0.0036
9	ə. 0546	0.0388	0.0388	0.0397	0.0:77	6,0072	0.0098	0.0103
19	Ù. 0589	0.0372	0.0366	0.0376	0,0183	0.0052	0.0090	0.0076
11	0.0286	0.0212	0.0224	0.0238	0.0136	0.0073	0.0120	0.0100

189

語語な話書

···· ·· · · · · · · · ·

17-537-5



5-104. X 4-



	(b) Clos	ed rectange	ılar tunnel		
 		/ 12		L	
30	a)Ce		x	х <b>у</b>	
 0¢†	<b>1</b> 23	£2,5	ŝ	Ŕ	
 ين ن	ese h	No.	<b>v</b> čo		y
123	ля. П		<b>1</b> 23	. <i>C</i> C	
 UU U	فکه	NX4	فلآه	aX)	T
 ······································					



Fig. 3.1 Typical image systems

**\***\*\*

190

á

an Real Products

0.3 Type (I) (closed) Type (3) Type(2):::: (open) Type(4)[]] 0.5 0.1 80 0 -0.1 -0.2 -0.3L 0-50 0.75 1-25 b/h 1.00 1.50 1-75 2-00 0.6  $\Delta \alpha = \left(\delta_0 + \frac{\overline{\epsilon}}{2\beta h} \delta_1\right) \frac{SC_L}{bh}$ 0.4 0.2 8, 0 -0.5 -0-4L\_\_\_\_ 0-50 0.75 1.25 b/h i-00 i-50 1.75 2-00 Pig. 3. 2

ł

291

協

and the second subsection of the second statistic of the second statistic of the second statistic statistic statistics of the second statistics of the

i.

and the second second

Lift interference on small wings in rectangular tunnels



۴.,



ý

÷



15 d -

-----

a suspect of

\$



Pig. 3.4 Ratios  $\delta_1/\delta_0$  for small models in closed and open rectangular tunnels

-----

<del>در</del> ، در ا

2

PACENS" "1)



The second second

Ē.





N.

1 of the concernance

A CONTRACTOR OF THE CONTRACTOR

" JURA AWAY

1

$$\Delta \alpha = \left( \delta_0 + \frac{\lambda \overline{\epsilon} \delta_1}{2\beta h} \right) \frac{SC_L}{C}$$

Values of  $\lambda$  from lifting-surface theory for illustrated planforms are compared with those from two-dimensional strip theory.

A Lifting-surface theory

Ē/ē trom equation (3.38)



Fig.3.7 Lift-interference correction to measured incidence of small models of various planforms

-



and the second of the state of the second of the

t

197

A CONTRACTOR OF A CONTRACT OF

and the second state is a state of

-

a statistical

1

-,



Fig. 3.9 Lift interference on wings with elliptic spanwise loading in closed rectangular tunnels

12.00

ti T





# (b) Closed sides, open floor and roof. Correct





Fig. 3.10 Approximate representations of doubly infinite arrays of doublets to evaluate  $W_A$  for rectangular tunnels

199

- 7







A CALL BUT ALL STORE

•

1 4. 4. 14 M. M. M. A. 4.

5

1 ---- 1

MERCENSING ALL CONTRACT OF



Fig. 3. 12 Lift interference on wings with elliptic spanwise loading in closed octagonal tunnels

·202







ALL SALES

Fig. 3. 14 Vortex configurations defining  $\Phi_{p}$ ,  $\Phi_{1}^{(1)}$  and  $\Phi_{1}^{(2)}$  for circular tunnels

## 203

AL.

Levis and the state of the second



Fig. 3.15 Lift interference on uniformly loaded half-wings in closed and open circular tunnels with variable reflection planes



ł



i.

205

water aski . is down 21. Bearing

į

 فتغيينهم



ι

Fig. 3. 17 Interference corrections for wings of high aspect ratio in a closed rectangular tunnel (b/h = 10/7)

206

.

0.20  $\frac{\Delta \alpha}{C_{L}} = \left(\frac{S}{bh}\right)\delta$ ù-19 BA=2.5 0-18 0.17 δ S Bbh 0-16 0.3 0.15 0.2 0.14 0.1 0.05 0.13 Û 0.12 0-11 1.2 1-4 b/h 1-0 1-5 1·8 2.0

ł

and the second state of th



207

,ª

Ś

-----



Fig. 3.19 Comparative calculations of residual corrections to pitching moment on wings in closed rectangular tunnels for a range of Mach number



STATES STATES



. 1 .

and the second se



Interference correction to tailplane incidence on models of varying size Fig. 3. 21 in a closed dupler tunnel (M = 0)



÷

ł

ながに、ためまたとい

¢

1, r. S. ...

Fig. 3.22 Effect of wing spac on lift interference with ground simulation in three closed rectangular tunnels

CONTRACTOR OF

ころうちょうちょう ちょうちょう ちろうちょうちょう しょうちょう うちょうちょう

ł

TIS.



Fig. 3.23 Lift interference on ground effect on uniformly loaded wings in three closed rectangular tunnels

1000

Hand and a state of the state o
(a) Unswept wing in circular tunnel Section Plan 1.4 õ 01 C - 2R-1.2 С 1.0 с<sub>ц</sub> 8-0 0.6 Ref. 3.42 (NACA 4412) c=0-71R c=1-25R 0-4 Uncorrected -0--x---۵ Corrected + 0.2 0 (b) Swept wing in rectangular tunnel Section Plan 0 · 8 ה≠0.7b 0 . 6 CLL 0.4 Ret.3.57 (NACA 65-012 normal to I.e.) Uncorrected ( $A=45^{\circ}$ ) 0 0.2 Corrected (A =45") cos 45\* (C<sub>i</sub>) -2 C 2 6 a. (deg) i Q 12 14 8

Fig. 3. 24 Experiments on lift interference on models spanning closed tunnels

213

12-27

÷

CARLES CONTRACTOR OF

1-6



CALCORE STATUS

ï



Fig. 3.26 Comparison of aileron effectiveness of a complete model and a part-span model in a closed circular tunnel with reflection plane

DECEMBER OF THE STATE OF THE PARTY OF

100

and the state of the second second second

y v v - selvery v - starow y star - v - transk y vitrosovstv-selv- te Miter (<u>kuldens</u> best best



(a) Open circular tunnel

••



216

1111

il.

**North Assessment** 



ŝ



and a start of the second and

語言

## CHAPTER IV

• ~

Alighter Britsbergeranne

CANARA CAN

÷

1.427 2 ...

# INTERFERENCE EFFECTS IN UNSTEADY EXPERIMENTS

by

W.E.A. Acum

Ship Division, National Physical Laboratory, Feltham, Middlesex, England

• •

CONTENTS

;

. . .

		Page				
LIST	OF FIGURES	222				
NOTA	FION	223				
uef I M	NITIONS	227				
4. í	INTRODUCTION	229				
4.2	TWO-DIMENSIONAL WINGS IN INCOMPRESSIBLE FLOW	232				
4.3	TWO-DIMENSIONAL WINGS IN COMPRESSIBLE FLOW	236				
4.4	GENERAL THEORY FOR THREE-DIMENSIONAL WINGS					
	4.4.1 Tunnel Sesonance	248				
	4.4.2 Belations Between Steady and Unsteady Interference	241				
	4.4.3 General Interference Corrections	244				
4.5	APPLICATION TO THREE-DIMENSIONAL WINGS					
	IN CLOSED TUNNELS	247				
	4.5.1 Rectangular Tunnels	247				
	4.5.2 Circular Tanaels	259				
	4.5.3 Small Bings	259				
	4.5.4 Slender Wings	253				
	4.5.5 Numerical Examples	254				
4.6	PERFORATED AND SLOTTED WALLS	256				
4.7	STEADILY ROTATING NODELS	258				
	4.7.1 General Theory for Circular Tunnels	259				
	4.7.2 Slowly Holling Wings	262				
	4.7.3 Propellers	263				
4.8	APPLICATIONS TO CONNON EXPERIMENTAL SITUATIONS					
	4.8.1 Incompressible Flow	264				
	4.8.2 Compressible Flow	2\$5				
REFE	RENCES	256				
ADDIT	TIONAL REFERENCES	269				
FIGU	RES	278				

221

<u>.</u>...

The second s

ALL ALL

í

:

1

and the following the second for the second for the second se

THE PARTY

ملاتت كالملالا بالمزولية

2

## LIST OF FIGURES

In The State

Page

Fig. 4. 1	Boundary conditions for oscillating wing in two-dimensional closed tunnel	270
Fig. 4. 2	Tunnel corrections for wing pitching about its mid-chord point	271
Fig. 4.3	The effect of resonance on lift in a two-dimensional tunnel at $M = 0.7$	272
Fig. 4. 4	Interference parameters for small wings in closed rectangular wind tunnels	273
<b>Fig. 4.</b> 5	Planform of model wing used as an example in Section 4.5.5	274
Pig. 4.6	Effect of frequency on the interference upwash due to a horse-shoe vortex in a closed $9 \times 7$ tunnel	275
Fig. 4.7	Effect of number of slots on the pitching damping of an M-wing ( $\overline{\nu} \simeq 0.05$ )	277
Fig.4.8	Boundary conditions far downstream of a rolling wing	278

222

~ 🛓

. تەرىچ **ئ**ارىي

τ**-**•

the state of the

the second se

N.

### NOTATION

width of slot constants defining the lift distribution in Equation (4.34) 8, constants defined by Equations (4.26) Å, breadth of tunnel Ъ C chord of aerofoil ē mean chord of wing (S/2s) root chord c<sub>r</sub> tip chord c<sub>t</sub> C Theodorsen's lift function, defined in Equation (4.23) rolling moment coefficient = (rolling moment)/( $\rho U^2 S_3$ ) C, =  $\overline{C}_1 e^{i\omega t}$ , lift coefficient =  $(lift)/(\frac{1}{2}\rho U^2 S)$ C<sub>L</sub>  $\overline{C}_{p}e^{i\omega t}$ , pitching moment coefficient = (pitching moment)/ $(\frac{1}{2}\rho U^2 S\overline{c})$ C\_ = C, constants in Equation (4.19) representative length of tunnel cross-section đ ĩ free vorticity defined in Equations (4.13) Ē function defined in Equations (4.35) f,  $f_1$  functions defined in Equations (4.75) and (4.76) P function defined in Equation (4.27)  $F_1$ ,  $F_2$  functions defined in Equations (4.74) and (4.79) G function defined in Equation (4.80) height of a twanel of rectangular cross-section h prefix denoting imaginary part of Ī2 kernel of integral Equation (4.39), defined in Equations (4.40) and (4.41) X geometric slot parameter in Equation (4.98) ĸ



224	
ĩ	kernel of integral equation, defined in Equation (4.72)
I =	$le^{i\omega t}$ , non-dimensional wing loading = $(p_l - p_u)/(\frac{1}{2}\rho U^2)$
$l_{z}$ , $l_{z}$	direct heaving derivatives (see Definitions)
$l_{\alpha}, l_{\dot{\alpha}}$	derivatives of lift due to pitching (see Definitions)
L	lift force on wing
L(x)	lift per unit length in the streamwise direction
$\mathbf{L}_{\mathbf{R}}$ , $\mathbf{L}_{\mathbf{I}}$	spanwise lift distributions defined in Equation (4.62)
= <sub>z</sub> , =;	derivatives of pitching moment due to heaving (see Definitions)
<b>Ξ</b> α, Ξα	direct pitching derivatives (see Definitions)
M	Mach number of undisturbed stream
n	outward normal distance from tunnel boundary
N	number of slots in roof (or floor) of tunnel
P	pressure
p(x,y)	shape of a general mode of oscillation in Equation (4.65)
p <sub>æ</sub>	pressure of undisturbed stream
P <sub>c</sub>	pressure in plenum chamber
D <sub>a</sub>	weighting function for generalized force P <sub>n</sub>
P	wall perosity parameter defined in Equation (4.7)
Pa	generalized force $(n = 1, 2,)$ in Equation (4.65)
q =	$q_0 e^{i\omega t}$ , amplitude of a general mode * Equation (4.65)
r	radial cylindrical polar co-ord · in Equations (4.101)
Ŧ	non-dimensional radial distance = $\omega r/U$
R	radius of circular tunnel
R	non-dimensional radius = $\omega R/U$
R <sub>B</sub>	see Equation (4.41)
Re	prefix denoting real part of

semi-span of wing s s(x) local span of wing ŝ non-dimensional semi-span =  $\omega s/U$ S area of wing planform time t Т tunnel slot parameter in Equation (4.99) U velocity of undisturbed stream  $\overline{w}e^{i\omega t}$ , component of velocity in z direction  $\widetilde{w}_{i} e^{i\omega t}$  , part of w induced by tunnel walls = Wş streamwise distance z value of x at pitching axis X<sub>0</sub> spanwise distance T upward distance z wing mode shape defined by  $z = \operatorname{Re}\{\overline{z}(x, y)e^{i\omega t}\}$ ž non-dimensional amplitude of heaving mode (see Definitions) z, incidence of wing (in radians) α suplitude of pitching oscillation defined by  $\alpha = \alpha_n e^{j\omega t}$ α offective angle of incidence of a rotating wing α<sub>E</sub>  $(1 - H^2)^{\frac{1}{2}}$ β ř bound vorticity in Equations (4.13) Г  $\overline{\Gamma}e^{i\omega t}$ , strength of vortex; circulation in Section 4.7 = Ĩ, functions defined in Equation (4.20)  $\overline{\Gamma}_{\mathbf{R}}$  ,  $\overline{\Gamma}_{\mathbf{I}}$ vortex strengths defined in Equation (4.64) δ prefix denoting contribution due to wall interference Δ prefix denoting increment due to wall correction δ non-dimensional interference upwash defined in Equation (4.55) 225

and the second of the

226  $\delta_0$ ,  $\delta_1$  interference upwash parameters for a small wing defined in Equations (4.86)  $\delta_0', \delta_1'$  or (4.90) = y/d η angular chordwise parameter in Equation (4.22) θ θ sigular cylindrical polar co-ordinate in Equations (4.101) (Section 4.7.1) λ eigenvalue in Equation (4.37) frequency parameter =  $\omega d/U$ μ frequency parameter =  $\omega c/U$ ν v frequency parameter =  $\omega \bar{c}/U$ ξ x/d = local chordwise centres of pressure corresponding to  $\ L_{R}\,,\,L_{T}$ 51,51 density of undisturbed stream p s/đ σ = Weist. perturbation velocity potential in Equation (4.2) Φ Ξ **₽**, velocity potential of a steady horse-shoe vortex in Equation (4.43) **ቅ**\_ components of  $\overline{\Phi}$  for a rolling wing in Equation (4.105) **Q** see Equation (4.107) angular frequency of oscillation ω angular velocity of steadily rolling wing ω critical sngular frequency for tunnel resonance in Equations (4.28) and w<sub>2</sub> (4.38) Subscripts

- r denotes flow in free air.
- i denotes quantity induced by tunnel walls.
- l, u denotes lower, upper surface of wing.
- L, T denotes leading, trailing edge of wing.

DEFINITIONS

If the wing is performing oscillations such that its surface has the equation

$$z = -[z_0 c + \alpha_0 (x - x_0)]e^{i\omega t},$$

the derivative coefficients are defined by

$$\text{Lift} = \rho U^2 S \, e^{i\omega t} [(l_z + i\overline{\nu} l_z) z_0 + (l_\alpha + i\overline{\nu} l_\alpha) \alpha_0]$$
  
Pitching mement =  $\rho U^2 S \overline{c} e^{i\omega t} [(m_z + i\overline{\nu} m_z) z_0 \div (m_\alpha + i\overline{\nu} m_\alpha) \alpha_0]$ 

The pitching moment about  $x = x_0$  is reckoned as positive if it tends to raise the leading edge and depress the trailing edge.

For two-dimensional wings  $\overline{c}$  is the (constant) chord, and then  $\overline{\nu} = \nu = \omega c/U$ .

The usual convention about the interpretation of complex quantities in sinusoidal oscillations is assumed; each of the three equations above should be interpreted as having the prefix Re in its right-hand side, and similarly throughout the whole chapter.



11 - - - AS

## INTERPERENCE EFFECTS IN UNSTEADY EXPERIMENTS

#### W.E.A. Acuz

#### 4.1 INTRODUCTION

As might be expected, the calculation of unsteady interference effects on wings oscillating in wind tunnels is considerably more complicated than for steady flow; as a result not all cases of practical importance are covered by theory. Moreover, since it is possible for an oscillating model and a wind tunnel to form a resonating system, a new phenomenon not occurring in steady flow must be considered. Interference effects can, however, be calculated for many cases of practical interest, although here again a difference arises in that the calculation may be quite lengthy, and simple formulae cannot usually be given.

For the most part this chapter will be concerned with wings performing oscillations sinusoidal in time, but steadily rotating models will be dealt with in Section 4.7.

In most theoretical methods the equations, which have to be solved in order to determine the tunnel interference, are basically those of the small perturbation theory for oscillating thin wings in a free stream, but with modifications to the usual boundary conditions to account for the tunnel walls. The basis of linearized wing theory is set out in Reference 4.1. Thus for thin wings the perturbation velocity potential.  $\Phi(x,y,z,t)$ , is assumed to satisfy the linearized differential equation

$$\frac{\partial^2 \Phi}{\partial \mathbf{x}^2} + \frac{\partial^2 \Phi}{\partial \mathbf{y}^2} + \frac{\partial^2 \Phi}{\partial \mathbf{z}^2} = \frac{\mathbf{y}^2}{\mathbf{u}^2} \left( \mathbf{u}^2 \frac{\partial^2 \Phi}{\partial \mathbf{x}^2} + 2\mathbf{u} \frac{\partial^2 \Phi}{\partial \mathbf{x} \partial \mathbf{t}} + \frac{\partial^2 \Phi}{\partial \mathbf{t}^2} \right) . \tag{4.1}$$

Since the flow is oscillating with angular frequency  $\omega$ , we take

$$\Phi = \operatorname{Re}\{\overline{\Phi}(\mathbf{x},\mathbf{y},\mathbf{z})e^{i\omega t}\}, \qquad (4.2)$$

and obtain

$$(1 - \mathbf{M}^2) \frac{\partial^2 \overline{\Phi}}{\partial \mathbf{x}^2} + \frac{\partial^2 \overline{\Phi}}{\partial \mathbf{y}^2} + \frac{\partial^2 \overline{\Phi}}{\partial \mathbf{z}^2} - \frac{2i\omega \mathbf{M}^2}{\mathbf{u}} \frac{\partial \overline{\Phi}}{\partial \mathbf{x}} + \frac{\omega^2 \mathbf{M}^2}{\mathbf{u}^2} \overline{\Phi} = 0 . \qquad (4.3)$$

The linearized expression for the pressure corresponding to  $\Phi$  is

$$p - p_{ro} = -\rho \left( U \frac{\partial \Phi}{\partial x} + \frac{\partial \Phi}{\partial t} \right)$$
$$= -\rho \operatorname{Re} \left\{ \left( U \frac{\partial \overline{\Phi}}{\partial x} + i\omega \overline{\Phi} \right) e^{i\omega t} \right\}. \quad (4.4)$$

married to the state

As is customary in the theory of sinusoidal oscillations, the symbol Re for "the real part of" will be omitted unless there is any risk of ambiguity.

On the wing the vertical velocity of the fluid is determined by the wing motion. In the linearized theory this condition becomes

$$\frac{\partial \overline{\Phi}}{\partial z} = \mathbf{U} \frac{\partial \overline{z}}{\partial x} + \mathbf{i} \omega \overline{z}$$
(4.5)

on that part of the plane occupied by the planform, where the wing surface is assumed to be defined by  $z = \overline{z}(x,y)e^{i\omega t}$ , and the plane z = 0 is chosen so that  $\overline{z}$  is always small. At infinity  $\overline{\Phi}$  must represent outgoing waves; in the presence of tunnel walls this condition can be applied only in the upstream and downstream directions. The wake is assumed to lie in the strip, downstream of the trailing edge, defined by  $x > x_{\widehat{1}}$ , |y| < z, z = 0; the pressure must be continuous across the wake, so that, provided  $\overline{\Phi}$  is an odd function of z, its distribution over either side of the wake satisfies\*

$$\vec{\upsilon} \frac{\partial \Phi}{\partial \mathbf{x}} + i\omega \vec{\Phi} = 0 . \qquad (4.6)$$

At the trailing edge it is assumed that both  $\Phi$  and  $\partial \Phi / \partial x$  are continuous, so as to satisfy the Kutta-Joukowski condition.

On solid tunnel walls the normal velocity.  $\partial \Phi / \partial n$ , must be zero. In an open jet the pressure must be continuous at the edge of the jet and the boundary condition follows from Equation (4.4). These two cases are the limits of the boundary condition which is scriptimes applied to represent a porous wall, that is, that the velocity normal to the wall is proportional to the pressure difference across it. This may be written

$$\mathbf{p} - \mathbf{p}_{c} = \frac{\rho \mathbf{U}}{P} \frac{\partial \Phi}{\partial \mathbf{n}} \qquad (4.7)$$

where P is the porosity parameter, p is the pressure on the inside of the wall, and  $p_c$  the pressure on the outside. This  $p_c$  is normally the pressure in a plenum chamber. Solid walls correspond to P = 0 and free surfaces to  $P \to \infty^{\dagger}$ . The application of Equation (4.7) to steady flow is discussed in Section 6.7.

\* By symmetry  $\overline{\Phi}$  is an odd function of z for free-stream flow, and for most practical tunnel arrangements. If  $\overline{\Phi}$  is not an odd function of z, then the condition for the continuity of pressure across the wake becomes

$$: \quad \tilde{v} = (j\bar{\nabla} - u\bar{\Phi}) \left( \frac{1}{\sqrt{2}} + \frac{1}{\sqrt{2}} v \right)$$

it does not follow that either  $\bar{\Phi}_{n}$  or  $\bar{\Phi}_{j}$  separately satisfies Equation (4.6).

t There free surfaces are concerned these remarks apply only to wind tunnels. If a water tunnel has a free surface, the effect of gravity may have to be taken into consideration.

いいのないで、アインシス

بالأسط المعالي

In the theory it is usually assumed that  $p_c = p_{\infty}$  and that Equation (4.7 may be linearized by using Equation (4.4). It is conceivable that, to give a good representation of the conditions for an oscillating flow. P might have to be a complex number.

Equation (4.3) is valid for thin wings provided M is not too near to one. It is, however, not the only differential equation which can be used; for example, the approximation

$$\frac{\partial^2 \overline{\Phi}}{\partial y^2} + \frac{\partial^2 \overline{\Phi}}{\partial z^2} = 0$$
 (4.8)

may be used for very slender wings, while near M = 1 the unsteady tran onic equation

$$\frac{\partial^2 \tilde{\Phi}}{\partial \mathbf{y}^2} + \frac{\partial^2 \tilde{\Phi}}{\partial \mathbf{z}^2} - \frac{z_{1\omega} \mathbf{H}^2}{\mathbf{U}} \frac{\partial \tilde{\Phi}}{\partial \mathbf{z}} + \frac{\omega^2 \mathbf{H}^2}{\mathbf{U}^2} = 0 \qquad (4.9)$$

is more applicable. In fact little use has so far been made of them for problems of unsteady wall interference. Miles<sup>4,1</sup> gives a summary of the parameter ranges in which Equations (4.3), (4.8) and (4.9) are valid, and an account of their derivations. For two-dimensional flow Equation (4.3) still holds, but the derivative with respect to y vanishes.

In most theoretical treatments the tunnel is assumed to be cylindrical and to extend to infinity upstream and downstream, so that the determination of tunnel interference is equivalent to solving Equation (4.3) with boundary conditions such as those shown in Pigure 4.1, which represents a wing in a closed two-dimensional tunnel. Jordan (Ref. 4.2; 1953) has considered the effect of a two-dimensional vortex wake of finite length, but in general it seems essential to assume that the wake is infinite if a mathematical treatment is to be feasible.

For rectangular or two-dimensional tunnels the method of images, which is exactly analogous to that employed in steady flow, may be used. As in free-stream theory, it is then often possible to replace the differential equation by an equivalent integral equation. For the general cylindrical tunnel, with that of circular section as the most important example, it is not possible to construct an image system, and present methods of solution are restricted to the special cases of incompressible flow or low frequency.

Wind-tunnel experiments involving force measurements may be of two kinds. The object of the first is the determination of some force acting on a model, and the correction is applied to obtain a value appropriate to free-stream conditions. In such cases it is desirable that as much as possible of the measured data should be used in computing the interference corrections, in order to offset the in-accuracies of linearized theory. In particular, the calculation of a correction by taking the difference between

(i) free-stream linearized theory

and (ii) linearized theory modified to include the presence of the walls,

is not likely to be satisfactory unless the theory represents the experimental situation with an unusual degree of accuracy. The second kind of experiment has as

its objective an assessment of the accuracy of a theory. If this theory is the same in principle for both the free-stream and tunnel cases, then it is sufficient to compare (ii) with the uncorrected experiments.

Section 4.8 summarizes the methods recommended for typical configurations in incompressible and in compressible flow. There are difficult cases where no method of solution exists, and others where the numerical work might well be considered prohibitive.

#### 4.2 THO-DIMENSIONAL WINGS IN INCOMPRESSIBLE PLOW

Consider first an aerofoil of chord c on the centre line of a two-dimensional tunnel with solid walls distant h spart. Suppose that the motion of the aerofoil may be described by the equation

$$z = -[z_{0}c + \alpha_{0}(x - x_{0})]e^{i\omega t} , \qquad (4.10)$$

so that the motion is rigid pitching of amplitude  $\alpha_0$  about an axis  $x = x_0$ , superimposed on a rigid heaving motion of amplitude  $z_0c$ . If  $\alpha_0$  is not zero, Equation (4.10) also represents a rigid pitching motion about the axis  $x = x_0 - z_0c/\alpha_0$ .

Since the fluid is incompressible, Equation (4.3) reduces to

$$\nabla^2 \Phi = 0 , \qquad (4.11)$$

with boundary conditions (Fig. 4.1)

$\frac{\partial \bar{\Phi}}{\partial z} = - \mathbf{U} \alpha_0 - i \omega [\mathbf{z}_0 \mathbf{c} + \alpha_0 (\mathbf{x} - \mathbf{x}_0)]$	Ø	Z	= 0,	x <sub>L</sub> < x < x <sub>T</sub>	-	•
$\frac{\partial \mathbf{x}}{\partial \Phi} + \frac{\mathbf{r}}{\mathbf{r} \mathbf{x} \mathbf{y}} = 0$	on	Z <sup>:</sup>	= 0,	x > x <sub>?</sub>	} :	(4. 12)
$\frac{\partial z}{\partial \overline{\Delta}} = 0$	CER	Z	= ±12b	, -∞ <x<∝< td=""><td></td><td></td></x<∝<>		

furthermore at  $x = x_r$ ,  $\overline{\Phi}$  and  $\partial \overline{\Phi} / \partial x$  must be continuous with respect to x. For a general mode of oscillation the first of Equations (4.12) has to be replaced by Equation (4.5).

The alternative to the solution of the differential Equation (4.11) lies in the formulation of the corresponding integral equation. In this approach the vorticity distribution over the wing and wake is regarded as being the sum of two parts,  $\tilde{T}e^{\pm\omega t}$ , the "brund" vorticity, which is proportional to the local lift, and  $\tilde{E}e^{\pm\omega t}$ , the "tree" vorticity.  $\tilde{T}$  and  $\tilde{T}$  are defined by

anatomic strike and

21.14

$$\widetilde{\Gamma} + \widetilde{E} = \frac{\partial}{\partial x} (\overline{\Phi}_{u} - \overline{\Phi}_{l})$$

$$\widetilde{\Gamma} = \left(\frac{\partial}{\partial x} + \frac{i\omega}{v}\right) (\overline{\Phi}_{u} - \overline{\Phi}_{l})$$
(4.13)

so that, by Equation (4.4), the local loading per unit area is

$$l = \frac{p_{l} - p_{u}}{\frac{1}{2}\rho u^{2}} = \frac{2\overline{l}}{u} e^{i\omega t} , \qquad (4.14)$$

and, by Equations (4.13),

$$\widetilde{E}(\mathbf{x}) = -\frac{i\omega}{\mathbf{u}} \exp\left(-\frac{i\omega\mathbf{x}}{\mathbf{u}}\right) \int_{\mathbf{x}_{L}}^{\mathbf{x}} \widetilde{\Gamma}(\mathbf{x}') \exp\left(\frac{i\omega\mathbf{x}'}{\mathbf{u}}\right) d\mathbf{x}' \quad .$$
(4.15)

In the wake,  $x > x_T$ ,  $\tilde{\Gamma}$  is zero, so that for  $x > x_T$  the upper limit of integration in Equation (4.15) is  $x_T$ .

Now in the unbounded free stream the upwash corresponding to the vorticity distribution  $(\widetilde{\Gamma} + \widetilde{E})$  would be given by

$$\widehat{\mathbf{w}}(\mathbf{x}, \mathbf{z}) = -\frac{1}{2\pi} \int_{\mathbf{x}_{L}}^{\infty} \frac{(\widetilde{\Gamma} + \widetilde{E})(\mathbf{x} - \mathbf{x}')}{(\mathbf{x} - \mathbf{x}')^{2} + \mathbf{z}^{2}} \, d\mathbf{x}' \quad : \qquad (4.13)$$

this relation is a straightforward extension of that arising in the treatment of two-dimensional thin wings in steady incontressible flow, for esample that given in Chapter VIII of Reference 4.3.

In the wind-tunnel case it is necessary to modify this by including the contributions of the images which are assumed to be present in a way analogous to that for the corresponding stoudy flow (Chapter IV). This for a closed tunnel the upwash on z = 0 is that due to an infinite column of vortex sheets, of alternating sign, lying on  $z = \pm nh$ , n = 1, 2, 3, ... Equation (4.16) is then replaced by

$$\bar{\mathbf{x}}(\mathbf{x},0) = -\frac{1}{2\pi} \int_{\mathbf{x}_{L}}^{\infty} [\tilde{\Gamma}(\mathbf{x}') + \tilde{\mathbf{g}}(\mathbf{x}')] \left\{ \frac{1}{\mathbf{x} - \mathbf{x}'} + 2 \sum_{n=1}^{\infty} (-1)^{n} \frac{\mathbf{x} - \mathbf{x}'}{(\mathbf{x} - \mathbf{x}')^{2} + n^{2}h^{2}} \right\} d\mathbf{x}'$$

or, with the known sus of the infinite series,

$$\overline{\pi}(x,0) = -\frac{1}{2\pi} \int_{x_{L}}^{x_{L}} [\widetilde{\Gamma}(x') + \widetilde{E}(x')] \frac{\pi}{h} \operatorname{cosech} \frac{\pi(x-x')}{h} dx' . \quad (4.17)$$

S. 15 22. 6

### 234

The singularity at x' = x is accounted for by taking the Cauchy principal value. Since  $\overline{w}(x,0)$  is known from Equation (4.12) for  $x_L < x < x_T$ , Equation (4.17) is the required integral equation for  $(\widetilde{\Gamma} + \widetilde{E})$ .

Explicit analytical solutions of Equation (4.17) are available covering most cases of practical interest. However, it may be remarked that for experiments of the first type discussed at the end of Section 4.1 such solutions are not essential.

The equation for the interference upwash, due to the images only, is

$$\overline{\mathbf{x}}_{i}(\mathbf{x},0) = -\frac{1}{2\pi} \int_{\mathbf{x}_{L}}^{\infty} [\widetilde{\Gamma}(\mathbf{x}') + \widetilde{E}(\mathbf{x}')] \left\{ \frac{\pi}{h} \operatorname{cosech} \frac{\pi(\mathbf{x}-\mathbf{x}')}{h} - \frac{1}{\mathbf{x}-\mathbf{x}'} \right\} d\mathbf{x}' \quad (4.18)$$

Now if  $\tilde{l}$  were known,  $(\tilde{l} + \tilde{E})$  could be calculated from Equations (4.14) and (4.15) and  $\tilde{w}_i$  evaluated numerically using Equation (4.18). Unless c/h were large,  $\tilde{w}_i$  would be obtained sufficiently accurately if l were given its free-stream theoretical value, modified by a factor to make it consistent with the measured forces. The interference forces could then be calculated from free-stream theory.

Some progress towards a solution of Equation (4.17) may be made by the modification of the "vortex sheet" theory for two-dimensional oscillating wings in Reference 4.4 which is itself an extension of the simple theory for steady wings<sup>4.3</sup>. For the oscillating free-stream case, the bound vorticity is expanded in the form

$$\widetilde{\Gamma} = U \sum_{n=1}^{\infty} C_n \widetilde{\Gamma}_n , \qquad (4.19)$$

with

1

a statement at a sub-

「日本に、たちになっている」というないないないで、たちないのないないないで、たちないのできょうと

$$\widetilde{\Gamma}_{0} = 2\left[C(\frac{1}{2}\nu)\cot\frac{1}{2}\theta + \frac{1}{2}i\nu\sin\theta\right]$$

$$\widetilde{\Gamma}_{1} = -2\sin\theta + \cot\frac{1}{2}\theta + \frac{1}{2}i\nu\left[\sin\vartheta + \frac{1}{2}\sin2\theta\right]$$

$$\widetilde{\Gamma}_{2} = -2\sin(d^{2} + \frac{1}{1}d^{2})\left[\frac{\sin(\alpha+1)\vartheta}{\alpha+1} - \frac{\sin(\alpha-1)\theta}{\alpha-1}\right] \quad (n \ge 2)$$

$$(4.20)$$

Here

$$= \omega c/J$$
 , (4.21)

$$I = I_1 + \frac{1}{2}c(1 - cos\theta)$$
, (4.22)

and  $C(\frac{1}{2}\nu)$  is Theodorsen's lift function

ν

$$C(\frac{1}{2}\nu) = H_1^{(2)}(\frac{1}{2}\nu)/\{H_1^{(2)}(\frac{1}{2}\nu) + i H_0^{(2)}(\frac{1}{2}\nu)\}, \qquad (4.23)$$

where  $H_0^{(2)}$  and  $H_1^{(2)}$  are Hankel functions. The upwash distribution is then given by

in the second

$$\overline{\mathbf{w}} = -\overline{\mathbf{U}}\left[\mathbf{C}_{0} + \frac{1}{2}\mathbf{C}_{1} + \sum_{n=1}^{\infty} \mathbf{C}_{n} \cos n\theta\right] , \qquad (4.24)$$

and the unknown coefficients  $C_n$  may be determined by comparison with the first of Equations (4.12).

Now, if tunnel walls are present, Equation (4.19) will not lead to Equation (4.24) but to a modified form of it. There appears to be no simple method of working out what this is for general values of the ratio c/h, but Jones (Ref. 4.4; 1950) has given a method for small c/h . Equation (4.24) is replaced by

$$\vec{w} = -U \left[ (C_{e} + \frac{1}{2}C_{1} + A_{0}) + \sum_{n=1}^{\infty} (C_{n} + A_{n}) \cos n\theta \right],$$
 (4.25)

where

$$A_{0} = -\frac{\pi^{2}c^{2}}{24h^{2}} \left[ C_{0} \left( \frac{1}{2} - \frac{2iC}{\nu} \right) + \frac{C_{1} - C_{2}}{4} \right] + J_{0} (\frac{1}{2}\nu) FC_{0}$$

$$A_{n} = 2i^{n} J_{n} (\frac{1}{2}\nu) FC_{0} \quad (n \ge 1)$$

$$(4.26)$$

Here

$$F = \{CJ_{0}(\frac{1}{2}\nu) + i(1-C)J_{1}(\frac{1}{2}\nu)\}\left\{\frac{\pi^{2}c^{2}}{24h^{2}}\exp(-\frac{1}{2}i\nu)\left(1-\frac{2i}{\nu}\right) - \frac{1}{2}i\nu\int_{\frac{1}{2}\nu}^{\infty}\left[\frac{\exp(-iy)}{y}-\frac{\pi c}{\nu h}\frac{\exp(-iy)}{\sinh(\pi cy/\nu h)}\right]dy\right\}.$$
 (4.27)

 $C = C(\frac{1}{4}\nu)$  is given by Equation (4.23) and  $J_n$  is the Bessel function in the usual notation.

Lilley (Ref.4.5; 1952) has transformed Equation (4.16) into one of a type having known solutions, but unfortunately the result for general c/h ratio is too long and complicated to reproduce here, although some simplification occurs if c/h is of order one or larger. Certain modes of oscillation of cascades cause flows like those considered in tunnel interference; for example, the flow through an unstaggered cascade in which consecutive elements are oscillating in antiphase is the same as that past an aerofoil in a closed wind tunnel, as may be seen by considering the image system. Reference 4.6 could be applied in this way.

The alternative to an approach via vortex-sheet theory is the solution of the differential Equation (4.11) and associated boundary conditions. This has been cerried out by Rosenblat (Ref. 4.7; 1957) by a method involving conformal transformation of the interior of the tunnel, and also by Timman (Ref. 4.8; 1951) by a modification of Theodorsen's free-stress theory. The smalltical expressions are too lengthy for inclusion, but de Jager has presented Timasn's results in Reference 4.9.

algebraically and as graphs, for a rigid aerofoil with or without a trailing-edge control. It should be noted that, contrary to the statement made by Rosenblat in Reference 4.7, his results do in fact agree with Timman's<sup>\*</sup>.

It is likely that in most experiments c/n will be small enough for Equations (4.25) to (4.27) to apply. A comparison of the results from Reference 4.4 with the graphs of Reference 4.9 indicates that this may be expected to be good enough for c/h < 0.25. Figure 4.2 shows the variation with  $\nu$  of  $\Delta m_{\alpha}$  and  $\nu \Delta m_{\alpha}^{*}$  (where the prefix  $\Delta$  denotes free-stream value minus tunnel value) for a wing pitching about the mid-chord axis. The results from Equations (4.25) to (4.27) (Ref.4.4) for c/h = 0.21 fit in very well. Alternatively the tunnel corrections may be estimated from the graphs of Reference 4.9. A method of applying the theoretical quantities such as  $\Delta m_{\alpha}$  to practical situations is set out in Section 4.4.3.

A further simplification occurs when both the frequency parameter and c/h are small. If this is so, there are explicit expressions for the lift and pitching moment<sup>4,4</sup>; these will not be given here as they are merely special cases, with M = 0 and  $\beta = 1$ , of the corresponding expressions for compressible flow in Equations (4.35).

#### 4.3 TWO-DIMENSIONAL WINGS IN COMPRESSIBLE FLOW

Before considering the possibility of calculating the tunnel interference it is necessary to recognise that, if the fluid is compressible, the phenomenon of "tunnel resonance" is possible. It has been shown in References 4.10 and 4.11 that, according to linearized theory, the interference upwash due to a small oscillating model wing in a two-dimensional closed tunnel will become infinite for certain critical frequencies, the smallest of which is given by

$$\frac{\omega_{\rm rh}}{U} \simeq \frac{\pi\beta}{M} . \tag{4.23}$$

The physical significance of this particular frequency may be seen by considering the possibility of perturbation flows of the form

$$\Phi = e^{ikx} \sin(\pi z/h) , \qquad (4.29)$$

where k is a constant, as yet undetermined. This  $\Phi$  automatically satisfies the condition of no flow through the tunnel walls at  $z = \pm \frac{1}{2}h$ , but unless k is real  $\Phi$  will not remain finite at large distances upstream and downstream. Substitution of Equation (4.29) into Equation (4.3) gives

$$\left(\frac{\omega M}{U} + Mk\right)^2 = k^2 + \frac{\pi^2}{h^2} , \qquad (4.30)$$

\* The second equation is Section 14 (p.271) of Reference 4.7 should read

 $q = \frac{1}{32}k^2\left[1 + \frac{1}{2}k^2 + O(k^4)\right] .$ 

236

ALL LAND MARK

and the smallest value of  $\omega$  for which Equation (4.30), considered as an equation for k, has real roots, is in fact  $\omega_r$  as given by Equation (4.28). The lowest critical frequency for resonance is therefore the smallest for which a disturbance of the type Equation (4.29) can remain of finite amplitude at all large distances from the model.

According to Equation (4.28) the critical frequency is small if M is near to one, and measurements at high subsonic speeds are most likely to be affected. It might also appear that  $\omega_r$  should be small for a very large tunnel, but it must be borne in mind that Equation (4.28) is deduced on the assumptions that the tunnel is of infinito length and that viscosity is negligible, and it can therefore only roughly represent actual experiments. There is, nevertheless, strong experimental evidence, such as that in Figure 4.3, that the result is approximately correct.

The evaluation of the forces on a thin wing in a wind tunnel requires either the solution of Equation (4.1) with the appropriate boundary conditions, or else a solution of one of the equivalent integral equations. In free-stream flow the lift and upwash distributions are related by Possio's equation

$$\frac{\ddot{\mathbf{w}}(\mathbf{z})}{\mathbf{v}} = \frac{i}{8\beta} \lim_{\mathbf{z} \to 0} \int_{\mathbf{x}' c_{\mathbf{x}_{\mathrm{L}}}}^{\mathbf{x}_{\mathrm{I}}} \int_{-\infty}^{\mathbf{x}_{\mathrm{I}} \mathbf{z}'} \exp\left[-i\omega(\mathbf{x} - \mathbf{x}')\right] \exp\left(\frac{i\omega\xi}{\mathbf{v}\beta^2}\right) \times \frac{\partial^2}{\partial \mathbf{z}^2} H_0^{(2)} \left(\frac{\omega\mathfrak{H}}{\mathbf{v}\beta^2}\sqrt{\xi^2 + \beta^2 \mathbf{z}^2}\right) d\xi d\mathbf{x}' \quad (4.31)$$

where  $H_0^{(2)}(z)$  is a Hangel function of argument z; the velocity potential on the upper surface and the upwash are related by

$$\frac{\overline{\mathbf{w}}(\mathbf{x})}{\underline{\mathbf{v}}} = -\frac{\mathbf{i}}{2\beta \underline{\mathbf{v}}} \lim_{\overline{z} \to 0} \int_{\mathbf{x}_{\mathrm{L}}}^{\mathbf{w}} \Phi_{\mathbf{u}}(\mathbf{x}') \exp\left(-\frac{\mathbf{i}\omega \underline{\mathbf{w}}^{2}(\mathbf{x}-\mathbf{x}')}{\underline{\mathbf{v}}\beta^{2}}\right) \frac{\partial^{2}}{\partial z^{2}} H_{0}^{(2)} \left(\frac{\omega \underline{\mathbf{w}}}{\underline{\mathbf{v}}\beta^{2}} \sqrt{(\mathbf{x}-\mathbf{x}')^{2} + \beta^{2} \mathbf{z}^{2}}\right) d\mathbf{x}' .$$
(4.32)

These relations can be obtained from the corresponding three-dimensional integral equations by integrating spanwise with respect to y.

In the case of the closed wind tunnel the simulation of the walls by images introduces sums of the form

$$\sum_{\infty}^{\infty} (-1)^{n} H_{0}^{(2)} \left( \frac{\omega H}{U \beta^{2}} \sqrt{(z-z')^{2} + \beta^{2} (z-nh)^{2}} \right) .$$
 (4.33)

No analytical solutions of the integral equations modified in this way seem to be known, and calculations have to be made by approximate and numerical methods. Nevertheless, by analytical treatment of the series Equation (4.33), it was shown in References 4.10 and 4.11 that the sum is infinite when  $\omega h/U = (2m-1)\pi\beta/M$ , (m = 1,2,3...); obviously  $\omega_{\rm p}$  corresponds to m = 1.

237

Anthe Rend & Cash's dam to a bolo

In Reference 4.12 the integral equation corresponding to Equation (4.31) is studied and a numerical collocation method of solution is outlined, in which the usual expansion

$$\overline{l} = \mathbf{a}_0 \cot \frac{1}{2}\theta + \sum_{n=1}^{\infty} \mathbf{a}_n \sin n\theta \qquad (4.34)$$

is assumed and the upwash evaluated at a number of control points on the asrofoil. Three points were found to be enough for h/c = 3.8, and frequencies  $\omega < \omega_r$ . In this approach the aerofoil in a free stream is merely a particular case. Figure 4.3 refers to a wing pitching about the mid-chord axis and shows how the lift varies with frequency, in particular the marked effect as resonance is approached. The experimental phase angle agrees very well with the calculated value but the amplitude of the lift tends to be overestimated by theory. This behaviour is typical for the pitching moment also at Mach numbers from 0.35 to 0.70 (Ref.4.12). Evidently this method of calculation is reasonably satisfactory, provided the numerical work can be tolerated.

Nolyneux (Ref.4.39; 1964) has pointed out that the range of frequency for which the forces on the model are strongly affected by resonance becomes narrower if the size of the model is reduced while the height of the tunnel remains constant. This favours the use of small models. Molyneux also observes that the sharpness of the resonance decreases as M increases, particularly as regards changes in phase angle.

If the frequency is well below critical, the method described in Reference 4.10 may be used. This again is a collocation treatment, besed on the modified form of Equation (4.32), and involving an expansion of the kernel in powers of frequency; it is accordingly restricted in frequency range. When the frequency parameter is small and c/h is not large, the heaving and pitching derivatives for reference axis at the mid-chord (Fig. 4.1) are

$$l_{z} = 0 , \quad l_{z} = \frac{\pi}{\beta} \left( 1 + \frac{\pi^{2}c^{2}}{24\beta^{2}h^{2}} \right)$$

$$\mathbf{s}_{z} = 0 , \quad \mathbf{s}_{z} = \frac{\pi}{4\beta} \left( 1 + \frac{\pi^{2}c^{2}}{48\beta^{2}h^{2}} \right)$$

$$l_{\alpha} = l_{z} , \quad l_{\alpha} = \frac{\pi}{2\beta^{3}} \left[ \left( 1 + \frac{\pi^{2}c^{2}}{48\beta^{2}h^{2}} \right) \left( \frac{3\beta^{2} - 1}{2} \right) - \overline{E} \left( 1 + \frac{\pi^{2}c^{2}}{12\beta^{2}h^{2}} \right) \right]$$

$$\mathbf{s}_{\alpha} = \mathbf{s}_{z} , \quad \mathbf{s}_{\alpha} = -\frac{\pi}{8\beta^{3}} \left[ \left( 1 + \frac{\pi^{2}c^{2}}{16\beta^{2}h^{2}} \right) \overline{E} + \frac{\pi^{2}c^{2}\mathbf{M}^{2}}{32\beta^{2}h^{2}} \right]$$

$$\mathbf{r} = \overline{\mathbf{E}} = \log_{\mathbf{e}} \left\{ 2 \coth \left( \frac{\pi c}{4\beta h} \right) \right\} .$$
(4.35)

According to Equations (4.35)  $l_{\alpha}^{*}$  and  $\mathbf{m}_{\alpha}^{*}$  tend to infinity when h tends to infinity for a fixed c. This agrees with the fact that in a free stream

The

The subscription of the second s

 $(l_{\alpha})_{\mathrm{P}}$  and  $(\mathbf{m}_{\alpha})_{\mathrm{P}}$  do become infinitely large for very small  $\nu$ . Although the tunnel corrections are theoretically infinite, the forces in quadrature with the pitching motion are in fact proportional to  $\nu(l_{\alpha})_{\mathrm{P}}$  and  $\nu(\mathbf{m}_{\alpha})_{\mathrm{P}}$ , which remain finite.

Tabulated tunnel corrections for the compressible case are few in number, and in the nature of illustrative examples only, but enough evidence is available to show that the corrections can be significant in practice. For example, the following table, taken from Reference 4.10, refers to a wing for which h/c = 4.75 oscillating about its mid-chord axis in a stream with M = 0.7.

ν	(1 <sub>a</sub> ) <sub>F</sub>	lα	$(l_{\alpha}^{*})_{F}$	l;	(-m <sub>a</sub> ) <sub>P</sub>	- <b>m</b> α	(∽m <sub>a</sub> ) <sub>F</sub>	- <b>::</b> .
0	4.399	4.556	<del>4</del> 8	8.882	-1.100	-1.119	8	3.012
0.04	4.066	4.510	-12. 981	-8.715	-1.015	-1.106	4.030	2.969
0.08	3.757	4.339	- 8.903	-7.979	-0.933	-1.061	2.981	2.778
0.20	3.117	3.657	- 3.877	-5.084	-0.759	-0.880	1.669	2.023
0.40	2.638	2.975	- 1.274	-2.026	-0.617	-0.694	0.976	1.236

The values  $l_{\alpha}/(l_{\alpha})_{\rm F}$  and  $\tan^{-1}(\nu l_{\alpha}/l_{\alpha})$  have been used to plot a series of points in Figure 4.3. These agree quite well with the trend of variation predicted by Reference 4.12, although the magnitude of the correction is different since the value of c/h is different.

#### 4.4 GENERAL THEORY FOR THREE-DIMENSIONAL WINSS

1 14 S. Fr. 1

In this section the calculation of tunnel interference for three dimensions will be considered from a general viewpoint; corrections for tunnels with rectangular and circular sections will be left until Sections 4.5.1 and 4.5.2.

It will be assumed that the tunnel is a cylinder of infinite length with generators parallel to the x axis, and that linearized inviscid theory may be used. The model will be assumed to lie in the plane z = 0.

At the outset it may be pointed out that, even if the wing is represented by a lifting line, Prandtl's theorem relating the flow at the wing to that far downstream does not apply to oscillatory flow.

Three-dimensional lifting-surface theory has now reached the stage where the calculation of wing loading can be carried out with an accuracy sufficient for application to tunnel interference by one or other of the versions of the "kernel function" or "collocation" theory, for example References 4.13, 4.14 or 4.15. The success of free-stream lifting-surface theory depends on replacing the differential Equation (4.1) by an equivalent integral equation. With the added boundary conditions at the tunnel walls this integral equation becomes extremely complicated. In the absence of special simplifying features an analytical solution of the differential equation is not possible; although a numerical finite-difference treatment might

#### 239

語いの語言であることで

\* Y\*882 ....

#### 240

conceivably be attempted with a high-speed computer, no programme for doing so seems to exist. In practice, therefore, we are largely confined to tunnels for which the method of images is possible, or to flow conditions ( $N\overline{\nu} \ll 1$ ) for which the method suggested in Section 4.4.2 is available.

Even when the flow is such that the lifting-surface theory is of doubtful applicability, having for example local supersonic regions or separation from the leading edge of the wing, the theory may be used to estimate the far flow field. Provided that the measured derivatives are available to adjust the predictions of linearized theory, the conditions at the tunnel boundary, and hence the interference upwash, can be established fairly accurately.

#### 4.4.1 Tunnel Resonance

This phenomenon has already been discussed in connection with two-dimensional tunnels, but it seems reasonable to suppose that something similar will occur in three dimensions also.

In Reference 4.16 arguments are put forward to show that for a three-dimensional tunnel the critical frequency for resonance is given by

$$\frac{\omega_{\rm r}}{\rm U} = \frac{\beta \lambda}{\rm M} , \qquad (4.36)$$

where  $\lambda$  is the smallest eigenvalue of the equation

$$\frac{\partial^2 \phi}{\partial \mathbf{v}^2} + \frac{\partial^2 \phi}{\partial z^2} + \lambda^2 \phi = \mathbf{0} \quad . \tag{4.37}$$

This equation is to be satisfied inside the empty tunnel cross-section, with the boundary conditions that  $\partial \phi/\partial n = 0$  at a solid wall and  $\phi = 0$  at an open boundary. Modes for which the derivative of  $\phi$  normal to the plane of the wing is zero, may be ignored. Reference 4.16 includes some calculations for open and closed threedimensional tunnels and for three-dimensional slotted tunnels. Some experimental evidence supporting the view that resonance effects occur in three dimensions may be found in Reference 4.17.

For a rectangular tunnel with solid roof and floor the relevant critical frequency turns out to be the same as for the two-dimensional tunnel of the same height. If the tunnel has open roof and floor the critical frequency is doubled. For a closed . circular tunnel of radius R the critical frequency is given by

$$\frac{\omega_{\rm r} {\rm B}}{{\rm U}} = 1.84 \frac{\beta}{{\rm M}} , \qquad (4.38)$$

where x = 1.84 is the first zero of the Bessel function  $J'_{1}(x)$ .

For other cross-sections the ease or difficulty of finding the critical resonance frequency is simply that of determining the eigenvalue  $\lambda$  of Equation (4.37); in general this will have to be done numerically.

## 4.4.2 Relations Between Steady and Unsteady Interference

- -

The basic idea underlying the method given below is due to Goodman (Ref. 4. 18; 1953), but the development differs from his.

Let it be assumed that the wing and its wake lie in the plane z = 0. Then according to linearized theory<sup>4.13</sup> in an unbounded stream  $\overline{\Phi}$  is antisymmetric with respect to z, and

$$\bar{\Phi}(x,y,z) = -\frac{1}{2\pi} \iint \bar{\Phi}(x',y',+0) K(x-x',y-y',z) dx'dy', \quad (4.39)$$

where

$$K = \exp\left(\frac{i\omega M^{2}(x-x')}{U\beta^{2}}\right) \frac{\partial}{\partial z} \left[\exp\left\{-\frac{i\omega MR_{\beta}}{U\beta^{2}}\right\} / R_{\beta}\right]$$
(4.40)

with

$$R_{\beta} = [(x-x')^{2} + \beta^{2}(y-y')^{2} + \beta^{2}z^{2}]^{\frac{1}{2}} . \qquad (4.41)$$

The region of integration in Equation (4.39) is that part of the plane z = 0 in which  $\Phi(x', y', + 0) \neq 0$ , that is the wing and its wake. If the interference effect can be represented by an image system, Equation (4.39) provides formally a method of calculating the interference flow field due to the images, but the computation would be formidable. Nevertheless a practical procedure can be developed provided  $\omega M$  is small.

Consider first a steady horse-shoe vortex of circulation  $\overline{\Gamma}$  and span 2s in an incompressible flow. This may be represented by

$$\begin{split} \bar{\Phi}(\mathbf{x}, \mathbf{y}, +0) &= \frac{1}{2} \bar{\Gamma} \quad \text{for} \quad \mathbf{x} > 0 , \qquad |\mathbf{y}| < \mathbf{s} \\ &= 0 \quad \text{for} \quad \mathbf{x} < 0 \quad \text{or} \quad |\mathbf{y}| > \mathbf{s} \end{split} \right\} . \tag{4.42}$$

Then by Equation (4.39) the associated flow field is given by

$$\Phi = \Phi_{\mathbf{y}}(\mathbf{x}, \mathbf{y}, \mathbf{z}) = -\frac{1}{2\pi} \int_{\mathbf{x}'=0}^{\infty} \int_{\mathbf{y}'=-8}^{5} \frac{1}{2} \overline{\Gamma} \frac{\langle -\mathbf{z} \rangle}{[(\mathbf{x}-\mathbf{x}')^2 + \langle \mathbf{y}-\mathbf{y}' \rangle^2 + \mathbf{z}^2]^{3/2}} \, d\mathbf{x}' d\mathbf{y}' \quad (4.43)$$

Hence

$$\Phi_{\rm V} = \frac{\bar{\Gamma}_{\rm Z}}{4\pi} \int_{-\infty}^{\rm X} \frac{1}{\xi^2 + z^2} \left[ \frac{(s-y)}{[\xi^2 + (s-y)^2 + z^2]^{\frac{1}{2}}} + \frac{(s+y)}{[\xi^2 + (s+y)^2 + z^2]^{\frac{1}{2}}} \right] d\xi \quad . \quad (4.44)$$

so that

$$\frac{\partial \Phi_{y}}{\partial x} = \frac{\tilde{\Gamma}_{z}}{4\pi} \frac{1}{x^{2} + z^{2}} \left[ \frac{(s-y)}{\left[x^{2} + (s-y)^{2} + z^{2}\right]^{\frac{1}{2}}} + \frac{(s+y)}{\left[x^{2} + (s+y)^{2} + z^{2}\right]^{\frac{1}{2}}} \right] . \quad (4.45)$$

and the state of t

TO A REPORT OF THE PARTY OF THE

Now consider an oscillatory horse-shoe vortex of the same span with circulation  $\Gamma= \widetilde{\Gamma}^{i\omega t}$  , so that

$$\overline{\Phi}(\mathbf{x}^{t}, \mathbf{y}^{t}, + 0) = \frac{1}{2}\overline{\Gamma}\exp\left(-\frac{\mathbf{i}\omega\mathbf{x}^{t}}{3}\right) \quad \text{for} \quad \mathbf{x}^{t} > 0, \qquad |\mathbf{y}^{t}| < \mathbf{s}$$
$$= 0 \qquad \qquad \text{for} \quad \mathbf{x}^{t} < 0 \quad \text{or} \quad |\mathbf{y}^{t}| > \mathbf{s}$$
$$(4.46)$$

Hence, if the flow is incompressible,

$$\bar{\Phi} = -\frac{\bar{\Gamma}}{4\pi} \int_{x'=0}^{\infty} \exp\left(-\frac{i\omega x'}{U}\right) \int_{y'=-S}^{S} \frac{(-z)}{[(x-x')^2 + (y-y')^2 + z^2]^{3/2}} dx' dy' \quad (4.47)$$

$$= \frac{\bar{\Gamma}z}{4\pi} \int_{-\infty}^{x} \exp\left(\frac{i\omega(\xi-x)}{U}\right) \frac{1}{\xi^2 + z^2} \left[\frac{(s-y)}{[\xi^2 + (s-y)^2 + z^2]^{\frac{1}{2}}} + \frac{(s+y)}{[\xi^2 + (s+y)^2 + z^2]^{\frac{1}{2}}}\right] d\xi$$

$$= \int_{-\infty}^{x} \exp\left(\frac{i\omega(\xi-x)}{U}\right) \frac{\partial \Phi_{y}(\xi,y,z)}{\partial \xi} d\xi \quad . \quad (4.48)$$

Alternatively Equation (4.48) may be written

$$\overline{\Phi} = \Phi_{\mathbf{v}}(\mathbf{x}, \mathbf{y}, \mathbf{z}) - \frac{i\omega}{v} \exp\left(-\frac{i\omega \mathbf{x}}{v}\right) \int_{-\infty}^{\mathbf{x}} \exp\left(\frac{i\omega\xi}{v}\right) \Phi_{\mathbf{v}}(\xi, \mathbf{y}, \mathbf{z}) \, d\xi \quad . \tag{4.49}$$

Equation (4.49) refers to the free-stream values of  $\overline{\Phi}$  and  $\overline{\Phi}_{\mathbf{v}}$ . If  $\overline{\Phi}_{\mathbf{v}}$  is taken to represent the potential of a horse-shoe vortex in the presence of a tunnel. a function  $\overline{\Phi}$  may still be calculated from it by Equation (4.49). Now  $\nabla^2 \Phi_{\mathbf{v}} = 0$ and it follows from Equation (4.48) and the vanishing of  $\overline{\Phi}_{\mathbf{v}}$  and its derivatives at x=- $\infty$  that  $\nabla^2 \overline{\Phi} = 0$ . Moreover, if on the walls  $\partial \overline{\Phi}_{\mathbf{v}} / \partial \mathbf{n} = 0$ , it follows from Equation (4.48) or (4.49) that  $\partial \overline{\Phi} / \partial \mathbf{n} = 0$ ; similarly if  $\Phi_{\mathbf{v}} = 0$  at the tunnel boundary, then  $\overline{\Phi} = 0$ . Thus  $\Phi = \overline{\Phi} e^{i\omega t}$  is the oscillatory velocity potential in the presence of the tunnel walls. Since both the free-stream and wind-tunnel values of  $\overline{\Phi}$  and  $\Phi_{\mathbf{v}}$  are related by Equations (4.48) and (4.49), it follows that the interference fields satisfy the same relations.

The foregoing argument is based on the assumption that M = 0. If M is not zero then  $\overline{\Phi}$  is found by substituting from Equation (4.46) into Equation (4.39). This yields

$$\overline{\Phi}(\mathbf{x},\mathbf{y},\mathbf{z}) = -\frac{1}{2\pi} \int_{\mathbf{y}'=0}^{\infty} \int_{\mathbf{y}'=-\mathbf{s}}^{\mathbf{s}} \exp\left(-\frac{\mathbf{i}\omega\mathbf{x}'}{\mathbf{y}}\right) \mathbf{k}(\mathbf{x}-\mathbf{x}',\mathbf{y}-\mathbf{y}',\mathbf{z}) d\mathbf{x}' d\mathbf{y}'$$

The second of the

243

Strangenty.

ì

that is, by Equations (4.40) and (4.41)

$$\Phi = \frac{1}{2\pi} \int_{\mathbf{x}'=0}^{\infty} \int_{\mathbf{y}'=-6}^{9} \frac{\frac{1}{2} \widehat{\Gamma} \exp\left(-\frac{2\omega r}{U}\right) \exp\left(\frac{2\omega \mathbf{x}'}{U\beta^2} \exp\left(-\frac{2\omega \mathbf{x}'}{U\beta^2}\right) \exp\left(-\frac{2\omega \mathbf{x}}{U\beta^2}\right) \exp\left(-\frac{2\omega \mathbf{x}}{U\beta^2}\right) \exp\left(-\frac{2\omega \mathbf{x}}{U\beta^2}\right) \frac{\beta^2 z}{U\beta^2} \left(1 + \frac{2\omega \mathbf{x}}{U\beta^2}\right) \frac{\beta^2 z}{U\beta$$

Then by an argument similar to that in Appendix I of Reference 4.19 it can be shown that for small  $\omega$  Equation (4.53) may be explained by

$$\overline{\Phi} = \frac{\beta^2 \overline{\Gamma}_z}{4\pi} \int_{z'=0}^{\infty} \int_{y'=-s}^{s} \exp\left(\frac{i\omega z'}{\overline{u}}\right) e_{\varphi}\left(\frac{i\omega N^2(z-z')}{U\beta^2}\right) \frac{1}{N_{\beta}^3} dz' dz' + O(\omega^2 \log \omega) . \quad (4.51)$$

If the integration with respect to  $y^2$  is performed it follows from Equation (2.45) that  $ex/\beta$ 

$$\bar{\Phi} = \int_{-\infty}^{\Lambda/P} \exp\left(-\frac{i\omega z}{v}\right) \exp\left(\frac{i\omega \xi}{v\beta}\right) \frac{\partial \Phi_{v}(\xi, y, z)}{\partial \xi} d\xi + O(\omega^{2} \log \omega) \quad . \tag{4.52}$$

By virtue of the fact that  $\nabla^2 \Phi_0 = 0$  it follows that  $\overline{\Phi}$ , as given by the inversal in Equation (4.52), satisfies

$$\beta^2 \frac{\partial^2 \bar{\phi}}{\partial x^2} + \frac{\partial^2 \bar{\phi}}{\partial y^2} + \frac{\partial^2 \bar{\phi}}{\partial z^2} - \frac{2i\omega \mathbf{M}^2}{\mathbf{U}} \frac{\partial \bar{\phi}}{\partial x} - \frac{\omega^2 \mathbf{M}^2}{\mathbf{U}^2 \beta^2} = \mathbf{0} , \qquad (4.53)$$

so that, provided  $\omega^2$  is negligible,  $\overline{\Phi}$  satisfies Equation (4.5). Moreover as for incompressible flow  $\overline{\Phi}$  represents flow in the tunnel, or in a free stream, according as  $\Phi_{\rm V}$  does so; hence, if  $\Phi_{\rm V}$  represents the interference in steady incompressible flow, then  $\overline{\Phi}$  is the interference in oscillatory compressible flow with an error of order  $\omega^2 \log \omega$ . An alternative form of Equation (4.52) is

$$\bar{\Phi} = \exp\left(\frac{i\omega x^2 x}{v\beta^2}\right) \Phi_{\nu}\left(\frac{x}{\beta}, y, z\right) - \frac{i\omega}{v\beta} \exp\left(-\frac{i\omega x}{v}\right) \int_{-\infty}^{x/\beta} \exp\left(\frac{i\omega\xi^2}{v\beta}\right) \Phi_{\nu}(\xi, z, z) d\xi , \quad (4.54)$$

which reduces to Equation (4.49) if N = 0.

The interference upwash angle due to an oscillatory horse-shoe vertex of span 2s and circulation  $\Gamma = \overline{\Gamma}e^{i\omega t}$  will be written in the form

$$\frac{\mathbf{x}_{1}}{\mathbf{v}} = \frac{\overline{\mathbf{x}}_{1}}{\mathbf{v}} e^{i\omega t} = \frac{\overline{\mathbf{l}} e^{i\omega t}}{\mathbf{v} d} \delta \left( \frac{\mathbf{x}}{\mathbf{d}}, \frac{\mathbf{y}}{\mathbf{d}}; \frac{\mathbf{s}}{\mathbf{d}}, \frac{\omega \mathbf{i}}{\mathbf{v}} \right), \quad (4.55)$$

-----

where d is a representative length of the tunnel cross-section. Then Equation (4.54) becomes

Last

$$\delta(\xi, \eta; \sigma, \mu) = \exp\left(\frac{i\mu M^2 \xi}{\beta^2}\right) \delta\left(\frac{\xi}{\beta}, \eta; \sigma, \theta\right) - \frac{i\mu}{\beta} \exp\left(-i\mu\xi\right) \int_{-\infty}^{\xi/\beta} \exp\left(\frac{i\mu\theta}{\beta}\right) \delta(\theta, \eta; \sigma, \theta) d\theta$$
$$= \exp\left(-i\mu\xi\right) \int_{-\infty}^{\xi/\beta} \frac{\partial}{\partial \theta} \left[\delta(\theta, \eta; \sigma, \theta)\right] \exp\left(\frac{i\mu\theta}{\beta}\right) d\theta , \qquad (4.56)$$

where

$$\begin{cases} = \mathbf{x}/\mathbf{d} \\ \eta = \mathbf{y}/\mathbf{d} \\ \sigma = \mathbf{s}/\mathbf{d} \\ \mu = \mathbf{c}\mathbf{d}/\mathbf{U} \end{cases}$$
(4.57)

#### 4.4.3 General Interference Corrections

In order to apply the theory of Section 4.4.2, it is necessary to know the interference upwash due to a steady horse-shoe vortex, and moreover this must be known for all puints upstream of the model. It is then necessary to express the potential distribution round the oscillating model in terms of those of horse-shoe vortices.

Consider first a steady lift distribution symmetrical with respect to y. It may be shown<sup>4,20</sup> that the interference upwash is given by

$$\frac{\mathbf{w}_{\mathbf{i}}(\mathbf{x},\mathbf{y})}{\mathbf{u}} = \frac{1}{2d} \int_{\mathbf{y}'=0}^{\mathbf{s}} \int_{\mathbf{x}'=\mathbf{x}_{\mathbf{i}}(\mathbf{y}')}^{\mathbf{x}_{\mathbf{i}}(\mathbf{y}')} \overline{l}(\mathbf{x}',\mathbf{y}') \frac{\partial}{\partial \mathbf{y}'} \left[ \delta\left(\frac{\mathbf{x}-\mathbf{x}'}{\mathbf{d}},\frac{\mathbf{y}}{\mathbf{d}};\frac{\mathbf{y}'}{\mathbf{d}},\frac{\omega \mathbf{d}}{\mathbf{u}}\right) \right] d\mathbf{x}' d\mathbf{y}'$$

$$\approx \frac{1}{2} \int_{\eta'=0}^{\sigma} \int_{\xi'=\bar{c}_{\mathbf{i}}(\eta')}^{\xi'_{\mathbf{x}}(\eta')} \overline{l}(\xi',\eta') \frac{\partial}{\partial \eta'} \left[ \delta(\xi-\xi',\eta;\eta',\mu) \right] d\xi' d\eta' , \qquad (4.58)$$

where  $\delta$  is given by Equation (4.56). The planes y = 0 and  $\eta = 0$  need not be symmetrically situated in the tunnel, indeed the tunnel need not have any symmetry at all provided that it is cylindrical, but  $\delta$  must refer to a horse-shoe vortex lying in z = 0 and symmetrical about y = 0.

For a distribution of lift antisymmetrical with respect to y = 0, Equation (4.58) will still hold<sup>=.29</sup>, provided  $\delta(\xi, \eta; \sigma, 0)$  now defines the interference upwash angle due to the distribution of potential on  $z = \pm 0$  given by

A State of the second second second

$$\begin{array}{l} \overline{p} = \frac{y}{2|y|} \overrightarrow{\overline{l}} \quad \text{for} \quad x > 0 \quad \text{and} \quad |y| < s \\ = 0 \quad \text{for} \quad x < 0 \quad \text{or} \quad |y| > s \end{array} \right\} .$$
 (4.59)

The arguments of Section 4.4.2 apply to such a distribution; in particular Equation (4.56) still holds. Since any lift distribution may be split into symmetrical and antisymmetrical parts, the corresponding interference upwash may be calculated.

A CONTRACTOR OF THE AND A CONTRACT OF THE ADDRESS O

Of course in most experiments  $\overline{l}$  will not be measured in sufficient detail to apply Equation (4.58) directly. Generally one or more derivatives will be found, that is to say there will be available weighted integrals of  $\overline{l}$  over the planform. A possible simplification is to assume that the real and imaginary parts of  $\overline{l}$  may be concentrated on the curves passing through their respective theoretical centres of pressure at each spanwise position. Suppose the centres of pressure are defined by

$$\xi' = \xi_{\mathbf{n}}(\eta') \quad \text{and} \quad \xi' = \xi_{\mathbf{T}}(\eta') \quad . \tag{4.60}$$

Then Equation (4.58) becomes

$$\frac{\mathbf{v}_{\mathbf{j}}(\xi,\eta)}{\mathbf{U}} = \frac{1}{2} \int_{0}^{T} \left\{ L_{\mathbf{R}}(\eta') \frac{\partial}{\partial \eta'} \left[ \delta(\xi_{1},\eta;\eta',\mu) \right] + \frac{\partial}{\partial \eta'} \left[ \delta(\xi_{2},\eta;\eta',\mu) \right] \right\} d\eta' , \qquad (4.81)$$

where

$$L_{R}(\eta') + iL_{I}(\eta') = \int_{\xi_{L}(\eta')}^{\xi_{I}(\eta')} \bar{l}(\xi', \eta') d\xi' , \qquad (4.62)$$

and after the differentiation we put

$$\xi_1 = \xi - \xi_1(\eta')$$
,  $\xi_2 = \xi - \xi_1(\eta')$ . (4.63)

This should be a valid approximation provided the chord of the wing is small compared with the dimensions of the tunnel.

An alternative method may be found by observing that the circulation round any section of the wing is given by

$$\overline{\Gamma} = \overline{\Gamma}_{R} + i\overline{\Gamma}_{I} = \frac{\mathrm{Ed}}{2} \int_{\xi_{L}(\eta')}^{\xi_{T}(\eta')} \overline{l}(\xi', \eta') \exp \{i\mu[\xi - \xi_{T}(\eta')]\} d\xi' , \quad (4.64)$$

and then regarding  $\overline{\Gamma}_{\mathbf{R}}$  and  $\overline{\Gamma}_{\mathbf{I}}$  as being concentrated into vortices along the curves  $\xi' = \xi_{\mathbf{R}}(\eta')$  and  $\xi' = \xi_{\mathbf{I}}(\eta')$ . Then in Equation (4.61)  $L_{\mathbf{R}}$  must be replaced by



 $2\overline{\Gamma}_{R}/M$  and  $L_{-}$  by  $2\overline{\Gamma}_{I}/M$ . Provided that the chord is small, the difference in the calculated values of  $w_{i}/M$  will be scall. If the chord is large neither form will be mourate, but fortchately fairly crude representations of the model are usually satisfactory for interference calculations.

A further approximation is obtained by assuming that the span of the wing, as well as its chord, is small relative to the tunnel. This will be dealt with in Section 4.5.3.

The calculation is very much facilitated when a theoretical treatment by one of the available lifting-surface theories for oscillating wings (e.g. Refs. 4.13 to 4.15) is carried out in conjunction with the experiments. If such a calculation has been performed, there are available a theoretical lift distribution and the required theoretical derivatives. Suppose that the wing is oscillating in some mode

$$z = q_0 e^{i\omega t} p(x, y) = q p(x, y) ,$$
 (4.65)

and that measurements are made of the amplitude q and the generalized forces

$$P_{n} = \iint_{S} \overline{l}(x, y) p_{n}(x, y) dxdy \quad (n = 1, 2, ...) , \qquad (4.66)$$

where  $P_n$  is a weighting function. The theoretical free-stream lift distribution and derivatives  $(\partial P_n/\partial q)_p$  may be calculated, and hence the corresponding theoretical interference upwash  $\overline{w}_i$  is evaluated from Equation (4.58). Again by lifting-surface theory the distribution  $\overline{w}_i$  may be converted into theoretical increments  $(\partial P_n/\partial q)_i$ due to wall interference. A linear factor  $(P_1/q)/(\partial P_1/\partial q)_p$  is applied to the theoretical free-stream loading, so that it becomes consistent with the measured generalized force  $P_i$ . Then the contribution to  $P_n$  due to wall interference is

$$\delta P_{n} = \frac{P_{1}(\partial P_{n}/\partial q)_{i}}{(\partial P_{1}/\partial q)_{p}} . \qquad (4.67)$$

The wall interference can be regarded as a correction  $\Delta q$  to be applied to the measured amplitude q and residual corrections

$$\Delta P_n = -\delta P_n + \Delta q (\partial P_n / \partial q)_p \quad . \tag{4.68}$$

It may be convenient to choose

$$\frac{\Delta q}{P_{i}} = \frac{(\partial P_{i}/\partial q)_{i}}{(\partial P_{i}/\partial q)_{p}^{2}} .$$
so that
$$\Delta P_{i} = 0$$

$$\frac{\Delta P_{n}}{P_{i}} = \frac{1}{(\partial P_{i}/\partial q)_{p}^{2}} \left[ \left( \frac{\partial P_{q}}{\partial q} \right)_{p} \left( \frac{\partial P_{i}}{\partial q} \right)_{i} - \left( \frac{\partial P_{i}}{\partial q} \right)_{p} \left( \frac{\partial P_{n}}{\partial q} \right)_{i} \right] \quad (n \ge 2)$$

$$(4.69)$$

The right-hand sides of Equations (4.69) are theoretically determined complex numbers, so that in general  $\Delta q$  involves a phase shift as well as a change in amplitude.

If only one force has been measured there is no choice in the above procedure, but if several have been measured some care must be taken. For example, if  $P_n$  is a control hinge moment giving no information about  $\overline{l}(x,y)$  off the control surface, then it is probably best to take  $P_1$  to be the overall lift, if this is measured or can be estimated.

### 4.5 APPLICATION TO THREE-DIMENSIONAL WINGS IN CLOSED TUNNELS

The general theory of Section 4.4 is now applied to closed rectangular and circular tunnels (Sections 4.5.1 and 4.5.2). There are great simplifications for saall mings (Section 4.5.3), and these form the basis of an approximate treatment of wall interference for oscillating slender wings (Section 4.5.4). Some numerical examples are given in Section 4.5.5.

#### 4.5.1 Ecctangular Tunnels

States and the

Before applying the theory developed above we observe that, since the interference in a rectangular tunnel may be represented by an image system, expressions for the interference potential may be written down at once by a simple extension of freestream subsonic lifting-surface theory. For a wing symmetrically placed in a closed rectangular tunnel the kernel function in Equation (4.39) has to be replaced by

$$\sum_{m=-\infty}^{\infty} \sum_{m=-\infty}^{\infty} (-1)^{n} K(x-x', y-y'-mb, z-z'-nh) , \qquad (4.70)$$

where b is the breadth of the tunnel and h its height. The interference upwash for a given  $\overline{\Phi}(\mathbf{x}', \mathbf{y}', + 0)$  is obtained by omitting the term  $\mathbf{x} = \mathbf{n} = 0$  from the double summation; alternatively Equation (4.39) as modified by Equation (4.70) may be regarded as an integral equation for  $\overline{\Phi}(\mathbf{x}', \mathbf{y}', + 0)$ . The same remarks apply to an alternative form of lifting-surface theory such as Esference 4.15 which uses the loading  $\overline{l}$  as the unknown instead of  $\overline{\Phi}$ . For the wing symmetrically situated in a rectangular tunnel this gives

$$\frac{\overline{\mathbf{w}}(\mathbf{x},\mathbf{y},\mathbf{0})}{\mathbf{u}} = \lim_{\mathbf{z}\to\mathbf{0}} \left[ -\frac{1}{8\pi} \iint_{\mathbf{z}} \overline{l}(\mathbf{x}^{2},\mathbf{y}^{1}) \sum_{\mathbf{y}=-\infty}^{\infty} \sum_{\mathbf{n}=-\infty}^{\infty} (-1)^{\mathbf{r}} \widetilde{\mathbf{k}}(\mathbf{x}-\mathbf{x}^{1},\mathbf{y}-\mathbf{y}^{1}-\mathbf{z}\mathbf{b},\mathbf{z}-\mathbf{n}\mathbf{b}) \, \mathrm{d}\mathbf{x}^{1} \mathrm{d}\mathbf{y}^{1} \right] .$$

$$(4.71)$$

where

$$\widehat{\mathbf{X}}(\mathbf{x}-\mathbf{x}', \mathbf{y}-\mathbf{y}', \mathbf{z}) = \frac{\partial^2}{\partial \mathbf{z}^2} \int_{-\infty}^{\mathbf{1}-\mathbf{x}} \frac{\exp\left\{-\frac{i\omega}{\overline{\mathbf{U}}}\left[(\mathbf{x}-\mathbf{x}')-\frac{\widehat{\boldsymbol{\xi}}}{\beta^2}+\frac{\mathbf{x}}{\beta^2}\sqrt{\widehat{\boldsymbol{\xi}}^2+\beta^2(\mathbf{y}-\mathbf{y}')^2+\beta^2\mathbf{z}^2}\right]\right\}}{\sqrt{\widehat{\boldsymbol{\xi}}^2+\beta^2(\mathbf{y}-\mathbf{y}')^2+\beta^2\mathbf{z}^2}} d\widehat{\boldsymbol{\xi}}'.$$

(4.72)

Unfortunately both of the expressions Equations (4.70) and (4.71) are too unwieldy for practical computation, except in special cases. Moreover, in view of the remarks on tunnel resonance in Section 4.4.1, it may be expected that the sums will become infinite for critical values of  $\omega$ .

In order to apply the method proposed in Section 4.4.2 it is necessary to consider first the interference in the steady case. From Equation (4.55) and Equation (7) of Reference 4.20 it may be shown that\*

$$\delta(\xi,\eta;\sigma,0) = \frac{d}{b} \left\{ \mathbf{F}_1\left(\frac{\mathbf{x}}{b},\frac{\mathbf{y}-\mathbf{s}}{b}\right) - \mathbf{F}_1\left(\frac{\mathbf{x}}{b},\frac{\mathbf{y}+\mathbf{s}}{b}\right) \right\} \quad . \tag{4.73}$$

mbere

$$\mathbf{F}_{1}(\kappa,\lambda) = \frac{1}{4\pi} \frac{b}{h} \left\{ \mathbf{f}_{1}\left(\frac{b\kappa}{h}, \frac{b\lambda}{h}\right) + \pi \operatorname{ccsech} \frac{\pi b\lambda}{h} - \frac{h}{b\lambda} + \frac{\sum_{\mathbf{M}=1}^{\infty} \left[ \mathbf{f}\left(\frac{b\kappa}{h}, \frac{b(\lambda+\mathbf{m})}{h}\right) + \mathbf{f}\left(\frac{b\kappa}{h}, \frac{b(\lambda-\mathbf{m})}{h}\right) + \frac{\pi \operatorname{ccsech}}{h} \frac{\pi b(\lambda+\mathbf{m})}{h} + \pi \operatorname{ccsech} \frac{\pi b(\lambda-\mathbf{m})}{h} \right] \right\}, \quad (4.74)$$

$$f(x,y) = \sum_{n=\infty}^{\infty} (-1)^n \frac{xy}{\sqrt{x^2 + y^2 + n^2}} \left\{ \frac{1}{x^2 + n^2} + \frac{1}{y^2 + n^2} \right\} , \qquad (4.75)$$

and

$$f_1(x,y) = f(x,y) - \frac{xy}{\sqrt{x^2 + y^2}} \left\{ \frac{1}{x^2} + \frac{1}{y^2} \right\}$$
 (4.76)

Tables of f(x,y) may be found in Reference 4.20.

2

Then by Equations (4.73) and (4.56)

٢

$$\delta(\xi, \eta; \sigma, \mu) = \exp(-i\mu\xi) \int_{-\infty}^{\xi/\beta} \frac{d}{b} \frac{\partial}{\partial\theta} \left[ F_1\left(\frac{\theta d}{b}, \frac{d(\eta-\sigma)}{b}\right) - F_1\left(\frac{\theta d}{b}, \frac{d(\eta+\sigma)}{b}\right) \right] \exp\left(\frac{i\mu\theta}{\beta}\right) d\theta \quad . \tag{4.77}$$

\* In Reference 4.20 the typical length d was taken to be the tunnel breadth b ; in the present exposition d will be left general until numerical examples are given in Section 4.5.5.

248

CHART STAT

merer and the

No difficulty arises in these calculations since the series in Equation (4.74) is rapidly convergent, and the infinite integral in Equation (4.77) converges by virtue of the fact that  $\mathcal{F}_1 = O(\xi^{-2})$  as  $\xi \to \infty$ .

-

If 
$$\mu$$
 is small,  

$$\delta(\xi, \eta; \sigma, \mu) = \frac{d}{b} \left( 1 + \frac{i\mu M^2 \xi}{\beta^2} \right) \left[ F_1 \left( \frac{\xi d}{\beta b}, \frac{d}{b} (\eta - \sigma) \right) - F_1 \left( \frac{\xi d}{\beta b}, \frac{d}{b} (\eta + \sigma) \right) \right] - \frac{i\mu d}{\beta b} \int_{-\infty}^{\xi/\beta} \left[ F_1 \left( \frac{\partial d}{b}, \frac{d(\eta - \sigma)}{b} \right) - F_1 \left( \frac{\partial d}{b}, \frac{d(\eta + \sigma)}{b} \right) \right] d\theta . \quad (4.78)$$

Further, it follows from Equation (4.77) that  $\delta$  tends to zero as  $\mu$  tends to infinity.

The function required for the calculation of the interference upwash by Equation (4.58) or (4.61) is  $\partial[\delta(\xi,\eta;\eta',\mu)]/\partial\eta'$ . If we put

$$\mathbf{F}_{2}(\kappa,\lambda) = -\frac{\partial \mathbf{F}_{1}(\kappa,\lambda)}{\partial\lambda}$$
(4.79)

and

$$\mathbf{G}(\boldsymbol{\xi}, \boldsymbol{\eta}; \boldsymbol{\mu}) = \frac{\mathrm{d}^2}{\mathrm{b}^2} \exp\left(\frac{\mathrm{i}\boldsymbol{\mu} \mathbf{M}^2 \boldsymbol{\xi}}{\beta^2}\right) \mathbf{F}_2\left(\frac{\boldsymbol{\xi} \mathrm{d}}{\beta \mathrm{b}}, \frac{\boldsymbol{\eta} \mathrm{d}}{\mathrm{b}}\right) - \frac{\mathrm{i}\boldsymbol{\mu} \mathrm{d}^2}{\beta \mathrm{b}^2} \exp\left(-\mathrm{i}\boldsymbol{\mu} \boldsymbol{\xi}\right) \int_{-\infty}^{\boldsymbol{\xi}/\beta} \exp\left(\frac{\mathrm{i}\boldsymbol{\mu} \boldsymbol{\theta}}{\beta}\right) \mathbf{F}_2\left(\frac{\boldsymbol{\theta} \mathrm{d}}{\mathrm{b}}, \frac{\boldsymbol{\eta} \mathrm{d}}{\mathrm{b}}\right) \mathrm{d}\boldsymbol{\theta} \quad .$$
(4.80)

then

$$\frac{\partial}{\partial \eta'} \left[ \delta(\xi, \eta; \eta', \mu) \right] = \theta(\xi, \eta - \eta'; \mu) + \theta(\xi, \eta + \eta'; \mu) \quad (4.81)$$

and, by Equation (4.52), for a symmetrical lift distribution,

$$\frac{u_{1}(\xi,\eta)}{U} = \frac{1}{2} \int_{\eta'=-\sigma}^{\sigma} \int_{\xi'=\xi_{1}(\eta')}^{\xi_{1}(\eta')} \overline{l}(\xi',\eta') \mathfrak{g}(\xi-\xi',\eta-\eta';\mu) \, d\xi' d\eta' \quad (4.82)$$

The interference upwash will be required at whatever set of points is employed in the lifting-surface theory used to estimate  $\overline{I}$ . This fixes the set of values of  $(\xi,\eta)$  in Equation (4.82), and hence the region in which  $\Theta(\xi,\lambda;\mu)$  has to be determined. The tabulation of  $\Theta(\xi,\lambda;\mu)$  in this region is the main part of the computation involved, and has to be followed by integration over the planform. This method involves too such numerical work to be generally practicable.

The approximate Equation (4.61) may be written

17.74 G

$$\frac{\overline{\mathbf{w}}_{1}(\xi,\eta)}{\mathbf{u}} = \frac{1}{2} \int_{\eta'=-\sigma}^{\sigma} \{\mathbf{L}_{\mathbf{R}}(\eta')\mathbf{q}[\xi-\xi_{\mathbf{R}}(\eta'),\eta-\eta';\mu] + \mathbf{L}_{\tau}(\eta')\mathbf{q}[\xi-\xi_{\mathbf{R}}(\eta'),\eta-\eta';\mu]\} \, \mathrm{d}\eta' \quad (4.83)$$

This still involves the tabelation of  $G(\xi, \eta; \mu)$  for one or more values of  $\mu$ , although only a single integration is necessary after that. A few examples of calculations by this method are given in Reference 4.20.

For routine use it is desirable to represent the vorticity distribution over the wing by discrete horse-shoe vortices, as few in number as possible. The calculation then reduces to the determination of  $\delta$  for each of them. By Equation (4.77) this involves the calculation of  $F_1(\xi,\lambda)$  for several specified values of  $\lambda$  and all points,  $\xi$ , upstream of the model.

The calculation for antisymmetrical lift distributions is formally similar; the basic equations may be found in Reference 4.20.

#### 4.5.2 Circular Tunnels

Just as for the rectangular tunnel, the first point which arises is the availability of the interference upwash of a horse-shoe vortex in steady flow. In Reference 4.21 expressions are obtained for the interference upwash due to an elementary horse-shoe vortex arbitrarily situated in the tunnel, but they are rather complicated infinite series and only a few sumerical values are given. Eisenstadt (Ref. 4.22; 1947) has given tables of interference upwash for horse-shoe vortices covering a range of spans and upwash positions which should be adequate for the majority of wings. Eisenstadt's tables include the interference upwash for horse-shoe vortices with swept cross-stream parts and can mercover be used for antisymmetrical as well as symmetrical lift distributions.

Once the required steady data have been obtained, the procedure of Reference 4.23 can be applied; this is almost identical to that for a rectangular tunnel, and it is unnecessary to elaborate on it here. The results of a typical calculation are given in Section 4.5.5.

#### 4.5.3 Small Wings

If the overall chord and span are both small compared with the tunnel dimensions, the lift may be regarded as being concentrated at a single point. Consider first a closed rectangular tunnel and suppose the small wing is at the origin. Equation (4.82) then becomes

$$\frac{\overline{u}_{1}}{\overline{u}} = \frac{S\overline{C}_{L}}{2d^{2}} G(\xi, \eta; \mu) . \qquad (4.84)$$

-------

л: • ...

W. 11. 302 412 .

251

ŝ

Since the wing is small, only small values of  $\xi$  and  $\eta$  need be included. Take  $\eta = 0$  and suppose that  $\xi^2$  is negligibly small. Then by Equations (4.80) and (4.84).

ţ

$$\frac{\widetilde{\mathbf{w}}_{1}(\xi,0)}{\overline{\mathbf{u}}} = \frac{\widetilde{\mathbf{x}}_{L}}{2b^{2}} \left[ \mathbf{P}_{2}(0,0) + \frac{\xi d}{\beta b} \mathbf{F}_{2}^{\prime}(0,0) - \frac{i\mu}{\beta} \int_{-\infty}^{0} \mathbf{F}_{2}\left(\frac{\theta d}{b}, 0\right) \exp\left(\frac{i\mu\beta}{\beta}\right) d\theta - \frac{i\mu^{2} \xi}{\beta} \int_{-\infty}^{0} \mathbf{F}_{2}\left(\frac{\theta d}{b}, 0\right) \exp\left(\frac{i\mu\beta}{\beta}\right) d\theta - \frac{\mu^{2} \xi}{\beta} \int_{-\infty}^{0} \mathbf{F}_{2}\left(\frac{\theta d}{b}, 0\right) \exp\left(\frac{i\mu\beta}{\beta}\right) d\theta \right].$$
(4.85)

whers

- Ward and the state of the sta

$$\mathbf{F}_{2}^{\prime}(0,0) = \left[\frac{\partial \mathbf{F}_{2}(\kappa,\lambda)}{\partial \kappa}\right]_{\kappa=\lambda=0}$$

If also  $\mu$  is small, Equation (4.85) may be written as

$$\frac{\overline{w}_{1}(\xi,0)}{\overline{u}} = \frac{S\overline{C}_{L}}{bh} \left\{ \delta_{0} + \frac{x}{h} \delta_{1} + i \frac{\omega h}{\overline{u}} \left( \delta_{0}' + \frac{x}{h} \delta_{1}' \right) + 0 \left( \frac{x^{2}}{h^{2}} \right) \right\}, \quad (4.86)$$

where

$$\delta_{0} = \frac{h}{2^{h}} F_{2}(0,0)$$

$$\delta_{1} = \frac{h^{2}}{2\beta b^{2}} F_{2}'(0,0)$$

$$\delta_{0}' = -\frac{1}{2\beta} \int_{-\infty}^{0} F_{2}(\kappa,0) d\kappa$$

$$\delta_{1}' = -\delta_{0}$$
(4.87)

By Equations (4.74), (4.75), (4.76) and (4.79), and by virtue of the fact that

and the second second

$$\pi \operatorname{cosech} \pi y = \frac{1}{y} + 2 \sum_{n=1}^{\infty} (-1)^n \frac{y}{y^2 + n^2}$$
 (4.88)

it follows that
$\mathbf{F}_{2}(0,0) = \frac{b^{2}}{4\pi} \sum_{m=-\infty}^{\infty} \sum_{n=+\infty}^{\infty} (-1)^{n} \frac{m^{2}b^{2} - n^{2}h^{2}}{(m^{2}b^{2} + n^{2}h^{2})^{2}} , \quad (m,n \neq 0,0) , \quad (4.89)$ 

so that we recover the usual expression for steady interference (cf.  $\delta_0^{(1)}$  in Section 3.2.2). A more straightforward method of tabulating  $\delta_0$ ,  $\delta_1$ ,  $\delta_0'$ ,  $\delta_1'$  is described in Reference 4.23, which includes a table of their values. They are plotted in Figure 4.4.

Reference 4.23 also gives, for incompressible flow, the interference upwash for a small wing in a closed circular tunnel of radius R; by Equation (4.56) the compressibility factors are the same as in Equations (4.87), and hence

$$\overline{u}_{1} = \frac{S\overline{C}_{L}}{\pi R^{2}} \left[ \delta_{0} + \frac{x}{2R} \delta_{1} + \frac{i\omega R}{U} \left( \delta_{0}' + \frac{x}{2R} \delta_{1}' \right) \right] , \qquad (4.90)$$

where

$$\begin{aligned} \delta_0 &= 0.125 , & \delta_1 &= 0.25 \ \beta^{-1} \\ \delta_0' &= -0.07 \ \beta^{-1} , & \delta_1' &= -0.25 \end{aligned}$$
 (4.91)

Goodman<sup>4.15</sup> has given an analysis by a method basically that of Section 4.4.2 for the small wing in the circular tunnel, but without assuming that x and  $\omega R/U$  are small. The results are presented in the form of a chart which, apart from a number of misprints, appears to be consistent with Equations (4.91).

For a tunnel of any cross-section the quantities  $\delta_0$ ,  $\delta_1$ ,  $\delta_0'$  and  $\delta_1'$  may be found from Equation (4.56) provided  $\delta(\xi, \eta; \sigma, 0)$  is known.

An alternative approach, valid when M = 0, has been given by W.P.Jones (Ref. 4.24; 1943). His method may be illustrated by referring to Equation (4.71) which, for M = 0, leads to the following expression for the interference upwash

$$\frac{\overline{w}_{1}}{U} = -\frac{1}{8\pi} \iint_{S} \overline{l}(x', y') \int_{-\infty}^{x-x'} \exp\left(-\frac{i\omega(x-x')}{U}\right) \exp\left(\frac{i\omega\xi}{U}\right) \times \\ \times \sum_{B=-\infty}^{\infty} \sum_{n=-\infty}^{\infty} (-1)^{2} \frac{(2n^{2}h^{2} - (y' + ab)^{2} - \xi^{2})}{[\xi^{2} + (y' + ab)^{2} + n^{2}h^{2}]^{5/2}} d\xi dx' dy', \quad (a, n \neq 0, 0) .$$
(4.92)

where it is assumed that  $w_i/U$  is independent of y. When it is also assumed that the wing is small, Equation (4.92) leads to

 $\frac{\overline{u}_{i}}{\overline{u}} = -\frac{1}{8\pi} \iint_{S} \widetilde{l}(x',y') dx'dy' \left\{ \left(1 - \frac{i\omega x}{\overline{u}}\right) \int_{-\infty}^{0} exp\left(\frac{i\omega \xi}{\overline{u}}\right) \sum_{\underline{u}=-\infty}^{\infty} \int_{1}^{\infty} (-1)^{n} \frac{(2n^{2}h^{2} - \underline{u}^{2}b^{2} - \xi^{2})}{[\xi^{2} + \underline{u}^{2}b^{2} + n^{2}h^{2}]^{5/2}} d\xi + x \sum_{\underline{u}=-\infty}^{\infty} \sum_{\underline{u}=-\infty}^{\infty} (-1)^{n} \frac{2n^{2}h^{2} - \underline{u}^{2}b^{2}}{(\underline{u}^{2}b^{2} + n^{2}h^{2})^{5/2}} + o(x) \right\} \qquad (\underline{u}, n \neq 0, 0) . \quad (4.93)$ 

The second summation is that encountered as  $\delta_1^{(1)}$  in the treatment of steady interference given in Section 3.2.2. The integrals in Equation (4.93) are studied in Reference 4.24 and reduced to tractable expressions.

#### 4.5.4 Slender Wings

If the model is very slender, it may be regarded as made up of a number of small wings arranged in the streamwise direction. Consider a slender wing oscillating with small frequency parameter in a closed rectangular tunnel. It follows from Equation (4.86) that the interference upwash is given by<sup>4.25</sup>

$$\frac{\overline{w}_{i}}{\overline{u}}(\mathbf{x},0) = \frac{1}{bh} \iint_{S} \overline{l}(\xi,\mathbf{y}) \left[ \delta_{0} + \frac{\mathbf{x} - \xi}{h} \delta_{1} + \frac{i\omega h}{\overline{u}} \left( \delta_{0}' + \frac{\mathbf{x} - \xi}{h} \delta_{1}' \right) \right] d\xi d\mathbf{y}$$

$$= \frac{S}{bh} \left\{ \overline{C}_{L} \left[ \delta_{0} + \frac{\mathbf{x}}{h} \delta_{1} + \frac{i\omega h}{\overline{u}} \left( \delta_{0}' + \frac{\mathbf{x}}{h} \delta_{1}' \right) \right] + \frac{\overline{C}}{h} \overline{C}_{M} \left[ \delta_{1} + \frac{i\omega h}{\overline{u}} \delta_{1}' \right] \right\} , \quad (4.94)$$

where  $\delta_0$ ,  $\delta_1$ ,  $\delta_0'$  and  $\delta_1'$  are given in Equations (4.87), and  $\overline{C}_{\mathbf{n}}$  refers to the nose-up pitching moment about the axis  $\mathbf{x} = 0$ . A similar formula may be constructed for any tunnel for which  $\delta_0$ ,  $\delta_1$ ,  $\delta_0'$  and  $\delta_1'$  are known.

The interference upwash may be used to calculate the overall corrections to forces and moments as for general wings, but since the wing is slender the aerodynamics are considerably simplified, especially because  $\overline{w}_1$  may be regarded as independent of y. From the formula of unsteady slender-wing theory<sup>8.26</sup> the correction to lift per unit length in the streamwise direction corresponding to  $\overline{w}_1$  is

$$\Delta I_{i}(\mathbf{x}) = -\pi \rho \mathbf{U}^{2} \left\{ \mathbf{s}(\mathbf{x})^{2} \left[ \frac{\partial}{\partial \mathbf{x}} + \frac{i\omega}{\mathbf{u}} \right] \frac{\mathbf{\overline{\pi}}_{i}}{\mathbf{u}} + 2\mathbf{s}(\mathbf{x}) \frac{\mathbf{I}\mathbf{s}(\mathbf{x})}{\mathbf{I}} \frac{\mathbf{\overline{\pi}}_{j}}{\mathbf{u}} \right\} .$$
(4.95)

where s(x) is the local span.

An example of the application of Equations (4.94) and (4.95) may be found in Reference 4.25.

#### 4.5.5 Numerical Examples

200

Only a small amount of numerical data on the numerical corrections to threedimensional wings has been published, and this is entirely confined to incompressible flow. The reason is obviously the fact that heavy computation is involved unless drastic simplifying assumptions are mide. The earliest attempt on the problem seems to be that by W.P.Jones (Ref.4.24; 1943) described in Section 4.5.3. It is assumed that the wing is small although the frequency is feft general. Some numerical values are given for the corrected and uncorrected pitching derivatives of rectangular wings in a tunnel of square pross-section.

An example of calculation for which the wing was not assumed to be small is given in Reference 4.23. A model with the swept planform shown in Figure 4.5 was assumed to be performing low-frequency pitching oscillations in the closed circular tunnel shown there. Two pitching axes were considered, as shown in Figure 4.5. The theoretical pitching derivatives, as determined by the method of Reference 4.19, are given in the following table.

x <sub>o</sub> /c <sub>r</sub>	l <sub>a</sub>	l:	-≣ <sub>α</sub>	- <b>Ξ</b> .α
0.613	0.852	0.892	0.055	0.301
0.738	0.852	0.738	0.099	0.159

It was assumed that the parts of the circulation in phase and in quadrature with the incidence could be regarded as concentrated in horse-shoe vortices with their finite parts swept through 60° and semi-span equal to 0.35 of the tunnel radius. The position and strength of these vortices are then easily calculated from the pitching derivatives given above, and are plotted in Figure 4.5. Since the interference for steady horse-shoe vortices of this planform is tabulated in Reference 4.22, it is a straightforward matter to calculate the required oscillatory interference upwash distribution by Equation (4.56). Since the interference forces were also calculated by the method of Reference 4.19 with 2 chordwise and 5 speawise stations, these upwashes were in fact required at only six points on the half wing. The overall interference increments to the derivatives for the two pitching axes are given below.

<sup>1</sup> 0 <sup>/C</sup> r	$(l_{\alpha}) \qquad \delta(l_{\alpha})$	$\sum_{\alpha} \delta(-\pi_{\alpha})$	δ(-e <sub>α</sub> )
0.613 0	.051 0.00	9.968	0.001
0.738 C	051 -0.00	95 -0.701	0.009

It is interesting to compare these values with those cotained by the "ssail-wing" method of Section 4.5.3, which gives

254

このであるないないとういう

x_/c <sub>r</sub>	δ(l <sub>α</sub> )	δ(l;)	δ(-=)	δ(-==_α)
0.613	0.051	0. 013 <sub>5</sub>	0.009	0.093
0.738	0.051	0. 034	-C.001	0.061

The differences between the values computed by the two methods are negligible except for  $\delta l_{\alpha}$  when it is of order 0.01. Even this is only marginally significant compared with the accuracy of present experimental techniques. The inference is that, for wings up to the size shown in Figure 4.5, the small-wing approximation is probably adequate, when the frequency parameter is small.

The small-wing theories of Jones<sup>4,24</sup> and Goodman<sup>4,18</sup> both predict that the interference effects decrease rapidly to zero when the frequency parameter becomes large, but nevertheless remain significant over much of the practical range. This effect may be illustrated by Pigures 4.6(a) and (b) which show respectively the in-phase and quadrature components of interference upwash in a closed 9 x 7 rectangular tunnel due to a horse-shoe vortex with 2s/b = 0.4. Note how, for positive  $\xi$ , the in-phase part decreases as  $\mu$  increases, while the part in quadrature at first increases but eventually starts to decrease.

The effect of model size and frequency may be illustrated by some calculations from Reference 4.20. A cropped delta wing of aspect ratio 2s/c = 1.8 and taper ratio.  $c_t/c_r = 1/7$  is assumed to be performing pitching oscillations about an axis through its apex in a closed rectangular  $9 \times 7$  tunnel. The load distributions are calculated by the lifting-surface theories of References 4.19 and 4.27 for the low and finite frequency cases respectively, and the interference upwach is obtained from Equation (4.31) with the lift replaced by the circulation as suggested after Equation (4.64). The results obtained may be summarized as in the following table in the notation of Section 4.4.3.

<u>ωb</u> IJ	25 b		∆c̄ ī
→ 0	0.25	$0.006 - i\mu \ 0.003$	$0.001 - i\mu \ 0.001$
-0	0.50	$0.029 - i\mu \ 0.027$	$0.006 - 1\mu 0.008$
<b>→ 0</b>	0.75	0.076 - iµ 0.103	$0.020 - i\mu 0.025$
1.6	050	$0.000 - i\mu 0.008$	-0.001 - 1µ 0.002

These corrections are given as a correction to  $\alpha_0$  and a residual correction to  $\overline{C_m}$ . The complex value of  $\overline{C_L}$  would be determined from wind-tunnel measurements. We may note the rapid increase in the corrections as 2s/b increases, and that the correction to incidence involves a change of phase as well as magnitude. Indeed it is obvious that for 2s/b = 0.75 the model would be too large for measurements to be satisfactory. The effect of taking the frequency to be no longer shall is a large

255

「あるいたいとう

-

reduction in the corrections, as might be expected from the results of Reference 4.10 and Figure 4.6. Of course these remarks apply only to this particular configuration, and it would be resh to generalize them.

#### 4.8 PERFORATED AND SLOTTED WALLS

The use of ventilated tunnels is essential, if a fixed wall geometry is to be preserved for measurements in transonic flow. The usual way of achieving this is to use either a perforated tunnel, having a large number of small holes in one or more of its sides, or a slotted tunnel, usually with the slots parallel to the direction of flow.

In practice, tunnels with perforated walls are the less common, but most of the available cscillatory theories are in fact concerned with them and relate to the boundary condition in Equation (4.7). It is assumed that the perturbation velocities are small, so that Equation (4.7) is taken in its linearized form

$$\frac{\partial \overline{\Phi}}{\partial n} + P \left[ \frac{\partial \overline{\Phi}}{\partial x} \div \frac{i\omega}{u} \overline{\Phi} \right] = 0 . \qquad (4.96)$$

For perforated tunnels to be amenable to these theories, it is necessary to find a constant value of P to use in Equation (4.96), which then becomes a simple homogeneous boundary condition. The parameter P for steady flow in perforated tunnels is discussed in Section 6.7. It is not known whether the parameter should remain unchanged for oscillatory flow; possibly P should then be complex, and results obtained using Equation (4.96) should not be trusted if the frequency is large.

The two-dimensional problem of a slowly oscillating aerofoil symmetrically situated between perforated walls has been investigated by Drake<sup>4.28,4.29</sup>, using the boundary condition Equation (4.96). He derives an integral equation expressing the known upwash distribution on the aerofoil as a weighted integral of the lift distribution  $\overline{l}$ , and obtains solutions by expanding  $\overline{l}$  in a series similar to that in Equation (4.34) with coefficients assumed to be linear functions of frequency. Unfortunately the expressions for the derivatives given in Reference 4.29 are incorrect.

There is an alternative treatment due to Rosenblat<sup>4,30</sup>, which is restricted to low-frequency oscillations in incompressible flow. The derivatives are given as complicated analytical expressions and have not been evaluated,

Drake<sup>4.31</sup> has also treated some transmic flow problems involving perforated tunnels by solving Equation (4.9) for the following three cases

- (i) two-dimensional wing midway between two perforated walls,
- (ii) slender wing midway between infinite perforated roof and floor,
- (iii) slender wing in a rectangular tunnel with solid side-walls and perforated roof and floor.

256

÷

In all these cases mathematical difficulties require that the frequency parameter is small, and in (ii) and (iii) that the porceity parameter is small. The use of Equation (4.9) also imposes the restriction<sup>4,32</sup> that the thickness has to be very small. Drake's formulae for the derivatives are too lengthy to be given here, but are for the most part fairly straightforward to evaluate numerically.

ş

STATE CONTRACTOR

The homogeneous boundary condition for steady flow in slotted tunnels is given in Chapter VI as Equation (6.7). The corresponding condition for escillatory flow is not known with certainty, but from the analysis of Reference 4.40 it seems plausible to assume that it should be

$$\left(\frac{\partial}{\partial x} + \frac{1\omega}{\upsilon}\right) \left(\frac{\overline{\Phi}}{\overline{\Phi}} + K \frac{\partial \overline{\Phi}}{\partial n}\right) + \frac{1}{P} \frac{\partial \overline{\Phi}}{\partial n} = 0 , \qquad (4.97)$$

where, for rectangular tunnels with solid side-walls and the roof and floor each having N equally spaced slots of width a,

$$K = \frac{b}{\pi N} \log_e \csc \frac{\pi N a}{2b} . \qquad (4.93)$$

Equation (4.97) is perhaps harder to apply than the boundary condition for the perforated tunnel in Equation (4.96), while its validity is possibly even more dubious if the frequency is large.

It is now necessary to enquire if the steady and unsteady interference flow fields in ventilated tunnels are related by the simple Equation (4.58). This depends on whether or not the unsteady potential  $\overline{\Phi}$ , defined by Equation (4.48) or (4.52), satisfies the unsteady boundary condition when  $\Phi_{\rm V}$  satisfies the steady boundary condition. It may be deduced that this is not true for tunnels with boundary condition (4.96) or (4.97), because these both contain additional terms in  $\omega$  when the flow is oscillatory. Nevertheless, when viscosity is negligible so that  $P \to \infty$ , Equation (4.97) becomes

$$\overline{\Phi} + \mathbf{K} \frac{\partial \overline{\Phi}}{\partial n} = \mathbf{v} ,$$

and it follows that the oscillatory and steady interference fields do satisfy Equation (4.56). Alternatively, if the boundary of a longitudinally slotted tunnel may be regarded as a finite number of solid and open portions on which the boundary conditions are  $\partial \bar{\Phi} / \partial n = 0$  and  $\bar{\Phi} = 0$  respectively, then the argument following Equation (4.48) remains valid and Equation (4.56) is applicable.

There is experimental evidence that the interference in slotted-wall tunnels may be very large for some unsteady experiments. References 4.33 and 4.34 describe some measurements of pitching derivatives made in various rectangular tunnels with longitudinally slotted roof and floor. As shown in Figure 4.7, it has been found that by varying the number of slots very large changes are caused in the pitching damping derivative of a half-model M-wing in the NPL 25 in. by 20 in. Wind Tunnel. The parameter T is defined by

257

in the second second

$$t = \frac{h - K}{h + K}, \qquad (4.99)$$

so that T = -1 corresponds to a completely closed tunnel and T = +1 to a tunnel with solid mide-walls and open roof and floor. Remarkably dimilar results have been obtained for an unampt tapered half-model in the rame tunned; in both cases the presence of slots reduces  $-m_{cl}$  by the order of 25% at H = 0.60 to 40% or more at high subsonic Mach numbers. These large effects are not characteristic of all slotted tunnels and apparently do not occur on oscillating half-models in the NPL 35 in. by 14 in. Wind Tunnel. Nevertheless unsteady interference effects in ventilated tunnels pose an important problem. Unfortunately no satisfactory theoretical solutions for slotted tunnels are set available.

Garner and Moore<sup>4.41</sup> have found some explanation of the large interference effects in Figure 4.7 by considering the limiting ease  $T = \pm 1$ . The small-wing theory of Section 4.5.3. In particular Equation (4.86), way in principle be applied to any tunnel for which  $\delta_0$ ,  $\delta_1$ ,  $\delta_0^2$  and  $\delta_1^2$  are known. A half model in the NFL 25 in. by 20 in. Tunnel corresponds closely to a complete model in a duplex tunnel (b = 2h); for such a tunnel with solid side-walls and open floor and roof it is calculated that  $\delta_0^2 = \pm 0.159 \ \beta^{-1}$  as compared with  $\delta_0^2 = -0.004 \ \beta^{-1}$  for a completely closed tunnel of the same cross-section. The term in  $\delta_0^2$  in Equation (4.86) is particularly important, as it is inversely proportional to a linear dimension of the tunnel while the next terms are inversely proportional to tunnel area. It is now possible to make a theoratical estimate of the difference between  $-\pi_{\alpha}^2$  for T = -1 and for  $T = \pm 1$ on the hypothesis that the wing is small. Computations for the H-wing at Mach numbers 0.6 and 0.8 for the pitching axis indicated in Figure 4.7 give respective differences

$$[-\mathbf{u}_{d}]_{q_{m+1}} - [-\mathbf{u}_{d}]_{q_{m-1}} = -0.55$$
 and  $-0.81$ ;

these are consistent with Figure 4.7. The calculated effect of compressibility is slightly greater than the factor  $\beta^{-1}$  in  $\delta'_0$ ; this appears to explain the larger differences observed at N = 0.8 and above. The same conclusions are derived for the unswept tapered wing tested in the same tunnel.

#### 4.7 STEADILY ROTATING MODELS

Suppose that a wing of symmetrical planform has its line of symmetry in the direction of the airflow while rotating with a constant angular velocity  $\omega$  about the line of symmetry. It will be essumed that the vorticity shed at the trailing edge is convected downstream in a direction parallel to the velocity of the undisturbed fluid

Garmer, H.C. et al. The Theory of Interference Effects on Dynamic Measurements in Slotted-Wall Tunnels at Subscrite Speeds and Cosperison with Experiment. NFL Aero. Report 1211, ANC Report 18,339, 1966.

<sup>\*</sup> Reference 4.41 is supermeded by the fully theoretical and experimental study in the following paper:

flow, so that each element of the trailing worter sheet will be a helix with pitch inversely proportional to  $\omega$ . Obviously the theory of the rolling wing may be expected to be closely related to that of the screw propeller<sup>4.35</sup>; in fact the treatment of the circular tunnel given in Section 4.7.1 is an extension by Mandl and Pounder (Ref.4.36; 1951) of Goldstein's theory (Ref.4.37; 1929).

The assumption that the trailing vortices are exactly helical implies that the circulation is small, and therefore that the induced drag is small, and hence, in most cases, that the total drag is also small. By the momentum theory of wall interference on propellers (Ref. 4.3, Chap.XVII), if the thrust coefficient is small and negative, then the correction to the tunnel stream velocity may be neglected even in closed tunnels.

If the rotation is sufficiently slow, the helical nature of the vortex sheet will be negligible and the tunnel interference may be calculated by the theory for steady wings with antisymmetrical loading. However, in order to establish the range of  $\omega$ for which this method would be sufficiently accurate, it is necessary to consider the effect of non-zero angular velocity. Only for a circular tunnel cross-section is such an investigation practicable.

#### 4.7.1 General Theory for Circular Tunnels

The following analysis is essentially that of Reference 4.36\* with different notation and sign conventions. Both the flow in the tunnel and that in the free stream will be assumed to have a velocity potential satisfying Laplace's equation

$$\nabla^{2}\phi = \frac{\partial^{2}\phi}{\partial r^{2}} + \frac{1}{r}\frac{\partial\phi}{\partial r} + \frac{1}{r^{2}}\frac{\partial^{2}\phi}{\partial \theta^{2}} + \frac{\partial^{2}\phi}{\partial x^{2}} = 0 \qquad (4.100)$$

in cylindrical polar co-ordinates  $(z,r,\theta)$  where

$$\left. \begin{array}{c} \mathbf{y} = \mathbf{r}\cos\theta \\ \mathbf{z} = \mathbf{r}\sin\theta \end{array} \right\}$$
 (4.101)

such that the angular rolling velocity  $\omega$  is in the sense of  $\vartheta$  increasing. Far downstream the flow field will have belical symmetry so that  $\tilde{\Psi}$  must satisfy

$$\Phi(\mathbf{r}, \theta + \omega \delta t, \mathbf{x} + \mathbf{v} \delta t) = \Phi(\mathbf{r}, \theta, \mathbf{x})$$

where St is an arbitrary increment of time. Thus

$$\omega \frac{\partial \Phi}{\partial \theta} = -0 \frac{\partial \Phi}{\partial x} . \qquad (4.102)$$

and Equation (2.236) becomes

\* This paper contains several errors of sign.

259

ware a set the structure of the second ware set of the second second second second second second second second

and the state of the

$$\frac{\partial^2 \Phi}{\partial r^2} + \frac{1}{r} \frac{\partial \Phi}{\partial r} + \left(\frac{\omega^2}{v^2} + \frac{1}{r^2}\right) \frac{\partial^2 \Phi}{\partial \theta^2} = 0 \quad . \tag{4.103}$$

The boundary conditions are  $\partial \Phi / \partial r = 0$  on a solid wall,  $\Phi = 0$  at the boundary of an open jet, and for the free stream  $\Phi \to 0$  as  $r \to \infty$ .

Now consider  $\Phi$  at a time when the wing is in the position  $\theta = \pm \frac{1}{2}\pi$ . Let the circulation on the upper half  $(\theta = \frac{1}{2}\pi)$  be  $\Gamma(\mathbf{r})$ . Then the boundary conditions to be satisfied by  $\Phi$  on the line  $\theta = \pm \frac{1}{2}\pi$  are shown in Figure 4.8. Let the wing be a lifting line in the plane  $\mathbf{x} = 0$ . By considerations snalogous to those used in steady flow it follows that at the wing the normal velocity induced by the vortex system is

$$\frac{1}{2} \left( \frac{1}{r} \frac{\partial \Phi}{\partial \theta} \right)_{\theta = \frac{1}{2} \pi}$$

where  $\Phi$  has its value in the helical wake far downstream. Thus the problem of finding the interference velocity is reduced to solving Equation (4.163) subject to the boundary conditions in Pigure 4.8. The first step in obtaining a solution is the removal of the discontinuities across the line  $\theta = \pm \frac{1}{2}\pi$ . From the theory of Fourier series it follows that

$$\frac{\Gamma(\mathbf{r})}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^n}{n} \sin 2n\theta$$

has the required discontinuities, although it will not satisfy either Equation (4.103) or the other boundary conditions.

It is now convenient to change to the non-dimensional variables

$$\vec{r} = \omega r/U$$

$$\vec{s} = \omega s/U$$

$$\vec{R} = \omega R/U$$
(4.104)

Then Equation (4.103) becomes

$$\frac{\partial^2 \Phi}{\partial \overline{r}^2} + \frac{1}{\overline{r}} \frac{\partial \Phi}{\partial \overline{r}} + \left(1 + \frac{1}{\overline{r}^2}\right) \frac{\partial^2 \Phi}{\partial \theta^2} = 0 \quad .$$

Now put

$$\Phi = \frac{\overline{\Gamma(\overline{r})}}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^n}{n} \sin 2n\theta + \sum_{p=1}^{\infty} \Phi_n(\overline{r}) \sin 2n\theta \quad . \tag{4.105}$$

and it is found that the functions  $\tilde{\mathfrak{L}}_{n}(\overline{r})$  must satisfy

\* The first summation in Equation (4.105) has a linear variation between discontinuities and therefore contributes nothing to  $\partial^2 \Phi/\partial \theta^2$ .

$$\frac{\mathrm{d}^2 \bar{\Phi}_{\mathrm{B}}}{\mathrm{d}\bar{r}^2} + \frac{1}{\bar{r}} \frac{\mathrm{d}\bar{\Phi}_{\mathrm{B}}}{\mathrm{d}\bar{r}} - 4n^2 \left(1 + \frac{1}{\bar{r}^2}\right) \bar{\Phi}_{\mathrm{B}} + \frac{(-1)^{\mathrm{B}}}{107} \left(\frac{\mathrm{d}^2 \Gamma}{\mathrm{d}\bar{r}^2} + \frac{1}{\bar{r}} \frac{\mathrm{d}\bar{\Gamma}}{\mathrm{d}\bar{r}}\right) = 0 \quad . \tag{4.106}$$

If we put

$$\bar{\Psi}_{n}(\bar{r}) = \Phi_{n}(\bar{r}) + \frac{(-1)^{n}}{n\pi} \Gamma(\bar{r})$$
, (4.107)

then

$$\frac{d\tilde{r}^{2}\Psi_{n}}{d\tilde{r}^{2}} + \frac{1}{\tilde{r}}\frac{d\Psi_{n}}{d\tilde{r}} - 4n^{2}\left(1 + \frac{1}{\tilde{r}^{2}}\right)\Psi_{n} = -\frac{4n}{\pi}(-1)^{n}\left(1 + \frac{1}{\tilde{r}^{2}}\right)\Gamma(\tilde{r}) \quad . \tag{4.108}$$

This last step removes any difficulty associated with infinities in the derivatives of  $\Gamma$  at the wing tips.

When its right-hand side is zero, Equation (4.108) has the two fundamental solutions

 $\Psi_n = I_{2n}(2n\overline{r})$  and  $\Psi_n = K_{2n}(2n\overline{r})$ 

in the usual notation for Bessel functions of issginary argument. The solution for arbitrary  $\overline{\Gamma}(\overline{r})$  may be obtained by the method of "variation of parameters". It follows eventually that for an open-jet tunnel the increment to  $\Phi$  due to the tunnel boundaries is

$$\Phi_{1} = -\frac{4}{\pi} \sum_{\bar{R}=1}^{\infty} (-1)^{\bar{n}} n \sin 2n\theta \frac{\bar{K}_{2n}(2n\bar{R})}{I_{2n}(2n\bar{R})} I_{2n}(2n\bar{r}) \int_{0}^{\bar{n}} \frac{1+\xi^{2}}{\xi} I_{2n}(2n\xi) \Gamma(\xi) d\xi , \quad (4.109)$$

while for a closed tunnel the factor

$$\frac{K_{2n}(2n\bar{R})}{I_{2n}(2n\bar{R})} \text{ has to be replaced by } \frac{K_{2n}'(2n\bar{R})}{I_{2n}'(2n\bar{R})}$$

The interference velocity at the wing is thus

$$\mathbf{w}_{\mathbf{i}} = \frac{1}{2\pi} \left( \frac{\partial \Phi_{\mathbf{i}}}{\partial \theta} \right)_{\theta = \frac{1}{4\pi}} . \qquad (4.110)$$

This result only applies to an unswept wing of small chord. There appears to have been no development of a theory capable of dealing with rolling wings of significant streamwise extent.

It is now necessary to assess the effect of the interference velocity on the rolling moment. Mandl and Pounder<sup>5.36</sup> use simple strip theory as applied to wings of fairly high aspect ratio. Thus an element dr of the wing at a distance r from the tunnel axis is regarded as being at an incidence  $-\tan^{-1}(\omega r/\theta)$  to a free stream of speed  $\sqrt{(0^2 + \omega^2 r^2)}$ . The circumferential component of the interference velocity opposes  $\omega r$ , so that the effective angle of incidence

Sector devices and an and

and the second second

282

$$\alpha_{\rm g} = -\tan^{-1}\left(\frac{\omega r - w_{\rm i}}{U}\right) = -\tan^{-1}\left(\frac{\omega r}{U}\right) + \frac{Uw_{\rm i}}{U^2 + \omega^2 r^2} \quad . \tag{4.111}$$

when terms in  $w_1^2$  are neglected. Then the rolling moment coefficient is defined by

$$C_l = (C_l)_p + \delta C_l = \frac{2}{\rho U^2 Ss} \int_0^R r \frac{dL}{dr} dr$$
, (4.112)

where the element of lift dL corresponds to the local incidence  $\alpha_{\rm g}$  .

From the initial assumption that the circulation is  $\Gamma(\mathbf{r})$ , a factor

$$\frac{\text{USsC}_{l}}{2\int_{0}^{8}r\Gamma(r) \, dr}$$

must be applied to the right-band side of Equation (4.112), so that the assumed rolling moment is consistent with the measured  $C_1$ . To the first order in  $w_1$  the interference correction to  $C_1$  is given by

$$\frac{\Delta C_l}{C_l} = -\frac{\delta C_l}{C_l} = -\frac{\delta C_l}{(C_l)_p} \qquad (4.113)$$

Alternatively the correction

$$\Delta \omega = -\frac{\omega(\Delta c_1/c_1)}{1+(\Delta c_1/c_2)}$$
(4.114)

may be applied to the measured angular velocity.

#### 4.7.2 Slowly Rolling Wings

If the angular velocity is small enough, the helical nature of the wake may be neglected; the calculation of the interference then reduces to that for an antisymmetrically loaded wing in steady flow. This topic is treated in Section 3.3.5, and needs no further consideration here.

Evens (Bef.4.38; 1947) has treated the rolling wing in a circular tunnel by another method in which the trailing vorticity is convected downstream along straight lines. He assumes that the symmise distribution of circulation is

<sup>a</sup> Mandl and Pounder apperently include an axial component of interference velocity, which is taken to be  $\frac{1}{2}\partial\Phi_{i}/\partial x$  as given by Equations (4.102) and (4.105). The justification for this is obscure. Moreover, the effective stream velocity would appear to include a contribution of order  $\omega \pi w_{i}/U$ . These considerations do not influence the results for  $\omega = 0$ , but the calculated frequency effect in Beference 4.36 is regarded with suspicion.

263

$$\Gamma(\bar{r}) = \sum_{k=0}^{\infty} \Gamma_{k}(\bar{r}/\bar{s})^{k} , \qquad (4.115)$$

and evaluates  $\delta C_1$  by strip theory for wings of arbitrary taper ratio. At first sight his formula for  $\delta C_1$  is inconsistent with the limiting result from Reference 4.36 as  $\omega \to 0$ . However, Evans has implicitly assumed that the circulation at the wing tip is zero; in Appendix A of Reference 4.36, Mandl and Pounder have shown that the discrepancy disappears provided that

$$\Gamma(\bar{s}) = \sum_{m=0}^{\infty} \Gamma_m = 0$$
 (4.116)

#### 4.7.3 Propellers

Sevie Contraction and the

Rolling wings might be expected to lead on to corrections for airscrews in wind tunuels, but in fast this subject seems to have made relatively little progress since a method of correcting the longitudinal velocity for a model propeller in a closed circular wind tunnel was devised by McKinuon Wood and Harris<sup>4,42</sup> as long ago as 1920. This method appears to be satisfactory. The correction to free-stream velocity is presented in various standard works either graphically (as in Reference 4.43) or in a table (as in Reference 4.44). In practice it is sufficiently accurate to use the approximate formula

$$\frac{\Delta U}{U} = -\frac{\pi D^2}{8C} \frac{\tau}{(1+2\tau)^{\frac{1}{2}}}, \qquad (4.117)$$

where D is the dismeter of the propeller disc, C is the cross-sectional area of the tunnel and  $\tau$  is the thrust coefficient defined by

Thrust = 
$$\frac{1}{4}\pi\rho \bar{v}^2 D^2 \tau$$
. (4.118)

Young (Ref. 4.45; 1944) gives a correction for compressibility in which  $\tau$  is replaced by  $\tau/\beta^2$  in Equation (4.117).

Equation (4.117) is obtained by an extension of the axial momentum theory"."" and is normally used only when au is positive, so that the propeller has a slip-stream of the usual type and the closed tunnel gives a negative correction to free-streem speed. In the terminology of Chapter V it might be said that a positive thrust is equivalent to a negative drag and therefore produces a negative wake-blockage factor. The applicability of Equation (4.117) to other propeller operating conditions depends on how well the assumptions of momentum theory are satisfied. Presumably windmills  $(\tau < 0)$  would be covered.

Alternatively the correction to free-stream velocity may be found by measuring the axial velocity near the tunnel wall"."". This method has the advantage that it is applicable when the propeller is mounted on a body.

Theory points to the conclusion that the interference effect on a propeller in a circular open jet is negligible, and this is confirmed by experiment in Reference 4.44. Some recent empirical work by Berry and Whiting 4.36 on models of marine screw

S. 654

- 53

propellers in water suggests that the interference is much reduced in tunnels having longitudinally slotted walls.

#### 4.8 APPLICATIONS TO COMMON EXPERIMENTAL SITUATIONS

It is generally advisable to estimate the tunnel corrections before the experiments are carried out. The lift or vorticity distribution should be represented as simply as possible. The recommended precedures for oscillatory measurements in incompressible flow are summarized in Section 4.8.1. The availability of methods for compressible flow is indicated in Section 4.8.2. For numerical data or formulae reference should always be made to the original papers.

#### 4.8.1 Incompressible Flow

(i) Two-dimensional closed tunnel

For c/h small, use Equations (4.19) and (4.25).

For c/h not small, estimate l from free-stream theory, then use Equations (4.14), (4.15) and (4.18) to compute  $\overline{w}_i$ .

Corrections for rigid modes may be obtained from Equation (4.35) if  $\nu \ll 1$  and c/h is small, or may be estimated from graphs in Reference 4.9.

#### (ii) Closed rectangular tunnel

In general, represent the vorticity distribution as simply as possible and apply the theory of Section 4.5.1 (Ref.4.20).

A rough estimate of the interference upwash may be obtained from Equation (4.86) and Figure 4.4 by assuming that the wing is small, or from Equation (4.94) for slender wings.

#### (iii) Closed circular tunnel

Apply the theory outlined in Section 4.5.2 (Ref. 4.23).

Rough estimates may be made, as if the wing were small or slender.

(iv) Tunnels of other cross sections

The method of Section 4.4.2 may be applied, provided that the distribution of interference upwash can be estimated for steady incompressible flow.

Simplified formulae for small or slender wings then follow from Equations (4.56) and (4.58).

#### (v) Longitudinally slotted tunnels

------

When viscous effects at the slots are ignored, the slotted tunnel becomes a particular case of (iv).

Unfortunately the steady interference upwash (Chapter VI) has not yet been calculated in sufficient detail to be used in this way.

A PARTY OF THE OWNER OF THE OWNER

#### 4.8.2 Compressible Flow

In all the following applications the experimental frequency should be kept below that for tunnel resonance (Sections 4.3 and 4.4.1).

.

(i) Two-dimensional closed tunnel

In general, the interference may be evaluated by summing over the images the upwashes as given by Equation (4.31) or (4.32).

The chordwise loading  $\overline{l}$  may either be assumed to have its free-stream value, or the integral Equation (4.31) may be solved numerically<sup>4,12</sup>.

If the chord is small and  $\nu$  is small, then Equations (4.35) may be used for rigid modes.

#### (ii) Three-dimensional tunnels

There is no general method for three-dimensional tunnels. If the frequency is low, the method of Section 4.4.2 may be used provided that the steady interference upwash is known.

### (iii) Closed rectangular tunnel

In general, the evaluation of Equation (4.71) is required. For low frequency the theory of Sections 4.4.2 and 4.5.1 may be used.

For small or slender wings and low frequency the theory of Sections 4.5.3 and 4.5.4 may be used.

#### (iv) Closed circular tunnel

For low frequency the theory of Sections 4.4.2 and 4.5.2 may be used.

For small or slender wings and low frequency, use Section 4.5.3 or 4.5.4 and Equations (4.90) and (4.91).

265

•

Per V

A REAL BANK

man have

124

1 -.

New Section of the se

52.43

# REFERENCES

4.1 Miles, J.W.	The Potential Theory of Unsteady Supersonic Flow. Cambridge University Press, 1959.
4.2 Jordan, P.F.	The Harmonically Oscillating Wing with Finite Vortex Trail. ARC R&M 3038, 1953.
4.3 Glauert, H.	The Elements of Aerofoil and Airscrew Theory. Second Edition. Cambridge University Press, 1948.
4.4 Jones, W.P.	Wing Tunnel Interference Effects on Measurements of Aerodynamic Coefficients for Oscillating Aerofoils. ARC B&M 2786, 1950.
4.5 Lilley, G.M.	An Investigation of the Flexure-Torsion Flutter Characteristics of Aerofoils in Cascade. Coll. Aero. (Crenfield) Report 60. ARC Report 14,952, 1952.
4.6 Mendelson, A. Carroll, R.W.	Lift and Moment Equations for Oscillating Airfoils in an Infinite Unstaggered Cascade. NACA Tech. Note 3263, 1954.
4.7 Rosenblat, S.	The Aerodynamic Forces on an Aerofoil in Non-Uniform Unsteady Motion in a Closed Tunnel. Phil. Trans. Roy. Soc. A, Vol.250, 1957, pp.247-278.
4.8 Timen, R.	The Aerodynamic Forces on an Oscillating Aerofoil between Two Parallel Walls. App. sci. Res. A (Hague), Vol.3, 1951, pp.31-57.
4.9 de Jager, E.M.	The Aerodynamic Forces and Moments on an Oscillating Aerofoil with Control-Surface beisven Two Parallel Walls. NLL (Netherlands) Report F.140, 1953.
4.10 Jones, W.P.	Wind-Tunnel Wall Interference Effects on Oscillating Aerofoils in Subsonic Flow. ARC R&M 2943, 1953.
4.11 Runyen, H.L. Watkins, C.E.	Considerations on the Effect of Wind-Tunnel Walls on Oscillating Air Forces for Two-Dimensional Subsonic Compressible Flow. NACA Report 1150, 1953.
4.12 Runyan, H.L. et al.	Theoretical and Experimental Investigation of the Effect of Tunnel Walls on the Forces on an Oscillating Airfoil in Two-Dimensional Subsonic Compressible Flow. NACA Report 1262, 1956.
4.13 Acum. W.E.A.	Theory of Lifting Surfaces Oscillating at General Frequencies in a Stream of High Subsonic Mach Number. ARC Report 17,824, 1955. (To be ARC R&N.)

A STREET, STRE

4.14 Richau	rdson, J.R.	A Method for Calculating the Lifting Forces on Wings (Unsteady Subsonic and Supersonic Lifting-Surface Theory). ARC R&M 3157, 1955.
4.15 Watkin et al.	os, C.E.	A Systematic Kernel Function Procedure for Determining Aerodynamic Forces on Oscillating or Steady Finite Hings at Subsonic Speeds. NASA Tech. Report R-48, 1959.
4.16 Acum,	Ħ.E.A.	A Simplified Approach to the Phenomenon of Wind-Tunnel Resonance. APC R&N 3371, 1962.
4.17 Widma et al.	yer, E.	Some Measurements of Aerodynamic Forces and Moments at Subsonic Speeds on a Rectangular Wing of Aspect Ratio 2 Oscillating about the Midchord. NACA Tech. Note 4249, 1958.
4.18 Goodm	an, T.R.	The Upwash Correction for an Oscillating Wing in a Wind Tunnel. J. mero. Sci., Vol.29, 1953, pp.383-386, 406.
4.19 Garner	r, H.C.	Multhopp's Subsonic Lifting-Surface Theory of Wings in Slow Pitching Oscillations. ARC R&M 2885, 1952.
4.20 Acum,	W.E.A.	Wall Corrections for Wings Oscillating in Wind Tunnels of Closed Rectangular Section. ARC R&M 3312, 1958.
4.21 von Ka Burger	ármán, T. rs, J.M.	General Aerodynamic Theory Perfect Fluids. Airfoils and Airfoil Systems of Finit' Span. "Aerodynamic Theory" (Ed. W.F.Durand), Vol.II, Div.E. Chapter IV, Sect.42 to 44. Julius Springer, Berlin, 1935.
4.22 Eisen:	stadt, B.J.	Boundary-Induced Upwash for Yawed and Swept-Back Wings in Closed Circular Wind Tunnels. NACA Tech. Note 1265, 1947.
4.23 Acum, Garnes	W.E.A. r, H.C.	Approximate Wall Corrections for an Oscillating Swept Wing in a Wind Tunnel of Closed Circular Section. ARC CP 184, 1954.
4.24 Jones,	, <b>W.P.</b>	Wind Tunnel Interference Effect on the Values of Experi- mentally Determined Derivative Coefficients for Oscillating Aerofoils. ARC R&M 1912, 1943.
4.25 Acum,	W.E.A.	A Note on the Estimation of the Effect of Wind Tunnel Walls on the Forces on Slowly Oscillating Slender Wings. ARC CP 707, 1963.
4.26 Garrie	ck, I.E.	Some Research on High-Speed Flutter. Anglo-American Aeronautical Conference, Brighton, 1951, pp.419-446. (Royal Aero. Soc.)

Barris and the state of the second second

267

1278**4** 

An II SUNGAN

- 「「「「「「「「「」」」」

\* \*\* \*\*

........

. .

. . . . . . . . . . .

Adding a strange and a strange

前日

----



268	
4.27 Lehrien, Doris, E.	Calculation of Stability Derivatives for Oscillating Wings. ARC R&M 2922, 1953.
4.28 Draže, D.G.	The Oscillating Two-Dimensional Aerofoil between Porous Walls. Aero. Quart., Vol.3, 1957, pp.220-239.
4.29 Drake, D.G.	Quasi-Steady Derivatives for the Subsonic Flow past an Oscillating Aerofoil in a Porous Wind Tunnel. Aero. Quart., Vol.10. 1959, pp.211-229.
4.30 Roseablat, S.	The Aerodynamic Forces on an Aerofoil in Unsteady Motion between Porous Walis. Quart. J. Mech. appl. Math., Vol.12, 1959, pp.151-174.
4.31 Draže, D.G.	Wind-Tanuel Interference for Oscillating Wings at Transonic Speeds. ARC Report 21,489, 1959 (unpublished).
4.32 Landahl, M.T.	Unsteady Transonic Flow. Forgazon Press, 1961.
4.33 Acus, R.E.A. Garner, H.C.	The Estimation of Oscillatory Fing and Control Derivatives. AGARD Report 340, 1961.
4.34 Wight, X.C.	A Review of Slotted-Wall Wind Tunnel Interference Effects on Oscillating Models in Subsonic and Transonic Flows. J. Roy. meron. Soc., Vol.68, 1964, pp.670-674.
4.35 Theodorsen, Th.	Theory of Propellers. McGraw-Hill, 1948.
4.36 Mandl, P. Pounder, J.R.	Wind Tunnel Interference on Rolling Moment of a Rotating Wing. NRC (Canada) Aeronautical Report AR-10, 1951.
4.37 Goldstein, S.	On the Vortez Theory of Screw Propellers. Proc. Roy. Soc., Series A, Vol.123, 1929, pp.449-465.
4.38 Evans, J.M.	Stability Derivatives. Wind Tunnel Interference on the Lateral Derivatives $l_p$ , $l_r$ and $l_v$ with Particular Reference to $l_p$ . Report ACA-33 (Australia), 1947.
4.39 Mglyneux, W.G.	Wind Tunnel Interference in Dynamic Measurements, RAE Tech. Report 64069, ARC Report 26,673, 1964. Presented at a short course on the Use of Flexible Nodels in Aeroelastic Research, Von Kármán Institute of Fluid Dynamics, May 1964.
4.40 Davis, D.D.' Moore, D.	Analytical Study of Blockage- and Lift-Interference Corrections for Slotted Tunnels Obtained by the Sub- stitution of an Equivalent Homogeneous Soundary for the Discrete Slots. NACA RM 153E07b (NACA/TIB/3792), 1953.

ł

٠,

\*\*\*\*\*\*\*\*

۰.

1

Ĺ

And the second sec

۲. ۲.

4.41 SE MO	ore, Á.¥.	Notes on Unsteady Stotted-Wall Junnel Interference. NPL Aero. Note 1039, ARC Report 27,486, 1965.
4.42 %c	od. R.KcK.	Some Notes on the Theory of an Airscrew Working in a
Ha	rris, R.G.	Wind Channel. ARC R & M 662, 1920.
4.43 Ps	akhurst, R.C.	Wind-Tunnel Technique. Pitman, London, 1952 (reprinted
Ho	Ider, D.W.	in 1955).
4.44 Cl	auert, H.	Airplane Propellers. The Experimental Study of Propellers. "Aerodynamic Theory" (Ed. W.F.Durand), Vol.IV, Div.L, Chap.IX, Julius Springer, Berlin, 1935, pp.293-310.
4.45 Yo	ung, A.D.	Note on the Application of the Linear Perturbation Theory to Determine the Effect of Compressibility on the Wind Tunnel Constraint on a Propeller. ARC R&M 2113, 1944.
4.46 Be	erry, L.W.	Channel Speed Correction for No.1 Tunnel. NPL Ship
Wh	iting, D.	Division Tech. Memo. 71, 1964 (unpublished).

BHIDKENIA BRAISTAANNAN

÷

: ::

# ADDITIONAL REFERENCES FOR CHAPTER IV

Drake, D.G.	The Motion of an Oscillating Aerofoil in a Compressible Free Jet. J. Roy. meron. Soc., Vol.60, 1956, pp.621-523.
Meller, A.G. Berton, A.	Étude des Vibrations Aéro-Elastiques des Gouvernes en Subsonique Élevé. Rech. aeron. No.47, September/October 1955, pp.27-35.
Miles, J.W.	The Compressible Flow past an Oscillating Airfoil in a Wind Tunnel. J. mero. Sci., Vol.23, 1956, pp.671-678.
Reissner, E.	Wind Tunnel Corrections for the Two-Dimensional Theory of Oscillatory Airfoils. Cornell Aero. Lab. Report SB-318-S-3, 1947.
Whiteheod, D.S.	The Vibration of Air in a Duct with a Subsonic Mean Flow. Aero. Quart., Vol.12, 1961, pp.34-40.
Noods, L.C.	The Aerodynomic Forces on an Oscillating Aerofoil in a Free Jet. Froc. Roy. Soc., Series A. Vol.229, 1955, pp.235-250.

「明書語る」







Fig.4.2 Tannel corrections for wing pitching about its mid-chord point

271

;-

.





Fig. 4.4 Interference parameters for small wings in closed rectangular wind tunnels

All street of



all and a state of the state of

2.00



AND THE OFFICE AND A STREET

S. Harry Konner and Marine and

Plg. 4. 8(a) Efrect of frequency on the interference upwash due to a horse-shoe vortex in a closed 9 x 7 tunnel

275

• • • •



A ALLER AND A CONTRACT OF A

31. 4 . . . A . .

Ŧ

ŝ

and the second many and the second second

.



Reproduced from Fig.3 of Ref. 434

ł

The substantion the second second

1



Closed

The States of the States and the second

Open

277

and a subset half the real of the



CHAPTER V

ł

and the state of the second second second

a

a sector as the property of the sector of the sector of the sector of

A CALLER A CONTRACTOR DESIGNATION AND AND A CONTRACTOR

Maria Same

言語

STORE STRATE

# BLOCKAGE EFFECTS IN CLOSED OR OPEN TUNNELS

Dy

E. W. E. Rogers

Aerodynamics Division, National Physical Laboratory, Teddington, Middlesex, England

# **BLANK PAGE**

No. Contraction of the second s

~~~~~

# CONTENTS

-

ł

ŧ

Page

| LIST | OF TABLES   | 283 |
|------|---|-----|
| list | OF FIGURES  | 283 |
| NUTA | rion .  | 284 |
| 5.1  | INTRODUCTION                                      | 287 |
| 5.2  | SOLID BLOCKAGE FACTORS IN CLOSED RECTANGULAR      |     |
|      | TUNNELS   | 289 |
|      | 5.2.1 Two-Dimensional Aerofoil                    | 289 |
|      | 5.2.2 Body of Revolution                          | 294 |
|      | 5.2.3 Wings and Wing-Body Combinations '          | 297 |
| 5.3  | SOLID BLOCKAGE FACTORS IN CLOSED NON-RECTANGULAR  | 299 |
|      | TUNNELS   |     |
|      | 5.3.1 Uctagenai Tunneis                           | 299 |
|      | 5.3.2 Circular and Elliptical Tunneis             | 300 |
| 5.4  | WAKE BLOCKAGE FACTORS IN CLOSED TUNNELS           | 384 |
|      | 5.4.1 Two-Dimensional Aerofoil in a Rectangular   |     |
|      | Tunnel  | 384 |
|      | 5.4.2 Three-Dimensional Models in a Rectangular   |     |
|      | Tunnel  | 396 |
|      | 5.4.3 Closed Non-Rectangular Tunnels              | 308 |
|      | 5.4.4 Wake Blockage Gradient                      | 388 |
| 5.5  | BLOCKAGE FACTORS IN OPEN-JET TUNNELS              | 388 |
|      | 5.5.1 Solid Blockage for Models Spanning the      |     |
|      | Tunnel  | 308 |
|      | 5.5.2 Three-Dimensional Solid Blockage            | 389 |
|      | 5.5.3 Wake Blockage Effects                       | 311 |
| 5.6  | ESTINATION OF BLOCKAGE FRON WALL NEASUREMENTS     | 311 |
|      | 5.6.1 Two-Dimensional Aerofoil                    | 312 |
|      | 5.6.2 Three-Dimensional Models                    | 313 |
| 5.7  | BLOCKAGE CORRECTIONS AT VERY HIGH SUBSONIC SPEEDS |     |
|      | IN CLOSED TUNNELS                                 | 315 |
|      | 5.7.1 Modification of Simple Compressibility      |     |
|      | Theory  | 315 |
|      | 5.7.2 Regions of Local Supersonic Flow            | 316 |

. •

| ~    | ~~  |
|------|-----|
| - 27 | R.7 |
| -    |     |
| -    |     |

| 5.8  | USE OF BLOC  | KAGE FACTORS IN CORRECTING MEASURED |     |
|------|--------------|-------------------------------------|-----|
|      | QUANTITIES   |                                     | 317 |
|      | 5.8.1 Corr   | ections to Stream Quantities        | 317 |
|      | 5.8.2 Corr   | ections to Model Quantities         | 319 |
| 5.9  | SUMMARY OF   | PRINCIPAL BLOCKAGE FORMULAE         | 321 |
|      | 5.9.1 Clos   | ed Rectangular Tunnel               | 321 |
|      | 5.9.2 Clos   | ed Non-Rectangular Tunnels          | 322 |
|      | 5.9.3 Open   | -Jet Tunnels                        | 323 |
|      | 5.9.4 Corr   | ections to Stream Quantities        | 323 |
|      | 5.9.5 Corr   | ections to Model Quantities         | 324 |
| BEFE | RENCES       |                                     | 324 |
| ADDI | TIONAL REFER | ENCES                               | 329 |
| TABI | E            |                                     | 330 |
| FIG  | RES          |                                     | 331 |
|      |              |                                     |     |

Page

3.22

2

ĸ

;

## LIST OF TABLES

なけのとうだいとないないとないない。ないない

ł

ł

A Paralle and Martin

283

and the second second second

AL LAGING

| TABLE 5.1 | Functions Giving the Distribution of Solid Blockage |     |
|-----------|---|-----|
|           | along Bodies in Circular Tunnels (Ref. 5.33)        | 330 |

#### LIST OF FIGURES

| Fig. 5.1       | Aerofoil-shape parameter  | 331     |
|----------------|---|---------|
| Fig. 5.2       | Body-shape parameter  | , 331   |
| Fig. 5.3       | Annel-shape parameter for small models  | 332     |
| Fig. 5. 4      | Tunnel-shape parameter from various formulae  | 332     |
| <b>Fig.5.5</b> | Tunnel-shape parameter for large-span models  | 333     |
| Fig. 5. 6      | Effect of model span on T for various closed tunnels  | 333     |
| Pig. 5.7       | Distribution of solid blockage along bodies in closed<br>circular tunnel                                  | 334     |
| Fig. 3. 8      | Approximate values of tunnel-shape parameters for long<br>bodies in closed circular tunnels               | 334     |
| Fig. 5.9       | Blockage factors for semi-infinite bodies of revolution<br>in circular tunnels                            | 335     |
| Fig. 5.10      | Blockage at centre section of wing spanning an elliptical tunnel  | 335     |
| Fig. 5. 11     | Glauert's two-dimensional wake-blockage factor  | 336     |
| Fig. 5. 12     | Simple compressibility factors used in blockage<br>equations  | 336     |
| Fig. 5. 13     | Tunnel-shape parameters for small models in open<br>rectangular tunnels                                   | 337     |
| Fig. 5. 14     | Effect of model span and sweepback on wall-velocity ratio $2_{g}(b/h = \sqrt{2})$                         | 338-339 |
| Fig. 5, 15     | Examples of calculated and experimental Mach number increments at side-wall of 10 ft $\times$ 7 ft tunnel | 349     |

# NOTATION

٢

ç

۰,

|                  | distance between source and sink                                      |
|------------------|---|
| <b>a</b> *       | critical speed of sound   |
| A                | cross-sectional cres of aerofcil                                      |
| A <sub>22</sub>  | maxigum transverse cross-section of model                             |
| b                | breadth of tunnel   |
| c                | chord of aerofoll   |
| C                | cross-sectional area of tunnel  |
| C <sub>D</sub>   | drag coefficient = $D/\frac{1}{2}\rho U^2 S$                          |
| C <sub>DO</sub>  | drag coefficient at zero lift   |
| C <sub>p</sub>   | pressure coefficient = $(p - p_{\infty})/\frac{1}{2}\rho U^2$         |
| đ                | distance of model from floor of tunnel                                |
| D                | dreg of model   |
| Ĩ                | fineness ratio = $l/t$  |
| h                | height of tunnel  |
| R, I             | functions given in Table 5.1  |
| k(X)             | moment of inertia of aerofoil about chordwise position $x = constant$ |
| ž                | moment of inertia about x axis  |
| k <sub>2</sub> = | A/tc  |
| k <sub>3</sub> = | V/t <sup>2</sup> !  |
| l                | length of body  |
| 2                | zajor axis of elliptical tunnel                                       |
| M                | Mach number of undisturbed stream                                     |
| ¥* =             | U/a*  |
| r.               | minor axis of elliptical tunnel                                       |
|                  |   |

14.88 W.

AN A STATISTICS

284

1

-----

-

おちょうざ いうおん

treversity breaking and

SCOL-1

- 5

.

25

÷

: 3

A DESCRIPTION OF THE PARTY OF

.

.

| P                       |   | local static pressure   |
|-------------------------|---|---|
| $\mathfrak{P}_{\infty}$ |   | static pressure of undisturbed stream   |
| P                       |   | strength of doublet   |
| q                       |   | local surface velocity  |
| Q                       |   | strength of source  |
| R                       |   | radius of circular tunnel   |
| R <sub>N</sub>          |   | keynolds number   |
| 8                       |   | distance along body contour (Section 5.2.1)                                     |
| S                       |   | semi-span of wing   |
| S                       |   | reference area of model   |
| t                       |   | maximum thickness of model  |
| T                       |   | shape parameter of tunnel (cf. Figures 5.3 to 5.6)                              |
| u                       |   | x-component of velocity   |
| U                       |   | velocity of undisturbed stream  |
| v                       |   | volume of model   |
| x                       |   | distance downstream (origin at leading edge of merofoil or centre of body axis) |
| 2                       |   | spanwise distance normal to body axis   |
| z                       |   | distance upwards from body axis   |
| Z                       | = | $(\Delta U)/(u-U)_{wall}$   |
| α                       |   | incidence of model  |
| ß                       | = | $(1 - M^2)^{\frac{1}{2}}$   |
| <b>?</b>                |   | ratio of specific heats (taken as 1.400 for air)                                |
| Δ                       |   | prefix denoting increment due to wall correction                                |
| £                       |   | blockage factor = $(\Delta U)/U$  |
| ņ                       |   | Glauert's two-dimensional wake-blockage factor                                  |

. .::

ł



- $\lambda_3$  shape parameter of body (Fig. 5.2)
- Λ angle of sweepback
- $\rho$  density of undisturbed stream
- $\tau$  shape parameter of tunnel (cf. Figures 5.8 and 5.13)

## Subscripts

- B denotes effect of blockage.
- c denotes corrected value.
- denotes equivalent value.
- denotes effect of velocity gradient (buoyancy).

1340 235

The state of the s

- denotes solid blockage.
- denotes wake blockage.

#### BLOCKAGE EFFECTS IN CLOSED OR OPEN TUNNELS

#### E. W. E. Rogers

#### 5.1 INTRODUCTION

In the preceding chapters, the boundary interference arising from the model lift has been discussed in some detail. There exists, however, an additional interference effect associated with the fact that the model and its wake occupy a certain volume within the finite tunnel stream. The streamline pattern about the model is thereby distorted compared with free-air conditions, and this distortion exists even at zero model lift. The magnitude of the streamline displacement will depend, amongst other things, on the relative sizes of the model and tunnel working section, and on their cross-sections; the direction of the distortion (and hence the sign of the ensuing correction) depends on whether the tunnel boundaries are of the fixed-wall or open-jet type.

This interference is said to arise from the "blockage" of the model and its wake, that associated with the model itself being called "solid blockage", and that due to the wake "make blockage". For most purposes it is sufficiently accurate to assume that the two blockage components are independent both of each other and of the model lift. This simplifies the analysis and enables corrections for blockage effects to be evaluated at zero lift. Such a procedure implies that the model is small compared to the tunnel working section and that the lift is not too large; moreover it is assumed that the blockage only influences the longitudinal component of the flow about the model, and this is true only if the model is mounted in the centre of the tunnel.

In the simplest case, the flow about a small model in a tunnel corresponds to that about the same model in free air, but at a corrected velocity  $U + \Delta U_{\rm S}$ , where U is the nominal velocity of the tunnel stream and  $\Delta U_{\rm R}$  is the sum of velocity increments associated with the solid and wake blockages. For streamline flow it is convenient to express the ratio  $\Delta U_{\rm R}/U$  by the blockage factor  $\epsilon_{\rm B}$ , which is then the sum of corresponding solid and wake blockage factors

$$\epsilon_{\rm g} = \epsilon_{\rm g} + \epsilon_{\rm w} \,. \tag{5.1}$$

For bluff bodies this subdivision of blockage effects is not applicable. The principle of correspondence between the observed flow and some free-air flow at a different velocity is valid for a wide range of flow conditions. Sometimes, however, it may be necessary to determine the variation of the blockage factor along the model length, as for example in the case of a long slender body, or perhaps to consider whether the model blockage is affected by changes in the model incidence. These more complex analyses are usually less general in application than those concerned simply with some average correction applied over the whole volume occupied by the model.
The present chapter will be concerned with blockage effects in tunnels with either completely closed or completely open boundaries. Related problems for ventilated tunnels, having a combination of open and closed boundaries, are discussed in Chapter VI.

The walls of a completely closed tunnel impose a constraint on the flow about the model such that the flow near the wall follows that surface. The stream tubes surrounding the model do not take up their free-air shape but are compressed together, causing a corresponding increase in the local fluid velocities. A similar conclusion may be reached by equating the mass flow well upstream of the model to that at the model position. The velocity correction, and hence the solid-blockage factor, will therefore be positive in a closed tunnel.

The conditions at the boundary of an open jet are less easy to define precisely, but it is usual to assume that the jet boundary is one of constant pressure, equal to that of the surrounding motionless air. In practice, the edge of the jet may be difficult to specify because of mixing between the stream and the surrounding fluid and moreover it may be distorted by the presence of the model. Such difficulties are frequently overcome by assuming that the constant pressure condition may always be applied at the undisturbed boundary and that mixing effects can be ignored.

The assumption of a constant-pressure boundary also implies that the perturbation velocities, particularly the streamwise component, are small at the edge of the jet, since Bernouilli's equation may be applied to a boundary streamline extending from well upstream to a position opposite the model. Click the model induces supervelocities at this position, the constant-pressure condition may only be preserved by a reduction in the effective stream velocity approaching the model. Thus for the open-jet tunnel the solid-blockage factor will be negative.

Clauert (Ref. 5. 1; 1933) has pointed out an additional matter which makes the treatment of the open-jet tunnel less precise than its closed counterpart. In the latter case there is usually a considerable length of working section of almost constant cross-section upstream and downstream of the model. A free jet on the other hand frequently issues from a nozile placed at a relatively short distance upstream of the model and flows into a collector not far downstream. The assumption of an infinitely long jet inherent in much of the analysis is therefore violated<sup>5.57</sup> and the validity of the simple theoretical corrections must rest largely on experimental evidence. The boundary conditions appropriate to finite-length open-jet tunnels have been discussed by Katzoff et al. (Ref. 5.48; 1950) and in more general terms by Vandrey (Ref. 5.49; 1942), but up to the present the simple constant-pressure condition has been most widely used in the theory of wall interference.

It should perhaps be pointed out that the reasons for using short working sections are entirely practical. It is, for example, difficult to obtain a stable jet longer than about two jet diameters. In effect this imposes a restriction on the model size, since the ends of the model must be a moderate distance from both nezzle and collector, and the model length is hence unlikely to exceed about 0.3 jet diameters. In turn the small model size implies that the corrections will not be excessively large.

The simple boundary conditions of the open-jet tunnel suggest that the flow behind the model can enlarge to allow for the presence of a wake of reduced velocity. This

is not the case in a closed tunnel of constant cross-section where, to satisfy continuity upstream and downstream of the model, the flow velocity outside the wake must be greater than in the free stream shead of the model. The effect of the wake in the closed tunnel is to increase the effective stream velocity, so as to introduce the wake-blockage factor  $\epsilon_{\rm p}$  at the model position and to impose a longitudinal pressure gradient along the model. The latter budyancy effect is equivalent to a boundary-induced drag force on the model, for which a correction is required. In the open-jet tunnel, the wake-blockage velocity increment is abovent, but there is still a buoyancy correction.

When the flow departs significantly from the classical streamline flow, it becomes necessary to reconsider the mathematical model of the flow. The representation must incorporate the essential features of the separated flow and still remain simple enough to leave a tractable problem when there are closed boundaries. Such a theory for a non-lifting bluff body is developed in Chapter VII. The relatively large blockage factor for a bluff body is confirmed by experiment.

In the following sections methods of estimating the solid-blockage and wake-blockage factors will be discussed for a variety of tunnel cross-sections. The two important methods of determining the blockage factors are from theoretical calculations involving the model and tunnel geometry, tunnel Mach number and model drag (Sections 5.2 to 5.5), and from pressure measurements made at the tunnel wall (Section 5.6).

At high subsonic stream Mach numbers, shock waves may be present in the flow ebout the model, and some change in the form of the blockage factors discussed earlier may be required. In addition modifications may be necessary to the simple linear theory used to estimate the effect of stream compressibility. Matters of this kind are considered in Section 5.7.

The use of the blockage factors in correcting the measured quantities is discussed in Section 5.8, whilst the final Section 5.9 sets out the more important formulae used in the text.

# 5.2 SOLID BLOCKAGE FACTORS IN CLOSED REDTANGULAR TUNNELS

#### <sup>1</sup> 5.2.1 Two-Dimensional Aerofoll

Restaurantion and a start of the start

We will consider here an asroloil of chord c, maximum thickness t, spenning centrally a rectangular tunnel of height h. An early method of calculating the solid blockage of this configuration was given by Lock (Ref. 5.29; 1929: see also Reference 5.1). He represented the aerofoil by an equivalent doublet and the effect of the tunnel walls by an infinite array of dcublet images extending above and below the model and spaced at the tunnel beight h. The additional velocity induced by the images at the model position was fine be calculated and this is the velocity increment due to the solid blockage. The solid-blockage factor  $\epsilon_g$  in incompressible flow is then given by

$$\epsilon_{s} = \frac{\Delta U_{s}}{U} = \frac{\pi^{2}}{12} \left(\frac{c}{h}\right)^{2} \left(\frac{t}{c}\right)^{2} \lambda_{2} . \qquad (5.2)$$

289

. street

「日本のはななないないないない」 こちょうしょう くう

Hartsbeer Dealth .....

where  $\lambda_2$  is a parameter related to the profile thickness such that the model is replaced in the analysis by a cylinder of radius  $t/\lambda_2$ . Lock calculated  $\lambda_2$  for four typical body shapes (Fig.5.1), but a simple formula is obtained in only a few cases, e.g., ellipse<sup>5.1</sup> and Rankine oval<sup>5.2</sup>. An accurate determination of  $\lambda_2$ can be made if the surface pressure distribution about the model is known, so that the local surface velocity q can be found. Then for a symmetrical aerofoil

$$\lambda_2 = \frac{4}{\pi} \int \frac{q}{v} \frac{z(s)}{t^2} ds .$$
 (5.3)

where s is the distance along the body contour from the leading edge and z(s) is the surface ordinate from the chordline at s. The integral is taken along the curved surface from leading to trailing edge. For a combered profile it is sufficient to consider only the thickness distribution of the section.

Allen and Vincenti (Ref. 5.3; 1944) use the blockage factor

$$\Lambda = 4\left(\frac{t}{c}\right)^2 \lambda_2 \qquad (5.4)$$

so that

$$\epsilon_{\rm s} = \frac{\pi^2}{48} \left(\frac{c}{\rm h}\right)^2 \Lambda . \qquad (5.5)$$

Values of  $\Lambda$  for various basic aerofeil profiles will be found in References 5.3 and 5.27, and  $\lambda_2$  may easily be obtained from these sources. As Figure 5.1 shows, values of  $\lambda_2$  for typical families of NACA 6-series aerofoils are close to those for Joukowski-type profiles. The solid blockage of 2n aerofoil in an incompressible stream has also been considered by Toussmint (see Reference 5.54) using a complex potential method. His equation is similar to (5.2) but tends to give somewhat smaller values of  $\epsilon_{\rm g}$ .

In a compressible fluid when the stream Mach number is subsonic and not too near the choking value, the effect of compressibility may be allowed for sufficiently accurately by small-perturbation theory<sup>5.3,5.30,5.33</sup>. The velocity increment at the model position is then  $\beta^{-1}$  times the appropriate increment for an incompressible fluid in a tunnel of height  $\beta$ h; the model thickness and chord are assumed to remain unchanged. Thus Equations (5.2) and (5.5) must be multiplied by a factor  $\beta^{-3}$ . Lock's original equation then becomes

$$\varepsilon_{s} = \frac{\pi^{2}}{12} \left(\frac{c}{b}\right)^{2} \left(\frac{t}{c}\right)^{2} \frac{1}{\beta^{3}} \lambda_{2} . \qquad (5.6)$$

Sometimes however a more direct relationship between the blockage factor and model geometry is convenient. Thom (Ref. 5.4; 1943) replaced the single doublet of Lock's method by a series of sources and sinks distributed along the model chord. For a thin aerofoil the distribution need not be very complex and leads to the relationship

$$\epsilon_{\rm s} = \frac{\pi A}{6\beta^3 {\rm h}^2} = 0.524 \frac{A}{\beta^3 {\rm h}^2},$$
 (5.7)

where A is the cross-sectional area of the profile, related by some numerical factor to the product ct. If this factor is known, Equation (5.7) may be rearranged to a form similar to (5.6). For thin aerofoils the two equations yield similar blockage factors. Von Baranoff (Ref. 5.39; 1940) had earlier derived an expression for the solidblockage factor identical with Equation (5.7).

Alternatively, use may be made of a simple formula developed by Young and Equire (Ref. 5. 5; 1945), which is equivalent to

$$\epsilon_{\rm g} = 0.62 \, \frac{\rm A}{\beta^3 \rm h^2} \, . \tag{5.8}$$

The numerical factor is higher than in Equation (5.7) and Reference 5.5 suggests that Equation (5.8) may be preferable for thick sections. The effect of profile thickness has in fact been allowed for empirically by Thompson (Ref. 5.6; 1948), whose work suggests an equation of the form (see Reference 5.25\*)

$$\epsilon_{g} = 0.524 \left[ 1 + 1.2\beta \left( \frac{t}{c} \right) \right] \frac{A}{\beta^{3}h^{2}}$$
(5.9)

which agrees with Thom's relation for very thin aerofoils and with Young and Squire's equation when the thickness/chord ratio is about 0.2. An equation similar to (5.9) is probably the best simple approximation for the solid-blockage factor provided c/h is not too large, say about 0.25 for moderate subsolic Mach numbers.

It is interesting to note that, if  $A = x_2 ct$  and the aerofoil is replaced by an ellipse of the same area and thickness (but smaller chord), Equation (5.6) may be written in a similar form to Equation (5.9) but with the numerical factor 1.2 replaced by  $\frac{1}{4\pi}/k_2$ . If  $k_2 = 0.65$ , a typical value, then  $\frac{1}{2\pi}/k_2 = 1.21$ . This correspondence suggests that a more general form of Equation (5.9), or of Equation (5.6) is possible; in the latter case  $\lambda_2$  must be replaced by  $\frac{1}{2}[1 + \sqrt{4k_2}/\pi](c/t)]$ .

It is obviously possible to extend the preceding analyses to include terms up to the Jourth power in (c/h); this was done in incompressible flow by Goldstein (Ref. 5.7; 1942) who derived an equation which must be solved for each aerofoil. Allow and Vincenti<sup>5.3</sup> show that the first term in this is identical with Equation (5.7), so that Goldstein's relation becomes

$$\epsilon_{g} = \frac{\pi \Lambda}{6h^{2}} + \frac{\pi^{3}}{960} \left(\frac{c}{h}\right)^{\mu} \int_{0}^{\pi} \frac{z(\mathbf{x})}{c} \frac{\cos 4\theta}{\sin \theta} d\theta . \qquad (5.10)$$

where , with x measured from the leading edge, heta is defined by

$$x = \frac{1}{2}c(1 - \cos \theta)$$
. (5.11)

<sup>4</sup> The compressibility factor applied to the aerofoll thickness in Equation (5.9) was omitted in Thompson's original paper, and later inserted by Evans. Though not strictly in accord with Goldstein's theory in Equation (5.14) below, it is partially justified by experimental data and is recommended by the present author.

291

and the formation of the states and the states

and a strategic tendors and

E7.

The more complex solid-blockage corrections have also received considerable attention from Woods. In Reference 5.8 (1954), for example, he shows that the distribution of solid blockage along a thin asrofoil may be written as

$$\epsilon_{\rm g}({\bf x}) = \frac{\pi \Lambda}{6\beta^3 {\bf h}^2} \left[ 1 + \frac{N}{\beta} - \frac{{\bf k}({\bf x})}{5\Lambda} \left( \frac{\pi}{\beta {\bf h}} \right)^2 \right], \qquad (5.12)$$

where k(x) is the moment of inertia of the section about the position x. The parameter N is given by

$$N = -\frac{2}{\pi A} \int_{0}^{\infty} \int_{0}^{0} \frac{dz(x')}{dx'} \frac{z(x)}{x' - x} dx' dx . \qquad (5.13)$$

This topic is also discussed in Section 7.8 of Woods's book (Ref. 5.22; 1961). To the first approximation  $\epsilon_{g}(x)$  is constant along the chord and has the value calculated by Thom in Equation (5.7). This variation of  $\epsilon_{g}$  may be used to determine some mean value over the length of model, particularly if balance measurements have been made. In correcting the measured pressure distribution along the model, the local value of  $\epsilon_{g}$  may be used; care is required if  $\epsilon_{g}$  changes greatly along the model, as this may indicate that the pressure distribution itself may be distorted by the tuncel walls, thus reducing the validity of applying a blockage factor of the type contained in Equation (5.12). Hore complicated methods of calculating the blockage may then be required; some of these are discussed below.

To illustrate a particular application of Equation (5.12), an elliptical profile may be considered. Then

$$\epsilon_{g}(\mathbf{x}) = \frac{\pi A}{6\beta^{3}\mathbf{h}^{2}} \left[ 1 + \frac{\mathbf{i}}{\beta} \left( \frac{\mathbf{t}}{\mathbf{c}} \right) \right] - \frac{\pi^{3}}{1920} \left( \frac{\mathbf{c}}{\mathbf{h}} \right)^{3} \frac{1}{\beta^{3}} \left( \frac{\mathbf{t}}{\mathbf{c}} \right) \left[ 1 + 4 \cos^{2} \theta \right] . \quad (5.14)$$

The solid blockage is therefore a maximum at the mid-chord of the profile and falls off towards the leading and trailing edges. The first term is in fact similar to that derived by Thompson in Equation (5.9) apart from the compressibility factor for thickness effect. The second term of Equation (5.14) in incompressible flow might be expected to agree with that derived from Equation (5.10) for an ellipse. At the midchord position, Woods' equation has a term of magnitude  $-(\pi^5/1920)(c/h)^4(t/c)$ , whereas Goldstein's equation has a (c/h)<sup>4</sup> term of zero value. Similar discrepancies occur for a biconvex profile.\*

Knechtel (Ref. 5.52; 1953) has made an experimental investigation in which the ratio of aerofoil chord to tunnel height was varied progressively. Near zero-lift conditions at Mach numbers up to 0.85 satisfactory corrections can be made for boundary-interference effects with only terms in  $(c/h)^2$  unless c > 0.15h. For larger models the simple corrections become increasingly questionable.

In allowing for the effect of compressibility Woods uses hodograph-type equations to avoid the limitations of simple small-perturbation theory. Considerations of this type have led Klunker and Harder (Ref. 5.9; 1951) to use the so-called Prandtl-Basemann

Contraction of the second second

\* At present the reason for this disagreement is not apparent.

iteration method for small disturbances to calculate a second approximation to the blockage corrections for a thin symmetrical body at zero incidence. It is assumed that the potential function of the non-linear equations of motion may be expanded in a series form where each successive term is small compared with its predecessor. It is possible in this way to calculate the blockage effect over the complete model surface and to estimate the errors which may arise, when  $c/\beta h$  is large, from ignoring the chordwise changes in induced velocity.

.m. \_\_\_\_\_\_

The occasional need for more precise solid-blockage corrections than result from the simple theories has given rise to several diverse approaches, a few of which may be mentioned briefly. Thom and Klanfer (Refs. 5.10 and 5.11; 1946-47) have developed a "squares" method for solving the complete field about an aerofoil in a wind tunnel. A similar relaxation technique has been used by Emmons (Ref. 5.12; 1948) and by Epstein and Albers (Ref. 5.13; 1948). In the latter case a 10-per-cent-thick symmetrical aerofoil of two sizes,  $c = \frac{1}{2}h$  and c = h, was considered at stream Mach numbers of zero and 0.5. Tsien and Lees (Ref. 5.45; 1945) have combined linear perturbation theory with a representation of the aerofoil by a source-sink distribution and consider the correction to the maximum surface velocity on a 16-per-cent-thick parabolic-arc profile for two different heights of tunnel. Whitehead (Ref. 5.17; 1950) gives an additional method for calculating the solid blockage of very large bodies, and the effect of the tunnel walls on the critical Mach number of a circular cylinder spanning a tunnel is discussed in a paper by Suzuki (Ref. 5.53; 1941) who uses the hodograph method.

These examples perhaps serve to illustrate that in seeking more accurate corrections for a large model much simplicity is lost and extensive calculation may be needed for each case. It is unlikely therefore that use of such techniques for routine test work will be wide-spread. With large models indeed it may no longer be sufficient to calculate the distribution of blockage along the model and assume that this is independent of model lift. What may then be required is the distortion of the surface pressure distribution due to the presence of the wind-tunnel walls.

A process of conformal mapping is valuable for obtaining this information. No general expressions can be derived and a fresh calculation must be made for each merofoil and test incidence. Both the symmetrical merofoil at zero lift and the arbitrary lifting merofoil have been discussed by Perl and Moses in References 5.15 and 5.16; the results, obtained by a series of flow transformations, are compared with those obtained from image methods of icck<sup>5.1</sup> and Goldstein<sup>5.7</sup>. At incidence the solid blockage is calculated as well as the modified surface pressures. It is claimed that the conformal mapping method is very convenient to use if more exact corrections are required than would be given by equations using terms up to  $(c/h)^4$ . A similar though less complicated method of mapping was developed some years earlier by Franke and Weinig (Ref. 5.20; 1939). A full discussion of the use of a complex potential sethod for determining the combined lift and blockage interference of a two-dimensional aerofoil is given by Earbieux (Ref. 5.54; 1955), where comparison is made with the simple equations derived by lock. Thos, Allen and Vincenti and others. Barbieux gives results from which the distribution of blockage along the profile may be calculated.

Though the advanced general methods deal sdequately with the lifting merofoil, it is possible to adapt the simple theory to deal with the solid blockage of an merofoil at an incidence  $\alpha$ . This was done for incompressible flow by Batchelor (Ref. 5. 21; 1944), who showed that the solid-blockage factor must be increased by an amount proportional

293

to  $a^2$ . The relationship between the new blockage factor and that appropriate at

zero incidence may be written approximately as

$$\epsilon_{\mathbf{s}}(\alpha) = \epsilon_{\mathbf{s}} \left[ 1 + 1.1 \left( \frac{c}{t} \right) \alpha^2 \right],$$
 (5.15)

where  $\alpha$  is measured in radians. For c/t = 10 and  $\alpha = 10^{\circ}$  the blockage factor is about one-third greater than at  $\alpha = 0^{\circ}$ . This order of increase is near that derived from semi-empirical corrections to large models at incidence based on the measurement of wall pressures. One such approach is discussed in Reference 5.19 where the velocities at the walls of the tunnel close to the model position were measured and then averaged to remove the lift effect. The increase in this average value with model incidence was related to changes in the blockage. Such a technique can, of course, be regarded as an extension of more normal methods for estimating the solid blockage from the pressures on the tunnel walls, a subject discussed further in Section 5.6. For compressible flow, Equation (5.15) may still be used, though presumably with the additional factor  $\beta$  in the term in  $\alpha^2$ .

In all the foregoing discussion it has been assumed that the model is placed midway between the two tunnel walls. If the model is offset from the centre line by a distance  $(\frac{1}{2}h - d)$ , so that d is the distance of the aerofoil from the tunnel wall, a small increase in the solid-blockage factor occurs. This effect has been estimated for low-speed flow by Batchelor<sup>5-21</sup> and the relationship between the off-centre and centre-line blockage factors may be written as

$$\epsilon_{\mathbf{g}}(\mathbf{d}) = \epsilon_{\mathbf{g}} \left[ 1 + \frac{3}{4} \cot^2 \frac{\pi \mathbf{d}}{\mathbf{h}} \right];$$
 (5.16)

this equation is also valid in subsonic compressible flow. Thus, a model offset from the tunnel centre line by 0.1h will be subject to about 8% increase in solid blockage. It should be pointed out that, when the model is offset, the transverse component of the velocity induced by the images of the model is no longer zero; strictly therefore there should be an incidence (and camber) correction due to solid blockage. This is usually very small, especially when compared with the corresponding correction due to the lift on the model. The merofoil displaced from the tunnel centre line is also discussed in Reference 5.7.

## 5.2.2 Body of Revolution

The analysis for a body of revolution, though more complex, is in many ways similar to that for an aerofoil. In the early work Lock (Refs.5.1; 5.29) replaced a small body by a doublet whose strength was related to  $\overset{\circ}{}$  e-stream velocity by a factor  $\lambda_3$ , analogous to  $\lambda_2$  in the preceding section of  $\lambda_3$ , where  $A_{\rm m}$  is the maximum cross-section of the body normal to the stream and, corresponding to Equation (5.3),

$$\lambda_3 = 4 \int_{\mathcal{T}} \frac{q}{v} \frac{[z(s)]^2}{t^3} ds . \qquad (5.17)$$

The images of this doublet persist to infinity on all sides of the model, so that a doubly-infinite summation is required to obtain the solid-blockage factor. For a

$$\epsilon_{\rm g} = \tau \left(\frac{\Lambda_{\rm m}}{\rm hh}\right)^{3/2} \lambda_{\rm g} , \qquad (5.18)$$

where

$$\tau = \frac{1}{2\pi^{3/2}} \sum_{-\infty}^{\infty} (\frac{bh}{n^2b^2 + n^2h^2})^{3/2} .$$

model centrally placed in a tunnel of height h and breadth b

Here  $\sum_{-\infty}^{\tau}$  indicates that (m, n) takes all integral pairs except (0,0).  $\tau$  is a

numerical coefficient which must be determined for each ratio of tunnel height to breadth; for a square tunnel  $\tau = 0.309$  (Ref. 5.1). Lock calculated  $\lambda_3$  for a Rankine ovoid and a spheroid and this work was later extended by Borden (Ref. 5.2; 1954) and Harriot (Ref. 5.23; 1950). The latter reference contains tables from which  $\lambda_3$  can be estimated for many shapes of body. The variation of  $\lambda_3$  with body fineness ratio is shown in Figure 5.2.

Young (Ref. 5.5; 1945) has also estimated  $\lambda_3$  for bodies having pointed tails, for a range of maximum-thickness position. If the fineness ratio of the body is denoted by f, the equation

$$\lambda_3 = 0.4 \div 0.49f$$
 (5.19)

approximately represents Young's results (Fig. 5.2) and gives a rather lower value than for symmetrical bodies of the same fineness ratio. A more recent discussion of Lock's method, which includes an approximate method of deriving  $\lambda_3$  for streamline bodies has been given by Vasy (Ref. 5.24; 1957).

Despite the existence of fairly extensive data for estimating  $\lambda_3$  for any given body. many authors have adopted the alternative approach whereby the blockage factor is related to the body volume V. This is analogous to the use of the cross-sectional area A in the case of a tro-dimensional zerofoil. It can be shown that

$$2\lambda_{-}A_{-}^{3/2} = \sqrt{\pi}V, \qquad (5.20)$$

and that the strength P of the doublet representing the model is given by the product VU. Thus Equation (5.18) may be rewritten

$$\epsilon_{g} = \left(\tau \frac{\sqrt{\pi}}{2}\right) \left(\frac{1}{bh}\right)^{3/2} V = T \left(\frac{1}{bh}\right)^{3/2} V, \qquad (5.21)$$

where the tunnel shape factor

;

$$T = \frac{1}{4\pi} \sum_{-\infty}^{\infty} \left( \frac{bh}{a^2 b^2 + n^2 h^2} \right)^{3/2} .$$

Small perturbation theory allows the effect of compressibility to be estimated in the same way as for an serofoll<sup>\*</sup>. Thus

$$\epsilon_{g} = T \left(\frac{1}{bh}\right)^{3/2} \frac{V}{\beta^{3}}. \qquad (5.22)$$

Evans<sup>5.25</sup>, using some earlier work of Thompson<sup>5.6</sup>, includes an additional term to allow for the effect of body fineness ratio, so that Equation (5.22) becomes\*\*

$$\epsilon_{\rm s} = T \left(\frac{1}{bh}\right)^{3/2} \frac{V}{\beta^3} (1 + 0.4\beta/f) .$$
 (5.23)

If it is assumed that  $V = k_3 lt^2$ , where l is the length of the body and t its maximum d'ameter, this modification is equivalent to putting in Equation (5.18)

$$\lambda_3 = \frac{4}{\pi} k_3 [f + 0.4] , \qquad (5.24)$$

which, for a typical value  $k_3=0.45$  , gives values of  $\lambda_3$  close to those derived more directly.

In Section 5.2.1 it was suggested that the solid-blockage factor of an aerofoil could be determined readily if the model was replaced by an equivalent ellipse. Similarly, the body of revolution may be replaced by a spheroid having the same volume and maximum thickness; the length of the equivalent spheroid will usually be less than l, and the equivalent fineness ratio will be

$$f_e = \frac{6k_3}{\pi} f$$
. (5.25)

and the second second second second

The value of  $\lambda_3$  for the body may now be found directly from Figure 5.2. As an example, consider a body of actual fineness ratio 10, where the equivalent-spheroid approach yields the value  $\lambda_3 = 5.91$  in close agreement with  $\lambda_3 = 5.96$  from Equation (5.24).

So far we have considered the contribution of the model geometry to the blockage factor. The tunnel shape also influences  $\epsilon_{\rm B}$  through the parameter T or  $\tau$  and values for specific tunnel cross-sections are given in zeveral papers (e.g., References 5.1, 5.4, 5.23). The most complete calculations for the tunnel-shape parameter are probably those by Herriot<sup>5.23</sup>, covering values of b/h between 0.29 and 3.5 with corresponding changes in  $\tau$  from 0.31 to 1.73. These results are replotted in Figures 5.3 and 5.4 in terms of T, where use is made of the fact that for a small model the value of  $\tau$  is independent of the orientation of the model with respect to

• In some early papers there was confusion about the precise form of the boundary conditions at the model and hence of the compressibility factor. It is now generally accepted that  $\beta^{-3}$  is correct.

\*\* The remarks contained in the footnote, indicated just before Equation (5.9), about the compressibility factor for thickness effect of two-dimensional profiles apply to three-dimensional bodies also.

the walls. Approximate formulae for T have been suggested as an alternative to the more precise calculations. Thus, for  $\frac{1}{2} < b/h < 2$ , Thompson<sup>5.6</sup> recommends

$$\mathbf{T} = 0.36 \left[ \frac{\mathbf{b}}{\mathbf{h}} + \frac{\mathbf{h}}{\mathbf{b}} \right]. \tag{5.26}$$

As Figure 5.4 shows, this lies very close to Herriot's curve. For 1.2 < b/h < 2.0Young and Squire<sup>5.5</sup> have suggested the following formula

$$\Gamma = 0.65 \left(\frac{b}{h}\right)^{\frac{1}{2}}$$
, (5.27)

which, though far less satisfactory, leads to a very simple form of Equation (5.22),

$$\epsilon_{\rm s} = 0.65 \frac{V_{\rm m}}{\beta^3 {\rm h}^2 {\rm b}}.$$
 (5.28)

It is claimed that this gives the blockage correction to within  $\pm 5\%$  for any threedimensional model. For a body of revolution specifically, Young and Squire suggest that the numerical factor is taken as 0.68. There would seem to be little to be savea in using their form in place of the more exact relation (5.22) or (5.23), combined with (5.26).

As in the case of the two-dimensional aerofoil, it is sometimes necessary to use more complex expressions for the solid blockage, if the body is long compared with the tunnel dimensions. This problem has been considered, for example, by Evans (Ref. 5.25; 1949), whose method is essentially a development and extension of the earlier papers by Thom and Thompson. A thin body of revolution may be represented approximately by a distribution of sources and sinks along the body axis, and the walls by an infinite array of images. The variation of the blockage factor along the body may now be calculated; this is done by Evans for a tunnel where b/h = 1.43. The method, though more complex than the simple analyses discussed earlier, may easily be applied to any value of b/h so that the validity of the simple equation can be checked in any particular case.

A model mounted off the tunnel axis is considered in Reference 5.57.

## 5.2.3 Wings and Wing-Body Combinations

For the purpose of calculating the solid-blockage effects the small wing or wing-body combination of finite span may be regarded as identical to a body of revolution of the same volume. This implies, however, that the wing is replaced by an equivalent sphere or spheroid, and not, as one might feel is more appropriate, by an equivalent cylinder. A rectangular wing of 6 ft (1.83 m) span, 1 ft (0.305 m) chord with t/c = 0.125 would be replaced in the analysis by a sphere of diameter about 11.8 ins (0.30 m); the equivalent spheroid, having the same frontal area as the wing, would not be greatly different from this sphere.

Though precision is not possible in such matters a "small" wing may conveniently be regarded as one in which the ratio of the sing span (23) to the tunnel breadth (b) is

CONTRACTOR OF THE STREET

wax.

#### 298

below about 0.5. Within this range, Equation (5.22) may be used and Equation (5.23) monified to become

$$\epsilon_{\rm s} = T \left(\frac{1}{\rm c}\right)^{3/2} \frac{V}{\beta^3} \left[1 + 1.2\beta \left(\frac{\rm t}{\rm c}\right)\right], \qquad (5.29)$$

where C = bh. It is assumed in this equation that the wing effects will be more important than those arising from the body.

When the model is no longer small compared with the tunnel dimensions, its representation by a doublet becomes invalid. For a slender delta planform a distribution of sources and sinks lying along the model axis should be used and the calculation made as for a long body of revolution<sup>5.25</sup>. In practice, a single source-sink arrangement is often sufficient. If the wing is unswept or only moderately swept, it may be better to represent it by a series of finite line sources and sinks running either parallel to the span, or parallel to the leading edge. Such an approach was made by Thom<sup>5.4</sup>, who chose to keep the strength of the sources and sinks constant along the span despite the fact that this arrangement strictly entails some distortion of the planform and a reduction in wing thickness near the tip. The axial velocity induced in the tunnel by the images of the distribution representing the model may now be found, though in some cases the calculation may be complex.

The representation of the wing by line sources and sinks was also adopted by Herriot<sup>5.23</sup>, whose results cover a wide range of tunnel shape (0.29 < b/h < 3.5) and model span (0 < 2s/b < 1) in compressible flow. Herriot's results may be expressed in terms of the tunnel-shape parameter  $\tau$  and the body-shape parameters  $\lambda_2$  and  $\lambda_3$ . In the notation of the present text his equation for the solid-blockage factor at the centre of a finite wing of frontal area  $A_{\mu}$  may be written as

$$\epsilon_{\rm s} = \frac{\pi^{3/2} \tau \mathbf{A}_{\rm s} t \lambda_2}{4\beta^3 (\rm bh)^{3/2}} , \qquad (5.30)$$

where  $\tau$  now depends on 2s/b.

Equation (5.30) may be transformed to yield equations similar to (5.22) and (5.29) above, but where T is now a function of b/h and 2s/b, as shown in Figures 5.5° and 5.6 which are based on Reference 5.23; when 2s/b < 0.5, T does not vary greatly with 2s/b. The general forms of Equations (5.23) and (5.29) may be applied to wingbody combinations as well as simple wings, since the effect of the wing is generally of greatest importance in determining the blockage. In cases where this is less obvious, the contributions of the wing and body separately may be estimated and their effects added.

When the wing is swepthack, and perhaps tapered, the lines of sources and sinks may no longer be normal to the stream and the infinite summations are less easy. Evans<sup>5.25</sup> has shown that most wings may be represented by a uniform non-tapered wing, having the

• For 2s/b = 1, a comparison can be made with two-dimensional results by putting Y = Ab in Equation (5.22). In this case

$$\Gamma = \frac{\pi}{6} \left( \frac{b}{b} \right)^{2} + \frac{1}{\pi} \left( \frac{b}{b} \right)^{2/2}$$

where the first term corresponds to Equation (5.7) and the second term arises from the side-wall images.

•-

RANGE AND A SUBMIC

same volume, mean sweep and thickness-chord ratio as the original wing, but with a span equal to  $2\sqrt{3} k_x$ , where  $k_x$  is the radius of gyration of the original wing about the x axis. In Reference 5.25, Evans is concerned with a tunnel with b/h = 1.43 and carries out summations for various mean sweeps, so that  $\epsilon_s$  may be calculated over the model. The method, however, is of very general application.

When a half-span model, mounted directly on to the tunnel side-wall, is being tested, the solid blockage at the model position is approximately that due to a complete model in a tunnel of twice the breadth.

It may sometimes be necessary to calculate the solid-blockage effects of model supports, such as struts and stings. Some care is required, because the strut is roughly equivalent to a half-wing model and the sting to a very long solid of revolution; moreover the corrections to stream velocity are required at the model position and not at the centre of the support. It is sometimes preferable to attempt to measure directly this velocity increment associated with the support gear in the absence of the model.

# 5.3 SOLID BLOCKAGE FACTORS IN CLOSED NON-RECTANGULAR TUNNELS

#### 5.3.1 Octagonal Tunnels

Many wind tunnels, though basically of rectangular cross-section, have corner fillets to form what may be described as an octagonal working section. The effect of the corner fillets in modifying the results presented in Section 5.2 is therefore of some importance, though it is likely that significant differences will only arise if the fillets are large. The analysis for an octagonal tunnel is less straightforward than for a rectangular tunnel and only very approximate methods are at present available. Batchelor<sup>5.21</sup>, considering incompressible flow, uses the fact that the tunnel height varies along the span of a two-dimensional aerofoil and argues that there is a consequent spanwise variation in  $\epsilon_s$  which locally is given by Equation (5.2). This means that when the overall force on the wing is measured the interference must be averaged across the span to give an equivalent height, as defined by Equation (2.29) of Chapter II. Details of the results for the Melbourne 9 ft  $\times$  7 ft tunnel will be found in Reference 5.21: some care is required if the forces are measured on only part of the wing span. Batchelor's approach probably overestimates the interference present on a full-span model, and it would seem safer to adopt the alternative equivalent height suggested in Equation (2.30) of Chapter II. The fillets can probably be ignored in relation to measurements of pressure at the mid-span position.

Batchelor also discusses the interference likely to arise from three-dimensional models within the octagonal working section. It is assumed that the solid blockage is the sum of that due to the basic rectangular section and a contribution from the corner fillets. The condition of zero flow across the fillet boundary is only approximately satisfied by the chosen distribution of sources and sinks because of the complexity of the analysis. Working in terms of the tunnel-shape parameter  $\tau$  and the three-dimensional model-shape parameter  $\lambda_3$ . Batchelor estimates the blockage factor from the following equation

$$\tilde{\epsilon}_{s} = \tau_{v} \left(\frac{A_{m}}{bh}\right)^{3/2} \lambda_{3} + \tau_{1} \left(\frac{A_{m}}{C}\right)^{3/2} \lambda_{3} , \qquad (5.31)$$

where  $\tau_0$  is the value for the basic rectangular tunnel,  $\tau_1$  is the calculated increment due to the fillets, and C is the cross-sectional area of the ortagonal working section. Equation (5.31) may be rewritten as

$$f_{s} = \left[\tau_{0}\left(\frac{c}{bh}\right)^{3/2} + \tau_{1}\right]\left(\frac{A_{m}}{c}\right)^{3/2}\lambda_{3} . \qquad (5.32)$$

which is in an analogous form to Equation (5.18). For the Melbourne tunnel  $\tau_0 = 0.83$ , and the first term in the square bracket has a value of 0.71.  $\tau_1$  is estimated to be about 0.04 which is sufficiently small to suggest that, for moderately small fillets, a sufficiently good answer may be obtained by simply calculating the blockage factor appropriate to the basic rectangular tunnel, i.e., by putting  $\tau_1 = 0$  in Equation (5.32). If this procedure is valid, then any of the formulae in Section 5.2 may be used and the effect of compressibility is allowed for by the factor  $\beta^{-3}$ .

The lift interference on a model in an octagonal tunnel was also discussed by Batchelor in Reference 5.21, and this approach was later extended by Loos (Ref. 5.51; 1951), who represented the tunnel walls in a simpler but rather more approximate fashion. It seems possible that a similar analysis could be made for the blockage correction.

#### 5.3.2 Circular and Elliptical Tonuels

Tunnels having a closed circular cross-section are now less widely employed than in the past; as a consequence there is less literature on blockage effects than for the rectangular working section.

The calculation of the solid blockage attributable to an aerofoil spanning a closed circular tunnel is far from easy and seems first to have been discussed by Vincenti and Graham (Ref. 5. 27; 1946): they pointed out that the interference cannot be found by the image method, since no system of images satisfies the appropriate conditions at the tunnel boundary. The flow field of the non-lifting two-dimensional model was represented by a distribution of horse-shoe vortex elements of infinitesimal span for which the interference field in a circular tunnel is known<sup>5.55</sup>. In the present notation, these authors obtained, for the solid-blockage factor at the tunnel centre,

$$\epsilon_{\rm s} = 1.356 \left(\frac{\rm c}{2\rm R}\right)^2 \left(\frac{\rm t}{\rm c}\right)^2 \frac{1}{\beta^3} \lambda_2$$
, (5.33)

where R is the radius of the tunnel. This is identical to Equation (5.6) for an aerofoil spanning a two-dimensional rectangular tunnel of height 1.558R. It follows that, provided this transformation of tunnel height and diameter is made, the several blockage equations of Section 5.2.1 may be used. Thompson's Equation (5.9) becomes, for example,

$$\epsilon_{\rm g} = 0.216 \left[ 1 + 1.2\beta \left( \frac{t}{c} \right) \right] \frac{A}{\beta^3 R^2} . \qquad (5.34)$$

Vincenti and Graham show that, because of the symmetry of the system, no vertical velocity is induced at any position along the span of the aerofoil, nor is there a streamwise pressure-gradient associated with the solid blockage.

The blockage effect of a body of revolution inside a circular tunnel was one of the earliest problems of this type to be discussed. Lamb (Ref. 5.26; 1926) considered a Rankine ovoid. The series of Bessel functions which arise in the analysis scre subsequently reduced to a more usable form by Watson (Ref. 5.28; 1936). Glavert<sup>5.2</sup> and Lock<sup>5.29</sup> derived the solid-blockage factor for incorrespible flow in the simple form

$$\epsilon_{s} = 0.797 \left(\frac{\lambda_{z}}{C}\right)^{3/2} \lambda_{3} , \qquad (5.25)$$

where  $C = \pi R^2$ . Hence

 $\epsilon_{\rm s} = 0.149 \left(\frac{\lambda_{\rm s}}{\kappa^2}\right)^{3/2} \lambda_{\rm s} ; \qquad (5.36)$ 

 $\lambda_3$  may be derived for a wide range of bodies from the tables in Seference 5.23, or alternatively estimated from Pigure 5.2. The numerical factor in Equation (7.35) is, of course, a tunnel-shape parameter analogous to  $\tau$ ; its value is close to that for a square tunnel (0.31).

Croce again it is sometimes convenient to use the model volume V and a tunnel-shape parameter corresponding to  $T(=\frac{1}{2}\sqrt{\pi}\tau)$ . With allowance for compressibility, Equations (5.35) and (5.36) then become

$$\epsilon_{\rm g} = 0.706 \left(\frac{1}{c}\right)^{3/2} \frac{\rm v}{\beta^3}$$
 (5.37)

and

$$\epsilon_{\rm s} = 0.128 \frac{V}{\beta^2 R^3}$$
. (5.38)

The tunnel parameter T thus has a value of 0.706 for circular tunnels. Equations of this type have been derived by several authors<sup>5,23,5,30-5,32</sup> for small bodies of revolution, small wings or wing-body combinations. Thompson's thickness modification may also be applied by multiplying the equations by either  $[1 + 1.2\beta(t/c)]$  or  $(1 \div 0.4\beta/f)$ .

When the model span is no longer small compared with the tunnel breadth, the authorical constants in Equations (5.35) and (5.37) must be modified<sup>5.23</sup> as shown in the following table and Figure 5.5.

| 2s/0 | 0     | <b>ə</b> . 25 | J. 50  | Q. 75  |
|------|-------|---------------|--------|--------|
| T    | 0.797 | 0.812         | 0, 855 | 0,855  |
| T    | 0.705 | 0.720         | 0.734  | 0. 761 |

301

「あるので、「「ないない」というない」をなったいです。

A WHICH DESIGNATION OF THE PARTY OF THE PART

The actual variation in T with model size is very close to that for a square tunnel. Herriot's theory is not really applicable at the limit 2s/b = 1, but a similar difference to that noted for rectangular tunnels (footnute in Section 5.2.3) occurs if his limiting results are compared with the quasi-two-dimensional Equation (5.34).

Fuchshiber (Ref. 5.59; 1961) gives a mathematical solution of the axisymmetrical problem without numerical results. The solid-blockage corrections associated with a body of revolution at incidence in the centre of a circular tunnel has been considered in general terms by Cremer and Kolberg (Ref. 5.58; 1960). Even the expression for the velocity increment at the centre-line of the tunnel is quite complicated, and the authors suggest the use of a computer in any applications.

The distribution of solid blockage along a body of length l placed at the axis of a circular tunnel has been discussed by Vandrey (Ref. 5.33; 1951) in a paper which may be regarded as the counterpart of Borden's analysis<sup>5.2</sup> for rectangular tunnels. Vandrey replaces the body by a Rankine ovoid having the same volume and maximum crozssection and obtains an expression for the solid-blockage factor at a distance x from the body centre

$$\epsilon_{s}(x) = [H(\xi_{1})] - H(\xi_{2})] \frac{A_{s}}{c},$$
 (5.39)

varte

$$5_1 = \frac{x + \frac{1}{2}x}{R}$$

$$5_2 = \frac{x - \frac{1}{2}x}{R}$$

and  $H(\zeta)$  is given in Table 5.1. Distributions of  $\epsilon_g(x)$  for bodies of different lengths are illustrated in Figure 5.7.

The maximum blockage factor is reached at the centre of the body (x = 0), where

 $\epsilon_{s}(0) = 2\left[H\left(\frac{a}{2R}\right)\right]\frac{A_{s}}{C},$  (5.49)

where a is the distance between the source and the sink that represent the body. For a Rankine ovoid the shape parameter  $\lambda_3$  is approximately equal to the body financess ratio<sup>5.2</sup>. Equation (5.49) may be rewritten as

$$\epsilon_{g}(0) = \left\{ \frac{4R}{tf} H\left(\frac{a}{2R}\right) \right\} \lambda_{3} \left(\frac{A_{a}}{C}\right)^{3/2}$$
$$= \tau \left(\frac{A_{a}}{C}\right)^{3/2} \lambda_{3} . \qquad (5.41)$$

which is formally similar to Equation (5.18). The parameter  $\tau$  will now vary with the ratio of the body length to tunnel diameter (Fig. 5.8). For a very small body  $\tau$ 

303

the last of the second in which we also and a mere and a second of the second of the second second to be a second second

approaches Lock's value of 0.797. It is at once apparent that use of this small-body value for long bodies will seriously overestimate the maximum solid-blockage correction.

ţ

Though the analysis is strictly only applicable to Rankine ovoids, it should also apply approximately to bodies of rather similar shape. Thus Equation (5.41) may be rewritten as

$$\varepsilon_{\rm g}(0) = T \left(\frac{1}{C}\right)^{3/2} \frac{V}{\beta^3}$$
 (5.42)

where  $T(=\frac{1}{2}\sqrt{\pi}\tau)$  is also given in Figure 5.8. It is interesting to note the marked similarity between Wandrey'r work and an earlier discussion by Lock and Johansen (Ref. 5.34; 1931). The value of  $\varepsilon_g(0)$  is by definition the maximum schieved on the body and not the average over its length. It may be argued that for many purposes, such as the determination of critical flow Gorditions, it is this maximum value which is of greatest importance; in other cases, such as the correction of drag and pressure measurements, the mean blockage may well be dated.

Vandrey's method may also be applied to find the solid-blockage factor for a semi-infinite body; the variation ahead of, and along, the body is shown in Figure 5.9. The maximum value of  $\epsilon_g$  has almost been reached by one tunnel diameter from the node of the body. For this case the origin can convertently be changed to the source position near the body nose. If x, is the distance from this origin, then Equation (5.39) reduces to

$$\epsilon_{\mathbf{s}}(\mathbf{x}_{1}) = \left[0.5 + \mathrm{H}/\xi_{3}\right] \left(\frac{\mathrm{A}_{\mathbf{s}}}{\mathrm{C}}\right), \qquad (5.45)$$

where

$$\xi_3 = z_1/R.$$

Since the velocity gradiest at the origin due to a source corresponds in magnitude to the velocity induced by a doublet, the tangent to the care shown in Figure 5.9 may be regarded as equivalent to Lock's small-body Equation (5-35) if the fineness ratio is assumed to equal  $\lambda_{\rm s}$ .

Though several authors have considered the lift interference associated with a model mounted in a tunnel of elliptical cross-section, the corresponding blockage problem seems to have been neglected. To estimate the solid-blockage factor at the centra section of a two-dimensional aerofoil spanning an elliptical tunnel, it would seem possible to apply rectangular-tunnel corrections for an effective tunnel hoight  $h_e$ . If a and n are the major (spanwise) and minor area of the tunnel, then  $h_e$  will be in the range between the values for circular and rectangular tunnels  $0.779 < h_e/n < 1$ ; in the absence of other evidence, a simple linear relationship between  $h_e/n$  and n/m is suggested in Figure 5.10. For balance measurements on the full aerofoil span,  $h_e$  may be taken as equal to  $\frac{1}{2}mn$ .

For a small three-dimensional model the tunnel-shape parameter T or  $\tau$  for a circular tunnel is within 2% of that for a square tunnel, and Pigure 5.6 shows that the variation of the tunnel-shape parameter with increasing model size is similar for the two shapes of working section. In the absence of direct information it is suggested that, for three-dimensional models, the approximate value of T or  $\tau$  for an

elliptical tunnel may be taken as that for a restangular tunnel of breadth to height ratio m/n. It is important to rescaber that, in adapting the formulae of Sections 5.2.2 and 5.2.3, the cross-sectional area of the elliptical tunnel  $C(=\frac{1}{4}mmn)$  should be used in place of bh.

#### 5.4 WARE BLOCKAGE FACTORS IN CLOSED TUNNELS

# 5.4.1 Two-Disensions) Acrofcii in a Rectangular Junzel

An early attempt to allow for the wake blockage of a two-dimensional aerofoil in incompressible flow was made by Fage (Ref. 5.55; 1929) and Lock (Ref. 5.29; 1929), who introduced an empirical factor bised on the effective width of the wake, into the solid-blockage correction. Later disavet<sup>5.1</sup> drew an analogy with the discontinuous flow behind a bluff body in a constricted stream and suggested that the effect of the wake-invared longitudinal velocity increment on the measured drag force (D) could be represented by

$$D_{c} = D \left[ 1 - \frac{\eta t}{h} \right]^{2}$$
, (5.44)

where  $D_{c}$  is the corrected drag;  $\eta t$  is the effective width of the cake some distance downstream of the model and is assumed to vary with thickness/chord ratio in the manner shown in Figure 5.11. This curve is based on the experimental evidence available when Reference 5.1 was written (1982). Equation (5.44) corresponds to a wake-blockage factor

$$\epsilon_{\psi} = \eta \left(\frac{t}{c}\right) \left(\frac{c}{h}\right).$$
 (5.45)

if terms in  $\eta^2$  are reglected.

Subsequent authors have left that the use of the factor  $\eta$  was in many ways unGatisfactory, particularly for aeroluits and slender bodies and it is now usual to relate  $\sigma_{\rm r}$  to the measured boay drag when this is known or can be estimated. Use had often been made of Frandtl's suggestion that the wake can be represented by an equivalent source situated at space point on the aerofoil (say, aid-chord or to diling edgo). The strength of this source Q is related to the screfoil drag by

$$\mathbf{q} = \frac{\mathbf{D}}{\mathbf{\rho}\mathbf{u}} \,. \tag{5.46}$$

A sink of equal strength roat be placed for downstress. The velocity increment due to wake blockage may then be calculated as that effectively induced at the model position by the infinite array of source and sink images. The appropriate wake-blockage factor in incompressible flow

$$\epsilon_{g} = \frac{1}{4} \left( \frac{c}{k} \right) \tilde{c}_{g} \tag{5.47}$$

has been derived by many authors. For modern acrossil sections, Equation. (5.47) gives a lower value of  $\epsilon_{\pm}$  than Equation (5.45) and is to be preferred.

Equation (5.47) is derived in slightly different ways in the literature, mainly because of inherent difficulties in satisfying the conditions well downstream of the model\*. These differences are of little importance in incompressible flow, but may affect the compressibility factor that is applied. Thom<sup>5.4</sup> and Göthert<sup>5.31</sup> recommend multiplying Equation (5.47) by  $\beta^{-2}$  to give

$$\epsilon_{\rm gr} = \frac{1}{4} \left( \frac{c}{n} \right) \frac{1}{\beta^2} C_{\rm D} . \qquad (5.43)$$

Allen and Vincenti<sup>5.3</sup> consider that the source representing the model wake should give the same pressure drop far downstream and not, as in Reference 5.4, the same drag as the model in the tunnel. Allen and Vincenti obtain a different compressibility factor which, for the ratio of specific heats  $\gamma = 1.4$ , gives

$$\xi_{\rm F} = \frac{1}{4} \left( \frac{c}{h} \right) \frac{1 + 0.4k^2}{\beta^2} C_{\rm D} . \qquad (5.49)$$

This compressibility factor lies between  $\beta^{-2}$  and  $\beta^{-3}$  (Fig. 5.12). Thompson<sup>5.6</sup> doubts whether this more complex factor is theoretically better than the simpler form, because the representation of the wake by a source is in any case difficult to justify rigorously. The use of Equation (5.49) sometimes leads to better agreement with experiment at high subsonic Mach numbers than does Equation (5.48), since in practice the compressibility factor  $\beta^{-2}$  often underestimates the effect of stream Mach number; for this reason alone, Equation (5.49) would seem to be preferable.

Goldstein<sup>5-7</sup> has also considered the wake-blockage effect in incompressible flow and concludes that the source-image system of representing the wake and walls is valid if  $C_0$  is small, even though c/h is comparatively large. In such cases, however, terms up to  $(c/b)^3$  may be required. Although no direct expression for a wake-blockage factor is contained in Reference 5.7, the correction is implicit in the equations derived for finding the stream velocity from measurements made at the tunnel wall.

Equation (5.48) is the first term of an equation due to Woods<sup>5.8</sup>, who gives a somewhat different approach both in allowing for stream compressibility and for representing the wake. This is replaced by a sting of thickness g attached to the trailing edge of the model and extending to infinity downstream. In an unbounded stream

$$s = \frac{1}{2} c c_{\rm p}$$
, (5.50)

and this remains true in the tunnel provided terms in  $C_D^2$  can be neglected. Woods shows that  $\epsilon_w$  may vary along the model chord and derives the expression, valid in the neighbourhood of the zerofoil,

$$\epsilon_{w}(\mathbf{x}) = \frac{1}{4} \left( \frac{c}{h} \right) \frac{1}{\beta^{2}} C_{p} \left[ 1 + \frac{\pi}{3\beta h} (\mathbf{x} - c) \right].$$
 (5.51)

\* Compare for example, the methods of References 5.3 and 5.4. Thom, in the latter paper, discusses the position of the mink and concludes that it need not be at infinity provided it is further away from the model than 5h; at smaller distances  $\epsilon_{\mu}$  is affected.

こう e" , 111,11 í , although the active with the solution of

. . . . . . . .

The largest value of  $\epsilon_{y}$  along the chord of the model occurs at the trailing edge (x = c) and is that given by Equation (5.48). Equation (5.51) may be used to calculate an approximate mean wake-blockage factor for a large model, for say, the correction of balance results. If the variation of  $\epsilon_{y}$  along the model is large this may indicate that the pressure field around the model is distorted and does not correspond to some free-air pressure field. The validity of using the local value of  $\epsilon_{y}$  to correct the measured pressure distribution then seems doubtful. As far as is known, the distortion of the model surface pressures due to the presence of a large wake has not been considered.

# 5.4.2 Three-Dimensional Models is a Rectangular Tunnel

Originally Glauert<sup>5.1</sup> suggested that his empirical wake-blockage factor  $\eta$  might be applied in a modified form for three-dimensional bodies, particularly solids of revolution. A new factor

$$\eta' = \eta^2 \tag{5.52}$$

was suggested,  $\eta$  being obtained from an equivalent two-dimensional body of the same fineness ratio. Equation (5.52) implies that the wake-blockage effect is far less important for small three-dimensional models than for aerofoils. Such arguments are less forceful for a large wing, and later authors have once more preferred to relate the wake-blockage factor to the model drag. As in two-dimensional flow, the wake is replaced by a source at the model position. The corresponding equation to (5.48) above is

$$f_{\rm g} = \frac{1}{4} \left( \frac{s}{bh} \right) \frac{1}{\beta^2} C_{\rm D} , \qquad (5.53)$$

where 5 is the model area on which the aerodynamic coefficients are based (e.g., the planform area of a wing). Alternatively, following Allen and Vincenti<sup>5.3</sup> and Herriot<sup>5.23</sup>, we have

$$\epsilon_{\rm w} = \frac{1}{4} \left( \frac{5}{bb} \right) \frac{1 + 0.40t^2}{\beta^2} C_{\rm p} \,. \tag{5.54}$$

These equations are usually held to apply to any three-dimensional model but some care is needed if this is a lifting wing; the drag is then partly due to the trailing-vortex system and partly to surface friction and incomplete pressure recovery over the alter part of the model. It seems reasonable to ignore the influence of the vortex drag on the wake blockage, so that in Equations (5.53) and (5.54).  $C_{\rm D0}$ , the zero-lift drag should be used. Such an approach is of course consistent with the general assumption that the blockage is independent of lift, provided the latter is not large. With separated flow, however, an alternative method is required and this is discussed in Chapter VII.

Methods so far discussed have largely ignored the actual structure of the wake. This matter was considered by Evans<sup>5.25</sup> who, starting from the momentum and continuity equations of the flow, showed that the form of the velocity distribution within the wake had little influence on the velocity increment experienced by the flow outside

er styre

the wake. The velocity inside the wake can therefore be considered as constant at  $\Omega$  times the uncorrected free-stream velocity U. The velocity increment sufficiently far downstream from the model (where the wake blockage will be higher than at the model position) may be written as

 $(\delta u)_{wake} = \left[\frac{1}{4}\left(\frac{s}{bb}\right)\frac{1}{\beta^2}C_{D}\right]2UW$ , (5.55)

where

$$\mathbf{W} = \frac{1}{\Omega} \left[ 1 + \frac{1+\Omega}{2} (\gamma - 1) \mathbf{M}^2 \right].$$

A source of strength Q placed on the tunnel axis at the model position gives an increment

$$(\delta u)_{ake} = \frac{Q}{\beta^2 bh} . \qquad (5.56)$$

Thus the source strength equivalent to Equation (3.55) is

$$Q_{\mu} = \frac{1}{2} \# SC_{\mu} U$$
 (5.57)

Now  $\Omega$  is about 0.9 in the region downstream of the model and hence as the stream Mach number increases from zero to unity W changes from 1.1 to about 1.5. Evans in fact suggests that a value of 1.4 is appropriate for most high-speed tunnel tests, if ho more exact information is available. Thom<sup>5.8</sup> and others, using Equation (5.46), way be regarded as putting W equal to unity in Equation (5.57).

The form of Equation (5.53) advocated by Evans then becomes

$$\epsilon_{\mathbf{w}} = \frac{1}{4} \left( \frac{s}{bb} \right) \left[ \frac{1 + 0.2(1 + \Omega) \mathbf{w}^2}{\Omega \beta^2} \right] \mathbf{C}_{\mathrm{p}} .$$
 (5.58)

If  $\Omega$  is taken as unity for all stream Mach numbers, this equation reduces to a form identical with Equation (5.54) above.

For a long body, Evans suggests that the source be placed at the model centre-ofvolume and the wake-blockage velocity increment calculated not only along the length of the body but along the tunnel walls as well. As Section 5.6 above, the latter information is valuable in providing a method of checking the validity of blockagecorrection theory. Although Reference 5.25 is concerned with a tunnel where b/h = 1.43, the method is of general application.

As in the case of the solid blockage, the wake blockage for a half-span model mounted directly on the side-wall of a rectangular tunnel should be calculated from the value appropriate to a complete model in a tunnel of twice the breadth. The wake blockage of struts supporting a model may scattimes be significant and these should be regarded as equivalent to half-wings mounted on the tunnel floor. It is usually sufficient to assume that the velocity increment calculated from present techniques may be applied at the actual model position.

A DESCRIPTION OF THE OWNER OWNER OF THE OWNER OWNE

ひょうちょうちょう いっちょう ちょうちょうちょう しんかんかく ちんいちょう ANGALINAN .....

#### 5.4.3 Closed Non-Sectangular Tunnels

Eatchelor<sup>5.21</sup> and Vincenti and Graham<sup>5.27</sup> discuss the appropriate wake-blockage factors for two-dimensional aerofoils spanning octagonal and circular tunnels respectively; they recommend, as for solid blockage (Sections 5.3.1 and 5.3.2), that a similar formula should be used to that for a rectangular tunnel, but with an equivalent height  $k_e$  in place of h. At the centre section it seems best to ignore corner fillets, to take  $h_e = 1.558R$  for a circular tunnel, and to use Figure 5.10 for an elliptical tunnel. When balance results on a full-span model have been obtained and require correction, it would seem best to put

$$\mathbf{h}_{\mathbf{e}} = \frac{\mathbf{C}}{\mathbf{b}}, \qquad (5.59)$$

as suggested in Equation (2.30) of Chapter II; this would give a slightly larger value  $h_e = 1.571E$  for a circular tunnel, or generally for an elliptical tunnel  $h_e = 0.785n$  in place of Figure 5.10. The relevant wake-blockage equation in all cases will be

$$\epsilon_{\rm H} = \frac{1}{4} \left( \frac{\rm c}{\rm h_e} \right) \left( \frac{1+0.4{\rm M}^2}{\beta^2} \right) C_{\rm D0} , \qquad (5.60)$$

where C<sub>D0</sub> is the zero-lift drag.

For bodies of revolution, wings and wing-body combinations, it is suggested that Equation (5.54) be used to determine the wake-blockage factor, but with the tunnel cross-sectional area C in place of bh. If the wake characteristics are known, Equation (5.58), similarly modified, may be employed.

#### 5.4.4 Wake Blockage Gradient

In the preceding sections, the discussion has mainly been concerned with the calculation of the effective velocity increment at the model position caused by the images of the source and sink, that simulate the walls of the tunnel. This velocity increment varies along the tunnel axis, and hence along the model length, but the calculated value at the source position usually represents an average increment over the model unless this is very long.

Unlike the solid-blockage velocity increment, the wake-blockage effect is not symmetrical about the model but increases continuously from nose to tail. The model is therefore subject to a longitudinal velocity gradient. This buoyancy effect imposes a drag force on the model which, since it would be absent in free air, must be corrected for. The method of doing this is discussed in Section 5.8.2.

## 5.5 BLOCKAGE FACTORS IN OPEN-JET TUNNELS

# 5.5.1 Solid Blockage for Models Spanning the Tunnel

For an open jet of rectangular shape the method of images may be used to represent the model and the jet boundaries. The doublet images are no longer of the same sign,

303

۳<sup>---</sup>--

but alternate with distance from the model. The induced velocity is therefore smaller at the model position than for a closed tunnel and of opposite sign, as shown by the equation (Refs. 5.1, 5.29)

$$\epsilon_{\mathbf{s}} = -\frac{\pi^2}{24} \left(\frac{\mathbf{c}}{\mathbf{h}}\right)^2 \left(\frac{\mathbf{t}}{\mathbf{c}}\right)^2 \lambda_2 , \qquad (5.61)$$

where  $\lambda_2$  is given in Figure 5.1. In magnitude this is half the blockage correction appropriate to a closed tunnel; accordingly the equations of Section 5.2.1 may be modified simply. Equation (5.9), for example, becomes

$$\epsilon_{\rm s} = -0.262 \left[ 1 + 1.2\beta \left(\frac{\rm t}{\rm c}\right) \right] \frac{\rm A}{\beta^3 \rm h^2} \quad . \tag{5.62}$$

In Reference 5.22, Section 7.21, Woods has considered the distribution of blockage factor along the axis of an open-jet tunnel due to the presence of a two-dimensional aerofoil. At the centre of the model the value obtained is identical to that given by the simple theories.

The blockage effect on an aerofoil spanning a circular jet seems to have received little consideration. By analogy with Equation (5.62) one possible assumption is that the blockage factor is half that of the corresponding closed tunnel, and of opposite sign, so that in place of Equation (5.34)

$$\epsilon_{\rm s} = -0.108 \left[ 1 + 1.2\beta \left(\frac{t}{c}\right) \right] \frac{A}{\beta^3 R^2} .$$
 (5.63)

A similar assumption may be made for elliptical working sections.

#### 5.5.2 Three-Dimensional Solid Blockage

The boundary conditions for a small model at the centre of a rectangular free jet may be represented by a doubly-infinite set of images, as for a closed tunnel. As in the two-dimensional case the signs of the doublets alternate, this time in rows and columns, so that the velocity induced at the model position is smaller than if the boundaries were closed. Lock<sup>5.29</sup> evaluated the blockage factor for a square jet; the equation is similar to (5.18) above

$$\epsilon_{\rm g} = \tau \left(\frac{\Lambda_{\rm m}}{{\rm h}^2}\right)^{3/2} \lambda_3 , \qquad (5.64)$$

but now  $\tau = -0.238$  compared with its value 0.809 for a closed square section. The magnitude of blockage effect in a square open jet is thus only about 0.29 of that present in a closed tunnel. In terms of the model volume, .ith allowance for compressibility effects, a simple equation for the square jet tunnel is

$$\epsilon_{\rm g} = -0.211 \frac{V}{\beta^3 h^3}.$$
 (5.65)

309

-62

The general case of a rectangular jet has only recently but considered, both at the NPL and by Wuest<sup>5.5?</sup>. In the former work, values of  $\tau$  were evaluated from the double summation in the notation of Equation (5.18)

$$\tau = \frac{1}{2\pi^{3/2}} \sum_{-\infty}^{\infty} (-1)^{2 + n} \left( \frac{bh}{a^2 b^2 + n^2 h^2} \right)^{3/2}$$
(5.66)

and are given, together with  $T(=\frac{1}{2}\sqrt{\pi}\tau)$  in the following table:

| b<br>h | 2, 00<br>0, 50 | 1.80<br>0.55 <sub>6</sub> | 1.66,<br>0.60 | 1.60<br>0.62 <sub>5</sub> | 1.40<br>6.71 <sub>4</sub> | 1.25<br>0.60 | 1.10<br>0.90 <sub>9</sub> | 1.0    |
|--------|----------------|---------------------------|---------------|---------------------------|---------------------------|--------------|---------------------------|--------|
| τ      | 0. 461         | -0.397                    | -0.357        | -0.339                    | -0. 289                   | -0.260       | -0.242                    | -0.238 |
| T      | -0.409         | -0.352                    | -0.316        | -9.300                    | ~0.256                    | -0.231       | -0.214                    | -0.211 |

The results are plotted in Figure 5.13. The percentage increase in the tunnel-shape factor in the range 1 < b/h < 2 is such larger than for similar closed rectangular tunnels (see Figure 5.4), but the actual change in value is this range is little different for the two types of boundary.

Whest has also considered the effects of displacing the model from the tunnel centre line.

The circular open jet containing a small model was discussed in Lock's original paper<sup>5.29</sup> and led to z similar equation to that for the closed tunnel but with a different numerical factor, and of opposite sign,

$$\epsilon_{\rm s} = -0.206 \left( \frac{\Lambda_{\rm s}}{C} \right)^{3/2} \lambda_{\rm s} , \qquad (5.67)$$

where  $\lambda_3$  is given in Figure 5.2. The ratio of solid blockage in open and closed circular tunnels is therefore ~0.26. a value close to that for square tunnels. A simple equation in terms of the stream Mach number, tunnel diameter, and medel volume is

$$\epsilon_{\rm g} = -0.0333 \left(\frac{\rm V}{\rm R^2}\right) \frac{1}{\beta^3}$$
 (5.68)

Cremer and Kolberg (Ref. 5.58; 1960) have considered the solid-blockage effects due to a body of revolution placed at incidence in a circular open jet. General expressions are given for the velocity increments induced in the vicinity of the model and for the deformation of the streamlines. These expressions are very complicated, however, and are not readily evaluated without mechanized computation.

Vandrey<sup>5,33</sup> has considered the solid-blockage correction for long Rankine ovoids in circular open jets and obtains an analogous expression to Equation (5.39)

$$\epsilon_{s}(x) = [I(\xi_{1}) - I(\xi_{2})] \frac{A_{s}}{c},$$
 (5.69)

where  $\xi_i$  and  $\xi_2$  are defined, and values of  $I(\xi)$  listed, in Table 5.I. It is interesting to note that the largest value of  $\epsilon_g(x)$  is reached at the centre of the body only if the body length is less than 6R. For longer bodies  $\epsilon_g(x)$  is larger near the ends of the body than at the centre and on a semi-infinite body this maximum is achieved at a distance 1.2R downstream from the nose of the body (Fig. 5.9). Fuchshuber (Ref. 5.59; 1961) gives a complete axisymmetrical mathematical solution and has evaluated the flow along the boundary of the jet. For body lengths greater than about 1.5R the maximum velocity increment is obtained at two positions fore and aft of the centre; these maxima sharpen and their positions approach the ends of the body as its length increases. Because of the difficulties of obtaining long stable jets it seems that these results will seldom be needed.

Many open-jet tunnels are of elliptical cross-section and some simple blockage correction may therefore be required. In the absence of more direct information it is suggested that an equation of the form

$$\epsilon_{\rm g} = \langle {\bf 1}_{\rm R} + 0.029 \rangle \left( \frac{1}{\rm C} \right)^{3/2} \frac{\rm V}{\beta^3} , \qquad (5.70)$$

may be used, where  $T_R$  is that for a rectangular open jet of breadth/neight ratio equal to m/n (Fig. 5.13) and  $C = \frac{1}{4}\pi mmn$ .

#### 5.5.3 Wake Blockage Effects

It is often stated that were interference effects are zero in open-jet tunnels, but this is true only in a restricted sense. If the model wake is represented by a source, the effect of the tunnel boundaries may be simulated by an array of images alternating in sign with the original. The velocity increment associated with the wake blockage due to the presence of these images is zero, because the alternately positive and negative infinite sets of images produce zero velocity increment at the model position and at infinity downstream or upstream. The velocity gradient along the model length is not zero, however, and is opposite in sign to, and smaller than, that in the corresponding closed tunnel. Thus for a small model one may conclude that  $\epsilon_{w} = 0$ , but for a long model a distribution of  $\epsilon_{w}$  corresponding to the velocity gradient, will exist. A correction must therefore be applied to the measured drag to allow for the wall-induced horizontal buoyancy. This is discussed in Section 5.8.2.

The distribution of the wake-blockage factor along the chord of a two-dimensional aerofoil spanning a two-dimensional open jet is discussed by Woods in Section 7.21 of Reference 5.22. At the mid-chord position the blockage factor becomes zero, though the gradient is finite.

## 5.6 ESTINATION OF BLOCKAGE FROM WALL MEASUREMENTS

In the preceding sections we have been concerned with the calculation of the blockage effects from the geometry of the model and tunnel, and from the state of the tunnel stream. The existing theory may also be used to estimate the incremental blockage velocities anywhere within the working section, for example, at the tunnel-speed hole. The changes in stream velocity induced by the model and wake and their images may also

#### 

be calculated at the tunnel well immediately opposite, say, the midpoint of the model. The ratio (Z) of the velocity induced at the model position by the images alone, to 12; total induced velocity at this wall position is important, since, if this is known on theoretical grounds, the blockage of the model may be found very simply from measurements at the wall. An approach of this kind is therefore most valuable and has special advantages. It is of course of greatest application to tunnels having closed boundaries and these will be mainly considered in relation to two-dimensional (Section 5 6.1) and three-dimensional (Section 5.6.2) models.

In discussing the methods of estimating blockage from well measurements it will be assumed that the model is at zero lift. If this is not the case, the lift effect on the wall pressures must be removed by averaging the pressures obtained above and below the model.

## 5.6.1 Two-Dimensional Aerofeil

For a small aerofoil placed at the centre of z closed rectangular tunnel the solidblockage velocity increment at the wall immediately above or below a non-lifting model is about 1.78° times that at the model position. The velocity increment at the wall deduced from pressure measurements will, however, include the increase in velocity due to the model itself, as well as that due to the images. The velocity ratio  $(Z_3)$  for this case is exactly 1/3; that is to say, the solid-blockage factor at the model will be given by

$$Z_{\rm s} = \frac{U\epsilon_{\rm s}}{(u-U)_{\rm max}} = \frac{1}{3}$$
 (5.71)

where  $(u - U)_{pax}$  is the maximum supervelocity at the tunnel well opposite the model and may be estimated from the well pressure distribution.

The value of  $Z_g$  is unaffected by compressibility of the tunnel stream (Refs. 5.4, 5.2S), but at high Mach numbers the velocity peak on the wall tends to move downstream from a position opposite the maximum thickness of the model. For large models  $Z_g$  may fall below 1/3; Mocds<sup>5.8</sup> has considered the relation between the distribution of  $\epsilon_g$  along the model and the velocity increments along the wall.

As the distance upstream or downstream of the model increases, the supervelocity at the wall due to the solid blockage becomes smaller and finally vanishes. At infinity downstream, however, a velocity increment will be recorded which is due entirely to the wake blockage and is found to be twice  $\epsilon_{\rm m} U$ . If  $(u - U)_{\infty}$  is estimated from the wall pressures well downstream, then

$$Z_{\mathbf{w}} = \frac{\vartheta \epsilon_{\mathbf{w}}}{(\mathbf{u} - \vartheta)_{\infty}} = \frac{1}{2}. \qquad (5.72)$$

AND ANY ANY ANY ANY

\* Thus<sup>5.4</sup> discusses the variation scross the tunnel and gives this ratio as  $3(n^2-4)/n^2$ . See also Reference 5.54.

Goldstein<sup>5.7</sup> gives a general expression for the velocity at the wall in incompressible flow

$$\left(\frac{u}{b}\right)_{wall} = 1 + \frac{\pi}{2} \frac{A}{h^2} \operatorname{sech}^2 \chi + \frac{1}{4} \left(\frac{c}{h}\right) C_{\mathrm{p}} [1 + \tanh \chi] , \qquad (5.73)$$

where

$$\chi = \frac{\pi(x-\frac{1}{2}z)}{h}.$$

In terms of the solid and wake blockage factors this expression becomes

$$\left(\frac{u}{u}\right)_{\text{well}} = 1 + (3 \operatorname{sech}^{2} \chi) \epsilon_{g} + (1 + \tanh \chi) \epsilon_{w}, \qquad (5.74)$$

thus enabling the solid and wake blockage to be estimated from measurements made at any two stations; one of these should preferably be close to the model and one downstream. Because tanh X rapidly approaches its asymptotic value as x increases, whilst sech<sup>2</sup>X decreases, the latter station need not be further downstream than about 1.5 tunnel heights in order to make an accurate estimate of  $\epsilon_{y}$ . This is of great convenience in view of the limited length of tunnel working sections. The solid-blockage contribution at this position will be less than 1% of the total.

The values of  $Z_{ij}$  and  $Z_{j}$  for an warofoil spanning a circular tunnel have only been discussed in a limited manner<sup>5.36</sup>; it seems likely that the methods of Reference 5.27 could be extended to cover this case.

# 5.6.2 Three-Dimensional Models

For a rectangular tunnel containing a three-dimensional model scall enough to be represented by a doublet, the value of  $\ddot{z}_{\rm g}$  is different on the roof and pide-walls, unless the tunnel cross-section is square. In an early paper Thom<sup>5.4</sup> showed that  $Z_{\rm g}$  was respectively 0.28 and 0.56 for the roof and side-walls of a tunnel with b/h = 1.43, when the model was represented by a simple doublet. For a wing composed of line sources and sinks, whose span is 0.6 of the tunnel breadth, these values become 0.36 and 5.33. A body of revolution in a square tunarl gives  $Z_{\rm g} = 0.43$ ; in a circular tunnel  $Z_{\rm g} = 0.45$  (Ref. 5.30).

Ther's work for a rectangular tunnel was later extended by Hensel (Ref. 5.40; 1951), who represented a finite-span wing by a distribution of line sources and sinks, and a body of revolution by a distribution of point sources and sinks. The finite swept wing however presents certain analytical difficulties, and discrete doublets were used in this case. Hensel's method is quite general and can be applied to any closed rectangular tunnel. The numerical computation is nonvever complicated, and in Reference 5.40 results are given only for b/h =  $\sqrt{2}$ . The effect of speep on  $Z_g$  for both the side-salls and rect is shown in Figures 5.14(a) and (b). Although  $Z_g$  is larger for swept than for unswept wings, it should be remembered that the actual wall velocity increments are smaller in the latter case, so that the trend in  $Z_g$  with sweep is not necessarily that of  $C_g$ . The value of  $Z_g$  at the roof for an unswept finite wing almost completely spanning the tunnel exceeds that for a two-dimensional wing. A similar effect occurs in the direct calculation of the solid blockage (see feature in Section 5.2,3).

For compressible flow Hensel recommends the use of an equivalent sweep angle  $\Lambda_e$  related to the geometric sweep angle  $\Lambda$  by

$$3 \tan \Lambda_e = \tan \Lambda; \qquad (5.75)$$

 $\Lambda_{a}$  can then be used in the charts or equations of Reference 5.40.

Hensel's analysis assumes that the body length or wing chord is small compared with the tunnel dimensions. This will often be the case for models tested in high-speed wind tunnels. Exceptionally long bodies of revolution may be investigated and then the method developed by Evans<sup>5.25</sup> seems preferable, though the calculations and experiments described in the reference are primarily concerned with a tunnel where b/h = 1.43. Evans shows that  $Z_g$  is not constant for large models but depends on the length of the model and the Nach number of the stream.

Hensel points out that, for larger wings, the velocity ratio should be related to the mean blockage incressent over the wing span and not to a local value on the centre section. This mean is comparatively easy to calculate for a straight wing, but for a sweptback planform each spanwise position lies in a different longitudinal position and this complicates the analysis. The effect of wing taper on both swept and unswept wings cannot readily be obtained by Hensel's approach, though he suggests that the effects of ignoring both wing taper and spanwise variation of  $Z_g$  tend to cancel each other. Evans considers that any wing, unless of most unorthodox shape, may be represented by a non-tapered swept wing of equal volume, mean angle of sweep and thickness/ chord ratio, but with a span  $2/3k_x$ , where  $k_x$  is the radius of gyration of the original wing about the x axis. Slender delta wings are best replaced by an equivalent body of revolution.

Evans also suggests an allowance for the effect of small supersonic flow regions adjacent to the model surface, an effect likely to be present at high subsonic stream speeds. This aspect is briefly discussed in Section 5.8.2. In general Reference 5.25 shows that very good agreement may be obtained between the calculated and observed wall pressures, and that the latter may be used reliably to estimate the solid-blockage factor.

The wake-blockage factor for small three-dimensional models may be found from the velocity increments far downstream in the same way as for a two-dimensional aerofoil by Equation (5.72).

The velocity-ratio method of obtaining blockage factors for models in circular closed tunnels has been discussed by Göthert<sup>5.30</sup> and Schwits<sup>5.36</sup>. As was mentioned earlier, the value of  $Z_g$  for a small wing or body is 0.45, which is close to the value for a square tunnel. For an unswept wing whose span is equal to the tunnel radius,  $Z_g$  equals 0.50.

References 5.35 and 5.59 both consider the case of an open-jet tunnel containing a body of revolution, though it seems unlikely that the velocity-ratio method will be used to find the solid-blockage factor in this type of tunnel.

# 5.7 BLOCKAGE COBBECTIONS AT VEBY HIGH SUBSONIC SPEEDS IN CLUSED TUNNELS

The effects of compressibility have been allowed for in the earlier sections by means of a small perturbation theory which essentially becomes less valid as flow Mach numbers approach unity. This means that the compressibility factors become unreliable as the tunnel Mach number reaches high subsonic values. Moreover, in such flows, supersonic velocities may be reached locally about the model and the associated shock waves violate the assumption that the flow is everywhere isentropic. Unfortunately it is just in this region of stream Nach number that the model cheracteristics show the greatest sensitivity and a precise value of equivalent free-air Mach number (and bence of the appropriate blockage correction) is important. The appricate of the blockage corrections will in any case be larger than at lower stream Mach numbers because of the form of the simple compressibility factors (Fig. 5.12), and also because the model drag used in the wake-blockage correction factor may increase very rapidly with N . The highest Mach number obtainable in a closed-wall tunnel is that achieved when the tunnel choked; no additional mass flow can then be passed through the working section by increasing the fan speed or tunnel power. The exact value of the choking Mach number depends on the geometry of the model and the tunnel, but a simple onedimensional consideration often gives a useful guide to this limiting speed (see Reference 5.56, for example).

The presence of a maximum Mach number, the clash between the increasing megnitude and importance of the blockage corrections and the uncertainties inherent in their determination, led to the development of wind tunnels having slotted or perforated walls. The checking effect was completely overcome and the blockage effects could be considerably reduced (see Chapter VI). The success of these ventilated working sections has been such us to reduce the demand for more accurate blockage corrections at high subsonic species. Nevertheless such corrections will be required occasionally, and a short discussion of some available techniques is justified.

## 5.7.1 Modification of Simple Comprensibility Theory

Experimental evidence suggests that by using the simple compressibility factors given by small perturbation theory the true compressibility effect on blockage is frequently underestimated; some modification of the factors may therefore be considered, or alternatively a new theoretical approach attempted. Blockage corrections applicable near the choking condition have been put forward by Thom and Jones (Ref. 5. 18; 1947: Ref. 5. 41; 1946); a noticeable feature is a new term in the solid-blockage relation proportional to  $\beta^{-5}$ . This method is not easy to apply in general, and perhaps the most valuable and simple suggestion was made by Svans<sup>5.25</sup>: that the corrected Mach number should be used in calculating the factors  $\beta^{-2}$  and  $\beta^{-3}$  of the simple theory. This implies, of course, that the final value of the corrected stream Mach number is reached by successive approximations; usually two or three iterations are sufficient. This technique is amply justified by the experimental evidence of Evans's paper, which considers both the calculated blockage and the use of the wall-velocity increments. Examples illustrating agreement between theory and experimental are given in Figure 5.15.

Drougge (Ref. 5. 14; 1951) has developed an approximation (due to Orwatitsch and valid near H=1) useful for analysing the transconic flow past an zerofoil, and from which a transonic similarity law relating the acrofoil pressure drag and a free-stream

915

「おおお」の「いった」」というないです。

an and the second second second

Mach number (M<sup>\*</sup>) based on the critical velocity of sound (a<sup>\*</sup>) may be derived. Drougge suggests that this law may be applied to the blockage problem at high transonic speeds, if tests results are available on two similar models of different sizes. An extrapolation to zero model size\* valid at transcuic speeds may then be made.

## 5.7.2 Regions of Local Supersonic Flow

The presence of regions of local supersonic flow on the model further complicates the problem of blockage corrections at high subsonic speeds. Thompson<sup>5.6</sup> considers that potential theory can still be used to calculate the velocity increment at the model position due to blockage from an array of simple doublet images, but that the theory is no longer valid for finding the strength of the equivalent doublet for the model. This suggests that the effect may be simulated by an increase in the effective volume of the model in the solid-blockage equations. Evans<sup>5.25</sup> in fact argues that the influence of the supersonic regions on the induced velocity at some distant point is similar to that produced by a local bulge on the model surface. The extra volume may be related roughly to the increase in drag coefficient ( $\delta C_D$ ) that occurs after the drag rise has begun. From an analysis of experimental results on three unswept wings in the RAE 10 ft x 7 ft High Speed Tunnel, Evans deduced an equation for high subsonic speeds which may be rewritten, in a form analogous to Equation (5.29) above,

$$\epsilon_{g} = T \left(\frac{1}{bb}\right)^{3/2} \frac{V}{\beta_{c}^{3}} \left[1 + 1.2\beta_{c}\left(\frac{t}{c}\right) + 14\delta C_{p}\right]. \quad (5.76)$$

where  $\beta_{\rm C} = (1 - M_{\rm C}^2)^{\frac{1}{2}}$  corresponds to the corrected Mach number  $M_{\rm C}$ . The actual value of the numerical factor in the last term is rather uncertain and is best determined directly for each tunnel and class of model. Shender bodies of revolution develop extensive supersonic flow fields only at test Mach numbers close to the choking condition and for this limited speed range an equation similar to Equation (5.23) above, but with  $\beta_{\rm C}$  in place of  $\beta$ , is probably sufficient.

The growth of the supersonic region and the lateral extension of shock waves will widen the wake behind the body, thus increasing the wake-blockage effect. The usual wake-blockage equations contain the model drag, which increases rapidly with stream Mach number in this region, but in certain cases they may seriously underestimate the true wake blockage<sup>5.42</sup>. The wall pressure downstream of the model is then more reliable. If the theoretical wake-blockage equations are used, the corrected value of the stream Mach number should be used in the compressibility factor.

The use of the wall-velocity increments to deduce solid and wake blockage is very valuable at these high subsonic speeds, because, though small perturbation theory may no longer apply at the model. it may still he valid at some distance away. Göthert<sup>5,39</sup> indeed considers that the velocity-ratio method may be employed up to the choking speed of the tunnel for this very reason.

\* Drougge also considers the more simple case of extrapolation to zero model size at moderate subsonic Mach numbers. The technique has also been discussed by Ackeret, Degen and Rott (Sef.3.43; 1950) for both straight and swept wings, and by Petersohn (Ref.5.50; 1948) for two-dimensional models spanning a circular tunnel.

# 5.8 USE OF BLOCKAGE FACTORS IN CORRECTING Measured quantities

We have so far been concerned with the estimation of the blockage factors  $\epsilon_{g}$  and  $\epsilon_{w}$  associated with the presence of the model and its wake within the tunnel. It is now necessary to consider how these factors are used to correct the measured quantities. Once more it is assumed that the model is small enough for the blockage effects to be additive; in addition the stream Mach number must not be so high as to invalidate the simple linear-perturbation theory.

The quantities to be corrected fall conveniently into two main groups; those, such as velocity and density, associated with the working fluid (Section 5.8.1), and the forces, moments and surface pressures measured on the model (Section 5.8.2).

#### 5.8.1 Corrections to Stream Quantities

It is one of the important assumptions of blockage theory that the model behaves within the tunnel as it would in free air at some speed slightly different from the nominal tunnel velocity. It is the purpose of the theory to establish means of estimating the necessary increment  $\Delta U_g$  which must be applied to the tunnel velocity U. The latter is of course that appropriate to the model position in the empty turnel and must itself contain any corrections required by the tunnel calibration. Thus the corrected stream velocity is given by

$$\mathbf{U}_{\mathbf{C}} = \mathbf{U} + \Delta \mathbf{U}_{\mathbf{a}} \,. \tag{5.77}$$

In terms of the solid and wake blockage factors this becomes

$$\mathbf{v}_{c} = (1 + \epsilon_{s} + \epsilon_{s})\mathbf{v} . \tag{5.78}$$

Usually, however, the tunnel speed is determined from the pressure at some reference hole on the tunnel wall upstreas of the model position. In some cases this pressure may itself be influenced by the presence of the model, so that the undisturbed freestream velocity will be unknown; the effects of both the model lift and the blockage must then be considered. The influence of the lift can, in principle, be removed by meraging the pressures on the top and bottom walls of the tunnel, or its magnitude may be calculated by replacing the model by a vortex of the appropriate strength. The solid-blockage influence on the tunnel-speed hole has been considered by Goldstein<sup>5.7</sup> and Thom<sup>5.4</sup> amongst others. The latter shows that, for a small twodimensional wing at zero lift in free air, a small negative velocity would be indoced at positions ahead of the wing and the magnitude of this perturbation decreases with distance from the wing. When the model is in a tunnel with closed rectangular boundaries the effect is counterbalanced by the influence of the images which induce a small positive velocity. Ahead of about 1.5 tunnel heights from the model centre the two effects almost cancel out, and a wall hole upstream of this position should similarly indicate a velocity very close to the postulated undisturbed free-stream value U. Goldstein shows that the effect of the wake blockage is negligible beyond shout one tunnel height upstress of the model.

Schultz<sup>5.36</sup> has also chasidered the corrections required in a rectangular tunnel, and his results are in general agreement with earlier authors. In addition, he

317

were constructed from

and the state of t

discusses the closed circular tunnel containing a body of revolution and shows that the correction is very small at more than one tunnel diameter ahead of the model. Open-jet tunnels of rectangular and circular cross-section are considered in References 5.36 and 5.59 and it is interesting to note that in these cases the velocity increments induced at a position on the edge of the jet corresponding to the tunnelspeed hole in a closed tunnel may be positive and thus of opposite sign to that induced at the model position. This effect is probably of little practical importance, since the stream velocity in open-jet tunnels is often determined from readings of wall pressure within the nozzle, a calibration having been made with the tunnel empty. If, however, a direct measurement of tunnel velocity is made in the stream ahead of the model, then this may be subject to blockage interference.

For compressible flow the stream velocity is of rather less importance than the stream Mach number H. The correction to H can be obtained very simply by differentiating the adiabatic isentropic flow equations. For  $\gamma = 1.4$  the correction to the nominal Mach number is given by

$$\Delta \mathbf{M} = (1 + 0.2\mathbf{M}^2) \mathbf{M} \in_{\mathbf{R}} . \tag{5.79}$$

and the set of the second s

Then

$$\mathbf{M}_{\mathbf{c}} = \mathbf{M} + \Delta \mathbf{M} , \qquad (5.80)$$

where  $\Delta N$  takes the sign of  $\epsilon_{\rm B}$ , which is positive for closed tunnels and negative for open tunnels.

Once the blockage factors are known, corrections may also be made to the stream static pressure (p), density ( $\rho$ ) and static temperature ( $\Theta$ ) on the assumption that the flow is isentropic and adiabatic. For  $\gamma = 1.4$ .

$$\Delta p_{\rm m} = -1.4 {\rm M}^2 p_{\rm m} \epsilon_{\rm R} , \qquad (5.81)$$

$$\Delta \rho = -\underline{\mathtt{M}}^2 \rho \epsilon_{\mathtt{R}} \,, \tag{5.82}$$

$$\Delta \Theta = -0.4 \Re^2 \Theta \epsilon_{\rm B} \,. \tag{5.83}$$

In incompressible flow, Equation (5 81) will be replaced by

$$\Delta \mathbf{p}_{m} = -\rho \mathbf{U}^{2} \boldsymbol{\epsilon}_{n} . \tag{5.84}$$

It follows that the correction to the stream kinetic pressure is given by

$$\Delta(\frac{1}{2}\rho \mathbf{U}^2) = (2 - \mathbf{N}^2)(\frac{1}{2}\rho \mathbf{U}^2)\epsilon_{\mathbf{R}}.$$
 (5.85)

The test Reynolds number  $(R_N)$  is also modified by the blockage effects. Usually such changes can be ignored, but a correction may be calculated from a knowledge of the changes in stream velocity, density and temperature. If a suitable, but simple, relationship between fluid viscosity and temperature is used, a simple correction equation may be obtained. For example, if it is assumed that the viscosity is proportion to  $\theta^{3/4}$ , then

$$\Delta R_{\rm N} = (1 - 0.7 {\rm M}^2) R_{\rm N} \epsilon_{\rm B} \,. \tag{5.86}$$

## 5.8.2 Corrections to Model Quantities

The lift interference discussed in Chapters II and III gives rise to corrections to the measured model incidence, forces and moments, but not to the stream velocity at the model position unless this is off the tunnel centre line. The correction to the stream velocity, because of the solid and wake blockage effects, has already been discussed, together with the change in stream kinetic pressure. In forming the non-dimensional coefficients from the measured forces and moments, use can be made directly of the corrected kinetic pressure; the resulting coefficients will then be corrected for blockage effects. Alternatively, it is sometimes convenient to form these coefficients using the uncorrected values of stream density and velocity and then to apply a correction to the coefficient itself. Thus if  $C_A$  is a typical non-dimensional coefficient, the corrected value will be

$$C_{A} + \Delta C_{A} = C_{A} [1 - (2 - N^{2})\epsilon_{B}],$$
 (5.87)

if squares of small quantities are neglected. Equation (5.87) is obtained by using Equation (5.85) above and may be applied to any measured force or moment coefficient, provided due regard is paid to the restrictions implicit in the derivation of  $\epsilon_{\rm R}$ .

An extra correction is required to the drag coefficient, because, as was mentioned in Sections 5.4.4 and 5.5.3, the images of the source representing the wake impose a longitudinal pressure gradient along the tunnel and hence a longitudinal buoyancy force on the model. The measured drag is higher than the free-air value in a closed tunnel, and lower in an open tunnel. The pressure gradient may conveniently be regarded as linear along the length of the model. The buoyancy force  $D_g$  may then be taken as the product of the pressure gradient and the effective volume of the model  $V_e$ , where this is the sum of the actual volume and a virtual volume  $V_{\psi}$  corresponding to the virtual mass of the body in the accelerated flow along the tunnel axis. Thus

$$\bar{v}_{g} = -\left(\frac{dp}{dx}\right)_{B} V_{e} = -\left(\frac{dp}{dx}\right)_{B} (V + V_{v}) , \qquad (5.88)$$

Now  $V_y$  may be related to the shape parameter  $\lambda_2$  or  $\lambda_3$  and to the doublet strength P representing the potential flow past the body (see References 5.1, 5.23, for example). Thus for a slender body of revolution in incompressible free

$$P = V_{a}U = \frac{1}{4}\pi\lambda_{a}t^{3}U. \qquad (5.89)$$

On the basis of linear theory the pressure gradient may be written as

$$\left(\frac{dp}{dx}\right)_{R} = -\rho \eta \frac{\partial \eta_{R}}{\partial x} = -\rho \eta^{2} \frac{\partial \epsilon_{R}}{\partial x}.$$
 (5.90)

If  $\Delta U_s$  denotes the solid-blockage effect of a doublet of strength P, then the gradient  $(\partial/\partial x) (\epsilon_y U)$  due to a source of strength Q is precisely  $Q(\Delta U_g)/P$  (References 5.37 and 5.38). Moreover the source strength used in the wake-blockage analysis is equal to  $D/\rho U$ . Thus

$$\left(\frac{d\mathbf{p}}{d\mathbf{x}}\right)_{\mathbf{g}} = -\frac{\mathbf{D}\mathbf{0}\boldsymbol{\epsilon}_{\mathbf{g}}}{\mathbf{P}} \,. \tag{5.91}$$

315

and a constrained to a plate and a constrained on a

With the substitution of  $V_e$  and (dp/dx) from Equations (5.89) and (5.91), Equation (5.88) gives a Simple expression for  $\cdots D_p$ , which can be rewritten non-dimensionally

as a correction to drag coefficient

 $(\Delta C_p)_g = -C_p \epsilon_s$  (5.92)

This equation is independent of corrections to the kinetic pressure if second-order terms in  $\epsilon_s$  are neglected. Goldstein<sup>5</sup> <sup>7</sup> derives a more complex relation for the wake-buoyancy correction in incompressible flor, but shows that Equation (5.92) is valid if the square of the drag coefficient is small enough to be neglected.

The influence of compressibility on the wake buoyancy correction has sometimes caused difficulty. Equation (5.88) is unaffected by simple compressibility theory, and some authors have argued that no compressibility factor is required in Equation (5.91) other than that inherent in  $\epsilon_s$  itself. Others<sup>5.3, 5.23</sup> consider that a compressibility correction to the source strength should be included, as in Equation (5.49), so that in place of Equation (5.92)

$$(\Delta C_{\rm p})_{\rm g} = -C_{\rm p} (1 \pm 0.4 \mu^2) \epsilon_{\rm g} \,. \tag{5.93}$$

Ludwieg<sup>5.37</sup>, moreover, points out that the virtual volume is also modified slightly by compressibility and Herriot<sup>5.23</sup> considers it worthwhile allowing for this in the drag correction. For most wind tunnels and models this particular effect will be small even at high Mach numbers.

The sign of the bucyancy force follows that of  $\epsilon_s$  and hence is different in open and closed tunnels. For a two-dimensional rectangular section the magnitude of the open-jet correction is half that of the closed tunnel: for three-dimensional models this ratio is about one-quarter.

The full correction to the measured drag coefficient is obtained by combining the corrections obtained in Equations (5.37) and (5.92) or (5.93). Thus, in two-dimensional flow, either

$$\Delta C_{\rm p} = -\left[\epsilon_{\rm s} + (2 - \mathbf{M}^2)\epsilon_{\rm p}\right] C_{\rm p} \tag{5.94}$$

or

 $\Delta C_{\rm p} = - \left[ (1+0.4M^2) \epsilon_{\rm s} + (2-M^2) \epsilon_{\rm B} \right] C_{\rm p} . \qquad (5.95)$ 

In three dimensions  $C_D$  is replaced by  $C_{D0}$  in Equations (5.92) and (5.93) and there is corresponding adjustment to the first terms in Equations (5.94) and (5.95).

Allen and Vincenti<sup>5.3</sup> rezark that it is not at once apparent that these equations apply to drag coefficients determined by the wake-traverse method<sup>5.46</sup> and, after discussing the matter, they suggest that, for normal-sized aerofoil models, the error in using Equation (5.95) is about the same order a. the experimental error. This view has sometimes been questioned. The matter is extremely complex and each term in the formulae used in calculating the zerofoil drag from the experimental data needs to be considered with great care. Moreover the physical significance of the formula must be clear; for example the final calculated drag from the form coefficient may be obtained

320

÷.

321

A CONTRACTOR OF THE OWNER

a and the contract of the second second second second second and the second second second second second second

in a way which allows for some wall-interference effects. In the opinion of the present author no wake-buoyancy correction should be applied to a drag coefficient obtained from wake-traverse measurements if these are reduced in the usual way.

The pressure coefficient  $C_p$  on the body will also require correction because of changes in both the static and kinetic pressures of the tunnel stream. The requisite correction may be written as

$$\Delta C_{p} = \frac{p - (p_{\omega} + \Delta p_{\omega})}{\frac{1}{2}\rho v^{2} + \Delta (\frac{1}{2}\rho v^{2})}$$
  
=  $[2 - (2 - H^{2})C_{p}]\epsilon_{B}$ . (5.96)

Such an expression assumes that the observed pressures are those which would be obtained in the corrected free-air conditions. That is to say, the use of Equation (5.96) is equivalent to the use of Equations (5.81) and (5.84) before the pressure is nondimensionalized. The actual distortion of the pressure distribution about the body because of buoyancy effects and other higher-order well corrections cannot be obtained in this manner. Methods such as the conformal mapping of References 5.15 and 5.16 or the higher-order calculations of Goldstein<sup>5.7</sup> must be used. Provided that the model is not too large, the stream Mach number is not near unity and the model incidence is moderate, then the distortion will be small.

# 5.9 SUMMARY OF PRINCIPAL BLOCKAGE FORMULAE

It is convenient to list together the simple formulae which may be most frequently required in calculating the blockage effects and which seem most satisfactory in use. Only those which give explicit answers will be presented; for more complicated expressions requiring extensive calculations the text of this report and the appropriate references should be consulted.

#### 5.9.1. Closed Rectangular Tunnel

For two-dimensional aerofoils

 $\epsilon_{\rm s} = \frac{\pi^2}{12} \left(\frac{\rm c}{\rm h}\right)^2 \left(\frac{\rm t}{\rm c}\right)^2 \frac{1}{\beta^3} \lambda_2$  (see Pigure 5.1)

OT

$$f_s = \frac{\pi}{6} \left[ 1 + 1.2\beta \left( \frac{t}{c} \right) \right] \frac{A}{\beta^3 h^2}$$

නාර

$$\varepsilon_{w} = \frac{1}{4} \left( \frac{c}{h} \right) \left( \frac{1+0.4M^2}{\beta^2} \right) C_{\mathrm{p}} .$$

If incidence effects sust be allowed for, put

$$\epsilon_{\mathbf{s}}(\alpha) = \epsilon_{\mathbf{s}}\left[1+1.1\beta\left(\frac{c}{t}\right)\alpha^{2}\right].$$

. چي

If the model is at a distance d from the wall, put

$$\epsilon_{s}(d) = \epsilon_{s}\left[1 + \frac{3}{4}\cot^{2}\frac{\pi d}{h}\right].$$

For three-dimensional models

$$\epsilon_{\rm g} = \tau \left(\frac{A_{\rm m}}{\rm bh}\right)^{3/2} \frac{1}{\beta^3} \lambda$$

$$\epsilon_{\rm g} = T \left(\frac{1}{\rm bh}\right)^{3/2} \frac{1}{\beta^3} VG$$

or

where for wings

$$\lambda = \lambda_2 \quad (\text{Fig. 5.1})$$

$$G = 1 + 1.2\beta \left(\frac{t}{c}\right)$$

and for bodies

$$\left. \begin{array}{l} \lambda = \lambda_3 \quad (\text{Fig. 5.2}) \\ \\ \mathbf{G} = \mathbf{1} + \mathbf{0.4}\beta/\mathbf{f} \end{array} \right\} \,.$$

Values of  $\tau (=2T/\sqrt{\pi})$  and T depend on ratios b/h and 2s/b (Figures 5.5 and 5.6). For large wing-body combinations, it may be better to add the separate contributions of the wing and body.

$$\epsilon_{w} = \frac{1}{4} \left( \frac{S}{bh} \right) \left( \frac{1 + 0.4M^2}{\beta^2} \right) C_{DO} .$$

If the wake structure is known, use Equation (5.58) instead. For lifting wings, use zero-lift drag coefficient.

## 5.9.2 Closed Non-Rectangular Tunnels

For two-dimensional aerofoils, use the solid-blockage equations for the rectangular tunnel of an equivalent height  $(h_e)$ . The value of  $h_e$  depends on both the tunnel shape and whether a correction is required at the mid-span position or as an average over the complete span.

| T            | Value of h <sub>e</sub> |         |  |  |
|--------------|-------------------------|---------|--|--|
| tunnet Snape | At Mid-Span             | Average |  |  |
| Circalar     | 1.558R                  | 1.571R  |  |  |
| Octagonal    | h (at centre)           | C/b     |  |  |
| Elliptical   | Fig. 5. 10              | 0.785n  |  |  |

11 A A

A CARLES AND A CARLES

For the wake-blockage factor, in all cases use

$$\epsilon_{\rm w} = \frac{1}{4} \left( \frac{\rm c}{\rm h_e} \right) \left( \frac{\rm I + 0.4M^2}{\beta^2} \right) \rm C_{\rm D} ,$$

where  $h_e$  is the value used in calculating the solid-blockage. For lifting wings, use zero-lift drag coefficient.

For three-dimensional models, use equations for rectangular tunnel, with appropriate values of  $\tau$  or T and with bh replaced by the tunnel cross-sectional area C.

#### 5.9.3 Open-Jet Tunnels

For a two-dimensional aerofoil in a rectangular jet

 $\epsilon_{-} = 0$ .

 $\epsilon_{\rm s} = -\frac{\pi^2}{24} \left(\frac{\rm c}{\rm h}\right)^2 \left(\frac{\rm t}{\rm c}\right)^2 \frac{1}{\beta^3} \lambda_2 \qquad (\text{see Pigure 5.1})$   $\epsilon_{\rm s} = -\frac{\pi}{12} \left[1 + 1.2\beta \left(\frac{\rm t}{\rm c}\right)\right] \frac{\rm A}{\beta^3 {\rm i} {\rm n}^2}$ 

or

and

For three-dimensional models, use equations for  $\epsilon_{\rm g}$  for closed rectangular tunnels, but with values of T and  $\tau$  given in the table in Section 5.5.2. For non-rectangular tunnels, replace bh in these equations by C: for a circular tunnel T = -0.182 and  $\tau = -0.206$ : for an elliptical tunnel take  $T_{\rm R} + 0.029$  (or  $\tau_{\rm R} + 0.032$ ) where  $T_{\rm R}$  (or  $\tau_{\rm R}$ ) corresponds to a rectangular open jet of breadth/height ratio equal to m/n. The wake-blockage factor

 $\epsilon_{w} = 0$ .

5.9.4 Corrections to Stream Quantities

For  $\gamma = 1.4$ .

$$\Delta \mathbf{U} = (\epsilon_{g} + \epsilon_{w})\mathbf{U} = \epsilon_{g}\mathbf{U}$$

$$\Delta \mathbf{M} = (1 + 0.2\mathbf{N}^{2})\mathbf{M}\epsilon_{g}$$

$$\Delta p_{\infty} = -1.4\mathbf{M}^{2}p_{\omega}\epsilon_{g}$$

$$\Delta \rho = -\mathbf{M}^{2}\rho\epsilon_{g}$$

$$\Delta \theta = -0.4\mathbf{M}^{2}\theta\epsilon_{g}$$

$$\Delta (\frac{1}{2}\rho \mathbf{U}^{2}) = (2 - \mathbf{M}^{2})(\frac{1}{2}\rho \mathbf{U}^{2})\epsilon_{g}$$

$$\Delta \mathbf{R}_{N} = (1 - 0.7\mathbf{M}^{2})\mathbf{R}_{g}\epsilon_{g}$$

For incompressible flow

 $\Delta p_{\rm m} = -\rho v^2 \epsilon_{\rm B} \, .$
ると言語がい

į

inder and the restaurant for a substantian and the and the above of a distributed with the substantian and second

THE MARKEN AND AVAILANE TO THE SAME AND A SAME

5.9.5 Corrections to Bodel Quantities

 $\Delta C_{A} \simeq - (2 - \varkappa^{2}) C_{A} \epsilon_{B} ,$ 

where  $C_{\rm A}$  is a non-dimensional force or moment coefficient other than drag coefficient  $C_{\rm p}$  .

For tro-dimensional aerofoils

$$\Delta C_{\rm D} = - \left[ (1+0 \, 4 {\rm M}^2) \epsilon_{\rm s} + (2-{\rm M}^2) \epsilon_{\rm B} \right] C_{\rm D} \, .$$

For three-dimensional models

$$\Delta C_{\rm p} = -(1+0.4\mu^2)\epsilon_{\rm s}C_{\rm po} - (2-\mu^2)\epsilon_{\rm s}C_{\rm p} \ .$$

In both cases

5.2 Borden, A.

5.6 Thompson, J.S.

$$\Delta C_{\rm p} = [2 - (2 - \underline{M}^2)C_{\rm p}]\epsilon_{\rm b}$$

These corrections only apply if the uncorrected stream quantities (Section 5.9.4) are used in obtaining the non-dimensional coefficients.

#### REFERENCES

| 5.1 | Glauert, H. | Wind Tunnel Isterference on Wings, Bodies and Airscreus. |
|-----|-------------|--|
|     |             | ARC R & H 1566, 1933.                                    |
|     |             |  |

- Wall Corrections for Flow about Two- and Three-Dimensional Symmetrical Bodies in Rectangular Channels of Infinite and Finite Lengths. David W.Taylor Model Basin, Report 864, 1954.
- 5.3 Allen, H.J. Vincenti, W.G. NACK Report 782, 1944.

5.4 Those, A. Blockage Corrections in a Closed High-Speed Tunnel. ARC R & M 2033, 1943.

5.5 Young, A.D.Blockage Corrections in a Cloved Rectangular Tunnel.Squire, H.B.ABC R & M 1984, 1945.

Present Methods of Applying Blockage Corrections in a Closed Rectangular High Speed Vind Tunnel. ARC Report 11,325, 1948. (Unpublished.)

5.7 Goldstein, S. Tro-Disensional Hind-Tunnel Interference. ARC B & H 1902. 1945.

. ~ 2

Automatical and a second second

いたち いちんちんちちち ちょうちょう ちょうちょう ちんちんちちち しょう

÷.

the Hard and when the second of

新

| 5.3 ¥00ds, L.C.                         | The Second Order Terms in Two-Dimensional Tunnel Blockage.<br>Aeron, Quart. Vol.IV, 1954, pp.361-372.  |
|---|--|
| 5.9 Klunker, E.B.<br>Harder, K.C.       | Or the Second-Order Tunnel-Wall-Constriction Corrections<br>in Two-Dimensional Compressible Flow. NACA Tech. Note<br>2350, 1951.   |
| 5.10 Thoy, 5.                           | The Method of Influence Factors in Arithmetical Solutions<br>of Vertain Field Problems. Part I. ARC R & # 2440, 1946.  |
| 5.11 Klanfer, L.                        | The Method of Influence Factors in Arithmetical Solutions<br>of Certain Field Problems. Part II ARC R & M 2440, 1947.  |
| 5.12 Emmons, H.W.                       | Flow of a Compressible Fluid Past a Symmetrical Airfoil in<br>a Wind Tunnel and in Free Air. NACA Tech. Note 1746, 1946.   |
| 5.13 Epstein, H.T.<br>Albers, L.D.      | The Effects of Compressibility on the Two-Dimensional<br>Subsonic Wind-Tunnel Constriction Correction. J. merg.<br>Sci. Vol.15, 1948, pp.144-150.  |
| 5.14 Drougge, G.                        | Note on: Wall Interference in Two-Dimensional Flow at<br>Subsonic and Transonic Speeds. FPA (Sweden) Report 40,<br>1951.   |
| 5.15 Perl, T.<br>Moses, H.E.            | Velocity Distributions on Symmetrical Airfoils in Closed<br>Twanels by Conformal Mapping. NACA Tech. Note 1642, 1948.  |
| 5.16 Noses, H.E.                        | Velocity Distributions on Arbitrery Airfoils in Closed<br>Tunnels by Conformal Mapping. NACA Tech. Note 1899, 1949.  |
| 5.17 Whitehead, L.G.                    | Two-Dimensional Hind-Twnnel Interference. ARC R & M 2802,<br>1950.   |
| 5.18 Thos, A.                           | Tunnel Wall Effect from Mass Flow Consilerations. ARC R &<br>M 2442, 1967.   |
| 5.19 vor Doznhoff, A.E.<br>Abbott, P.T. | The Langley Two-Dimensional Low-Turbulence Pressure Tannel.<br>NACA Tech. Note 1283, 1947.   |
| 5.20 Pranke, A.<br>Weinig, F.           | Die Korrektur der Anströmgeschwindigkeit und des<br>Anströmminkels in einem Hochgeschwindigkeitskonal zit<br>geschlossener Keßetrecke infolge der Verdröngungsströmung<br>in Tragflögelmodellen. Deutsche Luftfahrtforschung<br>Forschungebericht 1171, 1939. (Translated am MAP R & T<br>259, ARC Report 10.268.) |
| 5.21 Batchelor, C.K.                    | Interference on Wings, Bodies and Airscrews in a Closed<br>Tunnel of Octagonal Section. Repuit ACA-5 (Australia),<br>1944.   |
| 5.22 Woods, L.C.                        | The Theory of Subsonic Plane Flow. Combridge University<br>Press, 1961.  |

-

1

| 326                                 |  |
|-------------------------------------|--|
| 5.23 Herriot, J.G.                  | Blockage Corrections for Three-Dimensional-Flow Closed-<br>Throat Wind Tunnels, with Consideration of the Effect of<br>Compressibility. NACA Report 995, 1950.   |
| 5.24 Vasy, G.S.                     | The Drag of Streamline Bodies of Revolution in a Wind<br>Tunnel. Acta Tech. Acad. Sci. Hung. (Hungary) Vol.18,<br>1957, pp.317-322.  |
| 5.25 Zreas, J.Y.G.                  | Corrections to ¥elocity for Wall Constraint in any 10 x 7<br>Rectangular Subsonic Wind Tunnel. ARC R & M 2662, 1949.   |
| 5.26 Lamb, H.                       | On the Effect of the Walls of an Experimental Tank on the Resistance of a Model. ARC R & M 1010, 1926.   |
| 5.27 Vincenti, W.G.<br>Graham, D.J. | The Effect of Wall Interference upon the Aerodynamic<br>Characteristics of an Airfoil Spanning a Closed-Throat<br>Circular Wind-Tunnel. NACA Report 849, 1946.   |
| 5.28 Watson, G.N.                   | The Use of a Series of Bessel Functions in Problems<br>Connected with Cylindrical Wind-Tunnels. Proc. Roy.<br>Soc. Series A. Vol.130, 1930, pp.29-37.  |
| 5.29 Lock, C.N.H.                   | The Interference of a Wind Tunnel on a Symmetrical Body.<br>ARC R & M 1275, 1929.  |
| 5.30 Gothert, B.                    | Windkanalkorrekturen bei halte Unterschallgeschwindig-<br>keiten unter besonderer Berücksichtigung des geschlassenen<br>Kreiskanals. Deutsche Luftfahrtforschung Forschungs-<br>bericht 1216, 1940. (Translated as NACA Tech. Maso, 1300.) |
| 5.31 Göthert, B.                    | Findkanalkorrekturen bei hohen Unterschallgeschwindig-<br>keiten. Lilienthal-Gesellschaft für Luftfahrtforschung,<br>Bericht 127, 1940, pp.114-12 <sup>4</sup>   |
| 5.32 Wieselsberger, C.              | Windkonalkorrekturen bei kompreszibler Strömung.<br>Lilienthal-Gesellschaft für Luftfahrtforschung, Bericht<br>127, 1940, pp.3-7. (Translated as MAP R & T 379, ABC<br>Report 11,199.)   |
| 5.33 Vendrey, F.                    | Theoretical Corrections for Long Rankine Ovoids in a<br>Closed or Open Tunnel of Circular Cross-Section. ARC<br>Report 14.239, 1951. (Ungublished.)  |
| 5.34 Lock, C.N.H.<br>Johansen, P.C. | Wind Tunnel Interference on Streamline Bodies. ARC<br>R & M 1451, 1931.  |
| 5.35 Fage, A.                       | On the Two-Dimensional Flow Past a Body of Symmetrical<br>Cross-Section Hounted in a Channel of Finite Breadth.  |

CALLER STREET

• •

5.38 Schmitz, G. Beitrag zur Geschwindigkeitskorrektur in Windkamälen bei koopressibler Unterschallströmung. Z. Flugwins, Vol.5, 1957, pp. 169-172. 5.37 Ludwieg, H. Widerstandskorrektur in Hachgeschwindigkeitskanälen. Deutsche Luftfahrtforschung Forschungsbericht 1955, 1944. (Translation in RACA Tech. Memo. 1163.) 5.38 Taylor, G.I. The Force Acting on a Body Placed in a Curved and Converging Stream of Fluid. ABC R & W 1166, 1928. 5.39 von Baranoff, A. Zur Frage der Kanalkorrektur bei kompressibler Unterschallströmung. Beutsche Luftfahrtforschung Porschungsbericht 1272, 1940. (Translated as NACA Tech. Memo. 1162.) Rectangular-Wind-Tunnel Blocking Corrections Using the 5.40 Hensel, R.W. Velocity-Ratio Method. NACA Tech. Note 2372, 1951. 5.41 Thes, A. Tunnel Blockage Near the Choking Condition. ARC R & M Jones, M. 2385, 1946. 5.42 Mair, W.A. The Effect of Model Size on Measurements in the RAE High Gemble, H.E. Speed Tunnel. Drag of Two-Dimensional Symmetrical Aerofoils at Zero Incidence. ABC R & H 2527, 1944. 3.43 Ackeret, J. Untersuchunzen an gepfeilten und ungepfeilten Flügeln bei hohen Unterschallgeschwindigkeiten. Z. ang. Math. Phys. et al. (ZAMF) Vol. 1, 1950, pp. 32-42. 5.44 Goldstein, S. The Linear Perturbation Theory of Compressible Flow, with Young, A.D. Applications to Wind-Tunneî Interference. ARC R & M 1909, 1943. 5.45 Taien, H.S.

113 mill descent

Lees, L.

5.46 Lock, C.N.H.

5.47 Adeeson, J.S.

5.48 Katzoff, S.

et al.

5.49 Vandrey, P.

et el.

The Glauert-Frandtl Approximation for Subscribe Flows of a Cospressible Fluid. J. asro Sci., Vol. 12, 1945. pp. 173-187, 202.

Determination of Profile Drog at High Speeds by a Pitot Traverse Method. ARC B & N 1971, 1980.

An Experimental Investigation of Wind-Tunnel Interference in the RAE 5 ft Open-Jet Circular Tunnel. ABC R & # 1897. 1941.

Linear Theory of Boundary Effects in Open Wind Tunnels with Finite Jet Lengths. NACA Report 976, 1950.

CONTRACTOR OF THE OWNER

Der Diseneinfluss auf die Nindkanalkorrekturen bei ebener Strömung. Jahlvuch der Deutschen Luftfahrtforschung, 1942, pp. 1785-1793.

**ストランオルビルにたいたいいのかか**な

AND SUPERING THE CONTRACTOR OF A CARDAN

BEERIN SALAREN KERENDER STATEREN DE TER TER TER

zerena picetric

| 5.51 Loos, H.G.                      | Tunnel-Wall Interference for Aerofoils and Aerofoil-<br>Fuselage Combinations in a Tunnel with an Octagonal<br>Section for Incompressible Flow. NLL (Netherlands) Report<br>A 1024, 1951.                    |
|--------------------------------------|--|
| 5.52 Knechtel, E.D.                  | Experimental Investigation of Two-Dimensional Tunnel-Wall<br>Interference as High Subsonic Speeds. NACA Tech. Note<br>3027, 1953.  |
| 5.53 Suzuki, K.                      | On the Subsonic Flow of a Compressible Fluid past a<br>Circular Cylinder Between Parallel Mails. Proc. Phys.<br>Math. Soc. Japan, Vol.23, 1941, pp.336-346.  |
| 5.54 Barbieux, J.                    | Contribution à l'Étude de l'Effet de Paroi en Écaulement<br>Plan Incompressible. Publ. Sci. Tech. Min. Air France<br>304, 1955.  |
| 5.55 von Kármán, T.<br>Burgers, J.M. | General Aerodynamic Theory - Perfect Fluids. Airfoils<br>and Airfoil Systems of Finite Span. "Aerodynamic 'theory"<br>(ed. W.F.Durand) Vol.II, Div. E, Chap. IV, Sect. 44,<br>Julius Springer, Berlin, 1935. |
| 5.56 Pankhurst, R.C.<br>Kolder, D.W. | Wind Tunnel Technique, Chap. 8, pp.396-399. Pitzan,<br>London, 1952. (Reprinted in 1965.)  |
| 5.57 Wuest, W.                       | Verdrängungkorrekturen für rechteckige Windkomäle bei<br>verschiedenen Strahlbegrenzungen und bei exzentrischer<br>Lage des Modells. Z. Plugwiss, Vol.9, 1961, pp.15-39<br>and Errata p.352.                 |
| 5.53 Cramer, H.<br>Kolberg, F.       | Windkanaikorrekturen augestellter Körper bei kompressibler<br>Unterschailströmung. Z. ang. Math. Mech. (ZAME) Vol.40,<br>1960, pp.65-74.   |
|                                      |  |

On Maximum Model Size for Wind Tunnel Investigations at High Subsonic Mach Numbers. FPA (Sweden) Report 23, 1948.

528

5.50 Petersohn, E.

5.59 Fuchshuber, R. Étude Mathématique de l'Influence des Dimensions Finies de la Veine d'Air autour d'un Modèle dans les Souffleries pour 16 Domaine de l'Écoulement Compressible Subsonique. Publ. Sci. Tech. Min. Air France 376, 1961.

ACTOR AND A CARD A COMPANY

;

ale and a serie with the series of the serie

AND A SOLO AND A

#### ADDITIONAL REFERENCES FOR CHAPTER V

Abdurahiman, P.V. Two-Dimensional Compressible Flow past a Solid Body Symmetrically Placed in a Channel. ANC B & M 2443, 1947. Bateman, T.E.B. A Comparison of Results in the ARA Transonic Tunnel on a Haines, A.B. Scall and a Large Model of a Slender Wing. ARC R & M 3287, 1961. Berndt, S.B. Approximate Calculation of the Influence of Wall Boundary Layers upon the Blockage Interference in a High Speed Wind Tunnel. FFA (Sweden) Report 45, 1952. Fuchshuber, R. Über die Berechnung randbedingter Geschwindigkeitskorrekturen bei Hindkonälen mit geschlossener und offener Meßstrecke im Bereich kompressibler Unterschallströmung. Dissertation Tech. Hoch., Aachen, 1956. Katzoff, S. Determination of Jet-Boundary Corrections to Cowling-Flap-Finz, R.S. Outlet Pressures by an Electrical Analogy Method. NACL. ARE 4823 (NACA/TIB/602), 1944. Sanders, J. Well Interference in Wind Tunnels of Closed Rectangular Section. NRC (Canada) Report AR-7, 1949. Pounder, J.E. Thos, A. Tunnel-Wall Effect on an Aerofoil at Subsonic Speeds. Klanfer, L. ARC R & M 2857, 1951. Whitehead, L.G. Teso-Dimensional Wind-Tunnel Interference. ARC R & M 2802, 1959. Compressible Subsonic Flow in Two-Dimensional Channels. Woods, L.C. The Direct and Indirect Problems. ARC Report 14,838, 1952. (Unpublished.) Woods, L.C. Compressible Subsonic Flow in Two-Dimensional Channels. Aeron. Quart. Vol.6, 1955, pp. 205-220, 254-276. Wright, B.H. Comparison of Two-Dimensional Air Flows about an NACA 0012 Donaldson, C. du P. Airfoil of 1-Inch Chord at Zero Lift in Open and Closed 3-Inch Jets and Corrections for Jet-Boundary Interference.

NACA Tech. Note 1955, 1946.

| TAB | LE | 5. | I |
|-----|----|----|---|
|-----|----|----|---|

Functions Giving the Distribution of Solid Blockage along Bodies in Circular Tunnels (Ref. 5.33)

| н      | -I  | ξ  | K  | I   |
|--------|---|--|--|---|
| 0.000  | 0.000   | 2.0  | <b>J. 439</b>  | 0.048   |
| C. 040 | 0.010   | 2.2  | 0.449  | 0.043   |
| 0.089  | 0.020   | 2.4  | 0.458  | 0.038   |
| 0.119  | 0.029   | 2.6  | 0.463  | 0.034   |
| 0. 155 | 0.038   | 2.8  | 0.468  | 0. 030  |
| 0. 190 | 0.045   | 3. Û   | 0.472  | 0.027   |
| 0.220  | 0.052   | 3.2  | 0.476  | 0.024   |
| 0.250  | 0.057   | 3.4  | 0.478  | 0.021   |
| 0.276  | 0.060   | 3.6  | Q. 481   | 0.019   |
| 0.300  | 0.063   | 3.8  | 0.483  | 0.017   |
| 0.322  | 0.064   | 4.0  | 0.485  | 0.015   |
| 0.342  | 0.054   | 4.2  | 0.486  | 0.014   |
| 0.359  | 0.054   | 4.4  | 0.487  | 0.013   |
| 0.374  | 0.063   | 4.6  | 0.488  | 0.012   |
| 0.387  | 0.062   | 4.8  | 0.489  | 0.011   |
| 0.399  | 0.060   | 5.0  | 0.490  | 0.010   |
| 0.409  | 0.058   | 6.0  | 0.493  | 0.097   |
| 0.418  | 0.055   | 7.0  | 0.495  | 0.005   |
| 0.426  | 0.052   | 8.0  | 0.496  | 0.094   |
| 0. 433 | 0.050   | 9.0  | 0.497  | 6.003   |
| 0.439  | 0.048   | 10.9   | 0.498  | 0.002   |
|        | H<br>0.000<br>C.040<br>0.080<br>0.115<br>0.155<br>0.190<br>0.220<br>0.250<br>0.276<br>0.309<br>0.322<br>0.342<br>0.359<br>0.374<br>0.359<br>0.374<br>0.387<br>0.387<br>0.387<br>0.387<br>0.387<br>0.429<br>0.418<br>0.426<br>0.433<br>0.439 | H         -I           0.000         0.000           0.040         0.619           0.089         0.020           0.119         0.029           0.155         0.038           0.190         0.045           0.220         0.052           0.250         0.057           0.250         0.063           0.322         0.064           0.342         0.064           0.359         0.054           0.359         0.054           0.374         0.063           0.387         0.062           0.399         0.060           0.403         0.052           0.418         0.055           0.426         0.052           0.433         0.050           0.439         0.048 | H $-I$ $\xi$ 0.0000.0002.00.0400.6192.20.0800.0202.40.1190.0292.60.1550.0382.80.1900.0453.00.2200.0523.20.2500.0573.40.2760.0603.60.3220.0644.00.3420.0644.20.3590.0644.40.3740.0634.60.3870.0624.80.3930.0505.00.4180.0557.00.4330.0509.00.4390.04810.9 | H $-I$ $\xi$ H0.0000.0002.0 $\overline{0}.439$ 0.0400.6102.2 $9.449$ 0.0800.0202.4 $9.458$ 0.1190.0292.6 $0.463$ 0.1550.0382.8 $0.469$ 0.1900.0453.6 $0.472$ 0.220 $9.052$ 3.2 $0.476$ 0.250 $0.057$ $3.4$ $0.478$ 0.276 $0.060$ $3.6$ $0.481$ 0.322 $0.064$ $4.0$ $0.485$ $0.342$ $0.064$ $4.2$ $0.486$ $0.359$ $0.062$ $4.8$ $0.487$ $0.387$ $0.062$ $4.8$ $0.489$ $0.399$ $0.052$ $5.0$ $0.490$ $0.426$ $0.052$ $8.0$ $0.495$ $0.433$ $0.050$ $9.0$ $0.497$ $0.439$ $0.048$ $10.9$ $0.458$ |

For closed and open circular tunnels respectively

$$\epsilon_{g}(x) = [\Re(\xi_{1}) - \Re(\xi_{2})] \frac{A_{g}}{C}$$
 (5.39)

snd

330

$$\epsilon_{s}(\mathbf{x}) = [I(\xi_{1}) - I(\xi_{2})] \frac{A_{s}}{C}.$$
 (5.59)

where

$$\xi_1 = \frac{x + \frac{1}{2}x}{R}$$
$$\xi_2 = \frac{x - \frac{1}{2}x}{R}$$

x is measured from the centre of the body, and for slender bodies x = 1 (Ref. 5.2). Note that H(-x) = -H(x), I(-x) = -I(x).

in we converted the E



1

Contra States







331





記録の

Fig.5.3 Tunnel-shape parameter for small models





Orthone and the



;

and the second s

-----

75.

i

ころのないの



¢...\*

1.184 · 3m

f

•

THE SHARE 



1- 1-2-1 -1

1

1

į

-

ö

335

¢. ×

100 A. S.

i

ï

;

:

•

÷



The second second

WE COM

N TELEVIS

.

336

Alexandre al

man and the second s

----

- - -----

٦,

こうう 金融しの こういっていた



Fig. 5.13 Tunnel-shape parameters for small models in open rectangular tunnels

.-

337

-

A STARTAN AND A CONTRACT OF A STARTAN





A State of the sta



Based on Fig. 5 of Ref. 5.40



20000. -----

339

14. ·

2



Fig. 5.15 Examples of calculated and experimental Mach number increments at side-wall of 10 ft × 7 ft tunnel

CONTRACTOR OF

340

ŧ

CHAPTER VI

# WALL INTERFERENCE IN TUNNELS WITH VENTILATED WALLS

by

E.W.E.Rogers

k

Aerodynamics Division, National Physical Laboratory, Teddington, Niddlesex, England

20.000

# BLANK PAGE

# CONTENTS

light Juny Holiza

とれてたいまで

1.11

|       |   | Page |
|-------|---|------|
| LIST  | OF TABLES   | 345  |
| LIST  | OF FIGURES  | 345  |
| NOTAI | NOI   | 348  |
| 6. í  | INTRODUCTION  | 351  |
| 6.2   | BOUNDARY CONDITIONS FOR WALLS WITH LONGITUDINAL SLOTS   | 353  |
| 6.3   | TWO-DIMENSIGNAL BLOCKAGE IN TUNNELS WITH                |      |
|       | LONGITUDINAL SLOTS                                      | 357  |
|       | 6.3.1 Solid Blockage                                    | 357  |
|       | 6.3.2 Longitudinal Pressure Gradient Due to Solid       |      |
|       | Blockage  | 360  |
|       | 6.3.3 Wake Blockage                                     | 361  |
| 6.4   | THREE-DIMENSIONAL BLOCKAGE IN TUNNELS WITH LONGITUDINAL |      |
|       | SLOTS   | 362  |
|       | 6.4.1 Circular Tunnel                                   | 362  |
|       | 6.4.2 Rectangular Tunnel with Slotted Roof and Floor    | 365  |
|       | 6.4.3 Rectangular Tunnel with Slots on Each Wall        | 367  |
| 6.5   | LIFT INTERFERENCE IN TUNNELS WITH LONGITUDINAL SLOTS    | 369  |
|       | 6.5.1 Two-Dimensional Tunnel                            | 369  |
|       | 6.5.2 Circular Tunnel                                   | 372  |
|       | 6.5.3 Rectangular Tunnel with Slotted Roof and Floor    | 375  |
|       | 6.5.4 Rectangular Tunnel with Slots on Each Wall        | 378  |
| 6.6   | SOME GENERAL REMARKS ON INTERFERENCE EFFECTS IN         |      |
|       | TURNELS WITH LONGITUDINAL SLOTS                         | 379  |
| S.7   | THE PERFORATED WALL                                     | 380  |
| 6.8   | TWO-DIMENSIONAL BLOCKAGE IN TUNNELS WITH PERFORATED     |      |
|       | WALLS   | 383  |
|       | 6.8.1 Solid Blockage                                    | 383  |
|       | 6.8.2 Wake Blockage                                     | 384  |
| 6.9   | THREE-DIMENSIONAL BLOCKAGE IN TUNNELS WITH PERFORATED   |      |
|       | VALLS   | 385  |
|       | 6.9.1 Circular Tunnel                                   | 385  |
|       | 6.9.2 Rectangular Tunnel                                | 386  |



\*\*\*

4

ż,

٤

ì

CENTRAL N

|       |  | Page  |
|-------|--|-------|
| 6.10  | LIFT INTERFERENCE IN TUNNELS WITH PERFORATED WALLS | 386   |
|       | 6.10.1 Two-Dimensional Tunnel                      | 386   |
|       | 6.10.2 Circular Tunnel                             | 387   |
|       | 6.10.3 Rectangular Tunnel                          | 388   |
| 6.11  | APPLICATION OF CORRECTION FACTORS                  | 389   |
| REFER | ences  | 391   |
| ADDIT | IONAL REFERENCES                                   | 395   |
| TABLE |  | , 397 |
| FIGUR | ES   | 398   |

LIST OF TABLES

# Page

397

Table 6.1 Summary of Results

.

### LIST OF FIGURES

| Fig. 6. 1  | Notation and geometry for longitudinal slots   | 398 |
|------------|--|-----|
| Fig. 6.2   | Solid-blockage factors for two-dimensional aerofoil in a slotted tunnel  | 399 |
| Fig. 6.3   | Values of slot and porosity parameters required to give zero<br>solid blockage for a two-dimensional aerofoil                              | 400 |
| Fig. 6.4   | Effect of aerofoil size on solid-blockage factor in ideal slotted tunnel   | 401 |
| Fig. 6.5   | Effect of model size on slot parameter for zero blockage $(F_0)$ in ideal slotted tunnel   | 402 |
| Pig.6.6    | Distribution of solid-blockage factor along two-dimensional ideal slotted tunnel   | 403 |
| Fig.6.7    | Longitudinal pressure gradient at the model associated with solid blockage   | 404 |
| Fig.6.8    | Solid-blockage factor for small three-dimensional model in circular slotted tunnel   | 405 |
| Fig.6.9    | Values of slot and porosity parameters required to give zero<br>solid blockage for a small three-dimensional model in a<br>circular tunnel | 406 |
| Fig. 6. 10 | Solid-blockage factor for a small model in rectangular<br>tunnels with ideal slotted real and floor  | 407 |
| Pig. 5. 11 | Solid-blockage factor for large wings in rectangular tunnels<br>with ideal slotted roof and floor  | 408 |
| Fig. 6. 12 | Estimated solid-blockage factor for small model in square<br>tunnel with ideal slotted walls   | 409 |
| Fig.6.13   | Collapse of existing results for solid-blockage interference<br>in ideal slotted tunnels   | 410 |

٢



1

Ś

All and the second

and the second

のないで、「「「「」」」である。

アレビンビン

.

うち かんないかん ちかい うい

3

| Fig.6.14   | Influence of porosity parameter on lift interference for small<br>two-dimensional aerofoil in a slotted tunnel                                  | 411 |
|------------|---|-----|
| Fig.6.15   | Lift interference for two-dimensional aerofoil in an ideal slotted tunnel   | 412 |
| Pig. 6. 16 | Values of slot and porosity parameters required to give zero<br>upwash interference for a small three-dimensional model in a<br>circular tunnel | 413 |
| Fig.6.17   | Lift-interference parameter for circular and square ideal slott, tunnels  | 414 |
| Fig. 6. 18 | Lift-interference parameter for models in a square ideal slotted tunnel with solid side-walls   | 415 |
| Fig. 6. 19 | Lift-interference garameter for models in idea "lotted<br>tunnel with solid side-walls  | 416 |
| Fig. 6.20  | Streamline-curvature correction in ideal slotted tunnel w.tl.<br>solid side-walls   | 417 |
| Fig. 6.21  | Influence of viscosity on lift-interference parameter for<br>models in square slotted tunnel with solid side-walls                              | 418 |
| Fig. 6. 22 | Influence of viscosity on lift-interference parameter for<br>models in slotted tunnel with solid side-walls                                     | 419 |
| Fig. £. 23 | Values of slot and porosity parameters required to give zero<br>lift interference in slotted rectangular tunnels with solid<br>side-walls       | 420 |
| Pig. 6, 24 | Effect on lift interference of discrete slots in the roof<br>end floor  | 421 |
| Fig. 6. 25 | Lift-interference parameter for small wings in rectangular<br>tunnels with ideal slotted walks  | 422 |
| Fig. 6.26  | Variation of blockage with wall perosity in two-dimensional perforated tunnel   | 423 |
| Fig. 6. 27 | Distribution of solid-blockage factor along two-dimensional perforated tunnel   | 424 |
| Pig. 6. 28 | Blockage factor for small model in perforated circular tunnel   | 425 |
| Fig.6 29   | Lift interference in a two-dimensional perforated tunnel  | 425 |
|            |   |     |

•

·· ASSAULTS AND ····

A STATE AND A STAT

Page

347 Page ł

•

\* \*\* \*\*

| Fig.6.30 | Lift interference for a small model in a perforated circular tunnel  | 426 |
|----------|--|-----|
| Fig.6.31 | Lift-interference parameter for small wing in special types<br>of perforated turnel                          | 427 |
| Fig.6.32 | Comparison of exact and approximate values of $\delta_0^{}$ for closed and open rectangular tunnels          | 428 |
| Fig.6.33 | Estimated lift-interference parameter for small wing in perforated rectangular tunnels of different porosity | 429 |

ï,

:

14

SI MANAGARAN

NOTATION

width of slot (Fig. 6. 1) A cross-sectional area of aerofoil A A<sub>e</sub> equivalent value of A (Section 6.3.2) ۸<sub>n</sub> coefficient in Equation (6.67) breadth of tunnel b chord of aerofoil C ē geometric mean chord of wing Ē serodynamic sean chord of wing in Equation (6.64) C cross-sectional area of tuanel drag coefficient =  $(drag)/\frac{i}{z}\rho U^2S$ C<sub>D</sub> lift coefficient =  $(lift)/\frac{1}{2}\rho U^2 S$ C<sub>L</sub> = (pitching moment about quarter-chord axis) $\frac{1}{2}\rho U^2 S \tilde{c}$ C, periodic spacing of slots (Fig.6.1) đ thickness of slat (Fig. 6.1) f \$ non-dimensional slot parameter (Fig. 6. 1) value of P giving zero interference F, h height of tunnel I<sub>0</sub>,I<sub>1</sub> modified Bessel functions (Ref. 6.27) Integrands in Equations (6.18), (6.23), (6.28), (6.51) IA, B. C. D Integrands in Equations (6.52), (6.58), (6.84), (6.95) I<sub>2,7,6,8</sub> geometric slot parameter in Equation (6.2) K modified Bessel functions (Ref.6.27) K<sub>0</sub>, K<sub>1</sub> distance between source and sink ( $\simeq$  c) l Mach number of undisturbed stream ч

Ē

N

p

P<sub>c</sub>

P

σ

r

R

s

S

Т

U

70

v

V<sub>e</sub>

₩<sub>£</sub>.

x

y

z

ä

β

δ

**S**1

outward normal distance from tunnel boundary number of equally spaced slots in a single wall pressure pressure in plenum chamber porosity parameter defined by Equation (6.6) variable of integration radial distance from axis of circular tunnel radius of circular tunnel seti-span of wing reference area of model shape parameter of tunnel (Chapter V) velocity of undisturbed stream velocity normal to perforated wall volume of model equivalent volume of model (Chapter V) interference upwash velocity distance downstream of model spanwise distance from axis of tunnel distance upwards from model incidence of serofoil (in radians)  $= (1 - \underline{M}^2)^{\frac{1}{2}}$ list interference parameter for a small model lift interference parameter associated with streamline curvature

Sp pressure drop across a slot

 $\Delta$  prefix denoting increment due to wall correction

 $\epsilon$  blockage factor,  $\Delta U/U$  (Section 6.3)

349

10-10-17-Y

.

| 350                           |   |   |
|-------------------------------|---|---|
| ζ                             | = | $\tan^{-1}(\beta/P)$  |
| θ                             |   | cylindrical co-ordinate   |
| μ                             | = | $\cot^{-1}(\beta/P)$  |
| ρ                             |   | density of undisturbed stream                                     |
| σ                             | = | 2s/b  |
| Φ                             |   | perturbation velocity potential                                   |
| Φ <sub>r</sub>                | = | $\Phi(row)$ in a tunnel of infinite height                        |
| $\Phi_{\mathbf{s}}$           | = | $(\Phi - \Phi_r)$ , contribution from slotted roof and fluor      |
| Ψ                             |   | solid-blockage parameter for slotted tunnels in Equation (6.46)   |
| Ω <sub>s</sub> Ω <sub>w</sub> |   | blockage-factor ratio in Equations (6.16), (6.24)                 |
| Ω'                            |   | limit of $\Omega_{\mathbf{g}}$ as slotted walls become fully open |

# Subscripts

ALCONOM: N

ALLER .......

ì

.

ł

ļ

Ę

| 8 | denotes effect of blockage.                     |
|---|---|
| c | denotes closed turnel.                          |
| Ľ | denotes effect of velocity gradient (buoyancy). |
| i | denotes quantity induced by tunnel walls.       |
| • | denctes mode? alone.                            |
| 3 | denotes solid blockage.                         |

A STATE OF THE OWNER OF THE OWNER

STATION .

denotes wake blackage.

#### WALL INTERFERENCE IN TUNNELS WITH VENTILATED WALLS

#### E.W.E.Rogers

#### **G.1** INTRODUCTION

and the second states and second states in

The use of wind tunnels with partially-open, or ventilated, walls for transonic investigations is now widespread, and such tunnels are inevitably exployed for tests at subsonic speeds, a fixed wall geometry usually being retained throughout the speed range. Ventilated walls may be of several different types, but frequently the openings consist of longitudinal slots or circular perforations; these allow the air to pass between the working section and the surrounding plenum chamber or tank. The ventilated walls may constitute the complete tunnel boundary or, alternatively, the side-walls of the tunnel may be solid, with glass windows, thus easing the problems of viewing or supporting the model.

The boundaries of the ventilated wind tunnel influence the flow about the model in a similar manner to the boundaries of the more conventional closed or open-jet tunnel. Indeed, these two types may often be regarded as the extreme limits within which the amount of wall ventilation may be varied. Fortunately, many of the wall-interference effects are of opposite sign in the two extreme geometries, and hence by using a partially-open tunnel it is possible to reduce, or even eliminate, these effects. Such considerations sometimes determine the design of the wall geometry to be used; frequently, however, the need to reduce shock-wave reflection effects at transonic and low supersonic speeds may be more important and become the basis of the tunnel design 5.32. Indeed for tunnels with perforated walls, where shock-wave reflection can be considerably reduced by careful choice of wall geometry, some degree of wall interference may have to be accepted, because the criteria for wave-reflection elimination and zero subsonic wall interference will protably be incompatible.

In general it is not possible to remove all types of subscnic wall interference by choosing a particular wall geometry; corrections are therefore required to many of the measured quantities, but very often these are not applied. One important reason for this apparent neglect is that the necessary corrections are frequently very small, partly because the model may be small in relation to the tunnel, and partly because the chosen wall geometry may well lie close to that needed to eliminate the particular interference effect. In addition, the theoretical potential-flow approach to the problems of ventilated-wall interference is sometimes felt to be less soundly based than when applied to tunnels with uniform boundaries, particularly since the effects of viscosity on the flow through the ventilations may be of considerable significance, yet difficult to estimate. The approach developed by, say, Baldwin et al. (Ref.6.1; 1954), in which the general effects of viscosity (particularly on the flow through longitudinal slots) may be allowed for, is of great value in meeting this type of criticism. The analysis is inevitably made more complicated, however, and knowledge

351

A CONTRACTOR OF A CONTRACTOR

ş

こうちょう

of the porosity characteristics of the wall is required. This can often only be found by direct experiment.

Though much analysis has been undertaken in the general field of ventilated windtunnel interference, the supporting experimental checks on the theory (as opposed to tests demonstrating that certain interference effects are small) are at present insufficient to confirm or modify much of the theoretical work. Theory, however, does form a useful and general frame of reference against which the tunnel operator must often make his own assessment.

Interest in the use of mixed boundaries to minimize wall interference was aroused as early as 1931, when Theodorsen<sup>6.2</sup> discussed the interference experienced by a small lifting wing in tunnels having a combination of open and closed boundaries, and suggested that constraint effects could be reduced considerably by using these unconventional types of working section. These ideas were subsequently extended by many authors<sup>6.3-6.12</sup>, both lift and blockage interference being considered. A few experimental tests were also made<sup>6.4,6.10,6.11</sup>. The blockage factors appropriate to rectangular tunnels having one or more completely open walls have recently been given by Wuest (Ref.6.50; 1961) who used an electronic computer and by this means was able to calculate the effects of displacing the model from the tunnel axis.

In terms of present-day tunnel configurations, these papers may be regarded as dealing with working sections having one or two longitudinal slots of considerable width. There are, however, certain disadvantages in using a small number of slots. For example, the flow in the working section may deteriorate due to jet instability<sup>6.2</sup>, and the flow field in the neighbourhood of the model will tend to be rather distorted. The power requirements of the tunnel may also be large. For such reasons, working sections having multiple longitudinal slots have been adopted and much of the modern analysis has been concerned with this type of boundary.

The use of perforated walls to produce a partly-open boundary seems to have arisen from considering the characteristics of percus walls, where the flow through the wall corresponds to a slow viscous motion, and in which the pressure drop across the wall is proportional to the mass flow passing through it.

It is convenient therefore to divide the discussion of wall interference into two sections, dealing with tunnels having (a) longitudinal slots and (b) perforations. This is largely a matter of convenience since, as Maeder and Wood (Ref. 6.13; 1956) show so elegantly, both types of ventilation are particular cases of a more general boundary condition; the proposed arrangement however does correspond to the major division in wind-tunnel construction, even though there are certain tunnels with walls which in effect combine slots and perforations.

Historically, and perhaps also practically, the elimination of solid-blockage effects has been considered of more importance than the removal of lift-interference effects, probably because it was felt that the test Mach number could then be accurately determined, particularly at high subsonic speeds. It is proposed therefore to discuss blockage interference initially and subsequently to consider lift interference, particularly the value likely to be present in a tunnel designed for zeroblockage conditions. It will be assumed that the lift and blockage corrections are independent and additive, and hence that it is sufficient to calculate the blockage

effects of zero model lift. A summary of the results to be presented in this chapter is included at its end, as Table 6.1.

#### 6.2 BOUNDARY CONDITIONS FOR WALLS WITH LONGITUDINAL SLOTS

The first attempt to calculate the lift interference experienced by a small wing in a total having multiple longitudinal slots was made in 1940 by Pistolesi<sup>6,18</sup> for a circular funnel. Similar results were obtained in 1953 by Matthews<sup>6,15</sup> in an independent analysis, which was more detailed than the earlier work and allowed for variations in slot configuration, wing span and loading. The blockage of a slender body in a slotted circular wind tunnel was discussed by Wright and Ward (Ref. 6.16; 1948), again using the exact boundary conditions. An extension of this method to a two-dimensional aerofoil was made in 1950 by Tomlinson<sup>6,51</sup>.

It proved difficult however to apply these methods in practice due to the complexity associated with the mixed nature of the boundary; for example, to satisfy the correct coundary conditions for a source-sink doublet placed on the axis of a circular-slotted tunnel, an infinite set of linear simultaneous equations arises in the calculation of the interference functions. Even with certain simplifying assumptions, extensive numerical work is often required. Bifficulties of this type led to the consideration of a simpler, though less exact, wall condition by an approach attributed in Reference 5.17 to Busemenn. This assumed that the real slotted wall could be replaced by an equivalent homogeneous boundary whose influence near the model would be very similar to that of the real wall. Since this influence may be imagined as a weighted mean of the effects due to the slots and the solid slats, it seems that the equivalent wall would give a close representation of the flow near the tunnel centre. provided that the dimensions of the tunnel were large compared with the distance between adjacent slots. The calculations and discussions of Davis and Moore (Ref. 6. 17; 1953) confirm the validity of this argument, and show that the simpler analysis based on the equivalent boundary gives, in general, correction effects which are in very close agreement with those calculated by the earlier exact methods.

The linearized boundary condition of the equivalent wall has been derived by more than one method. (See References 5.1, 5.13, 6.17 to 6.20). In most cases the crossflow approaching a single slot and slat is considered, and this leads to a simple homogeneous boundary condition for subsenic flow, which is assumed to apply along the entire length of the tunnel working section and is thus independent of x:

$$\frac{\partial \Phi}{\partial x} + \mathbf{K} \frac{\partial^2 \Phi}{\partial x \partial n} = 0, \qquad (6.1)$$

where  $\Phi$  is the perturbation velocity potential, x is measured in the stream direction and n along the outward normal to the wall surface. The parameter X, having the dimensions of length, is defined by

$$K = \frac{d}{\pi} \log_e \operatorname{cosec} \frac{\pi z}{2d}.$$
 (6.2)

353

Name and a state of the state

Here, the slats are assumed to be of small thickness normal to the cross-flow direction, d is the periodic spacing of the slots, and a is the slot width (Fig.6.1); the open area ratio a/d must be small, d heing a small fraction of the tunnel dimensions. Equation (6.1) should be compared with the exact conditions, represented usually by

$$\frac{\partial \Phi}{\partial n} = 0 \quad \text{at the slats} \\ \frac{\partial \Phi}{\partial x} = 0 \quad \text{at the slots}$$
(6.3)

The boundary condition used by Woods in Sections 7.30 to 7.32 of Reference 6.19 (1961) is somewhat different and overcomes certain difficulties involved in the assumption that the calculations are being made for a position whose distance from the wall is large compared with the slot spacing. This alternative boundary condition would seem to be of most value when only a few widely spaced slots are present. Goethert (Kef. 6.18; 1957) in a more exact analysis of the cross-flow near the slots shows that Equation (6.2) may be regarded as the first term of a power series in the open-area ratio a/d; unless this ratio is small, higher-order terms may be significant.

Wood<sup>6.57</sup> discusses the validity of the linearized boundary condition in Equation (6.1) with particular emphasis on the large transverse velocities which can occur in the neighbourhood of narrow slots when large models are tested, and on the significance of the displacement of the free boundary from the contour of the slotted wall. These considerations lead to the non-linear homogeneous boundary condition

$$\frac{\partial \Phi}{\partial \mathbf{x}} + \left(\frac{\pi + 2}{\pi}\right)^2 \frac{\mathrm{d}^2}{2\mathrm{U}\mathbf{a}^2} \left(\frac{\partial \Phi}{\partial \mathbf{n}}\right)^2 = 0 , \qquad (6.4)$$

which implies a similarity between the transverse flow in the slot and a twodimensional Helmholtz jet. Reference 6.57 includes a measured velocity distribution along the wall, that appears to justify Wood's non-linear calculations for a nonlifting two-dimensional aerofoil. Further applications of this non-linear boundary condition have not so far been made, and the analysis in the present chapter is based on linear boundary conditions.

So far the flow has been assumed inviscid, so that there is no pressure drop across the slot boundary, apart from that associated with changes in fluid velocity. In practice, the effects of viscosity may well be of importance, particularly if the slot width is small compared with the wall boundary-layer thickness. Equation (6.1) then needs to be modified. Baldwin et al.<sup>6.1</sup> do this by adopting the homogeneous boundary condition for a porous wall<sup>6.21,6.44</sup>, in which the average velocity normal to the wall is assumed to be proportional to the pressure drop ( $\delta p$ ) through the wall (see Section 6.7). This is a linearized approximation to the viscous flow through a porous medium and leads to the boundary condition

$$\frac{\partial \Phi}{\partial x} + \frac{1}{P} \frac{\partial \Phi}{\partial x} = 0. \qquad (6.5)$$

Core-Classics

•

大きななないたいないのでなってきます。

where P is a porosity parameter defined by

$$\delta p = \frac{\rho_0}{P} \frac{\partial \Phi}{\partial n} . \qquad (6.6)$$

Here U is the free-stream velocity,  $\partial \Phi/\partial n$  the velocity normal to the wall, and  $\rho$  the fluid density. As mentioned in Section 6.7, P may be determined experimentally for any given wall by measuring  $\delta p$  for various values of  $\rho(\partial \Phi/\partial n)$ .

It is assumed therefore that, for a slotted wall having viscous effects within the slots, a boundary condition

$$\frac{\partial \Phi}{\partial x} + \frac{x}{\kappa} \frac{\partial^2 \Phi}{\partial x \partial n} + \frac{1}{P} \frac{\partial \Phi}{\partial n} = 0$$
 (6.7)

may be used; alternatively, if the porosity parameter is found not to be linear with  $\partial \Phi/\partial n$ , a more complex condition may be required (Appendix B of Reference 6.1). Solutions based on Equation (6.7) will include the following special cases:

> the closed wall  $(K \to \infty \text{ or } P = 0)$ , the open jet  $(K = 0 \text{ and } P \to \infty)$ , the perforated wall (K = 0), the invisciá slotted wall  $(P \to \infty)$ .

The perforated wall will be discussed in Sections 6.7 to 6.10. The last case will henceforward be referred to as an "ideal" slotted wall.

Alternatively the effects of viscosity in the slot region may be represented by considering the slots to be covered with a porous or perforated material; a more complicated boundary condition is then required  $^{6.13}$ . In some wind tunnels perforated cover plates have been fixed across the slots, so that the pressure drop can be increased artificially to match the local outflow characteristics in free air.

Much of the existing analysis for interference effects in slotted-wall tunnels is based on an inviscid boundary condition similar to Equation (6.1), and it is now possible to estimate the required corrections for a wide range of models and tunnel cross-sections for this ideal slotted wall. Some of the experimental checks that have been made in References 6.23 and 6.24 suggest however that the ideal slotted wall condition  $P \rightarrow \infty$  may not always be achieved; indeed for typical working-section configurations P may be of order unity and not greatly influenced by the stream Mach number. Much more information is required on this point, and it is here perhaps sufficient to point out that the ideal slotted-wall correction factors may sometimes need to be applied with caution; fortunately interference corrections have been evaluated in several cases for the more general boundary condition (6.7), and these are discussed below. The assumption of an ideal slotted wall, however, is valuable in enabling a general picture to be obtained of wall-interference effects in slotted tunnels, especially the trends with slot geometry and model size.

It will also be assumed (unless stated to the contrary) that the slotted working section is long compared with the tunnel height or breadth and that the slots are

AND THE REAL PROPERTY AND THE REAL PROPERTY OF A DESCRIPTION OF A DESCRIPR

355

ل المالية من مالية المالية من مالية م

uniformly spaced, have sharp edges, small thickness and no longitudinal taper. Further it will be assumed that the model is mounted centrally in the working section.

• • •

In certain cases, the restrictions on slot geometry may be overcome. For example, if the walls have tapered slots, Equation (6.7) may be used as the boundary condition but P, instead of representing the viscous effects in the slot flow, may now be related to the slot taper by

$$\frac{1}{P} = \frac{dK}{dx}, \qquad (6.8)$$

K being given by Equation (6.2); this matter is discussed further in Reference 5.1.

If the slats have a thickness f, a boundary condition of similar form to Equation (6.1) may be derived, but with a somewhat different definition of the parameter K (Ref. 6.52; 1957). Unfortunately, as f tends to zero this new definition of K becomes inconsistent with Equation (6.2) above. The differences are quite large and seem to arise from the way in which the slats are represented in the analysis; by doublets, when the slats have finite thickness and, in effect, by sources for the case of very thin slats. Since in practice most wind-tunnel slats are thin compared with their width, the definition of Equation (6.2) seems a reasonable one to use in such cases.

The discrete boundary conditions at a slotted wall may be satisfied if the rheoelectric analogy is employed; in this the electric potential in an electrolytic tank may be identified with the interference velocity potential, the tank being contoured to represent the tunnel section. In the exact analogical model the solid portions of the wall are insulators, whilst the slots are made of conducting material, so that a wall consisting only of a few slots and slats may readily be represented. If required, the linear homogeneous boundary condition of Equation (6.5) can also be represented by a simple analogy (Tirumaless, Ref. 6.55; 1959). The actual profile and incidence of a two-dimensional aerofoil may be simulated in the tank, a considerable advantage if the model is a large one, and this espect is discussed in detail in Reference 6.55.

At Birmingham University, Rushton (Ref. 6.60; 1965) has developed an electrical analogue computer to study slotted-wall interference effects. The computer consists of a rectangular array of resistors which form, in effect, a model of the wing and tunnel cross-section. The equations of the electrical network are made identical to the finite-difference form of the two-dimensional differential equation governing the velocity potential, and the resistance network can then be used to estimate the interference upwash at the wing for different types of tunnel boundary. Rushton in fact considered the interference experienced by small lifting wings in rectangular tunnels having both roof and floer slotted or all four walls slotted. With such an analogue it is possible to compare results obtained under the linear homogeneous boundary condition, discussed earlier, and for walls with ideal slots and slats; this comparison is discussed below in Section 6.5.3. The analogue method can also be used to check the significance of assumptions necessary to secure an analytical solution to a particular problem, and to give, if required, improved data for the interference corrections. Examples of this are contained in Sections 6.5.3 and 6.5.4.

## 6.3 TWO-DIMENSIONAL BLOCKAGE IN TUNNELS WITH LONGITUDINAL SLGTS

Blockage interference arises from both the model and its wake, the two components being called solid and wake blockage respectively. It manifests itself as an increment  $\Delta U_{\rm B} = \epsilon_{\rm B} U$  in stream velocity at the model position. The total blockage factor  $\epsilon_{\rm B}$  is conveniently expressed as the sum of the solid-blockage factor  $\epsilon_{\rm S}$  and the wake-blockage factor  $\epsilon_{\rm W}$ , which are derived independently at zero model lift. Associated with both the solid- and wake-blockage increments are longitudinal velocity gradients, which impose a corresponding drag force on the model.

The enalysis for slotted tunnels is rather less extensive in scope than for the closed and open tunnels, although most of the important simple cases have now been discussed. The present section is concerned with two-dimensional models spanning a slotted tunnel of height h; the tunnel side-walls are assumed to act merely as reflection planes contributing nothing to the flow about the model.

#### 6.3.1 Solid Blockage

If the model is small and thin, it may be represented by a two-dimensional doublet whose velocity potential in a free stream of subsonic Mach number M is

$$\Phi_{g} = \frac{AU}{2\pi\beta} \left[ \frac{x}{x^2 + \beta^2 z^2} \right] .$$
 (6.9)

where  $\Lambda$  is the cross-sectional area of the model, and  $\beta = (1 - \frac{1}{2}^2)^{\frac{1}{2}}$ . The origin of the co-ordinate system is at the centroid of the model. In the tunnel the perturbation potential is given by

$$\Phi = \Phi_n + \Phi_i , \qquad (6.10)$$

 $\Phi_i$  being the potential of the wall constraint. The latter must satisfy the general linearized equation of flow

$$\beta^{2} \frac{\partial^{2} \Phi_{i}}{\partial x^{2}} + \frac{\partial^{2} \Phi_{i}}{\partial z^{2}} = 0 . \qquad (6.1^{1})$$

The potential  $\Phi$  must also satisfy the boundary condition (6.7) at the slotted walls, so that

$$\left(\frac{\partial \Phi_{i}}{\partial x} \pm K \frac{\partial^{2} \Phi_{i}}{\partial x \partial z} \pm \frac{1}{p} \frac{\partial \bar{\Phi}_{i}}{\partial z}\right)_{z=\pm\frac{1}{2}b} = -\left(\frac{\partial \Phi_{g}}{\partial x} \pm K \frac{\partial^{2} \Phi_{g}}{\partial x \partial z} \pm \frac{1}{p} \frac{\partial \Phi_{g}}{\partial z}\right)_{z=\pm\frac{1}{2}b}$$
(6.12)

From Equations (6.9), (6.11) and (6.12), the interference potential may be found. Different techniques have been employed in the solution for  $\hat{\Psi}_i$ . Reference 6.1, for example, uses Fourier transforms

BERE SALES

$$G(g, z) = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \Phi(x, z) e^{i\xi z} dx$$

$$\Phi(x, z) = \int_{-\infty}^{+\infty} \ell(g, z) e^{i\xi z} dz$$
(6.13)



for both  $\Phi_{\rm p}$  and  $\Phi_{\rm i}$ . Once the solution for  $\Phi_{\rm i}$  is known, the solid-blockeys factor can be obtained from the expression

$$\epsilon_{s} = \frac{\Delta U_{s}}{U} = \frac{1}{U} \left( \frac{\partial \Phi_{i}}{\partial x} \right)_{x=z=0} , \qquad (6.14)$$

conditions at the origin being taken as the mean of the distributed effect along the model. In the general  $case^{6.1}$ 

$$\epsilon_{\rm s} = -\frac{5}{\pi^2} \left( \frac{\pi A}{6\beta^3 h^2} \right) \int_{0}^{\infty} \left\{ \frac{\left[ 1 - F^2 q^2 - (\beta/P)^2 \right] + \left[ (1 - Fq)^2 + (\beta/P)^2 \right] e^{-2q}}{(\cosh q + Fq \sinh q)^2 + (\beta/P)^2 \sinh^2 q} \right\} q \, dq \,, \qquad (6.15)$$

where  $q = \frac{1}{2}\beta_{sh}^{2}$ . The effect of slot geometry is represented by the parameter P (= 2%/h), and the influence of viscosity on the slot flow by the ratio  $\beta/P$ . Thus, if P is assumed to be independent of stream Mach number (as certain tests suggest), the influence of the viscous slot flow will vary with M; moreover, as the stream Mach number approaches unity,  $\beta/P$  tends to zero, and the ideal slotted tunnel condition is approached.

Following Reference 6.17, it is convenient to use a parameter  $(1 \div F)^{-\frac{1}{2}}$  to illustrate the variation of the solid-blockage effect with slot geometry. This parameter has a value of zero for the closed tunnel and unity for the open tunnel. It seems reasonable, too, to consider the ratio of  $\epsilon_s$  in the slotted tunnel to its value  $(\epsilon_s)_c$  in a similar tunnel with closed walls. The latter quantity is that in brackets outside the integral in Equation (6.15). Thus

$$\Omega_{\rm s} = \epsilon_{\rm s} / (\epsilon_{\rm s})_{\rm c} = \left(\frac{6\beta^3 b^2}{\pi A}\right) \epsilon_{\rm s}$$
(6.16)

is considered as a function of  $(1 + F)^{-\frac{1}{2}}$ . This type of presentation may be of value in extending the present results for simple model shapes to more complex representations, by assuming that for a given tunnel the value of  $\Omega_s$  is unchanged if the more sophisticated methods discussed in Chapter V are used for determining  $(\epsilon_s)_c$ .

The variation of  $\Omega_s$  with  $(1 \div F)^{-\frac{1}{2}}$  for a two-dimensional zerofoil is shown in Figure 6.2 for two values of  $\beta/P$ . One, taken from Reference 6.17, corresponds to the ideal slotted tunnel (F -  $\infty$ ) and indicates an algost linear variation of  $\Omega_s$ from the fully closed tunnel to open-jet tunnel. The other, broken, curve has been calculated for M = 0.8, P = 1.0; the latter value, though arbitrarily chosen, may well be typical of many tunnels.

Zero solid blockage will correspond to  $\,\Omega_{\rm S}^{}=0$  , and this is achieved in an ideal slotted tunnel for the condition

 $(1 + F)^{-\frac{1}{2}} = 0.67,$  (6.17) P =  $F_0 = 1.18$ .

whence

For the broken curve,  $P_0 = 1.0_5$ . The variation of  $P_c$  with  $\beta/P$  to give zero blockage has been estimated by Baldzin et al.<sup>5.1</sup> and is reproduced as Figure 6.3.

The definition of F contained in Figure 6.1 indicates that for each value of the ratio of tunnel height to slot spacing (h/d) there is a critical value of the openarea ratio for zero solid blockage, as illustrated in the following table, where the ideal slotted tunnel is compared with one having  $\beta/P = 1$ .

| <u>h</u><br>đ                                      | 1    | 2   | 4                         | 6                                      | 8                                      | 12                                      |
|--|------|-----|---------------------------|--|--|---|
| $100 \frac{a}{d}$ for $\frac{\beta}{P} = 0$        | 10.0 | 1.6 | 0.4<br>× 10 <sup>-1</sup> | 0.9<br>× 10 <sup>-3</sup>              | 0.2<br>× 10 <sup>-‡</sup>              | 0.1<br>× 10 <sup>-7</sup>               |
| $100 \frac{a}{d}  \text{for}  \frac{\beta}{p} = 1$ | 20.5 | 6.4 | 0.64                      | 0.6 <sub>5</sub><br>× 10 <sup>-1</sup> | 0.6 <sub>5</sub><br>× 10 <sup>-2</sup> | 0. ೮ <sub>5</sub><br>≮ 10 <sup>−4</sup> |

Percentage Open-Area Natio for Zero Solid Blockage

These values are remarkably small for the slot configurations usually employed, and it seems doubtful whether the theoretical approach can be considered entirely valid for the two-dimensional case. On this evidence, slot configurations usual in two-dimensional tunnels would be often too open, and the solid-blockage factor would be negative. Experimentally, Pearcey et al. (Ref. 6.24; 1959) found zero-blockage conditions for a particular slot configuration h/d = 6.37, a/d = 1.65, for which P = 0.37. As Figure 6.3 shows, this would correspond to a value  $\beta/P = 1.20$ , though no attempt was made to measure this quantity directly.

The alternative boundary condition suggested by Woods in Section 7.31 of Reference 5.19 is claimed to be more valid for h/d < 2, but for greater values of this ratio there is little difference between his results and those obtained by other authors.

Maeder (Ref. 6.25; 1953) discusses the case in which the model is too large to be represented sufficiently accurately by a doublet and where use of a Rankine oval (discrete source and sink) is preferable. His theoretical curves for the ideal slotted tunnel are reproduced in Figure 6.4 for different values of the ratio  $1/\beta h$ , where lis the source-sink separation distance. When the aerofoil is thin, this ratio is approximately  $c/\beta h$ , where c is the model chord; a more accurate relationship between l and c will be found in Reference 6.26. Figure 6.4 shows a discrepancy between Waeder's curve for a very small model and that given by Davis and Where<sup>6.17</sup>; this is due to an approximation contained in the analysis of Reference 6.25t. The discrepancy does not affect the estimated value of  $F_0$  required to give zero solid blockage (Fig. 6.5) and, since most wind tunnels operate in the region between this condition and that for an open-jet, the difference between the two methods of analysis may usually be unimportant.

• For example, if h is 10 units and the slot spacing d is 1 unit, then the calculated slot width for zero interference is  $6 \times 10^{-9}$  units for the ideal slotted tunnel and about  $7 \times 10^{-6}$  units for  $\beta/P = 1.0$ .

<sup>†</sup> There is a similar discrepancy for a small three-dimensional model in a tunnel with solid sidewalls and slotted roof and floor (Section 6.4.2); this is discussed by Acum (Ref. 6.34; 1961).

levere and a real state that have an
The problem of an aerofoil whose chord may no longer be regarded as small compared with the tunnel height is also discussed by Tirumalesa, using the rhecelectric analogy<sup>6.55</sup>. Nis results in Figure 3 of Reference 6.55 are in broad agreement with the theoretical calculations of Maeder and confirm that, as the model size increases relative to the tunnel,  $F_0$  also increases (Fig.6.5). For a fixed slot spacing, this . corresponds to a reduction in open area ratio.

### 6.3.2 Longitudinal Pressure Gradient Due to Solid Blockage

In the preceding section, the velocity increment at the model position due to the solid blockage has been discussed. This increment, however, varies both along and across the working sections. The latter variation is small in the region of the model and may usually be neglected. The longitudinal velocity gradient has been detected experimentally by Wright (Ref. 6.28; 1959) and is of importance for two reasons. Firstly, if the stream velocity is measured at some position on the tunnel wall upstream of the model, this position may be itself subject to a small interference effect<sup>\*</sup>. Secondly, consideration must be given to the velocity gradient, and hence the corresponding pressure gradient, along the model to see whether this influences the measured drag.

In an ideal slotted tunnel with the linearized, homogeneous boundary condition, the solid-blockage velocity increment varies symmetrically on either side of the position of a small model, so that the longitudinal pressure gradient there is zero (Fig. 6.6). Hence there is no horizontal buoyancy force, and no correction to be applied to the measured drag. On the other hand,  $\text{Wood}^{5,57}$  shows that for large interference velocities, where the linear boundary condition may no longer be valid, the interference velocity may be antisymmetric on either side of the model, giving rise to a buoyancy force.

The interference sympetry is also lost if viscous effects appear at the slots and the linearized boundary condition (6.7) with finite P is used. The longitudinal velocity gradient for a two-dimensional aerofoil may be derived from the general solution for  $\hat{\Phi}_i$  in Equation (19) of Reference 6.1. At the model position

$$\frac{\partial \epsilon_s}{\partial x} = \frac{24}{\pi^2 \beta h} (\epsilon_s)_c \frac{\beta}{p} \int_0^\infty I_A \, dq \, . \qquad (6.18)$$

stere

$$I_{A} = \frac{q^{2}}{(\cosh q + Fq \sinh q)^{2} + (\beta/P)^{2} \sinh^{2}q}$$

and  $(\epsilon_{\rm g})_{\rm C} = \pi A/(6\beta^3 h^2)$  is the value of  $\epsilon_{\rm g}$  for a closed tunnel. The pressure gradient,  $\partial p/\partial x$ , is given by  $-\rho U^2(\partial \epsilon_{\rm g}/\partial x)$ , and the magnitude of this gradient for a particular ratio,  $\beta/P = 0.6$ , is illustrated in Pigure 6.7.

• In many tunnels, the stream speed is determined from the pressure in the pleaus chamber surrounding the working section. The influence of the model on the pleaus-chamber pressure is discussed briefly in Reference 6.49.

The pressure gradient associated with the solid blockage imposes an unwanted drag force on the model, which is given by minus the product of the pressure gradient and the equivalent area of aerofoil cross-section  $(A_{\rm f})$ . Now  $A_{\rm e}$  may be shown to be equivalent to  $6\beta^3 h^2(\epsilon_{\rm s})_c/\pi$  (see Reference 6.43, for example), and it follows that the required correction to the measured drag coefficient for the wall-induced velocity gradient is

$$(\Delta C_{\rm p})_{\rm sg} = \frac{\partial p}{\partial x} \frac{A_{\rm e}}{\frac{1}{2} \rho U^2 c}$$

$$= -\frac{288}{\pi^3} \frac{\beta^2 h}{c} (\epsilon_{\rm s})_c^2 \frac{\beta}{p} \int_0^\infty I_{\rm A} \, dq , \qquad (6.19)$$

where  $I_A$  is the integrand in Equation (6.18).

### 6.3.3 Wake Blockage

As discussed in Section 5.4.4, the presence of the wake downstream of the model gives rise to two constraint effects when the tunnel walls are solid. One is an effective velocity increment at the model position: the other is the horizontal buoyancy associated with the longitudinal velocity gradient at the model position. In the openjet tunnel, the first effect (but not the second) is usually taken to be zero, because the boundaries of the jet can be displaced away from the tunnel axis in order to compensate for the low-velocity air in the model wake. There is then no need to increase the stream velocity in the flow outside the wake in order to maintain the same mass flow upstream and downstream of the model. Similar considerations would apply to the slotted tunnel.

Formally, the model wake is replaced by a source. The subsequent analysis has been made for an ideal slotted tunnel by Maeder<sup>6.13,6.25</sup> and for a tunnel with a porogity parameter P by Wright<sup>6.28</sup>, using the techniques of Reference 6.1.

When the model is a two-dimensional wereful, the effective interference velocity\*,  $\epsilon_{\rm w}$ U, at the model is found to be zero, provided that the slot flow corresponds to that in an ideal slotted tunnel. The velocity gradient at the model position for this case is

 $\frac{\partial \epsilon_{\rm g}}{\partial x} = -\frac{C_{\rm B}c}{\pi\beta^2 h^2} \int_0^\infty \frac{(1-Fq)e^{-q}q \, dq}{\cosh q + Fq \sinh q} \,. \tag{6.20}$ 

The velocity gradient associated with a single source has the same magnitude as the velocity due to a doublet of equal strength. Provided that the boundary conditions are independent of x, as in Equation (6.7) when  $P \rightarrow \infty$ , then the wake-blockage gradient can be identified with the solid-blockage factor  $\epsilon_{\rm g}$ , apart from a numerical factor. Thus for the condition  $\beta/P = 0$ , it follows from Equations (6.15), (6.16) and (6.20) that

• There may be an induced velocity at some reference point upstream of the model and in consequence an equal and opposite effective interference velocity at the model.

361

ないではないまです。

. . . .

$$\frac{\partial \epsilon_{w}}{\partial x} = \frac{\pi}{12} \Omega_{s} \frac{C_{b}c}{\beta^{3}h^{2}} , \qquad (6.21)$$

which becomes zero when  $\Omega_g$  vanishes. Hence, in an ideal slotted tunnel, the wakeblockage velocity increment and horizontal bucyancy are eliminated if zero solidblockage conditions are present. If this is not the case, the correction to the measured drag is

$$(\Delta C_{\rm D})_{\rm wg} = -\Omega_{\rm s}(\epsilon_{\rm s})_{\rm c}C_{\rm D} , \qquad (6.22)$$

the derivation being similar to that described in Section 5.8.2.

If the tunnel does not conform to ideal slotted conditions, the velocity increment far upstream is still zero, but at the model position by Equation (23) of Reference 6.28 there is a wake-blockage factor

$$\mathbf{f}_{\mathbf{y}} = -\frac{2}{\pi} \left[ \frac{\mathbf{C}_{\mathbf{p}} \mathbf{c}}{4\beta^2 \mathbf{h}} \right] \frac{\beta}{\mathbf{p}} \int_{0}^{\infty} \mathbf{I}_{\mathbf{g}} \, \mathrm{d}\mathbf{q} \, . \tag{6.23}$$

where

$$I_3 = \frac{l(1 + Fq) \sinh q + \cosh q le^{-q}}{(\cosh q + Fq \sinh q)^2 + (\beta/P)^2 \sinh^2 q}$$

The quantity in square brackets outside the integral in Equation (6.23) is  $(\epsilon_{y})_{c}$ , the value of  $\epsilon_{y}$  for a closed tunnel, so that the ratio of the wake-blockage factors may be written as

$$\Omega_{\pi} = \frac{\epsilon_{\pi}}{(\epsilon_{\pi})_{c}} = -\frac{2}{\pi} \frac{\beta}{p} \int_{0}^{\infty} \mathbf{I}_{\mathbf{B}} \, \mathrm{d}\mathbf{q} \, . \tag{6.24}$$

The velocity gradient at the model and the associated drag correction are still given by Equations (6.21) and (6.22), and it follows that these vanish if the tunnel is operating under zero solid-blockage conditions.

# 6.4 THREE-DIMENSIONAL BLOCKAGE IN TUNNELS WITH LONGITUDINAL SLOTS

There are three main types of slotted working-section used for testing threedimensional models: (i) circular; (ii) rectangular with slots on the top and bottom walls; and (iii) rectangular with all walls slotted. These will be considered in turn.

### 6.4.1 Circular Tannel

A small three-dimensional model may be replaced by its equivalent doublet, whose potential in free air is given approximately by

$$\Phi_{\rm m} = \frac{UV}{4\pi} \left[ \frac{x}{(x^2 + \beta^2 r^2)^{3/2}} \right], \qquad (6.25)$$

and the state of the

And the bolie terms and the second second

and thereases a

where V is the volume of the model, and  $r = (y^2 + z^2)^{\frac{1}{2}}$  is the radial distance perpendicular to the tunnel axis. In cylindrical co-ordinates the velocity potential satisfies

$$\beta^2 \frac{\partial^2 \Phi}{\partial \mathbf{x}^2} + \frac{1}{r} \frac{\partial}{\partial r} \left( r \frac{\partial \Phi}{\partial r} \right) = \mathbf{0} . \qquad (6.26)$$

As in Section 6.3.1, we write  $\Phi = \Phi_{\mathbf{R}} + \Phi_{\mathbf{i}}$ , which satisfies Equation (6.26) and the boundary condition at the wall (r = R),

$$\left(\frac{\partial \Phi_{i}}{\partial x} + K \frac{\partial^{2} \Phi_{i}}{\partial x \partial r} + \frac{1}{P} \frac{\partial \Phi_{i}}{\partial r}\right)_{r=R} = - \left(\frac{\partial \Phi_{R}}{\partial x} + K \frac{\partial^{2} \Phi_{R}}{\partial x \partial r} + \frac{1}{P} \frac{\partial \Phi_{R}}{\partial r}\right)_{r=R}.$$
 (6.27)

This corresponds to Equation (6.12). Fourier integral expressions for  $\Phi_{\underline{s}}$  and  $\Phi_{\underline{i}}$  may again be used to deduce the perturbation velocity at the model position and hence the blockage factor. Baldwin et al.<sup>6.1</sup> give this in a form which may be rewritten as

$$\epsilon_{\rm g} = -0.40 \left[ \frac{0.797V}{2\pi\beta^3 R^3} \right] \int_0^\infty I_{\rm c} \, dq \,, \qquad (6.28)$$

with

$$I_{c} = \frac{\left\{ K_{0}(q)I_{0}(q) - qF[K_{1}(q)I_{0}(q) - K_{0}(q)I_{1}(q)] - \left(q^{2}F^{2} + \frac{\beta^{2}}{P^{2}}\right)K_{1}(q)I_{1}(q)\right\}q^{2}}{\left[I_{0}(q) + qFI_{1}(q)\right]^{2} + \left[\frac{\beta}{P}I_{1}(q)\right]^{2}} \right]^{2}$$

where for the circular tunnel F = K/R and the modified Bessel functions  $I_0$ .  $I_1$ .  $K_0$  and  $K_1$  are tabulated in Reference 6.27. For the ideal slotted tunnel  $P \rightarrow \infty$ , and  $I_C$  becomes

$$I_{c} = \frac{[K_{0}(q) - qFK_{1}(q)]q^{2}}{[I_{0}(q) + qFI_{1}(q)]} . \qquad (6.29)$$

The expression in brackets in Equation (6.28) is the solid-blockage factor for a closed circular tunnel, so that we may write

$$\widehat{u}_{\mathbf{s}} = \frac{\epsilon_{\mathbf{s}}}{(\epsilon_{\mathbf{s}})_{\mathbf{c}}} = -0.40 \int_{0}^{\infty} \mathbf{I}_{\mathbf{c}} \, \mathrm{dq} \quad . \tag{6.30}$$

The variation of  $\Omega_{\rm g}$  with slot geometry is shown in Figure 6.8 for the ideal slotted tunnel only. Davis and Moore<sup>5.17</sup> used these results to demonstrate the validity of the approximate homogeneous boundary condition (6.27) by showing that they are in good agreement with the exact calculations made by Wright and Ward<sup>6.16</sup>.

Zero blockage in the ideal slotted tunnel occurs when the abscissa  $(1+P)^{-\frac{1}{2}} = 0.798$ . i.e. when  $P = P_0 = 0.57$ . Now for a circular tunnel with N equally-spaced slots

.

$$\frac{\mathrm{d}}{\mathrm{R}} = \frac{2\pi}{\mathrm{N}} , \qquad (6.31)$$

so that the non-dimensional slot parameter F becomes (Fig. 6.1)

$$\mathbf{F} = \frac{\mathbf{K}}{\mathbf{R}} = \frac{2}{N} \log_e \operatorname{cosec} \frac{\pi \mathbf{a}}{2\mathbf{d}} \,. \tag{6.32}$$

The variation of the open-area ratio with the number of slots to give zero blockege in an ideal tunnel is illustrated in the following table corresponding to  $F = F_c = 0.57$ .

| N                 | 2    | 4    | 6    | 8   | 12  | 16  | 24   |
|-------------------|------|------|------|-----|-----|-----|------|
| $100 \frac{a}{d}$ | 38.3 | 20.7 | 11.6 | 6.5 | 2.1 | 0.7 | 0.67 |

These values are far larger than those for a two-dimensional model and give reasonable slot widths even when the number of slots is large.

The effect of viscosity on  $F_0$  is discussed in Reference 6.1. from which the present Figure 6.9 is reproduced<sup>\*</sup>; certain values have been transferred to Figure 6.8 to indicate the possible trend of the curves for  $\Omega_g$ . There seems to be little direct evidence as to whether the assumption of zero viscosity at the slots is reasonable or not in typical slotted circular tunnels. Nevertheless the tests of Reference 6.16, made in a tunnel with 10 slots and open-area ratio 12.5%  $[(1 + P)^{-\frac{1}{2}} = 0.86]$  exhibit an interference of about the magnitude expected from the curve of Figure 6.8. It seems therefore that, if there is no information about the appropriate value of  $\beta/P$  for a given tunnel and the wall geometry is not far removed from that required to give zero blockage for an ideal tunnel, then it may be assumed that  $\beta/P = 0$ . The longitudinal pressure gradient of the model due to the solid blockage is absent for this condition only; when  $\beta/P$  has a finite value, the resulting velocity gradient, and hence the pressure gradient, may be calculated from Equation (34) of Reference 6.1.

The wake-blockage effects of a small three-dimensional model in a circular slotted tunnel are discussed in Reference 6.28, where the wake is replaced by a three-dimensional point source whose strength is proportional to the model drag. The expressions for the axial interference velocity ratio and the velocity gradient along the tunnel, at the model position, are

$$\Omega_{w} = \frac{\epsilon_{w}}{(\epsilon_{w})_{c}} = -\frac{1}{\pi} \frac{\beta}{p} \int_{0}^{\infty} \frac{dq}{[\bar{i}_{0}(q) + qFI_{1}(q)]^{2} + [\frac{\beta}{p}I_{1}(q)]^{2}}$$
(6.33)

• Attention is drawn to the Errata to Reference 6.1, which corrects the original Figure 4.

and

$$\frac{\partial \epsilon_{\mathbf{w}}}{\partial \mathbf{x}} = -\frac{1}{\pi\beta_{\mathbf{R}}} \left(\epsilon_{\mathbf{w}}\right)_{\mathbf{c}} \int_{0}^{\infty} \mathbf{I}_{\mathbf{c}} \, \mathrm{d}\mathbf{q} = \frac{\left(\epsilon_{\mathbf{w}}\right)_{\mathbf{c}} \Omega_{\mathbf{s}}}{\partial 40\pi\beta_{\mathbf{R}}} \,, \qquad (6.34)$$

1.8. 200

where I<sub>c</sub> is the integrand of Equation (6.28) for the solid-blockage factor and

$$(\epsilon_{\mathbf{w}})_{\mathbf{c}} = \frac{C_{\mathbf{D}}}{4\beta^2} \left( \frac{\mathbf{S}}{\pi \mathbf{R}^2} \right)$$

S being the reference area on which the model drag coefficient is based. It follows from Equations (6.30) and (6.34) that, if the solid-blockage interference is zero, then the velocity gradient at the model due to the wake will be eliminated. Similarly it may be shown that for this flow condition  $\epsilon_w$  is zero too.

Then zero solid blockage is not obtained, the wake blockage may be found from Equation (6.33); the buoyancy correction to the measured drag is precisely that contained in Equation (6.22).

### 6.4.2 Rectangular Tunnel with Slotted Roof and Floor

In this case the model may once again be represented by the doublet whose potential in free air  $(\Phi_{g2})$  is given in Equation (6.25). The solid side-walls require a condition of zero flow velocity normal to them and can therefore be represented by a horizontal row of doublet images spaced at distances equal to the tunnel breadth (b). The potential  $\Phi_r$  of such an image-row can be derived simply from  $\Phi_{g2}$  (Ref. 6.17). Another potential  $\Phi_s$  may be added to  $\Phi_r$  to satisfy the conditions at the slotted walls. Thus in an ideal slotted tunnel  $\Phi = \Phi_r + \Phi_s$  will satisfy

$$\frac{\partial \Phi}{\partial y} = 0 \qquad (y = \pm \frac{1}{2}b)$$

$$\frac{\partial \Phi}{\partial x} \pm \kappa \frac{\partial^2 \Phi}{\partial x \partial z} = 0 \qquad (z = \pm \frac{1}{2}b)$$
(6.35)

The formal solution for  $\Phi_s$  is given in Reference 6.17. The solid-blockage incremental velocity at the model position has two components, one associated with the slotted, and the other with the solid walls; thus

-----

$$\Delta U_{g} = \frac{\partial \tilde{c}_{g}}{\partial x} + \frac{\partial}{\partial x} \left( \tilde{\Phi}_{r} - \tilde{\Phi}_{r} \right) . \qquad (6.36)$$

After substitution and some manipulation<sup>6.34</sup> it follows that

$$\epsilon_{s} = \frac{\Delta v_{s}}{v} = \frac{V}{\beta^{3}b^{3}} \left\{ \frac{1}{2\pi} \sum_{u=1}^{\infty} \frac{1}{n^{3}} + \frac{2b^{2}}{\pi b^{2}} \int_{0}^{\infty} \frac{qe^{-q}(Fq-1)}{\cosh q + Fq \sinh q} dq + \right. \\ \left. + \sum_{n=1}^{\infty} \frac{4b^{2}}{\pi b^{2}} \int_{q_{0}}^{\infty} \frac{(q^{2} - q_{0}^{2})^{\frac{1}{2}}e^{-q}(Fq-1)}{\cosh q + Fq \sinh q} dq \right\} .$$
(6.37)

where  $q_0 = mh/b$  and, from Figure 6.1,

$$\mathbf{F} = \frac{2b}{\pi Nh} \log_e \operatorname{cosec} \frac{\pi Na}{2b} . \tag{6.38}$$

The first term in the bracket expresses the influence of the solid side-walls, and the remainder the influence of the slotted roof and floor. In the special case F = 0, corresponding to a tunnel with open roof and floor, Equation (6.37) becomes<sup>6.34</sup>

$$(\epsilon_{g})_{p=0} = \frac{V}{\beta^{3}b^{3}} \left\{ \frac{1}{2\pi} \sum_{n=1}^{\infty} \frac{1}{n^{3}} - \frac{\pi b^{2}}{12h^{2}} + \frac{4b}{h} \sum_{t=1}^{\infty} t \left[ \sum_{n=1}^{\infty} \frac{(-1)^{n}}{n} K_{1} \left( \frac{2\pi nth}{b} \right) \right] \right\} , \quad (6.39)$$

and for  $P \rightarrow \infty$ , corresponding to a closed rectangular tunnel,

$$(\epsilon_{\rm s})_{\rm c} = \frac{V}{5^3} \left\{ \frac{1}{2\pi} \sum_{\rm n=1}^{\infty} \frac{1}{{\rm n}^3} + \frac{\pi {\rm b}^2}{6{\rm h}^2} \div \frac{4{\rm b}}{{\rm h}} \sum_{\rm t=1}^{\infty} {\rm t} \left[ \sum_{\rm n=1}^{\infty} \frac{1}{{\rm n}} \kappa_1 \left( \frac{2\pi {\rm nth}}{{\rm b}} \right) \right] \right\} \quad . \tag{6.40}$$

T roomential behaviour of the Bessel function  $K_1(2\pi nth/b)$  ensures rapid convergence of the series (6.39) and (6.40) which are most convenient for computation. In Figure 6.10 values of  $\Omega_g = \epsilon_g/(\epsilon_g)_c$  from Equations (6.37) and (6.40) have been plotted as functions of  $(1+F)^{-\frac{1}{2}}$  for representative values of h/b. As might be expected, the effect of varying F is greatest for small values of h/b; for h > 1.17b zero blockage cannot be achieved. This confirms an earlier prediction of Wieselsberger (Ref. 6.7; 1942). Numerical values of  $\beta^{3}b^{3}\epsilon_{g}/(8V)$  are given in Table 1 of Reference 6.34 for the complete range of  $(1+F)^{-\frac{1}{2}}$  and h/b; alternatively, the solid-blockage factor for small wings may be obtained from Figure 6.10 with the aid of the tunnel-shape parameter T from Figure 5.3 or 5.4 of Chapter V.

Macder<sup>6.25</sup> has used a somewhat different technique to evaluate solid-blockage corrections and has extended his method to include the effect of model span. His result for a small model is obtained by taking only the first two terms inside the bracket in Equation (6.37). It can be shown that for large  $q_0$ 

$$\frac{4b^2}{mh^2} \int_{q_0}^{\infty} \frac{(q^2 - q_0^2)^{\frac{1}{2}} e^{-q} (Fq - 1)}{\cosh q + Fq \sinh q} \, dq = O(q_0^{1/2} e^{-2q_0})$$
(6.41)

STOR SHOW FOR THE

and will be small. If h = 1.25b, the difference between Maeder's values of  $\epsilon_g$  and those from Equation (6.37) is less than  $0.0006V/(\beta b)^3$ , but for smaller values of h/b it becomes appreciable. In effect, Maeder assumes the influence of the slotted roof and floar to be independent of model span. Then, for a model of span 2s, the first term of the bracket in Equation (6.37) is replaced by

$$\frac{1}{2\pi}\sum_{n=1}^{\infty}\frac{n}{(n^2-\frac{1}{4}\sigma^2)^2} = 0.1913 + \frac{\sigma^2(8-\sigma^2)}{2\pi(4-\sigma^2)^2} + 0.0029\sigma^2 + 0.0002\sigma^2 + \dots , \quad (6.42)$$

where  $\sigma = 2s/b$ . The effect of model span, shown in Figure 5.11, is obtained by adding the terms involving  $\sigma$  on the right-hand side of Equation (6.42) to the bracketed

expression in Equation (6.37). The curves of  $\epsilon_{\rm g}/(\epsilon_{\rm S})_{\rm C,\,\sigma=0}$  against  $(1 + F)^{-\frac{1}{2}}$  are thus displaced upwards by an amount independent of F, but depending on  $\sigma$  and h/b. For a square tunnel zero blockage can only just be achieved if  $\sigma$  is as large as 0.8.

The wake-blockage effects associated with a small model in an ideal slotted tunnel with ventilated roof and floor are discussed in Reference 6.28. It is there shown that  $\epsilon_{\rm w}$  may be taken as zero, whatever the slot geometry, and that the velocity gradient at the model vanishes when the wall is designed to eliminate solid blockage. The influence of viscous flow within the slots on the foregoing analysis, cf. Equation (6.18), does not appear to have been considered for this type of tunnel.

Maeder's treatment does not hold when h/b is small and should be used with caution

### 6.4.3 Rectangular Tunnel with Slots on Each Wall

when h/b < 1.

This particular case does not seem to have been treated directly, though several authors have suggested that the solid-blockage effects for a square tunnel should be very similar to those obtained in a circular tunnel of the same cross-sectional area. If the solid-blockage factor for a small model takes the form

$$\epsilon_{\rm s} = T\left(\frac{\rm v}{\beta^3 c^{3/2}}\right) \tag{6.43}$$

where C is the tunnel cross-sectional area, then T will depend on the tunnel shape and slot parameter F. For example, T has values of 0.706 and 0.717 for the circular and square closed tunnels respectively, and for the corresponding open tunnels, these values are -0.182 and -0.211. Thus, when the tunnel walls are close to the open-jet condition, the error in using the circular-tunnel corrections may be of consequence.

In the absence of direct information, it seems more realistic to use the existing knowledge of the blockage corrections for the open and closed square tunnels, which correspond to values of  $\Omega_s$  of 1.0 and -0.294. These values may be joined by a straight line<sup>6.33</sup> or, better still, by a scaled version of the curve for a circular tunnel. These estimated curves for the ideal slotted square tunnel are shown in Figure 6.12 and are almost coincident in the important region near zero blockage at approximately  $F = F_0 = 0.67_5$ . If F is assumed to be equal to K/R', where R' =  $\sqrt{(C/\pi)}$  is the radius of a circular tunnel having the same cross-sectional area, then for a square tunnel

$$F = \frac{\sqrt{\pi K}}{h} = \frac{d}{\sqrt{\pi h}} \log_e \csc \frac{\pi a}{2d}$$
$$= \frac{1}{\sqrt{\pi N}} \log_e \csc \frac{\pi a}{2d} . \qquad (6.44)$$

mestioner.

where N is the number of slots on each wall. For zero solid blockage, the open-area ratio a/d is given in the following table.

| N                 | 1             | 2   | 4    | 6    | 8                   | 10     |
|-------------------|---------------|-----|------|------|---------------------|--------|
| $100 \frac{a}{d}$ | 19 <u>,</u> 5 | 5.8 | 0.53 | 0.05 | 0. 094 <sub>5</sub> | 0.0004 |

Around the tunnel periphery there will be 4N slots, and the open-area ratio corresponding to this number of slots in the table for the circular tunnel in Section 6.4.1 is about the same, suggesting once more the correspondence between the two types of tunnel. It seems reasonable therefore to apply to the square slotted tunnel similar influences of slot porosity to those aiready discussed for the circular tunnel.

The foregoing procedure may of course be applied to a tunnel of general rectangular cross-section; the appropriate values of  $\epsilon_s$  for fully open and closed boundaries (Chapter V) would replace the end values of Figure 6.12. If the correspondence with the equivalent circular tunnel is maintained, then in place of Equation (6.44)

$$F = \frac{\sqrt{\pi K}}{\sqrt{bh}} = \frac{d}{\sqrt{\pi bh}} \log_e \operatorname{cosec} \frac{\pi a}{2d} . \qquad (6.45)$$

One interesting point, which perhaps strengthens the use of an approximate method similar to that at present being discussed, is revealed by replotting the known results for slotted tunnels in terms of a new parameter

$$\Psi = \frac{\Omega_{\rm s} - \Omega_{\rm s}'}{1 - \Omega_{\rm s}'} , \qquad (6.46)$$

where  $\Omega'_s$  is the value of  $\Omega_s$  for the condition F = 0. Thus  $\Psi$  varies between unity and zero as the slotted walls change from fully closed to fully open. As Figure 6.13 shows, most of the known results lie fairly close together. Except perhaps for tunnels having a large ratio of breadth to height, some mean curve could be used to estimate the solid-blockage factor for other configurations including, for example, elliptical and octagonal sections. In particular, for rectangular tunnels with slots on each wall the blockage factor for the open-jet case must be determined first; typical values of  $\Omega'_s$  for a small centrally-situated model are listed below.

| $\frac{h}{b} \text{ or } \frac{b}{h}$ | 2.00  | 1.80  | 1.60  | 1.40  | 1.25  | 1.10  | 1.00  |
|---------------------------------------|-------|-------|-------|-------|-------|-------|-------|
|                                       | 0.449 | 0.413 | 0.376 | 0.337 | 0.313 | 0.296 | 0.293 |

The wake-blockage effects in tunnels with all four walls slotted do not seem to have been discussed. For tunnels of near-square cross-section, it would seem reasonable to calculate the appropriate velocity increment and drag correction from Equations (6.33) and (6.22) for a circular tunnel of the same cross-sectional area. For an ideal slotted tunnel, it may be assumed, by analogy with the circular tunnel and that having a slotted roof and floor, that the wake-blockage factor is zero and that the gradient at the model vanishes for the wall geometry giving zero solid blockage.

# 6.5 LIFT INTERFERENCE IN TUNNELS WITH LONGITUDINAL SLOTS

The induced upwash associated with the walls of a circular, slotted tunnel containing a small lifting wing was one of the first problems of this type to be considered, and in the early work of Pistolesi<sup>6,14</sup> the exact boundary conditions were employed. Much simplification results from the use of the equivalent homogeneous boundary conditions discussed in Section 6.2, which are valid provided that there is a reasonable number of slots; by this means the lift interference has now been evaluated for two- and three-dimensional models in the types of tunnel most frequently employed.

The lifting model is usually replaced by a simple vortex system. for example, a single bound vortex if the model is a two-dimensional aerofoll, or a trailing vortex pair in the case of a small wing. This type of representation will be sufficient for most purposes. The actual interference experienced by the model may in general be regarded as composed of two parts; an upwash (or incidence) correction at the model position, and a "streamline curvature" effect associated with the variation in the wall-induced upwash along the model length. The latter becomes of importance if this length is of the same order as the tunnel height.

Once again it is convenient to discuss separately different sorts of models and working sections. First, the two-dimensional tunnel will be considered, and then three types of working section suitable for testing three-dimensional models.

### 6.5.1 Two-Dimensional Tunnel

As before, the aerofoil is considered to span the tunnel breadth, the solid sidewalls acting merely as reflection planes. The interference is due to the pressure of the slotted boundaries above and below the aerofoil which is replaced by a two-dimensional vortex of strength  $\frac{1}{2}UCC_{r}$ .

Wright<sup>6.28</sup> has developed for this particular case the approach contained in Reference 6.1, which allows for viscous effects within the slots. The potential of the vortex in free air is

$$\Phi_{\rm m} = -\frac{\rm UcC_{\rm L}}{4\pi} \tan^{-1}\frac{\beta z}{x} . \qquad (6.47)$$

At the slotted boundaries, the interference potential  $\Phi_j$  must satisfy Equation (6.12). The method of solution is then similar to that used for the solid-blockage interference, outlined in Section 6.3.1. In the general case, the induced upwash angle at the vortex position becomes

$$\frac{\mathbf{w}_{\mathbf{i}}}{\mathbf{v}} = \frac{1}{\mathbf{v}} \frac{\partial \Phi_{\mathbf{i}}}{\partial z} = -\frac{cC_{\mathbf{L}}}{2\pi h} \frac{\beta}{P} \int_{0}^{\infty} \frac{dq}{[\sinh q + Fq \cosh q]^{2} + \left[\frac{\beta}{P} \cosh q\right]^{2}} . \quad (6.48)$$

Consider first the ideal slotted tunnel ( $P \rightarrow \infty$ ); in this case Equation (6.48) reduces to the limit

2 MARTIN CARESONN - TA

369

`?`

$$\frac{w_1}{U} = -\frac{cC_L}{4h(1+F)} , \qquad (6.49)$$

which is independent of stream Mach number. This lift interference becomes zero when the tunnel walls are closed  $(F \to \infty)$  and has the value  $-\frac{1}{4}(c/h)C_{L}$  for an open-jet tunnel (F = 0). For the condition of zero solid blockage in an ideal slotted tunnel  $(F_{0} = 1.18)$  the lift interference is about 0.46 of that predicted for a two-dimensional open jet. Because it is proportional to c/h, the interference may be of considerable magnitude.

Holder (Ref. 6.58; 1962) has evaluated Equation (6.48) for a range of values of  $\beta/P$ . His results are presented in Figure 6.14 in terms of the parameter

$$\delta_0 = \frac{hw_1}{UcC_t} . \tag{6.50}$$

The induced upwash interference disappears for the condition P = 0, which represents a wall with extremely high resistance to cross-flow. In this condition, an important limiting case is the closed tunnel (P = 9,  $P - \omega$ ) in which  $\delta_0$  is known to be zero from earlier work on upwash interference. One important feature revealed by Figure 6.14 is that, for a given slot geometry (constant F), the upwesh interference decreases as the viscous effects grow. This suggests that in practice the actual interference is likely to be smaller than that predicted by Equation (6.49). The experiments described in Reference 6.24 provide evidence of such a trend, but the effect may not be attributed with certainty to the influence of viscous flow within the slots.

The general expression for the induced upwash angle along the tunnel axis may sometimes be required; Equation (6) of Reference 6.28 simplifies to give

$$\frac{w_{i}}{U} = -\frac{cC_{L}}{2\pi h} \int_{0}^{\infty} I_{D} dq , \qquad (6.51)$$

wherc

$$I_{\rm D} = \frac{\frac{\beta}{P}\cos\left(\frac{2qx}{\beta h}\right) \div \left[(1 - Pq)(\sinh q + Pq\cosh q) - \left(\frac{\beta}{P}\right)^2 \cosh q\right] e^{-q} \sin\left(\frac{2qx}{\beta h}\right)}{[\sinh q + Pq\cosh q]^2 \div (\beta/P)^2 \cosh^2 q}$$

For an ideal slotted tunnel  $(P \to \infty)$  Equations (6.43) and (6.5i) are inconsistent. It is explained in Reference 6.28 that the uniform upwash corresponding to Equation (6.49) must be added to Equation (6.51) to make w. vanish in the limit as  $x \to -\infty$ . Fortunately, if P is finite, there is no singularity in  $I_D$  at q = 0 and concequently no difficulty at  $x = -\infty$ .

For a completely closed tunnel, the upwash vanishes both far upstream and at the model position. The well-known lift-interference corrections for an aerofoil in a tunnel with closed boundaries result from the streamline curvature and are proportional to  $(c/h)^2$ , when higher-order terms are neglected. The streamline curvature effect in a slotied-wall tunnel at x = 0, can be obtained by differentiating Equation (6.51):

$$\frac{c}{J}\frac{\partial m_{1}}{\partial z} = -\frac{c^{2}C_{L}}{\pi \beta h^{2}}\int_{0}^{\infty}I_{E} c_{E} , \qquad (6.52)$$

where

$$I_{\rm E} = \frac{\left[(1-{\rm Pq})(\sinh q + {\rm Pq}\cosh q) - (\beta/{\rm P})^2 \cosh q\right] q e^{-q}}{\left[\sinh q + {\rm Pq}\cosh q\right]^2 + (\beta/{\rm P})^2 \cosh^2 q}$$

As Glauert<sup>6, 43</sup> points out, the correction to incidence,  $\Delta \alpha$  is the value of  $w_i/U$  at the mid-chord of the aeroful. The representative vortex must be placed at the centre of pressure, and the mid-chord is approximately at a distance  $c(\frac{1}{4} + C_g/C_L)$  downstream. Then

$$\Delta \alpha = \frac{\mathbf{w}_{i}}{\mathbf{u}} + \frac{c}{\mathbf{u}} \frac{\partial \mathbf{z}_{i}}{\partial \mathbf{x}} \left( \frac{1}{4} + \frac{c_{\mathbf{g}}}{c_{\mathbf{L}}} \right)$$
$$= \frac{cc_{\mathbf{L}}}{\mathbf{h}} \delta_{0} + \frac{c^{2}}{\beta \mathbf{h}^{2}} \delta_{1} (\frac{1}{4}c_{\mathbf{L}} + c_{\mathbf{g}}) , \qquad (6.53)$$

where, by Equations (6.49), (6.51) and (6.52),

$$\delta_{c} = \frac{hw_{i}}{UcC_{L}} = -\frac{1}{4(1 + F)} \quad \text{when} \quad P \to \infty$$

$$\delta_{a} = -\frac{1}{2\pi} \int_{0}^{\infty} I_{p} \, dq \quad \text{when} \quad P_{q} \text{ is finite}$$

$$\delta_{i} = \frac{\beta h^{2}}{UcC_{r}} \frac{\partial w_{i}}{\partial x} = -\frac{1}{\pi} \int_{0}^{\infty} I_{E} \, dq \quad \text{for all } P$$
(2.54)

The residual corrections due to streamline curvature are conveniently expressed as equivalent increments to the coefficients of lift and pitching moment about the quarter-chord point.

$$\Delta \mathbf{C}_{\mathbf{L}} = -\frac{\pi}{2\beta} \frac{\mathbf{c}}{\mathbf{u}} \frac{\partial \mathbf{x}_{\mathbf{j}}}{\partial \mathbf{x}} = -\frac{\pi}{2} \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^2 \delta_{\mathbf{i}} \mathbf{C}_{\mathbf{L}} \\ \Delta \mathbf{C}_{\mathbf{x}} = \frac{\pi}{8\beta} \frac{\mathbf{c}}{\mathbf{u}} \frac{\partial \mathbf{x}_{\mathbf{i}}}{\partial \mathbf{x}} = \frac{\pi}{8} \left( \frac{\mathbf{c}}{\beta \mathbf{h}} \right)^2 \delta_{\mathbf{i}} \mathbf{C}_{\mathbf{L}}$$
(6.55)

For a closed tunnel  $\delta_6 \approx 0$  and  $\delta_1 = \pi/24$ , and Equations (6.53) and (6.55) becase identical to Equations (2.13) and (2.17) of Chapter II.

For the ideal slotted tunnel  $\beta/P = 0$  and

$$I_g = \frac{(1 - Fq)qe^{-q}}{\sinh q + Fq \cosh q} . \qquad (6.56)$$

For zero solid blockage when  $F = F_0 = 1.18$ , Wright<sup>6.28</sup> has evaluated  $\delta_1 = -0.027$ , which is small relative to the corresponding values for closed and open tunnels. Values of  $\delta_0$  and  $\delta_1$  at other values of F may be obtained from Figure 6.15, where the abscissa  $(1 \div F)^{-1}$  is a useful parameter for presenting lift-interference curves. Zero streamline curvature occurs when F = 1.59.

Tirumalesa<sup>6.55</sup> has used the rhecelectric snalogy to treat the case where the model chord is of the same order as the tunnel height. The distribution of interference upw/sh along a two-dimensional model, displaced from the axis of the tunnel, is discussed in Reference 6.30.

#### 6.5.2 Circular Tunnel

Matchews<sup>6,13</sup> has extended the early fork of Pistolesi<sup>C,14</sup> and provides a general solution for the correct boundary conditions; In special circumstances the method of Reference 6.15 may be of great value, and in addition it provides a check on the use of the lass precise homogeneous boundary condition (Section 6.2). The linear form of this condition has been employed by several authors for a lifting wing in the centre of a circular slotted tunnel, and the method devised by Kaldwin et al.<sup>6.1</sup> is again of significance because of the way in which it can allow for viscous effects at the slots.

In this analysis the wing is replaced by a small horse-shoe vortex, whose potential in free air is

$$\Phi_{\rm y} = \frac{\rm USC_L}{5\pi} \left( 1 + \frac{\rm x}{\sqrt{\rm x^2 + \beta^2 r^2}} \right) \frac{\sin\theta}{\rm r} , \qquad (6.57)$$

where S is the wing planform area and x, r and  $\theta$  are the cylindrical co-ordinates. The fact that the actual span (2s) is finite introduces higher-order terms which are negligible at distances large compared with the model size. The interference potential  $\Phi_i$  sust satisfy Equation (6.26) and the boundary condition (6.27); in Reference 6.1,  $\Phi_i$  and its derivative  $\partial \Phi_i / \partial z$  are obtained by a Fourier transform method.

The upwash angle at the model position may be expressed as

$$\frac{\Psi_1}{U} = \frac{SC_L}{8\pi R^2} \left[ 1 - \frac{\beta}{\pi P} \int_0^\infty I_P \, \mathrm{d}q \right] . \qquad (6.58)$$

where

$$I_{F} = \frac{q^{2}}{q^{2}[(1 - F)I_{1}(q) + QFI_{0}(q)]^{2} + (\beta P)^{2}[I_{1}(q) - qI_{0}(q)]^{2}}$$

and, as in Equation (6.28), F = K/3. I<sub>0</sub> and I<sub>1</sub> are modified Bessel functions. The incidence correction is unaffected by compressibility, except through  $\beta/P$ .

For a small model it is convenient to express the lift interference as

$$\Delta \alpha = \frac{w_1}{U} = \frac{SC_L}{C} \delta_{g} , \qquad (6.59)$$

C being the tunnel cross-sectional area. In the present case  $C=\pi R^2$ , and the interference parameter

$$\delta_{0} = \frac{1}{8} \left[ 1 - \frac{\beta}{\pi P} \int_{0}^{\infty} I_{P} dq \right]. \qquad (6.60)$$

The values of  $F = F_0$  required to give zero lift interference are shown in Figure 6.16 against the slot porosity parameter  $\beta/P$ .

Devis and Moore<sup>6.17</sup> have shown that a wing of uniform spanwise loading, represented as a horse-shoe vortex, in an ideal slotted tunnel gives an interference parameter

$$\delta_0 = \frac{1}{8} \sum_{1}^{\infty} \frac{P_1 - 1}{r_1(P_1 + 1)} \left(\frac{s}{R}\right)^{2n-2} \qquad (r_1 = 1, 3, 5, ...) \quad . \tag{6.61}$$

which is obtained as an average along the semi-span  $\hat{\upsilon} \leq y \leq s$ . Even for quite large values of s/R, only the first term need be considered, so that it is sufficient to write

$$\delta_0 = \frac{1}{8} \left[ \frac{F-1}{F+1} \right] . \tag{(8.62)}$$

which is plotted as a straight line against  $(1 + F)^{-1}$  in Figure 6.17. It can be shown that Equation (6.60) tends to this result, when the ideal slotted tunnel is considered by the limiting process  $P \to \infty$ . In Reference 6.17 Equation (6.62) is compared with the exact values obtained by Matthews<sup>6.15</sup> for tunnels having 4, 8 and 12 slots; excellent agreement is obtained between the two methods of solution.

The condition for zero upwash interference corresponds to the value  $F_0 = 1.0$ , compared with a value of 0.57 to eliminate solid-blockage effects. It is apparent therefore that it is not possible to eliminate both upwash and blockage interference simultaneously in a circular ideal slotted tunnel, but, as Figure 6.16 shows, this may will be achieved if  $\beta/P$  has a value around 0.7. At F = 1.0, the relationship between the total number of slots and the open-area ratio is as follows.

| N                             | 2    | 4           | 6   | 8   | 12    | 16   | 24     |
|-------------------------------|------|-------------|-----|-----|-------|------|--------|
| $130\left(\frac{z}{d}\right)$ | 24.0 | <b>8.</b> 6 | 3.2 | 1.2 | 0. 16 | 6.02 | 0.0004 |

The ideal zero-blockage condition may well be the one for which the tunnel is designed;  $\delta_0$  is then ~0.034, which is of the opposite sign to the correction is a solid-wall tunnel and about 27% of its magnitude.

If the model chord is of the same order as the tunnel disacter, some correction for streamline curvature may be required. For the case where  $\beta/P$  is not zero the curvature can be obtained as

1.164 Star

$$\frac{\overline{o}}{\overline{o}} \frac{\overline{\partial \mathbf{x}_1}}{\overline{\partial \mathbf{x}}} = \frac{\overline{o}}{\overline{o}} \left( \frac{\overline{\partial^2 \Phi_1}}{\overline{\partial z \partial \mathbf{x}}} \right)_{\mathbf{x} = \mathbf{r} = 0}$$

where  $\bar{c}$  is the geometric mean wing chord and  $\bar{\Phi}_i$  is given by Equation (46) of Reference 6.1. Reynolds et al.<sup>6,23</sup> give curves of the corrections for a value of F = 0.191 appropriate to the Cooperative Wind Tunnel in California with different values of the porosity parameter P. The total correction to incidence takes the form

$$\Delta \alpha = \frac{SC_L}{\pi R^2} \left( \delta_0 + \frac{\overline{c}}{4\beta R} \delta_1 \right) , \qquad (6.63)$$
$$\delta_1 = \frac{2\pi\beta R^3}{USC_L} \frac{\partial w_1}{\partial x}$$

where

is independent of Mach number. The second term in Equation (6.63) represents the increment in  $w_1/0$  associated with a displacement  $\frac{1}{2}\overline{c}$  in the streamwise direction, where the aerodynamic mean chord

$$\overline{\overline{c}} = \int_0^1 \frac{[c(y)]^2}{\overline{c}} d\left(\frac{y}{s}\right) . \qquad (6.64)$$

This choice of  $\Delta \alpha$  approximately eliminates any residual correction to lift, but there remains an approximate pitching-moment correction

$$\Delta C_{\mu} = \frac{\partial C_{L} \lambda x_{1}}{\partial \alpha} \frac{\partial w_{I}}{\partial u}, \qquad (6.65)$$

where the planform parameter  $\lambda x_1/c$  is discussed in Sections 3.2.4 and 3.3.4 and typical values are given in Table 3.111 of Chapter III.

According to Guderley<sup>5.20</sup>, the curvature correction is very small in an ideal slotted tunnel designed for zero solid blockage. Wright<sup>5.28</sup> estimates it to be about -1% of the correction appropriate to a closed circular tunnel; he adds that no adequate experimental data exist for checking the interference corrections in a circular slotted tunnel.

If the wing is so large and slender that the wing and tunnel may be treated as a slender configuration, then the Munk-Jones theory<sup>6.41</sup> may be used. In this case Tirumalesa<sup>6.37</sup> has derived a value of the lift-interference parameter,

$$\frac{C(\Delta \alpha)}{SC_{L}} = S_{0} = \frac{1}{4} \left[ \frac{F-1}{F+1} \right]. \qquad (6.66)$$

which is just twice the value given for a small wing in Equation (6.52).

ちょうろうろうというというという

Carry . The first france of a burner and a lar a second fraction for the

weather and a solution of the

# 6.5.3 Rectangular Tunnel with Slotted Roof and Floor

As in the calculation of the solid-blockage interference, the effects of the solid side-walls may be adequately represented by a horizontal row of vortices extending in a plane containing the wing span. To the potential  $\hat{\Psi}_r$  associated with this row may be added a potential  $\hat{\Psi}_g$ , so that the total  $\hat{\Psi} = \hat{\Psi}_r + \hat{\Phi}_g$  will satisfy the boundary conditions of Equation (6.35).

The solution for a uniformly-loaded wing in an ideal elotted tunnel is set out in Equation (43) of Reference 6.17, where an average interference upwosh in the plane of the wing is derived. When the analysis is completed by substituting the coefficients of the Fourier cosine series for  $\Phi_{\mu}$  and  $\partial \Phi_{\mu}/\partial z$ , we obtain, corresponding to Equation (6.59), the interference parameter

$$\delta_{0} = \frac{h}{4\pi\sigma^{2}b} \left[ \log_{e} \left( \frac{\pi\sigma}{\sin \pi\sigma} \right) - \sum_{n=1}^{\infty} A_{n} \sin (n\pi\sigma) - \frac{\pi\sigma^{2}b}{h(1+F)} \right], \quad (6.67)$$

where

ind Marial Automotive Service

$$\sigma = 2s/b$$
.

$$nA_{n} = \frac{4(b - n\pi hF) \sin (n\pi r)}{(b + n\pi hF)e^{2n\pi h/b} - (b - n\pi hF)}$$

In the limiting case of small span this reduces to

$$\delta_{0} = \frac{\pi h}{24b} - \frac{\pi h}{b} \sum_{n=1}^{\infty} \frac{h}{\theta_{n} e^{2n\pi h/b} - 1} - \frac{1}{4(1+F)} , \qquad (6.68)$$

where

The influence of the model size on the correction factor is illustrated in Figures  
6.18 and 6.19 for tunnels with 
$$b/b$$
 equal to 1.0 and 0.5. As in the case of a circular  
ideal slotted tunnel, the conditions for zero upwash interference do not correspond to  
those required for zero blockage interference. The induced upwash in this will condi-  
tion is guite large: for example, a square tunnel designed for zero solid blockage in

 $\theta_{n} = \frac{b + n\pi bF}{b - n\pi bF}$ 

calculated to have a fift interference of about -0.63 of the value for a closed tunnel. Maeder et al. (Rev. 6.29; 1955) give a similar equation to (6.67) above for the

and the correct expression is

$$\delta_{\sigma} = \frac{h}{4\sigma b} \left( \frac{2}{n\sigma} - \cot \frac{\pi \sigma}{2} \right) - \frac{h}{4\sigma b} \sum_{B=1}^{\infty} bA_{B} - \frac{1}{G(1+F)} . \qquad (6.69)$$

where  $M_{\rm p}$  is defined in Equation (6.07). They point cut, however, that the first term is essentially associated with the side-wall image system and this does not vary with the slot parameter F. The third term is identically the incidence correction for a two-dimensional zerofoil in Equation (6.54) and thus may be imagined as giving

the influence due to an aerofoil spanning the tunnel breadth and having the same total lift as the wing. These authors conclude therefore that the summation term in Equation (6.67) may often be neglected, as it merely indicates the difference in the slotted-wall case between a mean two-dimensional vortex system and an exact vortex system. The approximate equation for  $\delta_0$  then becomes

10

$$\delta_0 = (\delta_0)_c - \frac{1}{4} \left( \frac{1}{1+F} \right)$$
 (6.70)

which is independent of stream Mach number;  $(\delta_0)_c$  denotes the value of  $\delta_0$  in a closed tunnel, to be found in Chapter III for different value of h/b. For a square tunnel, the discrepancy between the two equations is small, even when the ratio of model span to tunnel breadth is as small as 0.1 (Fig. 6.18). At h/b = 0.5, however, the approximate formula is no longer valid (Fig. 6.19) due to the dominant effect of the slotted boundaries. Katzoff and Barger (Ref. 6.30; 1959) have discussed the limits of this approximate method and give corrections which may be applied to values of  $\delta_0$  obtained in this way. By using the results contained in Reference 6.30, it is possible to obtain easily the upwash distribution along the tunnel axis.

Wright<sup>6.23</sup> discusses an interesting difference between the approach of References 6.25 and 6.29 on the one hand and Reference 6.30 on the other. Katzoff and Barger ascribe the slotted-wall effect mainly to the interference on the bound vortex, whilst the effect on the trailing vortices is used merely to derive a small correction. Number and his associates<sup>6.25,6.29</sup> consider the main correction to come from the interference of the slotted walls in the presence of trailing vortices.

The streamline curvature at the model is also discussed in Reference 6.29. Similarly to Equation (6.63), the total correction to incidence is expressed as

$$\Delta \alpha = \frac{SC_{L}}{C} \left( \delta_{0} + \frac{1}{2C} \frac{\overline{c}}{\beta_{1}} \delta_{1} \right) , \qquad (6.71)$$
$$\dot{\delta}_{1} = \frac{\beta_{1}C}{USC_{L}} \frac{\partial \alpha_{1}}{\partial x} .$$

where

The correction term in Reference 6.29 is equivalent to

$$\delta_1 = (\delta_1)_c - \frac{h}{2\pi b} \int_0^\infty \left[ \frac{1}{be^q - 1} + \frac{1}{e^q + 1} \right] q \, dq , \qquad (6.72)$$

where

$$= \frac{h + Fbq}{h - Fbq}$$

B

and  $(\delta_1)_c$  denotes the value for a closed turnel (Chapter III). The variation with slot geometry for a turnel with h/b = 0.85 is shown in Figure 6.20 as a full line. It should be noted that this result is very approximate; the limiting value  $\delta_1 = -0.12$ for F = 0 compares with the exact value  $\delta_1 = -0.21$  for an open floor and roof (Chapter III). Note accurate results have been obtained by Rushton<sup>6,61</sup> using a threedimensional electrical analogue, and these are shown in Figure 6.20. The difference

376

between the two curves is quite large, particularly when the wall is designed to give zero solid blockage.

For a square ideal slotted turnel designed to operate at zero solid blockage, the streamline curvature is about -0.28 of the value obtained in a square closed tunnel, if the model is reasonably small<sup>6,28</sup>.

ころうちょうちょうちょうちょうちょうちょう シンティーション シートレー しょうちょうちょうちょうちょうちょうちょう ちょうしょう

いってい いっち

STATES PE

377

The influence of viscous flow within the slots on the results discussed in this section has been considered by Holder<sup>6.58,5.59</sup>. He has extended the analysis of Reference 6.17 to satisfy the boundary condition (6.7). The average interference parameter in the transverse plane of a uniformly-loaded wing can be written as\*

 $\delta_0 = (\delta_0)_c - \frac{\beta}{2\pi^p} \int_{-\infty}^{\infty} \frac{dq}{\left[\sinh q + Fq \cosh q\right]^2 + \left[\frac{\beta}{p} \cosh q\right]^2} -\frac{\beta}{\pi P} \sum_{n=1}^{\infty} \left[ \frac{\sin(n\pi\sigma)}{n\pi\sigma} \right]^2 \int_{0}^{\infty} \frac{dq}{\left(\frac{q}{\alpha_n}\right)^2 \left[\sinh\alpha_n + F\alpha_n \cosh\alpha_n\right]^2 + \left[\frac{\beta}{P} \cosh\alpha_n\right]^2} \quad (6.73)$  $x_{n} = \left[ q^{2} + \left( \frac{n\pi n}{h} \right)^{2} \right]^{\frac{1}{4}}$ where

の日本のないのためになったいのかないという

The second term is precisely the interference upwash of a two-dimensional merofoil in Equation (6.48). For an ideal slotted tunnel ( $P \rightarrow \varpi$ ), this reduces to the second term of Equation (6.70) and the third term of Equation (6.73) therefore represents a correction to Equation (6.70). It can be shown that Equation (6.73) reduces to Equation (6.67) in the limit as  $P \to \infty$ .

Equation (6.73) has been evaluated by Holder for a range of values of  $\beta'P$ . for a square tunnel cross-section and one where h = 0.5b. These results are shown in Figures 6.21 and 6.22 for the case of a very small model ( $\sigma = 0$ ), and also where the wodel span is half the tunnel breadth ( $\sigma = 0.5$ ). For a given slot geometry, the effect of the viscous flow in the slots is to increase the induced upwash, and for values of  $\beta/P$  greater than about unity it may no longer be possible to obtain the zero liftinterference condition. The value of Po for this condition is also affected by the span of the wing, the effect dying particularly marked for the duplex tunnel crosssection. The variation of  $P_n$  with  $\beta/P$  is shown in Pigure 6.23.

The preceding discussion in this section has been based on the use of the honogeneous boundary condition at the slotted walls. By employing an electrical analogue technique, however, it is possible to obtain results for ideal slot flow in which the discrete slots and slats are represented. Thus a direct comparison between the use of exact and sporoxiBate boundary conditions may be made, and this has been done by Rushton<sup>6,60</sup> for the case of a small lifting wing in the centre of a rectangular tunnel with slotted roof and floor. The exact boundary conditions are represented in a linearized form by

\* An error in Equation (9) of Reference 6.59 has been corrected here.

Equations (6.3). The interference potential  $\Phi_i$  satisfies Laplace's equation subject to the discrete boundary conditions at the roof and floor in place of the homogeneous condition (6.12). The resistance network automatically solves the finite-difference form of Laplace's equation, and in Reference 6.60 the boundary conditions are applied to the voltages in finite-difference form, if necessary. Then the electrical potentials are measured on the network, and these may be equated to the velocity potentials.

Rushton has investigated the validity of the homogeneous boundary condition for an ideal slotted wall by considering a rectangular tunnel with h/b = 0.5. The slotted roof and floor had an open-area ratio (a/d) equal to 0.125. Two symmetrical arrangements of the slots are possible, with either a slot or a slat directly above and below the small wing. Four cases of each arrangement were considered, with a total of N = 2,3,4 or 6 slots forming the ventilated wall and, if necessary, half slots at the corners. For the fixed open-area ratio the slot parameter F is inversely proportional to N . The interference parameter  $\delta_0$  obtained with these discrete arrangements of slots may be compared with those deduced from Equation (6.68), and this comparison is shown in Figure 6.24 against  $(1 + P)^{-1}$ . It is clear that, as the number of slots is increased. the differences between the exact and the homogeneous boundary conditions become smaller, and for  $N \ge 6$  these differences may be neglected. Further, the value of  $\delta_{n}$  deviates towards that for an open or closed tunnel according as a slot or slat lies directly above the wing. Some points obtained from the electrical analogue, but with the homogerrous boundary condition at the slotted wall, are plotted in Figure 6.24. These agree well with the analytical solution from Equation (6.68) and illustrate the validity of the analogue technique.

It is perhaps worth stressing that these results are for the small ratio n/b = 0.5and for one particular value of the open-area ratio (0.125). More favourable results would be expected for larger h/b: many tunnels operate with an open-area ratio not very different from 0.125, and it is unusual to have fewer than six slots in a ventilated wall. One may conclude then, that in most cases of practical importance the use of the simple homogeneous boundary condition is satisfactory.

### 6.5.4 Rectangular Tunnel with Slots on Each Wall

The lift interference experienced by a small wing at the centre of an ideal slotted rectangular tunnel has been considered by Davis and Moore<sup>6,17</sup>, who use a method which involves the transformation from a uniformly slotted circular tunnel. The widths of the individual slots in the rectangular tunnel vary, and their dimensions may be found by calculating the points which correspond to the slot edges in the circular tunnel. The technique therefore is less precise than that used for other cross-section shapes and geometries.

The general solution for the interference parameter  $\delta_0$  in the slotted tunnal involves elliptic functions, but very simple approximations for  $\delta_0$  may be derived. For a square tunnel, Davis and Noore suggest that

$$\delta_0 \simeq 0.137 \left[ \frac{F-1}{F+1} \right]$$
 (e.74)

where F is defined in Figure 6.1; and for extreme rectangular cross-sections

378

:: Ì

ş

$$\delta_0 \simeq \frac{\pi}{48} \frac{b}{h} \left[ \frac{F-2}{F+1} \right] \qquad \left( \frac{h}{b} \leq 0.5 \right) \tag{6.75}$$

and

$$\delta_0 \simeq \frac{\pi}{48} \frac{h}{b} \left[ \frac{2F-1}{F+1} \right] \qquad \left( \frac{h}{b} \ge 2.0 \right) \qquad (6.76)$$

These equations give good approximations to the limiting values of  $\delta_c$  for completely closed or open tunnels. A straight line joining these limiting values results if  $\delta_0$ is plotted against  $(1 + F)^{-1}$ . Equation (6.74) is very similar in form to that given in Equation (6.62) for a circular ideal slotted tunnel. As shown in Figure 6.17, both these results are close to that for a square tunnel with solid side-walls and ideal slotted roof and floor. This would seem to justify in part an assumption, commonly made, that the interference in a square slotted tunnel is equal to that obtained in a circular slotted tunnel of the same cross-sectional area. Reynolds et al.<sup>6.23</sup> suggest that the curvature correction for a near-square tunnel with all four walls slotted may be based on the analogous Equations (6.63) and (6.65) above for the slotted circular tunnel having the same cross-sectional area.

The method of analysis proposed by Davis and Moore is rather unsatisfactory, and it is therefore particularly valuable to have results from the electrical-analogue technique for rectangular tunnels with all four walls slotted. These results, presented by Kushton in Reference 6.60, correspond to the idea! homogeneous boundary condition for slotted walls. Three tunnel geometries, with h/b = 0.5, 1.0 and 1.6, are considered over the whole range of slot parameter F. In Figure 6.25 the full curves from the analogue experiments are compared with the results deduced from the linear relationship suggested by Davis and Moore. Between h/b = 0.5 and h/b = 1.0 the linear approximation does not lead to large discrepancies, but more serious errors are incurred when h/b = 1.6. The data given by Rushton are probably sufficient to allow corrections to be estimated for most tunnel geometries at prese ' in use.

Zero upwash interference in a square slotted tunnel does not correspond to zero blockage; at the former condition the interference correction due to blockage is about 0.12 of the value appropriate to a square tunnel (i.e.,  $\Omega_s = 0.12$  in Figure 6.12). Conversely, for the square tunnel operating at zero blockage, the upwash will be about -0.28 of the closed-tunnel value. The evidence in Figures 6.13 and 6.25 does suggest, however, that both interference effects could be eliminated in a rectangular tunnel having ideal slots on each wall with F = 0.8 approximately and a ratio h/b between 1.4 and 1.5.

# 6.6 SOME GENERAL REMARKS ON INTERFEBENCE EFFECTS IN TUNNELS WITH LONGITUDINAL SLOTS

In the preceding pages, an attempt has been made to outline the techniques and results at present available to correct aerodynamic data obtained in wind tunkels with longitudinally-slotted walls. It is apparent that the most important types of working section geometry and shape have been analysed, for small models at least, in sufficient detail to enable the required calculations to be made. A summary of the relevant figures and equations is given in Table 6.1.

and the state of t

The theory of Reference 6.1 is particularly valuable, in that for certain cases it enables some estimate to be made of the influence of viscous effects on the slots. The theory indicates that in compressible flow the primary parameter is not 1/P, but  $\beta/P$ , so that the conditions required for zero blockage or zero lift interference will vary with Mach number, even if P remains constant. As the stream Mach number approaches unity,  $\beta/P$  will tend to zero, so that the ideal slotted tunnel conditions will be approached. However, it is just in this range that the theory itself becomes less reliable, and the magnitude of the interference effects is probably best determined by experimental comparisons. An allowance for the viscous effects, though complicating the interference correction, is of great value in suggesting a mechanism by which in real tunnels both blockage and lift interference may be eliminated or miumized for a given slot arrangement (Figs. 6.16 and 6.23). The viscous effects, linked as they are to  $\beta/P$ , may also be responsible for the disagreement sometimes observed between the trend with Mach number of the experimental results and that predicted for an ideal slotted tunnel<sup>6.24, 6.25, 6.31</sup>.

It is likely that the theoretical corrections based on the linear boundary conditions of Equations (6.1) and (6.7) would have most validity when applied to reasonably small models, for with these any distortion of the flow field which may correspond to a second-order effect is likely to be small also. For larger disturbances at the boundary, the non-linear condition of  $Wood^{6.57}$  must be considered. Wright<sup>6.28</sup> also suggests that the lift interference is less well determined than the blockage interference in a slotted tunnel, due to uncertainties in the boundary conditions for the former. This consideration is likely to be of greatest significance for two-dimensional flows because of the appearance in these cases only of the conditions at infinity upstream. Wright argues that any significant downward displacement of the tunnel flow resulting from the lift imposed on the model will cause an increase in the inflow on the wall above the wing and hence an increased boundary-layer thickness. The actual boundary condition in this region might then approach that for an open tunnel or resemble the non-linear condition of Equation (6.4).

Despite reservations of this type, the progress made in recent years towards understanding the general problem of slotted-wall interference has been considerable, due mainly to the use of the equivalent homogeneous boundary conditions. Their validity for most types of problem has been established beyond doubt, provided the disturbance due to the model is small.

It will be apparent from the foregoing text that, for a tunnel designed to have zero-blockage with a small model, the interference corrections will be considerably smaller than those appropriate to either 'closed or an open tunnel. In many cases the effects of the tunnel walls can be neglected altogether; it is important, however, to consider whether this action is justified for each combination of model and tunnel, and whether the interference quantities could become comparable with the permitted telerances in the observations.

### 6.7 THE PERFORATED WALL

The other form of ventilated boundary in common use contains regularly-spaced openings into the surrounding plenum chamber, the dimensions of the openings being small compared with the tunnel height or breadth. Such walls are usually called

The second s

'perforated'. The perforations may be of various shapes, but most frequently they are circular, with the axes of the holes normal to the plane of the wall. Occasionally, for reasons connected with the need to minimize wave reflection at low supersonic stream speeds, the hole axes may be inclined away from the normal.

There is an important difference in the behaviour of a wall containing an array of small perforations and one having longitudinal slots along the working section. In the former case any pressure difference between the working section and the plenum chamber is primarily due to viscous effects in the slot, the resulting pressure change being given by Equation (6.6). With a perforated wall a pressure difference is associated with a cross-flow, even if viscosity is ignored. For convenience, the solid part of the wall may be likened to a lattice of small lifting wings whose incidence is the flow inclination to the wall  $(v_n/U)$ . Thus one may write approximately

$$\delta p \propto \frac{1}{2}\rho U^2 \frac{v_n}{U} \propto \rho U v_n$$
 (6.77)

so that

A STATE AND A STAT

$$\delta p = p - p_c = \frac{\rho u}{p} v_n = \frac{\rho u}{p} \frac{\partial \Phi}{\partial n}. \qquad (6.78)$$

where  $p_c$  is the pressure in the plenum chamber. This is precisely the Equation (C.6) used for estimating the influence of viscosity on interference effects in tunnels with longitudinal slots. Unless otherwise stated, it will be assumed that a perforated wall has the linear pressure-drop characteristics, implied by Equation (6.78), where the porosity parameter, P, is mainly dependent on the wall geometry and stream velocity. In general, P must be determined experimentally for the particular wall geometry of interest and at the required stream Mach number. Theoretical values for P have so far only been obtained for single and multiple slots transverse to the flow direction (Refs. 6.19, 6.32, 6.42, 6.47). In the case of the single transverse slot it has been shown<sup>6.54</sup>, contrary to Reference 6.57, that separation on the plenum-chamber side of the wall does not affect the linear form of the pressure-drop equation.

The experimental cross-flow characteristics of perforated walls are discussed in some detail by Goethert in Reference 6.32, and further information is contained in References 6.22, 6.38 and 6.47. For circular holes with axes normal to the wall there may be a small decrease in P as the stream Mach number rises. The wall boundary layer, which develops more repidly along a perforated wall than along a longitudinallyslotted wall 6. 46, 6. 48, 6. 53, 6. 56, may have a marked influence on the observed porosity parameter and on the range over which the linear variation of pressure diff nce with flow inclination is maintained. Markedly non-linear effects can be obtained at small cross-flows, if the displacement thickness of the boundary layer is large compared with the hele diameter. Similar deviations may be obtained if the wall thickness is comparable with the hole size; in addition, the hole gay now act as a diffuser, so that the overall pressure change across the wall is reduced. It has also been found experimatally<sup>6,52</sup>, that the cross-flow characteristics of a perforated wall are changed, if the working-section length is short compared with the tunnel height. These and other aspects of perforated-wall behaviour are considered in Reference 6.32; it is sufficient here to note that the theoretical analysis of boundary interference in perforated-wall tunnels assumes a pressure-drop relationship of the linear form given by Equation (5.78)

and a simple linearized homogeneous boundary condition at the perforated wall, which is equivalent to Equation (6.5), viz.

$$\frac{\partial \Phi}{\partial x} + \frac{1}{P} \frac{\partial \Phi}{\partial n} = 0 . \qquad (6.79)$$

It is perhaps necessary to consider briefly the difference in the behaviour of a perforated wall, and one which is truly porces. In the latter case friction effects dominate the flow, and the pressure drop is directly proportional to the mass flow through the wall  $^{6.94}$ . The equation analogous to (6.78) is then

$$\delta p = p - p_c = \frac{\rho U}{P} v_n = \frac{\rho U}{(P^2 U)} \frac{\partial P}{\partial n}$$
 (6.80)

The porosity parameter P for a perforated wall is replaced by P'U and thus depends directly on the stream velocity. This is an undesirable characteristic which is partly responsible for the limited use made of percus-wall tunnels. Analyses for wall interference in tunnels with porous walls (e.g., Ref. 6.21) are immediately applicable to the perforated case, provided that the porosity parameter is defined by means of Equation (6.78). Conversely, the results set out in the subsequent sections of the present text, with perforated tunnels in mind, may be used for tunnels with percous walls.

It is apparent that an actual perforated well may, in certain flow conditions, not conform to the behaviour of the idealized wall assumed in the analysis; to this extent it may be argued (as in Reference 6.28) that the corrections for wali-interference effects in perforated tunnels are less soundly based than in the case of longitudinallyslotted boundaries. Nevertheless, it is felt that the simple theory at present available does give a valuable indication of the approximate magnitude and trends of the interference effects in wind tunnels having perforated boundaries.

As will be shown below, the parameter determining the interference-free flow is  $\beta/P$ , where P is defined by Equation (6.78) and is dependent mainly on the wall geometry. Thus, to maintain a constant value of  $\beta/P$  over the complete range of test Mach number, the well geometry would need to be continuously varied. In contrast, the interference-free condition for an ideal slotted tunnel i. usually independent of stream Mach number and attained simply through a specific value of the geometric parameter  $P = P_0$  (Pig.6.1). This distinction is sometimes used as an argument, that the perforated wall is less suitable as a boundary for subsonic tunnels than one having longitudinal slots. However, when viscous effects are present, so that  $F_0$  is now a function of  $\beta/P$ , experimental evidence suggests that the value of P for slotted walls is not greatly infinenced by stream Mach number, and hence, as in the case of a perforated wall, continuous variation of vall geometry is strictly required to maintain interference-free conditions. The relative advantages of the two types of wall are than less obvious.

As the stream Mach mumber approaches unity,  $\beta/P$  tends to zero, and the perforated tunnel tehaves more and more like an open jet. In this flow regime the linearized theory, on which the analysis is based, is of limited validity; the interference present must therefore be calculated from flow equations with the proper transonic approximations.

382

However, the choice of the geometry of a perforated wall may be influenced by considerations other than the need to minimize subsonic wall interference. For tunnels which are required to operate at transonic and low supersonic speeds, a perforated wall can be designed to reduce shock-wave reflections<sup>6.32</sup>; the open-area ratio of such a wall (as well as the inclination of the holes to the wall surface) may well be very different from that which would be chosen solely to satisfy the need for smell wall interference at subsonic speeds. More specifically, the volue of  $\beta/P$  for a wavecancellation wall may be as low as 0.3, so that the tunnel behaves rather like an open-jet configuration and corrections for wall interference are far from negligible. In some cases, however, there may be an incidental advantage in operating the tunnel away from zero-interference conditions; the longitudinal pressure gradient clong the tunnel axis and the streamline curvature may then be scaller, so that a more uniform interference effect is achieved over the whole length of the model.

It is clear that the actual design of a perforated wall can be a compromise between conflicting requirements. Nevertheless in the present text some stress will continue to be laid on conditions which achieve zero wall interference at subsonic speeds. Sufficient data are presented, however, to enable the corrections to be calculated for a wide range of  $\beta/P$ .

# 6.8 THE-DIVENSIONAL BLOCKAGE IN TUNNELS WITH PERFORATED WALLS

### 6.8.1 Solid Blockage

The solid-blockage interference experienced by a two-dimensional model in a perforated tunnel was considered independently by Goodman (Ref. 6.21; 1950) and Kassner (Ref. 6.35; 1952). Both authors use the boundary condition (6.79). Later, a solution was obtained by Baldwin et al.<sup>6.1</sup> as a special case (K = 0) of the homogeneous boundary condition (6.7; for longitudinal slots with viscous flow. The interference potential  $\Phi_1$ , and hence the longitudinal interference velocity increment at the model position  $\Delta \hat{U}_q$ , may be found similarly to that for lon tudinal slots, discussed in Section 6.3.1.

Alternatively a Fourier analysis method, or, following Kassner, a modified image method, may be employed. The solid-blockage factor  $\epsilon_s$  is most conveniently expressed in terms of  $\Omega_s$ , the ratio of its value in the ventilated tunnel to that in a geometrically-similar tunnel with closed walls. For a two-dimensional model in a perforated-wall tunnel a simple equation for this ratio is

$$\Omega_{\rm s} = 1 - 6\left(\frac{\mu}{\pi}\right) + 6\left(\frac{\mu}{\pi}\right)^2$$
, (6.81)

where  $\mu = \cot^{-1}(\beta/P)$  may vary between zero and  $\frac{1}{2}\pi$ , the value for an open-jet tunnel. The variation of  $\Omega_s$  with poresity parameter is shown in Figure 6.26. Zero solid blocksge occurs when  $\beta/P = 1.28$ ; a result close to this, but derived in a different way, was obtained by Brescia (Ref. 6.40; 1952).

The boundaries of a perforated tunnel, unlike ideal slotted walls, induce a velocity gradient along the model due to the solid blockage. At the model position this gradient may be obtained from Equation (6.18) with F = 0.

383

and a second second

$$\frac{\partial \epsilon_{\rm s}}{\partial x} = \frac{24}{\pi^2 \beta h} (\epsilon_{\rm g})_{\rm c} \frac{\beta}{\rm P} \int_{0}^{c^2} \frac{q^2 dq}{\cosh^2 q + (\beta/{\rm P})^2 \sinh^2 q} . \tag{6.52}$$
$$\simeq \frac{12}{\pi^2 \beta h} (\epsilon_{\rm s})_{\rm c} .$$

if the tunnel is designed to give zero blockage ( $\beta/P = 1.28$ ). Figure 6.27, taken from Reference 6.35, shows how  $\epsilon_g$  varies along the tunnel for different wall porosities. The correction to measured drag may be calculated from Equation (6.12) with P = 0. The broken curve in Figure 6.27 suggests that  $(\Delta C_D)_{Sg}$  is roughly a maximum when  $\epsilon_g(3)$  is zero.

In Section 7.28 of Reference 6.19 Woods has considered the more general problem of a two-dimensional tunnel with perforated walls of finite length. He derives a simple, but lengthy, expression for  $\epsilon_g(x)$  which is shown to become consistent with Equation (6.51) when the perforated walls are of infinite streamwise extent.

# 6.8.2 Wake Blockage

Wright<sup>6.28</sup> has considered the wake-blockage effects experienced by an aerofoil in a perforated tunnel by methods similar to those of Reference 6.1. The ratio of the wake-blockage factor in the perforated tunnel to that in the closed tunnel may be denoted by  $\Omega_{\rm w}$ , and for the two-dimensional aerofoil

$$\Omega_{\rm w} = -\frac{2}{\pi} \frac{\beta}{P} \int_{0}^{\omega} \frac{\mathrm{d}q}{\cosh^2 q + (\beta/P)^2 \sinh^2 q} = -\frac{2}{\pi} \tan^{-1} \left(\frac{\beta}{P}\right) . \quad (6.83)$$

The variation of  $\Omega_w$  with  $\beta/P$  is shown in Figure 5.26. At zero solid blockage,  $\Omega_w$  is about -0.57; for  $\beta/P > 1.28$ , the wake blockage would tend to counter the solid blockage.

Whereas  $\Omega_s$  in Equation (6.81) tends to the correct limits  $-\frac{1}{2}$  and 1 for open  $(\beta/P = 0)$  and closed  $(\beta'P \to \infty)$  tunnels respectively,  $\Omega_w$  only tends to the correct limit in the former case. Just as for an ideal slotted tunnel  $\Omega_w = 0$  for finite values of F, so here  $\Omega_w$  is negative for finite values of  $\beta/P$ ; neither tends to the definition  $\Omega_w = 1$  for a closed tunnel. In the analysis of Reference 6.23, the model wake is represented by a single source and the interference velocity becomes zero far upstream of the model; the wake-blockage correction in Equation (6.83) is simply the interference velocity at the model position. The same is not true of a closed tunnel, unless the model wake includes a sink of equal strength far downstream.

The velocity gradient at the model position due to the wake images may also be derived formally<sup>6.23</sup>. The velocity gradient due to a source is closely related to the velocity increment associated with a doublet, and hence the wake-blockage gradient vanishes when the solid blockage is zero. In general,  $\partial \epsilon_{\rm w}/\partial x$  and the corresponding drag correction are given by Equations (6.21) and (6.22), where  $\Omega_{\rm g}$  now takes the appropriate value from Equation (6.81).

osta e o sta ale actuale de la la casta as suastantes de la casta de la casta de la casta de la casta de la cas

## 6.9 THREE-DIMENSIONAL BLOCKAGE IN TUNNELS WITH PERFORATED WALLS

### 6.9.1 Circular Tunnel

The solid blockage of a small body in a circular tunnel of radius R may be solved by similar acthods to those used in considering the slotted tunnel. Goodman<sup>6.36</sup> provided the first solution in 1951. At the model position, the blockage-factor ratio  $\Omega_{\rm g}$  becomes

$$\Omega_{\rm g} = -0.40 \, \int_0^\infty {\bf I}_{\rm G} \, d{\bf q} \qquad (6.84)$$

with

$$I_{G} = \frac{[K_{0}(q)I_{0}(q) - (\beta/P)^{2}K_{1}(u)I_{1}(q)]q^{2}}{[I_{0}(q)]^{2} + [(\beta/P)I_{1}(q)]^{2}}$$

where the modified Bessel functions  $I_0$ ,  $I_1$ ,  $K_0$  and  $K_1$  are tabulated in Reference 6.27. The variation of  $\Omega_s$  with  $\beta/P$  is shown in Figure 6.28, and zero solid blockage occurs for  $\beta/P$  equal to 1.22.

Once again the velocity gradient at the model position may be obtained directly by twice differentiating the interference velocity potential from Equation (34) of Reference 6.1; hence

$$\frac{\partial \epsilon_{s}}{\partial x} = (\epsilon_{s})_{c} \frac{0.40}{\beta R} \left[ \frac{\beta}{P} \int_{0}^{\infty} \frac{q^{2}}{[I_{0}(q)]^{2} + [(\beta/P)I_{1}(q)]^{2}} dq \right]. \quad (6.85)$$

According to Wright<sup>6, 28</sup> the expression in square brackets is equal to 1.47 when  $\beta/P$  has the value 1.22 appropriate to zero solid blockage. This is confirmed by independent calculation and, not unlike the two-dimensional case in Figure 6.27, the expression is close to its maximum value that occurs near  $\beta/P = 1.6$ .

In considering the say: blockage, the model wake may once more be represented by a three-dimensional source. The wake-blockage factor at the model position is obtained as a special case of Equation (6.33).

$$\Omega_{\rm g} = \frac{\epsilon_{\rm g}}{(\epsilon_{\rm g})_{\rm c}} = -\frac{1}{\pi} \frac{\beta}{\rm P} \int_0^\infty \frac{{\rm d}q}{[{\rm I}_0(q)]^2 + [(\beta/\rm P){\rm I}_1(q)]^2} . \tag{6.86}$$

A curve of  $\Omega_{\rm w}$  against  $\beta/P$  is shown in Figure 6.28; this only vanishes for an openjet tunnel  $(\beta/P=0)$ , and no perforated wall geometry will remove the wake-blockage effect. For a tunnel designed to have zero solid blockage  $\Omega_{\rm w}=-0.43$ . As in the two-dimensional case below Equation (6.83), the limit  $\Omega_{\rm w}=-1$  as  $\beta/P\to\infty$  cannot be identified with the definition  $\Omega_{\rm w}=1$  for a closed tunnel. An expression for the variation of the wake blockage along the axis of the tunnel for different values of  $\beta/P$  is given in Reference 6.45. Az in Equation (6.34) the gradient at the model position is

$$\frac{\partial \epsilon_{w}}{\partial x} = -\frac{1}{\pi \beta_{R}} (\epsilon_{w})_{c} \int_{0}^{\infty} I_{c} dq = \frac{(\epsilon_{w})_{c} \Omega_{s}}{0.40 \pi \beta_{R}}$$
(6.87)

0 38

?

RAMARIAN CONTRACTOR CONTRACT

where  $I_{g}$ , defined in Equation (6.84), is identical to  $I_{C}$  in Equation (6.28) with F = 0. It follows that the wake-blockage gradient vanishes with the solid blockage.

# 6.9.2 Rectangular Tannel

Blockage effects do not seem to have been considered for a rectangular tunnel having either two opposite, or all four, walls perforated. A working section with four perforated walls may be regarded as having similar interference characteristics to a circular tunnel of the same cross-sectional area, provided the height/breadth ratio is near unity. The results of Section 6.9.1, or Figure 6.28, may then be used for a small model mounted on the axis of the tunnel.

### 6.10 LIFT INTERPEBENCE IN TUNNELS WITH PERFORATED WALLS

### 6.10.1 Two-Dimensional Tunnel

The lifting aerofoil may be represented by a two-dimensional vortex having the same circulation. The subsequent analysis is then very similar to that for the solidblockage interference, and is set out in detail by Wright in Reference 6.26. The boundary-induced upwash at the model position, is

$$\frac{\mathbf{w}_1}{\mathbf{U}} = -\frac{cC_L}{2\pi\hbar} \cot^{-1}\left(\frac{\beta}{\mathbf{P}}\right) . \qquad (6.86)$$

Since the general expression for  $w_i/U$  becomes zero far upstream from the model. Equation (6.86) represents the complete interference upwash acting on a small aerofoil. For the closed tunnel (F = 0)  $w_i$  is zero, but for any other value of P some induced upwash must be present; in the limiting case  $\beta/P = 0$ , the value  $w_i/U = -\frac{1}{4}(c/h)C_L$  is consistent with an open tunnel. Equation (6.88) may be recast in terms of the interference parameter

$$\delta_0 = \frac{hw_1}{UcC_r} = -\frac{1}{2\pi} \cot^{-1}\left(\frac{\beta}{P}\right)$$
 (6.89)

which is plotted in Figure 6.29; at zero solid-blockage conditions, the correction is 0.42 of that appropriate to the open tunnel.

Wright has also considered the stressline curvature at the model position and shows that this may be expressed as

$$\frac{c}{\upsilon} \frac{\partial \mathbf{w}_{i}}{\partial \mathbf{x}} = -\frac{\pi c^{2} C_{i}}{12\beta \hbar^{2}} \left[ 1 - 6 \left(\frac{\zeta}{\pi}\right) + 6 \left(\frac{\zeta}{\pi}\right)^{2} \right] \qquad (6.90)$$

$$\zeta = \tan^{-1}(\beta/P) = \frac{i}{2}\pi - \mu \quad .$$

shere

The quantity in square brackets is identical in form to that arising in the expression for the solid blockage of a two-dimensional model in Equation (6.81), but now contains  $(\frac{1}{2}\pi - \mu)$  in place of  $\mu$ . It follows that the streamline curvature will be zero for a value  $\beta/P = 0.78$ , the reciprocal of that which eliminates the solid blockage. An

and a star with a set of the set

equation similar to (6.90) was obtained by Brescia<sup>6.44</sup>, using a different analytical approach. The interference parameter

$$\delta_1 = \frac{\beta h^2}{\text{UeC}_L} \frac{\partial w_1}{\partial x} = -\frac{\pi}{12} \left[ 1 - 5 \left(\frac{\zeta}{\pi}\right) + 6 \left(\frac{\zeta}{\pi}\right)^2 \right]$$
(6.91)

is also plotted as a function of  $\beta/P$  in Figure 6.29; the streamline curvature in the two-dimensional performted tunnel, designed to have zero solid blockage, is as much as 0.46 of the value appropriate to a closed tunnel. The corrections to incidence, lift and guarter-chord pitching moment are given by Equations (3.53) and (6.55).

$$\Delta c_{\mathbf{g}} = \frac{cC_{\mathbf{g}}}{b} \delta_{0} + \frac{c^{2}}{\beta h^{2}} \delta_{1} \left(\frac{1}{4}C_{\mathbf{g}} + C_{\mathbf{g}}\right)$$

$$\Delta C_{\mathbf{g}} = -\frac{\pi}{2} \left(\frac{c}{\beta h}\right)^{2} \delta_{3}C_{\mathbf{g}} \qquad (6.92)$$

$$\Delta C_{\mathbf{g}} = +\frac{\pi}{8} \left(\frac{c}{\beta h}\right)^{2} \hat{e}_{3}C_{\mathbf{g}}$$

where the quantities  $\delta_0$  and  $\delta_1$  for a given porosity parameter may be taken from Equations (6.89) and (6.91) or from Figure 6.29.

### 6.10.2 Circular Tunnel

The corrections for a small wing in a circular perforated tunnel may be considered as special cases of those derived in Reference 6.1 for the general ventilated wall having both porosity and longitudinal slots. The original solution by Goodman<sup>6.36</sup> considered only the perforated tunnel. The induced incidence at the model position may be expressed as

$$\frac{\mathbf{w}_{1}}{\mathbf{U}} = \frac{SC_{L}}{8\pi R^{2}} \left[ 1 - \frac{\beta}{\pi P} \int_{0}^{\infty} \mathbf{I}_{FC} \, \mathrm{d}\mathbf{q} \right] . \qquad (6.93)$$

where  $I_{F0}$  is the integrand of Equation (6.58) with the substitution F = 0. As in Equations (6.59) and (6.60), the interference parameter  $\delta_0$  becomes

$$\delta_0 = \frac{\pi R^2}{SC_L} \frac{\mathbf{w}_i}{U} = \frac{1}{g} \left[ 1 - \frac{\beta}{\pi P} \int_0^\infty \mathbf{I}_{P0} \, \mathrm{d}\mathbf{q} \right] \,. \tag{6.94}$$

The variation of  $\delta_0$  with  $\beta/P$  is shown in Figure 6.30. Zero lift interference occurs when  $\beta/P = 1.13$ , compared with  $\beta/P = 1.22$  required to remove the solidblockage velocity increment. Thus, at zero solid blockage,  $\delta_0$  will be only 0.04 of its value in a closed circular tunnel. The marked contrast between this case (F = 0) and the ideal slotted tunnel ( $\beta/P = 3$ ) is illustrated in Figure 6.16.

The flow curvature at the origin can be obtained directly from Equation (47) of Reference 6.1, which gives

the set of the set

$$\delta_{1} = \frac{2\pi/3R^{3}}{USC_{L}} \frac{\partial w_{1}}{\partial x} = -\frac{1}{4\pi} \int_{0}^{\omega} I_{H} dq , \qquad (6.95)$$

where

$$I_{H} = \frac{\{q^{2}K_{1}(q)I_{1}(q) + (\beta/P)^{2}[K_{1}(q) + qK_{0}(q)][I_{1}(q) - qI_{0}(q)]\}q^{2}}{[qI_{1}(q)]^{2} + (\beta/P)^{2}[I_{1}(q) - qI_{0}(q)]^{2}}$$

At zero solid-blockage conditions, the streamline curvature is large, 0.59 of its value in a closed tunnel<sup>5,28</sup>. The corrections to incidence and pitching moment follow from Equations (5.83) and (6.65) and may be evaluated with the aid of Figure 6.30.

Geodman<sup>5,34</sup> points out that for the perforated tunnel a solution to the upwash at the model position may not be deduced by considering the flow in a plane for downstream. In the closed or ideal slotted tunnel, for example, the upwash at this position is twice that at the model, and this analogy may be used to simplify the analysis. The besogeneous boundary condition for the perforated circular tunnel is

$$\frac{\partial \Phi}{\partial x} \div \frac{1}{P} \frac{\partial \Phi}{\partial r} = 0 ; \qquad (6.96)$$

since  $\partial \Phi / \partial x = 0$  in the distant wake, the boundary condition at infinity downstream reduces to

$$\frac{\partial \Phi}{\partial r} = 0 , \qquad (6.97)$$

the boundary condition for a closed tunnel. It is necessary therefore to consider the three-dimensional flow at the model position in solving the interference problem.

Goethert<sup>6,32</sup> suggests a physical reason for this behaviour of the perforated wall. The wall, acting like a lattice of elementary wings, generates free vortices at positions away from the axis of the tunnel. The distant-wake analogy holds true for each elementary vortex; but, since these are not all generated in the transverse plane of the model, x = 0, the analogy does not hold for the complete vortex system.

### 5.10.3 Rectangular Tamasl

The interference experienced by a small lifting wing in a rectangular tunnel seess only to have been partially solved, the sajor contribution coming from an early paper by Goodman (Ref. 6. 39; 1951). He considers the case of a wing mounted between two infinite performed walls, either vertical or borizontal. The wing itself is replaced in the analysis by an infinitesimal horse-shoe vortex. For vertical walls, the interference upwash at the wing is given by

$$\delta_{q} = \frac{c}{SC_{L}} \frac{w_{1}}{v}$$

$$= \frac{\pi h}{24b} \left[ 1 - \frac{12}{\pi^{2}} \int_{0}^{\infty} \frac{i\{(\beta/P)^{2} \sinh^{3}q + \cosh^{2}q\}^{\frac{1}{2}} - (\beta/P) \sinh q\} q \, dq}{\cosh^{2}q \sin q} \right] , \qquad (6.98)$$

and this may be expected to hold if h/b is large enough. Similarly, if h/b is small enough, Goodman's result for horizontal perforated walls may be used,

$$\delta_{0} = \frac{\pi b}{48h} \left[ 1 - \frac{24}{\pi^{2}} \int_{0}^{\infty} \frac{q \, dq}{\{(\beta/P)^{2} \cosh^{2}q + \sinh^{2}q\}^{\frac{1}{2}} \cosh q} \right] . \quad (6.99)$$

Values of  $(b/h)\delta_0$  from Equation (6.98) and of  $(h/b)\delta_0$  from Equation (6.99) are plotted against porosity parameter in Figure 6.31; zero lift interference occurs for  $\beta/P = 0.81$  and 1.50 respectively. The equations become identical to Equations (6.75) and (5.76) in the limiting cases of closed  $(\beta/P \to \infty)$  and open jet  $(\beta/P = 0)$  tunnels. Some idea of the range of applicability may be obtained from Figure 6.32 where these limiting results are compared with exact values calculated from the usual image methods (Chapter III). For the closed tunnel, it would seem that the horizontal walls have negligible influence when h/b > 1.25, and the vertical walls when h/b < 0.4. Similarly, for the open jet, the discrepancies can be ignored unless 0.8 < h/b < 2.5.

Unfortunately, many existing tunnels are nearly square, and it would seem best in these cases to estimate  $\delta_0$  from the curve of Figure 6.30 for a circular tunnel. This is reproduced in Figure 6.31 and suggests that the tunnel with vertical walls only is a good approximation for  $\beta/P > 2.0$ , and the tunnel with roof and floor only is useful for  $\beta/P < 0.5$ . For the intermediate range of  $\beta/P$ , curves of  $\delta_0$  against h/b in Figure 6.33 have been estimated graphically by identifying the square and circular tunnel and using the asymptotic formulae (6.93) and (6.99) for large and small h/b.

Goodman does not consider the curvature induced by the perforated boundaries, but it may be obtained as a direct extension of his analysis.

### 6.11 APPLICATION OF CORRECTION FACTORS

The ways in which blockage and lift interference may be used to correct the measured quantities are discussed in Sections 5.8 and 3.2.4 respectively.

The blockege factor  $\epsilon_{\rm B} = \epsilon_{\rm s} + \epsilon_{\rm w}$  determines the increment to the measured Mach number by means of the equation

$$\Delta M = (1 + 0.2M^2) M \epsilon_{\rm R} ; \qquad (6.100)$$

in addition, there are corrections to other stream quantities, such as density and Reynolds number. Alternatively, corrections may be applied to the measured force and pressure coefficients. If corrections, such as Equation (6.100), are applied to stream quantities, then there remains a correction to drag coefficient due to longitudinal pressure gradient.

$$\Delta C_{\rm p} = -\frac{2V_{\rm p}}{S} \frac{\partial \epsilon_{\rm B}}{\partial x}$$
$$= -\frac{2V_{\rm p}}{S} \frac{\partial \epsilon_{\rm s}}{\partial x} - C_{\rm p} \epsilon_{\rm s} , \qquad (6.101)$$

389

いたいたい たい おいたいないない 日本などの 日本 とうとう

and the second show a show a show a second show a second

Julian 1

A State and a state of the stat

where the equivalent volume of the model  $V_0$  may be estimated from Equation (5.89) and Figure 5.2. In two-dimensional flow  $(\Delta C_0)_{sg}$ , the first term on the right hand side of Equation (6.101), is given by Equation (6.19).

The interference upwach on a lifting sodel

$$\frac{\mathbf{x}_{1}}{\mathbf{v}} = \frac{SC_{L}}{C} \left( \sum_{0}^{2} \div \frac{\mathbf{x}}{\beta_{\mathrm{fl}}} \delta_{\mathrm{i}} \right)$$
 (6.102)

leads to incremental corrections to incidence, drag, lift and pitching moments. For a three-dimensional model in a rectangular tunnel

$$\Delta \alpha = \frac{SC_{L}}{C} \left( \delta_{0} + \frac{\overline{c}}{2\beta_{h}} \delta_{1} \right)^{2}$$

$$\Delta C_{D} = \frac{SC_{L}^{2}}{C} \delta_{0}$$

$$\Delta C_{L} = 0$$

$$\Delta C_{R} = \frac{\partial C_{L}}{\partial \alpha} \frac{SC_{L}}{C} \frac{\lambda x_{1}}{2\beta_{h}} \delta_{1}$$
(6.103)

where the planform parameter  $\lambda x_1/\bar{c}$  is discussed and tabulated in Chapter III. Corresponding formulae for a circular tunnel are given in Equations (6.63) and (6.65). In two-dimensional flow there is no correction to drag; the corrections to incidence, lift and pitching moment are given by Equations (6.92). In correcting the pitching moment for a complete aircraft model, care must be taken to evaluate the change in interference upwash botween the wing and the tail surface; the linear dependence of  $w_1/\bar{v}$  on x. assumed in Equation (6.102), may not be accurate enough.

The evaluation of interference corrections requires the knowledge of  $\epsilon_s$ ,  $\epsilon_w$ ,  $\partial \epsilon_g/\partial x$ ,  $\delta_0$  and  $\delta_1$ . Table 6.I is a guide to the available information for tunnels with slotted or perforated walls.

REFERENCES

| 6.1             | Baldwin, B.S.<br>et al.           | Wall Interference in Wind Tunnels with Slotted and Porous<br>Boundaries at Subsonic Speeds. NACA Tech. Note 3176,<br>1954.  |
|-----------------|-----------------------------------|---|
| 6.2             | Theodorsen, T.                    | The Theory of Wind-Thurnel Wall Interference. NACA Report<br>410, 1931.   |
| 6.3             | Rosenhead, L.                     | Interference Due to Walls of a Wind-Tunnel. Proc. Roy.<br>Soc. Series A, Vol.142, 1933, pp.308-320.   |
| 6.4             | Tani, I.<br>Taima, M.             | Two Notes on the Boundary Influence of Wind Tunnels of<br>Circular Cross Section. Aero. Res. Inst. (Tokyo) Report<br>121, 1935.   |
| 6.5             | de Haller, P                      | L'Influence des Limites de la Veine Fluide sur les Carac-<br>téristiques Aérodynamiques d'une Surface Portante.<br>Communication de l'Institut d'Aérodynamique de l'Ecole<br>Polytechnique Fédérale. (ETH. Zürich), 1934. |
| 6.6             | Toussaint, A.                     | Experimental Methods - Wind Tunnels. Influence of the<br>Dimensions of the Air Stream. "Aerodynamic Theory" (ed.<br>W.F.Durand), Vol. III, Div. I, Chap. III, Julius Springer,<br>Berlin, 1935, pp. 290-299.              |
| 6.7             | Wieselsberger, C.                 | Über den Einfluss der Windkanalbegrenzung auf den Wider-<br>stand insbesondere im Bereiche der kompressiblen Strömung.<br>Luftfahrtforschung, Vol. 19, 1942, pp. 124-128.   |
| <del>6</del> .8 | Kendo, K.                         | The Mall Interference of Wind Tunnels with Boundaries of<br>Circular Arcs. Aero. Res. Inst. (Tokyo) Report 126, 1935.   |
| 6.9             | Kondo, K.                         | Boundary interference of Partially Closed Wind Tunnels.<br>Aero. Res. Inst. (Tokyo) Report 137, 1936.   |
| 6.10            | Theodorsen, T.<br>Silverstein, A. | Experimental Verification of the Theory of Wind-Tunnel<br>Boundary Interference. NACA Report 478, 1934.   |
| 6. 11           | Van Schliestett, G.               | Experimental Verification of Theodorsen's Theoretical<br>Jet-Boundary Correction Factors. NACA Tech. Note 506,<br>1934.   |
| 6. 12           | Riegole, F.                       | Korrekturfaktoren für Windkanale elliptischen Querschnitts<br>mit tsilweise offener und teilweise geschlossener Meß-<br>strecke. Luftfahrtforschung, Vol. 16, 1939, pp. 25-30.<br>(Translated as NACA Tech. Memo. 1310.)  |
| 6. 13           | Maedor, P.F.<br>Wood, A.D.        | Trensonic Wind Tunnel Test Sections. Z. ang. Math. Phys.<br>Vol.7, 1956, pp.177-212.  |

391

日本の「日本の日本のない」

Í

\*\*\* \*\* \*\*\*

そうちしかういいや テレッス・ シリン・ション・シー

-

į.

22

and the second s

「日本アナナン」

ł

Š.

2

| d. 14 Pistolesi, E.             | <ul> <li>(i) On the Interference of a Wind Tunnel with a Mixed<br/>Boundary. Pont. Accad. Sci. Vol. 4, No. 9, 1940. (Unpub-<br/>lished translation by Cornell Aero. Lab., 1949.)</li> <li>(ii) On the Interference of a Wind Tunnel with Mixed<br/>Contour. Pont. Accad. Sci. Vol. 7, No. 5, 1943. (Unpub-<br/>lished translation by NACA, 1951.)</li> </ul> |
|---------------------------------|--|
| 6.15 Matthews, C.W.             | Theoretical Study of the Tunnel-Boundary Lift Interference<br>Due to Slotted Walls in the Presence of the Trailing-<br>Vortex System of a Lifting Midel. NACA Report 1221, 1955.   |
| 6.16 Wright, R.H.<br>Ward, Y.G. | NACA Transonic Wind-Tunnel Test Sections. NACA Report<br>1231, 1955. (First issued as R.N.L8J06, 1948.)  |
| 6.17 Davis, D.D.<br>Moore, D.   | Analytical Study of Blockage- and Lift-Interference<br>Corrections for Slotted Junnels Obtained by the Substi-<br>tution of an Equivalent Homogeneous Boundary for the<br>Discrete Slots. NACA RM 153E07b (NACA/TIB/3792), 1953.   |
| 6.18 Goethert, B.H.             | Properties of Test Section Walls with Longitudinal Slots<br>in Curved Flow for Subsonic and Supersonic Velocities<br>(Theoretical Investigations). Arnold Eng. Dev. Center,<br>ASDC-TN-55-56, 1957.  |
| 6.19 Woods, L.C.                | The Theory of Sabsonic Plane Flow. Cambridge University<br>Press, 1961.  |
| 5.20 Guderley, G.               | Simplifications of the Boundary Conditions at a Wind-<br>Tunnel Wall with Longitudinal Slots. Wright Air Dev.<br>Center, Tech. Report 53-150, 1953.  |
| 6.21 Goodman, T.R.              | The Porous Wall Wind Tunnel, Pt.II. Interference Effect<br>on a Cylindrical Body in a Two-Dimensional Tunnel at<br>Subsonic Speeds. Cornell Aero. Lab. Report AD-594-A-3,<br>1950.   |
| 6.32 Stokes, G.M.<br>et al.     | An Experimental Study of Porosity Characteristics of<br>Perforated Materials in Normal and Parallel Flow. NACA<br>Tech. Note 3085, 1954.   |
| 6.23 Reynolds, W.G.<br>et al.   | Boundary Interference Studies in the Slotted Test Section<br>of the Southern California Cooperative Wind Tunnel for<br>the Mach Number Range 0.4 to 1.3. CWE Report P-19, 1959.  |
| 6.24 Pearcey, H.H.<br>et al.    | Some Effects of Wind Turnel Interference Observed in<br>Tests on Two-Dimensional Aerofoils at High Subsonic and<br>Transonic Speeds. AGARD Report 296, 1959.   |
| 6.25 Maeder, P.F.               | Theoretical Investigation of Subsonic Wall Interference<br>in Rectangular Slotted Test Sections. Brown Univ. Div.<br>of Eng. Tech. Report WT-11, 1953.   |

200

1.1

T Constant

يغه Ŷ.

?

1

I will a give I all the construction and second and second and

| 6, 26 | Borden,              | Α.             | Wall Corrections for Flow About Two- and Three-Dimensional<br>Symmetrical Bodies in Rectangular Channels of Infinite and<br>Finite Length. David W. Taylor Model Basin, Report 804,<br>1954.   |
|-------|----------------------|----------------|--|
| 6.27  | Watson,              | G. N.          | A Treatise on the Theory of Bessel Functions. 2nd Edition.<br>Cambridge University Press. 1944.  |
| 6.28  | Wright,              | R. H.          | The Effectiveness of the Transonic Wind Tunnel as a Device<br>for Minimizing Tunnel-Boundary Interference for Model<br>Tests at Transonic Speeds. NASA paper presented at AGARD<br>meeting on Interference Effects. AGARD Proport 294, 1959. |
| 6. 29 | Maeder,<br>et al.    | P. F.          | Experimental Investigation of Subsonic Wall Interference<br>in Rectangular Slotted Test Sections. Brown Univ. Div.<br>of Eng. Tech. Report WT-16, 1955.  |
| 6.30  | Katzolf,<br>Barger   | S.<br>R.L.     | Boundary-Induced Downwash Due to Lift in a Two-Dimensional<br>Slotted Wind Tunnel. NASA Tech. Report R-25, 1959.   |
| 6. 31 | Rao, D.M<br>Ramaswam | I.<br>эу, ¥.А. | Measurements of Two-Dimensional Zero-Lift Interference of<br>Ventilated Boundaries at Speeds up to Sonic. 2. ang.<br>Math. Phys. Vol.11, 1960, pp.327-334.   |
| 8.32  | Goethert             | t, B.H.        | Transonic Wind Tunnel Testing. AGARDograph 49, 1961.   |
| 6.33  | Squire,              | L. C.          | A Comparison of Test Results from Three Transonic Tunnels<br>for Two Sets of Geometrically Similar Models. RAE Tech.<br>Note Aero. 2679, ARC Report 22,059, 1960.  |
| 6.34  | Acum, W.             | E. A.          | Note on the Evaluation of Solid-Blockage Corrections for<br>Rectangular Wind Tunnels with Slotted Walls. ARC R & M<br>3297, 1961.  |
| 6.35  | Kassner,             | R. R.          | Subsonic Flow over a Body between Porou: Walls. Wright<br>Air Dev. Center, Tech. Report 52-9, 1952.  |
| 6.36  | Goodaan,             | T. R.          | The Porous Wall Wind Tunnel, Pt.IV. Subsonic Interference<br>Problems in a Circular Tunnel. Cornell Aero. Lab. Report<br>AD-706-A-2, 1951.   |
| 6.37  | Tiramale             | esa, D.        | Lift Interference for Low Aspect Ratio Wings in Transonic<br>Wind Tunnels. J. mero. Soc. India, Vol.11, 1959, pp.79-82.  |
| 6.38  | Chew, W.             | L.             | Cross-Flow Calibration at Transcnic Speeds of Fourteen<br>Perforated Plates with Round Holes and Airflow Parallel<br>to the Piates. Arnold Eng. 7 Jenter AZDC-TR-54-65,<br>1955.   |
| 6.39  | Goodman.             | . <b>T. R.</b> | The Porous Wall Wind Tunnel, Pt.V. Subsonic Interference<br>on a Lifting Wing between Two Infinite Walls. Cornell<br>Aero. Lab. Report AD-706-A-3, 1951.   |



A state of the sta

ĝ

;

Party Construction of the

į

| J. 54 Hill, J. A.F.                | An Anal, tical Study of Flow through a Perforated Wall.<br>United Aircr. Corp. Report R-95630-9, 1953.  |
|------------------------------------|---|
| 6.35 Tirumalesa, D.                | Wall Interference in Transonic Wind Tunnels. J.Aero.<br>Soc. India, Vol.11, 1959, pp.53-67.   |
| 6.56 Maeder, P.F.<br>Thomsen, H.U. | On the Boundary Layer of Perforated Walls. Z. ang. Math.<br>Phys. Vol.95, 1958, pp.438-453.   |
| 6.57 ¥004, W.W.                    | Tunnel Interference from Slotted Weils. Quart. J. Mech.<br>appl. Math. Vol. 17, 1964, pp. 125-140.  |
| 6.58 Holder, D.R.                  | Upwash Interference in a Rectangular Wind Tunnel with<br>Closed Side Walls and Porous Slotted Floor and Roof.<br>ARC R & M 3322, 1962.                            |
| 6.59 Holder, D.R.                  | Upwash Interference on Wings of Finite Spon in a Rect-<br>angular Wind Tunnel with Closed Side Walls and Porous-<br>Slotted Floor and Roof. ARC R & M 3395, 1963. |
| 5.66 Rushton, K.R.                 | Studies of Slotted-Wall Interference Using an Electrical<br>Analogue, Part II. Particular Examples of Slotted-Wall<br>Tunnel Interference. ARC R & M 3452, 1965.  |
| 6.61 Eushton, K.R.                 | Private Commication, 1965.  |

and a supported by the second second

F

ł

# ADDITIONAL SEFERENCES FOR CHAPTER VI

| Brescie, R.  | Studio dell'Interferenza delle Galleric Aerodinaziche<br>con Poreti a Fessure in Correnti Coxpressibili. Atti  |
|--------------|--|
|              | della Accadesia delle Scienze di Torino, Vol. 53, 1953-54.   |
| Gale, A.E.   | Wall Corrections for Transonic Wind Yunnels, Part I:<br>A Lifting Model in a Circular Slotted Tennel. MIT Aero.<br>Eng. Dept. TACP Report 3, 1954.         |
| Guderley, G. | Theoretical Remarks on Wind Tunnels with Slotted Walls,<br>Wright Air Dev. Center Tech. Report 52-8, 1952.   |
| Guderley, C. | Wall Corrections for a Sind Tunnel with Longitudinal<br>Slots at Subsonic Velocities. Wright Air Dev. Center<br>Tech. Report 54-22, 1954.                  |
| Maeder, P.F. | Investigation of Tunnel Boundary Interference on Tro-<br>Disensional Airfoils Near the Speed of Suand. Brown<br>Univ. Div. of For. Tech. Report FI-6, 1981 |
A STATE OF THE PARTY OF THE PAR

10

An Experimental Investigation of the Zero-Lift Pressure Helson, W.J. Bloetscher, P. Distribution over a Wedge Airfoil in Closed, Slotted, and Open-Throat Tunnels at Transonic Mach Numbers. NACA RM L52C18 (NACA/TIB/3241), 1952. Some Aspects of Interference in Ventilated Wall Transonic Tirusaless, D. Wind Tunnels. NAL (Bangalore) Tech. Note TN-4E-12-61, 1961. Vandrey, F. Experiments on a Slotted Wall Working Section in a Wind Wieghardt, K. Tunnel. ABC CP 206, 1952. Thitcoub, C.F. An Experimental Investigation of Boundary Interference on Osborne, R.S. Force and Moment Characteristics of Lifting Models in the Langley 16- and 8-foot Transonic Tunnels. NACA PME L52L29 (NACA/TIB/3642), 1953. On the Theory of Tro-Dimensional Wind Tunnels with Porcus Toods, L.C. Walls. Proc. Roy. Soc., Series A, Vol. 223, 1955, pp. 74-90. On the Lifting Aerofoil in a Wind Tunnel with Porous Walls. Woods, L.C. Proc. Roy. Soc., Series A, Vol. 242, 1957, pp. 341-354. ¥right, R.H. Wind-Tunnel Lift Interference on Secriback Wings in Rect-Earger, R.L. angular Test Sections with Slotted Top and Bottom Walls. NASA Tech. Report R-241, 1966.

#### ا الله الي الرجع وحد الذي

-

17 . Amer 5-7 635 134 .

397

SULPARA STA

when it when

The second second second second

2

ī

Ì.

- 33

### TABLE 6.I

### Summary of Results

| Tunnei<br>section   | ¥alls                     | Model* | Blockage          |                | Lift              |                      |
|---------------------|---------------------------|--------|-------------------|----------------|-------------------|----------------------|
|                     |                           |        | Figs.             | Equs.          | Figs.             | Eqns.                |
| 170-<br>dizensional | Slotted                   | Small  | 6. 2<br>6. 3      | 6.15<br>6.23   | 5. 14<br>6. 15    | 6. 48                |
| Tro-<br>dimensional | Slotted                   | Large  | 6.4 to<br>6.7     | 6. 18<br>6. 20 | 6. 15             | 6.51<br>6.54         |
| Tro-<br>dizensionai | Perforated                | Szsll  | <b>6.</b> 26      | 6.81<br>6.83   | 6.29              | 6.89                 |
| Two-<br>dizensional | Perforated                | Large  | 6.27              | 6.84           | 6.29              | 6.91                 |
| Circular            | Slotted                   | Szall  | 6.8<br>6.9        | 6.28<br>6.33   | 6.15<br>6.17      | 6.58                 |
| Circular            | Slotted                   | Large  | ~                 | Ş. 34          | -                 | 6.61<br>5.63         |
| Circular            | Perforated                | Small  | <del>6</del> . 28 | 6.84<br>5.86   | \$, <b>3</b> 9    | 6. 94                |
| Circular            | Perforated                | Large  | -                 | 6.65<br>6.87   | -                 | 6.95                 |
| Rectangular         | Slotted roof<br>and floor | Stall  | 6. 10             | 6. 37          | 6.21 to<br>6.24   | 6.70<br>6,73         |
| Rectangular         | Slotted roof<br>and floor | Large  | 6.11              | -              | 6. 18 to<br>ಕ. 23 | 6.67<br>5.73<br>5.72 |
| Rectangular         | All slotted               | Seali  | 6. 12             | -              | 6.1?<br>6.25      | 6.74<br>6.75<br>6.76 |
| Rectangular         | Perforated                | Smal 3 | -                 | -              | 6.31<br>6.33      | 6.99<br>6.99         |
| General             | Slotted                   | Smil   | G. 33             | 5.46           | -                 | -                    |

• "Large" denotes that the effect of longitudinal pressure gradient, streamline curvature, length or span of model is considered.

and the second second

a hard a second s



Open area ratio is  $\frac{a}{d}$ 

\*\*\*\*\*\*\*\*\*\*\*

 $K = \frac{d}{\pi} \log_e \csc \frac{\pi d}{2d}$ 

(a) For a tunnel of height h and breadth b with solid side-walls

$$F = \frac{2K}{h} = \frac{2d}{\pi h} \log_e \operatorname{cosec} \frac{\pi a}{2d}$$
$$= \frac{2b}{\pi Nh} \log_e \operatorname{cosec} \frac{\pi a}{2d}$$

where N is the number of slots in the roof or floor

(b) For a circular tunnel of radius R

 $F = \frac{K}{R} = \frac{2}{N} \log_e \csc \frac{\pi d}{2d}$ 

(c) For a rectangular tunnel with evenly spaced slots on all four walls

$$F = \frac{\sqrt{\pi}K}{\sqrt{bh}} = \frac{d}{\sqrt{\pi bh}} \log_e \csc \frac{\pi d}{2d}$$

Fig. 6.1 Notation and geometry for longitudinal slots



;

CONTRACT)

12.55

La Atternation and

**ANKAR** 

e y y

ALC: NO DO



1 (j. - 1)

and a state of the state of the



ł

401

のないないであるのというのでのなると

ويزارهن الملك الكنكوالالمان المعامد ومعامه

ملالت معارضه بال

くとないの



: S

THE A

and the second



۰.

ł

403

S. With the second

14.2

\*\*\*

-----

the state of the s

**د.** ۲

404

のないたいないのである

-----

· · · h

STREET, BAR

i api

4

ŗ



÷

Fig. 6.7 Longitudinal pressure gradient at the model associated with solid blockage

A STATE OF A



:

405

and and the second second of the second s

:

\$

ŧ

đ.



· NW LANGER RIVE

14 14 A

÷.





•

407

×.

A THE OTHER PARTY AND THE AREA PARTY AND THE PARTY AND THE

**ANY AND** 



Contraction for the second of the

•

ì

And the second second second

The second se



The state of the second s

409

「二日町」に

1





The gat the second







a sai

**N**.4







-



i kili Erev

**.** 

Fig. 6.18 Lift-interforence parameter for models in a square ideal slotted tunnel with solid side-walls

١

「「「「ないな

a releases

3

and a second a support of a second a second se

4

And Angenters and a survey of

A CONTRACT



Fig. 6.19 Lift-interference rarameter for und 1. (n like) slotted tunnel with solid side wrije

. • •



Fig. 6.20 Streamline-curvature correction in ideal slotted tunnel with solid side-walls

417

6.72

and betanical a statistic man ----

11. N. S.

• •

.

•

26 Bernaria



Fig. 6.21 Influence of viscosity on lift-interference parameter for models in square slotted tunnel with solid side-walls

ŝ

ł

418

•7



.



419

tunnel with solid side-walls

and we are

3

n. R

. ;**`** 



Re



÷

421

きょうしょう いいちょう

:





ć

The Constant of the first states of a substances

1

į

Fig. 6.25 Variation of blockage with wall porosity in two-dimensional perforated tunnel

423

「「「「「「「「「「「「」」」」」」

1

2

「ないないのいない」





1-25

00-1

0-75

0-50

0.25

o x/Bh

-0.25

-0.50

-0.75

00.1-

- 1.0 [-.25

-0.5



ŝ

----

Ϋŧ







425

i

ŧ

and the second second





Fig. 6.30 Lift interference for a small model in a perforated circular tunnel



Fig. 6.31 Lift-interference parameter for small wing in special types of perforated tunnel



1.2

a the second

:

i,



Bernary ton



Fig.6.33 Estimated lift-interference parameter for small wing in perforated rectangular tunnels of different porosity

and the state of the second with the second state of the second st



## t second

- (Ectional

Ę

CHAPTER VII

# BLUFF BODIES AND HIGH-LIFT SYSTEMS

Ъy

### E.C.Maskell

Aerodynamics Department, Royal Aircraft Establishment, Farnborough, Hampshire, England

....




CONTENTS

|              |  | rage |
|--------------|--|------|
| lis <b>t</b> | OF FIGURES                                     | 434  |
| NOTA         | TIGN   | 435  |
| 7.1          | INTRODUCTION                                   | 437  |
| 7.2          | BLOCKAGE EFFECTS ON BLUFF BODIES               | 438  |
| 7.3          | LIFTING WINGS WITH SEPARATED FLOW              | 441  |
|              | 7.3.1 Stalled Wings                            | 441  |
|              | 7.3.2 Wings with Hinged or Split Flaps         | 442  |
|              | 7.3,3 Slender Wings with Leading-Edge Vortices | 443  |
| 7.4          | V/STOL CONFIGURATIONS                          | 444  |
|              | 7.4.1 Lifting Rotors                           | 444  |
|              | 7.4.2 Jet-Flaps in Closed Tunnels              | 446  |
|              | 7.4.3 Nore General Configurations              | 448  |
| REFE         | RENCES   | 449  |
| ADDI         | TIONAL REFERENCES                              | 450  |
| FICI         | 276  | 45.0 |

e

and a statistic the state

# LIST OF FIGURES

Page

| Fig.7.1         | Model of flow past a bluff body  | 452         |
|-----------------|--|-------------|
| Fig.7.2         | Invariance of $C_D/k^2$ for non-lifting square plates  | 452         |
| Fig.7.3         | Blockage effect on drag and base pressure for non-lifting square plates  | 453         |
| Pig. 7. 4       | Blockage effect on drag and base pressure for non-lifting<br>two-dimensional plates                              | 454         |
| Fig.7.5         | Variation of blockage factor with fineness ratio   | 455         |
| Fig.7.6         | Drag analysis for a lifting wing   | 456         |
| Fig.7.7         | Blockage corrections applied to lift and drag for two sizes of delta-wing-body combination                       | 457         |
| <b>Fig.7.</b> 8 | Wake of small rotor represented by a semi-infinite row of vertically directed doublets                           | 458         |
| Pig. 7.9        | Wake of a small rotor at non-zero angle of attack and its<br>image below the floor of a closed tunnel            | 459         |
| Fig. 7. 10      | Interference parameter against weke skew-angle for small<br>lifting rotors in closed rectangular tunnels         | 460         |
| Fig.7.11        | Interference parameter against breadth to height ratio of rectangular tunnels for rotors of different skew angle | 461.        |
| Fig. 7. 12      | Upwash interference against skew angle for a wake of horizontal<br>doublets in closed rectangular tunnels        | <b>4</b> 62 |
| Fig.7.13        | Effect of angle of attack on the interference parameter for small lifting rotors in a closed square tunnel       | 463         |
| Fig. 7. 14      | Assumed relationship between total downwash and interference upwash for a jet-flap wing                          | 464         |
| Fig. 7. 15      | Typical constraint corrections for a jet-flap wing with $50^{\circ}$ jet deflection                              | 465         |

434

۶.

# NOTATION

£

| A               | aspect ratio of wing                               |
|-----------------|--|
| b               | breadth of tunnel                                  |
| В               | cross-sectional area of wake at plane 2 (Pig. 7.1) |
| c               | chord of wing                                      |
| c               | cross-sectional area of turnel                     |
| C <sub>D</sub>  | drag coefficient = $D/qS$                          |
| C <sub>D0</sub> | profile drag coefficient                           |
| C <sub>Ds</sub> | separated-flow component of $C_{D}$ (Fig.7.6)      |
| C <sub>Dv</sub> | vortex-induced drag coefficient                    |
| с <sup>з</sup>  | jet momentum coefficient = momentum/qS             |
| C <sub>L</sub>  | lift coefficient = lift/qS                         |
| С <sub>р</sub>  | pressure coefficient                               |
| с <sub>рь</sub> | base-pressure coefficient                          |
| с <sub>т</sub>  | thrust coefficient = thrust/qS                     |
| D               | drag   |
| h               | height of tunnel                                   |
| k               | base-pressure parameter in Equation (7.1)          |
| l               | length of body                                     |
| l               | distance along wake of lifting rotor               |
| <b>R</b> =      | B/S  |
| ۶<br>b          | base pressure                                      |
| Q               | dynamic pressure of undisturbed stream             |
| Ŕ               | radius of vortex ring                              |
| S               | seni-span of wing                                  |
| S               | ares of planform of wing, reference area           |

435

÷

LINERCH NAME

1

A REAL PLAN AND A REAL PLAN AND A

12.21

\$ 19 K 3

- ----

| 436 |
|-----|
|-----|

| t | thickness | of | poda |
|---|-----------|----|------|
|---|-----------|----|------|

- U velocity of undisturbed stream
- w z-component of velocity
- w<sub>o</sub> upward velocity induced at centre of rotor
- w<sub>i</sub> interference upwash velocity
- W non-dimensional downwash in Equation (7.13)
- $\mathbf{W}_1 = \mathbf{W}$  for vertically directed doublets in Equation (7.14)
- $\mathbf{W}_{2} = \mathbf{W}$  for horizontally directed doublets in Equation (7.15)
- x streamwise distance
- y spanwise distance
- z upward distance
- a incidence of wing
- $\Gamma$  circulation
- $\delta_0$  upwash interference parameter in Equation (7.18)
- ô<sub>k</sub> upwash interference parameter for wake of horizontal doublets
- $\delta_{w}$  non-dimensional  $w_{i}$  in Equation (7.17)
- $\Delta$  prefix denoting increment due to wall interference
- $\epsilon$  blockage factor =  $\Delta U/U$
- $\zeta = 2z/h$
- $\eta = 2y/h$
- $\theta$  blockage factor for bluff-body flow in Equation (7.6)
- $\xi = 2x/h$
- $\rho = (\xi^2 + \eta^2 + \zeta^2)^{\frac{1}{2}}$
- r angle of deflection of jet flap
- x wake skew-angle of lifting rotor

Subscript

c denotes corrected value.

1.1

and the second sec

## BLUFF BODIES AND HIGH-LIFT SYSTEMS

#### E.C.Maskell

#### 7.1 INTRODUCTION

The extent to which it is possible to establish corrections for boundary constraint on a given flow depends largely on the extent to which a theory of the corresponding unlimited flow has been developed. Thus, for any unlimited flow that is well represented by classical aerofoil theory, it is usually possible to derive constraint corrections more or less rigorously, as the foregoing chapters have shown. But for other kinds of flow field, some of which have assumed great practical importance in recent years, the situation is much less satisfactory, because the mechanism of non-streamline flows is much less well understood. The difficulty in such cases is a fundamental one, that of establishing a methematical model of a flow that is sufficiently realistic and, at the same time, amenable to theoretical development; its resolution has been attempted in relation to only the few specific problems which form the basis of the present chapter.

The feature of an external flow that distinguishes one kind of flow field from another is the manner in which the wake, in particular the near wake, forms behind the body and interacts with the flow over the body itself. For most streamline flows this interaction is accounted for adequately, if the wake is represented simply as a plane sheet of streamwise trailing vortices. This is because the wake behind a streamline body is, by definition, thin, and because significant distortion from a plane sheet usually occurs too far downstream of the trailing edge to have much effect on the near field of the body. In the flows considered here at least one of these simplifying properties is absent. In consequence, the appropriate mathematical models are both more difficult to justify on physical grounds, and less tractable theoretically than the streamline flow model.

The flows in question fall into two distinct groups, the first of which is typified by the flow past a bluff body. The wake in this case is far from thin, and its interaction with the near field of the body is not well understood. However, constraint corrections in closed tunnels have been derived by Maskell (Ref. 7.1; 1963) by a method which relies heavily on limited experimental evidence. An interesting conclusion from this work is that distortion of the wake under constraint makes a significant contribution to the large blockage correction derived.

The second group of problems has arisen from increasing practical interest in recent years in V/STOL aircraft. The flows in this group are therefore of a kind in which lift is obtained, directly or indirectly, by the downward deflection of jets

• The slender wing is an important exception, in which the wake develops initially in the form of leading-edge vortices, often from the spex of the wing.

المحافظ والمحافظ والمحافظ والمحافظ والمحافظ المحافظ والمحافظ والمحاف

#### 437

ź

A HAR SHARE SHAR

or slipstreams. Here the paths of jets or slipstreams are important features of the flows, which might be greatly affected by boundary constraint. Realistic mathematical models can usually be constructed if the effects of mixing at the boundaries of the jets or slipstreams are ignored, so that, in principle, the appropriate interactions can be taken into account. In practice, however, the resulting mathematical models are intractable. Considerable simplification is therefore inevitable, and the problem is to establish what kinds of simplification are justifiable.

The most promising approaches to problems of this kind have been concerned with flows that can be regarded as essentially single-wake flows. Two examples are the flow past a helicopter, in which the dominant wake is evidently the slipstream of the rotor, and the flow past a jet-flap, in which the jet sheet simply replaces the usual trailing-vortex sheet. Heyson (Ref.7.2; 1960) has treated the helicopter problem. using a simple linear approximation to the jet path. The method is physically convincing at the limits of hovering flight, and high forward speed. But, in general, interaction between the slipstream and the external flow would be expected to lead to a curved slipstream. However, until the effect of this curvature has been investigated for the helicopter in an unlimited stream, its neglect in the constraint problem is plainly inevitable. The flow past a jet-flap wing is more closely related to streamline flow, and has been treated as a simple extension of classical aerofoil theory by Maskell and Spence (Ref. 7.3; 1959). The further extension to the problem of boundary constraint has been given by Maskell (Ref.7.4; 1959), who points out that distortion of the jet path by the constraint results in an effective change in the jet momentum coefficient. However, here again the approximation to the jet path employed in the theory is, to some extent, open to question, although there seems little doubt that a correction of the order of magnitude of that derived by Maskell is necessary.

More general V/STOL configurations employ multiple wakes, in that they usually consist of combinations of lifting wings with the deflected slipstreams of fans or jets in which the vortex wakes of the wings would be expected to follow markedly different paths from the slipstreams. The main problem, again, is to decide how necessary it is for these paths to be represented correctly. Heyson (Ref. 7. 5; 1962) has proposed an extension of his lifting-rotor theory to the general flow, and there is evidênce that the method has given plausible corrections in certain cases. However, the basic model is less convincing, on physical grounds, when applied to multiple-wake problems, which therefore seem to require much more attention. It is likely that the aerodynamics of such flows will often depend crucially on the mutual interference between the different wakes and slipstreams, and hence that it may be vital to determine, and interpret, the effect of boundary constraint on this interference. For purposes of this kind a more realistic mathematical model than Heyson's would be needed. But little further progress can be expected until the basic theory of such flows has advanced a great deal.

#### 7.2 BLOCKAGE EFFECTS ON BLUFF BODIES

Following a comment by Glauert on the nature of the blockage effect associated with a thick bluff-body wake, Fage and Johansen (Refs. 7.6 and 7.7; 1927-28) made what appears to have been the first serious attempt to establish the magnitude of the effect experimentally, for the particular case of a two-dimensional flat plate normal to a

wind stream. They tested several such plates, of different sizes, and so determined the drag coefficient corresponding to an unlimited stream by extrapolation of their measurements to zero size.

However, the attempt of Glauert to develop a correction formula in Equation (17.13) or (18.5) of Reference 7.8 was less successful - partly, perhaps, because he sought to make it too comprehensive. With greater understanding of the differences between the properties of the wakes of streamline and bluff bodies, it now seems unrealistic to suppose that the wall constraint on a bluff-body flow can be broken down into separate solid-blockage and wake-blockage effects. This sub-division is a physically realistic, and therefore useful, procedure when the influence of the wake on the form of the pressure field over the body can be regarded as a second-order effect, as it can be, for the most part, in streamline flow. But the pressure field over a bluff body depends crucially on the wake structure, and generally bears little relation to the inviscid, attached-flow, field of the body, from which the conventional solidblockage effect derives.

The bluff-body flow evidently needs to be represented, from the outset, by a mathematical model that is quite different from that appropriate to streamline flow. But since very little is known about the internal mechanics of a bluff-body wake, there is little prospect of constructing a flow model that bears more than a superficial resemblance to the observed flow. The best that can be done at the present time is to ensure that the external effects of the wake are adequately reproduced in the model, and to take such account as is necessary of the internal field through empirically determined auxiliary equations designed to match the behaviour of the model under constraint to observation.

Maskell (Ref.7.1; 1963). has pointed out that the observed properties of the forward part of a bluff-body wake (i.e. forward of any substantial pressure recovery) are well represented by a model incorporating a stream surface extending downstream from the edge of the body; the static pressure on the stream surface is supposed constant, and equal to the base pressure  $p_b$  as far as the plane 2 in Figure 7.1, where the crosssectional area of the wake is a maximum. The corresponding constant velocity he writes as kU. Then, if wall constraint is to be exactly equivalent to a simple increase in stream velocity, the form of the pressure distribution over the body must be invariant under constraint, and it follows that

$$\frac{C_{\rm D}}{k^2} = \frac{C_{\rm D}}{1 - C_{\rm nb}} = \text{constant}$$
(7.1)

independent of boundary constraint, where  $C_{pb}$  is the base-pressure coefficient,  $C_D$  is the drag coefficient D/qS, q the dynamic pressure of the undisturbed stream, and S a representative area of the body. It follows, also, that the velocity  $U_{z}$  of the unlimited stream which gives rise to a pressure distribution identical to that observed is such that  $k_c U_c = kU$ , and hence that

 $\frac{U_c^2}{U^2} = \frac{k^2}{k_c^2} = \frac{C_D}{C_{Dc}}.$  (7.2)

The relation (7.1) has been shown to hold experimentally (Fig.7.2) for a set of geometrically similar sharp-edged square plates, and is supposed to hold generally for incompressible flow in closed tunnels.

A further relation between the variables defining the flow model can de derived by considering the conservation of momentum within the control surface bounded by the walls of the wind tunnel, the surface of the body and the constant-pressure surface bounding the effective wake, and two planes normal to the undisturbed velocity vector - plane 1 lying upstream of the body, and plane 2 located where the cross-sectional dimensions of the wake are greatest. Neglecting certain supposedly small contributions, Haskell<sup>7.1</sup> has concluded that the momentum balance reduces approximately to

$$C_{\rm D} = m(k^2 - 1 - mS/C)$$
 (7.3)

where m = B/S, B is the cross-sectional area of the wake at the plane 2, and where  $(mS/C)^2$  is taken to be negligibly small.

In order to complete the set of equations necessary to define the blockage effect completely, one further relation is required, to account for the distortion of the wake under constraint. The auxiliary relation adopted by Maskell can be put in the form

$$\frac{E}{B_{c}} = 1 - \frac{C_{D} - C_{DC}}{(k^{2} - 1)(k_{c}^{2} - 1)} \frac{S}{C}$$
(7.4)

if terms of  $O(S/C)^2$  are ignored, and is well supported by limited experimental data obtained with a series of square flat plates.

Equations (7.1) to (7.4) yield the correction formula

$$\frac{k^2}{k_c^2} = 1 + \frac{C_D}{k_c^2 - 1} \frac{S}{C} + 0(S/C)^2$$
(7.5)

which may be written, alternatively, in the more conventional form

$$\frac{\Delta q}{q} = \theta \frac{c_0 s}{c}. \tag{7.6}$$

Here  $\Delta q = q_c - q$  is the effective increase in dynamic pressure of the undisturbed stream due to contraint,  $C_{\rm D}S/C$  is the usual wake-blockage parameter, and

$$\theta = \frac{1}{k_0^2 - 1}$$
(7.7)

A CONTRACTOR OF THE PARTY OF TH

is the so-called blockage factor for the bluff-body flow.

If both the drag and the average base-pressure coefficient  $C_{\rm pb} = 1 - k^2$  are measured,  $k_{\rm c}^2$ , and hence  $\theta$ , can be determined from Equation (7.5) - most conveniently, perhaps, by means of the iteration formula

$$(k_c^2)_n = k^2 \left\{ \frac{1}{1 + \frac{1}{(k_c^2)_{n-1} - 1} \frac{C_p S}{C}} \right\}^{-1}$$
 (7.8)

 A
 1
 2
 5
 10
 20
 ∞

 θ
 2.77
 2.70
 2.41
 2.13
 1.47
 0.96

where  $(k_c^2)_n$  is the nth approximation to  $k_c^2$ , and the initial value  $(k_c^2)_0 = k^2$ . In this way Fail, Lawford and Eyre (Ref. 7.9; 1957) have concluded that the blockage

factor for rectangular flat plates varies with aspect vatio A in the following manner:

It turns out that the blockage factor varies very little in the interval  $1 \le A \le 10$ , where it lies roughly in the range  $\theta = (5/2) \pm (1/4)$ . In this range, therefore, the constant value  $\theta = 5/2$  is unlikely to lead to appreciable error in practice.

The correction formulae from Equation (7.5) appropriate to drag and base pressure are given in both Figures 7.3 and 7.4. For square and two-dimensional flat plates respectively the theoretical straight lines against  $C_{\rm D}S/C$  are shown compared with experimental results. The blockage factor  $\bar{\theta}$  varies from about five times to twice the corresponding quantity for the wake blockage of streamline bodies (Chapter V).

More recent studies of blockage effects on bluff bodies are described in References 7.10 and 7.11. Lefebvre<sup>7.16</sup> gives results rather similar to those of Reference 7.1, and relates the geometric and aerodynamic blockage with particular reference to circular cones. Calvert<sup>7.11</sup> compares experiments on blunt-based bodies of revolution with the predictions of Reference 7.1 and concludes that the theory gives a good estimation of the blockage correction for fineness ratios l/t < 3, as shown in Figure 7.5. In the range 3 < l/t < 5.5 Maskell's theory still leads to reasonable results, but the methods of Chepter V are more appropriate to the more slender non-lifting bodies.

#### 7.3 LIFTING WINGS WITH SEPARATED FLOW

#### 7.3.1 Stalled Wings

The breakdown of streamline flow past a wing of finite spac gives rise to regions of separated flow that been some rescabiance to bluff-body wakes, and would therefore be expected to lead to blockage effects similar in kind to those treated in Section 7.2. In fact, experimental measurements by Kirby and Spence (Ref. 7.12; 1955) in the wekes of models of particular delta-wing and swept-wing aircraft suggest that, for wings of models of small aspect ratio, the localised regions of separated flow that develop as such wings begin to stall resemble axisymmetric bluff-body wakes surprisingly closely. Furthermore, Fail et al.<sup>7,9</sup> have observed a strong tendency to axial symmetry in the wakes of non-axisymmetric bluff bodies. And Mashell has been led to suppose the tendency to axial symmetry in the separated flow region to be universal, within the practical range of wing shapes, and to give rise to the blockage factor  $\hat{\theta} = 5/2$ appropriate to most bluff-body flows.

However, the formulae derived in Section 7.2 cannot be applied directly to the partially stalled lifting wing, because its total irag is only partially related to .

441

1. X. X.

the separated-flow regions. There are two other contributions to the total drag; the vortex-induced drag  $B_{\rm v}$ , and the momentum defect associated with the wake within the streamline regions of the flow and corresponding to the conventional profile drag  $D_0$ . Hence, the drag coefficient that occurs in the blockage parameter is not the total drag coefficient  $\mathcal{C}_{\rm D}$ , but its separated-flow component

$$C_{ijs} = C_{D} - C_{Dv} - C_{D0}$$
 (7.9)

There remains the problem of determining the drag coefficient  $C_{DS}$  from normal experimental observations of lift and drag. Since great accuracy is not usually necessary, Maskell suggests that the most logical procedure is to define  $C_{DS}$  by extrapolation from the measured properties of the unstalled wing. Linear extrapolation of the appropriate part of the measured  $C_0 \sim C_L^2$  relation, sketched in Figure 7.6, is usually sufficient. But visual observation of the flow development, allowing the caset of flow separation to be located, obviously helps to identify the point at which the bluff-body type of blockage effect begins to make its appearance. It will be noticed that the procedure ensures that the derived  $C_{DS}$  is zero for the unstalled wing, as it should be.

It is convenient, in practice, to combine the blockage correction corresponding to separated flow with that appropriate to streamline flow in the following manner,

$$\frac{q_{c}}{q} = 1 + \frac{1}{2} \frac{S}{c} (C_{DR} + C_{D0}) + \frac{5}{2} \frac{S}{c} (C_{D} - C_{DV} - C_{D0}) , \qquad (7.10)$$

where, for the sake of completeness, a drag coefficient  $C_{DR}$  associated with the support rig - and assumed here to correspond to streamline flow - has been included. The composite formula reduces automatically to the correct formula for streamline flow at incidences below that at which flow separation first occurs.

The effectiveness of the correction formula is indicated in Figure 7.7, which shows the results of applying it to data obtained with two models of an aircraft having a delta wing of aspect ratio 3.

For a stalled wing of infinite span there is no vortex-induced drag, and the drag coefficient appropriate to the blockage parameter is the total  $C_D$  if, as is usual, the total drag is predominantly due to the separated flow. The problem in this case is to assign a value to the blockage factor  $\theta$ , and no doubt the best procedure is to consume the pressure distribution over the upper surface of the aerofoil and to derive the eppropriate factor from Equation (7.5). However, from the limited experimental evidence available,  $\theta$  seems unlikely to differ greatly from unity once the stall is fully developed.

#### 7.3.2 Riags with Hinged or Split Flaps

Hinged or split flaps deflected near the trailing edges of wings also give rise to regions of separated flow similar to bluff-body wakes. It is reasonable to suppose, therefore, that the theory of Section 7.2 can again be used to calculate the blockage effect.

In this case the situation is much the same as that treated in Section 7.3.1, but with the important difference that some separated flow is present at all angles of incidence when the flaps are deflected. However, there is still a range of incidence over which the separated flow is restricted to the wakes of the flaps themselves, and in which the appropriate  $C_{\rm DV}$  can probably be defined with sufficient accuracy. Then, if  $C_{\rm D0}$  is determined from the clean wing (i.e. with flaps undeflected), Equation (7.10) can be used as before, with  $C_{\rm DS}$  (=  $C_{\rm D} - C_{\rm DV} - C_{\rm D0}$ ) non-zero at all angles of incidence.

The above procedure, though plausible, must be regarded as less well established than the application to stalled wings considered in Section 7.3.1, because it has been much less severely tested. In any event, for flap aspect ratios that are greater than 10, say, a blockage factor of 5/2 is most likely inappropriate. It would be reasonable to expect  $\theta$  to fall as flap aspect ratio increases, roughly according to the table in Section 7.2.

## 7.3.3 Slender Wings with Leading-Edge Vortices

At angles of incidence less than some critical value. flow separation from the leading edges of slender wings gives rise to a characteristic vortex flow that bears no resemblance whatever to the bluff-body wake. In fact this kind of "separated flow" is much better regarded as a generalized form of the classical streamline flow. It is essentially steady, with a thin wake, but one that originates from all the edges of the wing, and not only from the trailing edge. The rolling-up of the wake, which is taken to occur predominantly downstream of the trailing edge in the classical flow model to form the so-called tip vortices, therefore begins at the apex of a slender wing and forms leading-edge vortices. In consequence, the appropriate blockage correction is of the streamline-flow kind, viz.,

$$\frac{q_{c}}{q} = \frac{1}{3} C_{D0} \frac{S}{c} . \qquad (7.11)$$

where C<sub>ng</sub> is the profile drag coefficient.

However, at the critical angle of incidence, the leading-edge vortices are said to break down. Their structure changes from the orderly streamline form to cometning such more unsteady, and generally enclosing substantial regions of reversed flow which, once again, bear some resemblance to axisymmetric bluff-body wakes. At the critical incidence, the vortex breakdown occurs in the neighbourhood of the trailing edge of the wing and, as incidence is increased, the point of breakdown moves forward towards the apex. At angles of incidence below the critical, vortex breakdown may still occur, but not sufficiently close to the wing to affect its near field materially.

It seems reasonable to suppose that a blockage effect similar to that experienced by a bluff body will be felt by the slender wing at angles of incidence higher than the critical. Moreover, since the regions of separated flow in the wake of a partially stalled wing of moderate sweepback have a similar vortex structure to that which follows vortex breakdown on more slender wings, it also scenes reasonable to suppose that the same blockage factor  $\theta = 5/2$  will be appropriate in both cases.

The point at which the higher blockage factor becomes applicable is perhaps best identified from visual observations of the voržex flow. However, there is some

....

### 444

evidence to suggest that the procedure outlined in Section 7.3.1 may be satisfactory as most empirical  $C_D \sim C_L^2$  curves for slender wings appear to exhibit a linear region for a substantial range of incidence below the critical.

## 7.4 V/STOL CONFIGURATIONS

#### 7.4.1 Lifting Rotors

A wholly self-consistent mathematical model of a lifting rotor seems to follow from the apparently simple assumptions that the rotor itself can be represented as a uniformly loaded actuator disc, and that the effects of mixing between the slipstream and the external flow can be ignored. Thus a uniform rise in total head would be supposed to occur across both the rotor disc and the interface between the slipstream and the external flow. And the precise shape of this interface would be such as to make it a stream surface across which the static pressure varies continuously. However, such a model is not yet amenable to theoretical treatment, even for an unlimited mainstream, and further simplifying assumptions must be introduced.

In general, the interaction between slipstream and mainstream, implied in the boundary conditions at the interface, would be expected to result in a slipstream following a curved path and having cross-sections that become increasingly distorted downstream of the rotor disc. Of these two effects on the geometry of the slipstream, the former seems much the more important, since it implies a continuous transfer of downward momentum from the slipstream to the external flow downstream of the rotor. But theory has not yet advanced far enough to allow even this effect to be accounted for adequately, and the best that can be done at present is, following Heyson (Ref. 7.2: 1960), to replace the true path of the slipstream by an appropriate linear approximation.

Heydon represents the rotor slipstream by a skewed elliptic vortex cylinder of uniform strength  $d\bar{c}/dl$ , where  $\Gamma$  is the circulation associated with a length l along the skewed slipstream. The velocity field of the rotor in free air can then be calculated by integrating the Biot-Savart expression for the velocity field of a vortex element over the surface of the vortex cylinder. In particular, the upward induced velocity  $w_0$  at the centre of the rotor, and normal to its tip-path plane, is given by

$$\mathbf{F}_{0} = -\frac{1}{2} \frac{d\Gamma}{dl}.$$
 (7.12)

In general, the upwash w associated with the semi-infinite vortex cylinder can be written in the form

$$\mathbf{w} = \mathbf{w}_0 \frac{\mathbf{s}}{\mathbf{c}} \left\{ -\frac{2\mathbf{b}}{\pi \mathbf{h}} \, \mathbf{w} \left( \frac{2\mathbf{x}}{\mathbf{h}} \, \cdot \, \frac{2\mathbf{y}}{\mathbf{h}} \, \cdot \, \frac{2\mathbf{z}}{\mathbf{h}} \right) \right\} \, . \tag{7.13}$$

where 3 and h are respectively the breadth and height of the wind-tunnel test section, C is its cross-sectional area, and S is the area of the rotor disc.

An alternative expression for the singularities representing the slipstream follows from the fact that each elementary vortex ring is exactly equivalent to a uniform doublet sheet, circumscribed by the ring, of total strength  $\frac{1}{4}\Gamma R^2$ , where R is the radius of the circular element. In consequence, the far field of the rotor reduces to the field of a semi-infinite line of point doublets, directed vertically downwards if the rotor disc is in a horizontal plane. Equation (7.12) relates the strength of the line of doublets to the upwash at the centre of the rotor, and the function W in the expression (7.13) reduces, in the far field\*, to

T289Bbbbbb

$$\Psi_{1}(\xi,\eta,\zeta) = \frac{\xi^{2}+\eta^{2}}{\rho^{3}(\rho+\zeta\cos\chi-\xi\sin\chi)} - \left\{\frac{\zeta+\rho\cos\chi}{\rho(\rho+\zeta\cos\chi-\xi\sin\chi)}\right\}^{2}, \quad (7.14)$$

where  $\rho^2 = \xi^2 + \eta^2 + \zeta^2$  and  $\chi$  is the skew angle of the wake defined in Figure 7.8.

The upwash field thus derived is related strictly to Cartesian axes fixed in the rotor, such that the axis OZ is normal to the rotor disc with z measured positive along the thrust line. It gives the true upwash relative to wind axes only if the rotor disc is at zero incidence. However, if the elementary doublets representing the slipstream are supposed always to be directed normal to the rotor disc, they are readily resolved into components directed horizontally forwards and vertically downwards, in wind axes, and the associated component fields can be superposed to yield the complete upwash field of the rotor at incidence (Fig.7.9).

Heyson expresses the upwash field due to a line of horizontally-directed doublets of strength  $\frac{1}{4}\Gamma R^2$  in the form of Equation (7.13), as before, and the corresponding upwash function is

$$W_{2}(\xi,\eta,\zeta) = -\frac{\xi\zeta}{\rho^{3}(\rho+\zeta\cos\chi-\xi\sin\chi)} - \frac{(\xi-\rho\sin\chi)(\zeta+\rho\cos\chi)}{\rho^{2}(\rho+\zeta\cos\chi-\xi\sin\chi)^{2}}.$$
 (7.15)

The derivation of tunnel-wall corrections is now fairly straightforward, once the slipstream skew angle X is known. Following usual practice in helicopter theory, Heyson takes

$$\tan \chi = -\frac{U \cos \alpha}{U \sin \alpha \pm w_0}.$$
 (7.16)

an expression for which there is some experimental support, and ignores any possible effect of constraint on X in the immediate vicinity of the rotor. However, it is evident that, as the true slipstream approaches the floor of the wind tunnel its curvature increases rapidly. This effect is represented in Heyson's model by supposing that the slipstream follows its initial path until it strikes the tunnel floor, and then continues downstream coincident with the tunnel floor, as shown in Figure 7.9.

In consequence, there are two basic elements in the derivation of corrections. These are obtained by considering slipstreams composed of vertical and horizontal doublets, together with their first reflections in the tunnek floor, as illustrated in Figure 7.9. It will be noticed that the parts of the slipstreams coincident with the floor cancel

• Heyson<sup>7, 2</sup> gives also the general expression  $W(\xi,\eta,\zeta)$  for a thereed cylinder of radius R.

445

each other when the doublets are vertically directed, but reinforce each other in the second case. Their upwash fields are readily obtained by superposition of fields from Equations (7.14) and (7.15) respectively.

Heyson considers two wind-tunnel configurations, both having solid lower boundaries. He then takes the remaining boundaries to be either whelly solid or wholly free, and expresses the interference velocity at the centre of the test section in the form

$$\mathbf{w}_{i} = \delta_{\mathbf{w}} \frac{s}{c} \mathbf{w}_{c} . \qquad (7.17)$$

Some indication of its dependence on X,  $\alpha$  and b/h, for the two types of tunnel, is provided in Figures 7.10 to 7.13, where for consistency with the foregoing chapters

$$\delta_0 = -\frac{1}{4} \delta_{\mu} \tag{7.18}$$

is plotted. In particular, it should be noted that at high forward speeds, when slipstream skew angles are close to 90°, the corrections are the same as those for a wing, as they evidently should be: at very low speeds, i.e., suall skew angles, they are largely determined by the floor of the wind tunnel, and the interference upwash at the rotor is large.

Heyson points out that care is needed in the interpretation of the interference field. In hover, and at low forward speeds, the rotor performance is governed primarily by the change in inflow rather than by the change in effective incidence. He suggests, therefore, that a possible interpretation of the results is that the interference velocity corresponds to a change in rate of climb (or sink) between corresponding wind-tunnel and free-air conditions.

#### 7.4.2 Jet-flaps in Closed Tunnels

The main differences between a jet-flap and a conventional trailing-vortex wske are that the vortex representation of the jet sheet includes bound-vortex elements of strength proportional to the jet-momentum coefficient  $C_J$  and that there is local curvature of the sheet. The presence of these bound-vortex elements gives rise to a substantially larger circulation about the wing than would otherwise occur; in consequence, the effect of wind-tunnel constraint on their strength must be considered carefully. There will be additional curvature of the jet sheet due to constraint, leading to stronger bound vortices in the jet than would occur in an unlimited stream with the same jet-momentum coefficient  $C_J$  and jet-deflection angle  $\tau$ . It is essential, therefore, that this wall-induced curvature be taken into account.

Maskell (Ref.7.4; 1959) has considered the constraint problem for the jet-flap in a closed wind-tunnel, as an extension of the unlimited-stream theory given by Maskell and Spence<sup>7.3</sup>. The main problem is one of interpretation. The interference velocity field can be largely determined as for a three-dimensional wing, so that the interference upwash at the wing and at infinity downstleam are known. However, it is the variation of upwash between these limits that determines the curvature of the jet path, and hence the bound-vortex strength, and Maskell chooses to represent this variation by an interpolation that leads to a simple interpretation of the effect of

the induced curvature. He points out that, if the interference upwash  $w_i$  is related linearly to the total downwash -w in the manner shown in Figure 7.14, the induced curvature is equivalent to an increment  $\Delta C_J$  in jet-momentum coefficient. The particular choice of interpolation in  $w_i$  is, of course, open to question. But the conclusion that the effect can be interpreted as a correction to  $C_J$  seems saud, and the order of magnitude derived by Maskell is probably correct.

The first-order effects of constraint are obtained as a correction to incidence

$$\Delta \alpha = \delta_0 \frac{SC_L}{C} \left[ 1 + \frac{2C_J}{\pi A} \right]^{-1}.$$
 (7.19)

where  $\delta_0$  takes the classical value from Chapter III, and a correction to jet-momentum coefficient

$$\frac{\Delta c_{J}}{c_{J}} = \frac{\Delta \alpha}{\tau + \alpha + (w/U)_{\infty}}, \qquad (7.20)$$

where

$$\begin{pmatrix} \mathbf{w} \\ \mathbf{\tilde{U}} \\ \mathbf{w} \end{pmatrix}_{\infty} = -\frac{2\mathbf{C}_{\mathbf{L}}}{\pi\mathbf{A} + 2\mathbf{C}_{\mathbf{J}}}.$$
 (7.21)

The application of the corrections requires some care, since the lift and thrust forces measured in the wind tunnel are of the same order of magnitude. In consequence, when these forces are resolved normal and parallel to the effective stream direction, increments to lift and thrust coefficients result, of magnitude

$$\Delta C_{L} = C_{L} \Delta \alpha + \Delta C_{J} s^{i} n (\tau + \alpha)$$

$$\Delta C_{T} = -C_{L} \Delta \alpha + \Delta C_{J} cos (\tau + \alpha)$$
(7.22)

where  $\Delta \alpha$  and  $\Delta C_{J}$  are given in Equations (7.19) and (7.20). If a drag coefficient is defined according to

 $C_{\rm D} = C_{\rm J} - C_{\rm T}$ , (7.23)

then

2

 $\Delta C_{D} = \Delta C_{J} - \Delta C_{\overline{I}}$ 

and Equations (7.22) can be expressed in the alternative form

$$\frac{\Delta C_{L}}{\Delta \alpha} = -C_{D} + C_{J} \left( 1 + \frac{\sin (\tau + \alpha)}{\tau + \alpha + (w/U)_{\infty}} \right)$$

$$\frac{\Delta C_{D}}{\Delta \alpha} = C_{L} + C_{J} \left( \frac{1 - \cos (\tau + \alpha)}{\tau + \alpha + (w/U)_{\infty}} \right)$$
(7.24)

The corrections from Equations (7.19) to (7.22) for a typical wind-tunnel experiment<sup>7.13</sup> are plotted in Figure 7.15. For comparison, the effect of ignoring the correction to  $\Delta C_J$ , as in Glauert's<sup>7.8</sup> classical theory, is also shown. However, there is, as yet, no experimental confirmation that the magnitudes of the corrections are adequate in practice.

Reference should be made to Maskell's<sup>7.4</sup> original paper for a discussion of the effect of boundary constraint on the performance of the tailplane of a jet-flap aircraft.

## 7.4.3 Kore General Configurations

Nore general V/STOL configurations must be expected to require representation by mathematical models incorporating more than one wake or slipstream. If the overall performance of a configuration of this kind depends crucially on the mutual interference between the different lifting elements and their wakes or slipstreams, there seems little prospect of deriving adequate wind-tunnel corrections until the nature of the interference is better understood than it is at present. Much depends on the importance of bound vorticity of the kind included in a jet sheet and related to its curvature. But this problem has not yet been considered generally.

It may be that the jet-flap effect can be largely ignored in relation to jets and alipstreams of initially circular cross-section. Nevertheless, fair approximations to the true paths of the elements of the multiple wakes, and to their relative vortex or doublet strengths, would no doubt prove essential. Provided that Heyson's liftingrotor model is adequate for the single-rotor case, it can be extended, plausibly, to twin rotors with separate wakes. But the combination of, say, a fan with a conventional lifting wing appears to be too complex to handle in this manner.

Heyson (Ref. 7.5; 1962) has suggested that such general problems might be treated by a simple extension to his lifting-rotor theory, in which the multiple wakes are approximated by a single mean rotor wake, the parameters of which are determined by the measured lift and drag. He has given an extremely comprehensive theory based on this simple concept and tables of interference factors in  $\varepsilon$  series of reports<sup>7.5, 7.14-7.17</sup>, to which reference should be made for the detailed corrections. But the theory would seem to be most plausible when a single wake or slipstream predominates.

Several experimental studies have been mad2 of wind-tunnel interference on V/STOL configurations, notably by Grunwald<sup>7.16,7.19</sup> and by Davenport and Kuhn<sup>7.20</sup>. The broad conclusion from these particular investigations seemed to be that Heyson's general theory accounts for most of the interference effects on lift and drag, but may fail to correct pitching moments adequately. Further work is needed before these conclusions can be generalized to other configurations. Similar caution is advocated in an independent review by Templin<sup>7.21</sup>, who discusses the magnitude of wall interference by Heyson's theory in relation to acceptable sizes of model and wind tunnel.

# REFERENCES

ť

2.

1

| 7.1   | Maskell, E.C.                  | A Theory of the Blockage Effects on Bluff Bodies and<br>Stalled Wings in a Closed Wind Tunnel. ARC R & M 3400,<br>1963.  |
|-------|--------------------------------|--|
| 7.2   | Heyson, H.H.                   | Jet-Boundary Corrections for Lifting Rotors Centered in<br>Rectangular Wind Tunnels. NASA Tech. Report R-71, 1960.   |
| 7.3   | Maskell, E.C.,<br>Spence, D.A. | A Theory of the Jet Flap in Three Dimensions. Proc. Roy.<br>Soc., Series A, Vol.251, 1959, pp.407-425.   |
| 7.4   | Maskell, E.C.                  | The Interference on a Three-Dimensional Jet-Flap Wing in<br>a Closed Wind Tunnel. ARC R & M 3219, 1959.  |
| 7.5   | Heyson, H.H.                   | Linearized Theory of Wind-Tunnel Jet-Boundary Corrections<br>and Ground Effect for VTOL-STOL Aircraft. NASA Tech.<br>Report R-124, 1962.   |
| 7.6   | Fage, A.,<br>Johansen, F.C.    | On the Flow of Air 5ehind an Inclined Flat Plate of<br>Infinite Span. ARC R&W 1104, 1927. Proc. Roy. Soc.,<br>Series A, Vol.116, 1927, pp.170-197.   |
| 7.7   | Fage, A.,<br>Johansen, F.C.    | The Structure of Vortex Sheets. ARC R & M 1143, 1928.<br>Phil. Mag., Vol.5, 1928, pp.417-441.  |
| 7.8   | Glauert, H.                    | Wind Tunnel Interference on Wings, Bodies and Airscrews.<br>ARC R & M 1566, 1933.  |
| 7.9   | Fail, R.,<br>et al.            | Low-Speed Experiments on the Wake Characteristics of<br>Flat Plates Normal to an Air Stream. ARC R & M 3120, 1957.   |
| 7.10  | Lefebvre, A.H.                 | A Method of Predicting the Aerodynamic Blockage of Bluff<br>Bodies in a Ducted Airstream. Coll. Aero. (Cranfield)<br>Report 188. ARC Report 27,859, 1965.  |
| 7.11  | Calvert, J.B.                  | Blockage Corrections for Blunt-Based Bodies of Revolution.<br>J. Roy. meron. Soc., Vol. 70, 1966, pp. 532-533.   |
| 7. 12 | Kirby, D.A.,<br>Spence, A.     | Low-Speed-Tunnel Model Tests on the Flow Structure behind<br>a Delta-Wing Aircraft and a 40 deg Swept-Wing Aircraft at<br>High Incidences. ARC R & M 3078, 1955.   |
| 7.13  | Foster, D.N.                   | Private communication, 1966.   |
| 7.14  | Heyson, H.H.                   | Tables of Interference Factors for Use in Wind-Tunnel<br>and Ground-Effect Calculations for VTOL-STOL<br>Aircraft. Part I - Wind Tunnels Having Width-Height<br>Batio of 20 NASA Tech. Note D-933, 1961. |

449

1. 1. Co 4.

, A K

And the state of the second se

CHINESE TON, ARTS

1

2

۰.

450 Tables of Interference Factors for Use in Wind-Turnel 7.15 Heyson, H.H. and Ground-Effect Calculations for VTOL-STOL Aircraft. Part II - Wind Tunnels Having Width-Height Batio of 1.5. NASA Tech. Note D-934, 1961. 7.16 Heyson, H.H. Tables of Interference Factors for Use in Wind-Tunnel and Ground-Effect Calculations for VTOL-STOL Aircraft. Part III - Wind Tunnels Having Width-Height Ratio of 1.0. NASA Tech. Note D-935, 1961. 7.17 Heyson, H.H. Tables of Interference Factors for Use in Wind-Tunnel and Ground-Effect Calculations for VTOL-STOL Aircraft. Part IV - Wind Tunnels Having Width-Height Ratio of 0.5. NASA Tech. Note D-936, 1961. Wall Effects and-Scale Effects in V/STOL Model Testing. 7.18 Grunwald, K.J. Presented at Aerodynamic Testing Conference at Washington, March 1964. 7.19 Grunwald, K.J. Experimental Study of Wind-Tunnel Wall Effects and Wall Corrections for a General-Research V/STOL Tilt-Wing Model with Flap. NASA Tech. Note D-2887, 1965. Wind-Tunnel-Wall Effects and Scale Effects on a VTOL 7.20 Davenport, E.E., Configuration with a Fan Mounted in the Fuselage. Kuhn, R.E. NASA Tech. Note D-2560, 1965. The Choice of Working Section Size and Shape for V/STOL 7.21 Templia, R.J. Wind Tunnels. NRC (Canada) Quarterly Bulletin DME/NAE 1965(4), 1965.

## ADDITIONAL REFERENCES FOR CHAPTER VII

Butler, S.F.J., Williams, J.

Duquenne, R., Werlé, H.

Ganzer, V.M., Rae, W.H.

Heyson, H.H.

Further Comments on High-Lift Testing in Wind Tunnels with Particular Reference to Jet-Blowing Models. AGARD Report 304, 1959.

Effets de Paroi sur une Aile avec Soufflage. AGARD Report 305, 1959.

An Experimental Investigation of the Effect of Wind Tunnel Walls on the Aerodynamic Performance of a Helicopter Rotor. NASA Tech. Note D-415, 1960.

Wind-Tunnel Wall Interference and Ground Effect for VTOL-STOL Aircraft. J. 4. Helicopt. Soc., Vol.6, 1961, pp.1-9.

| de Jager, E.M.  | Two-Dimensional Tunnel-Wall Corrections for a Wing with<br>a Blown Flap between Two Parallel Walls. NLR<br>(Netherlands) Report W.8, 1961. |
|-----------------|--|
| de Jager, E.M., | Calculated Tunnel-Wall Corrections for Two-Dimensional   |
| van Spiegel, E. | High-Lift Wings. NLL (Netnerlands) Report MP.181, 1959.  |
| Kuhn, R.E.,     | Tunnel-Wall Effects Associated with VTOL-STOL Model  |
| Naeseth, R.L.   | Testing. AGARD Report 303, 1959.   |
| Wells, W.J.     | Wind Tunnel Corrections for a Helicopter Rotor in<br>Forward Flight. United Aircr. Corp. Report M-0554-4,<br>1955.                         |

Į, į

**.** 

2

1

1

-





ł

rig. 7.3 Blockage effect on drag and base pressure for con-lifting square plates





.

10

1450 774

HUN YAN

Fig. 7.5 Variation of blockage factor with fineness ratio



:5



ľ



Ì I

Fig.7.7 Blockage corrections applied to lift and drag for two sizes of delta-wing-body combination!

457

managerily. ha

4

STATISTICS.

A PAGE

5-6







K MILLEN

460

્રેફે



シンシンにいくたちなとうでも

į

461

energe in

The second se

5



1.1

100 A. C. C.

1.

The substitution of the second se

1.1

- Min

**...** :



ć

and the second second second

Fig. 7.13 Effect of angle of attack on the interference parameter for small lifting rotors in a closed square tunnel





۶,

÷

£.,

Fig. 7.15 Typical constraint corrections for a jet-flap wing with 50° jet deflection

465

にはないという